SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM S-1

REGISTRATION STATEMENT

Under

The Securities Act of 1933

COMSCORE, INC.

(Exact name of Registrant as specified in its charter)

Delaware(State or other jurisdiction of incorporation or organization)

7389 (Primary Standard Industrial Classification Code Number)

11465 Sunset Hills Road

Reston, Virginia 20190 (703) 438-2000

(Address, including zip code, and telephone number, including area code, of Registrant's principal executive offices)

Magid M. Abraham, Ph.D.
President and Chief Executive Officer
comScore, Inc.
11465 Sunset Hills Road
Suite 200
Reston, Virginia 20190
(703) 438-2000

(Name, address, including zip code, and telephone number, including area code, of agent for service)

Copies to:

Christiana L. Lin, Esq. General Counsel 11465 Sunset Hills Road, Suite 200 Reston, Virginia 20190 Telephone: (703) 438-2000 Facsimile: (703) 438-2051 Andrew J. Pitts, Esq. Cravath, Swaine & Moore LLP Worldwide Plaza 825 Eighth Avenue New York, New York 10019 Telephone: (212) 474-1000 Facsimile: (212) 474-3700

54-19555550

(I.R.S. Employer Identification Number)

Jeffrey D. Saper, Esq. Robert G. Day, Esq. Wilson Sonsini Goodrich & Rosati, Professional Corporation 650 Page Mill Road Palo Alto, California 94304

Mark R. Fitzgerald, Esq. Wilson Sonsini Goodrich & Rosati, Professional Corporation 1700 K Street, N.W., Fifth Floor Washington, D.C. 20006 Telephone: (202) 973-8800 Facsimile: (202) 973-8899

Approximate date of commencement of proposed sale to the public: As soon as practicable after the effective date of this Registration Statement.

If any of the securities being registered on this Form are to be offered on a delayed or continuous basis pursuant to Rule 415 under the Securities Act of 1933, check the following box. o

If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, please check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. o

If this Form is a post-effective amendment filed pursuant to Rule 462(c) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this Form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

CALCULATION OF REGISTRATION FEE

Title of Each Class of Securities to be Registered	Proposed Maximum Aggregate Offering Price(1)(2)	Amount of Registration Fee
Common Stock, \$0.001 par value per share	\$86,250,000	\$2,648

- (1) Estimated solely for the purpose of computing the amount of the registration fee pursuant to Rule 457(o) under the Securities Act of 1933, as amended.
- (2) Includes shares of common stock that may be purchased by the underwriters to cover over-allotments, if any.

The Registrant hereby amends this Registration Statement on such date or dates as may be necessary to delay its effective date until the Registrant shall file a further amendment which specifically states that this Registration Statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act of 1933 or until the Registration Statement shall become effective on such date as the Commission, acting pursuant to said Section 8(a), may determine.

The information in this prospectus is not complete and may be changed. We may not sell these securities until the registration statement filed with the Securities and Exchange Commission is effective. This prospectus is not an offer to sell these securities and it is not soliciting an offer to buy these securities in any jurisdiction where the offer or sale is not permitted.

SUBJECT TO COMPLETION, DATED MARCH 30, 2007 PRELIMINARY PROSPECTUS

Shares



Prior to this offering, there has been no public market for our common stock. The initial public offering price of the common stock is expected to be between \$ and \$ per share. We intend to apply to list our common stock on The NASDAQ Global Market under the symbol "SCOR."

We are selling shares of common stock and the selling stockholders are selling shares of common stock. We will not receive any of the proceeds from the shares of common stock sold by the selling stockholders.

The underwriters have an option to purchase a maximum of additional shares from us and the selling stockholders to cover over-allotments of shares. The underwriters can exercise this right at any time within 30 days from the date of this prospectus.

Investing in our common stock involves risks. See "Risk Factors" on page 8.

	Price to Public	Underwriting Discounts and Commissions	Proceeds to comScore	Proceeds to Selling Stockholders
Per Share	\$	\$	\$	\$
Total	\$	\$	\$	\$

Neither the Securities and Exchange Commission nor any state securities commission has approved or disapproved of these securities or determined if this prospectus is truthful or complete. Any representation to the contrary is a criminal offense.

Delivery of the shares of common stock will be made on or about , 2007.

Credit Suisse

Deutsche Bank Securities

William Blair & Company

Friedman Billings Ramsey

Jefferies & Company

The date of this prospectus is , 2007

TABLE OF CONTENTS

	rage
Prospectus Summary	1
RISK FACTORS	8
CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS AND INDUSTRY DATA	26
Use of Proceeds	27
DIVIDEND POLICY	27
CAPITALIZATION	28
DILUTION	30
SELECTED CONSOLIDATED FINANCIAL DATA	32
MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS	37
Business	54
Management	72
COMPENSATION DISCUSSION AND ANALYSIS	77
EXECUTIVE COMPENSATION	81
CERTAIN RELATIONSHIPS AND RELATED PARTY TRANSACTIONS	89
PRINCIPAL AND SELLING STOCKHOLDERS	90
DESCRIPTION OF CAPITAL STOCK	93
ELEARING OF CHILDRE SALE	99
U.S. FEDERAL TAX CONSEQUENCES TO NON-U.S. HOLDERS	101
UNDERWITING	104
CINICENNI INC. NOTICE TO CANADIAN RESIDENTS	108
NOTICE TO CANADIAN RESIDENTS LEGAL MATTERS LEGAL PARTERS	109
ERORI MATTERS Experts	109
EAPERIS WHERE YOU CAN FIND MORE INFORMATION	110
WHERE TOO CAN FIRD MORE INFORMATION INDEX TO CONSCIDENCE PIRANCIAL STATEMENTS	F-1
INDEX TO CONSOLIDATED FINANCIAL STATEMENTS EX-3.2	F-1
EX-4.2	
EX-4.3	
EX-4.4	
<u>EX-4.5</u>	
<u>EX-4.6</u>	
EX-4.7	
EX-4.8	
<u>EX-4.9</u>	
<u>EX-4.10</u>	
<u>EX-4.11</u>	
<u>EX-4.12</u>	
<u>EX-4.13</u>	
<u>EX-4.14</u>	
<u>EX-4.15</u>	
<u>EX-4.16</u>	
<u>EX-10.1</u>	
<u>EX-10.2</u>	
<u>EX-10.3</u>	
<u>EX-10.4</u>	
<u>EX-10.5</u>	
<u>EX-10.6</u>	
<u>EX-10.7</u>	
EX-10.8	
<u>EX-10.9</u>	
EX-10.10	
<u>EX-10.11</u>	
EX-10.12	
<u>EX-10.13</u>	
EX-10.14	
<u>EX-21.1</u>	
<u>EX-23.1</u>	

You should rely only on the information contained in this document or to which we have referred you. We have not authorized anyone to provide you with information that is different. This document may only be used where it is legal to sell these securities. The information in this document may only be accurate on the date of this document.

Dealer Prospectus Delivery Obligation

Until , 2007 (25 days after the commencement of this offering) all dealers that effect transactions in these securities, whether or not participating in this offering, may be required to deliver a prospectus. This is in addition to the dealers' obligation to deliver a prospectus when acting as underwriters and with respect to their unsold allotments or subscriptions.

[&]quot;comScore", "Media Metrix" and "MyMetrix" are registered trademarks in the U.S. and several other countries. Our unregistered trademarks and service marks include: "Ad Metrix", "Campaign R/F", "Campaign Metrix", "comScore Marketing Solutions", "Marketing Solutions", "Plan Metrix", "qSearch", "Video Metrix" and "World Metrix".

PROSPECTUS SUMMARY

This summary highlights information contained elsewhere in this prospectus. This summary does not contain all of the information you should consider before buying shares in this offering. Therefore, you should read this entire prospectus carefully, including the "Risk Factors" section beginning on page 8 and our consolidated financial statements and the related notes. Unless the context requires otherwise, the words "we," "us," "our" and "comScore" refer to comScore, Inc. and its consolidated subsidiaries.

comScore, Inc

We provide a leading digital marketing intelligence platform that helps our customers make better-informed business decisions and implement more effective digital business strategies. Our products and solutions offer our customers deep insights into consumer behavior, including objective, detailed information regarding usage of their online properties and those of their competitors, coupled with information on consumer demographic characteristics, attitudes, lifestyles and offline behavior.

Our digital marketing intelligence platform is comprised of proprietary databases and a computational infrastructure that measures, analyzes and reports on digital activity. The foundation of our platform is data collected from our comScore panel of more than two million Internet users worldwide who have granted us explicit permission to confidentially measure their Internet usage patterns, online and certain offline buying behavior and other activities. By applying advanced statistical methodologies to our panel data, we project consumers' online behavior for the total online population and a wide variety of user categories.

We deliver our digital marketing intelligence through our comScore Media Metrix product family and through comScore Marketing Solutions. Media Metrix delivers digital media intelligence by providing an independent, third-party measurement of the size, behavior and characteristics of Web site and online advertising network audiences among home, work and university Internet users as well as insights into the effectiveness of online advertising. Our Marketing Solutions products combine the proprietary information gathered from the comScore panel with the vertical industry expertise of comScore analysts to deliver digital marketing intelligence, including the measurement of online advertising effectiveness, customized for specific industries. We typically deliver our Media Metrix products electronically in the form of weekly, monthly or quarterly reports. Customers can access current and historical Media Metrix data and analyze these data anytime online. Our Marketing Solutions products are typically delivered on a monthly, quarterly or ad hoc basis through electronic reports and analyses.

In 2006, we generated revenues of \$66.3 million and had cash flow from operations of \$10.9 million. We derive our revenues primarily from the fees that we charge for subscription-based products and customized projects. A significant characteristic of our business model is our large percentage of subscription-based contracts. Subscription-based revenues have grown to 75% of our total revenues in 2006. See "Management's Discussion and Analysis of Financial Condition and Results of Operations" contained in this prospectus for a discussion of how we determine subscription-based revenues.

Our Industry

The Internet is a global digital medium for commerce, content, advertising and communications. According to IDC, the number of global Internet users is projected to grow from approximately 968 million in 2005 to over 1.7 billion in 2010. As the online population continues to grow, the Internet is increasingly becoming a tool for research and commerce and for distributing and consuming media.

The interactive nature of digital media on the Internet enables businesses to access a wealth of user information that was virtually unavailable through offline audience measurement and marketing intelligence techniques. Digital media provide businesses with the opportunity to measure detailed user activity, such as how users interact with Web page content; to assess how users respond to online marketing, such as which online ads users click on to pursue a transaction; and to analyze how audiences and user behavior compare

across various Web sites. This type of detailed user data can be combined with demographic, attitudinal and transactional information to develop a deeper understanding of user behavior,

We believe that the growth in the online and digital media markets for digital commerce, content, advertising and communications creates an unprecedented opportunity for businesses to acquire a deeper understanding of both their customers and their competitive market position. Businesses can use accurate, relevant and objective digital marketing intelligence to develop and validate key strategies and improve performance.

The comScore Digital Marketing Intelligence Platform

We provide a leading digital marketing intelligence platform that enables our customers to devise and implement more effective digital business strategies.

Key attributes of our platform include

Panel of global Internet users. Our ability to provide digital marketing intelligence is based on information continuously gathered from a broad cross-section of more than two million Internet users worldwide who have granted us explicit permission to confidentially measure their Internet usage patterns, online and certain offline buying behavior and other activities.

Scalable technology infrastructure. We developed our databases and computational infrastructure to support the growth in online activity among our global Internet panel and the increasing complexity of digital content formats, advertising channels and communication applications. The design of our technology infrastructure is based on distributed processing and data capture environments that allow for the collection and organization of vast amounts of data on online activity.

Benefits of our platform include:

Advanced digital marketing intelligence. We use our proprietary technology to compile vast amounts of data on Internet user activity and to organize that data into discrete, measurable elements that can be used to provide actionable insights to our customers.

Objective third-party resource for digital marketing intelligence. We are an independent company that is not affiliated with the digital businesses we measure and analyze, allowing us to serve as an objective third-party provider of digital marketing intelligence.

Vertical industry expertise. We have developed expertise across a variety of industries to provide digital marketing intelligence specifically tailored to the needs of our customers operating in specific industry sectors. We have dedicated personnel to address the automotive, consumer packaged goods, entertainment, financial services, media, pharmaceutical, retail, technology, telecommunications and travel industries.

Ease of use and functionality. The comScore digital marketing intelligence platform is designed to be easy to use by our customers. Our products are primarily available through the Internet using a standard browser; our customers do not need to install additional hardware or software to access our products.

Our Strategy

Our objective is to be the leading provider of global digital marketing intelligence products. We plan to pursue our objective through internal initiatives and, potentially, through acquisitions and other investments. The principal elements of our strategy are to:

- · deepen relationships with current customers;
- · grow our customer base;
- · expand our digital marketing intelligence platform;
- · address emerging digital media;

- · extend technology leadership;
- build brand awareness through media exposure; and
- · grow internationally.

Risks Related to Our Business

Our business is subject to a number of risks that you should be aware of before making an investment decision. These risks are discussed more fully in the section entitled "Risk Factors" immediately following this prospectus summary. We have a limited operating history, and we must continue to retain and attract customers. We must be able to maintain an Internet user panel of sufficient size in order to provide the quality of marketing intelligence demanded by our customers. Although we were profitable in each quarter of 2006, we were not profitable in 2005, and we had, at December 31, 2006, an accumulated deficit of \$99.5 million.

Company Information

We incorporated in August 1999 in Delaware. Our principal offices are located at 11465 Sunset Hills Road, Suite 200, Reston, Virginia 20190. Our telephone number is (703) 438-2000. You can access our Web site at www.comscore.com. Information contained on our Web site is not part of this prospectus and is not incorporated in this prospectus by reference.

comScore, Media Metrix and MyMetrix are registered trademarks in the U.S. and several other countries. Our unregistered trademarks and service marks include: Ad Metrix, Campaign R/F, Campaign Metrix, comScore Marketing Solutions, Marketing Solutions, Plan Metrix, qSearch, Video Metrix and World Metrix.

Campaign R/F, Campaign Metrix, comScore Marketing Solutions, Marketing Solutions, Plan Metrix, qSearch, Video Metrix and World Metrix.

The Offering

Common stock offered by us shares
Common stock offered by the selling stockholders shares
Total common stock offered shares
Common stock outstanding after this offering shares

Use of proceeds We intend to use the net proceeds from this offering for working capital, for capital expenditures and for other general corporate purposes. We may also use a portion of our net proceeds to fund potential acquisitions. We will not receive

any proceeds from the sale of shares of our common stock by the selling stockholders. See "Use of Proceeds."

Proposed NASDAQ Global Market symbol SCOF

Risk factors See "Risk Factors" and other information included in this prospectus for a discussion of factors you should carefully

consider before deciding to invest in shares of our common stock.

The number of shares of common stock that will be outstanding after this offering is based on 108,025,682 shares outstanding as of December 31, 2006 and excludes:

- 13,619,700 shares of common stock issuable upon exercise of options outstanding at a weighted-average exercise price of \$0.40 per share;
- 5,316,147 shares of common stock reserved for future issuance under our 1999 Stock Plan;
- 7,000,000 shares of common stock reserved for future issuance under our 2007 Equity Incentive Plan, which will be effective upon completion of this offering;
- 100,000 shares of common stock issuable upon the exercise of a warrant, which warrant shall terminate if not exercised prior to this offering, at an exercise price of \$1.00 per share: and
- 775,923 shares of common stock issuable upon the exercise of warrants, which total includes warrants for our preferred stock that will become exercisable for common stock after this offering, at a weighted-average exercise price of \$0.96 per share.

Unless otherwise indicated, all information in this prospectus assumes:

- a -for- reverse split of our common stock that will occur prior to the consummation of this offering;
- the conversion, in accordance with our certificate of incorporation, of all our shares of outstanding preferred stock into shares of our common stock;
- no exercise by the underwriters of their option to purchase up to additional shares to cover over-allotments, consisting of shares to be purchased from us and shares to be purchased from the selling stockholders; and
- the adoption of our amended and restated certificate of incorporation and bylaws that will occur immediately prior to the consummation of this offering.

Summary Historical Financial Data

You should read the summary historical financial data set forth below in conjunction with our consolidated financial statements, the notes to our consolidated financial statements and "Management's Discussion and Analysis of Financial Condition and Results of Operations" contained elsewhere in this prospectus. The consolidated statements of operations data and the consolidated statements of cash flows data for each of the three years ended December 31, 2004, 2005 and 2006 as well as the consolidated balance sheet data as of December 31, 2005 and 2006 are derived from our audited consolidated financial statements that are included elsewhere in this prospectus. Our historical results are not necessarily indicative of results to be expected for future periods.

	Year Ended December 31,					
		2004 2005 (In thousands)		2005 thousands)		2006
Consolidated Statement of Operations Data:			•	,		
Revenues	\$	34,894	\$	50,267	\$	66,293
Cost of revenues(1)		13,153		18,218		20,560
Selling and marketing(1)		13,890		18,953		21,473
Research and development(1)		5,493		7,416		9,009
General and administrative(1)		4,982		7,089		8,293
Amortization		356		2,437		1,371
Total expenses from operations		37,874		54,113		60,706
(Loss) income from operations		(2,980)		(3,846)		5,587
Interest (expense) income, net		(246)		(208)		231
(Loss) gain from foreign currency		_		(96)		125
Revaluation of preferred stock warrant liabilities		_		(14)		(224)
(Loss) income before income taxes and cumulative effect of change in accounting principle		(3,226)		(4,164)		5,719
(Benefit) provision for income taxes				(182)		50
Net (loss) income before cumulative effect of change in accounting principle		(3,226)		(3,982)		5,669
Cumulative effect of change in accounting principle		_		(440)		_
Net (loss) income		(3,226)		(4,422)		5,669
Accretion of redeemable preferred stock		(2,141)		(2,638)		(3,179)
Net (loss) income attributable to common stockholders	\$	(5,367)	\$	(7,060)	\$	2,490

(1) Amortization of stock-based compensation is included in the line items above as follows:

		Year Ended December 31,						
	2004			ousands)		2006		
Cost of revenues	\$	_	\$	_	\$	12		
Selling and marketing		_		_		82		
Research and development		_		_		13		
General and administrative		14		3		91		
Total	\$	14	\$	3	\$	198		

The following table presents consolidated balance sheet data as of December 31, 2006:

- on an actual basis without any adjustments to reflect subsequent or anticipated events;
- on a pro forma basis reflecting (i) the conversion of all outstanding shares of our Series A, Series B, Series C, Series C-1, Series D and Series E preferred stock into an aggregate of 86,286,697 shares of our common stock effective immediately prior to the completion of this offering, for a total of 108,025,682 shares of common stock, which amount includes 1,738,172 shares subject to put and (ii) the reclassification of our preferred stock warrant liabilities from current liabilities to additional paid in capital effective upon the completion of this offering; and
- on a pro forma as adjusted basis reflecting the conversion and reclassification described above and the receipt by us of the net proceeds from the sale of shares of common stock in this offering at an assumed initial public offering price of \$ per share, after deducting the estimated underwriting discounts and commissions and estimated offering expenses payable by us.

		As of December 31, 2006				
	_	Actual	Ctual Pro Forma (In thousands)		Pro Forma as Adjusted	
Consolidated Balance Sheet Data:						
Cash, cash equivalents and short-term investments	\$	16,032	\$	16,032		
Total current assets		31,493		31,493		
Total assets		42,087		42,087		
Total current liabilities		32,880		31,875		
Capital lease obligations, long-term		2,261		2,261		
Common stock subject to put		4,357		4,357		
Redeemable preferred stock		101,695		_		
Stockholders' equity (deficit)		(99,557)		3,143		

	=	2004 2005 (In thousands)			2006	
Consolidated Statement of Cash Flows Data:			(111	tilousulitusj		
Net cash provided by operating activities	\$	1,907	\$	4,253	\$	10,905
Depreciation and amortization		2,745		5,123		4,259
Capital expenditures		1,208		1,071		2,314

	2004	r Ended Decembe 2005 (In thousands)	2006
Other Financial and Operating Data (unaudited): Adjusted EBITDA(2)	\$ (221)	\$ 730	\$ 9,945

(2) We define Adjusted EBITDA as net income plus the (benefit) provision for income taxes, depreciation, amortization of purchased intangible assets and stock-based compensation; plus interest expense (income) and other income. Adjusted EBITDA is not a measure of liquidity calculated in accordance with GAAP, and should be viewed as a supplement to—not a substitute for — our results of operations presented on the basis of GAAP. Adjusted EBITDA does not purport to represent cash flow provided by, or used in, operating activities as defined by GAAP. Our statement of cash flows presents our cash flow activity in accordance with GAAP. Furthermore, Adjusted EBITDA is not necessarily comparable to similarly-titled measures reported by other companies.

We believe Adjusted EBITDA is useful to an investor in evaluating our operating performance for the following reasons:

- Adjusted EBITDA is widely used by investors to measure a company's operating performance without regard to items such as interest expense, taxes, depreciation and
 amortization, and stock-based compensation, which can vary substantially from company to company depending upon accounting methods and book value of assets, capital
 structure and the method by which assets were acquired; and
- · analysts and investors use Adjusted EBITDA as a supplemental measure to evaluate the overall operating performance of companies in our industry.

Our management uses Adjusted EBITDA:

- as a measure of operating performance, because it does not include the impact of items not directly resulting from our core operations;
- for planning purposes, including the preparation of our annual operating budget;
- · to allocate resources;
- to evaluate the effectiveness of our business strategies; and
- $\bullet \quad \text{in communications with our board of directors and stockholders concerning our financial performance}.$

A reconciliation of Adjusted EBITDA to net income, the most directly comparable GAAP measure, for each of the fiscal periods indicated is as follows:

		Ye	ar Ended	l December 31			
	=	2004		2005 ousands)		2006	
Net (loss) income	\$	(3,226)	\$	(4,422)	\$	5,669	
(Benefit) provision for income taxes		_		(182)		50	
Amortization		356		2,437		1,371	
Depreciation		2,389		2,686		2,888	
Stock-based compensation		14		3		198	
Interest expense (income), net		246		208		(231)	
Adjusted EBITDA	\$	(221)	\$	730	\$	9,945	

RISK FACTORS

An investment in our common stock offered by this prospectus involves a substantial risk of loss. You should carefully consider these risk factors, together with all of the other information included in this prospectus, before you decide to purchase shares of our common stock. The occurrence of any of the following risks could materially adversely affect our business, financial condition or operating results. In that case, the trading price of our common stock could decline, and you may lose part or all of your investment.

Risks Related to Our Business and Our Technologies

If we are not able to maintain a panel of sufficient size and scope, or if the costs of maintaining our panel materially increase, our business would be harmed.

We believe that the quality, size and scope of our Internet user panel are critical to our business. There can be no assurance, however, that we will be able to maintain a panel of sufficient size and scope to provide the quality of marketing intelligence that our customers demand from our products. If we fail to maintain a panel of sufficient size and scope, customers might decline to purchase our products or renew their subscriptions, our reputation could be damaged and our business could be materially and adversely affected. We expect that our panel costs may increase and may comprise a greater portion of our cost of revenues in the future. The costs associated with maintaining and improving the quality, size and scope of our panel are dependent on many factors, many of which are beyond our control, including the participation rate of potential panel members, the turnover among existing panel members and requirements for active participation of panel members, such as completing survey questionnaires. Concerns over the potential unauthorized disclosure of personal information or the classification of our software as "spyware" or "adware" may cause existing panel members to uninstall our software or may discourage potential panel members from installing our software. To the extent we experience greater turnover, or churn, in our panel than we have historically experienced, these costs would increase more rapidly. In addition, publishing content on the Internet and purchasing advertising space on Web sites may become more expensive or restrictive in the future, which could decrease the availability and increase the cost of advertising the incentives we offer to panel members. To the extent that such additional expenses are not accompanied by increased revenues, our operating margins would be reduced and our financial results would be adversely affected.

Our quarterly results of operations may fluctuate in the future. As a result, we may fail to meet or exceed the expectations of securities analysts or investors, which could cause our stock price to decline.

Our quarterly results of operations may fluctuate as a result of a variety of factors, many of which are outside of our control. If our quarterly revenues or results of operations do not meet or exceed the expectations of securities analysts or investors, the price of our common stock could decline substantially. In addition to the other risk factors set forth in this "Risk Factors" section, factors that may cause fluctuations in our quarterly revenues or results of operations include:

- · our ability to increase sales to existing customers and attract new customers;
- · our failure to accurately estimate or control costs;
- · our revenue recognition policies related to the timing of contract renewals, delivery of products and duration of contracts and the corresponding timing of revenue recognition;
- · the mix of subscription-based versus project-based revenues;
- the impact on our contract renewal rates, in particular for our subscription-based products, caused by our customers' budgetary constraints, competition, customer dissatisfaction or our customers' actual or perceived lack of need for our products;
- the potential loss of significant customers;
- the effect of revenues generated from significant one-time projects;
- · the amount and timing of capital expenditures and operating costs related to the maintenance and expansion of our operations and infrastructure;
- the timing and success of new product introductions by us or our competitors;

- variations in the demand for our products and the implementation cycles of our products by our customers;
- · changes in our pricing and discounting policies or those of our competitors;
- · service outages, other technical difficulties or security breaches;
- · limitations relating to the capacity of our networks, systems and processes;
- maintaining appropriate staffing levels and capabilities relative to projected growth;
- · adverse judgments or settlements in legal disputes;
- the timing of costs related to the development or acquisition of technologies, services or businesses to support our existing customer base and potential growth opportunities; and
- · general economic, industry and market conditions and those conditions specific to Internet usage and online businesses.

We believe that our quarterly revenues and results of operations on a year-over-year and sequential quarter-over-quarter basis may vary significantly in the future and that period-to-period comparisons of our operating results may not be meaningful. You should not rely on the results of prior quarters as an indication of future performance.

The market for digital marketing intelligence is at an early stage of development, and if it does not develop, or develops more slowly than expected, our business will be harmed.

The market for digital marketing intelligence products is at a relatively early stage of development, and it is uncertain whether these products will achieve high levels of demand and increased market acceptance. Our success will depend to a substantial extent on the willingness of companies to increase their use of such products. Factors that may affect market acceptance include:

- the reliability of digital marketing intelligence products;
- · public concern regarding privacy and data security;
- decisions of our customers and potential customers to develop digital marketing intelligence capabilities internally rather than purchasing such products from third-party suppliers like us;
- decisions by industry associations in the United States or in other countries that result in association-directed awards, on behalf of their members, of digital measurement contracts
 to one or a limited number of competitive vendors;
- · the ability to maintain high levels of customer satisfaction; and
- · the rate of growth in eCommerce, online advertising and digital media.

The market for our products may not develop further, or may develop more slowly than we expect, either of which could adversely affect our business and operating results.

We have a limited operating history and may not be able to achieve financial or operational success.

We were incorporated in 1999 and introduced our first syndicated Internet audience measurement product in 2000. Many of our other products were first introduced during the past few years. Accordingly, we are still in the early stages of development and have only a limited operating history upon which our business can be evaluated. You should evaluate our likelihood of financial and operational success in light of the risks, uncertainties, expenses, delays and difficulties associated with an early-stage business in an evolving market, some of which may be beyond our control, including:

- our ability to successfully manage any growth we may achieve in the future;
- · the risks associated with operating a business in international markets, including China; and
- our ability to successfully integrate acquired businesses, technologies or services.

We have a history of significant net losses, may incur significant net losses in the future and may not maintain profitability.

We have incurred significant losses in recent periods, including net losses of \$3.2 million and \$4.4 million in 2004 and 2005, respectively. Although we achieved net income of \$5.7 million in 2006, we cannot assure you that we will continue to sustain or increase profitability in the future. As of December 31, 2006, we had an accumulated deficit of \$99.5 million. Because a large portion of our costs are fixed, we may not be able to reduce or maintain our expenses in response to any decrease in our revenues, which would adversely affect our operating results. In addition, we expect operating expenses to increase as we implement certain growth initiatives, which include, among other things, the development of new products, expansion of our infrastructure, plans for international expansion and general and administrative expenses associated with being a public company. If our revenues do not increase to offset these expected increases in costs and operating expenses, our operating results would be materially and adversely affected. You should not consider our revenue growth in recent periods as indicative of our future performance, as our operating results for future periods are subject to numerous uncertainties.

Material defects or errors in our data collection and analysis systems could damage our reputation, result in significant costs to us and impair our ability to sell our products.

Our data collection and analysis systems are complex and may contain material defects or errors. In addition, the large amount of data that we collect may cause errors in our data collection and analysis systems. Any defect in our panelist data collection software, network systems, statistical projections or other methodologies could result in:

- · loss of customers;
- · damage to our brand;
- lost or delayed market acceptance and sales of our products;
- · interruptions in the availability of our products;
- the incurrence of substantial costs to correct any material defect or error;
- · sales credits, refunds or liability to our customers;
- · diversion of development resources; and
- · increased warranty and insurance costs.

Any material defect or error in our data collection systems could adversely affect our reputation and operating results.

Our business may be harmed if we deliver, or are perceived to deliver, inaccurate information to our customers or to the media.

If the information that we provide to our customers or the media is inaccurate, or perceived to be inaccurate, our brand may be harmed. The information that we collect or that is included in our databases and the statistical projections that we provide to our customers may contain inaccuracies. Any dissatisfaction by our customers or the media with our digital marketing intelligence, measurement or data collection and statistical projection methodologies could have an adverse effect on our ability to retain existing customers and attract new customers and could harm our brand. Additionally, we could be contractually required to pay damages, which could be substantial, to certain of our customers if the information we provide to them is found to be inaccurate. Any liability that we incur or any harm to our brand that we suffer because of actual or perceived irregularities or inaccuracies in the data we deliver to our customers could harm our business.

Our business may be harmed if we change our methodologies or the scope of information we collect.

We have in the past and may in the future change our methodologies or the scope of information we collect. Such changes may result from identified deficiencies in current methodologies, development of more advanced methodologies, changes in our business plans or expressed or perceived needs of our customers or potential customers. Any such changes or perceived changes, or our inability to accurately or adequately communicate to our customers and the media such changes and the potential implications of such changes on the data we have published or will publish in the future, may result in customer dissatisfaction, particularly if certain information is no longer collected or information collected in future periods is not comparable with information collected in prior periods. For example, in 2002, we integrated our existing methodologies with those of Jupiter Media Metrix, which we had recently acquired. As part of this process, we discontinued reporting certain metrics. Some customers were dissatisfied and either terminated their subscriptions or failed to renew their subscriptions because of these changes. Future changes to our methodologies or the information we collect may cause similar customer dissatisfaction and result in loss of customers.

We may lose customers or be liable to certain customers if we provide poor service or if our products do not comply with our customer agreements.

Errors in our systems resulting from the large amount of data that we collect, store and manage could cause the information that we collect to be incomplete or to contain inaccuracies that our customers regard as significant. The failure or inability of our systems, networks and processes to adequately handle the data in a high quality and consistent manner could result in the loss of customers. In addition, we may be liable to certain of our customers for damages they may incur resulting from these events, such as loss of business, loss of future revenues, breach of contract or loss of goodwill to their business.

Our insurance policies may not cover any claim against us for loss of data, inaccuracies in data or other indirect or consequential damages and defending a lawsuit, regardless of its merit, could be costly and divert management's attention. Adequate insurance coverage may not be available in the future on acceptable terms, or at all. Any such developments could adversely affect our business and results of operations.

The market for digital marketing intelligence is highly competitive, and if we cannot compete effectively, our revenues will decline and our business will be harmed.

The market for digital marketing intelligence is highly competitive and is evolving rapidly. We compete primarily with providers of digital media intelligence and related analytical products and services. We also compete with providers of marketing services and solutions, with full-service survey providers and with internal solutions developed by customers and potential customers. Our principal competitors include:

- · large and small companies that provide data and analysis of consumers' online behavior, including Compete Inc., Hitwise Pty. Ltd and NetRatings, Inc.;
- · online advertising companies that provide measurement of online ad effectiveness, including aQuantive, Inc., DoubleClick Inc., ValueClick, Inc. and WPP Group plc;
- companies that provide audience ratings for TV, radio and other media that have extended or may extend their current services, particularly in certain international markets, to the measurement of digital media, including Arbitron Inc., Nielsen Media Research, Inc. and Taylor Nelson Sofres plc;
- analytical services companies that provide customers with detailed information of behavior on their own Web sites, including Omniture, Inc., WebSideStory, Inc. and WebTrends Corporation:
- full-service market research firms and survey providers that may measure online behavior and attitudes, including Harris Interactive Inc., Ipsos Group, Taylor Nelson Sofres plc and The Nielsen Company; and
- specialty information providers for certain industries that we serve, including IMS Health Incorporated (healthcare) and Telephia, Inc. (telecommunications).

Some of our current competitors have longer operating histories, access to larger customer bases and substantially greater resources than we do. As a result, these competitors may be able to devote greater resources to marketing and promotional campaigns, panel retention, panel development or development of systems and technologies than we can. In addition, some of our competitors may adopt more aggressive pricing policies. Furthermore, large software companies, Internet portals and database management companies may enter our market or enhance their current offerings, either by developing competing services or by acquiring our competitors, and could leverage their significant resources and pre-existing relationships with our current and potential customers.

If we are unable to compete successfully against our current and future competitors, we may not be able to retain and acquire customers, and we may consequently experience a decline in revenues, reduced operating margins, loss of market share and diminished value from our products.

Concern over spyware and privacy, including any violations of privacy laws or perceived misuse of personal information, could cause public relations problems and could impair our ability to recruit panelists or maintain a panel of sufficient size and scope, which in turn could adversely affect our ability to provide our products.

Any perception of our practices as an invasion of privacy, whether legal or illegal, may subject us to public criticism. Existing and future privacy laws and increasing sensitivity of consumers to unauthorized disclosures and use of personal information may create negative public reaction related to our business practices. Public concern has increased recently regarding certain kinds of downloadable software known as "spyware" and "adware." These concerns might cause users to refrain from downloading software from the Internet, including our proprietary technology, which could make it difficult to recruit additional panelists or maintain a panel of sufficient size and scope to provide meaningful marketing intelligence. In response to spyware and adware concerns, numerous programs are available, many of which are available for free, that claim to identify and remove spyware and adware from users' computers. Some of these anti-spyware programs have in the past identified, and may in the future identify, our software as spyware or as a potential spyware application. We actively seek to prevent the inclusion of our software on lists of spyware applications or potential spyware applications, to apply best industry practices for obtaining appropriate consent from panelists and protecting the privacy and confidentiality of our panelist data and to comply with existing privacy laws. However, to the extent that we are not successful, or to the extent that new anti-spyware programs classify our software as spyware or as a potential spyware application, our brand may be harmed and users of these programs may uninstall our software. Any resulting reputational harm or decrease in the size or scope of our panel could reduce the demand for our products, increase the cost of recruiting panelists and adversely affect our ability to provide our products to our customers. Any of these effects could harm our business.

Any unauthorized disclosure or theft of private information we gather could harm our business.

Unauthorized disclosure of personally identifiable information regarding Web site visitors, whether through breach of our secure network by an unauthorized party, employee theft or misuse, or otherwise, could harm our business. If there were an inadvertent disclosure of personally identifiable information, or if a third party were to gain unauthorized access to the personally identifiable information we possess, our operations could be seriously disrupted and we could be subject to claims or litigation arising from damages suffered by panel members or pursuant to the agreements with our customers. In addition, we could incur significant costs in complying with the multitude of state, federal and foreign laws regarding the unauthorized disclosure of personal information. For example, California law requires companies that maintain data on California residents to inform individuals of any security breaches that result in their personal information being stolen. Finally, any perceived or actual unauthorized disclosure of the information we collect could harm our reputation, substantially impair our ability to attract and retain panelists and have an adverse impact on our business.

We may encounter difficulties managing our growth, which could adversely affect our results of operations.

We have experienced significant growth in recent periods. We have substantially expanded our overall business, customer base, headcount, data collection and processing infrastructure and operating procedures as our business has grown. We increased our total number of full time employees from 176 employees as of December 31, 2003 to 377 employees as of December 31, 2006, and we expect to continue to expand our workforce to meet our strategic objectives. In addition, during this same period, we made substantial investments in our network infrastructure operations as a result of our growth. We believe that we will need to continue to effectively manage and expand our organization, operations and facilities in order to accommodate our expected future growth. If we continue to grow, our current systems and facilities may not be adequate. Our need to effectively manage our operations and growth requires that we continue to assess and improve our operational, financial and management controls, reporting systems and procedures. If we are not able to efficiently and effectively manage our growth, our business may be impaired.

If the Internet advertising and eCommerce markets develop slower than we expect, our business will suffer.

Our future success will depend on continued growth in the use of the Internet as an advertising medium, a continued increase in eCommerce spending and the proliferation of the Internet as a platform for a wide variety of consumer activities. These markets are evolving rapidly, and it is not certain that their current growth trends will continue.

The adoption of Internet advertising, particularly by advertisers that have historically relied on traditional offline media, requires the acceptance of new approaches to conducting business. Advertisers may perceive Internet advertising to be less effective than traditional advertising for marketing their products. They may also be unwilling to pay premium rates for online advertising that is targeted at specific segments of users based on their demographic profile or Internet behavior. The online advertising and eCommerce markets may also be adversely affected by privacy issues relating to such targeted advertising, including that which makes use of personalized information. Furthermore, online merchants may not be able to establish online commerce models that are cost effective and may not learn how to effectively compete with other Web sites or offline merchants. In addition, consumers may not continue to shift their spending on goods and services from offline outlets to the Internet. As a result, growth in the use of the Internet for eCommerce may not continue at a rapid rate, or the Internet may not be adopted as a medium of commerce by a broad base of customers or companies worldwide. Because of the foregoing factors, among others, the market for Internet advertising and eCommerce may not continue to grow at significant rates. If these markets do not continue to develop, or if they develop slower than expected, our business will suffer.

Our growth depends upon our ability to retain existing large customers and add new large customers; however, to the extent we are successful in doing so, our ability to maintain profitability and positive cash flow may be impaired.

Our success depends in part on our ability to sell our products to large customers and on the renewal of the subscriptions of those customers in subsequent years. For the years ended December 31, 2004, 2005 and 2006, we derived over 38%, 41% and 39%, respectively, of our total revenues from our top 10 customers. The loss of any one or more of those customers could decrease our revenues and harm our current and future operating results. The addition of new large customers or increases in sales to existing large customers may require particularly long implementation periods and other costs, which may adversely affect our profitability. To compete effectively, we have in the past been, and may in the future be, forced to offer significant discounts to maintain existing customers or acquire other large customers. In addition, we may be forced to reduce or withdraw from our relationships with certain existing customers or refrain from acquiring certain new customers in order to acquire or maintain relationships with important large customers. As a result, new large customers or increased usage of our products by large customers may cause our profits to decline and our ability to sell our products to other customers could be adversely affected.

We derive a significant portion of our revenues from a single customer, Microsoft Corporation. For the years ended December 31, 2004, 2005 and 2006, we derived approximately 5%, 14% and 12%, respectively,

of our total revenues from Microsoft, If Microsoft were to cease or substantially reduce its use of our products, our revenues and earnings might decline.

If we fail to develop our brand, our business may suffer.

We believe that building and maintaining awareness of comScore and our portfolio of products in a cost-effective manner is critical to achieving widespread acceptance of our current and future products and is an important element in attracting new customers. We rely on our relationships with the media and the exposure we receive from numerous citations of our data by media outlets to build brand awareness and credibility among our customers and the marketplace. Furthermore, we believe that brand recognition will become more important for us as competition in our market increases. Our brand's success will depend on the effectiveness of our marketing efforts and on our ability to provide reliable and valuable products to our customers at competitive prices. Our brand marketing activities may not yield increased revenues, and even if they do, any increased revenues may not offset the expenses we incur in attempting to build our brand. If we fail to successfully market our brand, we may fail to attract new customers, retain existing customers or attract media coverage to the extent necessary to realize a sufficient return on our brand-building efforts, and our business and results of operations could suffer.

Failure to effectively expand our sales and marketing capabilities could harm our ability to increase our customer base and achieve broader market acceptance of our products.

Increasing our customer base and achieving broader market acceptance of our products will depend to a significant extent on our ability to expand our sales and marketing operations. We expect to continue to rely on our direct sales force to obtain new customers. We plan to continue to expand our direct sales force both domestically and internationally. We believe that there is significant competition for direct sales personnel with the sales skills and technical knowledge that we require. Our ability to achieve significant growth in revenues in the future will depend, in large part, on our success in recruiting, training and retaining sufficient numbers of direct sales personnel. In general, new hires require significant training and substantial experience before becoming productive. Our recent hires and planned hires may not become as productive as we require, and we may be unable to hire or retain sufficient numbers of qualified individuals in the future in the markets where we currently operate or where we seek to conduct business. Our business will be seriously harmed if the efforts to expand our sales and marketing capabilities are not successful or if they do not generate a sufficient increase in revenues.

We have limited experience with respect to our pricing model, and if the prices we charge for our products are unacceptable to our customers, our revenues and operating results will be harmed

We have limited experience in determining the prices for our products that our existing and potential customers will find acceptable. As the market for our products matures, or as new competitors introduce new products or services that compete with ours, we may be unable to renew our agreements with existing customers or attract new customers at the prices we have historically charged. As a result, it is possible that future competitive dynamics in our market may require us to reduce our prices, which could have an adverse effect on our revenues, profitability and operating results.

We derive a significant portion of our revenues from sales of our subscription-based digital marketing intelligence products. If our customers terminate or fail to renew their subscriptions, our business could suffer.

We currently derive a significant portion of our revenues from our subscription-based digital marketing intelligence products. Subscription-based products accounted for 70% and 75% of our revenues in 2005 and 2006, respectively. However, if our customers terminate their subscriptions for our products, do not renew their subscriptions, delay renewals of their subscriptions or renew on terms less favorable to us, our revenues could decline and our business could suffer.

Our customers have no obligation to renew after the expiration of their initial subscription period, which is typically one year, and we cannot assure you that current subscriptions will be renewed at the same or higher price levels, if at all. Some of our customers have elected not to renew their subscription agreements

with us in the past. If we experience a change of control, as defined in such agreements, some of our customers have the right to terminate their subscriptions. Moreover, some of our major customers have the right to cancel their subscription agreements without cause at any time. We have limited historical data with respect to rates of customer subscription renewals, so we cannot accurately predict future customer renewal rates. Our customer renewal rates may decline or fluctuate as a result of a number of factors, including customer satisfaction or dissatisfaction with our products, the prices or functionality of our products offered by our competitors, mergers and acquisitions affecting our customers' spending levels.

If we are unable to sell additional products to our existing customers or attract new customers, our revenue growth will be adversely affected.

To increase our revenues, we believe we must sell additional products to existing customers and regularly add new customers. If our existing and prospective customers do not perceive our products to be of sufficient value and quality, we may not be able to increase sales to existing customers and attract new customers, and our operating results will be adversely affected.

We depend on third parties for data that is critical to our business, and our business could suffer if we cannot continue to obtain data from these suppliers.

We rely on third-party data sources for information regarding certain offline activities of our panelists. The availability and accuracy of these data is important to the continuation and development of our products that link online activity to offline purchases. If this information is not available to us at commercially reasonable terms, or is found to be inaccurate, it could harm our reputation, business and financial performance.

System failures or delays in the operation of our computer and communications systems may harm our business.

Our success depends on the efficient and uninterrupted operation of our computer and communications systems and the third-party data centers we use. Our ability to collect and report accurate data may be interrupted by a number of factors, including our inability to access the Internet, the failure of our network or software systems, computer viruses, security breaches or variability in user traffic on customer Web sites. A failure of our network or data gathering procedures could impede the processing of data, cause the corruption or loss of data or prevent the timely delivery of our products.

In the future, we may need to expand our network and systems at a more rapid pace than we have in the past. Our network or systems may not be capable of meeting the demand for increased capacity, or we may incur additional unanticipated expenses to accommodate these capacity demands. In addition, we may lose valuable data, be unable to obtain or provide data on a timely basis or our network may temporarily shut down if we fail to adequately expand or maintain our network capabilities to meet future requirements. Any lapse in our ability to collect or transmit data may decrease the value of our products and prevent us from providing the data requested by our customers. Any disruption in our network processing or loss of Internet user data may damage our reputation and result in the loss of customers, and our business and results of operations could be adversely affected.

We rely on a small number of third-party service providers to host and deliver our products, and any interruptions or delays in services from these third parties could impair the delivery of our products and harm our business.

We host our products and serve all of our customers from two third-party data center facilities located in Virginia and Illinois. While we operate our equipment inside these facilities, we do not control the operation of either of these facilities, and, depending on service level requirements, we may not continue to operate or maintain redundant data center facilities for all of our products or for all of our data, which could increase our vulnerability. These facilities are vulnerable to damage or interruption from earthquakes, hurricanes, floods,

fires, power loss, telecommunications failures and similar events. They are also subject to break-ins, computer viruses, sabotage, intentional acts of vandalism and other misconduct. A natural disaster or an act of terrorism, a decision to close the facilities without adequate notice or other unanticipated problems could result in lengthy interruptions in availability of our products. We may also encounter capacity limitations at our third-party data centers. Additionally, our data center facility agreements are of limited durations, and our data center facilities have no obligation to renew their agreements with us on commercially reasonable terms, if at all. If we are unable to renew our agreements with the owners of the facilities on commercially reasonable terms, or if we migrate to a new data center, we may experience delays in delivering our products until an agreement with another data center facility can be arranged or the migration to a new facility is completed.

Further, we depend on access to the Internet through third-party bandwidth providers to operate our business. If we lose the services of one or more of our bandwidth providers for any reason, we could experience disruption in the delivery of our products or be required to retain the services of a replacement bandwidth provider. It may be difficult for us to replace any lost bandwidth on commercially reasonable terms, or at all, due to the large amount of bandwidth our operations require.

Our operations also rely heavily on the availability of electrical power and cooling capacity, which are also supplied by third-party providers. If we or the third-party data center operators that we use to deliver our products were to experience a major power outage or if the cost of electrical power increases significantly, our operations and profitability would be harmed. If we or the third-party data centers that we use were to experience a major power outage, we would have to rely on back-up generators, which may not function properly, and their supply may be inadequate. Such a power outage could result in the disruption of our business. Additionally, if our current facilities fail to have sufficient cooling capacity or availability of electrical power, we would need to find alternative facilities.

Any errors, defects, disruptions or other performance problems with our products caused by third parties could harm our reputation and may damage our business. Interruptions in the availability of our products may reduce our revenues due to increased turnaround time to complete projects, cause us to issue credits to customers, cause customers to terminate their subscription and project agreements or adversely affect our renewal rates. Our business would be harmed if our customers or potential customers believe our products are unreliable.

Because our long-term success depends, in part, on our ability to expand the sales of our products to customers located outside of the United States, our business will become increasingly susceptible to risks associated with international operations.

We have very limited experience operating in markets outside of the United States. Our inexperience in operating our business outside of the United States may increase the risk that the international expansion efforts we have begun to undertake will not be successful. In addition, conducting international operations subjects us to new risks that we have not generally faced in the United States. These risks include:

- recruitment and maintenance of a sufficiently large and representative panel both globally and in certain countries;
- · different customer needs and buying behavior than we are accustomed to in the United States;
- · difficulties and expenses associated with tailoring our products to local markets, including their translation into foreign languages;
- · difficulties in staffing and managing international operations;
- · longer accounts receivable payment cycles and difficulties in collecting accounts receivable;
- · potentially adverse tax consequences, including the complexities of foreign value-added taxes and restrictions on the repatriation of earnings;
- · reduced or varied protection for intellectual property rights in some countries;
- the burdens of complying with a wide variety of foreign laws and regulations;

- · fluctuations in currency exchange rates:
- · increased accounting and reporting burdens and complexities; and
- · political, social and economic instability abroad, terrorist attacks and security concerns.

Additionally, operating in international markets requires significant management attention and financial resources. We cannot be certain that the investments and additional resources required to establish and maintain operations in other countries will hold their value or produce desired levels of revenues or profitability. We cannot be certain that we will be able to maintain and increase the size of the Internet user panel that we currently have in various countries or that we will be able to recruit a representative sample for our audience measurement products. In addition, there can be no assurance that Internet usage and eCommerce will continue to grow in international markets. In addition, governmental authorities in various countries have different views regarding regulatory oversight of the Internet. For example, the Chinese government has recently taken steps to restrict the content available to Internet users in China.

The impact of any one or more of these risks could negatively affect or delay our plans to expand our international business and, consequently, our future operating results.

$If we fail to respond to technological developments, our products \ may \ become \ obsolete \ or \ less \ competitive.$

Our future success will depend in part on our ability to modify or enhance our products to meet customer needs, to add functionality and to address technological advancements. For example, online publishers and advertisers have recently started to use Asynchronous JavaScript and XML, or AJAX, a development technique that allows Web applications to quickly make incremental updates without having to refresh the entire Web page. AJAX may make page views a less useful metric for measuring the usage and effectiveness of online media. If our products are not effective at addressing evolving customer needs that result from increased AJAX usage, our business may be harmed. Similarly, technological advances in the handheld device industry may lead to changes in our customers' requirements. For example, if certain handheld devices become the primary mode of receiving content and conducting transactions on the Internet, and we are unable to adapt our software to collect information from such devices, then we would not be able to report on online activity. To remain competitive, we will need to develop new products that address these evolving technologies and standards. However, we may be unsuccessful in identifying new product opportunities or in developing or marketing new products in a timely or cost-effective manner. In addition, our product innovations may not achieve the market penetration or price levels necessary for profitability. If we are unable to develop enhancements to, and new features for, our existing products or if we are unable to develop new products that keep pace with rapid technological developments or changing industry standards, our products may become obsolete, less marketable and less competitive, and our business will be harmed.

The success of our business depends in large part on our ability to protect and enforce our intellectual property rights.

We rely on a combination of patent, copyright, service mark, trademark and trade secret laws, as well as confidentiality procedures and contractual restrictions, to establish and protect our proprietary rights, all of which provide only limited protection. While we have filed a number of patent applications and own one issued patent, we cannot assure you that any additional patents will be issued with respect to any of our pending or future patent applications, nor can we assure you that any patent issued to us will provide adequate protection, or that any patents issued to us will not be challenged, invalidated, circumvented, or held to be unenforceable in actions against alleged infringers. Also, we cannot assure you that any future trademark or service mark registrations will be issued with respect to pending or future applications or that any of our registered trademarks and service marks will be enforceable or provide adequate protection of our proprietary rights. Furthermore, adequate (or any) patent, trademark, service mark, copyright and trade secret protection may not be available in every country in which our services are available.

We endeavor to enter into agreements with our employees and contractors and with parties with whom we do business in order to limit access to and disclosure of our proprietary information. We cannot be certain

that the steps we have taken will prevent unauthorized use of our technology or the reverse engineering of our technology. Moreover, third parties might independently develop technologies that are competitive to ours or that infringe upon our intellectual property. In addition, the legal standards relating to the validity, enforceability and scope of protection of intellectual property rights in Internet-related industries are uncertain and still evolving, both in the United States and in other countries. The protection of our intellectual property rights may depend on our legal actions against any infringers being successful. We cannot be sure any such actions will be successful.

An assertion from a third party that we are infringing its intellectual property, whether such assertions are valid or not, could subject us to costly and time-consuming litigation or expensive licenses.

The Internet, software and technology industries are characterized by the existence of a large number of patents, copyrights, trademarks and trade secrets and by frequent litigation based on allegations of infringement or other violations of intellectual property rights, domestically or internationally. As we grow and face increasing competition, the probability that one or more third parties will make intellectual property rights claims against us increases. In such cases, our technologies may be found to infringe on the intellectual property rights of others. Additionally, many of our subscription agreements may require us to indemnify our customers for third-party intellectual property infringement claims, which would increase our costs if we have to defend such claims and may require that we pay damages and provide alternative services if there were an adverse ruling in any such claims. Intellectual property claims could harm our relationships with our customers, deter future customers from subscribing to our products or expose us to litigation. Even if we are not a party to any litigation between a customer and a third party, an adverse outcome in any such litigation could make it more difficult for us to defend against intellectual property claims by the third party in any subsequent litigation in which we are a named party. Any of these results could adversely affect our brand, business and results of operations.

One of our competitors has filed patent infringement lawsuits against others, demonstrating this party's propensity for patent litigation. It is possible that this third party, or some other third party, may bring an action against us, and thus cause us to incur the substantial costs and risks of litigation. Any intellectual property rights claim against us or our customers, with or without merit, could be time-consuming and expensive to litigate or settle and could divert management resources and attention. An adverse determination also could prevent us from offering our products to our customers and may require that we procure or develop substitute products that do not infringe on other parties' rights.

With respect to any intellectual property rights claim against us or our customers, we may have to pay damages or stop using technology found to be in violation of a third party's rights. We may have to seek a license for the technology, which may not be available on reasonable terms or at all, may significantly increase our operating expenses or may significantly restrict our business activities in one or more respects. We may also be required to develop alternative non-infringing technology, which could require significant effort and expense. Any of these outcomes could adversely affect our business and results of operations.

Domestic or foreign laws, regulations or enforcement actions may limit our ability to collect and use information about Internet users or restrict or prohibit our product offerings, causing a decrease in the value of our products and an adverse impact on the sales of our products.

Our business could be adversely impacted by existing or future laws or regulations of, or actions by, domestic or foreign regulatory agencies. For example, privacy concerns could lead to legislative, judicial and regulatory limitations on our ability to collect, maintain and use information about Internet users in the United States and abroad. Various state legislatures, including those of Utah and California, have enacted legislation designed to protect Internet users' privacy, for example by prohibiting spyware. In recent years, similar legislation has been proposed in other states and at the federal level and has been enacted in foreign countries, most notably by the European Union, which adopted a privacy directive regulating the collection of personally identifiable information online. These laws and regulations, if drafted or interpreted broadly, could be deemed to apply to the technology we use, and could restrict our information collection methods or decrease the amount and utility of the information that we would be permitted to collect. In addition, our ability to conduct business in certain foreign jurisdictions, including China, is restricted by the laws, regulations and agency actions of those jurisdictions. The costs of

compliance with, and the other burdens imposed by, these and other laws or regulatory actions may prevent us from selling our products or increase the costs associated with selling our products, and may affect our ability to invest in or jointly develop products in the United States and in foreign jurisdictions.

In addition, failure to comply with these and other laws and regulations may result in, among other things, administrative enforcement actions and fines, class action lawsuits and civil and criminal liability. State attorneys general, governmental and non-governmental entities and private persons may bring legal actions asserting that our methods of collecting, using and distributing Web site visitor information are illegal or improper, which could require us to spend significant time and resources defending these claims. For example, some companies that collect, use and distribute Web site visitor information have been the subject of governmental investigations and class-action lawsuits. Any such regulatory or civil action that is brought against us, even if unsuccessful, may distract our management's attention, divert our resources, negatively affect our public image or reputation among our panelists and customers and harm our business

The impact of any of these current or future laws or regulations could make it more difficult or expensive to attract or maintain panelists, particularly in affected jurisdictions, and could adversely affect our business and results of operations.

Laws related to the regulation of the Internet could adversely affect our business.

Laws and regulations that apply to communications and commerce over the Internet are becoming more prevalent. In particular, the growth and development of the market for eCommerce has prompted calls for more stringent tax, consumer protection and privacy laws in the United States and abroad that may impose additional burdens on companies conducting business online. The adoption, modification or interpretation of laws or regulations relating to the Internet or our customers' digital operations could negatively affect the businesses of our customers and reduce their demand for our products.

If we fail to respond to evolving industry standards, our products may become obsolete or less competitive.

The market for our products is characterized by rapid technological advances, changes in customer requirements, changes in protocols and evolving industry standards. For example, industry associations such as the Advertising Research Foundation, the Council of American Survey Research Organizations, the Internet Advertising Bureau and the Media Ratings Council have independently initiated efforts to either review online market research methodologies or to develop minimum standards for online market research. Any standards adopted by such organizations may lead to costly changes to our procedures and methodologies. As a result, the cost of developing our digital marketing intelligence products could increase. If we do not adhere to standards prescribed by such industry associations, our customers could choose to purchase products from competing companies that meet such standards. Furthermore, industry associations based in countries outside of the United States often endorse certain vendors or methodologies. If our methodologies fail to receive an endorsement from an important industry association located in a foreign country, advertising agencies, media companies and advertisers in that country may not purchase our products. As a result, our efforts to further expand internationally could be adversely affected.

The success of our business depends on the continued growth of the Internet as a medium for commerce, content, advertising and communications.

Expansion in the sales of our products depends on the continued acceptance of the Internet as a platform for commerce, content, advertising and communications. The use of the Internet as a medium for commerce, content, advertising and communications could be adversely impacted by delays in the development or adoption of new standards and protocols to handle increased demands of Internet activity, security, reliability, cost, ease-of-use, accessibility and quality-of-service. The performance of the Internet and its acceptance as a medium for commerce, content, advertising and communications has been harmed by viruses, worms, and similar malicious programs, and the Internet has experienced a variety of outages and other delays as a result of damage to portions of its infrastructure. If for any reason the Internet does not remain a medium for widespread commerce, content, advertising and communications, the demand for our products would be significantly reduced, which would harm our business.

We rely on our management team and need additional personnel to grow our business, and the loss of one or more key employees or the inability to attract and retain qualified personnel could harm our business.

Our success and future growth depends to a significant degree on the skills and continued services of our management team, including our founders, Magid M. Abraham, Ph.D. and Gian M. Fulgoni. Our future success also depends on our ability to retain, attract and motivate highly skilled technical, managerial, marketing and customer service personnel, including members of our management team. All of our employees work for us on an at-will basis. We plan to hire additional personnel in all areas of our business, particularly for our sales, marketing and technology development areas, both domestically and internationally, which will likely increase our recruiting and hiring costs. Competition for these types of personnel is intense, particularly in the Internet and software industries. As a result, we may be unable to successfully attract or retain qualified personnel. Our inability to retain and attract the necessary personnel could adversely affect our business.

We may expand through investments in, or acquisitions of, other companies, any of which may not be successful and may divert our management's attention.

Our business strategy may include acquiring complementary products, technologies or businesses. We also may enter into relationships with other businesses in order to expand our product offerings, which could involve preferred or exclusive licenses, discount pricing or investments in other companies.

Negotiating any such transactions could be time-consuming, difficult and expensive, and our ability to close these transactions may be subject to regulatory or other approvals and other conditions which are beyond our control. Consequently, we can make no assurances that any such transactions, if undertaken and announced, would be completed.

An acquisition, investment or business relationship may result in unforeseen operating difficulties and expenditures. In particular, we may encounter difficulties assimilating or integrating the businesses, technologies, products, personnel or operations of the acquired companies, particularly if the key personnel of the acquired company choose not to be employed by us, and we may have difficulty retaining the customers of any acquired business due to changes in management and ownership. Acquisitions may also disrupt our ongoing business, divert our resources and require significant management attention that would otherwise be available for ongoing development of our business. Moreover, we cannot assure you that the anticipated benefits of any acquisition, investment or business relationship would be realized or that we would not be exposed to unknown liabilities. In connection with any such transaction, we may:

- encounter difficulties retaining key employees of the acquired company or integrating diverse business cultures;
- · issue additional equity securities that would dilute the common stock held by existing stockholders;
- incur large charges or substantial liabilities;
- become subject to adverse tax consequences, substantial depreciation or deferred compensation charges;
- use cash that we may need in the future to operate our business; and
- · incur debt on terms unfavorable to us or that we are unable to repay.

The impact of any one or more of these factors could adversely affect our business or results of operations or cause the price of our common stock to decline substantially.

Changes in, or interpretations of, accounting rules and regulations, including recent rules and regulations regarding expensing of stock options, could result in unfavorable accounting charges or cause us to change our compensation policies.

Accounting methods and policies, including policies governing revenue recognition, expenses and accounting for stock options are continually subject to review, interpretation, and guidance from relevant accounting authorities, including the Financial Accounting Standards Board, or FASB, and the SEC. Changes

to, or interpretations of, accounting methods or policies in the future may require us to reclassify, restate or otherwise change or revise our financial statements, including those contained in this prospectus.

On December 16, 2004, the FASB issued SFAS No. 123R (revised 2004), Share-Based Payment, which is a revision of SFAS No. 123, Accounting for Stock-Based Compensation (SFAS No. 123R). SFAS No. 123R supersedes APB Opinion No. 25, Accounting for Stock Issued to Employees, and amends SFAS No. 95, Statement of Cash Flows. Generally, the approach in SFAS No. 123R is similar to the approach described in SFAS No. 123. However, SFAS No. 123R requires all share-based payments to employees, including grants of employee stock options, to be recognized in the income statement based on their fair values. We were required to adopt SFAS No. 123R on January 1, 2006, and have adopted it as of that date.

As permitted by SFAS No. 123, we accounted for share-based payments to employees through December 31, 2005 using APB Opinion No. 25's intrinsic value method and, as such, generally recognized no compensation cost for employee stock options. Accordingly, the adoption of SFAS No. 123R's fair value method has had a significant impact on the presentation of our results of operations, although it has not impacted our overall financial position. The long-term impact of adoption of SFAS No. 123R cannot be predicted at this time because it will depend on levels of share-based payments granted in the future and the assumptions for the variables which impact the computation of the fair value of any such grants.

Historically, we have used stock options as part of our compensation programs to motivate and retain existing employees and to attract new employees. Because we are now required to expense stock options, we may choose to reduce our reliance on stock options as part of our compensation packages. If we reduce our use of stock options, it may be more difficult for us to retain and attract qualified employees. If we do not reduce our use of stock options, our expenses in future periods may increase. Beginning in 2007, we expect to make use of restricted stock awards and reduce our use of stock options as a form of stock-based compensation, but we cannot be certain whether or how our stock-based compensation policy will change in the future

Investors could lose confidence in our financial reports, and our business and stock price may be adversely affected, if our internal control over financial reporting is found by management or by our independent registered public accounting firm to not be adequate or if we disclose significant existing or potential deficiencies or material weaknesses in those controls.

Section 404 of the Sarbanes-Oxley Act of 2002 requires us to include a report on our internal control over financial reporting in our Annual Report on Form 10-K for each year beginning with the year ending December 31, 2008. That report must include management's assessment of the effectiveness of our internal control over financial reporting as of the end of that and each subsequent fiscal year. Additionally, our independent registered public accounting firm will be required to issue a report on management's assessment of our internal control over financial reporting and on their evaluation of the operating effectiveness of our internal control over financial reporting.

We continue to evaluate our existing internal controls against the standards adopted by the Public Company Accounting Oversight Board, or PCAOB. During the course of our ongoing evaluation of our internal controls, we have in the past identified, and may in the future identify, areas requiring improvement, and may have to design enhanced processes and controls to address issues identified through this review. Remedying any significant deficiencies or material weaknesses that we or our independent registered public accounting firm may identify could require us to incur significant costs and expend significant time and management resources. We cannot assure you that any of the measures we may implement to remedy any such deficiencies will effectively mitigate or remedy such deficiencies. In addition, we cannot assure you that we will be able to complete the work necessary for our management report in a timely manner, or that we will be able to complete any work required for our management to be able to conclude that our internal control over financial reporting is operating effectively. If we are not able to complete the assessment under Section 404 in a timely manner or to remedy any identified material weaknesses, we and our independent registered public accounting firm would be unable to conclude that our internal control over financial reporting is effective as of December 31, 2008. If our internal control over financial reporting is

found by management or by our independent registered public accountant to not be adequate or if we disclose significant existing or potential deficiencies or material weaknesses in those controls, investors could lose confidence in our financial reports, we could be subject to sanctions or investigations by The NASDAQ Global Market, the Securities and Exchange Commission or other regulatory authorities and our stock price could be adversely affected.

A determination that there is a significant deficiency or material weakness in the effectiveness of our internal control over financial reporting could also reduce our ability to obtain financing or could increase the cost of any financing we obtain and require additional expenditures to comply with applicable requirements.

Our net operating loss carryforwards may expire unutilized or underutilized, which could prevent us from offsetting future taxable income.

We have experienced "changes in control" that have triggered the limitations of Section 382 of the Internal Revenue Code on our net operating loss carryforwards. As a result, we may be limited in the portion of net operating loss carryforwards that we can use in the future to offset taxable income for U.S. Federal income tax purposes.

At December 31, 2006, we had both federal and state net operating loss carryforwards of approximately \$81.2 million each which are available to offset future taxable income. The federal net operating loss carryforwards will begin to expire in 2020. The state net operating loss carryforwards began to expire in 2006.

In addition, at December 31, 2005 and 2006, we had net operating loss carryforwards for tax purposes related to our foreign subsidiaries of \$966,000 and \$703,000, respectively, which begin to expire in 2010.

In 2006, deferred tax assets, before valuation allowance, decreased approximately \$2.4 million due to our use of net operating loss carryforwards to offset taxable income.

We periodically assess the likelihood that we will be able to recover our deferred tax assets. We consider all available evidence, both positive and negative, including historical levels of income, expectations and risks associated with estimates of future taxable income and ongoing prudent and feasible profits. As a result of this analysis of all available evidence, both positive and negative, we concluded that a full valuation allowance against deferred tax assets should be applied as of December 31, 2006. To the extent we determine that all or a portion of our valuation allowance is no longer necessary, we will recognize an income tax benefit in the period such determination is made for the reversal of the valuation allowance. Once the valuation allowance is eliminated or reduced, its reversal will no longer be available to offset our current tax provision. These events could have a material impact on our reported results of operations.

We may require additional capital to support business growth, and this capital may not be available on acceptable terms or at all.

We intend to continue to make investments to support our business growth and may require additional funds to respond to business challenges, including the need to develop new products or enhance our existing products, enhance our operating infrastructure and acquire complementary businesses and technologies. Accordingly, we may need to engage in equity or debt financings to secure additional funds. If we raise additional funds through further issuances of equity or convertible debt securities, our existing stockholders could suffer significant dilution, and any new equity securities we issue could have rights, preferences and privileges superior to those of holders of our common stock. Any debt financing secured by us in the future could include restrictive covenants relating to our capital raising activities and other financial and operational matters, which may make it more difficult for us to obtain additional capital and to pursue business opportunities, including potential acquisitions. In addition, we may not be able to obtain additional financing on terms favorable to us or at all. If we are unable to obtain adequate financing or financing on terms satisfactory to us when we require it, our ability to continue to support our business growth and to respond to business challenges could be significantly limited. In addition, the terms of any additional equity or debt issuances may adversely affect the value and price of our common stock.

Risks Related to this Offering

We cannot assure you that a market will develop for our common stock or what the market price of our common stock will be.

Before this offering, there was no public trading market for our common stock, and we cannot assure you that one will develop or be sustained after this offering. If a market does not develop or is not sustained, it may be difficult for you to sell your shares of common stock at an attractive price or at all. We cannot predict the prices at which our common stock will trade.

The initial public offering price for our common stock will be determined through our negotiations with the underwriters, and may not bear any relationship to the market price at which our common stock will trade after this offering or to any other established criteria of the value of our business. The price of our common stock that will prevail in the market after this offering may be higher or lower than the price you pay, depending on many factors, some of which are beyond our control and may not be related to our operating performance. It is possible that, in future quarters, our operating results may be below the expectations of securities analysts or investors. As a result of these and other factors, the price of our common stock may decline, possibly materially. These fluctuations could cause you to lose all or part of your investment in our common stock. The public trading price for our common stock after this offering will be affected by a number of factors, including:

- price and volume fluctuations in the overall stock market from time to time;
- · volatility in the market price and trading volume of technology companies and of companies in our industry;
- actual or anticipated changes or fluctuations in our operating results;
- · actual or anticipated changes in expectations regarding our performance by investors or securities analysts;
- the failure of securities analysts to cover our common stock after this offering or changes in financial estimates by analysts;
- · actual or anticipated developments in our competitors' businesses or the competitive landscape;
- · actual or perceived inaccuracies in information we provide to our customers or the media;
- litigation involving us, our industry or both;
- · regulatory developments;
- · privacy and security concerns, including public perception of our practices as an invasion of privacy;
- · general economic conditions and trends;
- · major catastrophic events;
- · sales of large blocks of our stock;
- the timing and success of new product introductions or upgrades by us or our competitors;
- · changes in our pricing policies or payment terms or those of our competitors;
- · concerns relating to the security of our network and systems;
- · our ability to expand our operations, domestically and internationally, and the amount and timing of expenditures related to this expansion; or
- · departures of key personnel.

In addition, the stock prices of many technology companies have experienced wide fluctuations that have often been unrelated to the operating performance of those companies.

In the past, following periods of volatility in the market price of a company's securities, securities class action litigation has often been brought against that company. If our stock price is volatile, we may become the target of securities litigation, which could result in substantial costs and divert our management's attention and resources from our business.

Our stock price could decline due to the large number of outstanding shares of our common stock eligible for future sale.

Sales of substantial amounts of our common stock in the public market following this offering, or the perception that these sales could occur, could cause the market price of our common stock to decline. These sales could also make it more difficult for us to sell equity or equity-related securities in the future at a time and price that we deem appropriate.

Upon completion of this offering, we will have outstanding options or warrants after , 2007. The shares sold pursuant to this offering will be immediately tradable without restriction. Of the remaining shares:

- shares will be eligible for sale immediately upon completion of this offering, subject in some cases to volume and other restrictions of Rule 144 and Rule 701 under the Securities Act:
- shares will be eligible for sale upon the expiration of lock-up agreements, subject in some cases to volume and other restrictions of Rule 144 and Rule 701 under the Securities Act; and
- · shares will be eligible for sale upon the exercise of vested options after the expiration of the lock-up agreements.

The lock-up agreements expire 180 days after the date of this prospectus, provided that the 180-day period may be extended in certain cases for up to 34 additional days under certain circumstances where we announce or pre-announce earnings or a material event within approximately 17 days prior to, or approximately 16 days after, the termination of the 180-day period. Credit Suisse Securities (USA) LLC may, in its sole discretion and at any time without notice, release all or any portion of the securities subject to lock-up agreement. After the closing of this offering, we intend to register approximately shares of common stock that have been reserved for future issuance under our stock incentive plans.

Insiders will continue to have substantial control over us after this offering, which could limit your ability to influence the outcome of key transactions, including a change of control.

Our directors, executive officers and each of our stockholders who own greater than 5% of our outstanding common stock and their affiliates, in the aggregate, will beneficially own approximately % of the outstanding shares of our common stock after this offering. As a result, these stockholders, if acting together, would be able to influence or control matters requiring approval by our stockholders, including the election of directors and the approval of mergers, acquisitions or other extraordinary transactions. They may have interests that differ from yours and may vote in a way with which you disagree and which may be adverse to your interests. This concentration of ownership may have the effect of delaying, preventing or deterring a change of control of our company, could deprive our stockholders of an opportunity to receive a premium for their common stock as part of a sale of our company and might affect the market price of our common stock.

Our management will have broad discretion over the use of the proceeds from this offering and may not apply the proceeds of this offering in ways that increase the value of your investment.

Our management will have broad discretion to use the net proceeds we receive from this offering, and you will be relying on its judgment regarding the application of these proceeds. We expect to use the net proceeds from this offering for general corporate purposes, which may include working capital, capital expenditures, other corporate expenses and potential acquisitions of complementary products, technologies or businesses. We have not allocated these net proceeds for any specific purposes. However, management may not apply the net proceeds of this offering in ways that increase the value of your investment.

$If you \ purchase \ shares \ of \ our \ common \ stock \ in \ this \ offering, you \ will \ experience \ substantial \ and \ immediate \ dilution.$

If you purchase shares of our common stock in this offering, you will experience substantial and immediate dilution of \$ per share based on an assumed initial public offering price of \$ per share, the mid-point of the range shown on the cover of this prospectus, because the price that you pay will be substantially greater than the net tangible book value per share of the common stock that you acquire. This dilution is due in large part to the fact that our earlier investors paid substantially less than the initial public offering price when they purchased their shares of our capital stock. You will experience additional dilution upon the exercise of options to purchase common stock under our equity incentive plans, if we issue restricted stock to our employees under these plans or if we otherwise issue additional shares of our common stock. See "Dilution."

We will incur increased costs and demands upon management as a result of complying with the laws and regulations affecting a public company, which could adversely affect our operating results.

As a public company, we will incur significant legal, accounting and other expenses that we did not incur as a private company. In addition, the Sarbanes-Oxley Act of 2002, as well as rules implemented by the Securities and Exchange Commission and The NASDAQ Stock Market, requires certain corporate governance practices for public companies. Our management and other personnel will need to devote a substantial amount of time to public reporting requirements and corporate governance. We expect these rules and regulations to significantly increase our legal and financial compliance costs and to make some activities more time-consuming and costly. We will also incur additional costs associated with our public company reporting requirements. We are unable to currently estimate these costs with any degree of certainty. If these costs are not offset by increased revenues and improved financial performance, our operating results would be adversely affected. We also expect these rules and regulations to make it more difficult and more expensive for us to obtain director and officer liability insurance, and we may be required to accept reduced policy limits and coverage or incur substantially higher costs to obtain the same or similar coverage. As a result, it may be more difficult for us to attract and retain qualified people to serve on our board of directors or as executive officers.

Provisions in our certificate of incorporation and bylaws and under Delaware law might discourage, delay or prevent a change of control of our company or changes in our management and, therefore, depress the trading price of our common stock.

Our certificate of incorporation and bylaws contain provisions that could depress the trading price of our common stock by acting to discourage, delay or prevent a change of control of our company or changes in our management that the stockholders of our company may deem advantageous. These provisions:

- · establish a classified board of directors so that not all members of our board of directors are elected at one time;
- authorize "blank check" preferred stock that our board of directors could issue to increase the number of outstanding shares to discourage a takeover attempt;
- · prohibit stockholder action by written consent, which means that all stockholder actions must be taken at a meeting of our stockholders;
- · prohibit stockholders from calling a special meeting of our stockholders;
- provide that the board of directors is expressly authorized to make, alter or repeal our bylaws; and
- establish advance notice requirements for nominations for elections to our board of directors or for proposing matters that can be acted upon by stockholders at stockholder meetings.

Additionally, we are subject to Section 203 of the Delaware General Corporation Law, which prohibits a Delaware corporation from engaging in any of a broad range of business combinations with any "interested" stockholder for a period of three years following the date on which the stockholder became an "interested" stockholder and which may discourage, delay or prevent a change of control of our company.

CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS AND INDUSTRY DATA

This prospectus, including the sections entitled "Summary," "Risk Factors," "Management's Discussion and Analysis of Financial Condition and Results of Operations" and "Business," contains forward-looking statements. These statements may relate to, but are not limited to, expectations of future operating results or financial performance, capital expenditures, introduction of new products, regulatory compliance, plans for growth and future operations, as well as assumptions relating to the foregoing. Forward-looking statements are inherently subject to risks and uncertainties, some of which cannot be predicted or quantified. These risks and other factors include, but are not limited to, those listed under "Risk Factors." In some cases, you can identify forward-looking statements by terminology such as "may," "will," "should," "could," "expect," "plan," "anticipate," "believe," "estimate," "predict," "intend," "potential," "might," "would," "continue" or the negative of these terms or other comparable terminology. These statements are only predictions. Actual events or results may differ materially

We believe that it is important to communicate our future expectations to our investors. However, there may be events in the future that we are not able to accurately predict or control and that may cause our actual results to differ materially from the expectations we describe in our forward-looking statements. Except as required by applicable law, including the securities laws of the United States and the rules and regulations of the SEC, we do not plan to publicly update or revise any forward-looking statements after we distribute this prospectus, whether as a result of any new information, future events or otherwise. Potential investors should not place undue reliance on our forward-looking statements. Before you invest in our common stock, you should be aware that the occurrence of any of the events described in the "Risk Factors" section and elsewhere in this prospectus could harm our business, prospects, operating results and financial condition. Although we believe that the expectations reflected in the forward-looking statements are reasonable, we cannot guarantee future results, levels of activity, performance or achievements.

This prospectus also contains estimates and other information concerning our industry, including market size and growth rates of the markets in which we participate, that are based on industry publications, surveys and forecasts, including those generated by Forrester Research, IDC, JupiterResearch, Infonetics, the Internet Advertising Bureau and PriceWaterhouseCoopers. This information involves a number of assumptions and limitations, and you are cautioned not to give undue weight to these estimates. These industry publications, surveys and forecasts generally indicate that their information has been obtained from sources believed to be reliable, but do not guarantee the accuracy and completeness of their information. We have not independently verified their data and accordingly cannot guarantee their accuracy or completeness. The industry in which we operate is subject to a high degree of uncertainty and risk due to a variety of factors, including those described in "Risk Factors." These and other factors could cause actual results to differ materially from those expressed in these publications, surveys and forecasts.

USE OF PROCEEDS

We estimate that the net proceeds from the sale of the initial public offering price of \$ per share, the mid-point of the range on the front cover of this prospectus, after deducting underwriting discounts and commissions and estimated offering expenses. If the underwriters' over-allotment option is exercised in full, we estimate that we will receive additional net proceeds of approximately \$ million. We will not receive any proceeds from the sale of shares of our common stock by the selling stockholders.

The principal purposes of this offering are to obtain additional capital, to create a public market for our common stock and to facilitate our future access to the public equity markets.

We will use the net proceeds from this offering for general corporate purposes, which may include working capital, capital expenditures, other corporate expenses and acquisitions of complementary products, technologies or businesses. We currently have no agreements or commitments with respect to any such acquisitions. The timing and amount of our actual expenditures will be based on many factors, including cash flows from operations and the anticipated growth of our business. Pending these uses, we intend to invest the net proceeds of this offering primarily in short-term, investment-grade, interest-bearing instruments.

If we were to price the offering at \$ per share, the low end of the range on the cover of this prospectus, we estimate that we would receive net proceeds of \$ million, assuming the total number of shares offered by us remains the same and after deducting underwriting discounts and commissions and estimated offering expenses payable by us. If we were to price the offering at \$ per share, the high end of the range on the cover of this prospectus, then we estimate that we would receive net proceeds of \$ million, assuming the total number of shares offered by us remains the same and after deducting underwriting discounts and commissions and estimated offering expenses payable by us.

DIVIDEND POLICY

We have never declared or paid any dividends on our capital stock. We anticipate that we will retain any earnings to support operations and to finance the growth and development of our business. Accordingly, we do not expect to pay cash dividends on our common stock in the foreseeable future.

CAPITALIZATION

The following table sets forth our cash, cash equivalents and short-term investments and capitalization as of December 31, 2006:

- on an actual basis without any adjustments to reflect subsequent or anticipated events;
- on a pro forma basis reflecting (i) the conversion of all outstanding shares of our Series A, Series B, Series C, Series C-1, Series D and Series E preferred stock into an aggregate of 86,286,697 shares of our common stock effective immediately prior to the completion of this offering, for a total of 108,025,682 shares of common stock, which amount includes 1,738,172 shares subject to put and (ii) the reclassification of our preferred stock warrant liabilities from current liabilities to additional paid in capital effective upon the completion of this offering; and
- on a pro forma as adjusted basis reflecting the conversion and reclassification described above and the receipt by us of the net proceeds from the sale of shares of common stock in this offering at an assumed initial public offering price of \$ per share, after deducting the estimated underwriting discounts and commissions and estimated offering expenses payable by us.

You should read this table in conjunction with "Management's Discussion and Analysis of Financial Condition and Results of Operations" and our consolidated financial statements and related notes included elsewhere in this prospectus.

	As of December 31, 2006				
		Actual Pro Forma (In thousands, except share data)			Pro Forma as Adjusted lata)
Cash, cash equivalents and short-term investments	\$	16,032	\$	16,032	
Preferred stock warrant liabilities		1,005		_	
Redeemable preferred stock, \$0.001 par value, 73,673,224 shares authorized; 71,829,471 shares issued and outstanding actual;					
no shares issued or outstanding pro forma and pro forma as adjusted		101,695		_	
Common stock subject to put right, 1,738,172 shares outstanding		4,357		4,357	
Stockholders' equity (deficit):					
Common stock, \$0.001 par value; 130,000,000 shares authorized, 20,000,813 shares issued and outstanding actual; 150,000,000 shares authorized, 106,287,510 shares issued and outstanding pro forma and shares issued and					
outstanding pro forma as adjusted		20		106	
Additional paid-in capital		_		102,614	
Accumulated other comprehensive loss		(75)		(75)	
Accumulated deficit		(99,502)		(99,502)	
Total stockholders' equity (deficit)		(99,557)		3,143	
Total capitalization	\$	7,500	\$	7,500	

The table above excludes, as of December 31, 2006:

- 13,619,700 shares of common stock issuable upon exercise of options outstanding at a weighted-average exercise price of \$0.40 per share;
- 5,316,147 shares of common stock reserved for future issuance under our 1999 Stock Plan;
- 7,000,000 shares of common stock reserved for future issuance under our 2007 Equity Incentive Plan, which will be effective upon completion of this offering;

- 100,000 shares of common stock issuable upon the exercise of a warrant, which warrant shall terminate if not exercised prior to this offering, at an exercise price of \$1.00 per share; and
- 775,923 shares of common stock issuable upon the exercise of warrants, which total includes warrants for our preferred stock that will become exercisable for common stock after this offering, at a weighted-average exercise price of \$0.96 per share.

A \$1.00 decrease or increase in the offering price would result in an approximately \$ million increase or decrease in each of as adjusted cash and cash equivalents, as adjusted additional paid-in capital, as adjusted total stockholders' equity and as adjusted total capitalization, assuming the total number of shares offered by us remains the same and after deducting estimated underwriting discounts and commissions and estimated offering expenses payable by us.

DILUTION

If you invest in our common stock, your interest will be diluted to the extent of the difference between the public offering price per share of our common stock and the pro forma as adjusted net tangible book value per share of our common stock after this offering. Our pro forma net tangible book value as of December 31, 2006 was \$5.2 million, or \$0.05 per share of common stock. Pro forma net tangible book value per share represents total tangible assets less total liabilities, divided by the number of shares of common stock outstanding after giving effect to the conversion of all outstanding shares of our Series A, Series B, Series C, Series D and Series E preferred stock into an aggregate of 86,286,697 shares of our common stock effective immediately prior to the completion of this offering, for a total of 108,025,682 shares of common stock outstanding on December 31, 2006, which amount includes 1,738,172 shares subject to put. After giving effect to the sale by us of shares of our common stock in this offering at the assumed initial public offering price of \$ per share, the mid-point of the range on the front cover of this prospectus, and after deducting the underwriting discounts and commissions and our estimated offering expenses, our pro forma as adjusted net tangible book value as of December 31, 2006 would have been \$ million, or \$ per share. This represents an immediate increase in net tangible book value of \$ per share to our existing stockholders and an immediate dilution of \$ per share to our new investors purchasing shares of common stock in this offering. The following table illustrates this dilution on a per share hasis:

Assumed initial public offering price per share		\$
Pro forma net tangible book value per share as of December 31, 2006	\$ 0.05	
Increase in pro forma net tangible book value per share attributable to this offering per share to existing investors		
Pro forma as adjusted net tangible book value per share after this offering		
Dilution per share to new investors		\$

The following table sets forth as of December 31, 2006, on a pro forma as adjusted basis, the differences between the number of shares of common stock purchased from us, the total consideration paid, and the average price per share paid by existing stockholders and new investors purchasing shares of our common stock in this offering based on an assumed initial public offering price of \$ per share, the mid-point of the range on the front cover of this prospectus, and before deducting underwriting discounts and commissions and estimated offering expenses.

	Shares Purchased	Shares Purchased			ion	Average Price	
	Number	Percent		Amount	t Percent per Sha		er Share
Existing stockholders	108,025,682	%	\$	88,752,479	%	\$	0.82
New investors							
Total		100.0%	\$		100.0%	\$	

The table above excludes, as of December 31, 2006:

- 13,619,700 shares of common stock issuable upon exercise of options outstanding at a weighted-average exercise price of \$0.40 per share;
- 5.316.147 shares of common stock reserved for future issuance under our 1999 Stock Plan;
- 7,000,000 shares of common stock reserved for future issuance under our 2007 Equity Incentive Plan, which will be effective upon completion of this offering;
- 100,000 shares of common stock issuable upon the exercise of a warrant, which warrant shall terminate if not exercised prior to this offering, at an exercise price of \$1.00 per share; and
- 775,923 shares of common stock issuable upon the exercise of warrants, which total includes warrants for our preferred stock that will become exercisable for common stock after this offering, at a weighted-average exercise price of \$0.96 per share.

If the underwriters exercise their over-allotment option in full, the percentage of shares of common stock held by existing stockholders will decrease to approximately % of the total number of shares of our common stock outstanding after this offering, and the number of shares held by new investors will be increased to , or approximately % of the total number of shares of our common stock outstanding after this offering.

A \$1.00 decrease in the assumed offering price would decrease our net tangible book value after this offering by \$ million and dilution in net tangible book value per share to new investors by \$, assuming the total number of shares offered by us remains the same and after deducting estimated underwriting discounts and commissions and estimated offering expenses payable by us. A \$1.00 decrease in the assumed offering price would decrease each of total consideration paid by new investors in the offering and total consideration paid by all stockholders by \$ million, assuming the total number of shares offered by us remains the same and before deducting estimated underwriting discounts and commissions and estimated offering expenses payable by us.

A \$1.00 increase in the assumed offering price would increase our net tangible book value after this offering by \$ million and dilution in net tangible book value per share to new investors by \$, assuming the total number of shares offered by us remains the same and after deducting estimated underwriting discounts and commissions and estimated offering expenses payable by us. A \$1.00 increase in the assumed offering price would increase each of total consideration paid by new investors in the offering and total consideration paid by all stockholders by \$ million, assuming the total number of shares offered by us remains the same and before deducting estimated underwriting discounts and commissions and estimated offering expenses payable by us.

SELECTED CONSOLIDATED FINANCIAL DATA

You should read the selected consolidated financial data set forth below in conjunction with our consolidated financial statements, the notes to our consolidated financial statements and "Management's Discussion and Analysis of Financial Condition and Results of Operations" contained elsewhere in this prospectus.

The consolidated statements of operations data and the consolidated statements of cash flows data for the years ended January 31, 2003 and December 31, 2003 as well as the consolidated balance sheet data as of January 31, 2003 and December 31, 2003 and 2004 are derived from our audited consolidated financial statements not included in this prospectus. The consolidated statements of operations data and the consolidated statements of cash flows data for each of the three years ended December 31, 2004, 2005 and 2006 as well as the consolidated balance sheet data as of December 31, 2005 and 2006 are derived from our audited consolidated financial statements that are included elsewhere in this prospectus. In 2003, we changed our fiscal year to the twelve months ended December 31. The year ended January 31, 2003 and the year ended December 31, 2003 in the table below both include the results of operations for the month ended January 31, 2003. The pro forma basic net income per share data are unaudited and give effect to the conversion into common stock of all outstanding shares of our Series A, Series C, Series C-1, Series D and Series E preferred stock from their dates of original issuance. Our historical results are not necessarily indicative of results to be expected for future periods.

		Year Ended January 31,	Year Ended December 31,							
		2003		2003	ndo ave	2004 ept share and per sh	ava data)	2005		2006
Consolidated Statement of Operations Data:				(III tilousa	mus, exc	ept snare and per sn	are uata)			
Revenues	\$	15,400	\$	23,355	\$	34,894	\$	50,267	\$	66,293
	Ф		Þ		J.		Þ		D.	
Cost of revenues(1) Selling and marketing(1)		14,925 9,134		15,671 11.677		13,153 13,890		18,218 18,953		20,560 21,473
Research and development(1)		6,172		5.444		5,493		7,416		9.009
General and administrative(1)		4,431		4,124		4,982		7,416		8,293
Amortization		562		4,124 772		356		2,437		1,371
					_					
Total expenses from operations		35,224	_	37,688	_	37,874	_	54,113		60,706
(Loss) income from operations		(19,824)		(14,333)		(2,980)		(3,846)		5,587
Interest (expense) income, net		(885)		(595)		(246)		(208)		231
(Loss) gain from foreign currency		_		_		_		(96)		125
Revaluation of preferred stock warrant liabilities								(14)		(224)
(Loss) income before income taxes and cumulative effect of change										
in accounting principle		(20,708)		(14,928)		(3,226)		(4,164)		5,719
(Benefit) provision for income taxes								(182)		50
Net (loss) income before cumulative effect of change in accounting										
principle		(20,708)		(14,928)		(3,226)		(3,982)		5,669
Cumulative effect of change in accounting principle				_		_		(440)		_
Net (loss) income		(20,708)		(14,928)		(3,226)		(4,422)		5,669
Accretion of redeemable preferred stock		(2,216)		(2,476)		(2,141)		(2,638)		(3,179)
Net (loss) income attributable to common stockholders	\$	(22,924)	\$	(17,404)	\$	(5,367)	\$	(7,060)	\$	2,490
Net (loss) income attributable to common stockholders per common share:										
Basic and diluted	\$	(1.77)	\$	(1.29)	\$	(0.38)	\$	(0.46)	\$	0.00
Weighted-average number of shares used in per share calculations:										
Basic and diluted		12,918,989		13,451,440		14,358,561		15,650,969		19,236,064
Pro forma net (loss) income attributable to common stockholders										
per common share:										
Basic										
Diluted										
Pro forma weighted-average number of shares used in per share										
calculations:										
Basic										
Diluted										

(1) Amortization of stock-based compensation is included in the line items above as follows:

		Ended ary 31,			Year Ende	d December	31,		
	2	003	2	003	004 ousands)	2	005	2	2006
Cost of revenues	\$	_	\$	_	\$ _	\$	_	\$	12
Selling and marketing		_		_	_		_		82
Research and development		_		_	_		_		13
General and administrative		128		171	14		3		91
Total	\$	128	\$	171	\$ 14	\$	3	\$	198

	Ja	As of January 31, As of December 31,							
		2003 2003 2004 (In thousands)		2005		 2006			
Consolidated Balance Sheet Data:									
Cash, cash equivalents and short-term investments	\$	6,973	\$	9,557	\$	8,404	\$	9,174	\$ 16,032
Total current assets		11,778		15,482		15,678		20,792	31,493
Total assets		23,603		22,154		22,967		29,477	42,087
Total current liabilities		13,645		15,515		18,591		27,220	32,880
Equipment loan and capital lease obligations, long-term		4,072		2,421		1,438		1,283	2,261
Preferred stock warrant liabilities and common stock subject to put		404		349		2,218		4,997	5,362
Redeemable preferred stock		78,586		93,737		95,878		98,516	101,695
Stockholders' deficit		(73,735)		(89,919)		(95,230)		(102,294)	(99,557)

	Y	ear Ended								
	Ja	anuary 31,	Year Ended December 31,							
		2003	2003	003 2004		2005			2006	
				(In t	housands)					
Consolidated Statement of Cash Flows Data:										
Net cash provided by (used in) operating activities	\$	(12,653)	\$ (3,912)	\$	1,907	\$	4,253	\$	10,905	
Depreciation and amortization		5,865	6,604		2,745		5,123		4,259	
Capital expenditures		1,962	726		1,208		1,071		2,314	

	Year Ended January 31,				r 31,			
	 2003	_	2003		2004 nousands)	2	2005	 2006
Other Financial and Operating Data (unaudited):								
Adjusted EBITDA(2)	\$ (13,930)	\$	(7,558)	\$	(221)	\$	730	\$ 9,945

(2) We define Adjusted EBITDA as net income plus the (benefit) provision for income taxes, depreciation, amortization of purchased intangible assets and stock-based compensation; plus interest expense (income) and other income. Adjusted EBITDA is not a measure of liquidity calculated in accordance with GAAP, and should be viewed as a supplement to — not a substitute for — our results of operations presented on the basis of GAAP. Adjusted EBITDA does not purport to represent cash flow provided by, or used in, operating activities as defined by GAAP. Our statement of cash flows presents our cash flow activity in accordance with GAAP. Furthermore, Adjusted EBITDA is not necessarily comparable to similarly-titled measures reported by other companies.

We believe Adjusted EBITDA is useful to an investor in evaluating our operating performance for the following reasons:

- Adjusted EBITDA is widely used by investors to measure a company's operating performance without regard to items such as interest expense, taxes, depreciation and
 amortization, and stock-based compensation, which can vary substantially from company to company depending upon accounting methods and book value of assets, capital
 structure and the method by which assets were acquired; and
- analysts and investors use Adjusted EBITDA as a supplemental measure to evaluate the overall operating performance of companies in our industry.

Our management uses Adjusted EBITDA:

- · as a measure of operating performance, because it removes the impact of items not directly resulting from our core operations;
- for planning purposes, including the preparation of our internal annual operating budget;
- to allocate resources to enhance the financial performance of our business;
- · to evaluate the effectiveness of our operational strategies; and
- · in communications with the board of directors, stockholders, analysts and investors concerning our financial performance.

A reconciliation of Adjusted EBITDA to net income, the most directly comparable GAAP measure, for each of the fiscal periods indicated is as follows:

	ar Ended nuary 31,	Year Ended December 31,							
	 2003	_	2003	In thous	2004 ands)	_	2005	2	2006
Net (loss) income	\$ (20,708)	\$	(14,928)	\$	(3,226)	\$	(4,422)	\$	5,669
(Benefit) provision for income taxes	_		_		_		(182)		50
Amortization	562		772		356		2,437		1,371
Depreciation	5,303		5,832		2,389		2,686		2,888
Stock-based compensation	128		171		14		3		198
Interest expense (income), net	885		595		246		208		(231)
Adjusted EBITDA	\$ (13,930)	\$	(7,558)	\$	(221)	\$	730	\$	9,945

MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion and analysis of our financial condition and results of operations should be read in conjunction with our consolidated financial statements and the related notes to those statements included elsewhere in this prospectus. In addition to historical financial information, the following discussion and analysis contains forward-looking statements that involve risks, uncertainties and assumptions. Our actual results and timing of selected events may differ materially from those anticipated in these forward-looking statements as a result of many factors, including those discussed under "Risk Factors" and elsewhere in this prospectus. See "Cautionary Note Regarding Forward-Looking Statements."

Overview

We provide a leading digital marketing intelligence platform that helps our customers make better-informed business decisions and implement more effective digital business strategies. Our products and solutions offer our customers deep insights into consumer behavior, including objective, detailed information regarding usage of their online properties and those of their competitors, coupled with information on consumer demographic characteristics, attitudes, lifestyles and offline behavior.

Our digital marketing intelligence platform is comprised of proprietary databases and a computational infrastructure that measures, analyzes and reports on digital activity. The foundation of our platform is data collected from our comScore panel of more than two million Internet users worldwide who have granted us explicit permission to confidentially measure their Internet usage patterns, online and certain offline buying behavior and other activities. By applying advanced statistical methodologies to our panel data, we project consumers' online behavior for the total online population and a wide variety of user categories.

We deliver our digital marketing intelligence through our comScore Media Metrix product family and through comScore Marketing Solutions. Media Metrix delivers digital media intelligence by providing an independent, third-party measurement of the size, behavior and characteristics of Web site and online advertising network audiences among home, work and university Internet users as well as insight into the effectiveness of online advertising. Our Marketing Solutions products combine the proprietary information gathered from the comScore panel with the vertical industry expertise of comScore analysts to deliver digital marketing intelligence, including the measurement of online advertising effectiveness, customized for specific industries. We typically deliver our Media Metrix products electronically in the form of weekly, monthly or quarterly reports. Customers can access current and historical Media Metrix data and analyze these data anytime online. Our Marketing Solutions products are typically delivered on a monthly, quarterly or ad hoc basis through electronic reports and analyses.

Our company was founded in August 1999. By 2000, we had established a panel of Internet users and began delivering digital marketing intelligence products that measured online browsing and buying behavior to our first customers. We also introduced netScore, our initial syndicated Internet audience measurement product. We accelerated our introduction of new products in 2003 with the launch of Plan Metrix (formerly AiM 2.0), qSearch, the Campaign R/F (Reach and Frequency) analysis system and product offerings that measure online activity at the local market level. By 2004, we had built a global panel of over two million Internet users. In that year, in cooperation with Arbitron, we launched a service that provides ratings of online radio audiences. In 2005, we expanded our presence in Europe by opening an office in London. In 2006, we continued to expand our measurement capabilities with the launch of World Metrix, a product that provides worldwide data on digital media usage, and Video Metrix, our product that measures the audience for streaming online video.

We have complemented our internal development initiatives with select acquisitions. On June 6, 2002, we acquired certain Media Metrix assets from Jupiter Media Metrix, Inc. Through this acquisition, we acquired certain Internet audience measurement services that report details of Web site usage and visitor demographics. On July 28, 2004, we acquired the outstanding stock of Denaro and Associates, Inc, otherwise known as Q2 Brand Intelligence, Inc. or Q2, to improve our ability to provide our customers more robust survey research integrated with our underlying digital marketing intelligence platform. The total cost of the

acquisition was approximately \$3.3 million, consisting of cash and shares of our common stock. For the ninety-day period beginning July 28, 2007, the former shareholder of Q2 (or its transferees) has the right to sell 1,060,000 shares of our common stock back to us for an aggregate price of \$2.65 million, or \$2.50 per share. On January 4, 2005, we acquired the assets and assumed certain liabilities of SurveySite Inc., or SurveySite. Through this acquisition, we acquired proprietary Internet-based data-collection technologies and increased our customer penetration and revenues in the survey business. The total cost of the acquisition was approximately \$3.6 million, consisting of cash and shares of our common stock. For the ninety-day period beginning January 1, 2008, the former shareholders of SurveySite (or their transferees) have the right to sell 678,172 shares of our common stock back to us for an aggregate price of approximately \$1.8 million, or \$2.67 per share.

Our total revenues have grown from \$15.4 million during the fiscal year ending January 31, 2003 to \$66.3 million during the fiscal year ended December 31, 2006, a compounded annual growth rate of approximately 63%. By comparison, our total expenses from operations have grown from \$35.2 million to \$60.7 million over the same period, a compounded annual growth rate of approximately 20%. The growth in our revenues was primarily the result of:

- increased sales to existing customers, as a result of our efforts to deepen our relationships with these clients by increasing their awareness of, and confidence in, the value of our digital marketing intelligence platform;
- growth in our customer base through the addition of new customers;
- increases in the prices of our products and services;
- · the sales of new products to existing and new customers; and
- · growth in sales outside of the U.S. as a result of entering into new international markets.

As of December 31, 2006, we had 706 customers, compared to 334 as of January 31, 2003. We sell most of our products through our direct sales force.

Our Revenues

We derive our revenues primarily from the fees that we charge for subscription-based products and customized projects. We define subscription-based revenues as revenues that we generate from products that we deliver to a customer on a recurring basis. We define project revenues as revenues that we generate from customized projects that are performed for a specific customer on a non-recurring basis. We market our subscription-based products, customized projects and survey services within the comScore Media Metrix product family and through comScore Marketing Solutions.

A significant characteristic of our business model is our large percentage of subscription-based contracts. Subscription-based revenues accounted for 78% of our total revenues in 2004 and decreased to 70% of total revenues in 2005 primarily due to the acquisition of SurveySite. Subscription-based revenues increased to 75% of total revenues in 2006.

Many of our customers who initially purchased a customized project have subsequently purchased one of our subscription-based products. Similarly, many of our subscription-based customers have subsequently purchased additional customized projects.

Historically, we have generated most of our revenues from the sale and delivery of our products to companies and organizations located within the United States. We intend to expand our international revenues by selling our products and deploying our direct sales force model in additional international markets in the future. For the fiscal year ended December 31, 2006, our international revenues were \$5.7 million, an increase of \$2.4 million over international revenues of \$3.4 million for the fiscal year ended December 31, 2005. International revenues comprised approximately 9% of our total revenues in 2006 as compared to 7% of total revenues in 2005.

We anticipate that revenues from our U.S. customers will continue to constitute the substantial majority of our revenues, but we expect that revenues from customers outside of the U.S. will increase as a percentage of total revenues as we build greater international recognition of our brand and expand our sales operations globally.

Subscription Revenue

We generate a significant proportion of our subscription-based revenues from our Media Metrix product family. Products within the Media Metrix family include Media Metrix 2.0, Plan Metrix, World Metrix and Video Metrix. We intend to commercially launch Ad Metrix in the second quarter of 2007. These product offerings provide subscribers with intelligence on digital media usage, audience characteristics, audience demographics and online and offline purchasing behavior. Customers who subscribe to our Media Metrix products are provided with login IDs to our Web site, have access to our database and can generate reports at anytime.

We also generate subscription-based revenues from certain reports and analyses provided through comScore Marketing Solutions, if that work is procured by customers for at least a nine month period and the customer enters into an agreement to continue or extend the work. Through our Marketing Solutions products, we deliver digital marketing intelligence relating to specific industries, such as automotive, consumer packaged goods, entertainment, financial services, media, pharmaceutical, retail, technology, telecommunications and travel. This marketing intelligence leverages our global consumer panel and extensive database to deliver information unique to a particular customer's needs on a recurring schedule, as well as on a continual-access basis. Our Marketing Solutions customer agreements typically include a fixed fee with an initial term of at least one year. We also provide these products on a non-subscription basis as described under "Project Revenues" below.

In addition, we generate subscription-based revenues from survey products that we sell to our customers. In conducting our surveys, we generally use our global Internet user panel. After questionnaires are distributed to the panel members and completed, we compile their responses and then deliver our findings to the customer, who also has ongoing access to the survey response data as they are compiled and updated over time. These data include responses and information collected from the actual survey questionnaire and can also include behavioral information that we passively collect from our panelists. If a customer contractually commits to having a survey conducted on a recurring basis, we classify the revenues generated from survey products as subscription-based revenues. Approximately half of the revenues derived from survey products are generated on a subscription basis. Our contracts for survey services typically include fixed fee agreements that range from two months to one year.

Project Revenues

We generate project revenues by providing customized information reports to our customers on a non-recurring basis as part of our comScore Marketing Solutions. For example, a customer in the media industry might request a custom report that profiles the behavior of the customer's active online users and contrasts their market share and loyalty with similar metrics for a competitor's online user base. If this customer continues to request the report beyond an initial project term of at least nine months and enters into an agreement to purchase the report on a recurring basis, we begin to classify these future revenues as subscription-based.

In the second quarter of 2007, we intend to commercially launch Campaign Metrix, a product that will provide detailed information about online advertising campaigns. Project revenues from Campaign Metrix will be generated when a customer accesses or downloads a report through our Web site. Pricing for our Campaign Metrix product will initially be based on the scope of the information provided in the report generated by the customer.

Critical Accounting Policies and Estimates

Our discussion and analysis of our financial condition and results of operations are based on our consolidated financial statements, which have been prepared in accordance with accounting principles generally accepted in the U.S. The preparation of these financial statements requires us to make estimates, assumptions and judgments that affect the amounts reported in our financial statements and the accompanying notes. We base our estimates on historical experience and on various other assumptions that we believe to be reasonable under the circumstances. Actual results may differ from these estimates. While our significant accounting policies are described in more detail in the notes to our consolidated financial statements included in this prospectus, we believe the following accounting policies to be the most critical to the judgments and estimates used in the preparation of our consolidated financial statements

Revenue Recognition

We recognize revenues in accordance with Securities and Exchange Commission Staff Accounting Bulletin No. 104, Revenue Recognition (SAB 104). SAB 104 requires that four basic criteria must be met prior to revenue recognition: (i) persuasive evidence of an arrangement exists, (ii) delivery has occurred or the services have been rendered, (iii) the fee is fixed and determinable, and (iv) collection of the resulting receivable is reasonably assured. Certain of our arrangements contain multiple elements, consisting of the various services we offer. We account for these arrangements in accordance with Emerging Issues Task Force Issue No. 00-21, Revenue Arrangements with Multiple Deliverables. We have determined that we do not have objective and reliable evidence of fair value for any of our services. We therefore account for all elements in our multiple element service arrangements as a single unit of accounting and recognize the total value of the arrangement on a straight-line basis over the longest contract term.

We generate revenues by providing access to our online database or delivering information obtained from our database, usually in the form of periodic reports. Revenues are typically recognized on a straight-line basis over the period in which access to data or reports are provided, which generally ranges from three to 24 months.

We also generate revenues through survey services under contracts ranging in term from two months to one year. Our survey services consist of survey and questionnaire design with subsequent data collection, analysis and reporting. We recognize revenues on a straight-line basis over the estimated data collection period once the survey or questionnaire design has been delivered. Any change in the estimated data collection period results in an adjustment to revenues recognized in future periods.

Generally, our contracts are non-cancelable. A limited number of customers, however, have the right to cancel their contracts by providing us with written notice of cancellation. In the event that a customer cancels its contract, it is not entitled to a refund for prior services, and it will be charged for costs incurred plus services performed up to the cancellation date.

Goodwill and Intangible Assets

We record goodwill and intangible assets when we acquire other businesses. The allocation of acquisition costs to intangible assets and goodwill involves the extensive use of management's estimates and assumptions, and the result of the allocation process can have a significant impact on our future operating results. Under Statement of Financial Accounting Standards (SFAS) No. 142, Goodwill and Other Intangible Assets (SFAS 142), intangible assets with finite lives are amortized over their useful lives while goodwill and indefinite lived assets are not amortized, but rather are periodically tested for impairment. An impairment review generally requires developing assumptions and projections regarding our operating performance. In accordance with SFAS 142, we have determined that all of our goodwill is associated with one reporting unit as we do not operate separate lines of business with respect to our services. Accordingly, on an annual basis we perform the impairment assessment for goodwill required under SFAS 142 at the enterprise level by comparing the fair value of a reporting unit, based on estimated future cash flow, to its carrying value including goodwill recorded by the reporting unit. If the carrying value exceeds the fair value, impairment is measured by comparing the derived fair value of the goodwill to its carrying value and any impairment

determined is recorded in the current period. If our estimates or the related assumptions change in the future, we may be required to record impairment charges to reduce the carrying value of these assets, which could be material.

Long-lived assets

Our long-lived assets primarily consist of property and equipment and intangible assets. In accordance with SFAS No. 144, Accounting for the Impairment or Disposal of Long-Lived Assets, we evaluate the recoverability of our long-lived assets for impairment whenever events or changes in circumstances indicate the carrying value of such assets may not be recoverable. If an indication of impairment is present, we compare the estimated undiscounted future cash flows to be generated by the asset to its carrying amount. If the undiscounted future cash flows are less than the carrying amount of the asset, we record an impairment loss equal to the excess of the asset's carrying amount over its fair value. The fair value is determined based on valuation techniques such as a comparison to fair values of similar assets or using a discounted cash flow analysis. Substantially all of our long-lived assets are located in the United States. Although we believe that the carrying values of our long-lived assets are appropriately stated, changes in strategy or market conditions or significant technological developments could significantly impact these judgments and require adjustments to recorded asset balances. There were no impairment charges recognized during the years ended December 31, 2004, 2005, or 2006.

Allowance for Doubtful Accounts

We manage credit risk on accounts receivable by performing credit evaluations of our customers on a selective basis, by reviewing our accounts and contracts and by providing appropriate allowances for uncollectible amounts. Allowances are based on management's judgment, which considers historical experience and specific knowledge of accounts that may not be collectible. We make provisions based on our historical bad debt experience, a specific review of all significant outstanding invoices and an assessment of general economic conditions. If the financial condition of a customer deteriorates, resulting in an impairment of its ability to make payments, additional allowances may be required.

Income Taxes

We account for income taxes using the liability method in accordance with SFAS No. 109, Accounting for Income Taxes. We estimate our tax liability through calculations we perform for the determination of our current tax liability, together with assessing temporary differences resulting from the different treatment of items for income tax and financial reporting purposes. These differences result in deferred tax assets and liabilities, which are recorded on our balance sheet. Management then assesses the likelihood that deferred tax assets will be recovered in future periods. In assessing the need for a valuation allowance against the net deferred tax asset, we consider factors such as future reversals of existing taxable temporary differences, taxable income in prior carryback years, if carryback is permitted under the tax law, tax planning strategies and future taxable income exclusive of reversing temporary differences and carryforwards. To the extent that we cannot conclude that it is more likely than not that the benefit of such assets will be realized, we establish a valuation allowance to adjust the net carrying value of such assets.

To date, we have recorded a valuation allowance against our deferred tax assets, principally net operating loss carryforwards, due to uncertainty regarding our ability to generate future taxable income. Any current income tax benefit or provision to date has been offset by changes in the valuation allowance against our deferred tax assets. To the extent we determine that all or a portion of our valuation allowance is no longer necessary, we will recognize an income tax benefit in the period such determination is made for the reversal of the valuation allowance. Once the valuation allowance is eliminated, its reversal will no longer be available to offset our current tax provision. These events could have a material impact on our reported results of operations.

As of December 31, 2006, we had \$81.2 million of both federal and state net operating loss carryforwards each of which begin to expire in 2020 for federal and began to expire in 2006 for state income tax reporting

purposes. In addition, we had net operating loss carryforwards related to our foreign subsidiaries totalling \$966,000 as of December 31, 2005 and \$703,000 as of December 31, 2006, which begin to expire in 2010. Under Section 382 of the Internal Revenue Code, the utilization of net operating loss carryforwards is limited based on changes in percentage of our ownership. As a result of prior ownership changes, we believe that we will be limited in our use of net operating loss carryforwards to offset taxable income.

Stock-Based Compensation

Through December 31, 2005, as permitted by SFAS No. 123, Accounting for Stock-Based Compensation (SFAS 123), we applied the intrinsic value method for accounting for stock-based compensation as set forth in Accounting Principles Board Opinion No. 25, Accounting for Stock Issued to Employees (APB 25). For purposes of the pro forma disclosures required under SFAS 123, we used the minimum-value method to estimate the fair value of our stock-based awards. On January 1, 2006, we adopted SFAS No. 123R, Share-Based Compensation (SFAS 123R). Under SFAS 123R, a non-public company that previously used the minimum value method for pro forma disclosure purposes is required to adopt the standard using the prospective method. Under the prospective method, all awards granted, modified or settled after the date of adoption are accounted for using the measurement, recognition and attribution provisions of SFAS 123R. As a result, stock-based awards granted prior to the date of adoption of SFAS 123R will continue to be accounted for under APB 25 with no recognition of stock-based compensation in future periods, unless such awards are modified or settled.

Subsequent to the adoption of SFAS 123R, we estimate the fair value of our stock-based awards on the date of grant using the Black-Scholes option-pricing model. The determination of fair value using the Black-Scholes model requires a number of complex and subjective variables. One key input into the model is the estimated fair value of our common stock on the date of grant. Our board of directors has estimated the fair value of our common stock for the purpose of establishing exercise prices for our stock option grants. Our board has relied upon market place transaction history as well as the assistance of independent valuation specialists for purposes of estimating the fair value of our common stock.

Other key variables in the Black-Scholes option-pricing model include the expected volatility of our common stock price, the expected term of the award and the risk-free interest rate. In addition, under SFAS 123R, we are required to estimate forfeitures of unvested awards when recognizing compensation expense. If factors change and we employ different assumptions in the application of SFAS 123R in future periods, the compensation expense we record may differ significantly from what we have recorded during 2006.

At December 31, 2006, total estimated unrecognized compensation expense related to unvested stock-based awards granted prior to that date was \$1.3 million, which is expected to be recognized over a weighted-average period of 1.86 years.

We expect stock-based compensation expense to increase in absolute dollars as a result of the adoption of SFAS 123R as options that were granted at the beginning of 2006 and beyond vest. Beginning in 2007, we expect to make use of restricted stock awards and reduce our use of stock options as a form of stock-based compensation. The actual amount of stock-based compensation expense we record in any fiscal period will depend on a number of factors, including the number of shares subject to the stock options issued, the fair value of our common stock at the time of issuance and the expected volatility of our stock price over time.

Estimation of Fair Value of Warrants to Purchase Redeemable Convertible Preferred Stock

On July 1, 2005, we adopted FASB Staff Position 150-5 (FSP 150-5). Our outstanding warrants to purchase shares of our redeemable convertible preferred stock are subject to the requirements in FSP 150-5, which require us to classify these warrants as current liabilities and to adjust the value of these warrants to their fair value at the end of each reporting period. At the time of adoption, we recorded \$440,000 for the cumulative effect of this change in accounting principle to reflect the cumulative change in estimated fair value of these warrants as of that date. We recorded \$14,000 for the year ended December 31, 2005 and \$224,000 for the year ended December 31, 2006, to reflect further increases in the estimated fair value of the

warrants. We estimated the fair value of these warrants at the respective dates using the Black-Scholes option valuation model, based on the estimated market value of the underlying redeemable convertible preferred stock at the valuation measurement date, the contractual term of the warrant, risk-free interest rates and expected dividends on and expected volatility of the price of the underlying redeemable convertible preferred stock. These estimates, especially the market value of the underlying redeemable convertible preferred stock and the expected volatility, are highly judgmental and could differ materially in the future.

Upon the closing of this offering, all outstanding warrants to purchase shares of our preferred stock will become warrants to purchase shares of our common stock and, as a result, will no longer be subject to FSP 150-5. The then-current aggregate fair value of these warrants will be reclassified from liabilities to additional paid-in capital, a component of stockholder's equity, and we will cease to record any related periodic fair value adjustments. We anticipate that we will incur a non-cash charge relating to our outstanding warrants for preferred stock in the period in which this offering closes. Assuming that the price at which our common stock is valued for these purposes is the initial public offering price of \$ per share, the amount of that charge would be approximately \$. The exact amount of the charge may depend on the closing trading price of our common stock on The NASDAQ Global Market on , the expected date of the closing of this offering.

Seasonality

Historically, a slightly higher percentage of our customers have renewed their subscription products with us toward the end of the fourth quarter. While we execute projects for our customers throughout the year, we have historically experienced a slight upturn in our project-based business in the fourth quarter.

Results of Operations

The following table sets forth selected consolidated statements of operations data as a percentage of total revenues for each of the periods indicated.

		Ended December 31	
	2004	2005	2006
Revenues	100.0%	100.0%	100.0%
Cost of revenues	37.7	36.2	31.0
Selling and marketing	39.8	37.7	32.4
Research and development	15.7	14.8	13.6
General and administrative	14.3	14.1	12.5
Amortization	1.0	4.8	2.1
Total expenses from operations	108.5	107.7	91.6
(Loss) income from operations	(8.5)	(7.7)	8.4
Interest (expense) income, net	(0.7)	(0.4)	0.3
(Loss) gain from foreign currency	_	(0.2)	0.2
Revaluation of preferred stock warrant liabilities			(0.3)
(Loss) income before income taxes and cumulative effect of change in accounting principle	(9.2)	(8.3)	8.6
(Benefit) provision for income taxes		(0.4)	0.1
Net (loss) income before cumulative effect of change in accounting principle	(9.2)	(7.9)	8.6
Cumulative effect of change in accounting principle	_	(0.9)	_
Net (loss) income	(9.2)	(8.8)	8.6
Accretion of redeemable preferred stock	(6.1)	(5.2)	(4.8)
Net (loss) income attributable to common stockholders	(15.4)%	(14.0)%	3.8%

Years Ended December 31, 2004, 2005 and 2006

Revenues

		Year Ende	d December 31	,			Inc	rease			Percent	Change	
	 2004		2005		2006		2004 v. 2005		2005 v. 2006	_	2004 v. 2005	2005 v. 2006	-
	 				(D	ollars in	thousands)			_			_
Total revenues	\$ 34.894	\$	50.267	\$	66.293	\$	15.373	\$	16.026		44.1%	31.9	9%

Total revenues increased for the year ended December 31, 2006 as compared to the year ended December 31, 2005. This increase was largely driven by a combination of increased business with existing customers, continued growth in the number of customers for both our subscription and our project-based products and, to a lesser extent, general increases in our price levels from 2005 to 2006. Our customer base grew during this period from 565 as of December 31, 2005 to 706 as of December 31, 2006.

In 2005, total revenues increased from 2004 primarily due to the growth in our subscription revenues driven by deepening relationships with existing customers, the expansion of our customer base, revenues attributable to the acquisition of SurveySite and Q2, the introduction of new products, international expansion and, to a lesser extent, general increases in our price levels in 2005 from 2004. In addition, we were successful in growing our project revenues with existing and new customers during this period. Our customer base grew during this period from 469 as of December 31, 2004 to 565 as of December 31, 2005. Our 2005 revenues were positively impacted by the acquisitions of SurveySite and Q2. SurveySite, which we acquired on January 4, 2005, contributed \$5.1 million in revenues in 2005. Q2 contributed \$3.6 million in revenues in 2005 as compared to \$1.5 million in revenues in 2004, following our acquisition of Q2 on July 28, 2004.

Revenues from customers outside of the U.S. totaled \$3.4 million, or approximately 7% of total revenues, in 2005 and totaled approximately \$5.7 million, or approximately 9% of total revenues, in 2006, representing an increase of \$2.4 million. This increase was primarily due to our ongoing efforts to expand our international presence, which included the opening of a European office in London in the first half of 2005.

Cost of Revenues

		tear Ended December 51,		IIICI	casc	1 crcciii C	nange
				2004 v.	2005 v.	2004 v.	2005 v.
	2004	2005	2006 (Dollars	2005 s in thousands)	2006	2005	2006
			(Donar	, in thousands,			
Cost of revenues	\$ 13,153	\$ 18,218	\$ 20,560	\$ 5,065	\$ 2,342	38.5%	12.9%
As a percentage of revenues	37 7%	36.2%	31.0%				

Cost of revenues consists primarily of expenses related to operating our network infrastructure and the recruitment, maintenance and support of our consumer panels. Expenses associated with these areas include the salaries and related expenses of network operations, survey operations, custom analytics and technical support, all of which are expensed as they are incurred. Cost of revenues also includes data collection costs for our products and operational costs associated with our data centers, including depreciation expense associated with computer equipment.

Cost of revenues increased in 2006 as compared to 2005, primarily due to increased costs associated with supporting our consumer panel and data centers. Cost of revenues declined as a percentage of revenues over the same period primarily due to an increase in revenues and a moderation of the increases in costs to build and maintain our panel. The decline in cost of revenues as a percentage of revenues was offset in part by increases in bandwidth and data costs, which grew 9%. The headcount and costs associated with our technology staff grew at a lower rate than our growth in revenues.

Cost of revenues increased in 2005 as compared to 2004 primarily due to our acquisition of SurveySite and higher costs associated with data center operations and employee salaries, benefits and related costs required to support growth in our revenues and customer base during 2005. The cost of revenues as a percentage of revenues declined in 2005 compared to 2004 due to relatively flat panel costs and smaller

increases in bandwidth and data center costs, which did not grow at the same rate as our customer base and revenues. The headcount and costs associated with our technology staff grew at a lower rate than our growth in revenues.

We expect cost of revenues to increase in absolute dollar amounts as we seek to grow our business but vary as a percentage of revenues depending on whether we benefit from investments in our panel and network infrastructure.

Selling and Marketing Expenses

	Y	ear Ended December 31,		Inci	rease	Percent Change	
	2004	2005	2006	2004 v. 2005	2005 v. 2006	2004 v. 2005	2005 v. 2006
			(Dollars	in thousands)			
Selling and marketing expenses	\$ 13,890	\$ 18,953	\$ 21,473	\$ 5,063	\$ 2,520	36.5%	13.3%
As a percentage of revenues	39.8%	37.7%	32.4%				

Selling and marketing expenses consist primarily of salaries, benefits, commissions and bonuses paid to our direct sales force and industry analysts, as well as costs related to online and offline advertising, product management, industry conferences, promotional materials, public relations, other sales and marketing programs, and allocated overhead, including rent and depreciation. All selling and marketing costs are expensed as they are incurred. Commission plans are developed for our account managers with criteria and size of sales quotas that vary depending upon the individual's role. Commissions are paid to a salesperson and are expensed as selling and marketing costs when a sales contract is executed by both the customer and commScore. In the case of multi-year agreements, one year of commissions is paid initially, with the remaining amounts paid at the beginning of the succeeding years.

Selling and marketing expenses increased in 2006 as compared to 2005 in absolute dollars, primarily due to increased employee salaries and benefits and related costs resulting from additional account management personnel in our sales force, plus an increase in commission costs associated with increased revenues. Our selling and marketing headcount increased from 143 employees as of December 31, 2005 to 155 employees as of December 31, 2006. In addition, the expansion of our European office in London and increased marketing efforts in Europe contributed to our increase in selling and marketing expenses and headcount in 2006. The decrease in selling and marketing expenses as a percentage of revenues during this period reflects the increased productivity of our direct sales force and an increase in revenues.

Selling and marketing expenses increased in 2005 as compared to 2004, primarily due to an increase in the number of account managers, higher commissions associated with our growth in revenues and an increase in online and offline advertising and promotional efforts in support of building our brands. In addition, our selling and marketing headcount increased from 77 employees as of December 31, 2004 to 143 employees as of December 31, 2005. The acquisition of SurveySite and the opening of our first European office in London also contributed to our increase in selling and marketing expenses and headcount in 2005. The decrease in selling and marketing expenses as a percentage of revenues during this period reflected the increased productivity of our direct sales force.

We expect selling and marketing expenses to increase in absolute dollar amounts as we continue to grow our selling and marketing efforts but to vary in future periods as a percentage of revenues depending on whether we benefit from increased productivity in our sales force and from increased revenues resulting in part from our ongoing marketing initiatives.

Research and Development Expenses

	Yea	Year Ended December 31,			ease	Percent Change		
	2004	2005		2004 v. 2005 ollars in thousands)	2005 v. 2006	2004 v. 2005	2005 v. 2006	
Research and development expenses	\$ 5,493	\$ 7,416	\$ 9,009	\$ 1,923	\$ 1,593	35.0%	21.5%	
As a percentage of revenues	15.7%	14.8%	13.6%					

Research and development expenses include new product development costs, consisting primarily of compensation and related costs for personnel associated with research and development activities, and allocated overhead, including rent and depreciation.

Research and development expenses increased in 2006 as compared to 2005 primarily due to increased headcount and our continued focus on developing new products, such as World Metrix, Video Metrix, Campaign Metrix and Ad Metrix. Research and development costs decreased slightly as a percentage of revenues, primarily due to our growth in revenues.

The increase in research and development expenses in 2005 compared to 2004 was due to new product development activity, including the launch of a streaming media audience measurement product. The acquisition and integration of SurveySite's operations also contributed to the absolute dollar increase in research and development costs during this period.

We expect research and development expenses to increase in absolute dollar amounts as we continue to enhance and expand our product offerings. As a result of the size and diversity of our panel and our historical investment in our technology infrastructure, we expect that we will be able to develop new products with moderate increases in research and development spending as compared to our growth in revenues. We also expect research and development expenses to moderate due to our decision to outsource certain software development activities in 2005.

General and Administrative Expenses

	Ye.	ar Ended December 31		Inc	rease	Percent Change		
	2004	2005	2006	2004 v. 2005 (Dollars in thousands)	2005 v. 2006	2004 v. 2005	2005 v. 2006	
General and administrative expenses	\$ 4,982	\$ 7,089	\$ 8,293	\$ 2,107	\$ 1,204	42.3%	17.0%	
As a percentage of revenues	14.3%	14.1%	12.5%)				

General and administrative expenses consist primarily of salaries and related expenses for executive management, finance, accounting, human capital, legal, information technology and other administrative functions, as well as professional fees, overhead, including allocated rent and depreciation, and expenses incurred for other general corporate purposes.

General and administrative expenses increased in 2006 as compared to 2005, primarily due to increased professional fees and expanding our finance department. As a percentage of revenues, general and administrative expenses decreased in 2006 as compared to 2005, due primarily to our growth in revenues.

General and administrative expenses increased in 2005 as compared to 2004, primarily due to higher salaries, benefits and related costs associated with our existing employees plus an increase in our general and administrative headcount from 14 employees as of December 31, 2004 to 27 employees as of December 31, 2005. The higher headcount was due primarily to an increase in employees in such functions as finance, accounting, human capital and legal, as we built our staff and infrastructure to support our growth. Our acquisition of SurveySite also contributed to the increase in general and administrative expenses and related headcount in 2005. On a percentage of revenues basis, general and administrative expenses were flat in 2005 as compared to 2004, as the increase in headcount related to broadening our administrative support capabilities and the acquisition of SurveySite was offset by the growth in our customer base and revenues.

We expect general and administrative expenses to increase on an absolute basis in future annual periods as we incur increased costs associated with being a public company. Operating as a public company will present additional management and reporting requirements that will significantly increase our directors' and officers' liability insurance premiums and professional fees both in absolute dollars and as a percentage of revenues. We also anticipate hiring additional personnel to help manage future growth and our operations as a public company.

Amortization Expense

	,	Year Ended December	31,	Increa	ise	Percent Change					
	2004	2005	2006	2004 v. 2005 (Dollars in thousands)	2005 v. 2006	2004 v. 2005	2005 v. 2006				
Amortization expense	\$ 356	\$ 2,437	\$ 1,371	\$ 2,081	\$ (1,066)	584.6%	(43.7)%				
As a percentage of revenues	1.0%	4.8%	2.1%								

Amortization expense consists of charges related to the amortization of intangible assets associated with past acquisitions.

Amortization expense decreased during fiscal year 2006 over 2005 because certain intangible assets related to previous acquisitions were fully amortized as of that period.

The increase in amortization expense from 2004 to 2005 in absolute dollars is attributable primarily to the amortization expense relating to the Q2 acquisition on July 28, 2004 and the SurveySite acquisition on January 4, 2005.

Absent additional acquisitions, we expect amortization expense to continue to decline as the remaining amount of intangible assets related to previous acquisitions is amortized.

Interest (Expense) Income, Net

Interest income consists primarily of interest earned from short-term investments, such as auction rate securities, and our cash and cash equivalent balances. Interest expense is incurred due to capital leases pursuant to several equipment loan and security agreements and a line of credit that we have entered into in order to finance the lease of various hardware and other equipment purchases. Our capital lease obligations are secured by a senior security interest in eligible equipment.

Interest (expense) income, net was \$(246,000) in 2004, \$(208,000) in 2005 and \$231,000 in 2006. The year-to-year change from 2004 to 2005 and from 2005 to 2006 primarily reflects the net effect of interest income that we earned on our cash balances offset by the interest expense associated with the capital leases that we had in place in each year. Our net interest expense decreased from 2004 to 2005 due to our larger cash and investments balances and the lower amounts outstanding under our capital leases. We reported net interest income in 2006 due to a \$6.9 million increase in our cash and investments balance. We also continued to reduce the outstanding balance on our outstanding capital lease obligations.

Gain/Loss on Foreign Currency Transactions

Our gains and losses on foreign currency transactions arise from our Canadian and United Kingdom foreign subsidiaries that hold cash and receivables in currencies other than its functional currency. Our loss on foreign currency transactions in 2005 was \$96,000. We recorded a gain of \$125,000 in 2006 as a result of fluctuations in the exchange rate between the U.S. dollar and the Canadian dollar, Euro and British Pound.

Provision for Income Taxes

As of December 31, 2006, we had net operating loss carryforwards for federal income tax purposes in the amount of approximately \$81.2 million, which begin to expire in 2020 for federal and began to expire in 2006 for state income tax reporting purposes. In the future, we intend to utilize any carryforwards available to us to reduce our tax payments. These carryforwards may be subject to annual limitations based on changes in

percentage of our ownership as limited under Section 382 of the Internal Revenue Code. In 2005, we had an income tax benefit of \$182,000 related to a deferred tax liability of \$356,000 associated with a temporary difference related to certain acquired intangible assets of SurveySite. This compares to an income tax expense of \$50,000 in 2006 reflecting a payment of alternative minimum tax (AMT) partly offset by a decrease in the deferred tax liability.

Quarterly Results of Operations

The following tables set forth selected unaudited quarterly consolidated statement of operations data for each of the quarters indicated. The consolidated financial statements for each of these quarters have been prepared on the same basis as the audited consolidated financial statements included in this prospectus and, in the opinion of management, include all adjustments necessary for the fair presentation of the consolidated results of operations for these periods. You should read this information together with our consolidated financial statements and related notes included elsewhere in this prospectus. These quarterly operating results are not necessarily indicative of the results for any future period.

	Three Months Ended															
		Mar. 31, 2005		Jun. 30, 2005		Sept. 30, 2005		Dec. 31, 2005		Mar. 31, 2006		Jun. 30, 2006		Sept. 30, 2006		Dec. 31, 2006
							((In thousands)	(Unat	idited)	_		-		_	
Revenues	\$	11,135	\$	13,150	\$	12,953	\$	13,029	\$	14,985	\$	16,906	\$	16,165	\$	18,237
Cost of revenues(1)		3,936		4,863		4,602		4,817		5,148		5,205		4,977		5,230
Selling and marketing(1)		4,234		4,813		4,821		5,085		5,345		5,323		5,171		5,634
Research and development(1)		1,678		1,876		1,908		1,954		2,137		2,258		2,273		2,341
General and administrative(1)		1,489		1,804		1,779		2,017		1,918		2,176		1,897		2,302
Amortization		621		603		612		601		371		333		333		334
Total expenses from operations		11,958		13,959		13,722		14,474		14,919		15,295		14,651		15,841
(Loss) income from operations		(823)		(809)		(769)		(1,445)		66		1,611		1,514		2,396
Interest (expense) income, net		(58)		(71)		(39)		(40)		11		23		84		113
(Loss) gain from foreign currency		(21)		(1)		(72)		(2)		6		(33)		3		149
Revaluation of preferred stock warrant liabilities		_		_		(6)		(8)		2		(211)		(6)		(9)
(Loss) income before income taxes and cumulative effect of																
change in accounting principle		(902)		(881)		(886)		(1,495)		85		1,390		1,595		2,649
(Benefit) provision for income taxes		(53)		(52)		(38)		(39)								50
Net (loss) income before cumulative effect of change in							-									
accounting principle		(849)		(829)		(848)		(1,456)		85		1,390		1,595		2,599
Cumulative effect of change in accounting principle		_		_		(440)		_		_		_		_		_
Net (loss) income		(849)		(829)		(1,288)		(1,456)		85		1,390		1,595		2,599
Accretion of redeemable preferred stock		(611)		(643)		(675)		(709)		(742)		(777)		(812)		(848)
Net (loss) income attributable to common stockholders	\$	(1,460)	\$	(1,472)	\$	(1,963)	\$	(2,165)	\$	(657)	\$	613	\$	783	\$	1,751

(1) Amortization of stock-based compensation is included in the line items above as follows:

							Three Mont	hs Ende	d			
			Jun. 30, 2005		Sept. 30, 2005		ec. 31, 2005 1 thousands)	2	ar. 31, 2006 ted)	n. 30, 006	ept. 30, 2006	ec. 31, 2006
Cost of revenues	\$ _	\$	_	\$	_	\$	_	\$	_	\$ 2	\$ 4	\$ 6
Selling and marketing	_		_		_		_		6	26	23	27
Research and development	_		_		_		_		_	2	4	7
General and administrative	_		1		1		1		1	10	40	40
Total	\$ 	\$	1	\$	1	\$	1	\$	7	\$ 40	\$ 71	\$ 80

	As a Percentage of Total Revenues Three Months Ended													
	Mar. 31, 2005	Jun. 30, 2005	Sept. 30, 2005	Dec. 31, 2005 (Unaudi	Mar. 31, 2006	Jun. 30, 2006	Sept. 30, 2006	Dec. 31, 2006						
Revenues	100.0%	100.0%	100.0%	100.0%	100.0%	100.0%	100.0%	100.0%						
Cost of revenues	35.3	37.0	35.5	37.0	34.4	30.8	30.8	28.7						
Selling and marketing	38.0	36.6	37.2	39.0	35.7	31.5	32.0	30.9						
Research and development	15.1	14.3	14.7	15.0	14.3	13.4	14.1	12.9						
General and administrative	13.4	13.7	13.7	15.5	12.8	12.9	11.7	12.6						
Amortization	5.6	4.6	4.7	4.6	2.5	2.0	2.1	1.8						
Total expenses from operations	107.4	106.2	105.8	111.1	99.6	90.5	90.6	86.9						
(Loss) income from operations	(7.4)	(6.2)	(5.8)	(11.1)	0.4	9.5	9.4	13.1						
Interest (expense) income, net	(0.5)	(0.5)	(0.3)	(0.3)	0.1	0.1	0.5	0.6						
(Loss) gain from foreign currency	(0.2)	_	(0.6)	_	_	(0.2)	_	8.0						
Revaluation of preferred stock warrant liabilities						(1.2)								
(Loss) income before income taxes and cumulative effect of change														
in accounting principle	(8.1)	(6.7)	(6.8)	(11.4)	0.6	8.2	9.9	14.5						
(Benefit) provision for income taxes	(0.5)	(0.4)	(0.3)	(0.3)	_	_	_	0.3						
Net (loss) income before cumulative effect of change in accounting				<u> </u>										
principle	(7.6)	(6.3)	(6.5)	(11.1)	0.6	8.2	9.9	14.3						
Cumulative effect of change in accounting principle			(3.4)											
Net (loss) income	(7.6)	(6.3)	(9.9)	(11.1)	0.6	8.2	9.9	14.3						
Accretion of redeemable preferred stock	(5.5)	(4.9)	(5.2)	(5.4)	(5.0)	(4.6)	(5.0)	(4.6)						
Net (loss) income attributable to common stockholders	(13.1)	(11.2)	(15.1)	(16.6)	(4.4)	3.6	4.8	9.6						

Over the eight quarters of 2005 and 2006, revenues have generally increased due primarily to increases in subscription revenues from existing customers, growth in our customer base (both domestically and internationally), general increases in pricing for our products and the acquisition of SurveySite. In 2005, revenues increased sequentially from the first quarter to the second quarter before declining slightly in the third quarter and remaining relatively flat in the fourth quarter. Over these quarterly periods, fluctuations in project revenues partially offset the steady growth in subscription revenues and contributed to the relatively flat revenues on a sequential basis from the second through the fourth quarters of 2005. In 2006, revenues increased significantly

on a sequential basis in the first and second quarters before decreasing in the third quarter due to fluctuations in the closing of agreements relating to, and the execution of, projects. Revenues increased significantly in the fourth quarter of 2006 due to increased growth in subscription revenues for existing and new customers. Subscription revenues increased sequentially in each of the quarters presented.

Cost of revenues as a percentage of total revenues held relatively steady in each of the quarters in 2005 before declining in 2006. The decrease in cost of revenues on a percentage basis was due to the growth in revenues relative to the moderation in fixed costs to support our consumer panel, data center and technical infrastructure.

On an absolute basis, total expenses from operations increased significantly in the second quarter of 2005 due primarily to costs associated with the integration of the Q2 and SurveySite acquisitions and certain expenses for external data sources. Total expenses from operations remained relatively flat in the third quarter of 2005 and increased in the fourth quarter of 2005, primarily due to higher sales costs related to the opening of our first European sales office, located in London, and increased general and administrative costs in support of overall business growth. On an absolute basis, total expenses from operations declined slightly in the first quarter of 2006 before increasing in the second quarter of 2006, due to increases in general and administrative expenses associated with the hiring of new finance personnel and increases in professional services fees related to anticipated business expansion. In addition, expenses from operations increased in the second quarter of 2006 due to higher research and development costs tied to the development of several new products. After a decline in the third quarter, expenses from operations increased again in the fourth quarter of 2006, to to increased commissions tied to higher sales growth plus higher salaries, benefits and related costs associated with hiring additional personnel in our operations, technology, sales, research and development and general and administrative organizations to support the growth of our business. The total expenses from operations in 2006 increased at a lower rate than revenues and we were consequently able to better leverage our cost structure.

We became profitable on a net income basis in the first quarter of 2006, and were profitable on a net income basis every quarter in 2006 as our revenues increased significantly during these periods and our costs grew at a lower rate.

Liquidity and Capital Resources

The following table summarizes our cash flows:

	_	2004	(In t	2005 housands)	_	2006
Consolidated Cash Flow Data:						
Net cash provided by operating activities	\$	1,907	\$	4,253	\$	10,905
Net cash used in investing activities		(1,332)		(2,505)		(9,573)
Net cash used in financing activities		(952)		(1,092)		(1,381)
Effect of exchange rate changes on cash		25		(36)		(43)
Net increase (decrease) in cash and equivalents		(352)		620		(92)

For the Year Ended

Since our inception, we have funded our operations and met our capital expenditure requirements primarily with venture capital and private equity funding. In five separate issuances of preferred stock, from Series A on September 27, 1999 to Series E on August 1, 2003, we have raised over \$88 million from a number of institutional investors. The proceeds from all of these issuances have been used for general business purposes, with the exception of the Series E Preferred Stock offering, which was partially used to extinguish a \$1.5 million bank note. Each share of preferred stock is convertible into common stock at the respective conversion ratio for each series of preferred stock at any time, subject to adjustment triggered by changes in our capitalization such as a stock split. Conversion is automatic in the event of a public offering of common stock at a price of at least \$2.50 per share with gross proceeds of at least \$25 million. This conversion is expected to take place upon consummation of this offering.

Our principal uses of cash historically have consisted of payroll and other operating expenses and payments related to the purchase of equipment primarily to support our consumer panel and technical

infrastructure required to support our customer base. Since the beginning of 2004, we have purchased over \$4.6 million in property and equipment, made \$3.9 million in principal payments on capital lease obligations, and spent \$1.9 million as the cash component of consideration paid for acquisitions.

As of December 31, 2006, our principal sources of liquidity consisted of cash, cash equivalents and short-term investments of \$16.0 million.

Operatina Activities

Our cash flows from operating activities are significantly influenced by our investments in personnel and infrastructure to support the anticipated growth in our business, increases in the number of customers using our products and the amount and timing of payments made by these customers.

We generated approximately \$10.9 million of net cash from operating activities during 2006. The significant components of cash flows from operations were net income of \$5.7 million, \$4.3 million in non-cash depreciation and amortization expenses, a \$1.4 million increase in accounts payable and accrued expenses and a \$3.1 million increase in amounts collected from customers in advance of when we recognize revenues as a result of our growing customer base, offset by a \$3.9 million increase in accounts receivable.

We generated \$4.3 million of net cash from operating activities during 2005. The significant components of cash flows from operations were a \$6.4 million increase in amounts collected from customers in advance of when we recognized revenues as a result of our growing customer base, and \$5.1 million in non-cash depreciation and amortization expenses. These items were partially offset by a \$3.5 million net increase in accounts receivable related to our larger customer base, a net loss of \$4.4 million and other uses of cash in operations.

We generated \$1.9 million of net cash from operating activities in 2004. The significant components of cash flows from operations were a \$0.6 million increase in amounts collected from customers in advance of when we recognized revenues as a result of our growing customer base, a \$1.7 million net increase in accounts payable and accrued expenses due to the timing of payments to our vendors when compared to the same period in 2003 and \$2.7 million in non-cash depreciation and amortization expenses. These items were partially offset by a \$0.7 million net increase in accounts receivable due to our larger customer base, a net loss of \$3.2 million and other uses of cash in operations.

Investing Activities

Our primary investing activities have consisted of purchases of computer network equipment to support our Internet user panel and maintenance of our database, furniture and equipment to support our operations, and payments related to the acquisition of several companies. As our customer base continues to expand, we expect purchases of technical infrastructure equipment to grow in absolute dollars. The extent of these investments will be affected by our ability to expand relationships with existing customers, grow our customer base, introduce new digital formats and increase our international presence.

We used \$9.6 million of net cash in investing activities during 2006, a net \$7.0 million of which was used to purchase short-term investments, \$2.3 million of which was used to purchase property and equipment and \$0.3 million of which was used to pay contingent considerations associated with our Q2 and SurveySite acquisitions. We used \$2.5 million of net cash in investing activities during 2005, of which \$1.1 million was used to purchase property and equipment, \$0.9 million was used as part of the acquisition of SurveySite and \$0.3 million was used to pay contingent consideration associated with the Q2 acquisition. In 2004, we used \$1.3 million of net cash in investing activities, \$1.2 million of which was used to purchase property and equipment and \$0.9 million of which was used as part of the consideration for the acquisition of Q2, partially offset by \$0.8 million in net proceeds from the sale of short-term investments.

We expect to achieve greater economies of scale and operating leverage as we expand our customer base and utilize our Internet user panel and technical infrastructure more efficiently. While we anticipate that it will be necessary for us to continue to invest in our Internet user panel, technical infrastructure and technical personnel to support the combination of an increased customer base, new products, international expansion and new digital market intelligence formats, we believe that these investment requirements will be less than

the revenue growth generated by these actions. This should result in a lower rate of growth in our capital expenditures to support our technical infrastructure. In any given period, the timing of our incremental capital expenditure requirements could impact our cost of revenues, both in absolute dollars and as a percentage of revenues.

Financina Activities

Our primary financing activities since 2004 have consisted of financings to fund the acquisition of capital assets. We entered into an equipment lease agreement with GE Capital in 2003 and a line of credit agreement with GE Capital in 2005 to finance the purchase of hardware and other computer equipment to support our business growth. These borrowings were secured by a senior security interest in the equipment acquired under the facility. In December 2006, we entered into an equipment lease agreement with Banc of America Leasing & Capital, LLC to finance the purchase of new hardware and other computer equipment as we continue to expand our technology infrastructure in support of our business growth. This agreement includes a \$5 million line of credit available through December 31, 2007. Through December 31, 2006, we used this credit facility to establish an equipment lease for the amount of approximately \$2.9 million. The base term for this lease is three years and includes a small charge in the event of prepayment.

We used \$1.4 million of net cash in financing activities during 2006. We used \$1.6 million to make payments on our capital lease obligations partially offset by \$241,000 in proceeds from the exercise of our common stock options.

We used \$1.1 million of net cash from financing activities during 2005. We used \$1.2 million to make payments on our capital lease obligations partially offset by \$136,000 in proceeds from the exercise of our common stock options.

In 2004, we used approximately \$1.0 million of cash in financing activities. Substantially all of the use of this cash resulted from payments on our capital lease obligations.

We do not have any special purpose entities, and other than operating leases for office space, described below, we do not engage in off-balance sheet financing arrangements.

Contractual Obligations and Known Future Cash Requirements

Set forth below is information concerning our known contractual obligations as of December 31, 2006 that are fixed and determinable.

	Total	1 Year	(In thousands)	3-5 Years	5 Years
Capital lease obligations	\$ 4,418	\$ 1,986	\$ 2,432	_	_
Operating lease obligations	5,058	2,009	2,063	760	226
Total	\$ 9,476	\$ 3,995	\$ 4,495	\$ 760	\$ 226

Our principal lease commitments consist of obligations under leases for office space and computer and telecommunications equipment. We finance the purchase of some of our computer equipment under a capital lease arrangement over a period of 36 months. Our purchase obligations relate to outstanding orders to purchase computer equipment and are typically small; they do not materially impact our overall liquidity.

We currently have a line of credit for up to \$5.0 million available to us until December 31, 2007. We have used \$2.9 million of such line of credit to establish an equipment lease for the amount of approximately \$2.9 million bearing interest at a rate of 7.75% per annum.

Future Capital Requirements

We believe that our existing cash, cash equivalents, and short-term investments and operating cash flow, will be sufficient to meet our projected operating and capital expenditure requirements for at least the next twelve months. In addition, we expect that the net proceeds from this offering will provide us with the financial flexibility to execute our strategic objectives, including the ability to make acquisitions and strategic investments. Our ability to generate cash, however, is subject to our performance, general economic conditions, industry trends and other factors. To the extent that funds from this offering, combined with existing cash, cash equivalents, short-term investments and operating cash flow are insufficient to fund our future activities and requirements, we may need to raise additional funds through public or private equity or debt financing. If we issue equity securities in order to raise additional funds, substantial dilution to existing stockholders may occur.

For the ninety-day period beginning July 28, 2007, the former shareholder of Q2 has the right to sell its 1,060,000 shares back to us for an aggregate price of \$2.65 million, or \$2.50 per share. For the ninety-day period beginning January 1, 2008, the former shareholders of SurveySite have the right to sell their 678,172 shares back to us for an aggregate price of approximately \$1.8 million, or \$2.67 per share.

Quantitative and Qualitative Disclosures about Market Risk

Market risk represents the risk of loss that may impact our financial position due to adverse changes in financial market prices and rates. We do not hold or issue financial instruments for trading purposes or have any derivative financial instruments. To date, most payments made under our contracts are denominated in U.S. dollars and we have not experienced material gains or losses as a result of transactions denominated in foreign currencies. As of December 31, 2006, our cash reserves were maintained in money market investment accounts and fixed income securitities totaling \$16.0 million. These securities, like all fixed income instruments, are subject to interest rate risk and will decline in value if market interest rates increase. We have the ability to hold our fixed income investments until maturity and, therefore, we would not expect to experience any material adverse impact in income or cash flow.

Foreign Currency Risk

A portion of our revenues is derived from transactions denominated in U.S. dollars, even though we maintain sales and business operations in foreign countries. As such, we have exposure to adverse changes in exchange rates associated with operating expenses of our foreign operations, but we believe this exposure to be immaterial at this time. As such, we do not currently engage in any transactions that hedge foreign currency exchange rate risk. As we grow our international operations, our exposure to foreign currency risk could become more significant.

Recent Accounting Pronouncements

In June 2006, the Financial Accounting Standards Board issued FASB Interpretation No. 48 (FIN 48), Accounting for Uncertainty in Income Taxes, an interpretation of FASB Statement No. 109. This interpretation clarifies the accounting for income taxes by prescribing that a company should use a more-likely-than-not recognition threshold based on the technical merits of the tax position taken. Tax provisions that meet the more-likely-than-not recognition threshold should be measured as the largest amount of tax benefits, determined on a cumulative probability basis, which is more likely than not to be realized upon ultimate settlement in the financial statement. The interpretation also provides guidance on derecognition, classification, interest and penalties, accounting for interim periods, disclosure and transition, and explicitly excludes income taxes from the scope of SFAS No. 5, Accounting for Contingencies. FIN 48 is effective for fiscal years beginning after December 15, 2006. We are currently assessing the effect of FIN 48 on our consolidated financial statements.

In September 2006, the FASB issued SFAS No. 157, Fair Value Measurements. The purpose of this statement is to define fair value, establish a framework for measuring fair value and enhance disclosures about fair value measurements. The measurement and disclosure requirements are effective for us as of January 1, 2008 and are applied prospectively. We are currently evaluating the potential impact of adopting this new guidance on our results of operations and financial position.

BUSINESS

Overview

We provide a leading digital marketing intelligence platform that helps our customers make better-informed business decisions and implement more effective digital business strategies. Our products and solutions offer our customers deep insights into consumer behavior, including objective, detailed information regarding usage of their online properties and those of their competitors, coupled with information on consumer demographic characteristics, attitudes, lifestyles and offline behavior.

Our digital marketing intelligence platform is comprised of proprietary databases and a computational infrastructure that measures, analyzes and reports on digital activity. The foundation of our platform is data collected from our comScore panel of more than two million Internet users worldwide who have granted us explicit permission to confidentially measure their Internet usage patterns, online and certain offline buying behavior and other activities. By applying advanced statistical methodologies to our panel data, we project consumers' online behavior for the total online population and a wide variety of user categories.

We deliver our digital marketing intelligence through our comScore Media Metrix product family and through comScore Marketing Solutions. Media Metrix delivers digital media intelligence by providing an independent, third-party measurement of the size, behavior and characteristics of Web site and online advertising network audiences among home, work and university Internet users as well as insight into the effectiveness of online advertising. Our Marketing Solutions products combine the proprietary information gathered from the comScore panel with the vertical industry expertise of comScore analysts to deliver digital marketing intelligence, including the measurement of online advertising effectiveness, customized for specific industries. We typically deliver our Media Metrix products electronically in the form of weekly, monthly or quarterly reports. Customers can access current and historical Media Metrix data and analyze these data anytime online. Our Marketing Solutions products are typically delivered on a monthly, quarterly or ad hoc basis through electronic reports and analyses.

Industry Background

Growth of Digital Commerce, Content, Advertising and Communications

The Internet is a global digital medium for commerce, content, advertising and communications. According to IDC, the number of global Internet users is projected to grow from approximately 968 million in 2005 to over 1.7 billion in 2010. As the online population continues to grow, the Internet is increasingly becoming a tool for research and commerce and for distributing and consuming media. According to IDC, the global business-to-consumer eCommerce market is projected to grow from \$411\$ billion in 2005 to \$1 trillion in 2010. According to Jupiter Research, over 80% of online users in the United States research offline purchases using the Internet, making the Internet an important channel for both online and offline merchants. Consumers are also using the Internet to access an increasing amount of digital content across media formats including video, music, text and games. According to IDC, the domestic markets for online video and music consumption are projected to reach over \$1.7 billion and over \$3.3 billion, respectively, in 2010.

As consumers increasingly use the Internet to research and make purchases and to consume digital media, advertisers are shifting more of their marketing budgets to digital channels. According to the Internet Advertising Bureau and PriceWaterhouseCoopers, domestic online advertising spending, including search advertising, grew to \$16.8 billion in 2006, an increase of 34% over 2005. Despite the size and growth of the digital marketing sector, the shift of traditional advertising spending to the Internet has yet to match the rate of consumption of online media. According to Forrester Research, digital advertising represented only 6% of the total United States advertising market in 2004 despite consumers spending 16% of their available media time online. As advertisers spend more of their marketing budgets to reach Internet users, we believe that digital marketing will continue to grow.

In addition to the growth in online commerce, content and marketing, a number of new digital technologies and devices are emerging that enable users to access content and communicate in new ways.

Internet-enabled mobile phones allow users to access digital content such as games, music, video and news on their mobile devices through a wireless connection to the Internet. According to IDC, the worldwide number of shipments of converged mobile devices is projected to grow from 57 million in 2005 to 261 million in 2010, representing compounded annual growth of 36% over that period. Other digital communications technologies such as voice over Internet protocol (VoIP) utilize the Internet network infrastructure to enable efficient and cost-effective personal communications such as chat and VoIP-based telephony. According to Infonetics, the worldwide number of VoIP subscribers is projected to grow from 24.5 million in 2009. Delivery of digital television services over a network infrastructure using Internet Protocol, or IPTV, has a number of advantages over conventional television, including two-way communications, digital content and features, and interactivity. According to Infonetics, the worldwide number of IPTV subscribers is projected to grow from 2.4 million in 2005 to 68.9 million in 2009. We believe these and other new digital media and communications devices and services offer a similar opportunity as the Internet for us to measure and analyze user behavior.

Importance of Digital Marketing Intelligence

The interactive nature of digital media such as the Internet enables businesses to access a wealth of user information that was virtually unavailable through offline audience measurement and marketing intelligence techniques. Digital media provide businesses with the opportunity to measure detailed user activity, such as how users interact with Web page content; to assess how users respond to online marketing, such as which online ads users click on to pursue a transaction; and to analyze how audiences and user behavior compare across various Web sites. This type of detailed user data can be combined with demographic, attitudinal and transactional information to develop a deeper understanding of user behavior, attributes and preferences. Unlike offline media such as television and radio, which generally only allow for the passive measurement of relative audience size, digital media enable businesses to actively understand the link between digital content, advertising and user behavior.

We believe that the growth in the online and digital media markets for digital commerce, content, advertising and communications creates an unprecedented opportunity for businesses to acquire a deeper understanding of both their customers and their competitive market position. Businesses can use accurate, relevant and objective digital marketing intelligence to develop and validate key strategies and improve performance. For example, with a deep understanding of the size, demographic composition and other characteristics of its audience, an online content provider can better communicate the value of its audience to potential advertisers. With detailed metrics on the effectiveness of an online advertising campaign and how that campaign influences online and offline purchasing behavior, a business can refine its marketing initiatives. With insight into market share and customer behavior and preferences, a business can understand not only how its digital business is performing relative to its competitors but also the drivers behind such performance. Moreover, by using the appropriate digital marketing intelligence, businesses can refine their digital content, commerce, advertising and communications initiatives to enhance the effectiveness and return on investment of their marketing spending, enabling them to build more successful businesses.

Challenges in Providing Digital Marketing Intelligence

While the interactive and dynamic nature of digital markets creates the opportunity for businesses to gain deep insights into user behavior and competitive standing, there are a number of issues unique to the Internet that make it challenging for companies to provide digital marketing intelligence. Compared to offline media such as television or radio, the markets for digital media are significantly more fragmented, complex and dynamic. As of December 2006, we believe that there were more than 17,000 and 25,000 U.S. and global Web sites, respectively, that each receive more than 30,000 unique visitors per month, as compared to only a few hundred channels typically available with standard digital cable or satellite television and broadcast or satellite radio. The complexities of online user activity and the breadth of digital content and advertising make providing digital marketing intelligence a technically challenging and highly data-intensive process.

Digital media continues to develop at a rapid pace and includes numerous formats such as textual content, streaming and downloadable video and music, instant messaging, VoIP telephony, online gaming and email.

Digital advertising also includes multiple formats such as display, search, rich media and video. Detailed user activity such as viewing, clicking or downloading various components of a Web page across digital media or interacting with various advertising formats creates a substantial amount of data that must be captured on a continuous basis. The data must also be cleansed for quality, relevancy and privacy protection and be organized to enable companies to obtain relevant digital marketing intelligence. This capture of audience data can prove extremely challenging when it involves millions of Internet users with varying demographic characteristics accessing tens of thousands of Web sites across diverse geographies. In addition, the ongoing development of digital media programming languages and technologies contributes to the challenge of accurately measuring user activity. For example, online publishers and advertisers have recently started to use Asynchronous JavaScript and XML, or AJAX, a development technique that allows Web applications to quickly make incremental updates without having to refresh the entire Web page. Prior to AJAX, marketers relied heavily on page view statistics to plan and evaluate their online media spending programs. With AJAX, we believe marketers are beginning to question the definition of, and need for, page views, and are seeking alternative metrics for measuring the usage and effectiveness of online media. To maintain their relevance, audience and media measurement technologies must keep pace with the continued evolution and increasing complexity of digital media.

Need for Accuracy and Reliability. Relevant digital marketing intelligence requires access to accurate and reliable global data that measure online user activity. Existing data collection methodologies, including those that rely on third party sources, surveys or panels, face significant challenges and limitations. Survey or panel methodologies must measure a sufficiently large and representative sample size of Internet users to accurately capture data that is statistically projectable to the broader Internet population. In addition, the international composition of Internet audiences requires a geographically dispersed sample to accurately capture global digital activity. Digital marketing intelligence that depends on third-party sources to obtain Internet audience usage data has the potential to be biased, may be constrained by the data that the third party is capable of capturing, and may be limited in its application. For example, a solution that relies on data supplied by an Internet service provider, or ISP, may show a bias toward the demographic composition or other characteristics of that ISP's users. We believe that a meaningful digital media sourcing methodology must be based on data sourced from a large, representative global sample of online users that can be parsed, enhanced, mined and analyzed; must evolve rapidly and be flexible to adapt to changing technologies; and must be able to provide actionable digital marketing intelligence that can be used to improve business decision-making.

Need for Third-Party Objectivity. We believe that the availability of objective third-party data that measure digital audience size, behavior, demographic and attitudinal characteristics represents a key factor in the continued growth of digital content, advertising and commerce. This is similar to offline media markets, such as television and radio, whose development was significantly enhanced by the introduction of third-party audience measurement ratings that provided a basis for the pricing of advertising in those media. As the buying and selling of online advertising continues to grow, we believe that companies on both sides of the advertising transaction will increasingly seek third-party marketing intelligence to assess the value and effectiveness of digital media. In addition, as advertisers work with Web site publishers to target online advertising campaigns to reach a specific demographic or behavioral user profile, the need for objective audience and user information, unbiased by either party to the transaction, will become increasingly important.

Need for Competitive Information. In addition to the scope, complexity and rapid evolution of online digital media, the lack of data on competitors makes it difficult for companies to gain a comprehensive view of user behavior beyond their own digital businesses. While products and tools exist that enable companies to understand user activity on their own Web sites, these products are unable to provide a view of digital audience activity on other Web sites or offline. In order for publishers, marketers, merchants and service providers to benefit from accurate and comprehensive digital marketing intelligence they need to understand user activity on Web sites across the Internet and how online consumer behavior translates into offline actions

The comScore Digital Marketing Intelligence Platform

We provide a leading digital marketing intelligence platform that enables our customers to devise and implement more effective digital business strategies. Our platform is comprised of proprietary databases and a computational infrastructure that measures, analyzes and reports digital activity from our global panel of more than two million Internet users. We offer our customers deep insights into consumer behavior on their own online properties and those of their competitors, including objective, detailed information on users' demographic characteristics, attitudes. lifestyles and multi-channel buying activity. We also provide industry-specific metrics to our customers.

We deliver our digital marketing intelligence through our comScore Media Metrix product family and through comScore Marketing Solutions. Media Metrix provides intelligence on digital media usage, including a measurement of the size, behavior and characteristics of the audiences for individual Web sites and advertising networks within the global home, work and university Internet user populations as well as insight into the effectiveness of online advertising. Our Marketing Solutions products combine the proprietary information gathered from our user panel with the vertical industry expertise of comScore analysts to deliver digital marketing intelligence customized for specific industries. Media Metrix and Marketing Solutions products are typically delivered electronically in the form of periodic reports, through customized analyses or are generally available online via a user interface on the comScore Web site.

Key attributes of our platform include:

- Panel of global Internet users. Our ability to provide digital marketing intelligence is based on information continuously gathered from a broad cross-section of more than two
 million Internet users worldwide who have granted us explicit permission to confidentially measure their Internet usage patterns, online and certain offline buying behavior and
 other activities. Through our proprietary technology, we measure detailed Internet audience activity across the spectrum of digital content and marketing channels. Many
 comScore panelists also participate in online survey research that captures and integrates demographic, attitudinal, lifestyle and product preference information with Internet
 behavior data. The global nature of our Internet panel enables us to provide digital marketing intelligence for over 30 individual countries. Our global capability is valuable to
 companies based in international markets as well as to multi-national companies that want to better understand their global Internet audiences and the effectiveness of their global
 digital business initiatives.
- Scalable technology infrastructure. We developed our databases and computational infrastructure to support the growth in online activity among our global Internet panel and the increasing complexity of digital content formats, advertising channels and communication applications. The design of our technology infrastructure is based on distributed processing and data capture environments that allow for the collection and organization of vast amounts of data on online activity, including usage of proprietary networks such as AOL, instant messaging and audio and video streaming. Our database infrastructure currently captures approximately 182 million Web pages and 4.5 billion URL records each week from our global Internet panel, resulting in over 28 terabytes of data collected by our platform each month. We believe that our efficient and scalable technology infrastructure allows us to operate and expand our data collection infrastructure on a cost-effective basis. In recognition of the scale of our data collection and warehousing technology, we have received multiple awards, including the 2003, 2004 and 2005 Winter Corporation Grand Prize for Database Size on a Windows NT Platform.

Benefits of our platform include:

Advanced digital marketing intelligence. We use our proprietary technology to compile vast amounts of data on Internet user activity and to organize the data into discrete, measurable elements that can be used to provide actionable insights to our customers. We believe that our digital marketing intelligence platform enables companies to gain a deeper understanding of their digital audiences, which allows them to better assess and improve their company and product-specific competitive position. Because our marketing intelligence is based on a large sample of global Internet users and can incorporate

multi-channel transactional data, we are able to provide companies with an enhanced understanding of digital audience activity beyond their own Web sites and the ability to better assess the link between digital marketing and offline user activity. Digital content providers, marketers, advertising agencies, merchants and service providers can use the insights our platform provides to craft improved marketing campaigns and strategies and to measure the effectiveness and return on investment of their digital initiatives.

- Objective third-party resource for digital marketing intelligence. We are an independent company that is not affiliated with the digital businesses we measure and analyze, allowing us to serve as an objective third-party provider of digital marketing intelligence. Because businesses use our data to plan and evaluate the purchase and sale of online advertising and to measure the effectiveness of digital marketing, it is important that we provide unbiased data, marketing intelligence, reports and analyses. We deploy advanced statistical methodologies in building and maintaining the comScore global Internet user panel and utilize proven data capture, and computational practices in collecting, statistically projecting, aggregating and analyzing information regarding online user activity. We believe that our approach ensures that the insights we provide are as objective as possible and allows us to deliver products and services that are of value to our customers in their key business decision-making. We believe that the media industry views us as a highly recognized and credible resource for digital marketing intelligence. For example, between March 1 and December 31, 2006, our information on digital activity was cited more than 16,500 times by third-party media outlets, an average of approximately 55 citations per day. Our data are regularly cited by well-known media outlets such as the Associated Press, Reuters, Bloomberg, CNBC, The New York Times and The Wall Street Journal. Moreover, many of the leading Wall Street investment banks also purchase and cite our data in their published research reports prepared by financial analysts that cover Internet businesses.
- Vertical industry expertise. We have developed expertise across a variety of industries to provide digital marketing intelligence specifically tailored to the needs of our customers operating in specific industry sectors. We have dedicated personnel to address the automotive, consumer packaged goods, entertainment, financial services, media, pharmaceutical, retail, technology, telecommunications and travel sectors. We believe that companies across different industries have distinct information and marketing intelligence needs related to understanding their digital audiences and buyers, evaluating marketing initiatives and understanding company or product-specific competitive position. For example, a pharmaceutical company may want to understand how online research by consumers influences new prescriptions for a particular drug, while a financial services company may want to assess the effectiveness of its online advertising campaigns in signing up new consumers and how this compares to the efforts of its competitors. By working with companies in various industries over the course of multiple years, we have developed industry-specific applications of our data and our client service representatives have developed industry-specific knowledge and expertise that allow us to deliver relevant and meaningful marketing insight to our customers.
- Ease of use and functionality. The comScore digital marketing intelligence platform is designed to be easy to use by our customers. Our Media Metrix products are available through the Internet using a standard browser. Media Metrix customers can also run customized reports and refine their analyses using an intuitive interface available on our Web site. Our Marketing Solutions products are available either through the Internet or by using standard software applications such as Microsoft Excel, Microsoft PowerPoint or SPSS analytical software. Our customers do not need to install additional hardware or complex software to access and use our products.

Strategy

Our objective is to be the leading provider of global digital marketing intelligence products. We plan to pursue our objective through internal initiatives and, potentially, through acquisitions and other investments. The principal elements of our strategy are to:

- Deepen relationships with current customers. We intend to work closely with our customers to enable them to continuously enhance the value they obtain from our digital
 marketing intelligence platform. Many of our customers are Fortune 1000 companies that deploy multiple marketing initiatives, and we believe many of our customers would
 benefit from more extensive use of our product offerings to gain additional insights into their key digital initiatives. We will work to develop and expand our customer
 relationships to increase our customers' use of our digital marketing intelligence platform.
- Grow our customer base. As the digital media, commerce, marketing and communications sectors continue to grow, we believe the demand for digital marketing intelligence
 products will increase. To meet this increase in market demand, we intend to invest in sales, marketing and account management initiatives in an effort to expand our customer
 base. We intend to offer both general and industry-specific digital marketing products that deliver value to a wide range of potential customers in current and new industry
 verticals
- Expand our digital marketing intelligence platform. We expect to continue to increase our product offerings through our digital marketing intelligence platform. As digital markets become more complex, we believe that companies will require new information and insights to measure, understand and evaluate their digital business initiatives. We intend to develop new applications that leverage our digital marketing intelligence platform to be able to provide the most timely and relevant information to our customers. For example, in 2003 we were one of the first companies to offer data, analysis and reports on the fast-growing Internet search market.
- Address emerging digital media. The extension of digital media and communications to include new formats such as VoIP, IP television, content for mobile phones and next
 generation gaming consoles creates new opportunities to measure and analyze emerging digital media. We intend to extend our digital marketing platform to capture, measure and
 analyze user activity in these emerging digital media and communications formats.
- Extend technology leadership. We believe that the scalability and functionality of our database and computational infrastructure provide us with a competitive advantage in the
 digital media intelligence market. Accordingly, we intend to continue to invest in research and development to extend our technology leadership. We intend to continue to enhance
 our technology platform to improve scalability, performance and cost effectiveness and to expand our product offerings.
- Build brand awareness through media exposure. Our digital media, commerce and marketing information is frequently cited by media outlets. In addition, we proactively provide them with data and insights that we believe may be relevant to their news reports and articles. We believe that media coverage increases awareness and credibility of the comScore and Media Metrix brands and supplements our marketing efforts. We intend to continue to work with media outlets, including news distributors, newspapers, magazines, television networks, radio stations and online publishers, to increase their use of comScore data in content that discusses digital sector activity.
- Grow internationally. While we are currently in the early stages of providing customers with international services, we believe that a significant opportunity exists to provide our product offerings to multi-national and international companies. Approximately half of the existing comScore Internet user panel resides outside of the United States. In July 2006, we launched World Metrix, a product that measures global digital media usage. World Metrix is based on a sample of online users from countries that comprise approximately 95% of the global Internet population. We plan to expand our sales and marketing and account management presence outside the U.S. as we provide a broader array of digital marketing intelligence products that are tailored to local country markets as well as the global marketplace.

Our Product Offerings

We deliver our digital marketing intelligence through our comScore Media Metrix product family and through comScore Marketing Solutions.

comScore Media Metrix

Media Metrix provides its subscribers, consisting primarily of publishers, marketers, advertising agencies and advertising networks, with intelligence on digital media usage and a measurement of the size, behavior and characteristics of the audiences for Web sites and advertising networks among home, work and university Internet populations. Media Metrix also provides insights into the effectiveness of online advertising. Media Metrix data can be used to accurately identify and target key online audiences, evaluate the effectiveness of digital marketing and commerce initiatives, support the selling of online advertising by publishers, and to identify and exploit relative competitive standing. The vast majority of our Media Metrix subscribers access selected reports and analyses through the MyMetrix user interface on our Web site.

Our flagship product, Media Metrix 2.0, details the online activity and site visitation behavior of Internet users, including use of proprietary networks such as AOL, instant messaging, audio and video streaming, and other digital applications. Our customers subscribe to ongoing access to our digital marketing intelligence reports and analyses, including:

- comprehensive reports detailing online behavior for home, work and university audiences;
- demographic characteristics of visitors to Web sites and properties;
- buying power metrics that profile Web site audiences based on their online buying behavior;
- detailed measurement and reporting of online behavior for over 30 countries and over 100 U.S. local markets;
- measurement of key ethnic segments, including the online Hispanic population; and
- · reach and frequency metrics for online advertising campaigns that show the percent of a target audience reached and the frequency of exposure to advertising messages.

A representative MyMetrix screenshot, detailing the most visited online properties in the United States for December 2006, is shown on the following page.

1		_			•	_															7																					•	1
LOGOUT	trix	Find																																									•
KONAS	MyMetrix Contact Export Halp Logaria		Buying Four Index (BFI)	508F (16548)	100	1	117	125	187	114	181	128	108	131	162	145	161	99	142	138	127	217	153	125	117	840	150	230	114	242	988	181	180	980	558	282	100	2112	182	10.	40 00	100	ŧ
- AB	Cont	L	Averager Pages per Visitor	338T (TR(13)	2,588		187	20	100	92	8	æ 8	8 8	2	ş	ħ	8 8	2 9	8	m	17	2	8	28 8	ı E	•	2 :	2 2	8	100	200	15	ž	5 2	8	8	2 5	2 63	***	2 1	8 5	7	ŧ
TOM SUPPORT			Average Bloutes per Visitor	SERT (TRENE)	1,784.8	- 0.00	985.8	178.8	88.8	808.1	81.5	17.8	ND 4	15.6	12.8	13.0	11.0	2 2	18.8	es :	11.8	12.0	18.1	28.5	14.0	8	38.5	1 2	18.1	12.8	100.4	88.0	19.5	18.7	11,3	18.0	187	87.5	6.7	888	408	98	-
REPORTS CUSTOM			Average Usage Bays per Visitor		818		2 2	70	98	83	8.8	82	5 8	9	2	2	8	- C	84	2	PO GO	2	2	2:	200	2		9 9	200	100	2.0	2.1	98	88	80	2	e o	2	20	88	4 8	ă	•
¥	Î		Total Pages per Usage Jay	SHET (TREAT)	119.8	1 8	13.8	79	18.7	889	12.9	00 5	9 95	47	4.8	5	11.7	N C	85	8	148	10.0	18.0	118	2	4.7	10.7	13.0	7.8	10.8	11.0	81.4	8.0	4 0	10.01	888	14.8	11.1	4	10.3	7.0	6.7	
	Geograph' United States Locations All Locations The Period Secular SIL Trapt: Tead Audient SIL Media 1 for All Properties Media 1 for All Properties Media 1 for All States Media 2 for All States Media 3 for All States Media 4 for	Save to MyReports	Total Peges Viewed (WD)	SORT (TREAT)	461,044		18.800	17.958	13,387	41,450	2,083	1,711	2 597	708	481	679	1,153	889	944	88	427	9000	828	1.201	8	178	1,894	878	414	488	1,084 1,084	2,891	459		334	788	755	439	138	4	1000	818	
	Dography Teaching Teach	Save to	Average Rinutes per Usage Jay	SORT THEM	£19	1 0	38.4	18.4	F 5	84.0	7.4	8 5	181	42	20	88	200	2	8.1	8	18.0	. F	7.5	22	2	9	22	90	5.1	100	18.0	12.8	2.0	5 6	89	10.0	989	15.3	20	2;	4.5	Ş	
			Total Minutes (MD)	SBRT (TREAD)	307,442		31.784	20,247	7,888	14,881	1,222	887	9 380	624	803	201	100	308	623	8	305	287	431	719	341	配	888	548	870	545	123	1,685	322	100	28	355	919	808	80	470	200	2	
			Average Bally Visitors (000)	CHART CHARD	121,389	100.04	38.798	35,425	31,439	18,504	5,288	8,858	2,416	4.801	3,135	3,053	3,188	8.791	2,524	1,209	1,808	1,588	1,843	3,502	1,741	1,213	3,702	1.887	1,888	1,548	4,800	4,283	2,036	1.043	1,076	1,038	1,870	1.282	1,032	1,887	4,859	1,048	
			2 Reach	SORT THEN	100.0	W.	60.4	6.60	25.0	1.04	35.8	25 E	280	8	252	88	19.0	5	18.4	14.8	14.8	142	18.7	18.4	13.2	18.1	13.0	128	11.8	12	801	10.7	10.4	000	82	2	8 8	2	0.3	8 8	2 0	2	e e
			Total Brigue Visitors (DDD)	SSET TRIES	174,198	N/A	120,080	118,508	112,848	78,878	58,768	55,672	40.049	38,418	39,242	38,583	34,481	39,000	31,989	20,851	25,449	24,674	29,833	23,290	23,078	22,802	22,074	22,22	20,683	18,429	18.100	18,718	18,167	17,886	18,980	18,883	18,766	18,235	18,168	18,051	18,008	18,484	·
	KEY MEASURES REPORT TOP 100 PROPERTIES [UNDUP.]	Print Download Save to MyReports	Items 3 to 300 of 300 First Pew Next Last		O Total Internet: Total Audience	Top 100 Properties	O Time Margar Naturals	O Microsoft Sites	O Geogle Sites	O Fox Interactive Nedia	O Amazon Sites	O Ask Network	O Charles Market	O Agole Computer, Inc.	O New York Times Bigital	O Wikipedia Sites	O Varizon Communications Corporation	O Meather Channel The	O CNET Networks	O Adobe Sites	Corille Netion	O Best Buy Sites	O ATET, Inc.	O United Online, Inc	O Menster Meridalde	O Shapzilla.com Sites	O Bank of America	C Sears Silver	O Real-cre Network	O Expedie Inc	O PACEBOOK CON	O Concast Corporation	O Gannett Sites	O UPS Sites	O Circuit City Stores, Inc.	O OCPanney Sites	O Photosbucket.com LLC	O E.W.Scrippe	O Yelloupages.com Network	O NFL Internet Group	Meatherbug Property	O ARTISTEINECT Network	O EA Online
	KEY TOP 1	T.					- 0	. 00	4 .		4		. 5	-	12	13	7 :	9 5	17 B	2	P 8	<u>ه</u>	83	8 3	8 2	8	9 8	8 8	8	E 6	8 8	8	18	86	8	8	8 :	9		1	9 6	0	ę

In addition to our core offering, customers can subscribe to the following additional products in the Media Metrix product family:

Plan Metrix. Plan Metrix is a product that combines the continuously and passively observed Internet behavior provided by Media Metrix with comprehensive attitude, lifestyle and product usage data collected through online surveys of our U.S. Internet user panel. Plan Metrix provides advertising agencies, advertisers and publishers with multiple views of Web site audiences including their online behavior, demographics, lifestyles, attitudes, technology product ownership, product purchases and offline media usage. These data are used in the design and evaluation of online marketing campaigns. For example, an online auto retailer could use Plan Metrix to help understand which Web sites a prospective automobile purchaser is most likely to visit prior to making a purchase decision.

World Metrix. We provide insights into worldwide Internet activity through our World Metrix product, which delivers aggregate information about the behavior of online users on a global basis, for approximately 30 individual countries and for regional aggregations such as Latin America, Europe and Asia Pacific. For example, a content publisher can understand its market share of the global Internet audience using our World Metrix product.

Video Metrix. Video Metrix provides insights into the viewing of streaming video by U.S. Internet users. The product measures a wide range of video players and formats, including Windows Media, Flash, RealMedia and QuickTime. Video Metrix offers site-level measurement and audience ratings by demographics and time-of-day to assist agencies, advertisers and publishers in designing and implementing media plans that include streaming video. For example, an advertiser that is seeking to maximize the exposure of its streaming video ads to its target audience could use Video Metrix to help understand on which sites and at what times of the day its target audience is viewing the most streaming video.

Ad Metrix. Available through the Media Metrix client interface, Ad Metrix provides advertisers, agencies and publishers with a variety of online advertising metrics relating to impressions, or advertisements on a Web site that reach a target audience. Ad Metrix helps customers determine the impressions delivered by advertising campaigns across Web sites and online properties, including how many visitors are reached with advertisements and how often. In addition, Ad Metrix allows customers to determine the demographic profile of the advertising audience at a particular site, as well as how the volume of impressions changes over time on that site. The Ad Metrix data are consistent with offline media planning metrics such as GRPs, or gross rating points, which measure the percent of a target audience that is reached with an advertisement weighted by the number of exposures. For example, an advertiser might use Ad Metrix to plan the online portion of an advertising campaign for a sports product on sites that have previously successfully delivered advertising impressions to a target demographic audience. A publisher might use Ad Metrix data to measure its share of advertising impressions relative to competitive publishers. Ad Metrix was launched in early 2007 in beta format and we plan to commercially launch this product in the second quarter of 2007.

Some examples of Media Metrix digital marketing intelligence measurements and their customer uses are described in the following table.

Digital Marketing Intelligence Measurement **Examples of Customer Uses** Site Traffic & Usage Intensity • rank Web sites based on online usage metrics such as unique visitors, page views or minutes of use ${f \cdot}$ drill-down to standard or customer-defined site subsets such as channels or sub-channels (such as Yahoo! Finance and Yahoo! Sports) • analyze statistics over time such as trends in site visitors within demographic segments • assess which Web site audiences are growing or declining, which sites are most attractive to particular demographic segments or which sites or digital applications have the highest level of \bullet identify the source of traffic to a particular Web site or channel within a site profile site users based on life-stage or offline behavior such as panelist-reported TV usage, car ownership, health conditions or offline purchases Quantitative Consumer Information • efficiently identify and target a particular user segment (e.g., people who say they are likely to buy a car in the next six months) • quantify the audience overlap between different consumer segments or Web sites to identify the number of unique visitors reached • quantify the propensity of a particular Web site's audience to purchase certain categories of products Online Buying Power (e.g., consumer electronics) online Competitive Intelligence compare the standings of Web sites within particular content categories, such as finance or health ullet quantify audience size relative to competitors, including share of usage within a category and usage trends across competitors • track major competitors, quantify their growth, and identify initiatives to promote growth and market share Reach and Frequency • identify and quantify the size of audiences reached by individual Web sites and determine how often they reach those audiences ullet assist with the planning of online advertising campaigns that need to achieve specific reach or frequency objectives against a targeted audience across multiple Web sites ullet design the most cost-effective media plans that can achieve campaign objectives for reach and frequency

comScore Marketing Solutions

comScore Marketing Solutions products use our global database, computational infrastructure and our staff of experienced analytical personnel to help customers design more effective marketing strategies that increase sales, reduce costs, deepen customer relationships and ultimately enhance a customer's competitive position. We offer solutions tailored for specific industry verticals, including the automotive, consumer packaged goods, entertainment, financial services, media, pharmaceutical, retail, technology, telecommunications and travel sectors. Many of our Marketing Solutions products are delivered to subscribers on a recurring schedule such as monthly or quarterly. In some cases, we provide customized reports and analyses that combine our expertise with other proprietary information to address a specific customer need.

The core information products offered by comScore Marketing Solutions include:

Market Share Reports. These reports track a company's share of market as measured by industry-specific performance metrics. The metrics of choice vary by industry vertical, including as examples: share of online credit card spending for credit card issuers; share of online travel spending for travel companies; or share of subscribers for ISPs. In each case, market share reports provide an ongoing measurement of competitive performance and insight into the factors driving changes in market share.

Competitive Benchmark Reports. These reports allow customers to compare themselves to competitors using various industry-specific metrics. For example, retailers may look at metrics such as the rate of conversion of site visitors to buyers, average order size or rate of repeat purchases among existing customers. Banks may focus on the percentage of bank customers using online bill payment services, or compare the effectiveness of customer acquisition programs as reflected by the percentage of leads they acquire that ultimately sign up for an online account. In each case, a customer may define and obtain best-of-category metrics and use them as a benchmark to monitor its business performance over time.

Loyalty and Retention Analysis. These analyses provide an understanding of the extent to which consumers are also engaged with competitors, and identifies loyalty drivers to assist customers in capturing a higher share of the consumer's wallet. For example, a travel company might quantify the potential business lost when consumers visit its site, do not complete a purchase but then visit a competing site to book a travel reservation. Retention or churn analyses quantify consumer losses to competitors and the key drivers of such losses. For example, a narrowband Internet service provider may track the rate of attrition among its customer base, identify which competitors are capturing those lost customers, and analyze the characteristics of the lost customers in order to gain insight into ways to improve retention.

Customer Satisfaction Reports. These reports are based on panelist responses to survey questionnaires that ascertain the degree of satisfaction with various products or services offered to consumers. This information is often integrated with the online usage information that we collect from our panelists in order to identify which digital media usage activities affect customer satisfaction. For instance, a sports portal may use these reports to determine which features, such as participating in fantasy sports leagues or viewing streaming video clips, affect customer satisfaction and lovalty the most.

qSearch. This product is a monthly scorecard of the search market that provides a comparison of search activity across portals and major search engines. It helps identify the reach of a search engine, the loyalty of its user base, the frequency of search queries, and the effectiveness of sponsored links displayed on search result pages in driving referrals to advertiser sites. qSearch is used by major search engines and advertising agencies in planning search campaigns.

Campaign Metrix. This product provides detailed information about specific online advertising campaigns. These reports, available through a Web-based interface, describe for each advertising image, or "creative" within an advertising campaign, the size and demographic composition of the audience exposed to that particular advertisement, the average number of impressions delivered and other details regarding ad formats and ad sizes used in the campaign. An advertiser, agency or publisher could use Campaign Metrix to gain insight into the effectiveness of an online advertising campaign by examining the number of unique users exposed to the campaign, the number of times on average that a unique user was exposed to the campaign and

whether the campaign reached the targeted audience demographic. This product was launched in February 2007 in beta format and we plan to commercially launch this product in the second quarter of 2007.

Internet Advertising Effectiveness Studies. These studies provide an understanding of the effectiveness of particular advertising campaigns by measuring the online and offline behavior of a "target group" of comScore panelists, following their exposure to a particular advertisement, and comparing their behavior to that of a "control group" of comScore panelists who were not exposed to such advertisements. This type of a study allows a marketer to understand the impact of their advertising campaign and to estimate the return on their investment in online marketing.

Survey-Based Products. These products leverage our ability to administer surveys to our panel members to obtain valuable information that can be seamlessly integrated with online behavioral data to provide our clients with additional insights into the drivers of consumer behavior.

Customers

As of December 31, 2006, we had 706 customers, including over 100 Fortune 1000 customers. Our customers include:

- fifteen of the top twenty online properties, based on total unique visitors, as ranked by our Media Metrix database for the month of December 2006, including Microsoft, Yahoo!, AOL and Google;
- ten of the top twenty U.S. Internet service providers, based on the number of subscribers as of the third quarter of 2006, as ranked by ISP Planet;
- the top ten investment banks, based on 2006 revenues, as ranked by Dealogic;
- · 97 advertising and media buying agencies;
- five of the top six consumer banks, based on consolidated assets as of December 31, 2006, as ranked by the Federal Reserve System, National Information Center;
- seven of the top ten pharmaceutical companies, based on 2005 worldwide sales, as ranked by IMS Health; and
- seven of the top eight credit card issuers, based on total credit cards outstanding in 2006, as ranked by the 2006 Nilson Report.

One of our customers, Microsoft Corporation, accounted for 5% of our revenues in the year ended December 31, 2004, 14% of our revenues in the year ended December 31, 2005 and 12% of our revenues in the year ended December 31, 2006.

The following examples are provided as an illustration of the development and growth of our relationships with our customers:

• Microsoft is a leading provider of software, services and solutions. Since 2001, Microsoft's Internet division, MSN, has used our global panel data to better understand the needs of consumers, to help guide product planning strategies and to measure the impact of online marketing efforts, and has increased its use of our products in each subsequent year. Since 2004, MSN has purchased detailed Internet clickstream data patterns to study how consumers use MSN and competitive services, in order to better meet consumer needs. Since June 2005, MSN has used our gSearch product to measure and benchmark the behavior of consumers and competitors in the Internet search market. Since 2005, we have also provided MSN with advertising studies that it has used to measure the impact of MSN's online marketing campaigns and demonstrate to clients the effectiveness of online advertising. In addition, since 1999, Microsoft has been a customer of SurveySite, a company that we acquired on December 31, 2004. comScore SurveySite provides Microsoft with insights about their customers, partners and employees by conducting online qualitative research and quantitative surveys, including ongoing customer satisfaction tracking programs. comScore SurveySite has been a Premier Vendor for Online Research to Microsoft since 2002. comScore SurveySite was also the winner of the 2005

Microsoft Vendor Program Excellence Award in Technology in recognition of its innovative SiteRecruit system. In 2006, comScore SurveySite was also named a Relationship Marketing Specialty Vendor, a designation shared by only five market research vendors worldwide. comScore SurveySite has worked across all of Microsoft's principal business groups including Platform Products and Services, Business Products and Services and Entertainment and Devices.

- Verizon Communications is a leader in delivering broadband and other wireline and wireless communication innovations to business, government and wholesale and retail
 customers. Since 2001, Verizon Communications has used comScore Marketing Solutions products to better understand the competitive landscape in the Internet access industry
 and trends in broadband offerings. Starting with the purchase of an ISP market share analysis for two specific markets, Verizon Communications now uses our data and analyses
 in over 40 markets to not only understand its competitive position in the industry, but also to determine the efficacy of its broadband product line and to help guide marketing
 strategies. Verizon Communications also uses other comScore Marketing Solutions products to obtain answers to a variety of other business issues.
- Starcom USA is an independent operating unit of Starcom MediaVest Group, a global advertising and marketing agency. Starcom has been a customer of comScore's Marketing Solutions products since 2004, when it purchased an analysis to quantify the impact of a Fortune 500 client's online advertising on its share of consumer eCommerce spending during the 2003 holiday shopping season. In 2005, Starcom expanded the relationship to include comScore Marketing Solutions' online survey capabilities. Since 2004, Starcom's purchases of our products have expanded from purchasing surveys and holiday season eCommerce tracking to purchases covering almost the entire year. Starcom uses our digital market intelligence to analyze the impact of online advertising on its clients' share of consumer eCommerce spending at a total Internet and product category level. Starcom also uses our marketing solutions brand accountability analyses that we generate from survey results from our global consumer Internet panel.
- Yahoo! is a leading global Internet portal. Yahoo! became a customer when we acquired certain Media Metrix assets in 2002. Since then, Yahoo! has purchased additional Media Metrix products and in 2004 chose comScore as Yahoo!'s source of record for Internet audience measurement and search. Yahoo! has exclusively used Media Metrix for digital marketing intelligence in the U.S. since 2006. In 2002, our relationship with Yahoo! expanded with the launch of our qSearch product that tracks consumers' use of various search engines. qSearch information is used by Yahoo! in numerous aspects of managing its search business, including product development, market share tracking, competitive analysis, ad effectiveness and executive reporting. Yahoo! also commissioned us to conduct several analyses that measured the degree to which offline sales and latent online sales (sales made days or weeks after the initial click-through) were impacted by search advertising. In late 2005 and throughout 2006, Yahoo! integrated our advertising effectiveness testing products into its suite of advertiser products, thereby enabling its advertisers to analyze campaign effectiveness by measuring a variety of different metrics including offline sales, surveyed branding and awareness, online site usage and trademark search activity. In 2006, we completed two significant studies for Yahoo! entitled "Close the Loop" a study on the link between search and image advertising, and "Brand Advocates: The Impact of Search and Social Media on Branding." We became a preferred provider of services to Yahoo! In 2006. In 2007, our relationship with Yahoo! grew with the addition of international and worldwide data and ongoing adoption of certain of our new syndicated and custom comScore digital marketing intelligence products.

Selling and Marketing

We sell the majority of our products through a direct sales force. Sales of the comScore Media Metrix product suite to new clients are managed by sales representatives assigned specifically to new business development. A separate group of account managers within our sales organization is assigned to manage, renew and increase sales to existing Media Metrix customers. The comScore Marketing Solutions sales organization is organized vertically by industry with account executives dedicated to selling into the

automotive, consumer packaged goods, entertainment, financial services, media, pharmaceutical, retail, technology, telecommunications and travel sectors and other industries. Marketing Solutions account executives are tasked with both identifying and generating new business in specific verticals as well as servicing existing customers. Our sales and account representatives receive a base salary and are eligible for bonuses or commissions based on performance.

Our marketing communications staff is primarily focused on leveraging the use of comScore data and insights by the media and maximizing the number of times that comScore is cited as a source of information. We believe that the use of our data by general and industry-specific media outlets increases recognition of the comScore brand name and serves to help validate the value of the analyses and products we provide. In order to accomplish this goal, we seek to maintain relationships with key news distributors, publications, TV networks, reporters and other media outlets. We believe that the media views us as a highly recognized and credible resource for digital marketing intelligence. For example, between March 1 and December 31, 2006, comScore data were cited more than 16,500 times by third-party media outlets, an average of over 55 citations per day. Moreover, we are regularly cited by well-known news distributors, publications and TV networks such as the Associated Press, Reuters, Bloomberg, CNBC, The New York Times and The Wall Street Journal. We also target various industry conferences and tradeshows as part of our marketing efforts. These events are typically focused on a particular industry, allowing us to demonstrate to industry participants the value of our products to businesses in that industry.

Panel and Methodology

The foundation of our digital marketing intelligence platform is data collected from our comScore panel, which includes more than two million persons worldwide whose online behavior we have explicit permission to measure on a continuous, passive basis. We believe that our panel is one of the largest global panels of its kind, delivering a multi-faceted view of digital media usage and transactional activity as well as selected offline activity. By applying advanced statistical methodologies to our panel data, we project the behavior of the total online population.

We recruit our panel through a variety of online recruitment programs that have been tested and refined since our inception to ensure a diverse sample that sufficiently represents the broader global Internet population. In addition, in the United States we enlist a sub-sample of panelists through various offline recruiting methods. Participants in the comScore research panel receive a package of benefits that is designed to appeal to a broad variety of user categories. Examples of such benefits include, as of December 2006, free security applications such as server-based virus protection, encrypted file protection, encrypted network disk storage locations for user backups; free general purpose applications such as screensavers and games; sweepstakes; cash payments; and points that may be redeemed for prizes. Participants' data and privacy are protected by defined privacy policies that safeguard personally-identifiable information. This combination of recruiting methods allows us to maintain a panel large enough to provide statistically representative samples in most demographic segments.

We continuously determine the size, demographics and other characteristics of the online population using enumeration surveys of tens of thousands of persons annually, whereby respondents are asked a variety of questions about their Internet use, as well as demographic and other descriptive questions about themselves and their households. The sample of participants in each enumeration survey is selected using a random recruiting methodology. The result is an up-to-date picture of the population to which the comScore sample is then projected. We use the results from the enumeration surveys to weight and statistically project the panel data to ensure that the projected data reflect the characteristics of the Internet population.

Privacy

We believe that a key factor differentiating our digital marketing intelligence is our ability to track and analyze online usage behavior using the data collected from our panel. Since the founding of our company, we have endeavored to undertake such data collection and analysis responsibly and only with consumer permission. Participation in our research panel is voluntary. Participants must consent to our privacy and data

security practices before our software collects information on the user's online activity. In addition, we provide panelists with multiple opportunities and methods to remove themselves from our panel. We limit the type of information that we collect by identifying and filtering certain personal information from the data collected. The collected data is secured using multiple layers of physical and digital security mechanisms. Moreover, we maintain a strict policy of not sharing panelists' personally identifiable information with our customers. Understanding the sensitive nature of the data that we collect and analyze, each year we undergo an independent third-party audit of our privacy and data security practices by Ernst & Young, a leading digital-assurance firm. This audit involves a review of the disclosures we make and the consents obtained from our panelists, review of our data security practices, and testing of these practices, including attempts to penetrate our network and compromise our systems. Each year, since 2000, we have been certified as meeting the AICPA/CICA WebTrust criteria for online privacy.

Technology and Infrastructure

We have developed a proprietary system for the measurement of the activity of our global online panel. This system is continuously refined and developed to address the changing digital media landscape and to meet new customer business needs. The system is comprised of hundreds of servers that operate using software built on Microsoft and other technologies. Our technology infrastructure is operated in two third-party Tier-1 co-location facilities (one in Virginia and the other in Illinois). Our systems have multiple redundancies and are structured to ensure the continuation of business operations in the event of network failure or if one of our data centers has been rendered inoperable. As of December 31, 2006, our technology team (excluding employees devoted to research and development) was comprised of over 105 full-time employees (or full-time equivalents) working in four different geographic locations, who design, develop, maintain and operate our entire technology infrastructure. In addition, we have established a relationship with a third party firm for software development in an economically beneficial locale as a means to augment our technology efforts for discrete projects.

Our development efforts have spanned all aspects of our business. We have developed a data capture system that operates across our panelists' computers in almost 200 countries and is used for the real-time capture of consumer Internet behavior. We have built a large scale, efficient and proprietary system for processing massive amounts of data. Typically our systems handle and process data in excess of 10 billion input records per month. Despite the scale of processing required, these data are generally available on a daily basis for our business use. We have also developed a highly efficient and scalable system for the extraction and tabulation of all online activities of our panelists. Likewise, we have created a highly scalable data warehousing environment that allows ready access and analysis of the data we collect. This system, based on Sybase IQ, was awarded the 2003, 2004 and 2005 Grand Prize for the largest Microsoft-based decision support warehouse by the Winter Corporation. In December 2006, we were recognized as a 2007 Technology Pioneer by the World Economic Forum. We believe our scalable and highly cost-effective systems and processing methods provide us with a significant competitive advantage.

Our customers access our digital marketing intelligence product offerings through a variety of methods including MyMetrix, our proprietary, Web-based analysis and reporting system, which in the month of December 2006 was used by 4,020 users to produce more than 170,000 reports.

Research and Development

Our research and development efforts focus on the enhancement of our existing products and the development of new products to meet our customers' digital marketing intelligence needs across a broad range of industries and applications. Because of the rapidly growing and evolving use of the Internet and other digital mediums for commerce, content, advertising and communications, these efforts are critical to satisfying our customers' demand for relevant digital marketing intelligence. As of December 31, 2006, we had approximately 82 full-time employees (or full-time equivalents) working on research and development activities (excluding employees on our technology team cited under "Technology and Infrastructure" above). In addition, we involve management and operations personnel in our research and development efforts. In

2006, 2005 and 2004, we spent \$9.0 million, \$7.4 million and \$5.5 million, respectively, on research and development.

Intellectual Property

We rely on a combination of patent, trademark, copyright and trade secret laws in the United States and other jurisdictions together with confidentiality procedures and contractual provisions to protect our proprietary technology and our brand. We seek patent protection on inventions that we consider important to the development of our business. We control access to our proprietary technology and enter into confidentiality and invention assignment agreements with our employees and consultants and confidentiality agreements with other third parties.

Our success depends in part on our ability to develop patentable products and obtain, maintain and enforce patent and trade secret protection for our products, including successfully defending these patents against any third-party challenges, both in the United States and in other countries. We may be able to protect our technologies from unauthorized use by third parties to the extent that we own or have licensed valid and enforceable patents or trade secrets that cover them. However, the degree of future protection of our proprietary rights is uncertain because legal means afford only limited protection and may not adequately protect our rights or permit us to gain or keep our competitive advantage.

Currently, we own one issued U.S. patent and have twelve U.S. and foreign patent applications pending, and we intend to file, or request that our licensors file, additional patent applications for patents covering our products. However, patents may not be issued for any pending or future pending patent applications owned by or licensed to us, and claims allowed under any issued patent or future issued patent owned or licensed by us may not be valid or sufficiently broad to protect our technologies. Any issued patents owned by or licensed to us now or in the future may be challenged, invalidated, held unenforceable or circumvented, and the rights under such patents may not provide us with the expected benefits. In addition, competitors may design around our technology or develop competing technologies. Intellectual property rights may also be unavailable or limited in some foreign countries, which could make it easier for competitors to capture or increase their market share with respect to related technologies. Although we are not currently involved in any legal proceedings related to intellectual property, we could incur substantial costs to defend ourselves in suits brought against us or in suits in which we may assert our patent rights against others. An unfavorable outcome in any such litigation could have a material adverse effect on our business and results of operations.

In addition to patent and trade secret protection, we also rely on several trademarks and service marks to protect our intellectual property assets. We are the owner of numerous trademarks and service marks and service marks and in certain other countries to establish and protect our brand names as part of our intellectual property strategy. Some of our registered marks include comScore, Media Metrix and MyMetrix.

Our intellectual property policy is to protect our products, technology and processes by asserting our intellectual property rights where we believe it is appropriate and prudent. Any pending or future pending patent applications owned by or licensed to us (in the United States or abroad) may not be allowed or may in the future be challenged, invalidated, held unenforceable or circumvented, and the rights under such patents may not provide us with competitive advantages. Any significant impairment of our intellectual property rights could harm our business or our ability to compete. Protecting our intellectual property rights is costly and time consuming. Any increase in the unauthorized use of our intellectual property could make it more expensive to do business and harm our operating results.

There is always the risk that third parties may claim that we are infringing upon their intellectual property rights and, if successful in proving such claims, we could be prevented from selling our products.

For additional, important information related to our intellectual property, please review the information set forth in "Risk Factors — Risks Related to Our Business, Our Technologies and Our Industry."

Competition

The market for digital marketing intelligence is highly competitive and evolving rapidly. We compete primarily with providers of digital marketing intelligence and related analytical products and services. We also compete with providers of marketing services and solutions, with survey providers, as well as with internal solutions developed by customers and potential customers. Our principal competitors include:

- large and small companies that provide data and analysis of consumers' online behavior, including Compete Inc., Hitwise Pty. Ltd and NetRatings, Inc.;
- online advertising companies that provide measurement of online ad effectiveness, including aQuantive, Inc., DoubleClick Inc., ValueClick Inc., and WPP Group plc;
- companies that provide audience ratings for TV, radio and other media that have extended or may extend their current services, particularly in certain international markets, to the measurement of digital media, including Arbitron Inc., Nielsen Media Research, Inc. and Taylor Nelson Sofres plc;
- analytical services companies that provide customers with detailed information of behavior on their own Web sites, including Omniture, Inc., WebSideStory, Inc. and WebTrends Corporation;
- full-service market research firms and survey providers that may measure online behavior and attitudes, including Harris Interactive Inc., Ipsos Group, Taylor Nelson Sofres plc and The Nielsen Company; and
- specialty information providers for certain industries that we serve, including IMS Health Incorporated (healthcare) and Telephia, Inc. (telecommunications).

Some of our current competitors have longer operating histories, relationships with more customers and substantially greater resources than we do. As a result, these competitors may be able to devote more resources to marketing and promotional campaigns, panel retention and development techniques or technology and systems development than we can. In addition, some of our competitors may be able to adopt more aggressive pricing policies. Furthermore, large software companies, Internet portals and database management companies may enter the market or enhance their current offerings, either by developing competing services or by acquiring our competitors, and could leverage their significant resources and pre-existing relationships with our current and potential customers.

We believe the principal competitive factors in our markets include the following:

- the ability to provide actual and perceived high-quality, accurate and reliable data regarding Internet and other digital media audience behavior and activity in a timely manner, including the ability to maintain a large and statistically representative sample panel;
- · the ability to adapt product offerings to emerging digital media technologies and standards;
- · the breadth and depth of our products and their flexibility and ease of use;
- · the availability of data across various industry verticals and geographic areas and our expertise across these verticals and in these geographic areas;
- · the ability to offer survey-based information combined with digital media usage, eCommerce data and other online information collected from panelists;
- the ability to offer high-quality analytical services based on Internet and other digital media audience measurement information;
- the ability to offer products that meet the changing needs of customers and provide high-quality service; and
- the prices that are charged for products based on the perceived value delivered.

We believe that we compete favorably with our competitors on the basis of these factors. However, if we are unable to compete successfully against our current and future competitors, we may not be able to acquire

and retain customers, and we may consequently experience a decline in revenues, reduced operating margins, loss of market share and diminished value from our products.

Employees

As of December 31, 2006, we had 377 employees. None of our employees is represented by a labor union. We have experienced no work stoppages and believe that our employee relations are good.

Legal

Generally, we are involved in various legal proceedings arising from the normal course of business activities. Currently, we do not believe that resolution of these matters will have a material adverse impact on our consolidated results of operations, cash flows or our financial position. However, depending on the amount and timing, an unfavorable resolution of a matter could materially affect our future results of operations, cash flows or financial position in a particular period.

Facilities

Our corporate headquarters and executive offices are located in Reston, Virginia, where we occupy approximately 34,000 square feet of office space under a lease that expires in June 2008. We also lease space in various locations throughout the United States and in Toronto and London for sales and other personnel. If we require additional space, we believe that we would be able to obtain such space on commercially reasonable terms.

MANAGEMENT

Executive Officers and Directors

The following table sets forth certain information concerning our current executive officers and directors:

Name	Age	Position(s)
Executive Officers		
Magid M. Abraham, Ph.D.	48	President, Chief Executive Officer and Director
Gian M. Fulgoni	59	Executive Chairman of the Board of Directors
John M. Green	55	Chief Financial Officer
Gregory T. Dale	37	Chief Technology Officer
Christiana L. Lin	37	General Counsel and Chief Privacy Officer
Non-Employee Directors:		
Thomas D. Berman(1)(2)	49	Director
Bruce Golden(3)	48	Director
William J. Henderson(2)(3)	59	Director
Ronald J. Korn(1)(3)	66	Director
Frederick R. Wilson(1)(2)	45	Director

- (1) Member of the audit committee.
- (2) Member of the compensation committee.
- (3) Member of the nominating and governance committee.

Magid M. Abraham, Ph.D., one of our co-founders, has served as President, Chief Executive Officer and Director since September 1999. In 1995, Dr. Abraham founded Paragren Technologies, Inc., which specialized in delivering large scale Customer Relationship Marketing systems for strategic and target marketing, and served as its Chief Executive Officer from 1995 to 1999. Prior to founding Paragren, Dr. Abraham was employed by Information Resources, Inc. from 1985 until 1995, where he was President and Chief Operating Officer from 1993 to 1994 and later Vice Chairman of the Board of Directors from 1994 until 1995. Since May 2006, Dr. Abraham has also been a member of the board of directors of ES3, LLC, a storage and logistics services company. Dr. Abraham received the Paul Green Award in 1996 and the William F. O'Dell Award in 2000 from the American Marketing Association for a 1995 article that he co-authored in the Journal of Marketing Research. He received a Ph.D. in Operations Research and an M.B.A. from MIT. He also holds an Engineering degree from the École Polytechnique in France.

Gian M. Fulgoni, one of our co-founders, has served as Executive Chairman of the Board of Directors since September 1999. Prior to co-founding comScore, Mr. Fulgoni was employed by Information Resources, Inc., where he served as President from 1981 to 1989, Chief Executive Officer from 1986 to 1998 and Chairman of the Board of Directors from 1991 until 1995. Mr. Fulgoni has served on the board of directors of PetMed Express, Inc. since 2002 and previously served from August 1999 through November 2000. Mr. Fulgoni also serves on the board of directors of INXPO, LLC, an Illinois-based provider of virtual events, since July 2005. He also served on the board of directors of Platinum Technology, Inc. from 1990 to 1999, U.S. Robotics, Inc. from 1991 to 1994, and Yesmail.com, Inc. from 1999 to 2000. Mr. Fulgoni has twice been named an Illinois Entrepreneur of the Year. In 1992, he received the Wall Street Transcript Award for outstanding contributions as Chief Executive Officer of Information Resources, Inc. in enhancing the overall value of that company to the benefit of its shareholders. Educated in the United Kingdom, Mr. Fulgoni holds an M.A. in Marketing from the University of Lancaster and a B.Sc. in Physics from the University of Manchester.

John M. Green has served as Chief Financial Officer since May 2006. Prior to joining comScore, Mr. Green served as the Chief Financial Officer and U.S. Services Business Leader for BioReliance, a subsidiary of Invitrogen Corporation, from 2004 to March 2006. Prior to joining BioReliance, Mr. Green

served as the General Manager, Business Integrations at Invitrogen from September 2003 to April 2004. From March 2001 through August 2003, Mr. Green served as the Chief Financial Officer for InforMax, and as its Chief Operating Officer from October 2001 until the sale of InforMax and integration into Invitrogen in August 2003. Prior to 2001, Mr. Green held several financial and operating management roles, including serving as Executive Vice President of Operations at HMSHost Corporation, Senior Vice President of Finance and Corporate Controller at Marriott International Incorporated and Director of Business Planning and Director of Finance, Central Europe, at PepsiCo, Inc. Mr. Green received an M.Sc. in Economics from The London School of Economics and a B.A. in Political Science/International Relations from Tufts University.

Gregory T. Dale has served as Chief Technology Officer since October 2000. Prior to that, he served as Vice President, Product Management starting in September 1999. Prior to joining us, he served as Vice President of Client Service at Paragren Technologies, Inc., a company that specialized in enterprise relationship marketing. He holds a B.S. in Industrial Management from Purdue University.

Christiana L. Lin has served as General Counsel and Chief Privacy Officer since January 2006. Prior to that, she served as our Corporate Counsel and Chief Privacy Officer starting in March 2003. Prior to that, she served as our Deputy General Counsel starting in February 2001. Ms. Lin holds a J.D. from the Georgetown University Law Center and a B.A. in Political Science from Yale University.

Thomas D. Berman has served as a director since August 2001. Mr. Berman is a partner with Adams Street Partners, where he has led investments in information technology and business services companies since 1990. He served on the board of directors of PathScale, Inc. from May 2004 to April 2006 and has served on the board of directors of Adams Harris, Inc. since March 2006. Mr. Berman holds an S.B. in Electrical Engineering from MIT and an S.M. from the Sloan School of Management at MIT.

Bruce Golden has served as a director since June 2002. He is a partner at Accel Partners, which he joined in 1997. Mr. Golden has led a number of investments in enterprise software and Internet-related companies while at Accel and currently serves as a member of the boards of directors at several private companies. He holds an M.B.A. from Stanford University and a B.A. from Columbia University.

William J. Henderson has served as a director since August 2001. Mr. Henderson was the 71st Postmaster General of the United States. He served in that position from May 1998 until his retirement in May 2001. Mr. Henderson also served as the Chief Operations Officer of Netflix, Inc. from January 2006 until February 2007. Mr. Henderson also currently serves on the board of directors of Acxiom Corporation, where he has been a director since June 2001. Mr. Henderson holds a B.S. from the University of North Carolina at Chapel Hill and served in the U.S. Army.

Ronald J. Korn has served as a director since November 2005. Since 1991, he has served as the President of Ronald Korn Consulting, which provides business and marketing services. Mr. Korn served as a director, chairman of the audit committee, and member of the loan committee of Equinox Financial Corporation from 1999 until its acquisition in October 2005. Since 2002, he has served as a director, chairman of the audit committee and a member of the compensation and nominating and governance committees of PetMed Express, Inc. and since July 2003, he has served as a director, chairman of the audit committee and a member of the compensation committee of Ocwen Financial Corporation. Prior to that, Mr. Korn was a partner and employee of KPMG, LLP, from 1961 to 1991, where he was the managing partner of KPMG's Miami office from 1985 until 1991. Mr. Korn holds a B.S. from the University of Pennsylvania, Wharton School and a J.D. from New York University Law School.

Frederick R. Wilson has served as a director since August 1999. He has served as managing partner of Union Square Ventures since August 2003. He is also a managing partner of Flatiron Partners and has held that position since August 1996. He holds an M.B.A. from the Wharton School of Business at the University of Pennsylvania and an S.B. in Mechanical Engineering from MIT.

Board Composition

Upon completion of this offering, our directors will be divided into three classes serving staggered three-year terms. Class I, Class II and Class III directors will serve until our annual meetings of stockholders in

2008, 2009 and 2010, respectively. Upon expiration of the term of class of directors, directors in that class will be eligible to be elected for a new three-year term at the annual meeting of stockholders in the year in which their term expires. This classification of directors could have the effect of increasing the length of time necessary to change the composition of a majority of our board of directors. In general, at least two annual meetings of stockholders will be necessary for stockholders to effect a change in a majority of the members of our board of directors.

Our board of directors currently consists of seven members. Messrs. Abraham, Berman and Wilson are Class I directors and will serve for one year. Messrs. Henderson and Korn are Class II directors and will serve for two years. Messrs. Fulgoni and Golden are Class III directors and will serve for three years.

Board Committee

Our board of directors has established an audit committee, a compensation committee and a nominating and governance committee.

Audit Committee

Our audit committee consists of Messrs. Berman, Korn and Wilson, with Mr. Korn serving as chairman. Our audit committee oversees our corporate accounting and financial reporting process and internal controls over financial reporting. Our audit committee evaluates the independent registered public accounting firm's qualifications, independence and performance; engages and provides for the compensation of the independent registered public accounting firm; approves the retention of the independent registered public accounting firm to perform any proposed permissible non-audit services; reviews our consolidated financial statements; reviews our critical accounting policies and estimates and internal controls over financial reporting; and discusses with management and the independent registered public accounting firm the results of the annual audit and the reviews of our quarterly consolidated financial statements. We believe that our audit committee members meet the requirements for independence and financial literacy under the current requirements of the Sarbanes-Oxley Act of 2002, The NASDAQ Global Market and SEC rules and regulations. In addition, the board of directors has determined that Mr. Korn is qualified as an audit committee financial expert within the meaning of SEC regulations. We have made this determination based on information received by our board of directors, including questionnaires provided by the members of our audit committee. We believe that our audit committee complies with the applicable requirements of the Sarbanes-Oxley Act of 2002, The NASDAQ Global Market and SEC rules and regulations. We intend to comply with future requirements to the extent they become applicable to us. We have adopted an audit committee charter. We expect that the committee will meet no less frequently than quarterly. Our audit committee will meet no less frequently than quarterly. Our audit committee will meet no less

Compensation Committee

Our compensation committee consists of Messrs. Berman, Henderson and Wilson, with Mr. Henderson serving as chair. Our compensation committee reviews and recommends policy relating to compensation and benefits of our officers and employees, including reviewing and approving corporate goals and objectives relevant to compensation of the Chief Executive Officer and other senior officers, evaluating the performance of these officers in light of those goals and objectives and setting compensation of these officers based on such evaluations. The compensation committee also administers the issuance of stock options and other awards under our stock plans. We believe that the composition of our compensation committee meets the requirements for independence under, and the functioning of our compensation committee complies with, any applicable requirements of the Sarbanes-Oxley Act of 2002, The NASDAQ Global Market and SEC rules and regulations. We intend to comply with future requirements to the extent they become applicable to us. We have adopted a compensation committee charter. We expect that the committee will meet at least once a year. Our compensation committee has previously met on an annual basis to review key compensation decisions.

Nominatina and Governance Committee

Our nominating and governance committee consists of Messrs. Golden, Henderson and Korn, with Mr. Golden serving as chairman, each of whom the board of directors has determined is an independent director under the rules of The NASDAQ Global Market. The nominating and governance committee recommends to the board of directors nominees for election as directors, and meets as necessary to review director candidates and nominees for election as directors.

Code of Business Conduct and Ethics

Our board of directors has adopted a code of business conduct and ethics, which establishes the standards of ethical conduct applicable to all directors, officers and employees of our company. The code addresses, among other things, conflicts of interest, compliance with disclosure controls and procedures and internal controls over financial reporting, corporate opportunities and confidentiality requirements. The audit committee is responsible for applying and interpreting our code of business conduct in situations where questions are presented to the committee.

Compensation Committee Interlocks and Insider Participation

None of the members of our compensation committee is an executive officer or employee of our company. None of our executive officers serves as a member of the compensation committee of any entity that has one or more executive officers serving on our compensation committee.

Director Compensation

None of our non-employee directors are currently compensated for service on the board of directors. We do, however, reimburse director expenses for attending meetings of the board of directors.

We previously granted equity awards for the purchase of our common stock to two of our present non-employee directors, William Henderson and Ronald Korn, upon their initial appointment to our board of directors. A warrant to purchase 100,000 shares of our common stock at an exercise price of \$1.00 per share was issued on June 26, 2001 to Mr. Henderson, Such warrant shall terminate on the earlier of (i) June 26, 2011; (ii) the completion of this offering; or (iii) a change of control as defined in the warrant. In addition, Mr. Henderson was previously granted stock options for the purchase of 30,000 shares of our common stock at an exercise price of \$0.50 per share on April 9, 2002 and for the purchase of 50,000 shares of our common stock at an exercise price of \$0.90 per share on December 27, 2005. Mr. Korn was awarded stock options for the purchase of 100,000 shares of our common stock at an exercise price of \$0.85 per share on November 25, 2005. The warrant for the purchase of 100,000 shares of our common stock granted to Mr. Henderson, the stock options for the purchase of 80,000 shares of common stock granted to Mr. Henderson and the stock option for the purchase of 100,000 shares of common stock granted to Mr. Korn remain outstanding as of December 31, 2006.

Upon the closing of this offering, our non-employee directors will be entitled to an annual grant of restricted stock having a value of \$50,000 at the time of the grant. Non-employee directors will also be paid an annual cash retainer of \$25,000 for serving on our board of directors, an additional annual cash retainer of \$10,000 for serving as the chairman of our audit committee and \$7,500 for serving as the chair of our compensation committee.

Our non-employee directors did not receive any compensation for their services as directors in 2006.

Limitations on Director and Officer Liability and Indemnification

Our amended and restated certificate of incorporation as will be in effect upon completion of this offering limits the liability of our directors to the maximum extent permitted by Delaware law. Delaware law provides

that directors of a corporation will not be personally liable for monetary damages for breach of their fiduciary duties as directors, except liability for:

- · any breach of their duty of loyalty to the corporation or its stockholders;
- · acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law;
- · unlawful payments of dividends or unlawful stock repurchases or redemptions; or
- · any transaction from which the director derived an improper personal benefit.

Our amended and restated certificate of incorporation and our amended and restated bylaws will provide that we are required to indemnify our directors and officers, in each case to the fullest extent permitted by Delaware law. Any repeal of or modification to our amended and restated certificate of incorporation and our amended and restated bylaws may not adversely affect any right or protection of a director or officer for or with respect to any acts or omissions of such director or officer occurring prior to such amendment or repeal. Our bylaws will also provide that we shall advance expenses incurred by a director or officer in advance of the final disposition of any action or proceeding, and permit us to secure insurance on behalf of any officer, director, employee or other agent for any liability arising out of his or her actions in connection with their services to us, regardless of whether our bylaws permit such indemnification.

We have entered into separate indemnification agreements with our directors and executive officers, in addition to the indemnification provided for in our bylaws. These agreements, among other things, provide that we will indemnify our directors and executive officers for certain expenses (including attorney's fees), judgments, fines, penalties and settlement amounts incurred by a director or executive officer in any action or proceeding arising out of such person's services as one of our directors or executive officers, or any other company or enterprise to which the person provides services at our request. We believe that these provisions and agreements are necessary to attract and retain qualified persons as directors and executive officers.

The limitation of liability and indemnification provisions that will be contained in our amended and restated certificate of incorporation and our amended and restated bylaws may discourage stockholders from bringing a lawsuit against our directors for breach of their fiduciary duty. They may also reduce the likelihood of derivative litigation against our directors and officers, even though an action, if successful, might benefit us and other stockholders. Further, a stockholder's investment may be adversely affected to the extent that we pay the costs of settlement and damage awards against directors and officers as required by these indemnification provisions. There is no pending litigation or proceeding involving one of our directors or executive officers as to which indemnification is required or permitted and we are not aware of any threatened litigation or proceeding that may result in a claim for indemnification.

COMPENSATION DISCUSSION AND ANALYSIS

Our Philosophy

The objective of our compensation programs for employees is to retain and attract top talent. The plans are designed to reward, motivate and align employees to achieve business results and to reinforce accountability. In determining the compensation of senior executives, we are guided by the following key principles:

- · Competition. Compensation should reflect the competitive marketplace, so we can retain, attract, and motivate talented executives.
- Accountability for Business Performance. Compensation should be tied, in part, to financial performance, so that executives are held accountable through their compensation for contributions to the performance of the company as a whole through the performance of the businesses for which they are responsible.
- Accountability for Individual Performance. Compensation should be tied, in part, to the individual's performance to encourage and reflect individual contributions to our
 performance. Our board of directors considers individual performance as well as performance of the businesses and responsibility areas that an individual oversees, and weights
 these factors as appropriate in assessing a particular individual's performance.
- Alignment with Stockholder Interests. Compensation should be tied, in part, to our financial performance through equity awards to align executives' interests with those of our stockholders.
- Independence. An independent committee of our board of directors should be, and is, responsible for reviewing and establishing the compensation for our Chief Executive
 Officer and Executive Chairman, and for reviewing and approving the compensation recommendations made by our Chief Executive Officer for all of our other executive
 officers.

Application of our Philosophy

Our executive compensation and benefit program aims to encourage our management team to continually pursue our strategic opportunities while effectively managing the risks and challenges inherent to our business. Specifically, we have created an executive compensation package that balances short versus long-term components, cash versus equity elements, and fixed versus contingent payments, in ways we believe are most appropriate to incentivize our senior management and reward them for achieving the following goals:

- · develop a culture that embodies a passion for our business, creative contribution and a drive to achieve established goals and objectives;
- provide leadership to the organization in such a way as to maximize the results of our business operations;
- · lead us by demonstrating forward thinking in the operation, development and expansion of our business;
- effectively manage organizational resources to derive the greatest value possible from each dollar invested; and
- take strategic advantage of the market opportunity to expand and grow our business.

Our executive compensation structure not only aims to be competitive in our industry, but also to be fair relative to compensation paid to other professionals within our organization, relative to our short and long-term performance and relative to the value we deliver to our stockholders. We seek to maintain a performance-oriented culture and a compensation approach that rewards our executive officers when we achieve our goals and objectives, while putting at risk an appropriate portion of their compensation against the possibility that our goals and objectives may not be achieved. Overall, our approach is designed to relate the compensation of our executive officers to: the achievement of short and longer term goals and objectives; their willingness to challenge and improve existing policies and structures; and their capability to take advantage of unique opportunities and overcome difficult challenges within our business.

Role of Our Compensation Committee

Our compensation committee approves, administers and interprets our executive compensation and benefit policies, including our 1999 Stock Plan, our 2007 Equity Incentive Plan and our short-term compensation, long-term incentives and benefits programs. Our compensation committee is appointed by our board of directors, and consists entirely of directors who are "outside directors" for purposes of Section 162(m) of the Internal Revenue Code and "non-employee directors" for purposes of Rule 16b-3 under the Exchange Act. Our compensation committee is comprised of Messrs. Berman, Henderson and Wilson, and is chaired by Mr. Henderson.

Our compensation committee reviews and makes recommendations to our board of directors to ensure that our executive compensation and benefit program is consistent with our compensation philosophy and corporate governance guidelines and, subject to the approval of our board of directors, is responsible for establishing the executive compensation packages offered to our executive officers. We believe our executives' base salary, target annual bonus levels and long-term incentive award values are set at competitive levels.

Our compensation committee has taken the following steps to ensure that our executive compensation and benefit program is consistent with both our compensation philosophy and our corporate governance guidelines:

- · regularly reviewed the performance of and the total compensation earned by or awarded to our Chief Executive Officer and Executive Chairman independent of input from them;
- examined on an annual basis the performance of our other named executive officers and other key employees with assistance from our Chief Executive Officer and Executive
 Chairman, and approved compensation packages that are competitive in the marketplace; and
- · regularly held executive sessions of the compensation committee meeting without management present,

Components of our Executive Compensation Program.

Our executive compensation program consists of three components: short-term compensation (including base salary and annual performance bonuses), long-term incentives and benefits.

Short-term Compensation

We utilize short-term compensation, including base salary, annual adjustments to base salary and annual performance bonuses, to motivate and reward our key executives in accordance with our performance-based program. Each individual's short-term compensation components are tied to an annual assessment of his or her progress against established objectives.

Base salary is used to recognize the experience, skills, knowledge and responsibilities required of each executive officer, as well as competitive market conditions. In establishing the 2007 base salaries of the executive officers, our compensation committee and management took into account a number of factors, including the executive's seniority, position and functional role, level of responsibility and, to the extent such individual was employed by us for at least the prior six months, his or her accomplishments against personal and group objectives. For newly hired executives, we consider the base salary of the individual at his or her prior employment, any unique personal circumstances that motivated the executive to leave that prior position to join us and the compensation range for the particular role being filled. In addition, we consider the competitive market for corresponding positions within comparable geographic areas and industries.

The base salary of our executive officer group is reviewed on an annual basis and adjustments are made to reflect performance-based factors, as well as competitive conditions. Increases are considered within the context of our overall annual merit increase structure as well as individual and market competitive factors. We do not apply specific formulas to determine increases. Generally, executive officer salaries are adjusted effective the first quarter of each year based on a review of:

• their achievement of specific objectives established during the prior review;

- an assessment of their professional effectiveness, consisting of a portfolio of competencies that include leadership, commitment, creativity and organizational
 accomplishment; and
- their knowledge, skills and attitude, focusing on capabilities, capacity and the ability to drive results.

Annual performance bonuses for our executive officers are tied to the achievement of our annual company goals and objectives, functional area goals, and/or individual performance objectives. We set clearly defined goals for each executive officer; with an emphasis on quantifiable and achievable targets. A portion of each executive officer's bonus is clearly tied to the achievement of specific targets relative to the performance of the particular business segment or functional area for which they are responsible, with the remainder tied to similar targets relative to our overall financial performance. Individual awards under the program are based on a thorough review of the applicable performance results of the company, business, function or individual as compared to the applicable goals. Each individual may receive an award from zero to 100% of his or her target bonus based on the review.

Magid Abraham, our Chief Executive Officer, periodically reviews the performance of our executive officers on the basis noted above and recommends to the compensation committee any base salary changes or bonuses deemed appropriate.

For the 2005 and 2006 performance measurement years, executive bonuses were paid out in one installment during the month of February following the measurement year.

Long-term Compensation

Our long-term compensation program has historically consisted solely of stock options. Option grants made to executive officers are designed to provide them with incentive to execute their responsibilities in such a way as to generate long-term benefit to us and our stockholders. Through possession of stock options, our executives participate in the long-term results of their efforts, whether by appreciation of our company's value or the impact of business setbacks, either company-specific or industry based. Additionally, stock options provide a means of ensuring the retention of key executives, in that they are in almost all cases subject to vesting over an extended period of time.

Stock options are granted periodically, and are subject to vesting based on the executive's continued employment. Most options vest evenly over four years, beginning on the date of the grant. A portion of options granted to our executives vest according to defined performance milestones rather than solely based on time.

Upon joining us, each executive is granted an initial option award that is primarily based on competitive conditions applicable to the executive's specific position. In addition, the compensation committee considers the number of options owned by other executives in comparable positions within our company. We believe this strategy is consistent with the approach of other companies at the same stage of development in our industry and, in our compensation committee's view, is appropriate for aligning the interests of our executives with those of our stockholders over the long term.

Periodic awards to executive officers are made based on an assessment of their sustained performance over time, their ability to impact results that drive value to our stockholders and their organization level. Equity awards are not granted regularly or automatically to our executives on an annual basis. Magid Abraham, our Chief Executive Officer, periodically reviews the performance of our executive officers on the basis noted above and recommends to the compensation committee any equity awards deemed appropriate. The compensation committee reviews any such recommendations and presents them to our board of directors for approval, if appropriate.

During 2006, our board of directors granted stock options based upon the recommendations of our compensation committee. These grants were generally made during regularly scheduled board meetings. The exercise price of options was determined by our board of directors after taking into account a variety of factors, including the quality and growth of our management team and specific and general market comparables within our industry. In addition, our board of directors took into account the valuation opinion of our outside consultant, who provided valuations of our common stock at the end of each calendar quarter.

On March 18, 2007, we awarded an aggregate of 1,180,000 shares of restricted stock to our named executive officers based upon the recommendations of our compensation committee, taking into account the factors described above. Beginning in 2007, we expect to increase our use of restricted stock awards and reduce our use of stock options as a form of stock-based compensation.

Benefits

We provide the following benefits to our executive officers on the same basis as the benefits provided to all employees:

- · health and dental insurance;
- life insurance;
- · short-and long-term disability; and
- 401(k) plan

These benefits are consistent with those offered by other companies and specifically with those companies with which we compete for employees.

Our Competitive Market

The market for experienced management with the knowledge, skills and experience our organization requires is highly competitive. Our objective is to attract and retain the most highly qualified executives to manage each of our business functions. In doing so, we draw upon a pool of talent that is highly sought after by other companies in our industry and those industries that also produce the requisite skills we seek.

We believe that our ability to offer significant upside potential through restricted stock and/or other equity instruments gives us a competitive advantage. Nonetheless, we must also offer cash compensation to our existing and prospective employees through base salaries and cash bonuses that are competitive in the marketplace and allow them to satisfy their day to day financial requirements.

We also compete on the basis of our vision of future success, our culture and company values and the excellence of our management personnel. In all of these areas, we compete with other market research and technology companies.

Total Compensation

We intend to continue our strategy of compensating our named executive officers at competitive levels, with the opportunity to impact their total annual compensation through performance-based incentive programs that include both cash and equity elements. Our approach to total executive compensation is designed to drive results that maximize our financial performance and deliver value to our stockholders. In light of our compensation philosophy, we believe that the total compensation package for our executives should continue to consist of base salary, annual cash performance bonus and long-term equity-based incentives.

Evolution of our Compensation Approach

Our compensation approach is necessarily tied to our stage of development as a company. Accordingly, the specific direction, emphasis and components of our executive compensation program will continue to evolve as our company and its underlying business strategy continue to grow and develop. For example, we intend to reduce our executive compensation program's emphasis on stock options as a long-term incentive component in favor of other forms of equity compensation such as restricted stock awards. Similarly, we may revise how we measure senior executive performance to take into account the unique requirements of being a public company, including, but not limited to, strict compliance with the standards of the Sarbanes Oxley Act. In addition, we may engage a compensation consultant to assist our compensation committee in continuing to evolve our executive compensation program, and we may look to programs implemented by comparable public companies in refining our compensation approach.

EXECUTIVE COMPENSATION

Summary Compensation Table

The following table sets forth the summary information concerning compensation during 2006 for the following persons: (i) our chief executive officer, (ii) our current chief financial officer and any individual serving as our chief financial officer during 2006 and (iii) the three most highly compensated of our other executive officers who received compensation during 2006 of at least \$100,000 and who were executive officers on December 31, 2006. We refer to these persons as our "named executive officers" elsewhere in this prospectus.

Name and Principal Position	<u>Year</u>	Salary	Bonus	Option Awards	All Other Compensation	Total
Magid M. Abraham, Ph.D.	2006	\$ 297,612	\$ 117,273	_	\$ 3,072(1)	\$ 417,957
President, Chief Executive Officer and Director						
John M. Green	2006	156,731	47,019	\$ 87,366	42(2)	291,158
Chief Financial Officer						
Gian M. Fulgoni	2006	281,635	111,409	_	3,072(1)	396,116
Executive Chairman of the Board						
of Directors						
Gregory T. Dale	2006	222,115	44,423	_	3,072(1)	269,610
Chief Technology Officer						
Christiana L. Lin	2006	149,077	29,815	_	2,173(3)	181,065
General Counsel and Chief						
Privacy Officer						
Sheri Huston	2006	60,772	_	_	141,345(4)	202,117
Former Chief Financial Officer						

- (1) Includes discretionary matching contributions of \$3,000 each by us to Dr. Abraham's, Mr. Fulgoni's and Mr. Dale's respective 401(k) plan accounts and payment of life insurance premiums on behalf of each officer.
- (2) Represents life insurance premium paid by us on behalf of Mr. Green.
- (3) Includes discretionary matching contributions of \$2,000 by us to the Ms. Lin's 401(k) plan account and payment of life insurance premiums on behalf of Ms. Lin.
- (4) Includes payment of health insurance premiums for six months, payment of unused accrued vacation, discretionary matching contributions by us to the officer's 401(k) account, payment of life insurance premiums on behalf of the officer and a severance payment equivalent to six months salary.

Grants of Plan-Based Awards

Our board of directors approved awards under our 1999 Stock Plan to several of our named executive officers in 2006. See "Benefit Plans — 1999 Stock Plan" for more detail regarding these options.

The following table sets forth certain information concerning grants of plan-based awards to named executive officers in 2006:

Name	Grant Date	All Other Option Awards: Number of Securities Underlying Options	 Exercise or Base Price per Share of Option Awards
Magid M. Abraham, Ph.D.	_	_	_
President, Chief Executive Officer and Director			
John M. Green	5/9/2006	650,000(1)	\$ 1.50
Chief Financial Officer			
Gian M. Fulgoni	_	_	_
Executive Chairman of the Board of Directors			
Gregory T. Dale	_	_	_
Chief Technology Officer			
Christiana L. Lin	_	_	_
General Counsel and Chief Privacy Officer			
Sheri Huston	_	_	_
Former Chief Financial Officer			

⁽¹⁾ 1/48th of the total number of shares subject to option vest monthly.

Outstanding Equity Awards at December 31, 2006

	Option Awards				
<u>N</u> ame	Number of Se Underlyi Unexercised Exercisable	ng	Equity Incentive Plan Awards: Number of Securities Underlying Unexercised Unearned Options	Option Exercise Price	Option Expiration Date
Dr. Magid M. Abraham President, Chief Executive Officer and Director	1,083,465(1)	_	1,622,030(1)	\$ 0.05	12/16/2013
John M. Green Chief Financial Officer	81,248(2)	568,752(2)	_	1.50	5/9/2016
Gian M. Fulgoni Executive Chairman of the Board of Directors	_	_	1,166,725(3)	0.05	12/16/2013
Gregory T. Dale	170,633	_	_	0.05	4/28/2014
Chief Technology Officer	125	_	_	0.05	4/28/2014
	59,896	_	_	0.05	4/28/2014
	349	_	_	0.05	4/28/2014
	90,625	_	_	0.05	4/28/2014
	99,998(2)	50,002(2)	_	0.05	4/28/2014
	91,665(2)	108,335(2)	_	0.49	2/2/2015
	18,749(2)	56,251(2)	_	0.90	12/28/2015
Christiana L. Lin	5,417	_	_	0.05	4/28/2014
General Counsel and Chief	5,834	_	_	0.05	4/28/2014
Privacy Officer	21,879(4)	6,245(4)	_	0.05	4/28/2014
	25,402(2)	19,356(2)	_	0.05	4/28/2014
	12,499(2)	37,501(2)	_	0.90	12/28/2015
Sheri Huston Former Chief Financial Officer	_	_	_	_	_

- (1) Vesting for Dr. Abraham's option grant is based on the following milestones related to our performance, the completion of each shall be determined by a good faith determination of our board of directors:
 - 581,633 shares vested when we first achieved an EBITDA greater than \$0 for a full fiscal quarter;
 - \bullet 581,633 shares vested when we first achieved revenues of \$40 million or greater for a twelve month period;
 - 520,199 shares vested when we first achieved revenues of \$50 million or greater for a twelve month period;
 - $\bullet \quad 581,633 \text{ shares shall vest when we first achieve net income of greater than 0 for a twelve month period;}\\$
 - 520,199 shares shall vest when we first achieve pretax net income of \$5 million or greater for a twelve month period; and
 - 520,198 shares shall vest when we first achieve pretax net income of \$10 million or greater for a twelve month period.

 Any unvested shares remaining under the option shall vest on the earlier of (i) December 16, 2009 or (ii) the consummation of a change in control, provided that Dr. Abraham remains a service provider to us. Dr. Abraham has since exercised part of this option award for a total of 600,000 shares.
- (2) 1/48th of the total number of shares subject to option vest monthly.
- (3) Vesting for Mr. Fulgoni's option grant is based on the following milestones related to our performance, the completion of each shall be determined by a good faith determination of our board of directors:
 - 418,367 shares vested when we first achieved an EBITDA greater than \$0 for a full fiscal quarter;
 - 418,367 shares vested when we first achieved revenues of \$40 million or greater for a twelve month period;
 - 374,178 shares vested when we first achieved revenues of \$50 million or greater for a twelve month period;
 - 418,367 shares shall vest when we first achieve net income of greater than \$0 for a twelve month period;
 - 374,178 shares shall vest when we first achieve pretax net income of \$5 million or greater for a twelve month period; and
 - 374,178 shares shall vest when we first achieve pretax net income of \$10 million or greater for a twelve month period.

 Any unvested shares remaining under the option shall vest on the earlier of (i) December 16, 2009 or (ii) the consummation of a change in control, provided that Mr. Fulgoni remains a service provider to us. Mr. Fulgoni has since exercised part of this option award for a total of 1,210,912 shares.
- (4) 1/38th of the total number of shares subject to option vest monthly.

Option Exercises and Stock Vested Table

The following table presents certain information concerning the exercise of options by each of the named executive officers during the fiscal year ended December 31, 2006.

There was no public trading market for our common stock at the time of exercise of the options listed below. The values realized on exercise have been calculated based on the initial public offering price of \$\(\) , less the applicable exercise price.

	Option Awards		
Name	Number of Shares Acquired on Exercise	Value Realized on Exercise	
Magid M. Abraham Ph.D.	_	_	
President, Chief Executive Officer and Director			
John M. Green	_	_	
Chief Financial Officer			
Gian M. Fulgoni	836,734	\$	
Executive Chairman of the Board of Directors	374,178		
Gregory T. Dale	_	_	
Chief Technology Officer			
Christiana L. Lin	_	_	
General Counsel and Chief Privacy Officer			
Sheri Huston	114,581		
Former Chief Financial Officer	166,666		
	114,574		
	114,577		

Employment Agreements and Potential Payments upon Termination or Change-In-Control

We currently do not have an employment agreement with any of our named executive officers. We have offer letter agreements with Gregory T. Dale, our Chief Technology Officer, Christiana L. Lin, our General Counsel and Chief Privacy Officer, and John M. Green, our Chief Financial Officer. We also had an offer letter agreement with Sheri Huston, who was formerly our Chief Financial Officer. We do not have offer letter agreements or employment agreements with Magid M. Abraham, our President and Chief Executive Officer, or Gian M. Fulgoni, our Executive Chairman of the Board of Directors.

In September 1999, we entered into an offer letter agreement with Gregory T. Dale. The letter agreement set forth Mr. Dale's base salary of \$105,000 per year, an annual performance bonus of up to 15% of Mr. Dale's base salary and a grant of options for the purchase of 250,000 shares of our common stock. Mr. Dale's current annual base salary is \$225,000, and the compensation committee of our board of directors has approved an increase of his annual base salary to \$260,000 effective March 1, 2007. Mr. Dale is entitled to receive all normal benefits provided to our employees including health insurance and three weeks paid vacation. In December 1999, Mr. Dale was granted a stock option to purchase an aggregate of 275,000 shares of our common stock at an exercise price of \$0.10 per share pursuant to this agreement. The shares subject to the options vested over the next four years in equal monthly installments.

In December 2003, we entered into an offer letter agreement with Christiana L. Lin. The letter agreement set forth Ms. Lin's base salary of \$106,000 per year. Ms. Lin's current annual base salary is \$150,000, and the compensation committee of our board of directors has approved an increase of her annual base salary to \$200,000 effective March 1, 2007. Ms. Lin is entitled to receive all normal benefits provided to our employees including health insurance and twelve days paid vacation. The offer letter agreement provides that our employment relationship with Ms. Lin's employment is at will, and we or Ms. Lin may terminate the relationship at anytime.

In August 2002, we entered into an offer letter agreement with Sheri L. Huston. The letter agreement set forth Ms. Huston's base salary of \$215,000 per year, an annual performance bonus of up to 30% of Ms. Huston's base salary and a grant of options for the purchase of 250,000 shares of our common stock. In

October 2002, Ms. Huston was granted a stock option to purchase an aggregate of 250,000 shares of our common stock at an exercise price of \$0.25 per share pursuant to this agreement. The shares subject to the options vested over the next four years in equal monthly installments. On February 28, 2006, Ms. Huston terminated her employment and entered into a Separation Agreement with us. Pursuant to such Separation Agreement, we agreed to pay Ms. Huston severance benefits equivalent to six months base salary as well as Ms. Huston's 2005 performance bonus and the amount of her health insurance premiums in a lump sum payment upon her termination.

In May 2006, we entered into an offer letter agreement with John M. Green. The letter agreement set forth Mr. Green's base salary of \$250,000 per year, an annual performance bonus of up to 30% of Mr. Green's base salary and a grant of options for the purchase of 650,000 shares of our common stock. Mr. Green's current annual base salary is \$250,000, and the compensation committee of our board of directors has approved an increase of his annual base salary to \$270,000 effective March 1, 2007. In May 2006, Mr. Green was granted a stock option to purchase an aggregate of 650,000 shares of our common stock at an exercise price of \$1.50 per share pursuant to this agreement. The shares subject to the options vest over the four years following the start of Mr. Green's employment in equal monthly installments. Upon a change of control, if Mr. Green loses his position as Chief Financial Officer or is not provided an equivalent position, any remaining unvested shares under this option shall fully vest. Also, upon a change of control, if Mr. Green is provided with an alternative but diminished position, the lesser of either (i) any remaining unvested shares under this option or (ii) 162,500 shares under this option shall fully vest. The offer letter agreement provides that we may terminate Mr. Green's employment at any time with or without cause. In the event we terminate Mr. Green without cause, Mr. Green is entitled to severance for six pay periods. If we terminate his employment or he resigns, he is entitled to receive any unpaid prorated base salary along with all benefits and expense reimbursements to which he is entitled by virtue of his past employment with us.

Benefit Plans

The following section provides more details concerning our 1999 Stock Plan and our 2007 Equity Incentive Plan.

1999 Stock Plan

Our 1999 Stock Plan, as amended (the "1999 Stock Plan") was adopted by our board of directors and approved by our stockholders on September 23, 1999. The plan was last amended by our board of directors and approved by our stockholders on April 12, 2005. Our 1999 Stock Plan provides for the grant of incentive stock options, within the meaning of Section 422 of the Internal Revenue Code of 1986, as amended (the "Code"), to our employees and any parent and subsidiary corporations' employees, and for the grant of nonstatutory stock options and stock purchase rights to our employees, directors and consultants and any parent and subsidiary corporations' employees and consultants. We do not intend to grant any additional awards under our 1999 Stock Plan following this offering. However, our 1999 Stock Plan will continue to govern the terms and conditions of outstanding awards granted the resulter.

We have reserved a total of 26,760,284 shares of our common stock for issuance pursuant to the 1999 Stock Plan. As of December 31, 2006, options to purchase 13,619,700 shares of common stock were outstanding and 5,316,147 shares were available for future grant under this plan.

The compensation committee of our board of directors currently administers our 1999 Stock Plan. Under our 1999 Stock Plan, the plan administrator has the power to determine the terms of the awards, including the employees, directors and consultants who will receive awards, the exercise price, the number of shares subject to each award, the vesting schedule and exercisability of awards and the form of consideration payable upon exercise.

With respect to all incentive stock options granted under the 1999 Stock Plan, the exercise price must at least be equal to the fair market value of our common stock on the date of grant. With respect to all nonstatutory stock options granted under the 1999 Stock Plan, the exercise price must at least be equal to 85% of the fair market value of our common stock on the date of grant. The term of an option may not exceed ten

years, except that with respect to any participant who owns 10% of the voting power of all classes of our outstanding stock as of the grant date, the term must not exceed five years and the exercise price must equal at least 110% of the fair market value on the grant date. The administrator determines the terms of all other options.

After termination of an employee, director or consultant, he or she may exercise his or her option for the period of time stated in the option agreement. If termination is due to disability or death, the option will remain exercisable for no less than six months. In all other cases, the option will generally remain exercisable for at least thirty days. In the absence of a specified period of time in the option agreement, the option will remain exercisable for a period of three months following termination (or twelve months in the event of a termination due to death of disability). However, an option generally may not be exercised later than the expiration of its term.

Stock purchase rights may be granted alone, in addition to or in tandem with other awards granted under our 1999 Stock Plan. Stock purchase rights are rights to purchase shares of our common stock that vest in accordance with terms and conditions established by the administrator. The administrator will determine the number of shares subject to a stock purchase right granted to any employee, director or consultant. The administrator may impose whatever conditions to vesting it determines to be appropriate. Unless the administrator determines otherwise, we have a repurchase option exercisable upon termination of the purchaser's service with us. Shares subject to stock purchase rights that do not vest are subject to our right of repurchase or forfeiture

Our 1999 Stock Plan provides that in the event of certain change in control transactions, including our merger with or into another corporation or the sale of substantially all of our assets, the successor corporation will assume or substitute an equivalent award with respect to each outstanding award under the plan. If there is no assumption or substitution of outstanding awards, such awards will become fully vested and exercisable and the administrator will provide notice to the recipient that he or she has the right to exercise such outstanding awards for a period of fifteen days from the date of such notice. The awards will terminate upon the expiration of such stated notice period.

Unless otherwise determined by the administrator, the 1999 Stock Plan generally does not allow for the sale or transfer of awards under the 1999 Stock Plan other than by will or the laws of descent and distribution, and may be exercised only during the lifetime of the participant and only by such participant.

We have also established a U.K. sub-plan to our 1999 Stock Plan for option grants to U.K. residents

Our board of directors has the authority to amend, alter, suspend or terminate the 1999 Stock Plan provided such action does not impair the rights of any participant without the written consent of such participant.

2007 Equity Incentive Plan

Our board of directors and stockholders have adopted our 2007 Equity Incentive Plan (the "2007 Equity Incentive Plan"), to become effective upon the completion of this offering. Our 2007 Equity Incentive Plan provides for the grant of incentive stock options, within the meaning of Section 422 of the Code, to our employees and any parent and subsidiary corporations' employees, and for the grant of nonstatutory stock options, restricted stock, restricted stock units, stock appreciation rights, performance units and performance shares to our employees, directors and consultants and our parent and subsidiary corporations' employees and consultants.

We have reserved a total of 7,000,000 shares of our common stock for issuance pursuant to the 2007 Equity Incentive Plan, plus (a) any shares which have been reserved but not issued under our 1999 Stock Plan and are not subject to any awards granted thereunder, and (b) any shares subject to stock options or similar awards granted under the 1999 Stock Plan that expire or otherwise terminate without having been exercised in full and shares issued pursuant to awards granted under the 1999 Stock Plan that are forfeited to or repurchased by us. The maximum number of shares that may be added to the 2007 Equity Incentive Plan from the 1999 Stock Plan is 5,000,000 shares. In addition, our 2007 Equity Incentive Plan provides for annual

increases in the number of shares available for issuance thereunder on the first day of each fiscal year, beginning with our 2008 fiscal year, equal to the least of:

- 4% of the outstanding shares of our common stock on the last day of the immediately preceding fiscal year;
- 9 000 000 shares: o
- · such other amount as our board of directors may determine.

Our board of directors or a committee of our board administers our 2007 Equity Incentive Plan. In the case of options intended to qualify as "performance based compensation" within the meaning of Section 162(m) of the Code, the committee will consist of two or more "outside directors" within the meaning of Section 162(m) of the Code. The administrator has the power to determine the terms of the awards, including the exercise price, the number of shares subject to each such award, the exercisability of the awards and the form of consideration payable upon exercise. The administrator also has the authority to institute an exchange program whereby the exercise prices of outstanding awards may be reduced, outstanding awards may be surrendered or cancelled in exchange for awards with a higher or lower exercise price, or outstanding awards may be transferred to a third party.

The exercise price of options granted under our 2007 Equity Incentive Plan must at least be equal to the fair market value of our common stock on the date of grant. The term of an incentive stock option may not exceed ten years, except that with respect to any participant who owns 10% of the voting power of all classes of our outstanding stock as of the grant date, the term must not exceed five years and the exercise price must equal at least 110% of the fair market value on the grant date. The administrator determines the terms of all other options.

After termination of an employee, director or consultant, he or she may exercise his or her option for the period of time stated in the option agreement. Generally, if termination is due to death or disability, the option will remain exercisable for twelve months. In all other cases, the option will generally remain exercisable for three months. However, an option generally may not be exercised later than the expiration of its term.

Stock appreciation rights may be granted under our 2007 Equity Incentive Plan. Stock appreciation rights allow the recipient to receive the appreciation in the fair market value of our common stock between the exercise date and the date of grant. The administrator determines the terms of stock appreciation rights, including when such rights become exercisable and whether to pay the increased appreciation in cash or with shares of our common stock, or a combination thereof. Stock appreciation rights expire under the same rules that apply to stock options.

Restricted stock may be granted under our 2007 Equity Incentive Plan. Restricted stock awards are shares of our common stock that vest in accordance with terms and conditions established by the administrator. The administrator will determine the number of shares of restricted stock granted to any employee. The administrator may impose whatever conditions to vesting it determines to be appropriate. For example, the administrator may set restrictions based on the achievement of specific performance goals. Shares of restricted stock that do not vest are subject to our right of repurchase or forfeiture.

Restricted stock units may be granted under our 2007 Equity Incentive Plan. Restricted stock units are awards that will result in a payment to a participant at the end of a specified period only if performance goals established by the administrator are achieved or the award otherwise vests. The administrator may impose whatever conditions to vesting, restrictions and conditions to payment it determines to be appropriate. For example, the administrator may set restrictions based on the achievement of specific performance goals, on the continuation of service or employment or any other basis determined by the administrator. Payments of earned restricted stock units may be made, in the administrator's discretion, in cash or with shares of our common stock, or a combination thereof.

Performance units and performance shares may be granted under our 2007 Equity Incentive Plan. Performance units and performance shares are awards that will result in a payment to a participant only if performance goals established by the administrator are achieved or the awards otherwise vest. The

administrator will establish organizational or individual performance goals in its discretion, which, depending on the extent to which they are met, will determine the number and/or the value of performance units and performance shares to be paid out to participants. Performance units shall have an initial dollar value established by the administrator prior to the grant date. Performance shares shall have an initial value equal to the fair market value of our common stock on the grant date. Payment for performance units and performance shares may be made in cash or in shares of our common stock with equivalent value, or in some combination, as determined by the administrator.

Unless the administrator provides otherwise, our 2007 Equity Incentive Plan does not allow for the transfer of awards and only the recipient of an award may exercise an award during his or her lifetime.

Our 2007 Equity Incentive Plan provides that in the event of a change in control, as defined in the 2007 Equity Incentive Plan, each outstanding award will be treated as the administrator determines, including that the successor corporation or its parent or subsidiary will assume or substitute an equivalent award for each outstanding award. The administrator is not required to treat all awards similarly. If there is no assumption or substitution of outstanding awards, the awards will fully vest, all restrictions will lapse, and the awards will become fully exercisable. The administrator will provide notice to the recipient that he or she has the right to exercise the option and stock appreciation right as to all of the shares subject to the award, all restrictions on restricted stock will lapse, and all performance goals or other vesting requirements for performance shares and units will be deemed achieved, and all other terms and conditions met. The option or stock appreciation right will terminate upon the expiration of the period of time the administrator provides in the notice. In the event the service of an outside director is terminated on or following a change in control, other than pursuant to a voluntary resignation, his or her options and stock appreciation rights will fully vest and become immediately exercisable, all restrictions on restricted stock will lapse, and all performance goals or other vesting requirements for performance shares and units will be deemed achieved, and all other terms and conditions met.

Our 2007 Equity Incentive Plan will automatically terminate in 2017, unless we terminate it sooner. In addition, our board of directors has the authority to amend, alter, suspend or terminate the 2007 Equity Incentive Plan provided such action does not impair the rights of any participant.

CERTAIN RELATIONSHIPS AND RELATED PARTY TRANSACTIONS

Transactions and Relationships with Directors, Officers and 5% Stockholders

On August 1, 2003, we entered into a Licensing and Services Agreement with Citadel Investment Group, L.L.C., an entity affiliated with Citadel Equity Fund Ltd., a former stockholder that, until November 27, 2006, beneficially owned more than 5% of our outstanding voting stock and had a representative on our board of directors. Pursuant to the terms of the Licensing and Services Agreement, we granted Citadel Investment Group, L.L.C. a license to certain digital marketing intelligence data and products, subject to certain limitations. In each of 2004, 2005 and 2006, we received license fees of \$3 million and in 2007 we will receive an additional \$3 million.

In 2006, Linda Abraham, the spouse of our President and Chief Executive Officer, Magid Abraham, held the positions of acting Executive Vice President for Finance, Telecom and Pharmaceuticals and Executive Vice President for Product Management. In these positions, Ms. Abraham earned approximately \$143,564 in salary. Ms. Abraham remains employed as our Executive Vice President for Product Management.

Registration Rights Agreements

We and certain holders of our capital stock have entered into an agreement, pursuant to which these stockholders will have registration rights with respect to their shares of common stock following this offering. See "Description of Capital Stock — Registration Rights" for a further description of the terms of this agreement.

Indemnification Agreements

We have entered into an indemnification agreement with each of our directors and officers. The indemnification agreements and our amended and restated certificate of incorporation and bylaws require us to indemnify our directors and officers to the fullest extent permitted by Delaware law. See "Management — Limitations on Director and Officer Liability and Indemnification."

PRINCIPAL AND SELLING STOCKHOLDERS

The following table sets forth certain information with respect to the beneficial ownership of our common stock as of December 31, 2006 and as adjusted to reflect the sale of shares of our common stock offered by this prospectus, by:

- each beneficial owner of 5% or more of the outstanding shares of our common stock;
- · each of our directors;
- · each of our named executive officers;
- · each selling stockholder; and
- · all directors and executive officers as a group.

After giving effect to the conversion of all shares of our preferred stock into shares of our common stock, the table assumes the conversion of all shares of our preferred stock into shares of our common stock immediately prior to the completion of this offering. See "Description of Capital Stock — Preferred Stock". Beneficial ownership is determined in accordance with the rules of the SEC. In computing the number of shares beneficially owned by a person and the percentage ownership of that person, shares of common stock subject to options or warrants held by that person that are currently exercisable within 60 days of December 31, 2006 are deemed outstanding, but are not deemed outstanding for computing the percentage ownership of any other person. Percentage of beneficial ownership is based on 108,025,682 shares of common stock outstanding as of December 31, 2006 and shares of common stock outstanding after this offering.

To our knowledge, except as set forth in the footnotes to this table and subject to applicable community property laws, each person named in the table has sole voting and investment power with respect to the shares set forth opposite such person's name. Except as otherwise indicated, the address of each of the persons in this table is c/o comScore, Inc., 11465 Sunset Hills Road, Suite 200, Reston, Virginia 20190.

	Shares Beneficially Prior to the Off		Number of	Shares Beneficially Owned After the Offering	
Name of Beneficial Owner	Number	Percent	Shares Offered	Number	Percent
5% Stockholders:				<u> </u>	
Accel Partners(1)	29,514,275	27.3%			
J.P. Morgan Partners SBIC, LLC and related entities(2)	12,530,421	11.6			
Institutional Venture Partners(3)	10,949,164	10.1			
Lehman Brothers Inc.(4)	8,708,908	8.1			
Adams Street Partners(5)	8,505,767	7.9			
Topspin Partners, L.P.(6)	5,887,217	5.4			
Directors and Named Executive Officers:					
Magid M. Abraham, Ph.D.(7)	7,816,877	7.2			
Gian M. Fulgoni	6,696,019	6.2			
Gregory T. Dale(8)	828,633	*			
John M. Green(9)	108,333	*			
Sheri Huston	510,398	*			
Christiana L. Lin(10)	177,481	*			
Thomas D. Berman(11)	_	*			
Bruce Golden(12)	_	*			
William J. Henderson(13)	144,583	*			
Ronald J. Korn(14)	33,333	*			
Frederick R. Wilson(15)	_	*			
All directors and executive officers as a group (eleven persons)(16)	16,315,477	14.8			

- * Represents less than one percent (1%) of the outstanding shares of common stock.
- (1) Includes 21,486,401 shares held by Accel VII L.P., 5,371,593 shares held by Accel Internet Fund III L.P., and 2,656,281 shares held by Accel Investors '99 L.P. (together, the "Accel Funds"). Accel VII Associates L.L.C. is a general partner of Accel VII L.P. and has sole voting and dispositive power with respect to the shares held by Accel VII L.P. Accel Internet Fund III L.P. and has sole voting and dispositive power with respect to the shares held by Accel Internet Fund III L.P. James W. Breyer, Arthur C. Patterson, Theresia Gouw Ranzetta, James R. Swartz, and J. Peter Wagner are managing members of Accel VII Associates L.L.C. and Accel Internet Fund III Associates L.L.C. and clipsositive power with respect to the shares held by Accel Investors '99 L.P. The general partners and managing members of Accel Investors '99 L.P. and share voting and dispositive power with respect to the shares held by Accel Investors '99 L.P. The general partners and managing members disclaim beneficial ownership of the shares owned by the Accel Funds except to the extent of their proportionate pecuniary interest therein. The address for Accel Partners is 428 University Avenue, Palo Alto, California 94301.
- (2) Includes 10,988,417 shares held by J.P. Morgan Partners (SBIC), LLC ("JPMP SBIC") and 1,542,004 shares held by J.P. Morgan Partners (BHCA), L.P. ("BHCA"). The sole member of JPMP SBIC is BHCA. Pursuant to Rule 13d-3 under the Exchange Act, BHCA may be deemed to beneficially own the shares held by JPMP SBIC; however, the foregoing shall not be construed as an admission that BHCA is the beneficial owner of such shares. The general partner of BHCA is JPMP Mseter Fund Manager, L.P. ("JPMP MFM"). The general partner of JPMP MFM is JPMP Capital Corp. ("JPMP Capital"), a wholly owned subsidiary of JPMorgan Chase & Co. Each of JPMP MFM and JPMP Capital may be deemed, pursuant to Rule 13d-3 under the Exchange Act, to beneficially own the shares held by JPMP MFM and BHCA; however, the foregoing shall not be construed as an admission that JPMP SBIC or JPMP Capital is the beneficial owner of such shares. JPMP Capital exercises voting and dispositive power over the securities held by JPMP SBIC and BHCA. Voting and disposition decisions at JPMP Capital are made by an investment committee of three or more of its officers, and therefore no individual officer of JPMP Capital is the beneficial owner of the securities. The address for each of JPMP SBIC, BHCA, JPMP MFM and JPMP Capital is c/o J.P. Morgan Partners, LLC, 270 Park Avenue, New York, New York 10017.
- (3) Includes 8,968,827 shares held by Institutional Venture Partners X, L.P. and 1,980,337 shares held by Institutional Venture Partners X GmbH & Co. Beteiligungs KG. The address of Institutional Venture Partners is 3000 Sand Hill Road, Building 2, Suite 250, Menlo Park, California 94025.
- (4) Shares beneficially owned by Lehman Brothers Inc. includes shares held by the following wholly owned subsidiaries and affiliates of Lehman Brothers Inc.: 3,829,870 shares held by LB I Group Inc., 3,157,739 shares held by Lehman Brothers Venture Partners L.P., and 1,721,299 shares held by Lehman Brothers Venture Capital Partners I, L.P. The address for Lehman Brothers Inc. is 3000 Sand Hill Road, Building 3, Suite 190, Menlo Park, CA 94025.
- (5) Adams Street Partners, LLC, is the administrative member of BVCF IV, L.P., the entity that holds these shares. Adams Street Partners, LLC disclaims beneficial ownership of these shares except to the extent of its proportionate pecuniary interest therein. Mr. Thomas D. Berman is a partner of Adams Street Partners, LLC. Mr. Berman disclaims beneficial ownership of these shares except to the extent of his proportionate pecuniary interest therein.
- (6) Includes 5,621,116 shares held by Topspin Partners, L.P. and 266,101 shares held by Topspin Associates, L.P. The address for Topspin Partners is Three Expressway Plaza, Roslyn Heights, New York 11577.
- (7) Includes 2,909,375 shares held by the Abraham Family Trust and 1,083,465 shares subject to options that are immediately exercisable or exercisable within 60 days of December 31, 2006. Also includes 114,638 shares subject to options held by Mr. Abraham's wife, Linda Abraham, that are immediately exercisable or exercisable within 60 days of December 31, 2006.
- (8) Includes 545,047 shares subject to options that are immediately exercisable or exercisable within 60 days of December 31, 2006.
- (9) Includes 108,333 shares subject to options that are immediately exercisable or exercisable within 60 days of December 31, 2006.

- (10) Includes 77,614 shares subject to options that are immediately exercisable or exercisable within 60 days of December 31, 2006.
- (11) This total does not include 8,505,767 shares held by JP Morgan Chase Bank as custodian for BVCF IV, L.P. Mr. Berman is a partner of Adams Street Partners, LLC, the administrative member of BVCF IV, L.P. Mr. Berman disclaims beneficial ownership of these shares except to the extent of his proportionate pecuniary interest therein.
- (12) This does not include 29,514,275 shares owned by the Accel Funds. Bruce Golden is a general partner of Accel Partners. Mr. Golden disclaims beneficial ownership of any of the Accel Funds' shares except to the extent of his proportionate pecuniary interest therein.
- (13) Includes 44,583 shares subject to options that are immediately exercisable or exercisable within 60 days of December 31, 2006 and 100,000 shares subject to warrants that are immediately exercisable or exercisable within 60 days of December 31, 2006.
- (14) Includes 33,333 shares subject to options that are immediately exercisable or exercisable within 60 days of December 31, 2006.
- (15) Mr. Wilson is a managing member of Flatiron Partners and shares voting and dispositive power with respect to the 3,699,712 shares of common stock (assuming the conversion of all shares of preferred stock) owned by the Flatiron Funds and Flatiron Associates entities. Mr. Wilson disclaims beneficial ownership of these shares except to the extent of his proportionate pecuniary interest therein.
- (16) Includes 2,004,992 shares subject to options that are immediately exercisable or exercisable within 60 days of December 31, 2006 and 100,000 shares subject to warrants that are immediately exercisable or exercisable within 60 days of December 31, 2006.

DESCRIPTION OF CAPITAL STOCK

The following information describes our common stock and preferred stock, as well as options to purchase our common stock and provisions of our amended and restated certificate of incorporation and bylaws. This description is only a summary. You should also refer to our amended and restated certificate of incorporation and bylaws, which have been filed with the Securities and Exchange Commission as exhibits to our registration statement, of which this prospectus forms a part.

Conoral

Upon the completion of this offering, our authorized capital stock will consist of 150,000,000 shares of common stock with a \$0.001 par value per share, and 5,000,000 shares of preferred stock with a \$0.001 par value per share, all of which shares of preferred stock will be undesignated. Our board of directors may establish the rights and preferences of the preferred stock from time to time. As of December 31, 2006, after giving effect to the conversion of all outstanding preferred stock into shares of common stock, there would have been 108,830,901 shares of common stock issued and outstanding.

Common Stock

Each holder of our common stock is entitled to one vote for each share on all matters to be voted upon by the stockholders and there are no cumulative rights. Subject to any preferential rights of any outstanding preferred stock, holders of our common stock will be entitled to receive ratably the dividends, if any, as may be declared from time to time by the board of directors out of funds legally available therefor. If there is a liquidation, dissolution or winding up of our company, holders of our common stock would be entitled to share in our assets remaining after the payment of liabilities and any preferential rights of any outstanding preferred stock.

Holders of our common stock will have no preemptive or conversion rights or other subscription rights, and there will be no redemption or sinking fund provisions applicable to the common stock. All outstanding shares of our common stock will be fully paid and non-assessable. The rights, preferences and privileges of the holders of our common stock will be subject to, and may be adversely affected by, the rights of the holders of shares of any series of preferred stock which we may designate and issue in the future.

Pursuant to our acquisition of Q2 Brand Intelligence, Inc. and SurveySite Inc., we granted the former shareholders of these entities the right to sell a certain number of shares of our common stock back to us at an agreed-upon price. These rights transfer to any subsequent holder of these shares and are described in more detail in the "Overview" section of "Management's Discussion and Analysis of Financial Condition and Results of Operations."

Preferred Stock

Immediately prior to the completion of this offering, all outstanding shares of all series of our convertible preferred stock will be converted into shares of common stock according to the formula set forth in our current certificate of incorporation.

Under the terms of our amended and restated certificate of incorporation, our board of directors is authorized to issue shares of preferred stock in one or more series without stockholder approval. Our board of directors has the discretion to determine the rights, preferences, privileges and restrictions, including voting rights, dividend rights, conversion rights, redemption privileges and liquidation preferences, of each series of preferred stock.

The purpose of authorizing our board of directors to issue preferred stock and determine its rights and preferences is to eliminate delays associated with a stockholder vote on specific issuances. The issuance of preferred stock, while providing flexibility in connection with possible future acquisitions and other corporate purposes, will affect, and may adversely affect, the rights of holders of common stock. It is not possible to state the actual effect of the issuance of any shares of preferred stock on the rights of holders of common

stock until the board of directors determines the specific rights attached to that preferred stock. The effects of issuing preferred stock could include one or more of the following:

- · restricting dividends on the common stock;
- · diluting the voting power of the common stock;
- · impairing the liquidation rights of the common stock; or
- · delaying or preventing changes in control or management of our company.

We have no present plans to issue any shares of preferred stock.

Warrants

As of February 1, 2007, assuming the conversion of our convertible preferred stock into common stock, warrants for the purchase of an aggregate of 875,923 shares of our common stock were outstanding as follows:

- A warrant issued on June 9, 2000 to purchase 46,551 shares of our Series B Convertible Preferred Stock at an exercise price of \$2.90 per share. This warrant was issued in
 connection with the lease of certain of our equipment. Upon the automatic conversion of our convertible preferred stock immediately prior to the completion of this offering, the
 warrant shall be exercisable for 92,347 shares of our common stock at an exercise price of \$1.46 per share. The warrant shall terminate on the earlier of (i) June 9, 2010 or (ii) five
 years from the date of effectiveness of this registration statement. However, if this warrant is not exercised prior to termination and the fair market value of a share of our common
 stock exceeds the exercise price per share of this warrant immediately prior to termination, this warrant will automatically exercise prior to expiration.
- A warrant issued on July 31, 2000 to purchase 20,100 shares of our common stock to a consultant to us at an exercise price of \$2.50 per share. The warrant shall terminate on July 31, 2010
- A warrant issued on September 29, 2000 to purchase 9,694 shares of our Series B Convertible Preferred Stock at an exercise price of \$4.90 per share. This warrant was issued in connection with the lease of certain of our equipment. Upon the automatic conversion of our convertible preferred stock immediately prior to the completion of this offering, the warrant shall be exercisable for 19,231 shares of our common stock at an exercise price of \$2.47 per share. The warrant shall terminate on the earlier of (i) September 29, 2010 or (ii) five years from the date of effectiveness of this registration statement. However, if this warrant is not exercised prior to termination and the fair market value of a share of our common stock exceeds the exercise price per share of this warrant immediately prior to termination, this warrant will automatically exercise prior to expiration.
- A warrant issued on June 26, 2001 to purchase 100,000 shares of our common stock to William Henderson, a member of our board of directors, at an exercise price of \$1.00 per share. The warrant shall terminate on the earlier of (i) June 26, 2011; (ii) the completion of this offering; or (iii) a change of control as defined in the warrant.
- A warrant issued on November 30, 2001 to purchase 10,000 shares of our common stock to our landlord at an exercise price of \$5.90 per share. The warrant shall terminate on September 30, 2009.
- A warrant issued on July 3, 2002 to purchase 12,000 shares of our common stock to our landlord at an exercise price of \$3.00 per share. The warrant shall terminate on the earlier of (i) July 3, 2012; (ii) the receipt of prior written notice from an underwriter of this offering requesting exercise; or (iii) the closing of a merger as defined in the warrant. However, if this warrant is not exercised prior to termination and the fair market value of a share of our common stock exceeds the exercise price per share of this warrant immediately prior to termination, this warrant will automatically exercise prior to expiration.

- A warrant issued on July 31, 2002 to purchase 36,127 shares of our Series D Convertible Preferred Stock at an exercise price of \$0.8996 per share. This warrant was issued in connection with a promissory note. Upon the automatic conversion of our convertible preferred stock immediately prior to the completion of this offering, the warrant shall be exerciseable for 40,625 shares of our common stock at an exercise price of \$0.80 per share. The warrant includes certain registration rights under our fourth amended and restated investor rights agreement, but the holder of the warrant does not have a stand-alone right to demand registration of the shares. The warrant shall terminate on the later of (i) July 31, 2012 or (ii) five years from the completion of this offering. However, if this warrant is not exercised prior to termination and the fair market value of a share of our common stock exceeds the exercise price per share of this warrant immediately prior to termination, this warrant will automatically exercise prior to expiration.
- A warrant issued on July 31, 2002 to purchase 108,382 shares of our Series D Convertible Preferred Stock at an exercise price of \$0.8996 per share. This warrant was issued in connection with the lease of certain of our equipment originally. Upon the automatic conversion of our convertible preferred stock immediately prior to the completion of this offering, the warrant shall be exercisable for 121,875 shares of our common stock at an exercise price of \$0.80 per share. The warrant includes certain registration rights under our fourth amended and restated investor rights agreement, but the holder of the warrant does not have a stand-alone right to demand registration of the shares. The warrant shall terminate on the later of (i) July 31, 2012 or (ii) five years from the completion of this offering. However, if this warrant is not exercised prior to termination and the fair market value of a share of our common stock exceeds the exercise price per share of this warrant immediately prior to termination, this warrant will automatically exercise prior to expiration.
- A warrant issued on December 5, 2002 to purchase 45,854 shares of our Series D Convertible Preferred Stock at an exercise price of \$0.8996 per share. This warrant was issued in connection with a promissory note. Upon the automatic conversion of our convertible preferred stock immediately prior to the completion of this offering, the warrant shall be exercisable for 51,563 shares of our common stock at an exercise price of \$0.80 per share. The warrant includes certain registration rights under our fourth amended and restated investor rights agreement. The warrant shall terminate on December 4, 2012. However, if this warrant is not exercised prior to termination and the fair market value of a share of our common stock exceeds the exercise price per share of this warrant immediately prior to termination, this warrant will automatically exercise prior to expiration.
- A warrant issued on June 24, 2003 to purchase 100,000 shares of our common stock to our landlord at an exercise price of \$0.60 per share. The warrant shall terminate on the
 earlier of (i) June 24, 2013; (ii) the receipt of prior written notice from an underwriter of this offering requesting exercise; or (iii) the closing of a merger as defined in the warrant.
 However, if this warrant is not exercised prior to termination and the fair market value of a share of our common stock exceeds the exercise price per share of this warrant
 immediately prior to termination, this warrant will automatically exercise prior to expiration.
- A warrant issued on December 19, 2003 to purchase 240,000 shares of our Series E Convertible Preferred Stock at an exercise price of \$0.50 per share. This warrant was issued in connection with an equipment financing. Upon the automatic conversion of our convertible preferred stock immediately prior to the completion of this offering, the warrant shall be exercisable for 240,000 shares of our common stock at an exercise price of \$0.50 per share. The warrant includes certain registration rights under our fourth amended and restated investor rights agreement, but the holder of the warrant does not have a stand-alone right to demand registration of the shares. The warrant shall terminate on the later of (i) December 19, 2013; or (ii) five years from the completion of this offering. However, in the event that an underwriter of this offering provides prior written notice to the holder of the warrant requesting exercise, the warrant must either be exercised or waived. Furthermore, this warrant will expire upon the closing of a merger as defined in the warrant. However, if this warrant is not exercised prior to termination and the fair market value of a share of our common stock exceeds the exercise price per share of this warrant immediately prior to termination, this warrant will automatically exercise prior to expiration.

A warrant issued on April 29, 2005 to purchase 68,182 shares of our common stock to a creditor at an exercise price of \$1.10 per share. The warrant shall terminate on the later of
 (i) April 29, 2015 or (ii) five years after the closing of this offering. The warrant shall also terminate upon a merger as defined in the warrant. However, if the warrant is not
 exercised prior to termination and the fair market value of a share of our common stock exceeds the exercise price per share of this warrant immediately prior to termination, this
 warrant shall automatically exercise prior to expiration.

Registration Rights

In August 2003, we and the holders of all series of our convertible preferred stock entered into a fourth amended and restated investor rights agreement, which is included as an exhibit to the registration statement of which this prospectus is a part. Under the agreement, commencing 180 days after the closing of this offering, the holders of a majority of the shares of common stock issued upon the conversion of the shares of our Series A, B, C, C-1, D and E convertible preferred stock, which we refer to as "registrable securities", may require us to prepare and file a registration statement under the Securities Act, at our expense, covering the lesser of registrable securities with an aggregate anticipated offering price of at least \$10,000,000 or 3,000,000 shares of registrable securities. Under these demand registration rights, we are required to use our best efforts to cause the shares requested to be included in the registration statement, subject to customary conditions and limitations. We are not obligated to effect more than two of these demand registrations.

In addition, these holders have certain "piggyback" registration rights. If we propose to register any of our equity securities under the Securities Act other than specified excluded registrations, these holders are entitled to written notice of the registration and may require us to include all or a portion of their registrable securities in the registration and in any related underwriting. However, the managing underwriter has the right, subject to specified conditions, to limit the number of registrable securities such holders may include. Once we become eligible to file a registration statement on Form S-3, the holders of the registrable securities may require us to register these shares on Form S-3, if such registration will generate anticipated aggregate net proceeds of at least \$2,000,000, or consist of at least \$,000,000 shares. The holder of certain of our warrants that are exercisable for shares of our convertible preferred stock also have some or all of the registration rights described above. The registration rights described above terminate no later than five years after this offering. Registration of these shares under the Securities Act would result in these shares, other than shares purchased by our affiliates, becoming freely tradable without restriction under the Securities Act.

Effect of Certain Provisions of our Amended and Restated Certificate of Incorporation and Bylaws and the Delaware Anti-Takeover Statute

Amended and Restated Certificate of Incorporation and Bylaws

Some provisions of Delaware law and our amended and restated certificate of incorporation and bylaws contain provisions that could make the following transactions more difficult:

- · acquisition of us by means of a tender offer;
- · acquisition of us by means of a proxy contest or otherwise; or
- · removal of our incumbent officers and directors

These provisions, summarized below, are expected to discourage coercive takeover practices and inadequate takeover bids and to promote stability in our management. These provisions are also designed to encourage persons seeking to acquire control of us to first negotiate with our board of directors.

Undesignated Preferred Stock. The ability to authorize undesignated preferred stock makes it possible for our board of directors to issue one or more series of preferred stock with voting or other rights or preferences that could impede the success of any attempt to change control of comScore. These and other provisions may have the effect of deferring hostile takeovers or delaying changes in control or management of our company.

- · Stockholder Meetings. Our charter documents provide that a special meeting of stockholders may be called only by resolution adopted by the board of directors.
- Requirements for Advance Notification of Stockholder Nominations and Proposals. Our bylaws establish advance notice procedures with respect to stockholder proposals and the nomination of candidates for election as directors, other than nominations made by or at the direction of the board of directors or a committee of the board of directors.
- Board Classification. Our board of directors is divided into three classes. The directors in each class will serve for a three-year term, one class being elected each year by our stockholders. This system of electing and removing directors may tend to discourage a third party from making a tender offer or otherwise attempting to obtain control of us, because it generally makes it more difficult for stockholders to replace a majority of the directors.
- Limits on Ability of Stockholders to Act by Written Consent. We have provided in our certificate of incorporation that our stockholders may not act by written consent. This limit on the ability of our stockholders to act by written consent may lengthen the amount of time required to take stockholder actions. As a result, a holder controlling a majority of our capital stock would not be able to amend our bylaws or remove directors without holding a meeting of our stockholders called in accordance with our bylaws.
- Amendment of Certificate of Incorporation and Bylaws. The amendment of the above provisions of our amended and restated certificate of incorporation and bylaws requires approval by holders of at least two-thirds of our outstanding capital stock entitled to vote generally in the election of directors.

Delaware Anti-Takeover Statute

We are subject to Section 203 of the General Corporation Law of the State of Delaware, which prohibits a Delaware corporation from engaging in any business combination with any interested stockholder for a period of three years after the date that such stockholder became an interested stockholder, with the following exceptions:

- before such date, the board of directors of the corporation approved either the business combination or the transaction that resulted in the stockholder becoming an interested stockholder:
- upon completion of the transaction that resulted in the stockholder becoming an interested stockholder, the interested stockholder owned at least 85% of the voting stock of the corporation outstanding at the time the transaction began, excluding for purposes of determining the voting stock outstanding (but not the outstanding voting stock owned by the interested stockholder) those shares owned (i) by persons who are directors and also officers and (ii) employee stock plans in which employee participants do not have the right to determine confidentially whether shares held subject to the plan will be tendered in a tender or exchange offer; or
- on or after such date, the business combination is approved by the board of directors and authorized at an annual or special meeting of the stockholders, and not by written consent, by the affirmative vote of at least 66²/₃% of the outstanding voting stock that is not owned by the interested stockholder.

In general, Section 203 defines business combination to include the following:

- any merger or consolidation involving the corporation and the interested stockholder;
- any sale, lease, exchange, mortgage, transfer, pledge or other disposition of 10% or more of either the assets or outstanding stock of the corporation involving the interested stockholder:
- subject to certain exceptions, any transaction that results in the issuance or transfer by the corporation of any stock of the corporation to the interested stockholder;
- any transaction involving the corporation that has the effect of increasing the proportionate share of the stock of any class or series of the corporation beneficially owned by the interested stockholder; or

• the receipt by the interested stockholder of the benefit of any loans, advances, guarantees, pledges or other financial benefits by or through the corporation.

In general, Section 203 defines interested stockholder as an entity or person who, together with affiliates and associates, beneficially owns, or within three years prior to the determination of interested stockholder status did own, 15% or more of the outstanding voting stock of the corporation.

Listing on The NASDAQ Global Market

We intend to apply to list our common stock on The NASDAQ Global Market under the symbol "SCOR".

Transfer Agent and Registrar

The transfer agent and registrar for our common stock is Computershare Investor Services. Its address is P.O. Box 43078, Providence, RI 02940, and its telephone number is 1-800-942-5909.

SHARES ELIGIBLE FOR FUTURE SALE

We will have shares of common stock outstanding after the completion of this offering (shares, the shares of common stock sold in the offering (shares if the underwriters exercise their over-allotment option in full) will be freely transferable without restriction, unless purchased by persons deemed to be our "affiliates" as that term is defined in Rule 144 under the Securities Act. Any shares purchased by an affiliate may not be resold except pursuant to an effective registration statement or an applicable exemption from registration, including an exemption under Rule 144 promulgated under the Securities Act. The remaining shares of common stock to be outstanding immediately following the completion of this offering are "restricted," which means they were originally sold in offerings that were not registered under the Securities Act. Restricted shares may be sold through registration under the Securities Act or under an available exemption from registration, such as provided through Rule 144, which rules are summarized below. Taking into account the 180-day lock up agreements described below, and assuming the underwriters do not release any stockholders from these agreements, shares of our common stock will be available for sale in the public market as follows:

- shares will be eligible for sale immediately upon completion of this offering, subject in some cases to volume and other restrictions of Rule 144 and Rule 701 under the Securities Act;
- additional shares will be eligible for sale in the public market under Rule 144 or Rule 701 beginning 90 days after the date of this prospectus, subject to volume, manner of sale, and other limitations under those rules;
- additional shares will become eligible for sale, subject to the provisions of Rule 144, Rule 144(k) or Rule 701, beginning 180 days after the date of this prospectus, upon the
 expiration of agreements not to sell such shares entered into between the underwriters and such stockholders; and
- additional shares will be eligible for sale from time to time thereafter upon expiration of their respective one-year holding periods, but could be sold earlier if the holders
 exercise any available registration rights.

Subject to certain exceptions, each of our officers, directors and security holders has agreed not to offer, sell, contract to sell, pledge or otherwise dispose of, directly or indirectly, any shares of our common stock or securities convertible into or exchangeable or exercisable for any shares of our common stock, enter into a transaction which would have the same effect, or enter into any swap, hedge or other arrangement that transfers, in whole or in part, any of the economic consequences of ownership of our common stock, whether any such aforementioned transaction is to be settled by delivery of our common stock or such other securities, in cash or otherwise, or publicly disclose the intention to make any such offer, sale, pledge or disposition, or to enter into any such transaction, swap, hedge or other arrangement, without, in each case, the prior written consent of Credit Suisse Securities (USA) LLC for a period that shall continue and include the date 180 days after the date of this prospectus. In addition, without the prior written consent of Credit Suisse Securities (USA) LLC, such officers, directors and security holders will not make any demand for or exercise any right with respect to, the registration of any common stock or any security convertible into or exercisable or exchangeable for common stock during such lock-in period.

Notwithstanding the foregoing, for the purpose of allowing the underwriters to comply with NASD Rule 2711(f)(4), if (1) during the last 17 days of the initial 180-day lock-up period, we release earnings results or material news or a material event relating to us occurs or (2) prior to the expiration of the initial 180-day lock-up period, we announce that we will release earnings results during the 16-day period beginning on the last day of the initial 180-day lock-up period, then in each case the initial 180-day lock-up period will be extended until the expiration of the 18-day period beginning on the date of release of the earnings results or the occurrence of the material news or material event, as applicable, unless Credit Suisse Securities (USA) LLC waives, in writing, such extension.

After the offering, the holders of approximately 86,286,692 shares of our issued and outstanding common stock will be entitled to registration rights. For more information on these registration rights, see "Description of Capital Stock — Registration Rights."

In general, under Rule 144, as currently in effect, beginning 90 days after the effective date of this offering, a person (or persons whose shares are aggregated), including an affiliate, who has beneficially owned shares of our common stock for one year or more, may sell in the open market within any three-month period a number of shares that does not exceed the greater of:

- one percent of the then outstanding shares of our common stock (approximately shares immediately after the offering); or
- · the average weekly trading volume in the common stock on The NASDAQ Global Market during the four calendar weeks preceding the sale.

Sales under Rule 144 are also subject to certain limitations on the manner of sale, notice requirements and the availability of our current public information. In addition, a person (or persons whose shares are aggregated) who is deemed not to have been our affiliate at any time during the 90 days preceding a sale by such person and who has beneficially owned his or her shares for at least two years, may sell the shares in the public market under Rule 144(k) without regard to the volume limitations, manner of sale provisions, notice requirements or the availability of current public information we refer to above.

Any of our employees, officers, directors or consultants who purchased his or her shares before the completion of this offering or who holds options as of that date pursuant to a written compensatory plan or contract is entitled to rely on the resale provisions of Rule 701, which permits non-affiliates to sell their Rule 701 shares without having to comply with the public information, holding period, volume limitation or notice provisions of Rule 144 and permits affiliates to sell their Rule 701 shares without having to comply with Rule 144's holding-period restrictions, in each case commencing 90 days after completion of this offering. Neither Rule 144, Rule 144(k) nor Rule 701 supersedes the contractual obligations of our security holders set forth in the lock-up agreements described above.

Within three months following the completion of this offering, we intend to file a registration statement on Form S-8 under the Securities Act to register shares of common stock reserved for issuance under our 1999 Stock Plan and our 2007 Equity Incentive Plan, thus permitting the resale of such shares. Prior to the completion of this offering, there has been no public market for our common stock, and any sale of substantial amounts in the open market may adversely affect the market price of our common stock offered hereby.

U.S. FEDERAL TAX CONSEQUENCES TO NON-U.S. HOLDERS

This section summarizes certain material U.S. federal income and estate tax considerations relating to the ownership and disposition of common stock by non-U.S. holders. This summary does not provide a complete analysis of all potential tax considerations. The information provided below is based on existing authorities. These authorities may change, or the Internal Revenue Service ("IRS") might interpret the existing authorities differently. In either case, the tax considerations of owning or disposing of common stock could differ from those described below. For purposes of this summary, a "non-U.S. holder" is any holder that holds our common stock as a capital asset for U.S. federal income tax purposes and is any holder other than a citizen or resident of the United States, a corporation organized under the laws of the United States or any state, a trust that is (i) subject to the primary supervision of a U.S. court and the control of one of more U.S. persons or (ii) has a valid election in effect under applicable U.S. Treasury regulations to be treated as a U.S. person or an estate whose income is subject to U.S. income tax regardless of source. If a partnership or other flow-through entity is a beneficial owner of common stock, the tax treatment of a partner in the partnership or an owner of the entity will depend upon the status of the partner or other owner and the activities of the partnership or other entity. Accordingly, partnerships that hold our common stock and partners in such partnerships should consult their own tax advisors. The summary generally does not address tax considerations that may be relevant to particular investors because of their specific circumstances, or because they are subject to special rules (such as insurance companies, tax-exempt organizations, financial institutions, brokers, dealers in securities, partnerships, owners of 5% or more of our common stock and certain U.S. expartiates). Finally, the summary does not describe the effects of any applicable foreign, state, or local

Investors considering the purchase of common stock should consult their own tax advisors regarding the application of the U.S. federal income and estate tax laws to their particular situations and the consequences of foreign, state or local laws, and tax treaties.

Dividends

Any dividend paid to a non-U.S. holder on our common stock will generally be subject to U.S. withholding tax at a 30 percent rate. The withholding tax might not apply, however, or might apply at a reduced rate, if the non-U.S. holder satisfies the applicable conditions under the terms of an applicable income tax treaty between the United States and the non-U.S. holder's country of residence. A non-U.S. holder must demonstrate its entitlement to treaty benefits by providing a properly completed Form W-8BEN or appropriate substitute form to us or our paying agent. If the holder holds the stock through a financial institution or other agent acting on the holder's behalf, the holder will be required to provide appropriate documentation to the agent. The holder's agent will then be required to provide certification to us or our paying agent, either directly or through other intermediaries. For payments made to a foreign partnership or other flow through entity, the certification requirements generally apply to the partners or other owners rather than to the partnership or other entity, and the partnership or other entity must provide the partners' or other owners' documentation to us or our paying agent. Special rules, described below, apply if a dividend is effectively connected with a U.S. trade or business conducted by the non-U.S. holder.

Sale of Common Stock

Non-U.S. holders will generally not be subject to U.S. federal income tax on any gains realized on the sale, exchange, or other disposition of common stock. This general rule, however, is subject to several exceptions. For example, the gain would be subject to U.S. federal income tax if:

- the gain is effectively connected with the conduct by the non-U.S. holder of a U.S. trade or business (in which case the special rules described below under the caption "Dividends or Gains Effectively Connected with a U.S. Trade or Business" apply);
- subject to certain exceptions, the non-U.S. holder is an individual who is present in the United States for 183 days or more in the year of disposition, in which case the gain would be subject to a flat 30% tax, which may be offset by U.S. source capital losses, even though the individual is not considered a resident of the U.S.; or

• the rules of the Foreign Investment in Real Property Tax Act, or FIRPTA, described below, treat the gain as effectively connected with a U.S. trade or business.

The FIRPTA rules may apply to a sale, exchange or other disposition of common stock if we are, or were within five years before the transaction, a "U.S. real property holding corporation," or USRPHC. In general, we would be a USRPHC if interests in U.S. real estate comprised most of our assets. We do not believe that we are a USRPHC or that we will become one in the future.

Dividends or Gain Effectively Connected With a U.S. Trade or Business

If any dividend on common stock, or gain from the sale, exchange or other disposition of common stock, is effectively connected with a U.S. trade or business conducted by the non-U.S. holder, then the dividend or gain will be subject to U.S. federal income tax at the regular graduated rates. If the non-U.S. holder is eligible for the benefits of an income tax treaty between the United States and the holder's country of residence, any "effectively connected" dividend or gain would generally be subject to U.S. federal income tax only if it is also attributable to a permanent establishment or fixed base maintained by the holder in the United States. Payments of dividends that are effectively connected with a U.S. trade or business, and therefore included in the gross income of a non-U.S. holder, will not be subject to the 30 percent withholding tax. To claim exemption from withholding, the holder must certify its qualification, which can be done by filing a Form W-8ECI. If the non-U.S. holder is a corporation, that portion of its earnings and profits that is effectively connected with its U.S. trade or business would generally be subject to a "branch profits tax" in addition to any regular U.S. federal income tax on the dividend or gain. The branch profits tax rate is generally 30 percent, although an applicable income tax treaty might provide for a lower rate.

U.S. Federal Estate Tax

The estates of nonresident alien individuals are generally subject to U.S. federal estate tax on property with a U.S. situs. Because we are a U.S. corporation, our common stock will be U.S. situs property and therefore will be included in the taxable estate of a nonresident alien decedent. The U.S. federal estate tax liability of the estate of a nonresident alien may be affected by a tax treaty between the United States and the decedent's country of residence.

Backup Withholding and Information Reporting

The Internal Revenue Code of 1986, as amended, and the Treasury regulations promulgated thereunder require those who make specified payments to report the payments to the IRS. Among the specified payments are dividends and proceeds paid by brokers to their customers. The required information returns enable the IRS to determine whether the recipient properly included the payments in income. This reporting regime is reinforced by "backup withholding" rules. These rules require the payors to withhold tax from payments subject to information reporting if the recipient fails to cooperate with the reporting regime by failing to provide his taxpayer identification number to the payor, furnishing an incorrect identification number, or repeatedly failing to report interest or dividends on his returns. The withholding tax rate is currently 28 percent. The backup withholding rules do not apply to payments to corporations, whether domestic or foreign.

Payments to non-U.S. holders of dividends on common stock will generally not be subject to backup withholding, and payments of proceeds made to non-U.S. holders by a broker upon a sale of common stock will not be subject to information reporting or backup withholding, in each case so long as the non-U.S. holder certifies its nonresident status. Some of the common means of certifying nonresident status are described under "— Dividends." We must report annually to the IRS any dividends paid to each non-U.S. holder and the tax withheld, if any, with respect to such dividends. Copies of these reports may be made available to tax authorities in the country where the non-U.S. holder resides.

Information reporting and backup withholding also generally will not apply to a payment of the proceeds of a sale of common stock effected outside the United States by a foreign office of a foreign broker. However, information reporting requirements (but not backup withholding) will apply to a payment of the proceeds of a sale of common stock effected outside the United States by a foreign office of a broker if the broker (i) is a

United States person, (ii) derives 50 percent or more of its gross income for certain periods from the conduct of a trade or business in the United States, (iii) is a "controlled foreign corporation" as to the United States, or (iv) is a foreign partnership that, at any time during its taxable year is more than 50 percent (by income or capital interest) owned by United States persons or is engaged in the conduct of a U.S. trade or business, unless in any such case the broker has documentary evidence in its records that the holder is a non-U.S. holder and certain conditions are met, or the holder otherwise establishes an exemption. Payment by a United States office of a broker of the proceeds of a sale of common stock will be subject to both backup withholding and information reporting unless the holder certifies its non-United States status under penalties of perjury or otherwise establishes an exemption.

Any amounts withheld from a payment to a holder of common stock under the backup withholding rules can be credited against any U.S. federal income tax liability of the holder.

The preceding discussion of U.S. federal income tax considerations is for general information only and it is not tax advice. Each prospective investor should consult its own tax advisor regarding the particular U.S. federal, state, local and foreign tax consequences of purchasing, holding and disposing of our common stock, including the consequences of any proposed change in applicable laws.

UNDERWRITING

Under the terms and subject to the conditions contained in an underwriting agreement dated , 2007, we and the selling stockholders have agreed to sell to the underwriters named below, for whom Credit Suisse Securities (USA) LLC is acting as representative, the following respective numbers of shares of common stock:

<u>U</u> nderwriter	Number of Shares
Credit Suisse Securities (USA) LLC	
Deutsche Bank Securities Inc.	
Friedman, Billings, Ramsey & Co., Inc.	
Jefferies & Company, Inc.	
William Blair & Company, L.L.C.	
Total	

The underwriting agreement provides that the underwriters are obligated to purchase all the shares of common stock in the offering if any are purchased, other than those shares covered by the over-allotment option described below. The underwriting agreement also provides that, if an underwriter defaults, the purchase commitments of non-defaulting underwriters may be increased or the offering may be terminated.

We and the selling stockholders have granted to the underwriters a 30-day option to purchase on a pro rata basis up to additional shares from us and the selling stockholders at the initial public offering price less the underwriting discounts and commissions. The option may be exercised only to cover any over-allotments of common stock.

The underwriters propose to offer the shares of common stock initially at the public offering price on the cover page of this prospectus and to selling group members at that price less a selling concession of \$ per share. After the initial public offering Credit Suisse Securities (USA) LLC may change the public offering price and concession.

The following table summarizes the compensation and estimated expenses we and the selling stockholders will pay:

	Per Share			Total
	Without Over-allotment	With Over-allotment	Without Over-allotment	With Over-allotment
Underwriting Discounts and Commissions paid by us	\$	\$	\$	\$
Expenses payable by us	\$	\$	\$	\$
Underwriting Discounts and Commissions paid by selling stockholders	\$	\$	\$	\$
Expenses payable by the selling stockholders	\$	\$	\$	\$

Credit Suisse Securities (USA) LLC has informed us that they do not expect sales to accounts over which the underwriters have discretionary authority to exceed 5% of the shares of common stock being offered.

We have agreed that we will not offer, sell, contract to sell, pledge or otherwise dispose of, directly or indirectly, or file with the Securities and Exchange Commission, or SEC, a registration statement under the Securities Act relating to, any shares of our common stock or securities convertible into or exchangeable or exercisable for any shares of our common stock, or publicly disclose the intention to make any offer, sale, pledge, disposition or filing, without the prior written consent of Credit Suisse Securities (USA) LLC for a period of 180 days after the date of this prospectus, except (a) issuances by us pursuant to the exercise of employee stock options outstanding on the date hereof or pursuant to our dividend reinvestment plan and (b) up to shares of our common stock that may be sold at our permission by certain existing and former

employees designated by us. However, in the event that either (1) during the last 17 days of the "lock-up" period, we release earnings results or material news or a material event relating to us occurs or (2) prior to the expiration of the "lock-up" period, we announce that we will release earnings results during the 16-day period beginning on the last day of the "lock-up" period, then in either case the expiration of the "lock-up" will be extended until the expiration of the 18-day period beginning on the date of the release of the earnings results or the occurrence of the material news or event, as applicable, unless Credit Suisse Securities (USA) LLC waives, in writing, such an extension.

Subject to certain exceptions, our officers, directors and certain of our existing security holders have agreed that they will not offer, sell, contract to sell, pledge or otherwise dispose of, directly or indirectly, any shares of our common stock or securities convertible into or exchangeable or exercisable for any shares of our common stock, enter into a transaction that would have the same effect, or enter into any swap, hedge or other arrangement that transfers, in whole or in part, any of the economic consequences of ownership of our common stock, whether any of these transactions are to be settled by delivery of our common stock or other securities, in cash or otherwise, or publicly disclose the intention to make any offer, sale, pledge or disposition, or to enter into any transaction, swap, hedge or other arrangement, without, in each case, the prior written consent of Credit Suisse Securities (USA) LLC for a period of 180 days after the date of this prospectus. However, in the event that either (1) during the last 17 days of the "lock-up" period, we release earnings results or material news or a material event relating to us occurs or (2) prior to the expiration of the "lock-up" period, we announce that we will release earnings results during the 16-day period beginning on the last day of the "lock-up" period, then in either case the expiration of the "lock-up" will be extended until the expiration of the 18-day period beginning on the date of the release of the earnings results or the occurrence of the material news or event, as applicable, unless Credit Suisse Securities (USA) LLC waives, in writing, such an extension.

We and the selling stockholders have agreed to indemnify the underwriters against liabilities under the Securities Act, or contribute to payments that the underwriters may be required to make in that respect.

We intend to apply to list the shares of common stock on The NASDAQ Global Market under the symbol "SCOR".

Some of the underwriters and their affiliates have provided, and may provide in the future, investment banking and other financial services for us in the ordinary course of business for which they have received and would receive customary compensation.

Prior to this offering, there has been no public market for our common stock. The initial public offering price has been determined by a negotiation between us and Credit Suisse Securities (USA) LLC and will not necessarily reflect the market price of our common stock following the offering. The principal factors that were considered in determining the public offering price included:

- the information presented in this prospectus and otherwise available to the underwriters;
- · the history of and prospects for the industry in which we compete;
- · the ability of our management;
- · the prospects for our future earnings;
- · the present state of our development and our current financial condition;
- · the recent market prices of, and the demand for, publicly traded common stock of generally comparable companies; and
- · the general condition for the securities markets at the time of this offering.

In connection with the offering the underwriters may engage in stabilizing transactions, over-allotment transactions, syndicate covering transactions and penalty bids in accordance with Regulation M under the Securities Exchange Act of 1934, or the Exchange Act.

Table of Contents

- Stabilizing transactions permit bids to purchase the underlying security so long as the stabilizing bids do not exceed a specified maximum.
- Over-allotment involves sales by the underwriters of shares in excess of the number of shares the underwriters are obligated to purchase, which creates a syndicate short position. The short position may be either a covered short position or a naked short position. In a covered short position, the number of shares over-allotted by the underwriters is not greater than the number of shares that they may purchase in the over-allotment option. In a naked short position, the number of shares involved is greater than the number of shares in the over-allotment option. The underwriters may close out any covered short position by either exercising their over-allotment option and/or purchasing shares in the open market.
- Syndicate covering transactions involve purchases of the common stock in the open market after the distribution has been completed in order to cover syndicate short positions. In determining the source of shares to close out the short position, the underwriters will consider, among other things, the price of shares available for purchase in the open market as compared to the price at which they may purchase shares through the over-allotment option. If the underwriters sell more shares than could be covered by the over-allotment option, a naked short position, the position can only be closed out by buying shares in the open market. A naked short position is more likely to be created if the underwriters are concerned that there could be downward pressure on the price of the shares in the open market after pricing that could adversely affect investors who purchase in the offering.
- Penalty bids permit Credit Suisse Securities (USA) LLC to reclaim a selling concession from a syndicate member when the common stock originally sold by the syndicate
 member is purchased in a stabilizing or syndicate covering transaction to cover syndicate short positions.

These stabilizing transactions, syndicate covering transactions and penalty bids may have the effect of raising or maintaining the market price of our common stock or preventing or retarding a decline in the market price of the common stock. As a result the price of our common stock may be higher than the price that might otherwise exist in the open market. These transactions may be effected on The NASDAQ Global Market and, if commenced, may be discontinued at any time.

A prospectus in electronic format may be made available on the Web sites maintained by one or more of the underwriters, or selling group members, if any, participating in this offering and one or more of the underwriters participating in this offering may distribute prospectuses electronically. Credit Suisses Securities (USA) LLC may agree to allocate a number of shares to underwriters and selling group members for sale to their online brokerage account holders. Internet distributions will be allocated by the underwriters and selling group members that will make Internet distributions on the same basis as other allocations.

The common stock is being offered for sale in those jurisdictions in the United States, Europe and elsewhere where it is lawful to make such offers.

In relation to each Member State of the European Economic Area that has implemented the Prospectus Directive (each, a "Relevant Member State"), each underwriter represents and agrees that with effect from and including the date on which the Prospectus Directive is implemented in that Relevant Member State (the "Relevant Implementation Date") it has not made and will not make an offer of shares to the public in that Relevant Member State prior to the publication of a prospectus in relation to the shares which has been approved by the competent authority in that Relevant Member State or, where appropriate, approved in another Relevant Member State and notified to the competent authority in that Relevant Member State, all in accordance with the Prospectus Directive, except that it may, with effect from and including the Relevant Implementation Date, make an offer of shares to the public in that Relevant Member State at any time:

- (a) to legal entities that are authorized or regulated to operate in the financial markets or, if not so authorized or regulated, whose corporate purpose is solely to invest in securities;
- (b) to any legal entity that has two or more of (1) an average of at least 250 employees during the last financial year; (2) a total balance sheet of more than €43,000,000 and (3) an annual net turnover of more than €50,000,000, as shown in its last annual or consolidated accounts;

Table of Contents

- (c) to fewer than 100 natural or legal persons (other than qualified investors as defined in the Prospectus Directive) subject to obtaining the prior consent of the manager for any such offer; or
- d) in any other circumstances that do not require the publication by us of a prospectus pursuant to Article 3 of the Prospectus Directive.

For the purposes of this provision, the expression an "offer of shares to the public" in relation to any shares in any Relevant Member State means the communication in any form and by any means of sufficient information on the terms of the offer and the shares to be offered so as to enable an investor to decide to purchase or subscribe the shares, as the same may be varied in that Member State by any measure implementing the Prospectus Directive in that Member State and the expression "Prospectus Directive" means Directive 2003/71/EC and includes any relevant implementing measure in each Relevant Member State.

Each of the underwriters has represented and agreed that:

- (a) it has only communicated or caused to be communicated and will only communicate or cause to be communicated an invitation or inducement to engage in investment activity (within the meaning of section 21 of FSMA) to persons who have professional experience in matters relating to investments falling with Article 19(5) of the Financial Services and Markets Act 2000 (Financial Promotion) Order 2005 or in circumstances in which section 21 of FSMA does not apply to us; and
- (b) it has complied with, and will comply with, all applicable provisions of FSMA with respect to anything done by it in relation to the common stock in, from or otherwise involving the United Kingdom.

NOTICE TO CANADIAN RESIDENTS

Resale Restrictions

The distribution of the common stock in Canada is being made only on a private placement basis exempt from the requirement that we and the selling stockholders prepare and file a prospectus with the securities regulatory authorities in each province where trades of common stock are made. Any resale of the common stock in Canada must be made under applicable securities laws, which will vary depending on the relevant jurisdiction, and which may require resales to be made under available statutory exemptions or under a discretionary exemption granted by the applicable Canadian securities regulatory authority. Purchasers are advised to seek legal advice prior to any resale of shares of the common stock.

Representations of Purchasers

By purchasing common stock in Canada and accepting a purchase confirmation, a purchaser is representing to us, the selling stockholders and the dealer from whom the purchase confirmation is received that:

- the purchaser is entitled under applicable provincial securities laws to purchase the common stock without the benefit of a prospectus qualified under those securities laws;
- · where required by law, that the purchaser is purchasing as principal and not as agent;
- the purchaser has reviewed the text above under "Resale Restrictions"; and
- the purchaser acknowledges and consents to the provision of specified information concerning its purchase of common stock to the regulatory authority that by law is entitled to collect the information.

Further details concerning the legal authority for this information is available on request.

Rights of Action - Ontario Purchasers Only

Under Ontario securities legislation, certain purchasers who purchase a security offered by this prospectus during the period of distribution will have a statutory right of action for damages or, while still the owner of shares of common stock, for rescission against us and the selling stockholders in the event that this prospectus contains a misrepresentation without regard to whether the purchaser relied on the misrepresentation. The right of action for damages is exercisable not later than the earlier of 180 days from the date the purchaser first had knowledge of the facts giving rise to the cause of action and three years from the date on which payment is made for shares of common stock. The right of action for rescission, the purchaser will have no right of action for damages against us or the selling stockholders. In no case will the amount recoverable in any action exceed the price at which shares of common stock were offered to the purchaser and if the purchaser is shown to have purchased the securities with knowledge of the misrepresentation, we and the selling stockholders will have no liability. In the case of an action for damages, we and the selling stockholders will not be liable for all or any portion of the damages that are proven to not represent the depreciation in value of the common stock as a result of the misrepresentation relied upon. These rights are in addition to, and without derogation from, any other rights or remedies available at law to an Ontario purchaser. The foregoing is a summary of the rights available to an Ontario purchaser. Ontario purchasers should refer to the complete text of the relevant statutory provisions.

Enforcement of Legal Rights

All of our directors and officers as well as the experts named herein and the selling stockholders may be located outside of Canada and, as a result, it may not be possible for Canadian purchasers to effect service of process within Canada upon us or those persons. All or a substantial portion of our assets and the assets of those persons may be located outside of Canada and, as a result, it may not be

Table of Contents

possible to satisfy a judgment against us or those persons in Canada or to enforce a judgment obtained in Canadian courts against us or those persons outside of Canada.

Taxation and Eligibility for Investment

Canadian purchasers of our common stock should consult their own legal and tax advisors with respect to the tax consequences of an investment in the common stock in their particular circumstances and about the eligibility of the common stock for investment by the purchaser under relevant Canadian legislation.

LEGAL MATTERS

The validity of the shares of common stock offered hereby has been passed upon for comScore, Inc. by Wilson Sonsini Goodrich & Rosati, Professional Corporation, Washington, D.C. The underwriters have been represented in connection with this offering by Cravath, Swaine & Moore LLP, New York, New York. Certain members of, investment partnerships comprised of members of, and persons associated with, Wilson Sonsini Goodrich & Rosati, Professional Corporation beneficially hold an aggregate of 151,083 shares of our common stock on an as-converted basis.

EXPERTS

Ernst & Young LLP, independent registered public accounting firm, has audited our consolidated financial statements and schedule at December 31, 2005, and 2006, and for each of the three years in the period ended December 31, 2006, as set forth in their reports. We have included our consolidated financial statements and schedule in this prospectus and elsewhere in the registration statement in reliance on Ernst & Young LLP's reports, given on their authority as experts in accounting and auditing.

WHERE YOU CAN FIND MORE INFORMATION

We have filed a registration statement on Form S-1 with the SEC for the common stock we are offering pursuant to this prospectus. This prospectus does not include all of the information contained in the registration statement. You should refer to the registration statement and its exhibits for additional information. Whenever we make reference in this prospectus to any of our contracts, agreements or other documents, the references are summaries and are not necessarily complete and you should refer to the exhibits attached to the registration statement for copies of the actual contract, agreement or other document. When we complete this offering, we will also be required to file annual, quarterly and special reports, proxy statements and other information with the SEC.

You can read our SEC filings, including the registration statement, over the Internet at the SEC's Web site at www.sec.gov. You may also read and copy any document we file with the SEC at its public reference facilities at 100 F Street, N.E., Room 1580, Washington, D.C. 20549. You may also obtain copies of the documents at prescribed rates by writing to the Public Reference Section of the SEC at 100 F Street, N.E., Room 1580, Washington, D.C. 20549. Please call the SEC at 1-800-SEC-0330 for further information on the operation of the public reference facilities.

INDEX TO CONSOLIDATED FINANCIAL STATEMENTS

	Page
comScore, Inc. consolidated financial statements as of December 31, 2005 and 2006 and for the years ended December 31, 2004, 2005 and 2006	
Report of Independent Registered Public Accounting Firm	F-2
Consolidated balance sheets as of December 31, 2005 and 2006	F-3
Consolidated statements of operations for the years ended December 31, 2004, 2005 and 2006	F-5
Consolidated statements of stockholders' deficit for the years ended December 31, 2004, 2005	
and 2006	F-6
Consolidated statements of cash flows for the years ended December 31, 2004, 2005 and 2006	F-7
Notes to consolidated financial statements	F-8

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

Board of Directors comScore, Inc.

We have audited the accompanying consolidated balance sheets of comScore, Inc. (the Company) as of December 31, 2005 and 2006, and the related consolidated statements of operations, stockholders' deficit, and cash flows for each of the three years in the period ended December 31, 2006. These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the financial statements based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. We were not engaged to perform an audit of the Company's internal control over financial reporting. Our audits included consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the consolidated financial position of comScore, Inc. at December 31, 2005 and 2006, and the consolidated results of its operations and its cash flows for each of the three years in the period ended December 31, 2006, in conformity with U.S. generally accepted accounting principles.

As discussed in Note 2 to the consolidated financial statements, the Company adopted FASB Staff Position 150-5, Issuer's Accounting Under FASB Statement No. 150 for Freestanding Warrants and Other Similar Instruments on Shares That Are Redeemable, effective July 1, 2005, and changed its method of accounting for stock-based compensation in accordance with guidance provided in FASB Statement No. 123(R), Share-Based Payments, effective January 1, 2006.

/s/ Ernst & Young LLP

March 29, 2007 McLean, Virginia

CONSOLIDATED BALANCE SHEETS

	_	2005	nber 31, ousands)	2006
Assets				
Current assets:				
Cash and cash equivalents	\$	5,124	\$	5,032
Short-term investments		4,050		11,000
Accounts receivable, net of allowances of \$185 and \$188, respectively		10,328		14,123
Prepaid expenses and other current assets		1,029		1,068
Restricted cash		261		270
Total current assets		20,792		31,493
Property and equipment, net		4,480		6,980
Other non-current assets		786		1,267
Intangible assets, net		2,355		983
Goodwill		1,064		1,364
Total assets	\$	29,477	\$	42,087

CONSOLIDATED BALANCE SHEETS — (Continued)

	Decemb	
	2005 (In thou except sha	
Liabilities and stockholders' deficit		
Current liabilities:		
Accounts payable	\$ 1,048	\$ 1,353
Accrued expenses	4,185	6,020
Deferred revenues	19,588	22,776
Capital lease obligations	1,618	1,726
Preferred stock warrant liabilities	781	1,005
Total current liabilities	27,220	32,880
Capital lease obligations, long-term	1,283	2,261
Deferred tax liability	174	77
Other liabilities	362	374
Total liabilities	29,039	35,592
Commitments and contingencies		
Redeemable preferred stock:		
Series A preferred convertible stock, \$0.001 par value; 9,187,500 shares authorized; 9,187,500 shares issued and outstanding; liquidation preference		
of \$7,715 at December 31, 2006	8,443	8,154
Series B preferred convertible stock, \$0.001 par value; 3,535,486 shares authorized; 3,479,241 shares issued and outstanding; liquidation preference		
of \$14,315 at December 31, 2006	15,668	15,130
Series C preferred convertible stock, \$0.001 par value; 13,355,052 shares authorized; 13,236,018 shares issued and outstanding; liquidation		
preference of \$25,220 at December 31, 2006	27,565	26,633
Series C-1 preferred convertible stock, \$0.001 par value; 357,144 shares authorized; 357,144 shares issued and outstanding; liquidation preference of		
\$420 at December 31, 2006	458	443
Series D preferred convertible stock, \$0.001 par value; 22,238,042 shares authorized; 21,564,020 shares issued and outstanding; liquidation		
preference of \$40,723 at December 31, 2006	31,337	34,682
Series E preferred convertible stock, \$0.001 par value; 25,000,000 shares authorized; 24,005,548 shares issued and outstanding; liquidation		
preference of \$19,565 at December 31, 2006	15,045	16,653
Common Stock subject to put; 1,738,172 shares issued and outstanding	4,216	4,357
Stockholders' deficit:		
Common stock, \$0.001 par value; 130,000,000 shares authorized; 16,737,440 and 20,000,813 shares issued and outstanding at December 31, 2005		
and 2006, respectively	17	20
Additional paid-in capital	_	_
Deferred stock compensation	(6)	
Accumulated other comprehensive loss	(24)	(75)
Accumulated deficit	(102,281)	(99,502)
Total stockholders' deficit	(102,294)	(99,557)
Total liabilities and stockholders' deficit	\$ 29,477	\$ 42,087

CONSOLIDATED STATEMENTS OF OPERATIONS

	Years Ended December 31,					
	2004 2005 (In thousands, except share and per share of			J	2006	
	•	,		• •		
Revenues	\$	34,894	\$	50,267	\$	66,293
Cost of revenues (excludes amortization of intangible assets resulting from acquisitions shown below)(1)		13,153		18,218		20,560
Selling and marketing(1)		13,890		18,953		21,473
Research and development(1)		5,493		7,416		9,009
General and administrative(1)		4,982		7,089		8,293
Amortization of intangible assets resulting from acquisitions		356		2,437		1,371
Total expenses from operations		37,874		54,113		60,706
(Loss) income from operations		(2,980)		(3,846)		5,587
Interest (expense) income, net		(246)		(208)		231
(Loss) gain from foreign currency		_		(96)		125
Revaluation of preferred stock warrant liabilities				(14)		(224)
(Loss) income before income taxes and cumulative effect of change in accounting principle		(3,226)		(4,164)		5,719
(Benefit) provision for income taxes				(182)		50
Net (loss) income before cumulative effect of change in accounting principle		(3,226)		(3,982)		5,669
Cumulative effect of change in accounting principle		` —		(440)		_
Net (loss) income		(3,226)		(4,422)		5,669
Accretion of redeemable preferred stock		(2,141)		(2,638)		(3,179)
Net (loss) income attributable to common stockholders	\$	(5,367)	\$	(7,060)	\$	2,490
Net (loss) income attributable to common stockholders per common share:						
Basic and diluted	\$	(0.38)	\$	(0.46)	\$	0.00
Weighted-average number of shares used in per share calculation — common stock:						
Basic and diluted		14,358,561		15,650,969		19,236,064
Net (loss) income attributable to common stockholders per common share subject to put:						
Basic and diluted	\$	0.07	\$	0.08	\$	80.0
Weighted-average number of shares used in per share calculation — common share subject to put:						
Basic and diluted		457,596		1,738,172		1,738,172
(1) Amortization of stock-based compensation is included in the line items above as follows:						
Cost of revenues	\$	_	\$	_	\$	12
Selling and marketing		_		_		82
Research and development		_		_		13
General and administrative		14		3		91
Total	\$	14	\$	3	\$	198

CONSOLIDATED STATEMENTS OF STOCKHOLDERS' DEFICIT

	Common S Shares	Stock Amount	Additional Paid-In Capital	Deferred Stock Compensation (In thousands, e	Accumulated Other Comprehensive Income (Loss) except share data)	Accumulated Deficit	Total Stockholders' Deficit
Balance at December 31, 2003	13,729,967	\$ 14	s —	\$ (10)	\$ 30	\$ (89,953)	\$ (89,919)
Net loss	_	_	_	_	_	(3,226)	(3,226)
Foreign currency translation adjustment	_	_	_	_	(19)	<u> </u>	(19) 123
Exercise of common stock options	2,403,710	2	121	_	_	_	123
Repurchase of options previously issued	(928,125)	(1)	(45)	_	_	_	(46)
Amortization of deferred stock compensation	_	_	_	1	_	_	1
Issuance of common stock warrants	_	_	13	_	_	_	13
Accretion of redeemable preferred stock warrants	_	_	16	_	_	_	16
Accretion of redeemable preferred stock	_	_	(105)	_	_	(2,036)	(2,141)
Accretion of common stock subject to put						(32)	(32)
Balance at December 31, 2004	15,205,552	15	_	(9)	11	(95,247)	(95,230)
Net loss	_	_	_	_	_	(4,422)	(4,422)
Foreign currency translation adjustment	_	_	_	_	(35)		(35)
Exercise of common stock options	1,531,888	2	134	_	_	_	136
Amortization of deferred stock compensation	_	_	_	3	_	_	3
Issuance of common stock warrants	_	_	16	_	_	_	16
Accretion of redeemable preferred stock warrants	_	_	8	_	_	_	8
Accretion of redeemable preferred stock	_	_	(158)	_	_	(2,480)	(2,638)
Accretion of common stock subject to put						(132)	(132)
Balance at December 31, 2005	16,737,440	17		(6)	(24)	(102,281)	(102,294)
Net income		_	_	<u> </u>		5,669	5,669
Foreign currency translation adjustment	_	_	_	_	(51)		(51)
Exercise of common stock options	3,263,373	3	238	_		_	241
Amortization of deferred stock compensation	_	_	_	6	_	(3)	3
Amortization of stock-based compensation	_	_	195	_	_		195
Accretion of redeemable preferred stock	_	_	(433)	_	_	(2,746)	(3,179)
Accretion of common stock subject to put	_	_	_	_	_	(141)	(141)
Balance at December 31, 2006	20,000,813	\$ 20		s —	\$ (75)	\$ (99,502)	\$ (99,557)

CONSOLIDATED STATEMENTS OF CASH FLOWS

	Ye	Years Ended December 31,		
	2004	2005	2006	
		(In thousands)		
Operating activities	4 (2.000)	A (1.100)		
Net (loss) income	\$ (3,226)	\$ (4,422)	\$ 5,669	
Adjustments to reconcile net (loss) income to net cash provided by operating activities:				
Depreciation	2,389	2,686	2,888	
Amortization of intangible assets resulting from acquisitions	356	2,437	1,371	
Provisions for bad debts	12	90	212	
Stock-based compensation	14	3	198	
Revaluation of preferred stock warrant liabilities	_	14	224	
Cumulative effect of change in accounting principle	_	440	_	
Amortization of deferred finance costs	30	33	33	
Deferred tax benefit	_	(182)	(97)	
Changes in operating assets and liabilities, net of effect of acquisitions:				
Accounts receivable, net	(736)	(3,540)	(3,882)	
Prepaid expenses and other current assets	539	(157)	(311)	
Other non-current assets	174	539	30	
Accounts payable and accrued expenses	1,689	(245)	1,417	
Deferred revenues	608	6,427	3,139	
Other liabilities	58	130	14	
Net cash provided by operating activities	1,907	4,253	10,905	
Investing activities				
Payment of restricted cash	_	(41)	(9)	
Purchase of short-term investments	(5,600)	(8,960)	(14,900)	
Sale of short-term investments	6,400	8,810	7,950	
Purchase of property and equipment	(1,208)	(1,071)	(2,314)	
Acquisition of businesses, net of cash acquired of \$715 in 2005	(924)	(943)	_	
Payment of additional consideration for acquired businesses	_	(300)	(300)	
Net cash used in investing activities	(1,332)	(2,505)	(9,573)	
Financing activities	(/ /	()	(-)/	
Proceeds from the exercise of common stock options	123	136	241	
Repurchase of previously issued stock options	(46)	_	_	
Principal payments on capital lease obligations	(1,029)	(1,228)	(1,622)	
Net cash used in financing activities	(952)	(1,092)	(1,381)	
Effect of exchange rate changes on cash	25	(36)	(43)	
Net (decrease) increase in cash and cash equivalents	(352)	620	(92)	
Cash and cash equivalents at beginning of year	4,856	4,504	5,124	
Cash and cash equivalents at end of year	\$ 4,504	\$ 5,124	\$ 5,032	
Supplemental cash flow disclosures	¢ 252	¢ 214	¢ 240	
Interest paid	\$ 353	\$ 314	\$ 249	
Capital lease obligations incurred	\$ -	\$ 1,704	\$ 2,707	
Accretion of preferred stock	\$ 2,141	\$ 2,638	\$ 3,179	

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

1. Organization

comScore, Inc. (the Company), a Delaware corporation incorporated in August 1999, provides a digital marketing intelligence platform that helps customers make better-informed business decisions and implement more effective digital business strategies. The Company's products and solutions offer customers insights into consumer behavior, including objective, detailed information regarding usage of their online properties and those of their competitors, coupled with information on consumer demographic characteristics, attitudes, lifestyles and offline behavior.

The Company's digital marketing intelligence platform is comprised of proprietary databases and a computational infrastructure that measures, analyzes and reports on digital activity. The foundation of the platform is data collected from a panel of more than two million Internet users worldwide who have granted to the Company explicit permission to confidentially measure their Internet usage patterns, online and certain offline buying behavior and other activities. By applying advanced statistical methodologies to the panel data, the Company projects consumers' online behavior for the total online population and a wide variety of user categories.

2. Summary of Significant Accounting Policies

Basis of Presentation and Consolidation

The accompanying consolidated financial statements include the accounts of the Company and its wholly-owned subsidiaries. All significant intercompany transactions and accounts have been eliminated upon consolidation. The Company consolidates investments where it has a controlling financial interest as defined by Accounting Research Bulletin (ARB) No. 51, Consolidated Financial Statements, as amended by Statement of Financial Accounting Standards (SFAS) No. 94, Consolidation of all Majority-Owned Subsidiaries. The usual condition for controlling financial interest is ownership of a majority of the voting interest and, therefore, as a general rule, ownership, directly or indirectly, of more than 50% of the outstanding voting shares is a condition indicating consolidation. For investments in variable interest entities, as defined by Financial Accounting Standards Board (FASB) Interpretation No. 46, Consolidation of Variable Interest Entities, the Company would consolidate when it is determined to be the primary beneficiary of a variable interest entity. The Company does not have any variable interest entities.

Use of Estimates

The preparation of financial statements in conformity with accounting principles generally accepted in the United States requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ materially from those estimates.

Reclassification

Certain amounts in the prior years' financial statements have been reclassified to conform to the current year presentation.

Cash and Cash Equivalents, Short-Term Investments, and Restricted Cash

Cash and cash equivalents and restricted cash consist of highly liquid investments with an original maturity of three months or less at the time of purchase. Cash, cash equivalents, and restricted cash consists primarily of money market accounts.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Short-term investments, which consist principally of high-grade auction rate securities, are stated at fair market value, which approximates cost. These securities are accounted for as available-for-sale securities in accordance with SFAS No. 115, *Accounting for Certain Investments in Debt and Equity Securities*. The Company typically has the option to re-invest in its short-term investments every 30 days. The Company uses the specific identification method to compute realized gains and losses on its short-term investments.

Restricted cash is comprised of a certificate of deposit that is collateral for a letter of credit pertaining to the security deposit for an operating lease.

Interest income on short-term investments was \$100,000, \$133,000 and \$515,000 for the years ended December 31, 2004, 2005 and 2006, respectively.

Accounts Receivable

Accounts receivable are recorded at the invoiced amount and are non-interest bearing. The Company generally grants uncollateralized credit terms to its customers and maintains an allowance for doubtful accounts to reserve for potentially uncollectible receivables. Allowances are based on management's judgment, which considers historical experience and specific knowledge of accounts where collectibility may not be probable. The Company makes provisions based on historical bad debt experience, a specific review of all significant outstanding invoices and an assessment of general economic conditions. If the financial condition of a customer deteriorates, resulting in an impairment of its ability to make payments, additional allowances may be required.

Property and Equipment

Property and equipment is stated at cost, net of accumulated depreciation. Property and equipment is depreciated on a straight-line basis over the estimated useful lives of the assets, ranging from three to five years. Assets under capital leases are recorded at their net present value at the inception of the lease and are included in the appropriate asset category. Assets under capital leases and leasehold improvements are amortized over the shorter of the related lease terms or their useful lives. Replacements and major improvements are capitalized; maintenance and repairs are charged to expense as incurred. Amortization of assets under capital leases is included within the expense category on the Statement of Operations in which the asset is deployed.

Goodwill and Intangible Assets

Goodwill represents the excess of the purchase price over the fair value of identifiable assets acquired and liabilities assumed when other businesses are acquired. The allocation of the purchase price to intangible assets and goodwill involves the extensive use of management's estimates and assumptions, and the result of the allocation process can have a significant impact on future operating results. Under SFAS No. 142, Goodwill and Other Intangible Assets (SFAS 142), intangible assets with finite lives are amortized over their useful lives while goodwill and indefinite lived assets are not amortized but are evaluated for potential impairment at least annually by comparing the fair value of a reporting unit, based on estimated future cash flows, to its carrying value including goodwill recorded by the reporting unit. If the carrying value exceeds the fair value, impairment is measured by comparing the derived fair value of the goodwill to its carrying value, and any impairment determined is recorded in the current period. In accordance with SFAS 142, all of the Company's goodwill is associated with one reporting unit. Accordingly, on an annual basis the Company performs the impairment assessment for goodwill required under SFAS 142 at the enterprise level. The Company completed its annual impairment analysis for 2004, 2005 and 2006 and determined that there was no impairment of goodwill.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Intangible assets with finite lives are amortized using the straight-line method over the following useful lives:

	Useful Lives (Years)
Non-compete agreements	3 to 4
Customer relationships	1 to 3
Acquired methodologies/technology	1 to 3
Trademarks and brands	2

Impairment of Long-Lived Assets

Long-lived assets, including property and equipment, are reviewed for impairment whenever events or changes in circumstances indicate that the carrying amount should be addressed pursuant to SFAS No. 144, Accounting for the Impairment or Disposal of Long-Lived Assets (SFAS 144). Pursuant to SFAS 144, impairment is determined by comparing the carrying value of these long-lived assets to an estimate of the future undiscounted cash flows expected to result from the use of the assets and eventual disposition. In the event an impairment exists, a loss is recognized based on the amount by which the carrying value exceeds the fair value of the asset, which is generally determined by using quoted market prices or valuation techniques such as the discounted present value of expected future cash flows, appraisals, or other pricing models as appropriate. There were no impairment charges recognized during the years ended December 31, 2004, 2005 and 2006. In the event that there are changes in the planned use of the Company's long-term assets or its expected future undiscounted cash flows are reduced significantly, the Company's assessment of its ability to recover the carrying value of these assets could change.

Foreign Currency Translation

The Company applies SFAS No. 52, Foreign Currency Translation, with respect to its international operations. The functional currency of the Company's foreign subsidiaries is the local currency. All assets and liabilities are translated at the current exchange rate as of the end of the period, and revenues and expenses are translated at average exchange rates in effect during the period. The gain or loss resulting from the process of translating foreign currency financial statements into U.S. dollars is included as a component of other comprehensive income. The Company incurred a foreign currency transaction loss of \$96,000 for the year ended December 31, 2005 and a gain of \$125,000 for the year ended December 31, 2006. These gains and losses related to U.S. dollar denominated cash accounts and accounts receivable held by the Company's foreign subsidiaries. Foreign currency transaction losses were not material in 2004.

Business Segment Information

The Company is managed and operated as one business segment. A single management team reports to the chief operating decision maker who manages the entire business. The Company does not operate any material separate lines of business or separate business entities with respect to its services. The various products that the Company offers are all related to analyzing consumer behavior on the Internet. The same data source is used regardless of the product delivered. The Company's expenses are shared and are not allocated to individual products. Accordingly, the Company does not accumulate discrete financial information by product line and does not have separately reportable segments as defined by SFAS No. 131, Disclosure About Segments of an Enterprise and Related Information.

Revenue Recognition

The Company recognizes revenues in accordance with Securities and Exchange Commission Staff Accounting Bulletin (SAB) No. 104, Revenue Recognition (SAB 104). SAB 104 requires that four basic

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

criteria must be met prior to revenue recognition: (i) persuasive evidence of an arrangement exists, (ii) delivery has occurred or the services have been rendered, (iii) the fee is fixed and determinable and (iv) collection of the resulting receivable is reasonably assured. Certain of the Company's arrangements contain multiple elements, consisting of the various services the Company offers. These arrangements are accounted for in accordance with Emerging Issues Task Force (EITF) Issue No. 00-21, Revenue Arrangements with Multiple Deliverables. The Company has determined that there is not objective and reliable evidence of fair value for any of its services and, therefore, accounts for all elements in multiple elements arrangements as a single unit of accounting and recognizes the total value of the arrangement on a straight-line basis over the longest contract term

The Company generates revenues by providing access to the Company's online database or delivering information obtained from the database, usually in the form of periodic reports. Revenues are typically recognized on a straight-line basis over the period in which access to data or reports are provided, which generally ranges from three to 24 months.

Revenues are also generated through survey services under contracts ranging in term from two months to one year. Survey services consist of survey and questionnaire design with subsequent data collection, analysis and reporting. Revenues are recognized on a straight-line basis over the estimated data collection period once the survey or questionnaire has been delivered. Any change in the estimated data collection period results in an adjustment to revenues recognized in future periods.

Generally, contracts are non-cancelable. A limited number of customers, however, have the right to cancel their contracts by providing a written notice of cancellation. In the event that a customer cancels its contract, the customer is not entitled to a refund for prior services, and will be charged for costs incurred plus services performed up to the cancellation date.

Costs of Revenues

Cost of revenues consists primarily of expenses related to the operating network infrastructure and the recruitment, maintenance and support of consumer panels. Expenses associated with these areas include the salaries, stock-based compensation and related expenses of network operations, survey operations, custom analytics and technical support departments, and are expensed as they are incurred. Cost of revenues also includes data collection costs for the products and operational costs associated with the Company's data centers, including depreciation expense associated with computer equipment.

Selling and Marketing

Selling and marketing expenses consist primarily of salaries, stock-based compensation, benefits, commissions and bonuses paid to the direct sales force and industry analysts, as well as costs related to online and offline advertising, product management, seminars, promotional materials, public relations, other sales and marketing programs, and allocated overhead, including rent and depreciation. All selling and marketing costs are expensed as they are incurred.

Research and Development

Research and development expenses include new product development costs, consisting primarily of compensation, stock-based compensation and related costs for personnel associated with research and development activities, and allocated overhead, including rent and depreciation.

General and Administrative

General and administrative expenses consist primarily of salaries, stock-based compensation and related expenses for executive management, finance, accounting, human capital, legal, information technology and

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

other administrative functions, as well as professional fees, overhead, including allocated rent and depreciation and expenses incurred for other general corporate purposes.

Concentration of Credit Risk

Financial instruments that potentially subject the Company to concentrations of credit risk consist primarily of cash equivalents, short term investments and accounts receivable. Cash equivalents are held at financial institutions, which are regarded as highly creditworthy. Short term investments consist of high-grade auction rate securities which the Company has the option to re-invest in every 30 days. With respect to accounts receivable, credit risk is mitigated by the Company's ongoing credit evaluation of its customers' financial condition.

For the years ended December 31, 2004, 2005 and 2006, one customer accounted for 5%, 14% and 12%, respectively, of total revenues. No customer accounted for more than 10% of accounts receivable as of December 31, 2005 and 2006.

Advertising Costs

All advertising costs are expensed as incurred. Advertising expense, which is included in sales and marketing expense, totaled \$84,000, \$58,000 and \$210,000 for the years ended December 31, 2004, 2005 and 2006, respectively.

Stock-Based Compensation

In December 2004, the FASB issued SFAS No. 123(R), Share-Based Payment (SFAS 123R), which requires companies to expense the estimated fair value of employee stock options and similar awards. This statement is a revision to SFAS No. 123, Accounting for Stock-Based Compensation (SFAS 123), supersedes Accounting Principles Board Opinion No. 25 (APB 25), Accounting for Stock Issued to Employees, and amends SFAS No. 95, Statement of Cash Flows.

Prior to January 1, 2006, the Company accounted for its stock-based compensation plans under the recognition and measurement provisions of APB 25, and related interpretations, as permitted by SFAS 123. Effective January 1, 2006, the Company adopted SFAS 123R, including the fair value recognition provisions, using the prospective method. Under SFAS 123R, a non-public company that previously used the minimum value method for pro forma disclosure purposes is required to adopt the standard using the prospective method. Under the prospective method, all awards granted, modified or settled after the date of adoption are accounted for using the measurement, recognition and attribution provisions of SFAS 123R. As a result, stock-based awards granted prior to the date of adoption of SFAS 123R will continue to be accounted for under APB 25 with no recognition of stock-based compensation in future periods, unless such awards are modified or settled. Subsequent to the adoption of SFAS 123R, the Company estimates the value of stock-based awards on the date of grant using the Black-Scholes option-pricing model. For stock-based awards subject to graded vesting, the Company has utilized the straight-line ratable method for allocating compensation cost by period. For the year ended December 31, 2006, the Company recorded stock-based compensation expense of \$198,000 in accordance with SFAS 123R.

Cumulative Effect of Change in Accounting Principle

Effective July 1, 2005, the Company adopted the provisions of FASB Staff Position No. 150-5, Issuer's Accounting under Statement No. 150 for Freestanding Warrants and Other Similar Instruments on Shares that are Redeemable (FSP 150-5), an interpretation of SFAS No. 150, Accounting for Certain Financial Instruments with Characteristics of Both Liabilities and Equity (SFAS 150). Pursuant to FSP 150-5, freestanding warrants for shares that are either puttable or warrants for shares that are redeemable are

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

classified as liabilities on the consolidated balance sheet at fair value

Upon adoption of FSP 150-5, the Company reclassified the carrying value of its warrants to purchase shares of its redeemable convertible preferred stock from mezzanine equity to a liability and recorded a cumulative effect charge of approximately \$440,000 for the change in accounting principle to record the warrants at fair value on July 1, 2005. The Company recorded additional charges of approximately \$14,000 to reflect the increase in fair value between July 1, 2005 and December 31, 2005. In the year ended December 31, 2006, the Company recorded approximately \$224,000 of charges to reflect the increase in fair value between January 1, 2006 and December 31, 2006. The Company will continue to adjust the liabilities for changes in fair value until the earlier of the exercise of the warrants to purchase shares of its redeemable convertible preferred stock or the completion of a liquidation event, including the completion of an initial public offering, at which time the liabilities will be reclassified to stockholders' equity (deficit).

The pro forma effect of the adoption of FSP 150-5 on the results of operations for fiscal years 2004 and 2005 if applied retroactively, assuming FSP 150-5 had been adopted in these years, has not been disclosed as these amounts would not be materially different from the reported amounts.

Comprehensive (Loss) Income

Comprehensive (loss) income includes net (loss) income as well as the effects of foreign currency translation loss adjustments reflected in the table below:

rears Ended December 51,					
2004		2005 (In thousands)			2006
\$	(3,226)	\$	(4,422)	\$	5,669
	(19)		(35)		(51)
\$	(3,245)	\$	(4,457)	\$	5,618
	\$	\$ (3,226) (19)	\$ (3,226) \$ (19)	\$ (3,226) \$ (4,422) (19)(35)	2004 2005 (In thousands) (3,226) \$ (4,422) \$ (19) (35)

Vears Ended December 31

Income Taxes

Income taxes are accounted for using the liability method in accordance with SFAS No. 109, Accounting for Income Taxes. Deferred income taxes are provided for temporary differences in recognizing certain income, expense and credit items for financial reporting purposes and tax reporting purposes. Such deferred income taxes primarily relate to the difference between the tax bases of assets and liabilities and their financial reporting amounts. Deferred tax assets and liabilities are expected to be settled or realized

Earnings Per Share

The Company computes earnings per share in accordance with the provisions of FASB No. 128, Earnings Per Share (SFAS 128). The Company has issued shares of common stock in connection with business acquisitions (see Note 3) that give the holders the right to require the Company to repurchase the shares at a fixed price at a specified future date ("Common Stock Subject to Put"). The difference between the fair value

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

of the shares of Common Stock Subject to Put on the issuance date and the price at which the Company may be required to repurchase those shares is being accreted over the period from issuance to the first date at which the Company could be required to repurchase the shares as a dividend to the holders. EITF Topic D-98, Classification and Measurement of Redeemable Securities (EITF D-98) states that when a common shareholder has a contractual right to receive, at share redemption, an amount that is other than fair value, such shareholder has received, in substance, a preferential distribution. Under SFAS 128, entities with capital structures that include classes of common stock with different dividend rates are required to apply the two-class method of calculating earnings per share. Accordingly, the Company calculates earnings per share for its common stock and its Common Stock Subject to Put using a method akin to the two-class method under SFAS 128.

In addition, the Company's series of convertible redeemable preferred stock are considered participating securities as they are entitled to an 8% noncumulative preferential dividend before any dividends can be paid to common stockholders. The Company includes its participating preferred stock in the computation of earnings per share using the two-class method in accordance with EITF 03-06, *Participating Securities and the Two-Class Method under FASB Statement No. 128* (EITF 03-06).

The two-class computation method for each period allocates the undistributed earnings or losses to each participating security based on their respective rights to receive dividends. In addition to undistributed earnings or losses, the accretion to their redemption or put prices is also allocated to the Common Stock Subject to Put and the convertible redeemable preferred stock. In periods of undistributed losses, all losses are allocated to common stock in accordance with EITF 03-06 as the holders of Common Stock Subject to Put and participating preferred stock are not required to fund losses nor are their redemption or put prices reduced as a result of losses incurred. In periods of undistributed income, income is first allocated to the participating preferred stock for their preferential dividend, currently \$7.1 million per annum. Any undistributed earnings remaining are then allocated to holders of common stock. Common Stock Subject to Put and preferred stock (assuming conversion) on a pro rata basis. The total earnings or losses allocated to each class of common stock are then divided by the weighted-average number of shares outstanding for each class of common stock to determine basic earnings per share. EITF 03-06 does not require the presentation of basic and diluted earnings per share for securities other than common stock; therefore, earnings per share is only computed for the Company's common stock.

Diluted earnings per share for common stock reflects the potential dilution that could result if securities or other contracts to issue common stock were exercised or converted into common stock. Diluted earnings per share assumes the exercise of stock options and warrants using the treasury stock method and the conversion of the Company's convertible preferred stock using the if-converted method. No potentially dilutive securities are convertible or exercisable into shares of Common Stock Subject to Put.

For all periods presented, all potentially dilutive securities have been excluded from earnings per share calculations as their effect would have been anti-dilutive. The following is a summary of common stock equivalents for the securities outstanding during the respective periods that have been excluded from the earnings per share calculations:

		Year Ended December 31,	
	2004	2005	2006
Stock options	8,950,177	14,104,727	13,750,111
Convertible preferred stock warrants	565,643	565,643	565,643
Common stock warrants	1,948,660	1,994,800	576,786
Convertible preferred stock	86,286,744	86,286,744	86,286,744

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

The following table sets forth the computation of basic and diluted EPS:

	 2004	Year Er	nded December 31, 2005		2006
	 (In thousands, except share and per share data)				2000
Calculation of basic and diluted net income per share — two class method:					
Net (loss) income	\$ (3,226)	\$	(4,422)	\$	5,669
Accretion of redeemable preferred stock	(2,141)		(2,638)		(3,179)
Accretion of common stock subject to put	(32)		(133)		(138)
Undistributed (loss) earnings	(5,399)		(7,193)		2,352
Allocation of undistributed (loss) earnings:					
Common stock					
Earnings before cumulative effect of change in accounting principle	(5,399)		(6,753)		_
Cumulative effect of change in accounting principle	_		(440)		_
Common stock subject to put	_		_		_
Preferred stock	_		_		2,352
Total allocated (loss) earnings	\$ (5,399)	\$	(7,193)	\$	2,352
Net (loss) income attributable to common stockholders per common share:					
Basic and diluted	\$ (0.38)	\$	(0.46)	\$	0.00
Cumulative effect of change in accounting principle	\$ 0.00	\$	(0.03)	\$	0.00
Weighted average shares outstanding — common stock basic and diluted	14,358,561		15,650,969		19,236,064
Net (loss) income attributable to common stockholders per common share subject to put:					
Basic and diluted	\$ 0.07	\$	0.08	\$	0.08
Weighted average shares outstanding — common stock subject to put basic and diluted	457,596		1,738,172		1,738,172

Fair Value of Financial Instruments

SFAS No. 107, Disclosure about Fair Value of Financial Instruments, defines the fair value of financial instruments as the amount at which the instrument could be exchanged in a current transaction between willing parties. Cash equivalents, short-term investments, accounts receivable, accounts payable, accrued expenses and capital lease obligations reported in the consolidated balance sheets equal or approximate their respective fair values. The fair value of the Company's preferred stock warrants liabilities, convertible preferred stock and common stock subject to put is not practicable to determine, as no quoted market price exists for these instruments. The convertible preferred stock will be converted into common stock of the Company upon consummation of a qualified initial public offering.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Recent Pronouncements

In June 2006, the FASB issued FASB Interpretation No. 48 (FIN 48), Accounting for Uncertainty in Income Taxes, an interpretation of SFAS No. 109. This interpretation clarifies the accounting for income taxes by prescribing that a company should use a more-likely-than-not recognition threshold based on the technical merits of the tax position taken. Tax provisions that meet the more-likely-than-not recognition threshold should be measured as the largest amount of tax benefits, determined on a cumulative probability basis, which is more likely than not to be realized upon ultimate settlement in the financial statements. FIN 48 also provides guidance on derecognition, classification, interest and penalties, accounting for interim periods, disclosure and transition, and explicitly excludes income taxes from the scope of SFAS No. 5, Accounting for Contingencies. FIN 48 is effective for fiscal years beginning after December 15, 2006. The Company is currently assessing the effect of FIN 48 on its consolidated financial statements.

In September 2006, the FASB issued SFAS No. 157, Fair Value Measurements. The purpose of this statement is to define fair value, establish a framework for measuring fair value and enhance disclosures about fair value measurements. The measurement and disclosure requirements are effective for the Company as of January 1, 2008 and are applied prospectively. The Company is currently evaluating the potential impact of adopting this new guidance on its results of operations and financial position.

3. Acquisitions

Q2 Brand Intelligence, Inc.

On July 28, 2004, the Company acquired the outstanding stock of Denaro and Associates, Inc, otherwise known as Q2 Brand Intelligence, Inc. (Q2), to improve the Company's ability to provide customers more robust custom research integrated with its underlying digital marketing intelligence platform. The total cost of the acquisition was \$3,336,000, which included cash of \$873,000, the issuance of 1,060,000 shares of restricted common stock valued at \$2,412,000 and related costs incurred in the amount of \$51,000. The former sole shareholder of Q2 is entitled to receive up to an additional \$600,000 in cash based on the entity's achievement of certain performance criteria. No amounts were earned as of December 31, 2004. In 2005 and 2006, the performance criteria were met and the Company paid \$300,000 each year which was recorded as additional goodwill.

The Company accounted for the acquisition as a purchase in accordance with SFAS No. 141, *Business Combinations* (SFAS 141). Accordingly, the results of operations of Q2 have been included in the accompanying consolidated financial statements since the purchase date. In accordance with SFAS 141, the purchase price was allocated to the assets and liabilities of Q2 based on their estimated fair values.

The following table summarizes the estimated fair values of the tangible assets acquired and liabilities assumed at the date of acquisition:

	(In	thousands)
Accounts receivable	\$	917
Prepaids and other		24
Property and equipment		60
Total assets acquired		1,001
Accounts payable and accrued expenses		511
Deferred revenues		58
Net tangible assets acquired	\$	432

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

The common stock issued to the former sole shareholder of Q2 is subject to a restricted stock agreement that includes a put right at a price of \$2.50 per share to be effective for a ninety-day period beginning on the third anniversary of the closing date. The Company has valued the common stock subject to put at fair value on the date of issuance. The carrying value of the common stock subject to the put right is being accreted to the put obligation over the three year term using the effective interest rate method. For the years ended December 31, 2004, 2005 and 2006, the Company accreted a total of \$32,000, \$78,000 and \$80,000, respectively.

The non-tangible portion of the purchase price, including the payment of the contingent purchase consideration, was allocated as follows:

	<u> </u>	(In thousands)
Trademarks and brands	\$	338
Non-compete agreements		112
Customer relationships		1,249
Goodwill		1,364
Acquired methodology		451

Acquired trademarks and brand names were initially determined to have an indefinite life and, therefore, were not amortized. In July 2005, the Company determined that the trademarks and brand names would be phased out over the next six months so that the services could be branded under the Company's name. At the time of the decision, there were no indicators of impairment. Accordingly, the asset was amortized on a straight-line basis over its remaining six month useful life. The change in the estimated useful life resulted in additional amortization expense of \$290,000 for the year ended December 31, 2005. Acquired methodology and customer relationships are being amortized on a straight-line basis over four years.

SurveySite, Inc.

On January 4, 2005, the Company acquired the assets and assumed certain liabilities of SurveySite Inc., or SurveySite. Through this acquisition, the Company acquired proprietary data-collection technology and increased customer penetration and revenues in the survey business. The total cost of the acquisition was \$3.6 million, which included cash of \$1.7 million, the payment of additional purchase consideration of \$132,000, the issuance of 678,172 shares of restricted common stock valued at \$1.6 million and related costs incurred and adjustments in the amount of \$111,000.

The Company accounted for the acquisition as a purchase in accordance with SFAS 141. Accordingly, the results of operations of SurveySite have been included in the accompanying consolidated financial statements since the purchase date. In accordance with SFAS 141, the purchase price was allocated to the assets and liabilities of SurveySite based on their estimated fair values. Based on this analysis, the fair value of the identifiable tangible and intangible assets exceeded the cost of the acquired business by approximately \$790,000. Therefore, in accordance with SFAS 141, the Company reduced, on a pro rata basis, the value attributed to certain assets acquired.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

The following table summarizes the estimated fair values of the tangible assets acquired and liabilities assumed at the date of acquisition:

	 (In thousands)
Cash	\$ 715
Accounts receivable	606
Prepaid expense and other current assets	90
Property and equipment	 283
Total assets acquired	1,694
Accounts payable and accrued expenses	245
Deferred revenues	480
Deferred tax liability	 356
Net tangible assets acquired	\$ 613

The former shareholders of SurveySite are entitled to receive \$132,000 based on the entity's achievement of certain performance criteria. The performance criteria was achieved as of December 31, 2005 and the performance criteria was also expected to be achieved in 2006, therefore, the total contingent purchase consideration was paid in January 2006 and is included in the purchase price. The common stock issued is subject to a restricted stock agreement that includes a put right at a price of \$2.67 per share to be effective for a ninety-day period beginning on the third anniversary of the closing date. The Company has valued the common stock subject to put at fair value on the date of issuance. The carrying value of the common stock subject to the put right is being accreted to the put obligation over the three year term using the effective interest rate method. For the year ended December 31, 2005 and 2006, the Company accreted a total of \$55,000 and \$58,000, respectively.

The non-tangible portion of the purchase price, including the payment of the contingent purchase consideration, was allocated as follows:

	<u> </u>	(In thousands)
Trademarks	\$	323
Non-compete agreements		213
Customer relationships		2,228
Acquired methodologies/technology		237

Acquired methodology and customer relationships are being amortized on a straight-line basis over six months to three years. The trademarks and non-compete agreements are being amortized on a straight-line basis over two and three years, respectively.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

4. Property and Equipment

Property and equipment, including equipment under capital lease obligations, consists of the following:

	December 31			
	2005			2006
		(In thou	sands)	
Computer equipment	\$	15,165	\$	14,855
Computer software		3,220		2,816
Office equipment and furniture		1,178		1,159
Leasehold improvements		832		1,079
		20,395		19,909
Less: accumulated depreciation and amortization		(15,915)		(12,929)
	\$	4,480	\$	6,980
	\$	4,480	\$	6,980

Property and equipment financed through capital lease obligations, consisting of computer equipment, totaled \$4.5 million and \$4.6 million at December 31, 2005 and 2006, respectively. At December 31, 2005 and 2006, accumulated depreciation related to property and equipment financed through capital leases totaled \$2.2 million and \$1.1 million, respectively. During the year ended December 31, 2006, \$3.2 million of fully depreciated assets were written off. In addition, \$2.6 million of assets financed through capital leases terminated and were subsequently returned and written off.

For the years ended December 31, 2004, 2005 and 2006, total depreciation expense was \$2.4 million, \$2.7 million and \$2.9 million, respectively.

5. Goodwill and Intangible Assets

	December 31		
	 2005	_	2006
	(In th	ousands)	
Goodwill	\$ 1,064	\$	1,364
Intangible assets consist of the following:			
Trademarks and brands	\$ 662	\$	662
Non-compete agreements	326		326
Customer relationships	3,467		3,467
Acquired methodologies/technology	688		688
Total intangible assets	5,143		5,143
Accumulated amortization	 (2,788)		(4,160)
Intangible assets, net	\$ 2,355	\$	983

Amortization expense related to intangible assets was approximately \$356,000, \$2.4 million and \$1.4 million for the years ended December 31, 2004, 2005 and 2006, respectively.

Future expected amortization of intangible assets as of December 31, 2006, is as follows:

	(In thousands)
2007	\$967
2008	16

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

The weighted average amortization period by major asset class as of December 31, 2006, is as follows:

	(In years)
Trademarks and brands	1.7
Non-compete agreements	3.4
Customer relationships	2.7
Acquired methodologies/technology	2.0

6. Accrued Expenses

Accrued expenses consist of the following:

		Г	ecember 31	,
	_	2005		2006
		(I	n thousands)
Accrued payroll and related	\$	2,428	3 \$	3,118
Other	_	1,757	7	2,902
	\$	4,185	\$	6,020

7. Commitments and Contingencies

Leases

In December 2006, the Company entered into an equipment lease agreement with Banc of America Leasing & Capital, LLC to finance the purchase of new hardware and other computer equipment as the Company continues to expand its technology infrastructure in support of its business growth. This agreement includes a \$5.0 million line of credit available through December 31, 2007; its initial utilization of this credit facility was to establish an equipment lease for approximately \$2.9 million bearing interest at a rate of 7.75% per annum. The base term for this lease is three years and includes a nominal charge in the event of prepayment. Assets acquired under the equipment leases secure the obligations.

In addition to equipment financed through capital leases, the Company is obligated under various noncancelable operating leases for office facilities and equipment. These leases generally provide for renewal options and escalation increases. Future minimum payments under noncancelable lease agreements with initial terms of one year or more as of December 31, 2006 are as follows:

	C	apital Leases (In th	iousands)	Operating Leases
2007	\$	1,986	\$	2,009
2008		1,418		1,383
2009		1,014		680
2010		_		377
2011		_		383
Thereafter		_		226
Total minimum lease payments		4,418	\$	5,058
Less amount representing interest		(431)		
Present value of net minimum lease payments		3,987		
Less current portion		(1,726)		
Capital lease obligations, long-term	\$	2,261		

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Total rent expense was \$1.9 million, \$2.5 million and \$2.6 million for the years ended December 31, 2004, 2005 and 2006, respectively.

The Company is required to maintain a letter of credit in the amount of approximately \$256,000 as additional security deposit pertaining to an operating lease.

In June 2003, the Company modified its lease for its corporate headquarters resulting in (i) a reduction in the space rented, (ii) the lease termination date being revised from January 2011 to June 2008, and (iii) a reduction in the monthly lease rate. In connection with the modification, the Company relinquished its security deposit on the original lease and made certain cash payments which totaled \$2.0 million. The Company has treated the modification payments, net of a deferred rent liability of approximately \$300,000 associated with the vacated space, as prepaid rent and is recognizing the amount over the remaining lease term. The prepaid lease balance at December 31, 2005 and 2006 was approximately \$665,000 and \$386,000, respectively. The short-term portion is included in Prepaid Expenses and Other Current Assets and the long-term portion is included in Other Non-Current Assets in the Consolidated Balance

Contingencies

The Company has no asserted claims, but is from time to time exposed to unasserted potential claims encountered in the normal course of business. Although the outcome of any legal proceedings cannot be predicted with certainty, management believes that the final resolution of these matters will not materially affect the Company's financial position or results of operations.

8. Income Taxes

Income tax expense (benefit) is comprised of the following:

		Year Ended December 31,	
	2004	2005 (In thousands)	2006
Current:		(III tilotistilitis)	
	_		
Federal	\$ —	\$ —	\$ 147
State	_	_	_
Foreign			
Total	-	_	147
Deferred:			
Federal	_	_	_
State	_	_	_
Foreign	_	(182)	(97)
Total		(182)	(97)
Income tax expense (benefit)	-	(182)	50

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

A reconciliation of the statutory United States income tax rate to the effective income tax rate follows:

	Year Ended December 31,			
	2004	2005	2006	
Statutory federal tax rate	34.0%	34.0%	34.0%	
Nondeductible items	(0.9)	(1.2)	3.4	
State tax rate, net of federal benefit	4.5	2.6	5.6	
Foreign	_	0.4	(0.2)	
Change in valuation allowance	(37.6)	(31.2)	(41.9)	
Effective tax rate	0.0%	4.6%	0.9%	

Deferred income taxes reflect the net tax effects of temporary differences between the carrying amount of assets and liabilities for financial reporting purposes and the amounts used for income tax purposes. Significant components of the Company's net deferred income taxes are as follows:

	December 31		aber 31	
	_	2005		2006
		(In tho	usands)	
Deferred tax asset:				
Net operating loss	\$	34,498	\$	31,580
Tax credits		_		147
Accrued vacation and bonus		96		197
Deferred revenues		708		438
Acquired intangibles		287		673
Depreciation		345		525
Deferred rent		103		96
Other		102		90
Total deferred tax assets		36,139		33,746
Deferred tax liabilities:				
Intangibles		(174)		(77)
Less valuation allowance		(36,139)		(33,746)
Net deferred tax liability	\$	(174)	\$	(77)

As of December 31, 2005 and 2006, the Company had both federal and state net operating loss carryforwards for tax purposes of approximately \$88.5 million and \$81.2 million, respectively, which begin to expire in 2020 for federal and began to expire in 2006 for state income tax reporting purposes. In addition, at December 31, 2005 and 2006, the Company had net operating loss carryforwards for tax purposes related to our foreign subsidiaries of \$966,000 and \$703,000, respectively, which begin to expire in 2010.

Under the provisions of the Internal Revenue Code Section 382, certain substantial changes in the Company's ownership may result in a limitation on the amount of U.S. net operating loss carryforwards which could be utilized annually to offset future taxable income and taxes payable. Additionally, despite the net operating loss carryforward, the Company may have a future tax liability due to alternative minimum tax, foreign tax or state tax requirements.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Management believes that, based on a number of factors, the available objective evidence creates sufficient uncertainty regarding the realizability of the deferred tax assets such that a full valuation allowance is required. Such factors include the lack of a significant history of profits, recent increases in expense levels to support the Company's growth, the fact that the market in which the Company competes is intensely competitive and characterized by rapidly changing technology, and the lack of carryback capacity to realize deferred tax assets.

9. Convertible Preferred Stock

The Company's certificate of incorporation provides for the issuance of 9,187,500 shares of Series A Preferred Stock (Series A), 3,535,486 shares of Series B Preferred Stock (Series B), 13,355,052 shares of Series C Preferred Stock (Series C), 357,144 shares of Series C-1 Preferred Stock (Series C-1), 22,238,042 shares of Series D Preferred Stock (Series D) and 25,000,000 shares of Series E Preferred Stock (Series E).

The Series E ranks senior to all other classes of capital stock, with the exception of the Incentive Plan (see Note 11), on a distribution of assets upon liquidation, dissolution, or winding up of the Company. Upon such event, each share of Series E is entitled to a liquidation preference equal to 1.63 times the original purchase price of \$0.50 per share. In addition, each share of Series E is entitled to participate in any distribution pari passu with all classes of stock after \$88,392,465 (the Cap Amount) has been distributed to the holders of Series A through Series D preferred stock. The assets distributed to each share of Series E upon liquidation, dissolution or winding up of the Company shall not exceed five times the original purchase price of \$0.50 per share. Series E is convertible into common stock at a conversion price equal to the original issuance price, subject to adjustment.

The holders of Series E are entitled to dividends in preference to any class of capital stock of the Company at an annual rate of 8.0%. Following payment of any dividends to holders of Series E, holders of Series D are entitled to dividends in preference to any class of stock other than Series E at an annual rate of 8%. Following the payment of any dividends to the holders of Series D, holders of Series B, Series C and Series C-1 are entitled to dividends in preference to common stockholders at an annual rate of 8%. All dividends are noncumulative and are paid only when, if, and as declared by the Board of Directors. No dividend shall be paid on shares of common stock in any fiscal year unless (i) the noncumulative preferential dividends of the preferred stock have been paid in full and (ii) the holders of preferred stock participate in any such dividend on common stock on a pro rata basis assuming conversion of all preferred stock into common stock.

The Series A, B, C, C-1 and D (Series A-D) each has a liquidation preference senior to the common stock. In the event of any liquidation, dissolution, or winding up of the Company, each Series A-D share is entitled to a liquidation preference equal to a portion of the Cap Amount. The portion of the Cap Amount to which each share of Series A, B, C and C-1 is entitled is equal to the original purchase price for such share (plus all declared and unpaid dividends) multiplied by an adjustment factor set forth in the certificate of incorporation. The portion of the Cap Amount to which each share of Series D is entitled is equal to the original issue price (plus all declared and unpaid dividends) plus a 25% premium, compounded annually (but such total not to exceed 250% of the original issue price) multiplied by an adjustment factor set forth in the certificate of incorporation. The original purchase price per share for Series A, Series B, Series C, Series C-1 and Series D was \$1.00, \$4.90, \$2.27, \$1.40 and \$0.90, respectively. After the payment of the liquidation preference to the Series A-D, each share of Series A-D is entitled to participate in any distribution pari passu with all classes of stock. The assets distributed to each share of Series A-D upon liquidation, dissolution, or winding up of the Company shall not exceed 2.5 times the original purchase price of such shares.

Upon the occurrence of a Liquidation Event, defined as a consolidation, merger, or sale of the Company, Management shall be entitled to receive the first 10% of any liquidation proceeds pursuant to an Incentive Plan (see Note 11). The distribution of such proceeds shall be to the Incentive Plan participants (senior

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

management and Company's founders) based on both their respective equity ownership in the Company and a variable percentage which is subject to Board approval.

As a result of the issuance of Series E, the conversion prices of the Series A, Series B, Series C, Series C-1 and Series D were adjusted to the following rates: Series A \$0.86 per share, Series B \$2.47 per share, Series C \$1.50 per share, Series C-1 \$1.18 per share and Series D of \$0.80 per share.

Each share of preferred stock is convertible at any time into shares of common stock based on the conversion price then in effect. Conversion is automatic in the event of a public offering of common stock at a price of at least \$2.50 per share with gross proceeds of at least \$25 million. Each holder of preferred stock is entitled to the number of votes equal to the number of whole shares of common stock into which the shares held by the holder are then convertible at each meeting of the stockholders of the Company. All series of preferred stock have anti-dilution protection in the event the Company issues shares at a purchase price less than \$0.50.

All classes of preferred stock are redeemable by the holder on or after August 1, 2008. Series E ranks senior to all other classes of stock and may be redeemed at 1.63 times its original purchase price plus all declared but unpaid dividends. The aggregate redemption value for the Series A-D shares is equal to the Cap Amount. In the event that any series of preferred stock is converted into common stock prior to redemption, the aggregate redemption value of the remaining series of preferred stock remains equal to the Cap Amount. The redemption value for the Series A-D shares is equal to the liquidation preference in effect on the redemption date for each series of preferred stock as adjusted by a formula set forth in the certificate of incorporation. Upon the initiation of the Cap Amount, the carrying values of Series B, Series C and Series C-1 were in excess of their individual redemption value. The differences between the carrying value of each series of preferred stock and its respective redemption value (as adjusted for the Cap Amount for Series A-D) is being accreted as preferred stock dividends using the interest method over the period to the redemption date. Such accretion amounted to \$2.1 million, \$2.6 million and \$3.2 million for the years ended December 31, 2004, 2005 and 2006, respectively.

10. Convertible Preferred Stock Warrants

In prior years, the Company issued fully vested warrants to purchase 486,608 shares of preferred stock in connection with a master lease and various equipment lease agreements. The exercise prices of the warrants range from \$0.50 to \$4.90 per share and the warrants expire 10 years from the date of issue. The Company recorded the fair value of the warrants totaling \$383,000 as deferred financing costs with an offset to warrants to purchase redeemable preferred stock. The fair value of the warrants was estimated using the Black-Scholes option pricing model. The deferred financing costs are being amortized to interest expense over the respective agreement on a straight line basis. For each of the years ended December 31, 2004, 2005 and 2006, the Company recorded \$33,000 in interest expense.

Upon adoption of FSP 150-5 (July 1, 2005), the Company reclassified the carrying value of its warrants to purchase shares of its convertible preferred stock from mezzanine equity to a liability and adjusted the warrants to fair value. The fair value of the convertible preferred stock warrants at December 31, 2005 and 2006 was approximately \$781,000 and \$1.0 million, respectively. The fair value of warrants was estimated using the Black-Scholes option pricing model.

11. Stockholders' Deficit

1999 Stock Option Plan

In September 1999, the Company established the 1999 Stock Option Plan (the Plan) under which eligible employees and nonemployees may be granted options to purchase shares of the Company's common stock.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

The Plan provides for the issuance of a maximum of 26.8 million shares of common stock. The exercise price is determined by the Board of Directors, which is generally equal to fair value for incentive stock options and is determined on a per-grant basis for nonqualified options. The vesting period of options granted under the Plan is determined by the Board of Directors, generally ratably over a four-year period. The options expire 10 years from the date of the grant. As of December 31, 2006, 5,316,147 shares were available for grant under the plan.

Effective January 1, 2006, the Company adopted the fair value recognition provisions of SFAS 123R using the prospective transition method, which requires the Company to apply its provisions only to awards granted, modified, repurchased or cancelled after the effective date. Under this transition method, stock-based compensation expense recognized beginning January 1, 2006 is based on the following: (1) the grant-date fair value of stock option awards granted or modified beginning January 1, 2006; and (2) the balance of deferred stock-based compensation related to stock option awards granted prior to January 1, 2006, which was calculated using the intrinsic-value method as previously permitted under APB 25. Results for prior periods have not been restated.

In connection with the adoption of SFAS 123R, the Company estimates the fair value of stock option awards granted beginning January 1, 2006 using the Black-Scholes option-pricing formula and a single option award approach. The Company then amortizes the fair value of awards expected to vest on a straight-line basis over the requisite service periods of the awards, which is generally the period from the grant date to the end of the vesting period. The weighted-average expected option term for options granted during the year ended December 31, 2006 was calculated using the simplified method described in SAB No. 107, Share-Based Payment. The simplified method defines the expected term as the average of the contractual term and the vesting period. Estimated volatility for the year ended December 31, 2006 also reflected the application of SAB No. 107 interpretive guidance and, accordingly, incorporates historical volatility of similar entities whose share prices are publicly available. The risk-free interest rate is based on the yield curve of a zero-coupon U.S. Treasury bond on the date the stock option award is granted with a maturity equal to the expected term of the stock option award. The Company used historical data to estimate the number of future stock option for feitures.

As a result of adopting SFAS 123R on January 1, 2006, the Company's income before income taxes and net income for the year ended December 31, 2006 was \$198,000 less than if the Company had continued to account for stock-based compensation under APB No. 25. Basic and diluted net income per common share for the year ended December 31, 2006 would have been unaffected if the Company had not adopted SFAS 123R. As of December 31, 2006, total unrecognized compensation expense related to non-vested stock options, granted prior to that date is estimated at \$1.3 million, which the Company expects to recognize over a weighted average period of approximately 1.86 years. Total unrecognized compensation expense as of December 31, 2006 is estimated based on outstanding non-vested stock options and may be increased or decreased in future periods for subsequent grants or forfeitures. The following are the weighted-average assumptions used in valuing the stock options granted during the year ended December 31, 2006, and a discussion of the Company's assumptions.

Dividend yield	0.00%
Expected volatility	63.37%
Risk-free interest rate	4.76%
Expected life of options (in years)	6.02

Dividend yield — The Company has never declared or paid dividends on its common stock and does not anticipate paying dividends in the foreseeable future.

Expected volatility — Volatility is a measure of the amount by which a financial variable such as a share price has fluctuated (historical volatility) or is expected to fluctuate (expected volatility) during a period. The

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Company has used the historical volatility of its peer group to estimate expected volatility. The peer group includes companies that are similar in revenue size, in the same industry or are competitors.

Risk-free interest rate — This is the average U.S. Treasury rate (with a term that most closely resembles the expected life of the option) for the quarter in which the option was granted.

Expected life of the options — This is the period of time that the options granted are expected to remain outstanding. This estimate is derived from the average midpoint between the weighted average vesting period and the contractual term as described in the SAB No. 107.

The weighted average grant date fair value of options granted during the year ended December 31, 2006 was \$0.86. Options granted in the years ended December 31, 2004 and 2005 were issued prior to the adoption of SFAS 123R. The total fair value of shares vested during the year ended December 31, 2006 was \$178,000.

A summary of the Plan is presented below:

	Number of Shares	Weigh Exe	ited-Average rcise Price
Options outstanding at December 31, 2003	8,909,016	\$	0.12
Options granted	9,281,457		0.07
Options exercised	2,403,710		0.05
Options forfeited	481,733		0.15
Options expired	164,630		0.97
Options outstanding at December 31, 2004	15,140,400		0.09
Options granted	4,194,511		0.70
Options exercised	1,531,888		0.09
Options forfeited	878,210		0.22
Options expired	59,999		0.33
Options outstanding at December 31, 2005	16,864,814		0.23
Options granted	1,713,550		1.45
Options exercised	3,263,373		0.07
Options forfeited	1,509,284		0.45
Options expired	186,007		0.56
Options outstanding at December 31, 2006	13,619,700		0.40
Options exercisable at December 31, 2006	7.050.519		0.24

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

The following table summarizes information about options outstanding at December 31, 2006:

		Options O	utstanding			Options	Exercisable																									
Exercise Price	Options Outstanding	Av Ex	ghted- erage ercise rice	Weighted- Average Remaining Contractual Life	Options Exercisable	Av Ex	ighted- verage vercise Price	Weighted- Average Remaining Contractual Life																								
\$0.01 - \$0.50	9,791,048	\$	0.11	6.4	6,079,905	\$	0.11	5.9																								
0.51 - 1.00	2,414,903		0.87	8.5	741,666		0.85	8.1																								
1.01 - 1.50	896,639		1.50	8.9	171,228		1.50	7.4																								
1.51 - 2.00	517,110	1.75		1.75		1.75		1.75		1.75		1.7		1.75		1.75		1.75		1.75		1.75		1.75		1.75		9.3	57,720	1.83		6.8
	13,619,700		0.40	7.0	7,050,519		0.24	6.2																								

The aggregate intrinsic value of options exercised for the years ended December 31, 2004, 2005 and 2006 was \$1,747, \$1,072,511 and \$3,699,292, respectively. The aggregate intrinsic value for all options outstanding under the Company's stock plans as of December 31, 2006 was \$18,454,548. The aggregate intrinsic value for options exercisable under the Company's stock plans as of December 31, 2006 was \$10,665,346.

During 2003, the Company initiated an offer to exchange certain outstanding incentive stock options. Employees had the option to exchange all outstanding incentive stock options to purchase shares of the Company's common stock that had an exercise price equal to or greater than \$0.10 for new options with an exercise price equal to fair market value of the common stock to be granted the first business day that was six months and one day after the cancellation date. Employees tendered options to purchase 4,919,090 shares of common stock during the offer period. In April 2004, 4,436,009 stock options were granted in connection with the tender offer.

Incentive Plan

In connection with the Series E offering, the Company created a management incentive plan (the Incentive Plan) for certain officers, founders and key employees of the Company. Under the terms of the Incentive Plan, up to 10% of any liquidation proceeds from the consolidation, merger, or sale of the Company will be distributed to the plan participants. Of the potential payout to a plan participant, 75% is based on a predetermined formula with the remaining 25% of the payout at the discretion of the administrators of the Incentive Plan. The potential payout is reduced by any amounts the participant would receive in the liquidation through stock option exercises or stock ownership. The Incentive Plan terminates upon a qualifying initial public offering of the Company's common stock.

Common Stock Warrants

In prior years, the Company has granted an aggregate of 2,016,842 warrants to purchase common stock. The common stock warrants begin to expire in February 2006 through to April 2015 with exercise prices ranging from \$0.60 to \$4.90. As of December 31, 2006, warrants to purchase 310,282 shares of common stock were outstanding.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Shares Reserved for Issuance

At December 31, 2006, the Company has reserved for future issuance the following shares of common stock upon conversion of preferred stock and the exercise of options and warrants:

Series A	10,683,130
Series B	6,902,114
Series C	20,023,442
Series C-1	423,730
Series D	24,248,733
Series E	24,005,548
Common stock available for future issuances under the Plan	5,316,147
Common stock available for outstanding options	13,619,700
Common stock warrants	310,282
	105,532,826

In addition, the Company has reserved 111,579 Series B shares, 214,062 Series D shares and 240,000 Series E shares pursuant to outstanding warrants.

12. Employee Benefit Plans

The Company has a 401(k) Plan for the benefit of all employees who meet certain eligibility requirements. This plan covers substantially all of the Company's full-time employees. The Company made \$181,000 and \$221,000 in contributions to the 401(k) Plan for the year ended December 31, 2005 and 2006, respectively. No contributions were made for the year ended December 31, 2004.

13. Related Party Transactions

On August 1, 2003, the Company entered into a Licensing and Services Agreement with a counterparty that until November 27, 2006 was a stockholder of the Company. Pursuant to the terms of the Licensing and Services Agreement, the Company granted the counterparty a license to certain digital marketing intelligence data and products. In each of 2004, 2005 and 2006, the Company recognized revenues of \$3 million. In relation to this counterparty, there were no outstanding amounts included in our accounts receivable balance as of December 31, 2004, 2005 and 2006.

14. Geographic Information

The Company attributes revenues to customers based on the location of the customer. The composition of the Company's sales to unaffiliated customers between those in the United States and those in other locations for each year is set forth below:

_	2004	(T +	2005	_	2006
		(In t	nousanas)		
\$	33,096	\$	46,900	\$	60,550
	1,798		2,479		3,150
	_		888		2,593
\$	34,894	\$	50,267	\$	66,293
	\$	\$ 33,096 1,798 ——— \$ 34,894	\$ 33,096 \$ 1,798 \$ \$ 34,894 \$	2004 2005 (In thousands) \$ 33,096	(In thousands) \$ 33,096 \$ 46,900 \$ 1,798 2,479

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

The composition of the Company's property, plant and equipment between those in the United States and those in other countries as of the end of each year is set forth below:

		December 31,
	2005	In thousands)
United States	\$ 4,06	3 \$ 6,525
Canada	41	3 305
United Kingdom		4 150
Total	\$ 4,48	6,980

15. Quarterly Financial Information (Unaudited)

					Three Montl	hs Ended				
	Mar. 31, 2005	Jun. 30, 2005	Sept. 30, 2005		Dec. 31, 2005		Mar. 31, 2006	Jun. 30, 2006	Sept. 30, 2006	Dec. 31, 2006
				(In tho	usands, except sha	re and pe	r share data)			
Revenues	\$ 11,135	\$ 13,150	\$ 12,953	\$	13,029	\$	14,985	\$ 16,906	\$ 16,165	\$ 18,237
Cost of revenues(1)	3,936	 4,863	4,602		4,817		5,148	5,205	 4,977	 5,230
Selling and marketing(1)	4,234	4,813	4,821		5,085		5,345	5,323	5,171	5,634
Research and development(1)	1,678	1,876	1,908		1,954		2,137	2,258	2,273	2,341
General and administrative(1)	1,489	1,804	1,779		2,017		1,918	2,176	1,897	2,302
Amortization	 621	 603	 612		601		371	 333	 333	 334
Total expenses from operations	 11,958	13,959	13,722		14,474		14,919	 15,295	14,651	15,841
(Loss) income from operations	(823)	(809)	(769)		(1,445)		66	1,611	1,514	2,396
Interest (expense) income, net	(58)	(71)	(39)		(40)		11	23	84	113
(Loss) gain from foreign currency	(21)	(1)	(72)		(2)		6	(33)	3	149
Revaluation of preferred stock warrant liabilities		 	 (6)		(8)		2	(211)	 (6)	 (9)
(Loss) income before income taxes and cumulative effect of change in		(00.)								
accounting principle	(902)	(881)	(886)		(1,495)		85	1,390	1,595	2,649
(Benefit) provision for income taxes	(53)	(52)	(38)		(39)					50
Net (loss) income before cumulative effect of change in accounting										
principle	(849)	(829)	(848)		(1,456)		85	1,390	1,595	2,599
Cumulative effect of change in accounting principle			(440)							
Net (loss) income	(849)	(829)	(1,288)		(1,456)		85	1,390	1,595	2,599
Accretion of redeemable preferred stock	(611)	 (643)	(675)		(709)		(742)	(777)	 (812)	 (848)
Net (loss) income attributable to common stockholders	\$ (1,460)	\$ (1,472)	\$ (1,963)	\$	(2,165)	\$	(657)	\$ 613	\$ 783	\$ 1,751

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

		Mar. 31, 2005	_	Jun. 30, 2005		Sept. 30, 2005	(In the	Dec. 31, 2005 ousands, except sha	re and	Mar. 31, 2006 per share data)		Jun. 30, 2006	_	Sept. 30, 2006	_	Б	ec. 31, 2006
Net (loss) income attributable to common stockholders	\$	(1,493)	\$	(1,505)	\$	(1,996)	\$	(2,199)	\$	(691)	\$	579	\$	7-	48	\$	1,716
Net (loss) income attributable to common stockholders per common share: Basic and diluted	s	(0.10)	\$	(0.10)	\$	(0.13)	\$	(0.14)	\$	(0.04)	\$	0.00	S	0.	00	\$	0.00
Weighted-average number of shares used in per share calculation — common stock		(0.10)	•	(0.10)	Ψ	(0.15)	•	(0.11)	Ψ.	(0.01)		0.00	•	0.			0.00
Basic and diluted		15,256,120		15,608,104		15,752,664		15,977,938		18,049,639		19,217,897		19,790,2	95		19,860,437
Net (loss) income attributable to common stockholders per common share subject to put:	\$	0.02	\$	0.02	\$	0.02	\$	0.02	\$	0.02	\$	0.02	\$	0.	02	\$	0.02
Basic and diluted																	
Weighted-average number of shares used in per share calculation — common share subject to put:																	
Basic and diluted		1,738,172		1,738,172		1,738,172		1,738,172		1,738,172		1,738,172		1,738,1	72		1,738,172
(1) Amortization of stock-based compensation is include	ed in t	he line items a	bove	as follows:													
Cost of revenues			\$	— \$		— \$		— \$	_	- \$	_	\$	2	\$	4	\$	6
Selling and marketing				_		_		_	-	_	6		26		23		27
Research and development				_		_		_	_	_	_		2		4		7
General and administrative				_		1		1		1	1		10		40		40
Total			\$	— \$		1 \$		1 \$		1 \$	7	\$	40	\$	71	\$	80
		•															
						F-30											

Table of Contents

Until , 2007 (25 days after the commencement of this offering) all dealers that effect transactions in these securities, whether or not participating in this offering, may be required to deliver a prospectus. This is in addition to the dealers' obligation to deliver a prospectus when acting as underwriters and with respect to their unsold allotments or subscriptions.

comScore.

Common Stock **PROSPECTUS**

Credit Suisse

Friedman Billings Ramsey

Deutsche Bank Securities

Jefferies & Company

William Blair & Company

, 2007

PART II

Information not required in prospectus

ITEM 13. Other Expenses of Issuance and Distribution

The following table sets forth the costs and expenses, other than underwriting discounts and commissions, payable by comScore, Inc. in connection with the sale of the common stock being registered hereby. All amounts are estimates except the SEC Registration Fee, the NASD filing fee and The NASDAQ Global Market listing fee.

	 Amount to be Paid
Securities and Exchange Commission registration fee	\$ 2,648
NASD filing fee	9,125
The NASDAQ Global Market listing fee	100,000
Blue Sky fees and expenses	*
Printing and engraving expenses	*
Legal fees and expenses	*
Accounting fees and expenses	*
Transfer agent and registrar fees	*
Miscellaneous	*
Total	*

^{*} To be filed by amendment

ITEM 14. Indemnification of Directors and Officers

Section 145(a) of the Delaware General Corporation Law provides that a Delaware corporation may indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the corporation) by reason of the fact that such person is or was a director, officer, employee or agent of another corporation or enterprise, against expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by such person in connection with such action, suit or proceeding if he or she acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the corporation, and, with respect to any criminal action or proceeding, had no cause to believe his or her conduct was unlawful.

Section 145(b) of the Delaware General Corporation Law provides that a Delaware corporation may indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the corporation to procure a judgment in its favor by reason of the fact that such person acted in any of the capacities set forth above, against expenses (including attorneys' fees) actually and reasonably incurred by such person in connection with the defense or settlement of such action or suit if he or she acted under similar standards, except that no indemnification may be made in respect of any claim, issue or matter as to which such person shall have been adjudged to be liable to the corporation unless and only to the extent that the court in which such action or suit was brought shall determine that, despite the adjudication of liability but in view of all the circumstances of the case, such person is fairly and reasonably entitled to be indemnified for such expenses which the court shall deem proper.

Section 145 of the Delaware General Corporation Law further provides that: (i) to the extent that a former or present director or officer of a corporation has been successful in the defense of any action, suit or proceeding referred to in subsections (a) and (b) or in the defense of any claim, issue or matter therein, such person shall be indemnified against expenses (including attorneys' fees) actually and reasonably incurred by

him or her in connection therewith; (ii) indemnification provided for by Section 145 shall not be deemed exclusive of any other rights to which the indemnified party may be entitled; and (iii) the corporation may purchase and maintain insurance on behalf of any present or former director, officer, employee or agent of the corporation or any person who at the request of the corporation was serving in such capacity for another entity against any liability asserted against such person and incurred by him or her in any such capacity or arising out of his or her status as such, whether or not the corporation would have the power to indemnify him or her against such liabilities under Section 145.

Article X of our amended and restated certificate of incorporation authorizes us to provide for the indemnification of directors to the fullest extent permissible under Delaware law.

Article VI of our bylaws provides for the indemnification of officers, directors and third parties acting on our behalf if such person acted in good faith and in a manner reasonably believed to be in and not opposed to our best interest and, with respect to any criminal action or proceeding, the indemnified party had no reason to believe his or her conduct was unlawful.

We have entered into indemnification agreements with our directors, executive officers and others, in addition to indemnification provided for in our bylaws, and intend to enter into indemnification agreements with any new directors and executive officers in the future.

We have purchased and intend to maintain insurance on behalf of any person who is or was a director or officer against any loss arising from any claim asserted against him or her and incurred by him or her in any such capacity, subject to certain exclusions.

See also the undertakings set out in response to Item 17 herein.

ITEM 15. Recent Sales of Unregistered Securities

In the past three years, we have issued and sold the following securities as adjusted for the -for- stock split:

- 1. From December 7, 1999 through the date hereof, we have granted options to purchase 34,774,285 shares of our Common Stock at a weighted average exercise price of \$0.36 per share. As of March 21, 2007, 9,635,397 of these options had been exercised at prices ranging from \$0.05 to \$2.00 per share.
- 2. On March 18, 2007, we awarded an aggregate of 3,559,500 shares of our restricted stock to certain of our named executive officers and our employees based upon the recommendations of our compensation committee. The Company has a right of repurchase on such shares that lapses ratably over a 48 month period.
 - 3. In April 2005, we issued a warrant to purchase 68,182 shares of our common stock at a price of \$1.10 per share. That warrant has not been exercised as of the date hereof.

The sales of the above securities were deemed to be exempt from registration under the Securities Act with respect to items 1 and 2 above in reliance on Rule 701 promulgated under Section 3(b) of the Securities Act as transactions by an issuer not involving a public offering or transactions pursuant to compensatory benefit plans and contracts relating to compensation as provided under such Rule 701, and with respect to item 3 above in reliance on Section 4(2) of the Securities Act. The recipients of securities in each such transaction represented their intention to acquire the securities for investment only and not with a view to or for sale in connection with any distribution thereof and appropriate legends were affixed to the share certificates and warrants issued in such transactions. All recipients had adequate access, through their relationships with us, to information about us.

ITEM 16. Exhibits and Financial Statement Schedules

(a) Exhibits.

A list of exhibits filed herewith is contained in the exhibit index that immediately precedes such exhibits and is incorporated herein by reference.

(b) Financial Statement Schedule

SCHEDULE II — VALUATION AND QUALIFYING ACCOUNTS

	As of December 31,				
	 2004		2005		2006
	 	(In	thousands)		
Allowance for Doubtful Accounts					
Beginning Balance	\$ (298)	\$	(102)	\$	(185)
Additions	(12)		(90)		(212)
Reductions	208		7		209
Ending Balance	\$ (102)	\$	(185)	\$	(188)
Deferred Tax Valuation Allowance					
Beginning Balance	\$ (32,698)	\$	(33,056)	\$	(36,139)
Additions	(358)		(3,083)		_
Reductions	_		_		2,393
Ending Balance	\$ (33,056)	\$	(36,139)	\$	(33,746)

Report of Independent Registered Public Accounting Firm

Board of Directors comScore, Inc.

We have audited the consolidated financial statements of comScore, Inc. as of December 31, 2005 and 2006, and for each of the three years in the period ended December 31, 2006, and have issued our report thereon dated March 29, 2007 (including elsewhere in this Registration Statement). Our audits also included the financial statement schedule listed in Item 16(b) of Form S-1 of this Registration Statement. These schedules are the responsibility of the Company's management. Our responsibility is to express an opinion based on our audits.

In our opinion, the financial statement schedule referred to above, when considered in relation to the basic financial statements taken as a whole, presents fairly in all material respects the information set forth therein.

/s/ Ernst & Young LLP

McLean, VA March 29, 2007

ITEM 17. Undertakings

The undersigned Registrant hereby undertakes to provide to the underwriters at the closing specified in the underwriting agreement certificates in such denominations and registered in such names as required by the underwriters to permit prompt delivery to each purchaser.

Insofar as indemnification by the Registrant for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of the Registrant pursuant to the provisions described in Item 14 or otherwise, the Registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the Registrant of expenses incurred or paid by a director, officer or controlling person of the Registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the Registrant will, unless in the opinion of its counsel the matter has been settled by controlling persons of the Registrant to expense incurred or paid by a director, officer or controlling person in connection with the securities being registered, the Registrant will, unless in the opinion of its counsel the matter has been settled by controlling person of the Registrant to expense incurred or paid by a director, officer or controlling person in connection with the securities being registered, the Registrant public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.

The undersigned Registrant hereby undertakes that:

- (1) For purposes of determining any liability under the Securities Act, the information omitted from the form of prospectus filed as part of this registration statement in reliance upon Rule 430A and contained in a form of prospectus filed by the Registrant pursuant to Rule 424(b)(1) or (4) or 497(b) under the Securities Act shall be deemed to be part of this registration statement as of the time it was declared effective.
- (2) For the purpose of determining any liability under the Securities Act, each post-effective amendment that contains a form of prospectus shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof.

Signatures

Pursuant to the requirements of the Securities Act of 1933, the registrant has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in Reston, Commonwealth of Virginia, on the thirtieth day of March, 2007.

comScore, Inc.

/s/ Magid M. Abraham
Magid M. Abraham, Ph.D.
President, Chief Executive
Officer and Director

Power of Attorney

KNOW ALL PERSONS BY THESE PRESENTS, that each person whose signature appears below constitutes and appoints Magid M. Abraham, Ph.D. and John M. Green, and each of them, as his true and lawful attorneys-in-fact and agents, with full power of substitution and resubstitution, for him and in his name, place and stead, in any and all capacities, to sign any and all amendments (including post-effective amendments) to this Registration Statement (including any registration statement filed by a corporation that is a successor to comScore, Inc. by merger) and to sign any registration statement for the same offering covered by the Registration Statement that is to be effective upon filing pursuant to Rule 462 promulgated under the Securities Act of 1933, and to file the same, with all exhibits thereto, and other documents in connection therewith, with the Securities and Exchange Commission, granting unto said attorneys-in-fact and agents, and each of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection therewith and about the premises, as fully to all intents and purposes as he might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents, or any of them, or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1933, this Registration Statement has been signed by the following persons in the capacities and on the dates indicated:

Signature	<u>T</u> itle	Date
/s/ Magid M. Abraham Magid M. Abraham, Ph.D.	President, Chief Executive Officer (Principal Executive Officer) and Director	March 30, 2007
/s/ John M. Green John M. Green	Chief Financial Officer (Principal Financial and Accounting Officer)	March 30, 2007
/s/ Gian M. Fulgoni Gian M. Fulgoni	Executive Chairman of the Board of Directors	March 30, 2007
/s/ Thomas D. Berman Thomas D. Berman	Director	March 30, 2007
/s/ Bruce Golden Bruce Golden	Director	March 30, 2007

Signature	_Title	Date
/s/ William J. Henderson	Director	March 30, 2007
William J. Henderson		
/s/ Ronald J. Korn	Director	March 30, 2007
Ronald J. Korn		
/s/ Frederick R. Wilson	Director	March 30, 2007
Frederick R. Wilson	•	

EXHIBIT INDEX

	EAHIBIT INDEX
Exhibit Number	<u>Description</u>
1.1*	Form of Underwriting Agreement
3.1*	Amended and Restated Certificate of Incorporation currently in effect
3.2	Amended and Restated Bylaws currently in effect
3.3*	Form of Amended and Restated Certificate of Incorporation of the Registrant (to be effective upon the closing of the offering)
3.4*	Form of Amended and Restated Bylaws of the Registrant (to be effective upon the closing of the offering)
4.1*	Specimen Common Stock Certificate
4.2	Fourth Amended and Restated Investor Rights Agreement by and among comScore Networks, Inc. and certain holders of preferred stock, dated August 1, 2003
4.3	Warrant to purchase 46,551 shares of Series B Convertible Preferred Stock, dated June 9, 2000
4.4	Warrant to purchase 20,100 shares of common stock, dated July 31, 2000
4.5	Warrant to purchase 9,694 shares of Series B Convertible Preferred Stock, dated September 29, 2000
4.6	Warrant to purchase 100,000 shares of common stock, dated June 26, 2001
4.7	Warrant to purchase 10,000 shares of common stock, dated November 30, 2001
4.8 4.9	Warrant to purchase 12,000 shares of common stock, dated July 3, 2002 Warrant to purchase 36,127 shares of Series D Convertible Preferred Stock, dated July 31, 2002
4.10	Warrant to purchase 30,127 shares of Series D Convertible Preferred Stock, dated July 31, 2002 Warrant to purchase 108,382 shares of Series D Convertible Preferred Stock, dated July 31, 2002
4.10	Warrant to purchase 45,854 shares of Series D Convertible Preferred Stock, dated July 31, 2002
4.11	Warrant to purchase 100,000 shares of common stock, dated June 24, 2003
4.13	Warrant to purchase 240,000 shares of Series E Convertible Preferred Stock, dated December 19, 2003
4.14	Warrant to purchase 68,182 shares of common stock, dated April 29, 2005
4.15	Stock Restriction and Put Right Agreement by and between comScore Networks, Inc. and Lawrence Denaro, dated July 28, 2004
4.16	Stock Restriction and Put Right Agreement by and among comScore Networks, Inc., 954253 Ontario, Inc. and Rice and Associates Advertising Consultants, Inc., dated
	January 1, 2005
5.1*	Opinion of Wilson Sonsini Goodrich & Rosati, Professional Corporation
10.1	Form of Indemnification Agreement for directors and executive officers
10.2	1999 Stock Plan
10.3	Form of Stock Option Agreement under 1999 Stock Plan
10.4	Form of Notice of Grant of Restricted Stock Purchase Right under 1999 Stock Plan
10.5	Form of Notice of Grant of Restricted Stock Units under 1999 Stock Plan
10.6	2007 Equity Incentive Plan
10.7	Form of Notice of Grant of Stock Option under 2007 Equity Incentive Plan
10.8	Form of Notice of Grant of Restricted Stock under 2007 Equity Incentive Plan
10.9	Form of Notice of Grant of Restricted Stock Units under 2007 Equity Incentive Plan
10.10	Stock Option Agreement with Magid M. Abraham, dated December 16, 2003
10.11	Stock Option Agreement with Gian M. Fulgoni, dated December 16, 2003
10.12	Lease Agreement by and between comScore Networks, Inc. and Comstock Partners, L.C., dated June 23, 2003, as amended
10.13	Separation Agreement with Sheri L. Huston, dated February 28, 2006
10.14	Letter Agreement with John M. Green, dated May 8, 2006 List of Subsidiaries
21.1 23.1	Consent of Ernst & Young LLP
23.2*	Consent of Wilson Sonsini Goodrich & Rosati, Professional Corporation (included in Exhibit 5.1)
24.1	Power of Attorney (included on page II-4)
44.1	roner or mionicy (metaded on page 11-7)

^{*} To be filed by amendment.

BYLAWS

OF

COMSCORE NETWORKS, INC. (a Delaware corporation)

TABLE OF CONTENTS

ARTICLE I — COR	PORATE OFFICES	1
1.1	DEGICTEDED OFFICE	1
1.1 1.2	REGISTERED OFFICE	1
1.2	OTHER OFFICES	1
ARTICLE II — MEI	ETINGS OF STOCKHOLDERS	1
2.1	PLACE OF MEETINGS	1
2.2	ANNUAL MEETING	1
2.3	SPECIAL MEETING	1
2.4	NOTICE OF STOCKHOLDERS' MEETINGS	2
2.5	MANNER OF GIVING NOTICE; AFFIDAVIT OF NOTICE	2
2.6	QUORUM	2
2.7	ADJOURNED MEETING; NOTICE	3
2.8	VOTING	3
2.9	VALIDATION OF MEETINGS; WAIVER OF NOTICE; CONSENT	3
2.10	STOCKHOLDER ACTION BY WRITTEN CONSENT WITHOUT A MEETING	4
2.11	RECORD DATE FOR STOCKHOLDER NOTICE; VOTING; GIVING CONSENTS	4
2.12	PROXIES	4
2.13	INSPECTORS OF ELECTION	5
ARTICLE III — DII	RECTORS	5
3.1	POWERS	5
3.2	NUMBER AND TERM OF OFFICE	6
3.3	RESIGNATION AND VACANCIES	6
3.4	REMOVAL	7
3.5	PLACE OF MEETINGS; MEETINGS BY TELEPHONE	7
3.6	REGULAR MEETINGS	7
3.7	SPECIAL MEETINGS; NOTICE	7
3.8	QUORUM	7
3.9	WAIVER OF NOTICE	8
3.10	ADJOURNMENT	8
3.11	NOTICE OF ADJOURNMENT	8
3.12	BOARD ACTION BY WRITTEN CONSENT WITHOUT A MEETING	8
3.13	FEES AND COMPENSATION OF DIRECTORS	8
3.14	APPROVAL OF LOANS TO OFFICERS	8
ARTICLE IV — CO	MMITTEEC	9
ARTICLE IV — CO	INIMIT LEES	9
4.1	COMMITTEES OF DIRECTORS	9
4.2	MEETINGS AND ACTION OF COMMITTEES	9

5.1 5.2 5.3	OFFICERS ELECTION OF OFFICERS SUBORDINATE OFFICERS	9 10 10
5.4	REMOVAL AND RESIGNATION OF OFFICERS	10
5.5	VACANCIES IN OFFICES	10
5.6	CHAIRMAN OF THE BOARD	10
5.7	CHIEF EXECUTIVE OFFICER	10
5.8	PRESIDENT	10
5.9	VICE PRESIDENTS	11
5.10	SECRETARY	11
5.11	CHIEF FINANCIAL OFFICER	11
ARTICLE VI — INC	DEMNIFICATION OF DIRECTORS, OFFICERS, EMPLOYEES, AND OTHER AGENTS	12
6.1	INDEMNIFICATION OF DIRECTORS AND OFFICERS	12
6.2	INDEMNIFICATION OF OTHERS	12
6.3	INSURANCE	12
ARTICLE VII — RE	CORDS AND REPORTS	12
7.1	MAINTENANCE AND INSPECTION OF RECORDS	12
7.2	INSPECTION BY DIRECTORS	13
7.3	ANNUAL STATEMENT TO STOCKHOLDERS	13
7.4	REPRESENTATION OF SHARES OF OTHER CORPORATIONS	13
ARTICLE VIII — G	ENERAL MATTERS	13
8.1	RECORD DATE FOR PURPOSES OTHER THAN NOTICE AND VOTING	13
8.2	CHECKS; DRAFTS; EVIDENCES OF INDEBTEDNESS	14
8.3	CORPORATE CONTRACTS AND INSTRUMENTS: HOW EXECUTED	14
8.4	STOCK CERTIFICATES; PARTLY PAID SHARES	14
8.5	SPECIAL DESIGNATION ON CERTIFICATES	14
8.6	LOST CERTIFICATES	15
8.7	CONSTRUCTION; DEFINITIONS	15
ARTICLE IX — AM	ENDMENTS	15
ARTICLE X — DISS	GOLUTION	15
ARTICLE XI — CUS	STODIAN	16
11.1	APPOINTMENT OF A CUSTODIAN IN CERTAIN CASES	16
11.2	DUTIES OF CUSTODIAN	16
		-ji-

ARTICLE V — OFFICERS

BYLAWS OF

COMSCORE NETWORKS, INC.

(a Delaware corporation)

ARTICLE I

CORPORATE OFFICES

1.1 REGISTERED OFFICE

The registered office of the corporation shall be fixed in the Certificate of Incorporation of the corporation.

1.2 OTHER OFFICES .

The board of directors may at any time establish branch or subordinate offices at any place or places where the corporation is qualified to do business.

ARTICLE II

MEETINGS OF STOCKHOLDERS

2.1 PLACE OF MEETINGS .

Meetings of stockholders shall be held at any place within or outside the State of Delaware designated by the board of directors. In the absence of any such designation, stockholders' meetings shall be held at the registered office of the corporation.

2.2 ANNUAL MEETING

The annual meeting of stockholders shall be held each year on a date and at a time designated by the board of directors. In the absence of such designation, the annual meeting of stockholders shall be held on the first Wednesday of April in each year at 1:00 p.m. However, if such day falls on a legal holiday, then the meeting shall be held at the same time and place on the next succeeding full business day. At the meeting, directors shall be elected, and any other proper business may be transacted.

2.3 SPECIAL MEETING

A special meeting of the stockholders may be called at any time by the board of directors, or by the chairman of the board, by the president or by the chief executive officer, or by one or more stockholders holding shares in the aggregate entitled to cast not less than ten percent (10%) of the votes at that meeting.

If a special meeting is called by any person or persons other than the board of directors, the request shall be in writing, specifying the time of such meeting and the general nature of the business proposed to be transacted, and shall be delivered personally or sent by registered mail or by telegraphic or other facsimile

transmission to the chairman of the board, the president, chief executive officer, or the secretary of the corporation. No business may be transacted at such special meeting otherwise than specified in such notice. The officer receiving the request shall cause notice to be promptly given to the stockholders entitled to vote, in accordance with the provisions of Sections 2.4 and 2.5, that a meeting will be held at the time requested by the person or persons who called the meeting, not less than thirty-five (35) nor more than sixty (60) days after the receipt of the request. If the notice is not given within twenty (20) days after the receipt of the request, are person or persons requesting the meeting may give the notice. Nothing contained in this paragraph of this Section 2.3 shall be construed as limiting, fixing, or affecting the time when a meeting of stockholders called by action of the board of directors may be held.

2.4 NOTICE OF STOCKHOLDERS' MEETINGS

Except as set forth in Section 2.3, all notices of meetings of stockholders shall be sent or otherwise given in accordance with Section 2.5 of these bylaws not less than ten (10) nor more than sixty (60) days before the date of the meeting. The notice shall specify the place, date, and hour of the meeting and (i) in the case of a special meeting, the general nature of the business to be transacted (no business other than that specified in the notice may be transacted) or (ii) in the case of the annual meeting, those matters which the board of directors, at the time of giving the notice, intends to present for action by the stockholders (but any proper matter may be presented at the meeting for such action). The notice of any meeting at which directors are to be elected shall include the name of any nominee or nominees who, at the time of the notice, the board intends to present for election.

2.5 MANNER OF GIVING NOTICE; AFFIDAVIT OF NOTICE

Written notice of any meeting of stockholders shall be given either personally or by first-class mail or by facsimile, telegraphic or other written communication. Notices not personally delivered shall be sent charges prepaid and shall be addressed to the stockholder at the address of that stockholder appearing on the books of the corporation or given by the shareholder to the corporation for the purpose of notice. If no such address appears on the corporation's books or is given, notice shall be deemed to have been given if sent to that stockholder by mail or telegraphic or other written communication to the corporation's principal executive office, or if published at least once in a newspaper of general circulation in the county where that office is located. Notice shall be deemed to have been given at the time when delivered personally or deposited in the mail or sent by telegram or other means of written communication.

If any notice addressed to a stockholder at the address of that stockholder appearing on the books of the corporation is returned to the corporation by the United States Postal Service marked to indicate that the United States Postal Service is unable to deliver the notice to the stockholder at that address, then all future notices or reports shall be deemed to have been duly given without further mailing if the same shall be available to the stockholder on written demand of the stockholder at the principal executive office of the corporation for a period of one (1) year from the date of the giving of the notice.

An affidavit of the mailing or other means of giving any notice of any stockholders' meeting, executed by the secretary, assistant secretary or any transfer agent of the corporation giving the notice, shall be prima facie evidence of the giving of such notice.

2.6 <u>QUORUM</u> .

The presence in person or by proxy of the holders of a majority of the shares entitled to vote thereat constitutes a quorum for the transaction of business at all meetings of stockholders. The stockholders present at a duly called or held meeting at which a quorum is present may continue to do business until adjournment, notwithstanding the withdrawal of enough stockholders to leave less than a quorum, if any action taken (other than adjournment) is approved by at least a majority of the shares required to constitute a quorum.

2.7 ADJOURNED MEETING; NOTICE

Any stockholders' meeting, annual or special, whether or not a quorum is present, may be adjourned from time to time by the vote of the majority of the shares represented at that meeting, either in person or by proxy. In the absence of a quorum, no other business may be transacted at that meeting except as provided in Section 2.6 of these bylaws.

When any meeting of stockholders, either annual or special, is adjourned to another time or place, notice need not be given of the adjourned meeting if the time and place are announced at the meeting at which the adjournment is taken. However, if a new record date for the adjourned meeting is fixed or if the adjournment is for more than thirty (30) days from the date set for the original meeting, then notice of the adjourned meeting shall be given. Notice of any such adjourned meeting shall be given to each stockholder of record entitled to vote at the adjourned meeting in accordance with the provisions of Sections 2.4 and 2.5 of these bylaws. At any adjourned meeting the corporation may transact any business which might have been transacted at the original meeting.

2.8 VOTING

The stockholders entitled to vote at any meeting of stockholders shall be determined in accordance with the provisions of Section 2.11 of these bylaws, subject to the provisions of Sections 217 and 218 of the General Corporation Law of Delaware (relating to voting rights of fiduciaries, pledgors and joint owners, and to voting trusts and other voting agreements).

Except as may be otherwise provided in the Certificate of Incorporation, each outstanding share, regardless of class, shall be entitled to one vote on each matter submitted to a vote of the stockholders. Any stockholder entitled to vote on any matter may vote part of the shares in favor of the proposal and refrain from voting the remaining shares or, except when the matter is the election of directors, may vote them against the proposal; but, if the stockholder fails to specify the number of shares which the stockholder is voting affirmatively, it will be conclusively presumed that the stockholder's approving vote is with respect to all shares which the stockholder is entitled to vote.

If a quorum is present, the affirmative vote of the majority of the shares represented and voting at a duly held meeting (which shares voting affirmatively also constitute at least a majority of the required quorum) shall be the act of the stockholders, unless the vote of a greater number or a vote by classes is required by law or by the Certificate of Incorporation.

At a stockholders' meeting at which directors are to be elected, a stockholder shall be entitled to cumulate votes (i.e., cast for any candidate a number of votes greater than the number of votes which such stockholder normally is entitled to cast) if the candidates' names have been placed in nomination prior to commencement of the voting and the stockholder has given notice prior to commencement of the voting of the stockholder's intention to cumulate votes. If any stockholder has given such a notice, then every stockholder entitled to vote may cumulate votes for candidates in nomination either (i) by giving one candidate a number of votes equal to the number of directors to be elected multiplied by the number of votes to which that stockholder's shares are normally entitled or (ii) by distributing the stockholder's votes on the same principle among any or all of the candidates, as the stockholder thinks fit. The candidates receiving the highest number of affirmative votes, up to the number of directors to be elected, shall be elected; votes against any candidate and votes withheld shall have no legal effect.

2.9 <u>VALIDATION OF MEETINGS; WAIVER OF NOTICE; CONSENT</u>.

The transactions of any meeting of stockholders, either annual or special, however called and noticed, and wherever held, shall be as valid as though they had been taken at a meeting duly held after regular call and notice, if a quorum be present either in person or by proxy, and if, either before or after the meeting, each

person entitled to vote, who was not present in person or by proxy, signs a written waiver of notice or a consent to the holding of the meeting or an approval of the minutes thereof. The waiver of notice or consent or approval need not specify either the business to be transacted or the purpose of any annual or special meeting of stockholders. All such waivers, consents, and approvals shall be filed with the corporate records or made a part of the minutes of the meeting.

Attendance by a person at a meeting shall also constitute a waiver of notice of and presence at that meeting, except when the person objects at the beginning of the meeting to the transaction of any business because the meeting is not lawfully called or convened. Attendance at a meeting is not a waiver of any right to object to the consideration of matters required by law to be included in the notice of the meeting but not so included, if that objection is expressly made at the meeting.

2.10 STOCKHOLDER ACTION BY WRITTEN CONSENT WITHOUT A MEETING.

Unless otherwise provided in the Certificate of Incorporation, any action which may be taken at any annual or special meeting of stockholders may be taken without a meeting and without prior notice, if a consent in writing, setting forth the action so taken, is signed by the holders of outstanding shares having not less than the minimum number of votes that would be necessary to authorize or take that action at a meeting at which all shares entitled to vote on that action were present and voted.

Prompt notice of the taking of the corporate action without a meeting by less than unanimous written consent shall be given to those stockholders who have not consented in writing. If the action which is consented to is such as would have required the filing of a certificate under any section of the General Corporation Law of Delaware if such action had been voted on by stockholders at a meeting thereof, then the certificate filed under such section shall state, in lieu of any statement required by such section concerning any vote of stockholders, that written notice and written consent have been given as provided in Section 228 of the General Corporation Law of Delaware.

2.11 RECORD DATE FOR STOCKHOLDER NOTICE; VOTING; GIVING CONSENTS

For purposes of determining the stockholders entitled to notice of any meeting or to vote thereat or entitled to give consent to corporate action without a meeting, the board of directors may fix, in advance, a record date, which shall not be more than sixty (60) days nor less than ten (10) days before the date of any such meeting nor more than sixty (60) days before any such action without a meeting, and in such event only stockholders of record on the date so fixed are entitled to notice and to vote or to give consents, as the case may be, notwithstanding any transfer of any shares on the books of the corporation after the record date.

If the board of directors does not so fix a record date:

- (a) the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the business day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the business day next preceding the day on which the meeting is held; and
- (b) the record date for determining stockholders entitled to give consent to corporate action in writing without a meeting, (i) when no prior action by the board has been taken, shall be the day on which the first written consent is given, or (ii) when prior action by the board has been taken, shall be at the close of business on the day on which the board adopts the resolution relating to that action.

The record date for any other purpose shall be as provided in Article VIII of these bylaws.

2.12 PROXIES.

Every person entitled to vote for directors, or on any other matter, shall have the right to do so either in person or by one or more agents authorized by a written proxy signed by the person and filed with the secretary of the corporation, but no such proxy shall be voted or acted upon after three (3) years from its date, unless the proxy provides for a longer period. A proxy shall be deemed signed if the stockholder's name is placed on the proxy (whether by manual signature, typewriting, telegraphic transmission or otherwise) by the stockholder or the stockholder's attorney-in-fact. The revocability of a proxy that states on its face that it is irrevocable shall be governed by the provisions of Section 212(c) of the General Corporation Law of Delaware.

2.13 INSPECTORS OF ELECTION .

Before any meeting of stockholders, the board of directors may appoint an inspector or inspectors of election to act at the meeting or its adjournment. If no inspector of election is so appointed, then the chairman of the meeting may, and on the request of any stockholder or a stockholder's proxy shall, appoint an inspectors of election to act at the meeting. The number of inspectors shall be either one (1) or three (3). If inspectors are appointed at a meeting pursuant to the request of one (1) or more stockholders or proxies, then the holders of a majority of shares or their proxies present at the meeting shall determine whether one (1) or three (3) inspectors are to be appointed. If any person appointed as inspector fails to appear or fails or refuses to act, then the chairman of the meeting may, and upon the request of any stockholder or a stockholder's proxy shall, appoint a person to fill that vacancy.

Such inspectors shall:

- (a) determine the number of shares outstanding and the voting power of each, the number of shares represented at the meeting, the existence of a quorum, and the authenticity, validity, and effect of proxies;
- (b) receive votes, ballots or consents;
- (c) hear and determine all challenges and questions in any way arising in connection with the right to vote;
- (d) count and tabulate all votes or consents;
- (e) determine when the polls shall close;
- (f) determine the result; and
- (g) do any other acts that may be proper to conduct the election or vote with fairness to all stockholders.

ARTICLE II

DIRECTORS

3.1 POWERS .

Subject to the provisions of the General Corporation Law of Delaware and to any limitations in the Certificate of Incorporation or these bylaws relating to action required to be approved by the stockholders or by the outstanding shares, the business and affairs of the corporation shall be managed and all corporate powers shall be exercised by or under the direction of the board of directors.

3.2 NUMBER AND TERM OF OFFICE

The authorized number of directors shall be not less than eight (8) nor more than ten (10). The exact number of directors shall be nine (9) until changed, within the limits specified above, by a bylaw amending this Section 3.2, duly adopted by the board of directors or by the stockholders. The indefinite number of directors may be changed, or a definite number may be fixed without provision for an indefinite number, by a duly adopted amendment to the Certificate of Incorporation or by an amendment to this bylaw adopted by the vote or written consent of holders of a majority of the outstanding shares entitled to vote or by resolution of a majority of the board of directors.

No reduction of the authorized number of directors shall have the effect of removing any director before that director's term of office expires. If for any cause, the directors shall not have been elected at an annual meeting, they may be elected as soon thereafter as convenient at a special meeting of the stockholders called for that purpose in the manner provided in these Bylaws.

3.3 RESIGNATION AND VACANCIES

Any director may resign effective on giving written notice to the chairman of the board, the president, the secretary or the board of directors, unless the notice specifies a later time for that resignation to become effective. If the resignation of a director is effective at a future time, the board of directors may elect a successor to take office when the resignation becomes effective.

Vacancies in the board of directors may be filled by a majority of the remaining directors, even if less than a quorum, or by a sole remaining director; however, a vacancy created by the removal of a director by the vote or written consent of the stockholders or by court order may be filled only by the affirmative vote of a majority of the shares represented and voting at a duly held meeting at which a quorum is present (which shares voting affirmatively also constitute a majority of the required quorum), or by the unanimous written consent of all shares entitled to vote thereon. Each director so elected shall hold office until the next annual meeting of the stockholders and until a successor has been elected and qualified.

Unless otherwise provided in the Certificate of Incorporation or these bylaws:

- (a) Vacancies and newly created directorships resulting from any increase in the authorized number of directors elected by all of the stockholders having the right to vote as a single class may be filled by a majority of the directors then in office, although less than a quorum, or by a sole remaining director.
- (b) Whenever the holders of any class or classes of stock or series thereof are entitled to elect one or more directors by the provisions of the certificate of incorporation, vacancies and newly created directorships of such class or classes or series may be filled by a majority of the directors elected by such class or classes or series thereof then in office, or by a sole remaining director so elected.

If at any time, by reason of death or resignation or other cause, the corporation should have no directors in office, then any officer or any stockholder or an executor, administrator, trustee or guardian of a stockholder, or other fiduciary entrusted with like responsibility for the person or estate of a stockholder, may call a special meeting of stockholders in accordance with the provisions of the certificate of incorporation or these bylaws, or may apply to the Court of Chancery for a decree summarily ordering an election as provided in Section 211 of the General Corporation Law of Delaware.

If, at the time of filling any vacancy or any newly created directorship, the directors then in office constitute less than a majority of the whole board (as constituted immediately prior to any such increase), then the Court of Chancery may, upon application of any stockholder or stockholders holding at least ten (10) percent of the total number of the shares at the time outstanding having the right to vote for such directors,

summarily order an election to be held to fill any such vacancies or newly created directorships, or to replace the directors chosen by the directors then in office as aforesaid, which election shall be governed by the provisions of Section 211 of the General Corporation Law of Delaware as far as applicable.

3.4 REMOVAL

Subject to any limitations imposed by law or the Certificate of Incorporation, the Board of Directors, or any individual Director, may be removed from office at any time with or without cause by the affirmative vote of the holders of at least a majority of the then outstanding shares of the capital stock of the corporation entitled to vote at an election of Directors.

3.5 PLACE OF MEETINGS; MEETINGS BY TELEPHONE.

Regular meetings of the board of directors may be held at any place within or outside the State of Delaware that has been designated from time to time by resolution of the board. In the absence of such a designation, regular meetings shall be held at the principal executive office of the corporation. Special meetings of the board may be held at any place within or outside the State of Delaware that has been designated in the notice of the meeting or, if not stated in the notice or if there is no notice, at the principal executive office of the corporation.

Any meeting, regular or special, may be held by conference telephone or similar communication equipment, so long as all directors participating in the meeting can hear one another; and all such directors shall be deemed to be present in person at the meeting.

3.6 REGULAR MEETINGS

Regular meetings of the board of directors may be held without notice if the times of such meetings are fixed by the board of directors.

3.7 SPECIAL MEETINGS; NOTICE

Special meetings of the board of directors for any purpose or purposes may be called at any time by the chairman of the board, the president, any vice president, the secretary or any two directors.

Notice of the time and place of special meetings shall be delivered personally or by telephone to each director or sent by first-class mail or telegram, charges prepaid, addressed to each director at that director's address as it is shown on the records of the corporation. If the notice is mailed, it shall be deposited in the United States mail at least foru (4) days before the time of the holding of the meeting. If the notice is delivered personally or by telephone or by facsimile, it shall be delivered personally or by telephone or by facsimile, it shall be delivered personally or by telephone may be communicated either to the director or to a person at the office of the director who the person giving the notice has reason to believe will promptly communicate it to the director. The notice need not specify the purpose or the place of the meeting, if the meeting is to be held at the principal executive office of the corporation.

3.8 <u>QUORUM</u> .

A majority of the authorized number of directors shall constitute a quorum for the transaction of business, except to adjourn as provided in Section 3.11 of these bylaws. Every act or decision done or made by a majority of the directors present at a duly held meeting at which a quorum is present shall be regarded as the act of the board of directors, subject to the provisions of the Certificate of Incorporation and applicable law.

A meeting at which a quorum is initially present may continue to transact business notwithstanding the withdrawal of directors, if any action taken is approved by at least a majority of the required quorum for that meeting.

3.9 WAIVER OF NOTICE

Notice of a meeting need not be given to any director (i) who signs a waiver of notice or a consent to holding the meeting or an approval of the minutes thereof, whether before or after the meeting, or (ii) who attends the meeting without protesting, prior thereto or at its commencement, the lack of notice to such directors. All such waivers, consents, and approvals shall be filed with the corporate records or made part of the minutes of the meeting. A waiver of notice need not specify the purpose of any regular or special meeting of the board of directors.

3.10 ADJOURNMENT

A majority of the directors present, whether or not constituting a quorum, may adjourn any meeting to another time and place.

3.11 NOTICE OF ADJOURNMENT

Notice of the time and place of holding an adjourned meeting need not be given unless the meeting is adjourned for more than twenty-four (24) hours. If the meeting is adjourned for more than twenty-four (24) hours, then notice of the time and place of the adjourned meeting shall be given before the adjourned meeting takes place, in the manner specified in Section 3.8 of these bylaws, to the directors who were not present at the time of the adjournment.

3.12 BOARD ACTION BY WRITTEN CONSENT WITHOUT A MEETING

Any action required or permitted to be taken by the board of directors may be taken without a meeting, provided that all members of the board individually or collectively consent in writing to that action. Such action by written consent shall have the same force and effect as a unanimous vote of the board of directors. Such written consent and any counterparts thereof shall be filed with the minutes of the proceedings of the board.

3.13 FEES AND COMPENSATION OF DIRECTORS

Directors and members of committees may receive such compensation, if any, for their services and such reimbursement of expenses as may be fixed or determined by resolution of the board of directors. This Section 3.14 shall not be construed to preclude any director from serving the corporation in any other capacity as an officer, agent, employee or otherwise and receiving compensation for those services.

3.14 APPROVAL OF LOANS TO OFFICERS.

The corporation may lend money to, or guarantee any obligation of, or otherwise assist any officer or other employee of the corporation or of its subsidiary, including any officer or employee who is a director of the corporation or its subsidiary, whenever, in the judgment of the directors, such loan, guaranty or assistance may reasonably be expected to benefit the corporation. The loan, guaranty or other assistance may be with or without interest and may be unsecured, or secured in such manner as the board of directors shall approve, including, without limitation, a pledge of shares of stock of the corporation. Nothing contained in this section shall be deemed to deny, limit or restrict the powers of guaranty or warranty of the corporation at common law or under any statute.

ARTICLE IV

COMMITTEES

4.1 COMMITTEES OF DIRECTORS

The board of directors may, by resolution adopted by a majority of the authorized number of directors, designate one (1) or more committees, each consisting of two or more directors, to serve at the pleasure of the board. The board may designate one (1) or more directors as alternate members of any committee, who may replace any absent member at any meeting of the committee. The appointment of members or alternate members of a committee requires the vote of a majority of the authorized number of directors. Any committee, to the extent provided in the resolution of the board, shall have all the authority to (i) amend the Certificate of Incorporation (except that a committee may, to the extent authorized in the resolution or resolutions providing for the issuance of shares of stock adopted by the board of directors as provided in Section 151(a) of the General Corporation Law of Delaware, fix any of the preferences or rights of such shares relating to dividends, redemption, dissolution, any distribution of assets of the corporation or the conversion into, or the exchange of such shares for, shares of any other class or classes or any other series of the same or any other class or classes of stock of the corporation), (ii) adopt an agreement of merger or consolidation under Sections 251 or 252 of the General Corporation Law of Delaware, (iii) recommend to the stockholders the sale, lease or exchange of all or substantially all of the corporation's property and assets, (iv) recommend to the stockholders a dissolution of a dissolution, or (v) amend the bylaws of the corporation; and, unless the board resolution establishing the committee, the bylaws or the Certificate of Incorporation expressly so provide, no such committee shall have the power or authority to declare a dividend, to authorize the issuance of stock, or to adopt a certificate of ownership and merger pursuant to Section 253 of the General Corporation Law of Delaware.

4.2 MEETINGS AND ACTION OF COMMITTEES .

Meetings and actions of committees shall be governed by, and held and taken in accordance with, the provisions of Article III of these bylaws, Section 3.5 (place of meetings), Section 3.6 (regular meetings), Section 3.7 (special meetings and notice), Section 3.8 (quorum), Section 3.9 (waiver of notice), Section 3.10 (adjournment), Section 3.11 (notice of adjournment), and Section 3.12 (action without meeting), with such changes in the context of those bylaws as are necessary to substitute the committee and its members for the board of directors and its members; provided, however, that the time of regular meetings of committees may be determined either by resolution of the board of directors or by resolution of the committee, that special meetings of committees may also be called by resolution of the board of directors, and that notice of special meetings of committees shall also be given to all alternate members, who shall have the right to attend all meetings of the committee. The board of directors may adopt rules for the government of any committee not inconsistent with the provisions of these bylaws.

ARTICLE V

OFFICERS

5.1 OFFICERS.

The officers of the corporation shall be a president, a secretary, and a chief financial officer. The corporation may also have, at the discretion of the board of directors, a chairman of the board, one or more vice presidents, one or more assistant secretaries, one or more assistant treasurers, and such other officers as

may be appointed in accordance with the provisions of Section 5.3 of these bylaws. Any number of offices may be held by the same person.

5.2 ELECTION OF OFFICERS

The officers of the corporation, except such officers as may be appointed in accordance with the provisions of Section 5.3 or Section 5.5 of these bylaws, shall be chosen by the board, subject to the rights, if any, of an officer under any contract of employment.

5.3 SUBORDINATE OFFICERS

The board of directors may appoint, or may empower the president to appoint, such other officers as the business of the corporation may require, each of whom shall hold office for such period, have such authority, and perform such duties as are provided in these bylaws or as the board of directors may from time to time determine.

5.4 REMOVAL AND RESIGNATION OF OFFICERS .

Subject to the rights, if any, of an officer under any contract of employment, any officer may be removed, either with or without cause, by the board of directors at any regular or special meeting of the board or, except in case of an officer chosen by the board of directors, by any officer upon whom such power of removal may be conferred by the board of directors.

Any officer may resign at any time by giving written notice to the corporation. Any resignation shall take effect at the date of the receipt of that notice or at any later time specified in that notice; and, unless otherwise specified in that notice, the acceptance of the resignation shall not be necessary to make it effective. Any resignation is without prejudice to the rights, if any, of the corporation under any contract to which the officer is a party.

5.5 VACANCIES IN OFFICES .

A vacancy in any office because of death, resignation, removal, disqualification or any other cause shall be filled in the manner prescribed in these bylaws for regular appointments to that office.

5.6 CHAIRMAN OF THE BOARD .

The chairman of the board, if such an officer be elected, shall, if present, preside at meetings of the board of directors and exercise and perform such other powers and duties as may from time to time be assigned to him by the board of directors or as may be prescribed by these bylaws. If there is no chief executive officer, then the chairman of the board shall also be the chief executive officer of the corporation and shall have the powers and duties prescribed in Section 5.7 of these bylaws.

5.7 CHIEF EXECUTIVE OFFICER.

Subject to such supervisory powers, if any, as may be given by the board of directors to the chairman of the board, if there be such an officer, the chief executive officer shall be subject to the control of the board of directors and have general supervision, direction and control of the business. He or she shall preside at all meetings of the stockholders and, in the absence or non-existence of the chairman of the board, at all meetings of the board of directors. He or she shall have the general powers and duties of management usually vested in the office of the chief executive officer of a corporation, and shall have such other powers and perform such other duties as may be prescribed by the board of directors or these bylaws.

5.8 PRESIDENT

In the absence or disability of the chief executive officer, and if there is no chairman of the board, the president shall perform all the duties of the chief executive officer and when so acting shall have the power of, and be subject to all the restrictions upon, the chief executive officer. The president shall have such other powers and perform such other duties as from time to time may be prescribed for them respectively by the board of directors, these bylaws, the chief executive officer or the chairman of the board.

5.9 VICE PRESIDENTS

In the absence or disability of the president, the vice presidents, if any, in order of their rank as fixed by the board of directors or, if not ranked, a vice president designated by the board of directors, shall perform all the duties of the president and when so acting shall have all the powers of, and be subject to all the restrictions upon, the president. The vice presidents shall have such other powers and perform such other duties as from time to time may be prescribed for them respectively by the board of directors, these bylaws, the president or the chairman of the board.

5.10 SECRETARY

The secretary shall keep or cause to be kept, at the principal executive office of the corporation or such other place as the board of directors may direct, a book of minutes of all meetings and actions of directors, committees of directors and shareholders. The minutes shall show the time and place of each meeting, whether regular or special (and, if special, how authorized and the notice given), the names of those present at directors' meetings or committee meetings, the number of shares present or represented at shareholders' meetings, and the proceedings thereof.

The secretary shall keep, or cause to be kept, at the principal executive office of the corporation or at the office of the corporation's transfer agent or registrar, as determined by resolution of the board of directors, a share register, or a duplicate share register, showing the names of all shareholders and their addresses, the number and classes of shares held by each, the number and date of certificates evidencing such shares, and the number and date of cancellation of every certificate surrendered for cancellation.

The secretary shall give, or cause to be given, notice of all meetings of the shareholders and of the board of directors required to be given by law or by these bylaws. He shall keep the seal of the corporation, if one be adopted, in safe custody and shall have such other powers and perform such other duties as may be prescribed by the board of directors or by these bylaws.

5.11 CHIEF FINANCIAL OFFICER

The chief financial officer shall keep and maintain, or cause to be kept and maintained, adequate and correct books and records of accounts of the properties and business transactions of the corporation, including accounts of its assets, liabilities, receipts, disbursements, gains, losses, capital, retained earnings, and shares. The books of account shall at all reasonable times be open to inspection by any director.

The chief financial officer shall deposit all money and other valuables in the name and to the credit of the corporation with such depositaries as may be designated by the board of directors. He shall disburse the funds of the corporation as may be ordered by the board of directors, shall render to the president and directors, whenever they request it, an account of all of his transactions as chief financial officer and of the financial condition of the corporation, and shall have such other powers and perform such other duties as may be prescribed by the board of directors or these bylaws.

ARTICLE VI

INDEMNIFICATION OF DIRECTORS,

OFFICERS, EMPLOYEES, AND OTHER AGENTS

6.1 INDEMNIFICATION OF DIRECTORS AND OFFICERS.

The corporation shall, to the maximum extent and in the manner permitted by the General Corporation Law of Delaware, indemnify each of its directors and officers against expenses (including attorneys' fees), judgments, fines, settlements and other amounts actually and reasonably incurred in connection with any proceeding, arising by reason of the fact that such person is or was an agent of the corporation. For purposes of this Section 6.1, a "director" or "officer" of the corporation includes any person (i) who is or was a director or officer of the corporation, partnership, joint venture, trust or other enterprise, or (iii) who was a director or officer of a corporation which was a predecessor corporation or of another enterprise at the request of such predecessor corporation.

6.2 INDEMNIFICATION OF OTHERS

The corporation shall have the power, to the maximum extent and in the manner permitted by the General Corporation Law of Delaware, to indemnify each of its employees and agents (other than directors and officers) against expenses (including attorneys' fees), judgments, fines, settlements and other amounts actually and reasonably incurred in connection with any proceeding, arising by reason of the fact that such person is or was an agent of the corporation. For purposes of this Section 6.2, an "employee" or "agent" of the corporation (other than a director or officer) includes any person (i) who is or was an employee or agent of the corporation, (ii) who is or was serving at the request of the corporation and the request of such predecessor corporation or of another enterprise at the request of such predecessor corporation.

6.3 INSURANCE

The corporation may purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise against any liability asserted against him or her and incurred by him or her in any such capacity, or arising out of his or her status as such, whether or not the corporation would have the power to indemnify him or her against such liability under the provisions of the General Corporation Law of Delaware.

ARTICLE VII

RECORDS AND REPORTS

7.1 MAINTENANCE AND INSPECTION OF RECORDS

The corporation shall, either at its principal executive office or at such place or places as designated by the board of directors, keep a record of its stockholders listing their names and addresses and the number and class of shares held by each stockholder, a copy of these bylaws as amended to date, accounting books and other records.

Any stockholder of record, in person or by attorney or other agent, shall, upon written demand under oath stating the purpose thereof, have the right during the usual hours for business to inspect for any proper purpose the corporation's stock ledger, a list of its stockholders, and its other books and records and to make copies or extracts therefrom. A proper purpose shall mean a purpose reasonably related to such person's interest as a stockholder. In every instance where an attorney or other agent is the person who seeks the right to inspection, the demand under oath shall be accompanied by a power of attorney or such other writing that authorizes the attorney or other agent to so act on behalf of the stockholder. The demand under oath shall be directed to the corporation at its registered office in Delaware or at its principal place of business.

The officer who has charge of the stock ledger of a corporation shall prepare and make, at least ten (10) days before every meeting of stockholders, a complete list of the stockholders entitled to vote at the meeting, arranged in alphabetical order, and showing the address of each stockholder and the number of shares registered in the name of each stockholder. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting, during ordinary business hours, for a period of at least ten (10) days prior to the meeting, either at a place within the city where the meeting is to be held, which place shall be specified in the notice of the meeting, or, if not so specified, at the place where the meeting is to be held. The list shall also be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present.

7.2 INSPECTION BY DIRECTORS

Any director shall have the right to examine the corporation's stock ledger, a list of its stockholders and its other books and records for a purpose reasonably related to his or her position as a director. The Court of Chancery is hereby vested with the exclusive jurisdiction to determine whether a director is entitled to the inspection sought. The Court may summarily order the corporation to permit the director to inspect any and all books and records, the stock ledger, and the stock list and to make copies or extracts therefrom. The Court may, in its discretion, prescribe any limitations or conditions with reference to the inspection, or award such other and further relief as the Court may deem just and proper.

7.3 ANNUAL STATEMENT TO STOCKHOLDERS

The board of directors shall present at each annual meeting, and at any special meeting of the stockholders when called for by vote of the stockholders, a full and clear statement of the business and condition of the corporation.

7.4 REPRESENTATION OF SHARES OF OTHER CORPORATIONS.

The chairman of the board, the president, any vice president, the chief financial officer, the secretary or assistant secretary of this corporation, or any other person authorized by the board of directors or the president or a vice president, is authorized to vote, represent, and exercise on behalf of this corporation all rights incident to any and all shares of any other corporation or corporations standing in the name of this corporation. The authority herein granted may be exercised either by such person directly or by any other person authorized to do so by proxy or power of attorney duly executed by such person having the authority.

ARTICLE VIII

GENERAL MATTERS

8.1 RECORD DATE FOR PURPOSES OTHER THAN NOTICE AND VOTING.

For purposes of determining the stockholders entitled to receive payment of any dividend or other distribution or allotment of any rights or the stockholders entitled to exercise any rights in respect of any other

lawful action (other than action by stockholders by written consent without a meeting), the board of directors may fix, in advance, a record date, which shall not be more than sixty (60) days before any such action. In that case, only stockholders of record at the close of business on the date so fixed are entitled to receive the dividend, distribution or allotment of rights, or to exercise such rights, as the case may be, notwithstanding any transfer of any shares on the books of the corporation after the record date so fixed, except as otherwise provided by law.

If the board of directors does not so fix a record date, then the record date for determining stockholders for any such purpose shall be at the close of business on the day on which the board adopts the applicable resolution or the sixtieth (60th) day before the date of that action, whichever is later.

8.2 CHECKS; DRAFTS; EVIDENCES OF INDEBTEDNESS

From time to time, the board of directors shall determine by resolution which person or persons may sign or endorse all checks, drafts, other orders for payment of money, notes or other evidences of indebtedness that are issued in the name of or payable to the corporation, and only the persons so authorized shall sign or endorse those instruments.

8.3 CORPORATE CONTRACTS AND INSTRUMENTS: HOW EXECUTED.

The board of directors, except as otherwise provided in these bylaws, may authorize any officer or officers, or agent or agents, to enter into any contract or execute any instrument in the name of and on behalf of the corporation; such authority may be general or confined to specific instances. Unless so authorized or ratified by the board of directors or within the agency power of an officer, no officer, agent or employee shall have any power or authority to bind the corporation by any contract or engagement or to pledge its credit or to render it liable for any purpose or for any amount.

8.4 STOCK CERTIFICATES; PARTLY PAID SHARES .

The shares of a corporation shall be represented by certificates, provided that the board of directors of the corporation may provide by resolution or resolutions that some or all of any or all classes or series of its stock shall be uncertificated shares. Any such resolution shall not apply to shares represented by a certificate until such certificate is surrendered to the corporation. Notwithstanding the adoption of such a resolution by the board of directors, every holder of stock represented by certificates and upon request every holder of uncertificated shares shall be entitled to have a certificate signed by, or in the name of the corporation by, the chairman or vice-chairman of the board of directors, or the president or vice-president, and by the chief financial officer, the secretary or an assistant secretary of such corporation representing the number of shares registered in certificate form. Any or all of the signatures on the certificate may be a facsimile. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate has ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the corporation with the same effect as if he or she were such officer, transfer agent or registrar at the date of issue.

The corporation may issue the whole or any part of its shares as partly paid and subject to call for the remainder of the consideration to be paid therefor. Upon the face or back of each stock certificate issued to represent any such partly paid shares, upon the books and records of the corporation in the case of uncertificated partly paid shares, the total amount of the consideration to be paid therefor and the amount paid thereon shall be stated. Upon the declaration of any dividend on fully paid shares, the corporation shall declare a dividend upon partly paid shares of the same class, but only upon the basis of the percentage of the consideration actually paid thereon.

8.5 SPECIAL DESIGNATION ON CERTIFICATES .

If the corporation is authorized to issue more than one class of stock or more than one series of any class, then the powers, the designations, the preferences, and the relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights shall be set forth in full or summarized on the face or back of the certificate that the corporation shall issue to represent such class or series of stock; provided, however, that, except as otherwise provided in Section 202 of the General Corporation Law of Delaware, in lieu of the foregoing requirements there may be set forth on the face or back of the certificate that the corporation shall issue to represent such class or series of stock a statement that the corporation will furnish without charge to each stockholder who so requests the powers, the designations, the preferences, and the relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights.

8.6 LOST CERTIFICATES

Except as provided in this Section 8.6, no new certificates for shares shall be issued to replace a previously issued certificate unless the latter is surrendered to the corporation and canceled at the same time. The board of directors may, in case any share certificate or certificate for any other security is lost, stolen or destroyed, authorize the issuance of replacement certificates on such terms and conditions as the board may require; the board may require indemnification of the corporation secured by a bond or other adequate security sufficient to protect the corporation against any claim that may be made against it, including any expense or liability, on account of the alleged loss, theft or destruction of the certificate or the issuance of the replacement certificate.

8.7 CONSTRUCTION; DEFINITIONS

Unless the context requires otherwise, the general provisions, rules of construction, and definitions in the General Corporation Law of Delaware shall govern the construction of these bylaws. Without limiting the generality of this provision, the singular number includes the plural number includes the singular, and the term "person" includes both a corporation and a natural person.

ARTICLE IX

AMENDMENTS

The original or other bylaws of the corporation may be adopted, amended or repealed by the stockholders entitled to vote; provided, however, that the corporation may, in its Certificate of Incorporation, confer the power to adopt, amend or repeal bylaws upon the directors. The fact that such power has been so conferred upon the directors shall not divest the stockholders of the power, nor limit their power to adopt, amend or repeal bylaws.

ARTICLE X

DISSOLUTION

If it should be deemed advisable in the judgment of the board of directors of the corporation that the corporation should be dissolved, the board, after the adoption of a resolution to that effect by a majority of the whole board at any meeting called for that purpose, shall cause notice to be mailed to each stockholder entitled to vote thereon of the adoption of the resolution and of a meeting of stockholders to take action upon the resolution.

At the meeting a vote shall be taken for and against the proposed dissolution. If a majority of the outstanding stock of the corporation entitled to vote thereon votes for the proposed dissolution, then a certificate stating that the dissolution has been authorized in accordance with the provisions of Section 275 of the General Corporation Law of Delaware and setting forth the names and residences of the directors and officers shall be executed, acknowledged, and filed and shall become effective in accordance with Section 103 of the General Corporation Law of Delaware. Upon such certificate's becoming effective in accordance with Section 103 of the General Corporation Law of Delaware, the corporation shall be dissolved.

Whenever all the stockholders entitled to vote on a dissolution consent in writing, either in person or by duly authorized attorney, to a dissolution, no meeting of directors or stockholders shall be necessary. The consent shall be filed and shall become effective in accordance with Section 103 of the General Corporation Law of Delaware. Upon such consent's becoming effective in accordance with Section 103 of the General Corporation Law of Delaware, the corporation shall be dissolved. If the consent is signed by an attorney, then the original power of attorney or a photocopy thereof shall be attached to and filed with the consent. The consent filed with the Secretary of State shall have attached to it the affidavit of the secretary or some other officer of the corporation stating that the consent has been signed by or on behalf of all the stockholders entitled to vote on a dissolution; in addition, there shall be attached to the consent a certification by the secretary or some other officer of the corporation setting forth the names and residences of the directors and officers of the corporation.

ARTICLE XI CUSTODIAN

11.1 APPOINTMENT OF A CUSTODIAN IN CERTAIN CASES.

The Court of Chancery, upon application of any stockholder, may appoint one or more persons to be custodians and, if the corporation is insolvent, to be receivers, of and for the corporation when:

- (a) at any meeting held for the election of directors the stockholders are so divided that they have failed to elect successors to directors whose terms have expired or would have expired upon qualification of their successors; or
- (b) the business of the corporation is suffering or is threatened with irreparable injury because the directors are so divided respecting the management of the affairs of the corporation that the required vote for action by the board of directors cannot be obtained and the stockholders are unable to terminate this division; or
 - (c) the corporation has abandoned its business and has failed within a reasonable time to take steps to dissolve, liquidate or distribute its assets.

11.2 DUTIES OF CUSTODIAN .

The custodian shall have all the powers and title of a receiver appointed under Section 291 of the General Corporation Law of Delaware, but the authority of the custodian shall be to continue the business of the corporation and not to liquidate its affairs and distribute its assets, except when the Court of Chancery otherwise orders and except in cases arising under Sections 226(a)(3) or 352(a)(2) of the General Corporation Law of Delaware

CERTIFICATE OF ADOPTION OF BYLAWS

OF

COMSCORE NETWORKS, INC.

Adoption by Incorporator

The undersigned person appointed in the Certificate of Incorporation to act as the Incorporator of comScore Networks, Inc. hereby adopts the foregoing Bylaws, comprising sixteen (16) pages, as the Bylaws of the corporation.

Executed this 18th day of August, 1999.

/s/ Leonardo Murphy

Leonardo Murphy, Incorporator

Certificate by Secretary of Adoption by Incorporator

The undersigned hereby certifies that he is the duly elected Secretary of comScore Networks, Inc. and that the foregoing Bylaws, comprising sixteen (16) pages, were adopted as the Bylaws of the corporation on August 18, 1999, by the person appointed in the Certificate of Incorporation to act as the Incorporator of the corporation.

IN WITNESS WHEREOF, the undersigned has hereunto set his hand and affixed the corporate seal this 18th day of August, 1999.

/s/ Jeffrey D. Saper Jeffrey D. Saper,

Secretary

FOURTH AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT

THIS FOURTH AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT (this "Agreement") is made as of August 1, 2003, by and among comScore Networks. Inc., a Delaware corporation (the "Company"), the stockholders of the Company listed on the signature pages hereto (the "Founders"). Capitalized terms used but not defined herein shall have the meanings ascribed thereto in the Purchase Agreement (defined below),

RECITALS

WHEREAS, certain of the Purchasers (the "Prior Purchasers") hold shares of the Company's Series A Preferred Stock ("Series A Preferred"), Series B Preferred Stock ("Series B Preferred"), Series C Preferred Stock ("Series C Preferred") and Series D Preferred Stock ("Series D Preferred");

WHEREAS, certain of the Founders (the "Series C-1 Purchasers") hold shares of the Company's Series C-1 Preferred Stock ("Series C-1 Preferred");

WHEREAS, the Company, the Founders and the Prior Purchasers have previously entered into a Third Amended and Restated Investor Rights Agreement dated as of June 6, 2002, providing certain registration and other rights to the Prior Purchasers and certain of the Founders in their capacity as Series C-1 Purchasers (the "Prior Agreement");

WHEREAS, the Company and certain of the Purchasers (the "Series E Purchasers") are entering into that certain Series E Preferred Stock Purchase Agreement of even date herewith (the "Purchase Agreement") pursuant to which, among other things, the Series E Purchasers have agreed to purchase up to 24,005,548 shares of the Company's Series E Preferred Stock (the "Series E Preferred") at a purchase price of \$0.50 per share (the "Series E Price");

WHEREAS, in order to induce the Series E Purchasers to enter into the Purchaser Agreement and to contribute funds to the Company pursuant thereto, the Company, the Founders and the Prior Purchasers desire that the Company (and the Prior Purchasers, as applicable) grant to the Series E Purchasers certain registration rights and rights of participation, first refusal, co-sale, first offer, voting, information and other rights as set forth herein, and further desire to conform the existing registration and other rights and provisions applicable to the Prior Purchasers and certain of the Founders in their capacity as Series C-1 Purchasers to such rights and provisions as applied to the Series E Purchasers as set forth herein; and

WHEREAS, the Company, the Founders, the Series E Purchasers, and the Prior Purchasers holding not less than the minimum number of shares required to amend the Prior Agreement desire that this Agreement shall govern the registration and other rights of all Purchasers and shall supersede and replace the Prior Agreement in its entirety;

NOW, THEREFORE, in consideration of the promises and mutual covenants contained herein, the parties hereto hereby agree as follows:

- 1. Registration Rights. The Company covenants and agrees as follows:
 - (a) Definitions, For purposes of this Section 1:
 - (i) The term "1934 Act" means the Securities Exchange Act of 1934, as amended.
 - (ii) An "Affiliate" of a specified person means a person that directly, or indirectly through one or more intermediaries, controls or is controlled by, or is under common control with, the person specified.
 - (iii) The term "Act" means the Securities Act of 1933, as amended.
 - (iv) The term "Closing" has the meaning ascribed to it in the Purchase Agreement.
- (v) The term "Form S-3" means such form under the Act as in effect on the date hereof or any registration form under the Act subsequently adopted by the SEC which permits inclusion or incorporation of substantial information by reference to other documents filed by the Company with the SEC.
 - (vi) The term "Holder" means any person owning or having the right to acquire Registrable Securities or any assignee thereof in accordance with Section 1(k) hereof.
- (vii) The terms "register," "registered" and "registration" refer to a registration effected by preparing and filing a registration statement or similar document in compliance with the Act, and the declaration or ordering of effectiveness of such registration statement or document by the SEC.
- (viii) The term "Registrable Securities" means (i) Common Stock issuable or issued upon conversion of the Series A Preferred, the Series B Preferred, the Series C Preferred, the Series C-l Preferred, the Series D Preferred and the Series E Preferred, and (ii) any Common Stock of the Company issued as (or issuable upon the conversion or exercise of any warrant, right or other security which is issued as) a dividend or other distribution with respect to, or in exchange for or in replacement of the shares referenced in (i) above, excluding in all cases, however, (x) any Registrable Securities sold by a person in a transaction in which his rights under this Section 1 are not duly assigned as provided herein, (y) any Registrable Securities after such securities have been sold to the public or sold pursuant to Rule 144 promulgated under the Act, and (z) Registrable Securities held by any Holder of less than 1% of the aggregate outstanding Common Stock (on a fully-diluted, as-converted basis), but only so long as such securities are freely transferable by such Holder pursuant to SEC Rule 144(k) without restriction as to volume or otherwise. For purposes hereof, a person shall be deemed to be a holder of Registrable Securities whenever such person has the right to acquire such Registrable Securities (upon conversion, exchange or otherwise, but

disregarding any limitations or restrictions upon the exercise of such right) whether or not such acquisition has actually been effected.

- (ix) The number of shares of "Registrable Securities then outstanding" shall be determined by the number of shares of Common Stock which are Registrable Securities and (1) are then issued and outstanding or (2) are then issuable pursuant to the exercise or conversion of then outstanding and then exercisable options, warrants, Preferred Stock or other convertible securities
 - (x) The term "SEC" shall mean the Securities and Exchange Commission.
 - (b) Request for Registration
- (i) Registration Rights. If the Company shall receive at any time after the earlier of (i) three (3) years following the Closing, or (ii) six (6) months after the effective date of the first registration statement for a public offering of securities of the Company (other than a registration statement relating either to the sale of securities to employees of the Company pursuant to a stock option, stock purchase or similar plan or an SEC Rule 145 transaction), a written request from Holders of at least a majority of the Registrable Securities ("Initiating Holders") requesting that the Company file a registration statement under the Act covering the registration of a portion of the Registrable Securities than outstanding, then, provided that, such request must demand registration of the lesser of (x) 3,000,000 shares of Registrable Securities and (y) such number of shares of Registrable Securities anticipated to have, an aggregate offering price to the public (before deducting any underwriter discounts, commission or concessions) of at least \$10,000,000, the Company shall:
 - (1) within fifteen (15) days of the receipt thereof, give written notice of such request to all Holders; and
- (2) effect as soon as practicable, and in any event within ninety (90) days of the receipt of such request, the registration under the Act of all Registrable Securities which the Holders request to be registered, subject to the limitations of subsection 1(b)(ii), within fifteen (15) days of the mailing of such notice by the Company in accordance with Section 7(e).
- (ii) <u>Underwriting Requirements</u>. If the Initiating Holders intend to distribute the Registrable Securities covered by their request by means of an underwriting, they shall so advise the Company as a part of their request made pursuant to subsection 1(b)(i) and the Company shall include such information in the written notice referred to in subsection 1(b)(i). The underwriter will be selected by a majority in interest of the Initiating Holders, and shall be reasonably acceptable to the Company. In such event, the right of any Holder or other holder of securities of the Company to include securities in such registration shall be conditioned upon such Holder's or Holders' participation in such underwriting and the inclusion of such Holder's or Holders' securities in the underwriting (unless otherwise mutually agreed by a majority in interest of the Initiating Holders and such Holder or holder) to the extent provided herein. All Holders and other holders of securities of the Company proposing to distribute their securities through such underwriting shall (together with the Company as provided in subsection 1(e)(v)) enter into an

underwriting agreement in customary form with the underwriter or underwriters selected for such underwriting. Notwithstanding any other provision of this Section 1(b), if the underwriter advises the Initiating Holders in writing that marketing factors require a limitation of the number of shares to be underwritten, then the Initiating Holders shall so advise all Holders of Registrable Securities and other holders of registration rights which would otherwrite be underwritten pursuant hereto, and the number of securities that may be included in the underwriting on behalf of each Holder or other holder shall be allocated on a pro-rata basis among the selling stockholders and entitled to inclusion therein on the basis of a registration rights agreement with the Company; provided. however, that the number of shares of Registrable Securities to be included in such underwriting shall not be reduced unless all other securities are first entirely excluded from the underwriting and registration. For purposes of allocation of securities to be included in any offering, for any selling stockholder which is a partnership, corporation or other entity, the Affiliates, partners (or members) or retired partners (or retired members) of such holder, or the estates and family members of any persons and any trusts for the benefit of any of the foregoing persons shall be deemed to be a single "selling stockholder," and any pro-rata reduction with respect to such "selling stockholder," as defined in this sentence. To facilitate the allocation of shares in accordance with the above provisions, the Company may round the number of shares allocated to any Holder to the nearest 100 shares.

- (iii) Notwithstanding the foregoing, if the Company shall furnish to Holders requesting a registration statement pursuant to this Section 1(b), a certificate signed by the Chief Executive Officer of the Company stating that in the good faith judgment of the Board of Directors of the Company, it would be seriously detrimental to the Company and its stockholders for such registration statement to be filed and it is therefore essential to defer the filing of such registration statement, the Company shall have the right to defer taking action with respect to such filing; provided, however, that the Company may not utilize this right more than twice in any twelve-month period nor may the Company delay registration statements more than an aggregate of 120 days following receipt of requests from Initiating Holders in any twelve-month period.
 - (iv) In addition, the Company shall not be obligated to effect, or to take any action to effect, any registration pursuant to this Section 1(b):
 - (1) After the Company has effected two (2) registrations pursuant to this Section 1(b) and such registrations have been declared or ordered effective;
- (2) During the period starting with the date sixty (60) days prior to the Company's good faith estimate of the date of filing of, and ending on a date one hundred eighty (180) days after the effective date of, (x) the Company's initial registered offering of its securities to the general public (other than a registration statement relating to either the sale of securities to employees of the Company pursuant to a stock option, stock purchase or similar plan or an SEC Rule 145 transaction), (y) a previous registration subject to Section 1(b) hereof or (z) a previous registration subject to Section 1(c) hereof; provided that, the Company is actively employing in good faith reasonable efforts to cause such registration statement to become effective; or

(3) If the Initialing Holders propose to dispose of shares of Registrable Securities that may be immediately registered on Form S-3 (or any successor form) pursuant to a request made pursuant to Section 1(d) below (in which case, such Registrable Securities shall be registered pursuant to Section 1(d) below), unless the registration in question is to be underwritten and the managing underwriter is of the opinion that marketing factors would make a registration on Form S-1 advisable.

(c) Company Registration.

- (i) Registration Rights. If (but without any obligation to do so) the Company proposes to register (including for this purpose a registration effected by the Company for stockholders other than the Holders) any of its stock or other securities under the Act in connection with the public offering of such securities solely for cash (other than a registration relating solely to the sale of securities to participants in a Company stock plan, a registration pursuant to a Rule 145 transaction, a registration on any form which does not include substantially the same information as would be required to be included in a registration statement covering the sale of the Registrable Securities or a registration in which the only Common Stock being registered is Common Stock issuable upon conversion of debt securities which are also being registered), the Company shall, at such time, promptly give each Holder written notice of such registration. Upon the written request of each Holder given within fifteen (15) days after mailing of such notice by the Company in accordance with Section 7(e), the Company shall, subject to the provisions of paragraph (ii) below, cause to be registered under the Act all of the Registrable Securities that each such Holder has requested to be registered.
- (ii) <u>Underwriting Requirements</u>. In connection with any offering involving an underwriting of shares of the Company's capital stock, the Company shall not be required under this Section 1(c) to include any of the Holders' securities in such underwriting unless they accept the terms of the underwriting as agreed upon between the Company and the underwriters selected by it (or by other persons entitled to select the underwriters), and then only in such quantity as the underwriters determine in their sole discretion will not jeopardize the success of the offering by the Company. If the total amount of securities, including Registrable Securities, requested by stockholders to be included in such offering exceeds the amount of securities sold other than by the Company that the underwriters determine in their sole discretion is compatible with the success of the offering, then the Company shall be required to include in the offering only that number of such securities, including Registrable Securities, which the underwriters determine in their sole discretion will not jeopardize the success of the offering, but in no event shall the amount of Registrable Securities of the selling Holders included in the offering be reduced below ten percent (10%) of the total amount of securities included in such offering, unless such offering is the initial public offering of the Company's securities, in which case the Holders may be excluded entirely if the underwriters make the determination described above and no other stockholder's securities are included. Allocation of securities to be sold in any such offering shall be made on a pro-rata basis among the selling stockholders according to the total number of Registrable Securities held by each such selling stockholder and entitled to inclusion therein on the basis of a registration rights agreement with the Company; provided, however, that the number of shares of Registrable Securities to be included in such offering shall not be reduced unless all other securities (other than

included in any offering, for any selling stockholder which is a partnership, corporation or other entity, the Affiliates, partners (or members) or retired partners (or retired members) of such holder, or the estates and family members of any persons and any trusts for the benefit of any of the foregoing persons shall be deemed to be a single "selling stockholder," and any pro-rata reduction with respect to such "selling stockholder" shall be based upon the aggregate amount of shares carrying registration rights owned by all entities and individuals included in such "selling stockholder," as defined in this sentence.

- (d) <u>Form S-3 Registration</u>. In case the Company shall receive from Holders of the lesser of (x) at least 3,000,000 shares of Registrable Securities and (y) such number of shares of Registrable Securities and in the company effect a registration on Form S-3 and any related qualification or compliance with respect to all or a part of the Registrable Securities owned by such Holder or Holders, the Company will:
 - (i) promptly give written notice of the proposed registration, and any related qualification or compliance, to all other Holders; and
- (ii) as soon as practicable, effect such registration and all such qualifications and compliances as may be so requested and as would permit or facilitate the sale and distribution of all or such portion of such Holder's or Holders' Registrable Securities as are specified in such request, together with all or such portion of the Registrable Securities of any other Holder or Holders joining in such request as are specified in a written request given within fifteen (15) days after receipt of such written notice from the Company; provided, however, that the Company shall not be obligated to effect any such registration, qualification or compliance, pursuant to this Section 1(d): (1) if Form S-3 is not available for such offering by the Holders; (2) if the Company shall furnish to the Holders a certificate signed by the Chief Executive Officer of the Company stating that in the good faith judgment of the Board of Directors of the Company, it would be seriously detrimental to the Company and its stockholders for such Form S-3 Registration to be effected at such time, in which event the Company shall have the right to defer the filing of the Form S-3 registration statement; provided, however, that the Company shall not utilize this right more than twice in any twelve-month period, nor may the Company delay the filing of Form S-3 registration statements more than an aggregate of 120 days following receipt of the request of the Holder or Holders under this Section 1(d) during any twelve-month period; or (3) if the Company has, within the twelve-month period such request, already effected two (2) or more registrations on Form S-3 for the Holders pursuant to this Section 1(d).
- (iii) Subject to the foregoing, the Company shall file a registration statement covering the Registrable Securities and other securities so requested to be registered as soon as practicable after receipt of the request or requests of the Holders. Registrations effected pursuant to this Section 1(d) shall not be counted as demands for registration or registrations effected pursuant to Sections 1(b) or 1(c), respectively.

- (e) Obligations of the Company. Whenever required under this Section 1 to effect the registration of any Registrable Securities, the Company shall, as expeditiously as reasonably possible:
- (i) Prepare and file with the SEC (and afford one counsel for the selling Holders of Registrable Securities an opportunity to review) a registration statement with respect to such Registrable Securities and use its best efforts to cause such registration statement to become effective, and, upon the request of the Holders of a majority of the Registrable Securities registered thereunder, keep such registration statement effective for a period of up to 120 days or until the distribution contemplated in the Registration Statement has been completed; provided, however, that such 120-day period shall be extended for a period of time equal to the period the Holder refrains from selling any securities included in such registration at the request of an underwriter of Common Stock (or other securities) of the Company.
- (ii) Prepare and file with the SEC (and afford one counsel for the selling Holders of Registrable Securities an opportunity to review) such amendments and supplements to such registration statement and the prospectus used in connection with such registration statement as maybe necessary to comply with the provisions of the Act with respect to the disposition of all securities covered by such registration statement.
- (iii) Furnish to the Holders such numbers of copies of a prospectus, including a preliminary prospectus, in conformity with the requirements of the Act, and such other documents as they may reasonably request in order to facilitate the disposition of Registrable Securities owned by them.
- (iv) Use its best efforts to register and qualify the securities covered by such registration statement under such other securities or blue sky laws of such jurisdictions as shall be reasonably requested by the Holders; provided that the Company shall not be required in connection therewith or as a condition thereto to qualify to do business or to file a general consent to service of process in any such states or jurisdictions.
- (v) In the event of any underwritten public offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing underwriter of such offering. Each Holder participating in such underwriting shall also enter into and perform its obligations under such an agreement.
- (vi) Notify each Holder of Registrable Securities covered by such registration statement at any time when a prospectus relating thereto is required to be delivered under the Act of the happening of any event as a result of which the prospectus included in such registration statement, as then in effect, includes an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary to make the statements therein not misleading in the light of the circumstances then existing.
 - (vii) Cause all such Registrable Securities registered pursuant hereunder to be listed on each securities exchange on which similar securities issued by the Company are then listed.

- (viii) Provide a transfer agent and registrar for all Registrable Securities registered pursuant hereunder and a CUSIP number for all such Registrable Securities, in each case not later than the effective date of such registration.
- (ix) Make available for reasonable inspection by each seller, any underwriter participating in any distribution pursuant to such registration statement, and any attorney, accountant or other agent retained by such seller or underwriter, all financial and other records, pertinent corporate documents and properties of the Company, and cause the Company's officers, directors and employees to supply all information reasonably requested by any such seller, underwriter, attorney, accountant or agent in connection with such registration statement.
- (x) Use its best efforts (if the offering is underwritten and at the request of any seller of Registrable Securities), to furnish, at the request of any seller, on the date that Registrable Securities are delivered to the underwriters for sale pursuant to such registration, (1) an opinion of counsel representing the Company for the purposes of registration, addressed to the underwriters, stating that such registration statement has become effective under the Act and that (A) to the best knowledge of such counsel, no stop order suspending the effectiveness thereof has been issued and no proceedings for that purpose have been instituted or are pending or contemplated under the Act, (B) the registration statement, the related prospectus, and each amendment or supplement thereof, comply as to form in all material respects with the Act and the applicable rules and regulations of the SEC thereunder (except that such counsel need express no opinion as to financial statements, the notes thereto, the financial schedules and other financial, numerical, accounting and statistical data contained therein, other than opinions as to the capitalization of the Company) and (C) to such other effects as may reasonably be requested by counsel for the underwriters, and (2) a letter dated such date from the independent public accountants retained by the Company, addressed to the underwriters, stating that they are independent public accountants within the meaning of the Act and that, in the opinion of such accountants, the financial statements of the Company included in the registration statement or the prospectus, or any amendment or supplement thereof, comply as to form in all material respects with the applicable accounting requirements of the Act, and such letter shall additionally cover such other financial matters (including information as to the period ending no more than five business days prior to the date of such letter) with respect to the registration in respect of which such letter is being given, as such underwriters m
- (xi) In the event of any underwritten public offering, cooperate with the Holder requesting registration pursuant to this Section, the underwriters participating in the offering and their counsel in any due diligence investigation reasonably requested by the Holders or the underwriters in connection therewith, and participate, to the extent reasonably requested by the managing underwriter for the offering or the Holders, in efforts to sell the Registrable Securities under the offering (including, without limitation, participating in "roadshow" meetings with prospective investors) that would be customary for underwritten primary offerings of a comparable amount of equity securities by the Company.

(f) Furnish Information

- (i) It shall be a condition precedent to the obligations of the Company to take any action pursuant to this Section 1 with respect to the Registrable Securities of any selling Holder that such Holder shall furnish to the Company such information regarding itself, the Registrable Securities held by it, and the intended method of disposition of such securities as shall be reasonably required to effect the registration of such Holder's Registrable Securities.
- (ii) The Company shall have no obligation with respect to any registration requested pursuant to Section 1(b) or Section 1(d) if, due to the operation of subsection 1(f)(i), the number of shares or the anticipated aggregate offering price of the Registrable Securities to be included in the registration does not equal or exceed the number of shares or the anticipated aggregate offering price required to originally trigger the Company's obligation to initiate such registration as specified in subsection 1(b)(i) or Section 1(d), whichever is applicable.
- (g) Expenses of Demand. Company or S-3 Registration. All expenses (exclusive of underwriting discounts and commissions and stock transfer taxes) incurred in connection with registrations, filings or qualifications pursuant to this Section 1 including (without limitation) all registration, filing and qualification fees, printers' and accounting fees, fees and disbursements of counsel for the Company and the reasonable fees and disbursements of one counsel for all selling Holders of Registrable Securities shall be borne by the Company; provided, however, that the Company shall not be required to pay any expenses of a registration under Section 1(b) if such registration has been withdrawn at the request of the Holders and such registration has not been delayed by the Company pursuant to Section 1(b)(iii), unless the Holders of a majority of the Registrable Securities then outstanding agree in writing to forfeit one of the registrations to which the Holders are entitled under Section 1(b).
- (h) <u>Delay of Registration</u>. No Holder shall have any right to obtain or seek an injunction restraining or otherwise delaying any such registration as the result of any controversy that might arise with respect to the interpretation or implementation of this Section 1.
 - (i) Indemnification. In the event any Registrable Securities are included in a registration statement under this Section 1:
- (i) To the extent permitted by law, the Company will indemnify and hold harmless each Holder, any underwriter (as defined in the Act) for such Holder and each person, if any, who controls such Holder or underwriter within the meaning of the Act or the 1934 Act, against any losses, claims, damages, or liabilities (joint or several) to which they may become subject under the Act, the 1934 Act or other federal or state law, insofar as such losses, claims, damages, or liabilities (or actions in respect thereof) arise out of or are based upon any of the following statements, omissions or violations (collectively, a "Violation"): (i) any untrue statement or alleged untrue statement of a material fact contained in such registration statement, including any preliminary prospectus or final prospectus contained therein or any amendments or supplements thereto, (ii) the omission or alleged omission to state therein a material fact required to be stated therein, or necessary to make the statements therein not misleading, or (iii) any violation or alleged violation by the Company of the Act, the 1934 Act, any state securities law or any rule or regulation

promulgated under the Act, the 1934 Act or any state securities law; and the Company will pay to each such Holder, underwriter or controlling person any legal or other expenses reasonably incurred by them in connection with investigating or defending any such loss, claim, damage, liability, or action, as such expenses are incurred; provided, however, that the indemnity agreement contained in this subsection 1(i)(i) shall not apply to amounts paid in settlement of any such loss, claim, damage, liability, or action it is the Company be liable in any such case for any such loss, claim, damage, liability, or action to the extent that it arises out of or is based upon a Violation which occurs in reliance upon and in conformity with written information furnished expressly for use in connection with such registration by any such Holder, underwriter or controlling person.

- (ii) To the extent permitted by law, each selling Holder will indemnify and hold harmless the Company, each of its directors, each of its officers who has signed the registration statement, each person, if any, who controls the Company within the meaning of the Act, any underwriter, any other Holder selling securities in such registration statement and any controlling person of any such underwriter or other Holder, severally but not jointly, against any losses, claims, damages, or liabilities (joint or several) to which any of the foregoing persons may become subject, under the Act, the 1934 Act or other federal or state law, insofar as such losses, claims, damages, or liabilities (or actions in respect thereto) arise out of or are based upon any Violation, in each case to the extent (and only to the extent) that such Violation occurs in reliance upon and in conformity with written information furnished by such Holder expressly for use in connection with such registration; and each such Holder will pay any legal or other expenses reasonably incurred by any person intended to be indemnified pursuant to this subsection 1(i)(ii), in connection with investigating or defending any such loss, claim, damage, liability, or action; provided, however, that the indemnity agreement contained in this subsection 1(i)(ii) shall not apply to amounts paid in settlement of any such loss, claim, damage, liability or action if such settlement is effected without the consent of the Holder, which consent shall not be unreasonably withheld; provided that, in no event shall Holder's cumulative, aggregate liability under this subsection 1(i)(ii) under Section 1(i)(iv), or under such sections together, exceed the gross proceeds received by such Holder (net of underwriters' discounts and commissions and stock transfer taxes) from the offering out of which such Violation arises.
- (iii) Promptly after receipt by an indemnified party under this Section 1(i) of notice of the commencement of any action (including any governmental action), such indemnified party will, if a claim in respect thereof is to be made against any indemnifying party under this Section 1(i), deliver to the indemnifying party a written notice of the commencement thereof and the indemnifying party shall have the right to participate in, and, to the extent the indemnifying party so desires, jointly with any other indemnifying party similarly noticed, to assume the defense thereof with one counsel mutually satisfactory to the parties; provided, however, that an indemnified party (together with all other indemnified parties which may be represented without conflict by one counsel) shall have the right to retain one separate counsel, with the fees and expenses to be paid by the indemnifying party, if representation of such indemnified party by the counsel retained by the indemnifying party would be inappropriate due to actual or potential differing interests between such indemnified party and any other party represented by such counsel in such proceeding. The failure to deliver written notice to the indemnifying party within a reasonable time of the commencement of any such action shall not relieve such indemnifying party of any liability to the indemnified party under this Section 1(i) unless the failure to deliver notice is materially prejudicial to its ability to

defend such action. Any omission to so deliver written notice to the indemnifying party will not relieve it of any liability that it may have to any indemnified party otherwise than under this Section 1(i).

- (iv) If the indemnification provided for in this Section 1(i) is held by a court of competent jurisdiction to be unavailable to an indemnified party with respect to any loss, liability, claim, damage, or expense referred to therein, then the indemnifying party, in lieu of indemnifying such indemnifying party hereunder, shall contribute to the amount paid or payable by such indemnified party as a result of such loss, liability, claim, damage, or expense in such proportion as is appropriate to reflect the relative fault of the indemnifying party on the one hand, and of the indemnified party on the other, in connection with the statements or omissions that resulted in such loss, liability, claim, damage, or expense as well as any other relevant equitable considerations. The relative fault of the indemnifying party and of the indemnified party shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission to state a material fact relates to information supplied by the indemnifying party or by the indemnified party and the parties' relative intent, knowledge, access to information, and opportunity to correct or prevent such statement or omission.
- (v) Notwithstanding the foregoing, to the extent that the provisions on indemnification and contribution contained in the underwriting agreement entered into in connection with the underwritten public offering are in conflict with the foregoing provisions, the provisions in the underwriting agreement shall control.
 - (vi) The obligations of the Company and Holders under this Section 1(i) shall survive the completion of any offering of Registrable Securities in a registration statement under this Section 1, and otherwise.
- (j) Reports under Securities Exchange Act of 1934. With a view to making available to the Holders the benefits of Rule 144 promulgated under the Act and any other rule or regulation of the SEC that may at any time permit a Holder to sell securities of the Company to the public without registration or pursuant to a registration on Form S-3., the Company agrees to:
- (i) make and keep public information available, as those terms are understood and defined in SEC Rule 144, at all times after ninety (90) days after the effective date of the first registration statement filed by the Company for the offering of its securities to the general public;
- (ii) take such action, including the voluntary registration of its Common Stock under Section 12 of the 1934 Act, as is necessary to enable the Holders to utilize Form S-3 for the sale of their Registrable Securities, such action to be taken as soon as practicable after the end of the fiscal year in which the first registration statement filed by the Company for the offering of its securities to the general public is declared effective;
 - (iii) file with the SEC in a timely manner all reports and other documents required of the Company under the Act and the 1934 Act; and

- (iv) furnish to any Holder, so long as the Holder owns any Registrable Securities, forthwith upon request (i) a written statement by the Company that it has complied with the reporting requirements of SEC Rule 144 (at any time after ninety (90) days after the effective date of the first registration statement filed by the Company), the Act and the 1934 Act (at any time after it has become subject to such reporting requirements), or that it qualifies as a registrant whose securities may be resold pursuant to Form S-3 (at any time after it so qualifies), (ii) a copy of the most recent annual or quarterly report of the Company and such other reports and documents so filed by the Company, and (iii) such other information as may be reasonably requested in availing any Holder of any rule or regulation of the SEC which permits the selling of any such securities without registration or pursuant to such form.
- (k) <u>Assignment of Registration Rights</u>. The rights to cause the Company to register Registrable Securities pursuant to this Section 1 may be assigned (but only with all related obligations) by a Holder to, (i) in the case of any Holder that is a partnership, limited liability company or corporation, any current and former constituent partners, members, stockholders and Affiliates of that Holder, and (ii) in the case of any Holder, (w) to any other Holder, (x) a transferee or assignee of such securities who, after such assignment or transfer, holds (together with Affiliates) at least 1.000,000 shares of the Company's outstanding capital stock, or all of such transferring Holder's shares if a lesser amount is transferred, (y) a transferee or assignee who is a spouse, lineal descendant, father, mother, brother or sister (each, a "Family Member") of such transferring Holder or (z) or to a trust, the beneficiaries of which are exclusively the Holder and/or Family Members, <u>provided</u>, in each case, <u>that</u>: (a) the Company is, within a reasonable time after such transfer, furnished with written notice of the name and address of such transferee or assignee and the securities with respect to which such registration rights are being assigned; (b) such transferee or assignee agrees in writing to be bound by and subject to the terms and conditions of this Agreement, including without limitation the provisions of Section 1(1) below; and (c) such assignment shall be effective only if immediately following such transfer the further disposition of such securities by the transferee or assignee is restricted under the Act. For the purposes of determining the number of shares of Registrable Securities held by a transferee or assignee of a holder of Registrable Securities, the holdings of affiliated partnerships and other entities; <u>provided that</u> all assignees and transferees who would not qualify individually for assignment of registration rights shall have a single attorney-in-fact for the purpose of exercising any rights, rec
- (l) "Market Stand-Off" Agreement. Each Holder hereby agrees that, during the period of duration specified by the managing underwriter of common stock or other securities of the Company following the effective date of a registration statement for the initial public offering of the Company's Common Stock filed under the Act, it shall not, to the extent requested by such underwriter, directly or indirectly sell, offer to sell, contract to sell (including, without limitation, any short sale), grant any option to purchase or otherwise transfer or dispose of (other than to donees who agree to be similarly bound) any securities of the Company held by it at any time during such period except Common Stock included in such registration; provided, however, that:

- (i) all officers and directors of the Company enter into similar agreements or are bound by similar agreements with the Company;
- (ii) the Company uses all reasonable efforts to obtain from persons who hold in excess of one percent (1%) of the Company's outstanding capital stock, a lock-up agreement similar to that set forth in this Section 1(1); and
 - (iii) such market stand-off time period shall not exceed one hundred eighty (180) days.

Each Holder agrees to provide to the other underwriters of any such initial public offering such further agreements as such underwriter may reasonably request in connection with this market stand-off agreement, provided that the terms of such agreements are substantially consistent with the provisions of this Section 1(l). In order to enforce the foregoing covenant, the Company may impose stop-transfer instructions with respect to the Registrable Securities of each Holder (and the shares or securities of every other person subject to the foregoing restriction) until the end of such period.

Notwithstanding anything to the contrary contained herein, this Agreement shall not restrict Lehman Brothers, J.P. Morgan Chase & Co. or their respective Affiliates from engaging in any brokerage, investment advisory, financial advisory, anti-raid advisory, merger advisory, financing, asset management, trading, market making, arbitrage or other similar activities conducted in the ordinary course of its or its Affiliates' business, so long as such activities are not conducted with respect to any Registrable Securities.

Notwithstanding the foregoing, the obligations described in this Section 1 (I) shall not apply to a registration relating solely to employee benefit plans on Form S-1 or Form S-8 or similar forms which may be promulgated in the future, or a registration relating solely to an SEC Rule 145 transaction.

(m) <u>Termination of Registration Rights</u>. The right of any Holder to request registration or inclusion in any registration pursuant to this Section 1 shall terminate upon the earlier of (i) the date which is five (5) years after the effective date of the first registration statement for an initial public offering of securities of the Company (other than a registration statement relating either to the sale of securities to employees of the Company pursuant to a stock option, stock purchase or similar plan or an SEC Rule 145 transaction) or (ii) such date as (x) a public trading market shall exist for the Company's Common Stock and (y) all shares of Registrable Securities beneficially owned or subject to Rule 144 aggregation by such Holder may immediately be sold under Rule 144 (without regard to Rule 144(k)) during any 90-day period, provided that such Holder is not then an "affiliate" of the Company within the meaning of Rule 144 and such Holder's shares represent less than 1% of the Company's outstanding capital stock.

2. Covenants.

(a) Information and Inspection Rights.

- (i) The Company shall deliver to each holder of at least five hundred thousand (500,000) shares of Preferred (or Common Stock issuable upon conversion thereof; subject to appropriate adjustments for stock splits, stock dividends, combinations and other recapitalizations) (each, a "Major Holder"):
- (1) as soon as practicable, but in any event within one hundred twenty (120) days after the end of each fiscal year of the Company, an income statement for such fiscal year, a balance sheet of the Company and statement of stockholders' equity as of the end of such year, and a statement of cash flows for such year, such year-end financial reports to be in reasonable detail, prepared in accordance with generally accepted accounting principles ("GAAP"), and audited and certified by independent public accountants of nationally recognized standing selected by the Company;
- (2) as soon as practicable, but in any event within forty-five (45) days after the end of each quarter of each fiscal year of the Company, an unaudited profit or loss statement, a statement of cash flows for such fiscal quarter and an unaudited balance sheet as of the end of such fiscal quarter;
- (3) at the time of delivery of each statement pursuant to Section 2(a)(2), a management narrative report explaining all significant variances from forecasts and all significant current developments in staffing, marketing, sales and operations;
- (4) no later than 10 days subsequent to board approval of such documents, (consolidated) capital and operating expense budgets, cash flow projections and income and loss projections for the Company (and its subsidiaries, if any) in respect of such fiscal year, all itemized in reasonable detail and prepared on a monthly basis, and promptly after preparation, any revisions of the foregoing;
- (5) promptly following receipt by the Company, each audit response letter, accountants management letter and other written report submitted to the Company by its independent public accountants in connection with an annual or interim audit of the books of the Company or any of its subsidiaries;
- (6) promptly after the commencement thereof, notice of all actions, suits, claims, proceedings, investigations and inquiries that could materially adversely affect the Company or any of its subsidiaries, if any;
- (7) promptly upon sending, making available or filing the same, all press releases, reports and financial statements that the Company sends or makes available to its stockholders or directors or files with the SEC; and
 - (8) such other information relating to the financial condition, business, prospects or corporate affairs of the Company as the Holder or any assignee of the Holder

may from time to time request, <u>provided</u>, <u>however</u>, that the Company shall not be obligated under this subsection (a)(i)(8) or any other subsection of Section 2(a) to provide information which it deems in good faith to be a trade secret or similar confidential information unless the Holder or assignee receiving such information shall have first entered into a confidentiality agreement acceptable in form and substance to the Company.

- (ii) The Company shall afford each Major Holder, at the principal offices of the Company, reasonable access to material documents of the Company and rights to examine the books and records of the Company and to inspect the facilities and offices of the Company, upon at least five (5) days notice in advance of such visit to the Company from such Major Holder specifying which documents, offices and facilities such Major Holder wishes to inspect and the purpose of such inspection, but in any event not more than once every fiscal quarter.
- (iii) The Company shall deliver to each holder of at least three million (3,000,000) shares of Preferred (or Common Stock issuable upon conversion thereof; subject to appropriate adjustments for stock splits, stock dividends, combinations and other recapitalizations) copies of any materials distributed to members of the Board of Directors in connection with any meeting of the Board of Directors as and when distributed to such persons by the Company.
- (b) <u>Termination of Covenants</u>. The covenants set forth in Section 2(a) shall terminate and be of no further force or effect when the sale of securities pursuant to a registration statement filed by the Company under the Act in connection with the first firm commitment underwritten offering of its securities to the general public is consummated or when the Company first becomes subject to the periodic reporting requirements of Sections 12(g) or 15(d) of the 1934 Act, whichever event shall first occur.

(c) Right to Maintain Proportionate Ownership.

(i) In the event the Company desires to sell and issue shares of its capital stock or rights, options or other securities exercisable for or convertible into shares of its capital stock (directly or indirectly) and whether or not such right or option is immediately exercisable or convertible ("Shares"), then the Company shall first notify each Holder of the material terms of the proposed sale and shall permit such Holders to acquire, at the time of consummation of such proposed issuance and sale and on such terms as are specified in the Company's notice pursuant hereto, such number of Shares proposed for issuance and sale as would be required to enable each to maintain its voting and ownership rights in the Company following such issuance, on an as-converted, fully-diluted percentage basis, at a level maintained by it immediately prior to such proposed issuance. Each Holder shall have thirty (30) days after the date of any such notice to elect by notice to the Company to purchase any such Shares on such terms and at the time the proposed sale is consummated. If any Holder does not agree to purchase its full allotment of such Shares within such 30-day period (including by operation of a waiver under Section 7(g)), the Company shall notify each Holder that has elected to purchase its full allotment of such Shares of the number of Shares that were not subscribed for. Each such Holder shall then have ten (10) days to elect by notice to the Company to purchase its pro rata portion of the remaining Shares not purchased by such non-purchasing Holders. For the purposes of determining the number of Shares held by a transferee or assignee of Shares, the holdings of Affiliates, partners (or members) or retired partners

(or retired members) of such assigning or transferring Holder, or the estates and family members of any persons and any trusts for the benefit of any of the foregoing persons shall be aggregated together with such assigning or transferring Holder; provided that all assignees and transferes who would not qualify individually for assignment of rights hereunder shall have a single attorney-in-fact for the purpose of exercising any rights, receiving notices or taking any action under this Section 2(c).

- (ii) The right to maintain proportionate ownership set forth in this Section 2(c) shall not be applicable:
- (1) to the issuance or sale of shares (or options or warrants therefor) to officers, directors, employees, advisors, contractors or consultants, or to other providers of goods, services, technology, distribution channels or marketing opportunities pursuant to a stock purchase agreement, stock grant, option plan or purchase plan or other stock incentive program or arrangement approved by the Compensation Committee of the Board of Directors, or, in the absence of such a committee, the Company's Board of Directors;
- (2) to or after consummation of a bona fide, firm commitment underwritten public offering of shares of Common Stock, at a per share public offering price as shown on the cover page of the final prospectus relating to such offering (prior to underwriter discounts, commissions, concessions and expenses) of at least \$2.50 (as adjusted for stock splits, stock dividends, combinations and the like), registered under the Act pursuant to a registration statement on Form S-1 or SB-2, which results in aggregate gross proceeds to the Company of at least \$25,000,000 (a "Qualified IPO");
 - (3) to the issuance of securities pursuant to currently outstanding options, warrants, notes, or other rights to acquire securities of the Company;
- (4) to the issuance of shares upon conversion of the Series A Preferred, Series B Preferred, Series C Preferred, Series C-1 Preferred, Series D Preferred or Series E Preferred that is outstanding immediately following the closing under the Purchase Agreement;
- (5) to the issuance of securities in connection with a bona fide business acquisition of or by the Company approved by the Board of Directors, whether by merger, consolidation, sale of assets, sale or exchange of stock or otherwise;
- (6) to the issuance of securities to financial institutions, landlords or lessors in connection with commercial credit arrangements, real property leases, equipment financings or similar transactions undertaken for other than equity financing purposes (not to exceed 1,275,000 shares in the aggregate, excluding any shares issued or issuable to DoubleClick, Inc. ("DoubleClick") in connection with a Master Alliance Agreement between DoubleClick and the Company); or
 - (7) to the issuance of securities in connection with any stock split, stock dividend or recapitalization by the Company.

- (iii) The right to maintain proportionate ownership set forth in this Section 2(c) shall terminate immediately prior to the consummation of, and shall not be applicable with respect to, a Qualified IPO.
- (d) <u>Right of First Refusal</u>. In the event that any Founder proposes to offer any Shares for sale or transfer to a third party (other than any transfer or a sale of shares, that, when aggregated with all shares previously transferred by such Founder and not subject to the right of first refusal provisions of this Section 2(d), do not exceed 10% of the aggregate Shares held by such Founder on the date hereof (such transfer, an "*Exempt Transfer*")), the Founder proposing to offer such Shares (the "*Offering Founder*") shall make an offering of such Shares to the Company (first) and the Holders (second) prior to any transfer to any such third party, in accordance with the following provisions:
- (i) The Offering Founder shall deliver a notice in accordance with Section 7(e) (a "*Transfer Notice*") to the Company stating (i) its bona fide intention to offer such Shares, (ii) the number of such Shares to be offered and (iii) the price and terms, if any, upon which it proposes to offer such Shares.
- (ii) Within fifteen (15) days after giving of the Transfer Notice to the Company, the Company may elect to purchase by delivering notice of such election to the Offering Founder, at the price and on the terms specified in the Transfer Notice, all, but not less than all, of the Shares proposed to be offered.
- (iii) If the Company does not elect to purchase all of the shares proposed to be offered within the fifteen (15) day period specified in Section 2(d)(ii), the Offering Founder shall then deliver a Transfer Notice to each Holder.
- (iv) Within fifteen (15) days after giving of the Transfer Notice to the Holders, each Holder may elect to purchase by delivering notice of such election to the Offering Founder that number of Shares equal to the product obtained by multiplying the number of Shares to be offered by a fraction, (x) the numerator of which shall be the number of shares of Common Stock (on a fully-diluted, as-convened basis) at the time owned by such Holder and (y) the denominator of which shall be the number of shares of Common Stock (on a fully-diluted, as-converted basis) at the time owned by all Holders. If any Holder does not agree to purchase its full allotment of such Shares within such 15-day period, the Company shall notify each Holder which elects to purchase its full allotment of such Shares of the number of Shares that were not subscribed for, Each such Holder shall then have ten (10) days to elect by notice to the Company to purchase its pro rata portion of the remaining Shares not purchased by such non-purchasing Holders.
- (v) If all Shares referred to in the Notice are not elected to be purchased as provided in subsection 2(d)(ii) and 2(d)(iv) hereof, then, subject to Section 2(e), the Offering Founder may, within a sixty (60) day period following the expiration of the period provided in subsection 2(d)(iv) hereof, offer the remaining unsubscribed portion of such Shares to any person or persons at a price not less than, and upon terms no more favorable to the offeree, than those specified in the Notice. If the Offering Founder does not enter into an agreement for the sale of such Shares within such period, or if such agreement is not consummated within thirty (30) days of the execution

thereof, the right provided hereunder shall be deemed to be revived and such Shares shall not be offered unless first reoffered to the Company and the Holders in accordance with this Section 2(d).

(e) Co-Sale Rights.

- (i) Notice of Transfer. If a Founder (a "Selling Founder") receives one or more bona fide offers or otherwise undertakes to effect any transaction that would result in the transfer of Shares held by such Selling Founder (including a sale to which Section 2(d) may apply, other than a sale to the Company or a Holder or Holders pursuant to the exercise of rights under Section 2(d)) (collectively, a "Purchase Offer"), from any person to purchase from the Selling Founder any of such Selling Founder's Shares upon specific terms and conditions (including a specified purchase price payable in cash or other property), then the Selling Founder shall promptly notify each Holder in writing, in accordance with Section 7(e), of the terms and conditions of such Purchase Offer (a "Disposition Notice"), which notice, to the extent applicable, may be the same as the Transfer Notice.
- (ii) <u>Grant of Right</u>. To the extent that a Holder has not exercised its right of first offer set forth in Section 2(d) hereof, such Holder shall have the right, exercisable upon written notice to the Selling Founder within thirty (30) business days after effectiveness of the Disposition Notice of the Purchase Offer, to participate in the Selling Founder's sale of the Shares pursuant to the specified terms and conditions of such Purchase Offer. To the extent a Holder exercises such right of participation in accordance with the terms and conditions set forth below, the number of Shares that the Selling Founder may sell pursuant to such Purchase Offer shall be correspondingly reduced. The right of participation of the Holders shall be subject to the following terms and conditions:
- (1) Each Holder may sell all or any part of that number of Shares equal to the product obtained by multiplying the aggregate number of Shares covered by the Purchase Offer by a fraction, (x) the numerator of which shall be the number of shares of Common Stock (on an as-converted, fully-diluted basis) at the time owned by such Holder and (y) the denominator of which shall be the number of shares of Common Stock (on an as-converted, fully-diluted basis) then outstanding.
- (2) Each Holder may effect its participation in the sale by delivering to the Selling Founder, for transfer to the purchase offerer, one or more certificates, properly endorsed for transfer, that represents the number of Shares that such Holder elects to sell pursuant to this Section 2(e).
- (3) To the extent that any prospective purchaser or purchasers prohibit such exercise of co-sale right or otherwise refuses to purchase Shares from a Holder exercising its co-sale right hereunder, the Selling Founder shall not sell to such prospective purchaser or purchasers any Shares unless and until, simultaneously with such sale, the Selling Founder shall purchase such Shares from such Holder for the same consideration and on the same terms and conditions as the proposed transfer described in the Disposition Notice.

- (iii) Mechanics of Transfer. The stock certificates that the Holders deliver to the Selling Founder pursuant to Section 2(e)(ii) shall be transferred by such Selling Founder to the purchase offeror in consummation of the sale of the Selling Founder's Shares pursuant to the terms and conditions specified in the Disposition Notice, and the Selling Founder shall promptly thereafter remit to each Holder that portion of the sale proceeds to which such Holder is entitled by reason of its participation in such sale. In the event that less than all the shares represented by such a stock certificate are sold pursuant to Section 2(e)(ii), the Selling Founder shall instruct the Company to issue a new certificate to the Holder representing the shares not sold.
- (iv) No Effect on Subsequent Rights. The exercise or non-exercise of the rights of any Holder hereunder to participate in one or more sales of the Selling Founders Shares made by a Selling Founder shall not adversely affect the Holder's rights to participate in subsequent sales of Shares by a Selling Founder.
 - (f) Prohibited Founder Transfers. Exclusions from and Expiration of Co-Sale Rights and Rights of First Refusal.
- (i) <u>Agreement Not to Transfer</u>. Any attempt by a Founder to transfer any Shares in violation of Section 2(d) or 2(e) shall be void and the Company agrees that it will not effect such a transfer nor will it treat any alleged transferee as the holder of such shares without the written consent of a majority in interest of the Holders.
- (ii) <u>Repurchase</u>. In the event that a Founder should sell any Shares in contravention of the participation rights of the Company or the Holders under Section 2(d) or 2(e) (a "*Prohibited Founder Transfer*"), each Holder shall have the option to sell to such Founder a number of Shares equal to such Holder's pro rata share (as determined pursuant to Section 2(e)(ii)(l) above), <u>provided that</u>, the date of the Disposition Notice shall be understood to mean the date of the Prohibited Founder Transfer, on the following terms and conditions:
- (1) The price per share at which such Shares are to be sold to such Founder shall be equal to the price per share paid by the third-party purchaser or purchasers of the Shares (the "Contingent Purchaser") to such Founder.
- (2) Such Holder shall deliver to such Founder within ninety (90) days after such Holder has received notice from such Founder or otherwise become aware of the Prohibited Founder Transfer, the certificate or certificates representing Shares to be sold, each certificate to be properly endorsed for transfer.
- (3) Such Founder shall, immediately upon receipt of the certificates for the Shares, pay the aggregate purchase price therefore, by certified check or bank draft made payable to the order of such Holder, and shall reimburse such Holder for any reasonable additional expenses, including legal fees and expenses, incurred in effecting such purchase and sale.
- (iii) Exclusions. The participation rights of the Holders set forth in Sections 2(d) and 2(e) shall not pertain or apply to (i) any pledge of a Founder's Shares made by a Founder that creates only a security interest, (ii) any transfer to the ancestors, descendants or spouse

of a Founder or to trusts for the benefit of such persons or such Founder or (iii) any bona fide gift; provided that, the pledgee, transferee or donee shall furnish the Holders with a written agreement to be bound by and comply with all provisions of this Agreement applicable to the Holders.

- (iv) Expiration. The provisions of Section 2(d) and 2 (e) shall terminate as to any Series A Preferred, Series B Preferred, Series C Preferred, Series C-1 Preferred, Series D Preferred immediately prior to the closing of a Qualified IPO,
- (g) <u>Series E Purchaser Right of First Offer</u>. Except to the extent such transaction or series of related transactions constitutes a Liquidation Event as defined in the Company's Amended and Restated Certificate of Incorporation, in the event that any Holder proposes to offer any Shares for sale or transfer to a third party, the Holder proposing to offer such Shares (the "Offering Holder") shall make an offering of such Shares to the Series E Purchasers prior to any transfer to any such third party, in accordance with the following provisions:
- (i) The Offering Holder shall deliver a Transfer Notice in accordance with Section 7(e) to each Series E Purchaser stating (i) its bona fide intention to offer such Shares, (ii) the number of such Shares to be offered and (iii) the price and terms, if any, upon which it proposes to offer such Shares.
- (ii) Within fifteen (15) days after giving of the Transfer Notice to the Series E Purchasers, each Series E Purchaser may elect to purchase, by delivering notice of such election to the Offering Holder, that number of Shares equal to the product obtained by multiplying the number of Shares to be offered by a fraction, (x) the numerator of which shall be the number of shares of Common Stock (on a fully-diluted, asconverted basis) at the time owned by such Series E Purchaser and (y) the denominator of which shall be the number of shares of Common Stock (on a fully-diluted, as-converted basis) at the time owned by all Series E Purchasers. If any Series E Purchaser does not agree to purchase its full allotment of such Shares within such 15-day period, the Offering Holder shall notify each Series E Purchaser which elects to purchase its full allotment of such Shares of the number of Shares that were not subscribed for. Each such Series E Purchaser shall then have ten (10) days to elect by notice to the Offering Holder to purchase its pro rata portion of the remaining Shares not purchased by such non-purchasing Series E Purchasers.
- (iii) If all Shares referred to in the Transfer Notice are not elected to be purchased as provided in subsection 2(g)(ii) hereof, then the Offering Holder may, within a sixty (60) day period following the expiration of the period provided in subsection 2(g)(ii) hereof, offer all of such Shares referred to in the Transfer Notice to any person or persons at a price not less than, and upon terms no more favorable to the offeree, than those specified in the Transfer Notice. If the Offering Holder does not enter into an agreement for the sale of such Shares within such period, or if such agreement is not consummated within thirty (30) days of the execution thereof, the right provided hereunder shall be deemed to be revived and such Shares shall not be offered unless first reoffered to the Series E Purchasers in accordance with this Section 2(g).
- (iv) Exclusions. The participation rights of the Series E Purchasers set forth in this Section 2(g) shall not pertain or apply to any transfer, (i) in the case of a Holder that is a partnership, limited liability company or corporation, to any current and former constituent parties,

members, stockholders and Affiliates of that Holder, or (ii) in the case of any other Holders, to Family Members of such Holder or to a trust the beneficiaries of which are exclusively the Holder and/or Family Members of the Holder; provided that, the transferee shall furnish the Company and the Series E Purchasers with a written agreement to be bound by and shall comply with all provisions of this Agreement applicable to the Holders.

(v) The provisions of this Section 2(g) shall terminate immediately prior to the closing of a Qualified IPO.

(h) Prohibited Holder Transfers.

- (i) <u>Agreement Not to Transfer</u>. Any attempt by a Holder to transfer any Shares in violation of Section 2(g) shall be void and the Company agrees that it will not effect such a transfer nor will it treat any alleged transferee as the holder of such shares without the written consent of a majority in interest of the Series E Purchasers.
- (ii) Repurchase. In the event that a Holder should sell any Shares in contravention of the right of first offer of the Series E Purchasers under Section 2(g) (a "Prohibited Holder Transfer"), each Series E Purchaser shall have the option to sell to such Holder a number of Shares equal to such Series E Purchaser's pro rata share (as determined pursuant to Section 2(g)(ii) above), provided that, the date of the Transfer Notice shall be understood to mean the date of the Prohibited Holder Transfer, on the following terms and conditions:
 - (1) The price per share at which such Shares are to be sold to such Holder shall be equal to the price per share paid by the Contingent Purchaser to such Holder.
- (2) Such Series E Purchaser shall deliver to such Holder within ninety (90) days after such Series E Purchaser has received notice from such Holder or otherwise become aware of the Prohibited Holder Transfer, the certificate or certificates representing Shares to be sold, each certificate to be properly endorsed for transfer.
- (3) Such Holder shall, immediately upon receipt of the certificates for the Shares, pay the aggregate purchase price therefore, by certified check or bank draft made payable to the order of such Series E Purchaser, and shall reimburse such Series E Purchaser for any reasonable additional expenses, including legal fees and expenses, incurred in effecting such purchase and sale.

3 Voting

(a) <u>Board Composition</u>. During the term of this Section 3 and subject to Section 5 of Article IV(B) of the Company's Seventh Amended and Restated Certificate of Incorporation") and the relevant sections of the Company's Bylaws (if any), each Holder and each Founder agrees to vote all Shares now or hereafter directly or indirectly acquired (of record or beneficially) by such Holder or Founder, in such manner as may be necessary to elect (and maintain in office) as members of the Company's Board of Directors, the following individuals:

- (i) one (1) individual designated by the J.P. Morgan Purchasers (as indicated on the signature pages hereto) from time to time; provided that, the J.P. Morgan Purchasers continue to hold at least 1,000,000 shares of Preferred (or Common Stock issuable upon conversion thereof) (as adjusted for stock splits, stock dividends, combinations and the like);
- (ii) one (1) individual designated by the Accel Purchasers (as indicated on the signature pages hereto) from time to time, provided that, the Accel Purchasers continue to hold at least 1,000,000 shares of Preferred (or Common Stock issuable upon conversion thereof) (as adjusted for stock splits, stock dividends, combinations and the like);
- (iii) one (1) individual designated by the Adams Street Purchasers (as indicated on the signature pages hereto) from time to time, <u>provided that</u>, the Adams Street Purchasers continue to hold at least 1,000,000 shares of Preferred (or Common Stock issuable upon conversion thereof) (as adjusted for stock splits, stock dividends, combinations and the like);
- (iv) one (1) individual designated by the Topspin Purchasers (as indicated on the signature pages hereto) from time to time, provided that, the Topspin Purchasers continue to hold at least 750,000 shares of Preferred (or Common Stock issuable upon conversion thereof) (as adjusted for stock splits, stock dividends, combinations and the like);
- (v) one (1) individual designated by Citadel Equity Fund Ltd. or its affiliates ("Citadel"), who shall initially be a member of the senior executive management of Citadel; provided that, Citadel or its Affiliates continue to hold at least 3,000,000 shares of Preferred (or Common Stock issuable upon conversion thereof) (as adjusted for stock splits, stock dividends, combinations and the like);
- (vi) two (2) individuals designated by the holders of a majority of the Common Stock and acceptable to a majority of the members of the Company's Board of Directors other than those designated pursuant to this Section 3(a)(vi);
- (vii) that number of individuals as required to elect the remaining number of directors authorized to be elected as a director pursuant to the Company's Bylaws or Certificate of Incorporation, designated by the holders of a majority of the Preferred and the Common Stock, voting together as a single class.

The individuals designated pursuant to clauses (i) and (ii) above shall be the directors elected by the holders of the Series A Preferred pursuant to Section 5(g) of Article IV(B) of the Company's Certificate of Incorporation, the individual designated pursuant to clause (iii) above shall be the director elected by the holders of the Series C Preferred pursuant to Section 5(g) of Article IV(B) of the Company's Certificate of Incorporation, the individual designated pursuant to clause (iv) above shall be the director elected by the holders of Series D Preferred pursuant to Section 5(g) of Article IV(B) of the Company's Certificate of Incorporation and the individual designated pursuant to clause (v) above (the "Series E Director") shall be the director elected by the holders of the Series E Preferred pursuant to Section 5(g) of Article IV(B) of the Company's Certificate of Incorporation. The Company shall provide that the Series E Director designated pursuant to this Section 3 is appointed to the compensation committee of the Board of Directors. For purposes of this

Agreement: (i) any individual who is designated for election to the Company's Board of Directors pursuant to the foregoing provisions of this Section 3(a) is hereinafter referred to as a "Board Designee"; and (ii) any individual, entity, or group of individuals and/or entities who has the right to designate one or more Board Designees for election the Company's Board of Directors pursuant to the foregoing provisions of this Section 3(a) is hereinafter referred to as a "Designator" or as "Designators," as applicable.

- (b) Changes in Board Designees. From time to time during the term of this Agreement, a Designator or Designators shall, in their sole discretion, have the sole right to:
- (i) elect to remove from the Company's Board of Directors any incumbent Board Designee who occupies a Board seat for which such Designator or Designators are entitled to designate the Board Designee under Section 3(a); and/or
- (ii) designate a new Board Designee for election to a Board seat for which such Designator or Designators are entitled to designate the Board Designee under Section 3(a) (whether to replace a prior Board Designee or to fill a vacancy in such Board seat); provided that, such removal and/or designation of a Board Designee is approved in a writing signed by Designators who are entitled to designate such Board Designee under Section 3(a), in which case such election to remove a Board Designee and/or elect a new Board Designee will be binding on all such Designators. In the event of such a removal and/or designation of a Board Designee under this Section 3(b)(ii), the Holders and the Founders shall vote their shares of the Company's capital stock as provided in Section 3(a) to cause: (a) the removal from the Company's Board of Directors of the Board Designee or Designees so designated for removal by the appropriate Designators; and (b) the election to the Company's Board of Directors by the appropriate Designators.
- (iii) Notwithstanding anything to the contrary in this Section 3, if a Board Designee designated pursuant to Section 3(a)(vi) who is an employee of the Company subsequently ceases to be an employee of the Company, such Board Designee may be removed from the Company's Board of Directors by the holders of a majority of the Preferred and the Common Stock, voting together as a single class. If such a Board Designee is so removed, the Designators entitled to designate such Board Designee may designate a new Board Designee for election to the Company's Board of Directors in accordance with Sections 3(a)(vi) and 3(b)(ii).
- (c) Notice. The Company shall promptly give each of the Holders and the Founders written notice of any change in composition of the Company's Board of Directors and of any proposal by a Designator or Designators to remove or elect a new Board Designee.
- (d) <u>Further Assurances</u>. Each of the Holders, each of the Founders and the Company agree not to vote any shares of Company capital stock, or to take any other actions, that would in any manner defeat, impair, be inconsistent with or adversely affect the stated intentions of the parties under Section 3 of this Agreement.

(e) Term. The provisions of this Section 3 shall commence on the Closing and shall terminate upon the consummation of a Qualified IPO.

4. Regulatory Matters

- (a) Cooperation of Stockholders. Each Holder and each Founder agrees to cooperate with the Company in all reasonable respects in complying with the terms and provisions of the letter agreements (i) between the Company and JP Morgan Partners (SBIC), LLC ("IP Morgan SBIC") and (ii) between the Company and vSpring SBIC, L.P. ("vSpring"), copies of which are attached to the Purchase Agreement as Exhibit G and Exhibit H, respectively, regarding regulatory matters (the "Regulatory Side Letters"), including without limitation, voting to approve amending the Company's Certificate of Incorporation, the Company's by-laws or this Agreement in a manner reasonably acceptable to the Holders, Founders, JP Morgan SBIC, vSpring or any Affiliate of JP Morgan SBIC or vSpring entitled to make such request pursuant to the Regulatory Side Letters in order to remedy a Regulatory Problem (as defined in the relevant Regulatory Side Letter). Anything contained in this Section 4(a) to the contrary notwithstanding, no Holder or Founder shall be required under this Section 4(a) to take any action that would adversely affect in any material respect such Holder or Founder's rights under this Agreement or as a stockholder of the Company.
- (b) <u>Covenant not to Amend</u>. The Company and each Holder and Founder agree not to amend or waive the voting or other provisions of the Company's Certificate of Incorporation, the Company's By-laws or this Agreement if such amendment or waiver would cause any Regulated Holder to have a Regulatory Problem (as defined in the relevant Regulatory Side Letter). Each of JP Morgan SBIC and vSpring agree to notify the Company as to whether or not it would have a Regulatory Problem promptly after such party has notice of such amendment or waiver.
- 5. <u>HSR</u>. The Company will reimburse the Holders for all expenses, including filing fees and attorney fees, incurred in connection with any necessary Hart Scott Rodino filings with respect to this financing and any future financing transactions involving the Company.
- 6. <u>Observer Rights</u>. Each of Lehman Brothers Venture Partners L.P., vSpring and Citadel shall have the right to designate one individual as an observer at all meetings of the Board of Directors of the Company and any committees thereof, and to receive any materials distributed to members of the Board of Directors or of such committees, as the case may be, as and when distributed to such persons by the Company will reimburse the reasonable out-of-pocket expenses incurred by the observer in connection with attending any such meeting.

7 Miscellaneous

(a) <u>Successors and Assigns</u>. Except as otherwise provided herein, the terms and conditions of this Agreement shall inure to the benefit of and be binding upon the respective successors and assigns of the parties (including transferees of any shares of Registrable Securities). Nothing in this Agreement, express or implied, is intended to confer upon any party other than the parties hereto or their respective successors and assigns any rights, remedies, obligations, or liabilities under or by reason of this Agreement, except as expressly provided in this Agreement.

- (b) <u>Governing Law</u>. This Agreement shall be governed by and construed under the laws of the State of Delaware as applied to agreements among Delaware residents entered into and to be performed entirely within Delaware.
 - (c) Counterparts. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
 - (d) Titles and Subtitles. The titles and subtitles used in this Agreement are used for convenience only and are not to be considered in construing or interpreting this Agreement.
- (e) Notices. All notices and other communications required or permitted hereunder shall be in writing, shall be effective when given, and shall in any event be deemed to be given upon receipt or, if earlier, (a) five (5) days after deposit with the U.S. Postal Service or other applicable postal service, if delivered by first class mail, postage prepaid, (b) upon delivery, if delivered by hand, (c) one (1) business day after the business day of deposit with Federal Express or similar overnight courier, freight prepaid or (d) upon confirmation of receipt, if delivered by facsimile or electronic (email) transmission, and shall be addressed (i) if to a Purchaser, at such Purchaser's address as set forth beneath its signature to this Agreement and (ii) if to the Company, at the address of its principal corporate offices (attention: Chief Financial Officer), or at such other address as a party may designate by ten (10) days' advance written notice to the other party pursuant to the provisions above.
- (f) Expenses. Except as otherwise provided herein, if any action at law or in equity is necessary to enforce or interpret the terms of this Agreement, the prevailing party shall be entitled to reasonable attorneys' fees, costs and necessary disbursements in addition to any other relief to which such party may be entitled.
- (g) Amendments and Waivers. Except as otherwise specified, any term of this Agreement may be amended and the observance of any term of this Agreement may be waived (either generally or in a particular instance and either retroactively or prospectively), only with the written consent of (i) the Company, (ii) the holders of a majority of the Registrable Securities then outstanding and (iii) the holders of a majority of the Series E Preferred (or Common Stock issued upon conversion of the Series E Preferred) (with respect to (iii), the "Series E Consent"); provided that (A) adding new parties to this Agreement or the proportionate adjustments in rights that would result from adding new parties (including proportionate waivers of preemptive rights under Section 2(c) obtained in connection with adding such parties to the Agreement) shall not be deemed to be amendments requiring the Series E Consent and (B) no amendment shall be effective to the extent it disproportionately and adversely affects any class or series of equity securities vis-a-vis any other class or series, unless such amendment is approved by the holders of a majority in interest of the securities adversely affected. Any amendment or waiver effected in accordance with this paragraph shall be binding upon each holder of any Registrable Securities then outstanding, each future holder of all such Registrable Securities, and the Company. Notwithstanding the foregoing, the consent of Founders holding a majority of the total number of shares held by the Founders shall be required for any amendment of Sections 2(d)-2(f) which adversely affects the Founders; the consent of the J.P. Morgan Purchasers shall be required for any amendment of this Agreement that affects the

manner in which the director seat provided for in clause (i) of Section 3(a) hereof is designated or for any amendment of Section 4; the consent of the Accel Purchasers shall be required for any amendment of this Agreement that affects the manner in which the director seat provided for in clause (ii) of Section 3(a) hereof is designated; the consent of the Topspin Purchasers shall be required for any amendment of this Agreement that affects the manner in which the director seat provided for in clause (ii) of Section 3(a) hereof is designated; the consent of the Topspin Purchasers shall be required for any amendment of this Agreement that affects the manner in which the director seat provided for in clause (iv) of Section 3(a) hereof is designated; the consent of vSpring shall be required for any amendment of this Agreement; the consent of Citadel shall be required for any amendment of this Agreement; the the manner in which the director seat provided for in clause (v) of Section 3(a) hereof is designated or the provisions of Section 6 of this Agreement; and the consent of the holders of a majority of the Common Stock and a majority of the members of the Company's Board of Directors other than those designated pursuant to Section 3(a)(vi) shall be required for any amendment of this Agreement that affects the manner in which the director seats provided for in clause (vi) of Section 3(a) hereof are designated.

- (h) <u>Aggregation of Stock</u>. All shares of the Series A Preferred, Series B Preferred, Series C Preferred, Series D Preferred and Series E Preferred held or acquired (or Common Stock issuable upon conversion thereof) by entities, partnerships, former partnerships or persons affiliated with a Holder or Family Members of such Holder, or trusts the beneficiaries of which are affiliated entities or persons and/or Family Members of such Holder shall be aggregated together for the purpose of determining the availability or discharge of any rights of such Holder under this Agreement. Any such affiliated group shall be entitled to designate one person as representative of such group for the purpose of exercising any right or undertaking any obligation of such group hereunder, and the Company shall be entitled to rely on the representative for such purposes.
- (i) <u>Severability</u>. If one or more provisions of this Agreement are held to be unenforceable under applicable law, such provision shall be excluded from this Agreement and the balance of the Agreement shall be interpreted as if such provision were so excluded and shall be enforceable in accordance with its terms.
 - (j) Entire Agreement. This Agreement (including the Exhibits hereto, if any) constitutes the full and entire understanding and Agreement between the parties with regard to the subjects hereof and thereof.
- (k) <u>Waiver of Right to Maintain Proportionate Ownership</u>. By execution of this Agreement, each Prior Purchaser and Founder, on behalf of itself and each of the other Prior Purchasers and Founders, hereby waives the Right to Maintain Proportionate Ownership (including the notice requirements thereof and the over-allotment right contained therein) pursuant to Section 2(c) of the Prior Agreement to purchase any of the shares of Series E Preferred being offered pursuant to the Purchase Agreement (or the shares of Common Stock into which such Series E Preferred are convertible).

(1) <u>Termination of Prior Agreement</u> . By executing and delivering this Agreement, the Company, each Purchaser and each Founder agree that the Prior Agreement is hereby terminated and of no further force and effect.				
-27-				

IN WITNESS WHEREOF, the parties have executed this Fourth Amended and Restated Investor Rights Agreement as of the date first above written.

COMSCORE NETWORKS, INC.

 $\begin{array}{ll} \text{By:} & \underline{\textit{/s/Magid Abraham}} \\ & \underline{\text{Magid Abraham}} \\ & \text{President and Chief Executive Officer} \end{array}$

Address for Notices: 11465 Sunset Hills Road Reston, VA 20190

Citadel Purchasers

CITADEL EQUITY FUND LTD.

By: Citadel Limited Partnership, Portfolio Manager

By: GLB Partners, L.P., its General Partner

By: Citadel Investment Group, L.L.C., its

General Partner

By: /s/ Adam Cooper Name: Adam Cooper Title: General Counsel

Address for Notices: 131 South Dearborn 37th Floor Chicago, IL 60603 Tel: (312) 395-2100 Fax:(312) 977-0250

With a copy to:

131 South Dearborn

32nd Floor Chicago, IL 60603 Attn: General Counsel Tel: (312) 395-3067

Fax:(312)977-0280

Topspin Purchasers

TOPSPIN PARTNERS L.P,

By: Topspin Management, LLC By: LG Capital Appreciation, LLC

By: /s/ Leo Guthart

Leo Guthart Member, LG Capital Appreciation, LLC

Address for Notices: 3 Expressway Plaza Suite 100 Roslyn Heights, NY 11577 Attention: Leo Guthart Tel.: (516) 625-9400 Fax:(516) 625-9499

TOPSPIN ASSOCIATES, L.P.

By: Topspin Management, LLC By: LG Capital Appreciation, LLC

By: /s/ Leo Guthart

Leo Guthart

Member, LG Capital Appreciation, LLC

Address for Notices: 3 Expressway Plaza Suite 100 Roslyn Heights, NY 11577 Attention: Leo Guthart Tel: (516) 625-9400 Fax:(516) 625-9499

vSpring Purchaser

vSPRING SBIC, L.P.

vSpring SBIC, L.P., a Delaware limited partnership

By: vSpring SBIC Management, L.L.C., a Delaware limited liability company, its General Partner

By: /s/ Scott Petty
Scott Petty, Managing Member

Address for Notices: vSpring Capital 2795 East Cottonwood Parkway Suite 360 Salt Lake City, UT 84121 Attention: Scott Petty Tel.: (801) 942-8999 Fax: (801) 942-1636

Adams Street Purchasers

BVCF IV, L.P.

By: J. W. Puth Associates, LLC, its General Partner
By: Brinson Venture Management, LLC, its
Attorney-in-fact
By: Adams Street Partners, LLC, its
Administrative Member

By: /s/ Thomas D. Berman Thomas D. Berman Partner

Address for Notices for Adams Street Purchasers: 209 South LaSalle Street Chicago, IL 60604 Attention: Thomas D. Berman Tel: (312) 553-7866 Fax: (312) 553-7870

IVP Purchasers

INSTITUTIONAL VENTURE PARTNERS X, L.P.

By: Institutional Venture Management X, LLC

General Partner

By: /s/ Norm Fogelsong Managing Director

INSTITUTIONAL VENTURE PARTNERS X GmbH & CO. BETEILIGUNGS KG

By: Institutional Venture Management-X LLC Its: Managing Limited Partner

By: /s/ Norm Fogelsong

Managing Director

Address for Notices for IVP Purchasers: 3000 Sand Hill Road Building 2, Suite 290 Menlo Park, CA 94025 Attn: Todd Chaffee

Accel Purchasers

ACCEL VII L.P.

By: Accel VII Associates L.L.C. Its General Partner

 $\begin{tabular}{ll} By: & $\frac{\mbox{/s/ Tracy L. Sedlock}}{\mbox{ATTORNEY-IN-FACT}} \end{tabular}$

ACCEL INTERNET FUND III L.P.

By: Accel Internet Fund III Associates L.L.C. Its General Partner

By: /s/ Tracy L. Sedlock ATTORNEY-IN-FACT

ACCEL INVESTORS '99 L.P.

By: /s/ Tracy L. Sedlock
ATTORNEY-IN-FACT

J.P Morgan Purchasers

THE FLATIRON FUND 2000, LLC

By: /s/ Frederick Wilson Managing Member

FLATIRON ASSOCIATES II, LLCBy: Flatiron Partners, LLC Its Manager

By: /s/ Frederick Wilson Managing Partner

THE FLATIRON FUND 1998/99, LLC

By: /s/ Frederick Wilson

Managing Member

FLATIRON ASSOCIATES, LLC By: Flatiron Partners, LLC Its Manager

By: /s/ Frederick Wilson

Managing Partner

THE FLATIRON FUND 2001, LLC

By: /s/ Frederick Wilson

Managing Member

JP MORGAN PARTNERS (SBIC), LLC

/s/ Stephen P. Murray

Name: Stephen P. Murray Its: Managing Director

J.P. MORGAN PARTNERS (BHCA), L.P. By: JPMP Master Fund Manager, L.P.,

Its General Partner

By: JPMP Capital Corp, Its General Partner

By: /s/ Stephen P. Murray

Name: Stephen P. Murray Title: Managing Director

Addresses for Notices for J.P. Morgan Purchasers: c/o J.P. Morgan Partners, LLC 1221 Avenue of the Americas, 39th Floor New York, NY 10020

Attn: Official Notices Clerk FBO: Fred Wilson Ingrid Mazul

With a copy to:

O'Melveny & Myers LLP 30 Rockefeller Plaza

New York, NY 10112

Attn: Nicole J. Macarchuk

Lehman Brothers Purchasers

LEHMAN BROTHERS VENTURE CAPITAL PARTNERS I, L.P.

By: LBI Group Inc., as General Partner

By: /s/ Michael S. Castleman Name: Michael S. Castleman

Its: Vice President

LEHMAN BROTHERS VENTURE

PARTNERS L.P.

By: Lehman Brothers Venture G.P. Partnership

L.P., as General Partner

By: Lehman Brothers Venture Associates Inc.,

as General Partner

By: /s/ Michael S. Castleman

Name: Michael S. Castleman

Its: Vice President

LB I GROUP INC.

By: /s/ Michael S. Castleman

Name: Michael S. Castleman

Its: Vice President

Addresses for Notices for Lehman Purchasers: Michael S. Castleman

Lehman Brothers Inc.

399 Park Avenue, 9th Floor New York, NY 10022 Tel.: (212) 526-2969 Fax: (646) 758-4210

FOUNDERS AND SERIES C-I PURCHASERS

[COMDISCO LETTERHEAD]

FEDERAL EXPRESS

August 16, 2000

David Jones ComScore, Inc. 1761 Business Center Drive

Suite 250 Reston, VA 20190

Re: Preferred Stock Warrant Agreement Dated June 9, 2000 to the Master Lease Agreement Dated June 9, 2000, Equipment Schedule Nos. VL-1 and VL-2 Dated as of June 9, 2000 by and between Comdisco, Inc. ("Warrantholder") and ComScore, Inc. ("Company")

Dear David,

Pursuant to the closing of your Series B Preferred financing on July 5, 2000, this letter is to confirm that Comdisco, Inc., as Warrantholder, hereby agrees that the price per share shall be equal to \$2.90/sh providing the right to purchase 46,551 shares for an aggregate price of \$134,997.90 pursuant to the above referenced warrant.

Except as specifically set forth above, all other terms and conditions of the Warrant shall remain in full force and effect including any adjustments under Section 8.

Please indicate your acceptance by signing in the space provided below and returning to the undersigned and I will have it countersigned and will forward a copy to you to attach to your copy of the warrant. If you have any questions, please do not hesitate to call me at (650) 566-4912.

incerely,	ComScore, Inc.	
s/ Vika Tonga	By:	/s/David B. Jones
/ika Tonga nformation/Document Specialist	Title:	Controller
	Comdisco, Inc.	
	By:	/s/Jill C. Hanses
	Title:	/s/SVP

THESE SECURITIES HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 AS AMENDED, OR ANY STATE SECURITIES LAWS. THEY MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED, OR HYPOTHECATED IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT RELATED THERETO OR AN OPINION OF COUNSEL (WHICH MAY BE COMPANY COUNSEL) REASONABLY SATISFACTORY TO THE COMPANY THAT SUCH REGISTRATION IS NOT REQUIRED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY APPLICABLE STATE SECURITIES LAWS.

WARRANT AGREEMENT

To Purchase Shares of the Series B Preferred Stock, or Upon Certain Terms, the Series A Preferred Stock of COMSCORE, INC.

Dated as of June 9, 2000 (the "Effective Date")

WHEREAS, ComScore, Inc., a Delaware corporation (the "Company") has entered into a Master Lease Agreement dated as of June 9, 2000, Equipment Schedule No. VL-1 and VL-2 dated as of June 9, 2000, and related Summary Equipment Schedules (collectively, the "Leases") with Comdisco, Inc., a Delaware corporation (the "Warrantholder"); and

WHEREAS, the Company desires to grant to Warrantholder, in consideration for such Leases, the right to purchase shares of its Series B Preferred Stock if the Next Round, as defined below, is a private equity financing, or if the Next Round is a Merger Event, as defined below, or an Initial Public Offering, as defined below, then shares of its Series A Preferred Stock;

NOW, THEREFORE, in consideration of the Warrantholder executing and delivering such Leases and in consideration of mutual covenants and agreements contained herein, the Company and Warrantholder agree as follows:

1. GRANT OF THE RIGHT TO PURCHASE PREFERRED STOCK.

For Value received, the Company hereby grants to the Warrantholder, and the Warrantholder is entitled, upon the terms and subject to the conditions hereinafter set forth, to subscribe for and purchase from the Company that number of fully paid and non-assessable shares of the Company's Preferred Stock ("Preferred Stock") equal to One Hundred Thirty-five Thousand Dollars (\$135,000.00) ("Aggregate Purchase Price"), divided by the Exercise Price ("Exercise Price"). In the event the Next Round is a financing as defined in (i) below and successfully completed on or before August 31, 2000, Warrantholder shall have the right to purchase from the Company its Series B Preferred Stock, and the Exercise Price shall be defined as the sum of \$1.00 per share (the "Last Round Price") plus the product of (a) the difference between the price per share of the next round of equity financing (the "Next Round") and the Last Round, multiplied by (b) the fraction resulting from dividing (x) the number of days from the date of closing of the Last Round to the date of closing of the Next Round. Notwithstanding the foregoing, the price per share of the Next Round shall be capped at a Ninety-five Million Dollar Pre-money Valuation. "Nine-five Million Dollar Pre-Money Valuation" shall be calculated by dividing Nine-five Million Dollars (\$95,000,000.00) by the number of fully diluted shares of the Company's authorized Common Stock, Preferred Stock, warrants and options, as converted to Common Stock outstanding immediately prior to the closing of the Next Round. In the event the Next Round is an event as described in (ii) or (iii) below or the Next Round is not successfully completed by August 31, 2000, then Warrantholder shall have the right to purchase 135,000 shares of Series A Preferred Stock from the Company at an Exercise Price of \$1.00 per share. The number and purchase price of such shares are subject to adjustment as provided in Section 8 hereof.

"Next Round" shall be defined as the earlier to occur of (i) preferred stock financing of at least \$2,000,000.00, (ii) the sale, conveyance disposal, or encumbrance of all or substantially all of the Company's property or business or Company's merger into or consolidation with any other corporation (other than a wholly-owned subsidiary corporation) or any other transaction or series of related transactions in which more than fifty percent (50%) of the voting power of Company is disposed of ("Merger Event"), provided that a Merger Event shall not apply to a merger effected exclusively for the purpose of changing the domicile of the company, or (iii) an initial public offering of the Company's Common Stock which such public offering has been declared effective by the SEC.

2. TERM OF THE WARRANT AGREEMENT.

Except as otherwise provided for herein, the term of this Warrant Agreement and the right to purchase Preferred Stock as granted herein shall commence on the Effective Date and shall be exercisable for a period of (i) ten (10) years or (ii) five (5) years from the effective date of the Company's initial public offering, whichever is earlier.

3. EXERCISE OF THE PURCHASE RIGHTS.

(a) Exercise. The purchase rights set forth in this Warrant Agreement are exercisable by the Warrantholder, in whole or in part, at any time, or from time to time, prior to the expiration of the term set forth in Section 2 above, by tendering to the Company at its principal office a notice of exercise in the form attached hereto as Exhibit I (the "Notice of Exercise"), duly completed and executed. Promptly upon receipt of the Notice of Exercise and the payment of the purchase price in accordance with the terms set forth below, and in no event later than twenty-one (21) days thereafter, the Company shall issue to the Warrantholder a certificate for the number of shares of Preferred Stock purchased and shall execute the acknowledgment of exercise in the form attached hereto as Exhibit II (the "Acknowledgment of Exercise") indicating the number of shares which remain subject to future purchases, if any.

The Exercise Price may be paid at the Warrantholder's election either (i) by cash or check, or (ii) by surrender of Warrants ("Net Issuance") as determined below. If the Warrantholder elects the Net Issuance method, the Company will issue Preferred Stock in accordance with the following formula:

$$X = \underbrace{Y(A-B)}_{A}$$

Where: X = the number of shares of Preferred Stock to be issued to the Warrantholder,

Y = the number of shares of Preferred Stock requested to be exercised under this Warrant Agreement.

A = the fair market value of one (1) share of Preferred Stock.

B = the Exercise Price.

For purposes of the above calculation, current fair market value of Preferred Stock shall mean with respect to each share of Preferred Stock:

(i) if the exercise is in connection with an initial public offering of the Company's Common Stock, and if the Company's Registration Statement relating to such public offering has been declared effective by the SEC, then the fair market value per share shall be the product of (x) the initial "Price to Public" specified in the final prospectus with respect to the offering and (y) the number of shares of Common Stock into which each share of Preferred Stock is convertible at the time of such exercise;

(ii) if this Warrant is exercised after, and not in connection with the Company's initial public offering, and:

(a) if traded on a securities exchange, the fair market value shall be deemed to be the product of (x) the average of the closing prices over a five (5) day period ending three days before the day the current fair market value of the securities is being determined and (y) the number of shares of Common Stock into which each share of Preferred Stock is convertible at the time of such exercise; or

(b) if actively traded over-the-counter, the fair market value shall be deemed to be the product of (x) the average of the closing bid and asked prices quoted on the NASDAQ system (or similar system) over the five (5) day period ending three days before the day the current fair market value of the securities is being determined and (y) the number of shares of Common Stock into which each share of Preferred Stock is convertible at the time of such exercise:

(iii) if at any time the Common Stock is not listed on any securities exchange or quoted in the NASDAQ System or the over-the-counter market, the current fair market value of Preferred Stock shall be the product of (x) the highest price per share which the Company could obtain from a willing buyer (not a current employee or director) for shares of Common Stock sold by the Company, from authorized but unissued shares, as determined in good faith by its Board of Directors and (y) the number of shares of

Common Stock into which each share of Preferred Stock is convertible at the time of such exercise, unless the Company shall become subject to a merger, acquisition or other consolidation pursuant to which the Company is not the surviving party, in which case the fair market value of Preferred Stock shall be deemed to be the value received by the holders of the Company's Preferred Stock on a common equivalent basis pursuant to such merger or acquisition.

Upon partial exercise by either cash or Net Issuance, the Company shall promptly issue an amended Warrant Agreement representing the remaining number of shares purchasable hereunder. All other terms and conditions of such amended Warrant Agreement shall be identical to those contained herein, including, but not limited to the Effective Date hereof.

(b) Exercise Prior to Expiration. To the extent this Warrant is not previously exercised as to all Preferred Stock subject hereto, and if the fair market value of one share of the Preferred is greater than the Exercise Price then in effect, this Warrant shall be deemed automatically exercised pursuant to Section 3(a) above (even if not surrendered) immediately before its expiration. For purposes of such automatic exercise, the fair market value of one share of the Preferred Stock upon such expiration shall be determined pursuant to Section 3(a) above. To the extent this Warrant or any portion thereof is deemed automatically exercised pursuant to this Section 3(b), the Company agrees to promptly notify the Warrantholder of the number of Preferred Stock, if any, the Warrantholder is to receive by reason of such automatic exercise.

4. RESERVATION OF SHARES.

During the term of this Warrant Agreement, the Company will at all times have authorized and reserved a sufficient number of shares of its Preferred Stock to provide for the exercise of the rights to purchase Preferred Stock as provided for herein.

5. NO FRACTIONAL SHARES OR SCRIP.

No fractional shares or scrip representing fractional shares shall be issued upon the exercise of the Warrant, but in lieu of such fractional shares the Company shall make a cash payment therefor upon the basis of the Exercise Price then in effect.

6. NO RIGHTS AS SHAREHOLDER.

This Warrant Agreement does not entitle the Warrantholder to any voting rights or other rights as a shareholder of the Company prior to the exercise of the Warrant.

7. WARRANTHOLDER REGISTRY.

The Company shall maintain a registry showing the name and address of the registered holder of this Warrant Agreement.

8. ADJUSTMENT RIGHTS

The purchase price per share and the number of shares of Preferred Stock purchasable hereunder are subject to adjustment, as follows:

(a) Merger and Sale of Assets. If at any time there shall be a capital reorganization of the shares of the Company's stock (other than a combination, reclassification, exchange or subdivision of shares otherwise provided for herein), or a merger or consolidation of the Company with or into another corporation whether or not the Company is the surviving corporation, or the sale of all or substantially all of the Company's properties and assets to any other person (hereinafter referred to as a "Merger Event"), then, as a part of such Merger Event, lawful provision shall be made so that the Warrantholder shall thereafter be entitled to receive, upon exercise of the Warrant, the number of shares of preferred stock or other securities of the successor corporation resulting from such Merger Event, equivalent in value to that which would have been issuable if Warrantholder had exercised this Warrant immediately prior to the Merger Event. In any such case, appropriate adjustment (as determined in good faith by the Company's Board of Directors) shall be made in the application of the provisions of this Warrant Agreement with respect to the rights and interest of the Warrantholder after the Merger Event to the end that the provisions of this Warrant Agreement (including adjustments of the Exercise Price and number of shares of Preferred Stock purchasable) shall be applicable to the greatest extent possible.

- (b) <u>Reclassification of Shares</u>. If the Company at any time shall, by combination, reclassification, exchange or subdivision of securities or otherwise, change any of the securities as to which purchase rights under this Warrant Agreement exist into the same or a different number of securities of any other class or classes, this Warrant Agreement shall thereafter represent the right to acquire such number and kind of securities as would have been issuable as the result of such change with respect to the securities which were subject to the purchase rights under this Warrant Agreement immediately prior to such combination, reclassification, exchange, subdivision or other change.
- (c) <u>Subdivision or Combination of Shares</u>. If the Company at any time shall combine or subdivide its Preferred Stock, the Exercise Price shall be proportionately decreased in the case of a subdivision, or proportionately increased in the case of a combination.
- (d) <u>Stock Dividends</u>. If the Company at any time shall pay a dividend payable in, or make any other distribution (except any distribution specifically provided for in the foregoing subsections (a) or (b)) of the Company's stock, then the Exercise Price shall be adjusted, from and after the record date of such dividend or distribution, to that price determined by multiplying the Exercise Price in effect immediately prior to such record date by a fraction (i) the numerator of which shall be the total number of all shares of the Company's stock outstanding immediately prior to such dividend or distribution, and (ii) the denominator of which shall be the total number of slares of the Company's stock outstanding immediately after such dividend or distribution. The Warrantholder shall thereafter be entitled to purchase, at the Exercise Price resulting from such adjustment, the number of shares of Preferred Stock (calculated to the nearest whole share) obtained by multiplying the Exercise Price in effect immediately prior to such adjustment by the number of shares of Preferred Stock issuable upon the exercise hereof immediately prior to such adjustment and dividing the product thereof by the Exercise Price resulting from such adjustment.
- (e) <u>Right to Purchase Additional Stock</u>. If, the Warrantholder's total cost of equipment leased pursuant to the Leases exceeds \$3,000,000, Warrantholder shall have the right to purchase from the Company, at the Exercise Price (adjusted as set forth herein), an additional number of shares, which number shall be determined by (i) multiplying the amount by which the Warrantholder's total equipment cost exceeds \$3,000,000 by 4.5%, and (ii) dividing the product thereof by the Exercise Price per share referenced above.
- (f) <u>Antidilution Rights</u>. Additional antidilution rights applicable to the Preferred Stock purchasable hereunder are as set forth in the Company's Certificate of Incorporation, as amended through the Effective Date, a true and complete copy of which is attached hereto as **Exhibit IV** (the "Charter"). The Company shall promptly provide the Warrantholder with any restatement, amendment, modification or waiver of the Charter. The Company shall provide Warrantholder with prior written notice of any issuance of its stock or other equity security to occur after the Effective Date of this Warrant, which notice shall include (a) the price at which such stock or security is to be sold, (b) the number of shares to be issued, and (c) such other information as necessary for Warrantholder to determine if a dilutive event has occurred.
- (g) Notice of Adjustments. If: (i) the Company shall declare any dividend or distribution upon its stock, whether in cash, property, stock or other securities; (ii) the Company shall offer for subscription prorata to the holders of any class of its Preferred or other convertible stock any additional shares of stock of any class or other rights; (iii) there shall be any Merger Event; (iv) there shall be an initial public offering; or (v) there shall be any voluntary dissolution, liquidation or winding up of the Company; then, in connection with each such event, the Company shall send to the Warrantholder: (A) at least twenty (20) days' prior written notice of the date on which the bolders of Preferred Stock shall be entitled thereto) or for determining rights to vote in respect of such Merger Event, dissolution, liquidation or winding up; (B) in the case of any such Merger Event, dissolution, liquidation or winding up; at least twenty (20) days' prior written notice of the date when the same shall take place (and specifying the date on which the holders of Preferred Stock shall be entitled to exchange their Preferred Stock for securities or other property deliverable upon such Merger Event, dissolution, liquidation or winding up); and (C) in the case of a public offering, the Company shall give the Warrantholder at least twenty (20) days written notice prior to the effective date

Each such written notice shall set forth, in reasonable detail, (i) the event requiring the adjustment, (ii) the amount of the adjustment, (iii) the method by which such adjustment was calculated, (iv) the Exercise Price, and (v) the number of shares subject to purchase hereunder after giving effect to such adjustment, and shall be given by first class mail, postage prepaid, addressed to the Warrantholder, at the address as shown on the books of the Company.

(h) <u>Timely Notice</u>. Failure to timely provide such notice required by subsection (g) above shall entitle Warrantholder to retain the benefit of the applicable notice period notwithstanding anything to the contrary contained in any insufficient notice received by Warrantholder. The notice period shall begin on the date Warrantholder actually receives a written notice containing all the information specified above.

9. REPRESENTATIONS, WARRANTIES AND COVENANTS OF THE COMPANY.

- (a) Reservation of Preferred Stock. The Preferred Stock issuable upon exercise of the Warrantholder's rights has been duly and validly reserved and, when issued in accordance with the provisions of this Warrant Agreement, will be validly issued, fully paid and non-assessable, and will be free of any taxes, liens, charges or encumbrances of any nature whatsoever: provided, however, that the Preferred Stock issuable pursuant to this Warrant Agreement may be subject to restrictions or transfer under state and/or Federal securities laws. The Company has made available to the Warrantholder true, correct and complete copies of its Charter and Bylaws, as amended. The issuance of certificates for shares of Preferred Stock upon exercise of the Warranth Agreement shall be made without charge to the Warrantholder for any issuance tax in respect thereof, or other cost incurred by the Company in connection with such exercise and the related issuance of shares of Preferred Stock. The Company shall not be required to pay any tax which may be payable in respect of any transfer involved and the issuance and delivery of any certificate in a name other than that of the Warrantholder.
- (b) <u>Due Authority</u>. The execution and delivery by the Company of this Warrant Agreement and the performance of all obligations of the Company hereunder, including the issuance to Warrantholder of the right to acquire the shares of Preferred Stock, have been duly authorized by all necessary corporate action on the part of the Company, and the Leases and this Warrant Agreement are not inconsistent with the Company's Charter or Bylaws, do not contravene any law or governmental rule, regulation or order applicable to it, do not and will not contravene any provision of, or constitute a default under, any indenture, mortgage, contract or other instrument to which it is a party or by which it is bound, and the Leases and this Warrant Agreement constitute legal, valid and binding agreements of the Company, enforceable in accordance with their respective terms.
- (c) <u>Consents and Approvals</u>. No consent or approval of, giving of notice to, registration with, or taking of any other action in respect of any state, Federal or other governmental authority or agency is required with respect to the execution, delivery and performance by the Company of its obligations under this Warrant Agreement, except for the filing of notices pursuant to Regulation D under the 1933 Act and any filing required by applicable state securities law, which filings will be effective by the time required thereby.
- (d) <u>Issued Securities</u>. All issued and outstanding shares of Common Stock, Preferred Stock or any other securities of the Company have been duly authorized and validly issued and are fully paid and nonassessable. All outstanding shares of Common Stock, Preferred Stock and any other securities were issued in full compliance with all Federal and state securities laws. In addition, as of the date hereof:
 - (i) The authorized capital of the Company consists of (A) 50,000,000 shares of Common Stock, of which 12,122,396 shares are issued and outstanding, and (B) 9,187,500 shares of Series A Preferred Stock, of which 9,187,500 shares are issued and outstanding and are convertible into 9,187,500 shares of Common Stock at \$1.00 per share.
 - (ii) The Company has reserved 4,210,937 shares of Common Stock for issuance under its 1999 Stock Plan, under which 3,411,200 options are outstanding at an average price of \$0.10 per share. There are no other options, warrants, conversion privileges or other rights presently outstanding to purchase or otherwise acquire any authorized but unissued shares of the Company's capital stock or other securities of the Company.
 - (iii) In accordance with the Company's Certificate of Incorporation, no shareholder of the Company has preemptive rights to purchase new issuances of the Company's capital stock.
- (e) <u>Insurance</u>. The Company has in full force and effect insurance policies, with extended coverage, insuring the Company and its property and business against such losses and risks, and in such amounts, as are customary for corporations engaged in a similar business and similarly situated and as otherwise may be required pursuant to the terms of any other contract or agreement.
- (f) Other Commitments to Register Securities. Except as set forth in this Warrant Agreement, the Company is not, pursuant to the terms of any other agreement currently in existence, under any obligation to register under the 1933 Act any of its presently outstanding securities or any of its securities which may hereafter be issued.

- (g) Exempt Transaction. Subject to the accuracy of the Warrantholder's representations in Section 10 hereof, the issuance of the Preferred Stock upon exercise of this Warrant will constitute a transaction exempt from (i) the registration requirements of Section 5 of the 1933 Act, in reliance upon Section 4(2) thereof, and (ii) the qualification requirements of the applicable state securities laws.
- (h) <u>Compliance with Rule 144</u>. At the written request of the Warrantholder, who proposes to sell Preferred Stock issuable upon the exercise of the Warrant in compliance with Rule 144 promulgated by the Securities and Exchange Commission, the Company shall furnish to the Warrantholder, within ten days after receipt of such request, a written statement confirming the Company's compliance with the filing requirements of the Securities and Exchange Commission as set forth in such Rule, as such Rule may be amended from time to time.

10. REPRESENTATIONS AND COVENANTS OF THE WABRANTHOLDER.

This Warrant Agreement has been entered into by the Company in reliance upon the following representations and covenants of the Warrantholder:

- (a) <u>Investment Purpose</u>. The right to acquire Preferred Stock or the Preferred Stock issuable upon exercise of the Warrantholder's rights contained herein will be acquired for investment and not with a view to the sale or distribution of any part thereof, and the Warrantholder has no present intention of selling or engaging in any public distribution of the same except pursuant to a registration or exemption.
- (b) <u>Private Issue</u>. The Warrantholder understands (i) that the Preferred Stock issuable upon exercise of this Warrant is not registered under the 1933 Act or qualified under applicable state securities laws on the ground that the issuance contemplated by this Warrant Agreement will be exempt from the registration and qualifications requirements thereof, and (ii) that the Company's reliance on such exemption is predicated on the representations set forth in this Section 10.
- (c) <u>Disposition of Warrantholder's Rights</u>. In no event will the Warrantholder make a disposition of any of its rights to acquire Preferred Stock or Preferred Stock issuable upon exercise of such rights unless and until (i) it shall have notified the Company of the proposed disposition, and (ii) if requested by the Company, it shall have furnished the Company with an opinion of counsel (which counsel may either be inside or outside counsel to the Warrantholder) satisfactory to the Company and its counsel to the effect that (A) appropriate action necessary for compliance with the 1933 Act has been taken, or (B) an exemption from the exercise of such rights do not apply to transfers from the beneficial owner of any of the aforementioned securities to its nominee or from such nominee to its beneficial owner, and shall terminate as to any particular share of Preferred Stock when (1) such security shall have been effectively registered under the 1933 Act and sold by the holder thereof in accordance with such registration or (2) such security shall have been sold without registration in compliance with Rule 144 under the 1933 Act, or (3) a letter shall have been issued to the Warrantholder at its request by the staff of the Securities and Exchange Commission or a ruling shall have been issued to the Warrantholder at its request by such Commission, as the case may be, if such security is transferred without registration under the 1933 Act in accordance with the conditions set forth in such letter or ruling specifies that no subsequent restrictions on transfer are required. Whenever the restrictions imposed hereunder shall terminate, as hereinabove provided, the Warrantholder of a share of Preferred Stock then outstanding as to which such restrictions have terminated shall be entitled to receive from the Company, without expense to such holder, one or more new certificates for the Warrant or for such shares of Preferred Stock not bearing any restrictive legend.
- (d) <u>Financial Risk</u>. The Warrantholder has such knowledge and experience in financial and business matters as to be capable of evaluating the merits and risks of its investment, and has the ability to bear the economic risks of its investment.
- (e) <u>Risk of No Registration</u>. The Warrantholder understands that if the Company does not register with the Securities and Exchange Commission pursuant to Section 12 of the 1934 Act (the "1934 Act"), or file reports pursuant to Section 15(d), of the 1934 Act, or if a registration statement covering the securities under the 1933 Act is not in effect when it desires to sell (i) the rights to purchase Preferred Stock pursuant to this Warrant Agreement, or (ii) the Preferred Stock issuable upon exercise of the right to purchase, it may be required to hold such securities for an indefinite period. The Warrantholder also understands that any sale of its rights of the Warrantholder to purchase Preferred Stock or Preferred Stock which might be made by it in reliance upon Rule 144 under the 1933 Act may be made only in accordance with the terms and conditions of that Rule

(f) Accredited Investor. Warrantholder is an "accredited investor" within the meaning of the Securities and Exchange Rule 501 of Regulation D, as presently in effect.

11. TRANSFERS.

Subject to the terms and conditions contained in Section 10 hereof, this Warrant Agreement and all rights hereunder are transferable in whole or in part by the Warrantholder and any successor transferee, provided, however, in no event shall the number of transfers of the rights and interests in all of the Warrants exceed three (3) transfers. The transfer shall be recorded on the books of the Company upon receipt by the Company of a notice of transfer in the form attached hereto as **Exhibit III** (the "Transfer Notice"), at its principal offices and the payment to the Company of all transfer taxes and other governmental charges imposed on such transfer.

12. MISCELLANEOUS.

- (a) <u>Effective Date</u>. The provisions of this Warrant Agreement shall be construed and shall be given effect in all respects as if it had been executed and delivered by the Company on the date hereof. This Warrant Agreement shall be binding upon any successors or assigns of the Company.
- (b) <u>Attorney's Fees</u>. In any litigation, arbitration or court proceeding between the Company and the Warrantholder relating hereto, the prevailing party shall be entitled to attorneys' fees and expenses and all costs of proceedings incurred in enforcing this Warrant Agreement.
 - (c) Governing Law. This Warrant Agreement shall be governed by and construed for all purposes under and in accordance with the laws of the State of Illinois.
 - (d) Counterparts. This Warrant Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- (e) <u>Notices</u>. Any notice required or permitted hereunder shall be given in writing and shall be deemed effectively given upon personal delivery, facsimile transmission (provided that the original is sent by personal delivery or mail as hereinafter set forth) or seven (7) days after deposit in the United States mail, by registered or certified mail, addressed (i) to the Warrantholder at 6111 North River Road, Rosemont, Illinois 60018, <u>Attention</u>: Venture Lease Administration, cc: Legal Department, Attention: General Counsel, (and/or, if by Facsimile, (847) 518-5465 and (847)518-588) and (ii) to the Company at 1761 Business Center Drive, Suite 250, Reston, CA 20190, <u>Attention</u>: David Jones (and/or if by Facsimile, (703) 438-2091) or at such other address as any such party may subsequently designate by written notice to the other party.
- (f) <u>Remedies</u>. In the event of any default hereunder, the non-defaulting party may proceed to protect and enforce its rights either by suit in equity and/or by action at law, including but not limited to an action for damages as a result of any such default, and/or an action for specific performance for any default where Warrantholder will not have an adequate remedy at law and where damages will not be readily ascertainable. The Company expressly agrees that it shall not oppose an application by the Warrantholder or any other person entitled to the benefit of this Agreement requiring specific performance of any or all provisions hereof or enjoining the Company from continuing to commit any such breach of this Agreement.
- (g) No Impairment of Rights. The Company will not, by amendment of its Charter or through any other means, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, but will at all times in good faith assist in the carrying out of all such terms and in the taking of all such actions as may be necessary or appropriate in order to protect the rights of the Warrantholder against impairment.
- (h) <u>Survival</u>. The representations, warranties, covenants and conditions of the respective parties contained herein or made pursuant to this Warrant Agreement shall survive the execution and delivery of this Warrant Agreement.
- (i) <u>Severability</u>. In the event any one or more of the provisions of this Warrant Agreement shall for any reason be held invalid, illegal or unenforceable, the remaining provisions of this Warrant Agreement shall be unimpaired, and the invalid, illegal or unenforceable provision shall be replaced by a mutually acceptable valid, legal and enforceable provision, which comes closest to the intention of the parties underlying the invalid illegal or unenforceable provision.
 - (j) Amendments. Any provision of this Warrant Agreement may be amended by a written instrument signed by the Company and by the Warrantholder.

(k) Additional Documents. The Company, upon execution of this Warrant Agreement, shall provide the Warrantholder with certified resolutions with respect to the representations, warranties and covenants set forth in subparagraphs (a) through (d), (f) and (g) of Section 9 above. If the purchase price for the Leases referenced in the preamble of this Warrant Agreement exceeds \$1,000,000, the Company will also provide Warrantholder with an opinion from the Company's counsel with respect to those same representations, warranties and covenants. The Company shall also supply such other documents as the Warrantholder may from time to time reasonably request.

IN WITNESS WHEREOF, the parties hereto have caused this Warrant Agreement to be executed by its officers thereunto duty authorized as of the Effective Date.

COMPANY:	COMSCO	DRE, INC.
	By:	/s/Magid Abraham
	Title:	CEO
WARRANTHOLDER:	COMDIS	со
	By:	/s/Jill C. Hanses
	Title:	SVP
	Ω	

COMMON STOCK PURCHASE WARRANT

THIS WARRANT AND THE SHARES OF COMMON STOCK WHICH MAY BE PURCHASED PURSUANT TO THE EXERCISE OF THIS WARRANT HAVE BEEN ACQUIRED SOLELY FOR INVESTMENT AND HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR ANY STATE SECURITIES LAWS. SUCH SECURITIES MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED OR HYPOTHECATED IN THE ABSENCE OF SUCH REGISTRATION OR AN OPINION OF COUNSEL SATISFACTORY TO THE COMPANY AND ITS COUNSEL THAT SUCH SALE, OFFER, PLEDGE OR HYPOTHECATION IS EXEMPT FROM THE REGISTRATION AND PROSPECTUS DELIVERY REQUIREMENTS OF THE ACT AND OF ANY APPLICABLE STATE SECURITIES LAWS UNLESS SOLD PURSUANT TO RULE 144 OF THE ACT.

Void After July 31, 2010

COMSCORE NETWORKS, INC.

WARRANT TO PURCHASE SHARES OF COMMON STOCK

For value received, Kenneth Leiner (the "Holder") is entitled to subscribe for and purchase up to 20,100 shares (as adjusted pursuant to Section 3 hereof) of the Common Stock (the "Common Stock"), \$0.001 par value (the "Shares"), of comScore Networks, inc., a Delaware corporation (the "Company"), at the price of \$2.50 per share (the "Exercise Price") (as adjusted pursuant to Section 3 hereof), subject to the provisions and upon the terms and conditions hereinafter set forth.

1. Exercise and Payment.

- 1.1 Exercise. The purchase rights represented by this Warrant may be exercised by the Holder, in whole or in part, by the surrender of a duly executed exercise notice in the form attached hereto as **Exhibit A** at the principal office of the Company, and by the payment to the Company, by check or wire transfer, of an amount equal to the aggregate Exercise Price of the shares being purchased.
- 1.2 <u>Stock Certificate</u>. In the event of the exercise of this Warrant, a certificate for the shares of Common Stock so purchased shall be delivered to the Holder within a reasonable time, which shall in no event be later than thirty (30) days thereafter.
- 2. <u>Stock Fully Paid; Reservation of Shares</u>. All of the Shares issuable upon the exercise of this Warrant will, upon issuance and receipt of the Exercise Price therefor, be fully paid and nonassessable. During the period within which this Warrant may be exercised, the Company shall at all times have authorized and reserved for issuance sufficient shares of its Common Stock to provide for the exercise of this Warrant.
 - 3. Adjustment of Exercise Price and Number of Shares. Subject to Section 9 hereof, the number and kind of securities purchasable upon the exercise of this Warrant and the Exercise Price

therefor shall be subject to adjustment from time to time upon the occurrence of certain events, as follows:

- 3.1 <u>Reclassification, Consolidation or Merger</u>. In case of any reclassification or change of the Common Stock (other than a change in par value, or as a result of a subdivision or combination), or in case of any Merger Event (as defined herein), the Company or the successor corporation, as the case may be, shall execute a new warrant, providing that the Holder shall have the right to exercise such new warrant, and procure upon such exercise and payment of the same aggregate Exercise Price, in lieu of the shares of Common Stock theretofore issuable upon exercise of this Warrant, the kind and amount of shares of stock, other securities, money and property receivable upon such reclassification, change, or Merger Event by a holder of an equivalent number of shares of Common Stock. Such new warrant shall provide for adjustments which shall be as nearly equivalent as may be practicable to the adjustments provided for in this Section 3.
- 3.2 Stock Splits, Dividends and Combinations. In the event that the Company shall at any time subdivide the outstanding shares of Common Stock, or shall issue a stock dividend on its outstanding shares of Common Stock, the number of Shares issuable upon exercise of this Warrant immediately prior to such subdivision or to the issuance of such stock dividend shall be proportionately increased and the Exercise Price shall be proportionately decreased so that the Holder of the Warrant after such time shall be entitled to receive the number of shares of Common Stock which such Holder would have owned or been entitled to receive had such Warrant been exercised immediately prior to such event, and in the event that the Company shall at any time combine the outstanding shares of Common Stock, the number of Shares issuable upon exercise of this Warrant immediately prior to such combination shall be proportionately decreased and the Exercise Price shall be proportionately increased so that the Holder of the Warrant after such time shall be entitled to receive the number of shares of Common Stock which such Holder would have owned or been entitled to receive had such Warrant been exercised prior to such event, in either case effective at the close of business on the date of such subdivision, stock dividend or combination, as the case may be.
- 4. <u>Notice of Adjustments</u>. In the event that: (i) the Company shall declare any dividend or distribution upon its Common Stock, whether in cash, property, stock or other securities; (ii) the Company shall offer for subscription pro rata to the holders of its Common Stock any additional shares of stock of any class or other rights; (iii) there shall be any merger or consolidation of the Company with or into a third party pursuant to which the Company's stockholders prior to the transaction own less than fifty percent (50%) of the surviving entity or the sale of all or substantially all of the assets of the Company (a "Merger Event"); or (iv) there shall be any voluntary or involuntary dissolution, liquidation or winding up of the Company; then, in connection with each such event, the Company shall send to the Holder:
- (a) At least ten (10) days prior written notice of the date on which the books of the Company shall close or a record shall be taken for such dividend, distribution, subscription rights (specifying the date on which the holders of Common Stock shall be entitled thereto) or for determining rights to vote in respect of such Merger Event, dissolution, liquidation or winding up; and
 - (b) In the case of any such Merger Event, dissolution, liquidation or winding up, at least ten (10) days prior written notice of the date when the same shall take place (and specifying the date

on which the holders of Common Stock shall be entitled to exchange their Common Stock for securities or other property deliverable upon such Merger Event, dissolution, liquidation or winding up).

Each such written notice shall set forth, in reasonable detail, (i) the event requiring the adjustment, (ii) the amount of the adjustment, (iii) the method by which such adjustment was calculated, (iv) the Exercise Price, and (v) the number of shares subject to purchase hereunder after giving effect to such adjustment, and shall be given by first class mail, postage prepaid, addressed to the Holder, at the address as shown on the books of the Company.

- 5. <u>Fractional Shares</u>. No fractional shares of Common Stock will be issued in connection with any exercise hereunder. In lieu of such fractional shares the Company shall make a cash payment therefor based upon the Exercise Price then in effect.
- 6. Representations and Warranties of the Holder. The Holder hereby represents and warrants to the Company, with respect to its acquisition of the Warrant, as follows:
- 6.1 Experience. The Holder has sufficient knowledge and experience in financial and business matters so that he is capable of evaluating the merits and risks of his investment in the Company and has the capacity to protect his own interests.
- 6.2 <u>Investment</u>. The Holder is acquiring the Warrant and the Shares for investment for its own account, not as a nominee or agent, and not with the view to, or for resale in connection with, any distribution thereof. The Holder understands that the Warrant and the Shares have not been, and will not be, registered under the Act by reason of a specific exemption from the registration provisions of the Act, the availability of which depends upon, among other things, the bona fide nature of the investment intent and the accuracy of such Holder's representations as expressed herein.
- 6.3 Rule 144. The Holder acknowledges that the Warrant and the Shares must be held indefinitely unless subsequently registered under the Act or unless an exemption from such registration is available. The Holder is aware of the provisions of Rule 144 promulgated under the Act which permit limited resale of shares purchased in a private placement subject to the satisfaction of certain conditions, including, among other things, the existence of a public market for the shares, the availability of certain current public information about the Company, the resale occurring not less than one (1) year after a party has purchased and paid for the security to be sold, the sale being effected through a "broker's transaction" or in transactions directly with a "market maker" and the number of shares being sold during any three-month period not exceeding specified limitations.
- 6.4 No Public Market. The Holder understands that no public market now exists for any of the securities issued by the Company and that the Company has made no assurances that a public market will ever exist for the Company's securities.
 - 6.5 No Solicitation. The Holder knows of no public solicitation or advertisement of an offer in connection with the proposed issuance and sale of the Warrant or the Shares.
 - 6.6 Residence. The residence of the Holder for securities law purposes is set forth herein on page 6.

7. Restrictions on Transfer.

7.1 <u>Restrictive Legend</u>. Each certificate representing (i) the Shares and (ii) any other securities issued in respect of the Shares upon any stock split, stock dividend, recapitalization, merger, consolidation or similar event, (collectively, the "**Restricted Securities**") shall (unless otherwise permitted by the provisions of Section 7.2 below) be stamped or otherwise imprinted with a legend in substantially the following form (in addition to any legend required under applicable state securities laws);

THE SHARES REPRESENTED BY THIS CERTIFICATE HAVE BEEN ACQUIRED FOR INVESTMENT AND HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933. THESE SHARES MAY NOT BE SOLD OR TRANSFERRED IN THE ABSENCE OF SUCH REGISTRATION OR AN EXEMPTION THEREFROM UNDER SAID ACT.

7.2 Notice of Proposed Transfers. The holder of each certificate representing Restricted Securities by acceptance thereof agrees to comply in all respects with the provisions of this Section 7. Prior to any proposed transfer of any Restricted Securities, unless there is in effect a registration statement under the Act covering the proposed transfer, the holder thereof shall give written notice to the Company of such Holder's intention to effect such transfer. Each such notice shall describe the manner and circumstances of the proposed transfer in sufficient detail, and shall, if the Company so requests, be accompanied by either (i) an unqualified written opinion of legal counsel who shall be reasonably satisfactory to the Company, addressed to the Company and reasonably satisfactory in form and substance to the Company's counsel, to the effect that the proposed transfer of the Restricted Securities may be effected without registration under the Act, or (ii) a "No Action" letter from the Securities and Exchange Commission (the "Commission") to the effect that the transfer of such securities without registration will not result in a recommendation by the staff of the Commission that action be taken with respect thereto, whereupon the holder of such Restricted Securities shall be entitled to transfer such Restricted Securities in accordance with the terms of the notice delivered by the holder to the Company; provided, however, that no opinion or No Action letter need be obtained with respect to a transfer to (A) the immediate family of Holder upon the Holder's death, by will or intestacy, or to a trust for the benefit of the Holder's immediate family, (B) a partner, active or retired, of a holder of Restricted Securities, (C) the estate of any such partner, or (D) an "affiliate" of a holder of Restricted Securities as that term is defined in Rule 405 promulgated by the Commission under the Act, provided that in such cases the transferree agrees in writing to be subject to the terms hereof. Each certificate evidenc

8. <u>Rights of Stockholders</u>. No holder of this Warrant shall be entitled, as a Warrant holder, to vote or receive dividends or be deemed the holder of Common Stock or any other securities of the Company which may at any time be issuable on the exercise hereof for any purpose, nor shall anything contained herein be construed to confer upon the Holder, as such, any of the rights of a stockholder of the Company or any right to vote for the election of directors or upon any matter submitted to

stockholders at any meeting thereof, or to give or withhold consent to any corporate action (whether upon any recapitalization, issuance of stock, reclassification of stock, change of par value, consolidation, merger, conveyance, or otherwise) or to receive notice of meetings, or to receive dividends or subscription rights or otherwise until the Warrant shall have been exercised and the Shares purchasable upon the exercise hereof shall have become deliverable, as provided herein.

- 9. Expiration of Warrant. Notwithstanding any other provision of this Warrant, this Warrant shall expire and shall no longer be exercisable at 5:00 p.m., California time, on July 31, 2010.
- 10. Miscellaneous.
 - 10.1 Governing Law. This Agreement shall be governed in all respects by the laws of the State of Delaware.
- 10.2 <u>Successors and Assigns</u>. Except as otherwise provided herein, the provisions hereof shall inure to the benefit of, and be binding upon, the successors, assigns, heirs, executors, and administrators of the Company and the Holder.
- 10.3 Entire Agreement; Amendment. This Warrant constitutes the full and entire understanding and agreement between the parties with regard to the subjects hereof. Neither this Warrant nor any term hereof may be amended, waived, discharged, or terminated other than by a written instrument signed by the party against whom enforcement of any such amendment, waiver, discharge or termination is sought.
- 10.4 Notices, etc.. All notices and other communications required or permitted hereunder shall be in writing and shall be deemed effectively given upon delivery to the party to be notified in person or by courier service or by registered or certified mail, postage prepaid, addressed (a) to the Holder, at the address set forth on the last page of this Warrant or at such other address as such Holder shall have furnished the Company in writing, or (b) if to the Company, at the address set forth on the last page of this Warrant and addressed to the attention of the Chief Executive Officer, or at such other address as the Company shall have furnished to the Holder.

Issued this 31st day of July, 2000

COMSCORE NETWORKS, INC.

By:	/s/ Magid Abraham
Title:	C.E.O.
Kenneth	Leiner
Address:	17724 Lisa Drive Derwood, MD 20855
By:	
Title:	
6	

1. The undersigned hereby elects to purchase shares of Common Stock (the "Common Stock") of COMSCORE NETWORKS, INC. pursuant to the terms of the attached W

- 2. The undersigned elects to exercise the attached Warrant and tenders herewith payment in full for the purchase price of the shares being purchased, together with all applicable transfer taxes, if any.
- 3. Please issue a certificate or certificates representing said shares of Common Stock in the name of the undersigned or in such other name as is specified below:

(Name)

(Address)

4. The undersigned hereby represents and warrants that the aforesaid shares of Common Stock are being acquired for the account of the undersigned for investment and not with a view to, or for resale, in connection with the distribution thereof, and that the undersigned has no present intention of distributing or reselling such shares. In support thereof, the undersigned has executed and delivered herewith an Investment Representation Statement in substantially the form attached to the Warrant as **Exhibit B**.

(Signature)

Title:

(Date)

1

THESE SECURITIES HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 AS AMENDED, OR ANY STATE SECURITIES LAWS. THEY MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED, OR HYPOTHECATED IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT RELATED THERETO OR AN OPINION OF COUNSEL (WHICH MAY BE COMPANY COUNSEL) REASONABLY SATISFACTORY TO THE COMPANY THAT SUCH REGISTRATION IS NOT REQUIRED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY APPLICABLE STATE SECURITIES LAWS.

WARRANT AGREEMENT

To Purchase Shares of the Series B Preferred Stock of

COMSCORE NETWORKS, INC.

Dated as of September 29, 2000 (the "Effective Date")

WHEREAS, ComScore Networks, Inc. a Delaware corporation (the "Company") has entered into a Master Lease Agreement dated as of June 9, 2000, Equipment Schedule No. VL-3 and VL-4 dated as of September 29, 2000, and related Summary Equipment Schedules (collectively, the "Leases") with Comdisco, Inc., a Delaware corporation (the "Warrantholder"); and

WHEREAS, the Company desires to grant to Warrantholder, in consideration for such Leases, the right to purchase shares of its Series B Preferred Stock;

NOW, THEREFORE, in consideration of the Warrantholder executing and delivering such Leases and in consideration of mutual covenants and agreements contained herein, the Company and Warrantholder agree

1. GRANT OF THE RIGHT TO PURCHASE PREFERRED STOCK.

The Company hereby grants to the Warrantholder, and the Warrantholder is entitled, upon the terms and subject to the conditions hereinafter set forth, to subscribe for and purchase from the Company, 9,694 fully paid and non-assessable shares of the Company's Series B Preferred Stock ("Preferred Stock") at a purchase price of \$4.90 per share (the "Exercise Price"). The number and purchase price of such shares are subject to adjustment as provided in Section 8 hereof.

2. TERM OF THE WARRANT AGREEMENT

Except as otherwise provided for herein, the term of this Warrant Agreement and the right to purchase Preferred Stock as granted herein shall commence on the Effective Date and shall be exercisable for a period of (i) ten (10) years or (ii) five (5) years from the effective date of the Company's public offering, whichever is earlier.

3. EXERCISE OF THE PURCHASE RIGHTS.

(a) Exercise. The purchase rights set forth in this Warrant Agreement are exercisable by the Warrantholder, in whole or in part, at any time, or from time to time, prior to the expiration of the term set forth in Section 2 above, by tendering to the Company at its principal office a notice of exercise in the form attached hereto as Exhibit I (the "Notice of Exercise"), duly completed and executed. Promptly upon receipt of the Notice of Exercise and the payment of the purchase price in accordance with the terms set forth below, and in no event later than twenty-one (21) days thereafter, the Company shall issue to the Warrantholder a certificate for the number of shares of Preferred Stock purchased and shall execute the acknowledgment of exercise in the form attached hereto as Exhibit II (the "Acknowledgment of Exercise") indicating the number of shares which remain subject to future purchases, if any.

The Exercise Price may be paid at the Warrantholder's election either (i) by cash or check, or (ii) by surrender of Warrants ("Net Issuance") as determined below. If the Warrantholder elects the Net Issuance method, the Company will issue Preferred Stock in accordance with the following formula:

$$X = \underline{Y(A-B)}$$

Where: X = the number of shares of Preferred Stock to be issued to the Warrantholder

Y = the number of shares of Preferred Stock requested to be exercised under this Warrant Agreement.

A = the fair market value of one (1) share of Preferred Stock.

B = the Exercise Price.

For purposes of the above calculation, current fair market value of Preferred Stock shall mean with respect to each share of Preferred Stock:

- (i) if the exercise is in connection with an initial public offering of the Company's Common Stock, and if the Company's Registration Statement relating to such public offering has been declared effective by the SEC, then the fair market value per share shall be the product of (x) the initial "Price to Public" specified in the final prospectus with respect to the offering and (y) the number of shares of Common Stock into which each share of Preferred Stock is convertible at the time of such exercise;
 - (ii) if this Warrant is exercised after, and not in connection with the Company's initial public offering, and:
 - (a) if traded on a securities exchange, the fair market value shall be deemed to be the product of (x) the average of the closing prices over a five (5) day period ending three days before the day the current fair market value of the securities is being determined and (y) the number of shares of Common Stock into which each share of Preferred Stock is convertible at the time of such exercise; or
 - (b) if actively traded over-the-counter, the fair market value shall be deemed to be the product of (x) the average of the closing bid and asked prices quoted on the NASDAQ system (or similar system) over the five (5) day period ending three days before the day the current fair market value of the securities is being determined and (y) the number of shares of Common Stock into which each share of Preferred Stock is convertible at the time of such exercise:
- (iii) if at any time the Common Stock is not listed on any securities exchange or quoted in the NASDAQ System or the over-the-counter market, the current fair market value of Preferred Stock shall be the product of (x) the highest price per share which the Company could obtain from a willing buyer (not a current employee or director) for shares of Common Stock sold by the Company, from authorized but unissued shares, as determined in good faith by its Board of Directors and (y) the number of shares of Common Stock into which each share of Preferred Stock is convertible at the time of such exercise, unless the Company shall become subject to a merger, acquisition or other consolidation pursuant to which the Company is not the surviving party, in which case the fair market value of Preferred Stock shall be deemed to be the value received by the holders of the Company's Preferred Stock on a common equivalent basis pursuant to such merger or acquisition.

Upon partial exercise by either cash or Net Issuance, the Company shall promptly issue an amended Warrant Agreement representing the remaining number of shares purchasable hereunder. All other terms and conditions of such amended Warrant Agreement shall be identical to those contained herein, including but not limited to the Effective Date hereof.

(b) Exercise Prior to Expiration. To the extent this Warrant is not previously exercised as to all Preferred Stock subject hereto, and if the fair market value of one share of the Preferred is greater than the Exercise Price then in effect, this Warrant shall be deemed automatically exercised pursuant to Section 3(a) above (even if not surrendered) immediately before its expiration. For purposes of such automatic exercise, the fair market value of one share of the Preferred Stock upon such expiration shall be determined pursuant to Section 3(a) above. To the extent this Warrant or any portion thereof is deemed automatically exercised pursuant to this Section 3(b), the Company agrees to promptly notify the Warrantholder of the number of Preferred Stock, if any, the Warrantholder is to receive by reason of such automatic exercise.

4. RESERVATION OF SHARES

During the term of this Warrant Agreement, the Company will at all times have authorized and reserved a sufficient number of shares of its Preferred Stock to provide for the exercise of the rights to purchase Preferred Stock as provided for herein.

5. NO FRACTIONAL SHARES OR SCRIP.

No fractional shares or scrip representing fractional shares shall be issued upon the exercise of the Warrant, but in lieu of such fractional shares the Company shall make a cash payment therefor upon the basis of the Exercise Price then in effect.

6. NO RIGHTS AS SHAREHOLDER.

This Warrant Agreement does not entitle the Warrantholder to any voting rights or other rights as a shareholder of the Company prior to the exercise of the Warrant.

7. WARRANTHOLDER REGISTRY.

The Company shall maintain a registry showing the name and address of the registered holder of this Warrant Agreement.

8. ADJUSTMENT RIGHTS.

The purchase price per share and the number of shares of Preferred Stock purchasable hereunder are subject to adjustment, as follows:

- (a) Merger and Sale of Assets. If at any time there shall be a capital reorganization of the shares of the Company's stock (other than a combination, reclassification, exchange or subdivision of shares otherwise provided for herein), or a merger or consolidation of the Company with or into another corporation whether or not the Company is the surviving corporation, or the sale of all or substantially all of the Company's properties and assets to any other person (hereinafter referred to as a "Merger Event"), than, as a part of such Merger Event, lawful provision shall be made so that the Warrantholder shall thereafter be entitled to receive, upon exercise of the Warrant, the number of shares of preferred stock or other securities of the successor corporation resulting from such Merger Event, equivalent in value to that which would have been issuable if Warrantholder had exercised this Warrant immediately prior to the Merger Event. In any such case, appropriate adjustment (as determined in good faith by the Company's Board of Directors) shall be made in the application of the provisions of this Warrant Agreement with respect to the rights and interest of the Warrantholder after the Merger Event to the end that the provisions of this Warrant Agreement (including adjustments of the Exercise Price and number of shares of Preferred Stock purchasable) shall be applicable to the greatest extent possible.
- (b) Reclassification of Shares. If the Company at any time shall, by combination, reclassification, exchange or subdivision of securities or otherwise, change any of the securities as to which purchase rights under this Warrant Agreement exist into the same or a different number of securities or any other class or classes, this Warrant Agreement shall thereafter represent the right to acquire such number and kind of securities as would have been issuable as the result of such change with respect to the securities which were subject to the purchase rights under this Warrant Agreement immediately prior to such combination, reclassification, exchange, subdivision or other change.
- (c) <u>Subdivision or Combination of Shares</u>. If the Company at any time shall combine or subdivide its Preferred Stock, the Exercise Price shall be proportionately decreased in the case of a subdivision, or proportionately increased in the case of a combination
- (d) <u>Stock Dividends</u>. If the Company at any time shall pay a dividend payable in, or make any other distribution (except any distribution specifically provided for in the foregoing subsections (a) or (b)) of the Company's stock, then the Exercise Price shall be adjusted, from and after the record date of such dividend or distribution, to that price determined by multiplying the Exercise Price in effect immediately prior to such record date by a fraction (i) the numerator of which shall be the total number of all shares of the Company's stock outstanding immediately prior to such dividend or distribution, and (ii) the denominator of which shall be the total number of all shares of the Company's stock outstanding immediately after such dividend or distribution. The

Warrantholder shall thereafter be entitled to purchase, at the Exercise Price resulting from such adjustment, the number of shares of Preferred Stock (calculated to the nearest whole share) obtained by multiplying the Exercise Price in effect immediately prior to such adjustment by the number of shares of Preferred Stock issuable upon the exercise hereof immediately prior to such adjustment and dividing the product thereof by the Exercise Price resulting from such adjustment.

- (e) <u>Right to Purchase Additional Stock</u>. If, the Warrantholder's total cost of equipment leased pursuant to the Leases exceeds \$1,000,000, Warrantholder shall have the right to purchase from the Company, at the Exercise Price (adjusted as set forth herein), an additional number of shares, which number shall be determined by (i) multiplying the amount by which the Warrantholder's total equipment cost exceeds \$1,000,000 by 4.75%, and (ii) dividing the product thereof by the Exercise Price per share referenced above.
- (f) <u>Antidilution Rights</u>. Additional antidilution rights applicable to the Preferred Stock purchasable hereunder are as set forth in the Company's Certificate of Incorporation, as amended through the Effective Date, a true and complete copy of which is attached hereto as **Exhibit IV** (the "Charter"). The Company shall promptly provide the Warrantholder with any restatement, amendment, modification or waiver of the Charter. The Company shall provide Warrantholder with prior written notice of any issuance of its stock or other equity security to occur after the Effective Date of this Warrant, which notice shall include (a) the price at which such stock or security is to be sold, (b) the number of shares to be issued, and (c) such other information as necessary for Warrantholder to determine if a dilutive event has occurred.
- (g) Notice of Adjustments. If: (i) the Company shall declare any dividend or distribution upon its stock, whether in cash, property, stock or other securities; (ii) the Company shall offer for subscription prorata to the holders of any class of its Preferred or other convertible stock any additional shares of stock of any class or other rights, (iii) there shall be any Merger Event; (iv) there shall be an initial public offering; or (v) there shall be any voluntary dissolution, liquidation or winding up of the Company; then, in connection with each such event, the Company shall send to the Warrantholder: (A) at least twenty (20) days' prior written notice of the date on which the books of the Company shall close or a record shall be taken for such dividend, distribution, subscription rights (specifying the date on which the holders of Preferred Stock shall be entitled thereto) or for determining rights to vote in respect of such Merger Event, dissolution, liquidation or winding up; (B) in the case of any such Merger Event, dissolution or winding up, at least twenty (20) days' prior written notice of the date when the same shall take place (and specifying the date on which the holders of Preferred Stock shall be entitled to exchange their Preferred Stock for securities or other property deliverable upon such Merger Event, dissolution, liquidation or winding up); and (C) in the case of a public offering, the Company shall give the Warrantholder at least twenty (20) days written notice prior to the effective date thereof.

Each such written notice shall set forth, in reasonable detail, (i) the event requiring the adjustment, (ii) the amount of the adjustment, (iii) the method by which such adjustment was calculated, (iv) the Exercise Price, and (v) the number of shares subject to purchase hereunder after giving effect to such adjustment, and shall be given by first class mail, postage prepaid, addressed to the Warrantholder, at the address as shown on the books of the Company.

(h) <u>Timely Notice</u>. Failure to timely provide such notice required by subsection (g) above shall entitle Warrantholder to retain the benefit of the applicable notice period notwithstanding anything to the contrary contained in any insufficient notice received by Warrantholder. The notice period shall begin on the date Warrantholder actually receives a written notice containing all the information specified above.

9. REPRESENTATIONS, WARRANTIES AND COVENANTS OF THE COMPANY.

(a) Reservation of Preferred Stock. The Preferred Stock issuable upon exercise of the Warrantholder's rights has been duly and validly reserved and, when issued in accordance with the provisions of this Warrant Agreement, will be validly issued, fully paid and non-assessable, and will be free of any taxes, liens, charges or encumbrances of any nature whatsoever; provided, however, that the Preferred Stock issuable pursuant to this Warrant Agreement may be subject to restrictions on transfer under state and/or Federal securities laws. The Company has made available to the Warrantholder true, correct and complete copies of its Charter and Bylaws, as amended. The issuance of certificates for shares of Preferred Stock upon exercise of the Warrant Agreement shall be made without charge to the Warrantholder for any issuance tax in respect thereof, or other cost incurred by the Company in connection with such exercise and the related issuance of shares of Preferred Stock. The Company shall not be required to pay any tax which may be payable in respect of any transfer involved and the issuance and delivery of any certificate in a name other than that of the Warrantholder.

- (b) <u>Due Authority</u>. The execution and delivery by the Company of this Warrant Agreement and the performance of all obligations of the Company hereunder, including the issuance to Warrantholder of the right to acquire the shares of Preferred Stock, have been duly authorized by all necessary corporate action on the part of the Company, and the Leases and this Warrant Agreement are not inconsistent with the Company's Charter or Bylaws, do not contravene any law or governmental rule, regulation or order applicable to it, do not and will not contravene any provision of, or constitute a default under, any indenture, mortgage, contract or other instrument to which it is a party or by which it is bound, and the Leases and this Warrant Agreement constitute legal, valid and binding agreements of the Company, enforceable in accordance with their respective terms.
- (c) <u>Consents and Approvals</u>. No consent or approval of, giving of notice to, registration with, or taking of any other action in respect of any state, Federal or other governmental authority or agency is required with respect to the execution, delivery and performance by the Company of its obligations under this Warrant Agreement, except for the filing of notices pursuant to Regulation D under the 1933 Act and any filing required by applicable state securities law, which filings will be effective by the time required thereby.
- (d) <u>Issued Securities</u>. All issued and outstanding shares of Common Stock, Preferred Stock or any other securities of the Company have been duly authorized and validly issued and are fully paid and non-assessable. All outstanding shares of Common Stock, Preferred Stock and any other securities were issued in full compliance with all Federal and state securities laws. In addition, as of the date hereof:
 - (i) The authorized capital of the Company consists of (A) 100,000,000 shares of Common Stock, of which 12,179,896 shares are issued and outstanding, (B) 9,187,500 shares of Series A Preferred Stock, of which 9,187,500 shares are issued and outstanding and are convertible into 9,187,500 shares of Common Stock at \$1.00 per share, and (C) 6,326,531 shares of Series B Preferred Stock, of which 6,254,806 shares are issued and outstanding and are convertible into 6,254,806 shares of Common Stock at \$4.90 per share,
 - (ii) The Company has reserved 4,210,937 shares of Common Stock for issuance under its 1999 Stock Plan, under which 3,804,717 options are outstanding at an average price of \$0.50 per share. There are no other options, warrants, conversion privileges or other rights presently outstanding to purchase or otherwise acquire any authorized but unissued shares of the Company's capital stock or other securities of the Company.
 - (iii) In accordance with the Company's Certificate of Incorporation, no shareholder of the Company has preemptive rights to purchase new issuances of the Company's capital stock.
- (e) <u>Insurance</u>. The Company has in full force and effect insurance policies, with extended coverage, insuring the Company and its property and business against such losses and risks, and in such amounts, as are customary for corporations engaged in a similar business and similarly situated and as otherwise may be required pursuant to the terms of any other contract or agreement.
- (f) Other Commitments to Register Securities. Except as set forth in this Warrant Agreement, the Company is not, pursuant to the terms of any other agreement currently in existence, under any obligation to register under the 1933 Act any of its presently outstanding securities or any of its securities which may hereafter be issued.
- (g) Exempt Transaction. Subject to the accuracy of the Warrantholder's representations in Section 10 hereof, the issuance of the Preferred Stock upon exercise of this Warrant will constitute a transaction exempt from (i) the registration requirements of Section 5 of the 1933 Act, in reliance upon Section 4(2) thereof, and (ii) the qualification requirements of the applicable state securities laws.
- (h) <u>Compliance with Rule 144</u>. At the written request of the Warrantholder, who proposes to sell Preferred Stock issuable upon the exercise of the Warrant in compliance with Rule 144 promulgated by the Securities and Exchange Commission, the Company shall furnish to the Warrantholder, within ten days after receipt of such request, a written statement confirming the Company's compliance with the filing requirements of the Securities and Exchange Commission as set forth in such Rule, as such Rule may be amended from time to time.

10. REPRESENTATIONS AND COVENANTS OF THE WARRANTHOLDER.

This Warrant Agreement has been entered into by the Company in reliance upon the following representations and covenants of the Warrantholder:

- (a) <u>Investment Purposes</u>. The right to acquire Preferred Stock or the Preferred Stock issuable upon exercise of the Warrantholder's rights contained herein will be acquired for investment and not with a view to the sale or distribution of any part thereof, and the Warrantholder has no present intention of selling or engaging in any public distribution of the same except pursuant to a registration or exemption.
- (b) <u>Private Issue</u>. The Warrantholder understands (i) that the Preferred Stock issuable upon exercise of this Warrant is not registered under the 1933 Act or qualified under applicable state securities laws on the ground that the issuance contemplated by this Warrant Agreement will be exempt from the registration and qualifications requirements thereof, and (iii) that the Company's reliance on such exemption is predicated on the representations set forth in this Section 10.
- (c) <u>Disposition of Warrantholder's Rights</u>. In no event will the Warrantholder make a disposition of any of its rights to acquire Preferred Stock or Preferred Stock issuable upon exercise of such rights unless and until (i) it shall have notified the Company of the proposed disposition, and (ii) if requested by the Company, it shall have furnished the Company with an opinion of counsel (which counsel may either be inside or outside counsel to the Warrantholder) satisfactory to the Company and its counsel to the effect that (A) appropriate action necessary for compliance with the 1933 Act has been taken, or (B) an exemption from the registration requirements of the 1933 Act is available. Notwithstanding the foregoing, the restrictions imposed upon the transferability of any of its rights to acquire Preferred Stock or Preferred Stock issuable on the exercise of such rights do not apply to transfers from the beneficial owner of any of the aforementioned securities to its nominee or from such nominee to its beneficial owner, and shall terminate as to any particular share of Preferred Stock when (1) such security shall have been effectively registered under the 1933 Act and sold by the holder thereof in accordance with such registration or (2) such security shall have been sold without registration in compliance with Rule 144 under the 1933 Act, or (3) a letter shall have been issued to the Warrantholder at its request by the Staff of the Securities and Exchange Commission or a ruling shall have been issued to the Warrantholder at its request by such Commission, as the case may be, if such security is transferred without registration under the 1933 Act in accordance with the conditions set forth in such letter or ruling specifies that no subsequent restrictions on transfer are required. Whenever the restrictions imposed hereunder shall terminate, as hereinabove provided, the Warrantholder of a share of Preferred Stock then outstanding as to which such restrictions have terminated shall be entitled to receive from t
- (d) <u>Financial Risk</u>. The Warrantholder has such knowledge and experience in financial and business matters as to be capable of evaluating the merits and risks of its investment, and has the ability to bear the economic risks of its investment
- (e) <u>Risk of No Registration</u>. The Warrantholder understands that if the Company does not register with the Securities and Exchange Commission pursuant to Section 12 of the 1934 Act (the "1934 Act"), or file reports pursuant to Section 15(d), of the 1934 Act, or if a registration statement covering the securities under the 1933 Act is not in effect when it desires to sell (i) the rights to purchase Preferred Stock pursuant to this Warrant Agreement, or (ii) the Preferred Stock issuable upon exercise of the right to purchase, it may be required to hold such securities for an indefinite period. The Warrantholder also understands that any sale of its rights of the Warrantholder to purchase Preferred Stock or Preferred Stock which might be made by it in reliance upon Rule 144 under the 1933 Act may be made only in accordance with the terms and conditions of that Rule.
 - (f) Accredited Investor. Warrantholder is an "accredited investor" within the meaning of the Securities and Exchange Rule 501 of Regulation D, as presently in effect.

11. RIGHT OF FIRST OFFER.

In accordance with the provisions of Section 2 of the Investor Rights Agreement dated as of July 5, 2000 ("Investor Rights Agreement"), if the Company proposes to offer any shares of, or securities convertible into or exercisable for any shares of, any class of its capital stock ("Shares"), subject to the exceptions set forth thereof, the Company shall promptly provide Warrantholder with an offer to sell Warrantholder a portion of such Shares equal to the proportion that the number of shares of Preferred Stock to be issued upon exercise hereunder or number of shares of common stock upon conversion thereof, bears to the total number of shares of common stock of the Company then outstanding (assuming full conversion of all shares of Preferred Stock).

12. TRANSFERS.

Subject to the terms and conditions contained in Section 10 hereof, this Warrant Agreement and all rights hereunder are transferable in whole or in part by the Warrantholder and any successor transferee, provided, however, in no event shall the number of transfers of the rights and interests in all of the Warrants exceed three (3) transfers. The transfer shall be recorded on the books of the Company upon receipt by the Company of a notice of transfer in the form attached hereto as **Exhibit III** (the "Transfer Notice"), at its principal offices and the payment to the Company of all transfer taxes and other governmental charges imposed on such transfer.

13. MISCELLANEOUS.

- (a) <u>Effective Date</u>. The provisions of this Warrant Agreement shall be construed and shall be given effect in all respects as if it had been executed and delivered by the Company on the date hereof. This Warrant Agreement shall be binding upon any successors or assigns of the Company.
- (b) Attorney's Fees. In any litigation, arbitration or court proceeding between the Company and the Warrantholder relating hereto, the prevailing party shall be entitled to attorneys' fees and expenses and all costs of proceedings incurred in enforcing this Warrant Agreement.
 - (c) Governing Law. This Warrant Agreement shall be governed by and construed for all purposes under and in accordance with the laws of the State of Illinois.
 - (d) Counterparts. This Warrant Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- (e) <u>Notices</u>. Any notice required or permitted hereunder shall be given in writing and shall be deemed effectively given upon personal delivery, facsimile transmission (provided that the original is sent by personal delivery or mail as hereinafter set forth) or seven (7) days after deposit in the United States mail, by registered or certified mail, addressed (i) **to the Warrantholder** at 6111 North River Road, Rosemont, Illinois 60018, <u>Attention</u>: Venture Lease Administration, **cc: Legal Department**, <u>Attention</u>: General Counsel, (and/or, if by Facsimile, (847) 518-5465 and (847)518-5088) and (ii) **to the Company** at 1761 Business Center Drive, Suite 250, Reston, CA 20190, <u>Attention</u>: David Jones (and/or if by Facsimile, (703) 438-2033) or at such other address as any such party may subsequently designate by written notice to the other party.
- (f) Remedies. In the event of any default hereunder, the non-defaulting party may proceed to protect and enforce its rights either by suit in equity and/or by action at law, including but not limited to an action for damages as a result of any such default, and/or an action for specific performance for any default where Warrantholder will not have an adequate remedy at law and where damages will not be readily ascertainable. The Company expressly agrees that it shall not oppose an application by the Warrantholder or any other person entitled to the benefit of this Agreement requiring specific performance of any or all provisions hereof or enjoining the Company from continuing to commit any such breach of this Agreement
- (g) No Impairment of Rights. The Company will not, by amendment of its Charter or through any other means, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, but will at all times in good faith assist in the carrying out of all such terms and in the taking of all such actions as may be necessary or appropriate in order to protect the rights of the Warrantholder against impairment.
- (h) <u>Survival</u>. The representations, warranties, covenants and conditions of the respective parties contained herein or made pursuant to this Warrant Agreement shall survive the execution and delivery of this Warrant Agreement.
- (i) <u>Severability</u>. In the event any one or more of the provisions of this Warrant Agreement shall for any reason be held invalid, illegal or unenforceable, the remaining provisions of this Warrant Agreement shall be unimpaired, and the invalid, illegal or unenforceable provision shall be replaced by a mutually acceptable valid, legal and enforceable provision, which comes closest to the intention of the parties underlying the invalid, illegal or unenforceable provision.
 - (i) Amendments. Any provision of this Warrant Agreement may be amended by a written instrument signed by the Company and by the Warrantholder.

COMPANY:	COMSCORE NETWORKS, INC.
	By: /s/ Magid Abraham
	Title: /s/ CEO
WARRANTHOLDER:	COMDISCO, INC.
	Ву:
	Title:

IN WITNESS WHEREOF, the parties hereto have caused this Warrant Agreement to be executed by its officers thereunto duty authorized as of the Effective Date.

COMMON STOCK PURCHASE WARRANT

THIS WARRANT AND THE SHARES OF COMMON STOCK WHICH MAY BE PURCHASED PURSUANT TO THE EXERCISE OF THIS WARRANT HAVE BEEN ACQUIRED SOLELY FOR INVESTMENT AND HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR ANY STATE SECURITIES LAWS. SUCH SECURITIES MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED OR HYPOTHECATED IN THE ABSENCE OF SUCH REGISTRATION OR AN OPINION OF COUNSEL SATISFACTORY TO THE COMPANY AND ITS COUNSEL THAT SUCH SALE, OFFER, PLEDGE OR HYPOTHECATION IS EXEMPT FROM THE REGISTRATION AND PROSPECTUS DELIVERY REQUIREMENTS OF THE ACT AND OF ANY APPLICABLE STATE SECURITIES LAWS UNLESS SOLD PURSUANT TO RULE 144 OF THE ACT.

Void After June 26, 2011

COMSCORE NETWORKS, INC.

WARRANT TO PURCHASE SHARES OF COMMON STOCK

For value received, William Henderson (the "**Holder**") is entitled to subscribe for and purchase up to 100,000 shares (as adjusted pursuant to Section 3 hereof) of the Common Stock (the "**Common Stock**"), \$0.001 par value (the "**Shares**"), of comScore Networks, Inc., a Delaware corporation (the "**Company**"), at the price of \$1.00 per share (the "**Exercise Price**") (as adjusted pursuant to Section 3 hereof), subject to the provisions and upon the terms and conditions hereinafter set forth.

1. Exercise and Payment

- 1.1 <u>Vesting</u>. This warrant shall be exercisable, in whole or in part, according to the following vesting schedule: l/24th of the total number of shares subject to this warrant shall become exercisable on the last day of each calendar month beginning on June 26, 2001, until all such shares are exercisable, subject to the Holder continuing to be a member of the Board of Directors of the Company on such dates.
- 1.2 Exercise. The purchase rights represented by this Warrant may be exercised by the Holder, in whole or in part, by the surrender of a duly executed exercise notice in the form attached hereto as Exhibit A at the principal office of the Company, and by the payment to the Company, by check or wire transfer, of an amount equal to the aggregate Exercise Price of the shares being purchased.
- 1.3 <u>Stock Certificate</u>. In the event of the exercise of this Warrant, a certificate for the shares of Common Stock so purchased shall be delivered to the Holder within a reasonable time, which shall in no event be later than thirty (30) days thereafter.

- 2. <u>Stock Fully Paid; Reservation of Shares</u>. All of the Shares issuable upon the exercise of this Warrant will, upon issuance and receipt of the Exercise Price therefor, be fully paid and nonassessable. During the period within which this Warrant may be exercised, the Company shall at all times have authorized and reserved for issuance sufficient shares of its Common Stock to provide for the exercise of this Warrant.
- 3. <u>Adjustment of Exercise Price and Number of Shares</u>. Subject to Section 10 hereof, the number and kind of securities purchasable upon the exercise of this Warrant and the Exercise Price therefor shall be subject to adjustment from time to time upon the occurrence of certain events, as follows:
- 3.1 <u>Reclassification, Consolidation or Merger</u>. In case of any reclassification or change of the Common Stock (other than a change in par value, or as a result of a subdivision or combination), or in case of any Merger Event (as defined herein), the Company or the successor corporation, as the case may be, shall execute a new warrant, providing that the Holder shall have the right to exercise such new warrant, and procure upon such exercise and payment of the same aggregate Exercise Price, in lieu of the shares of Common Stock theretofore issuable upon exercise of this Warrant, the kind and amount of shares of stock, other securities, money and property receivable upon such reclassification, change, or Merger Event by a holder of an equivalent number of shares of Common Stock. Such new warrant shall provide for adjustments which shall be as nearly equivalent as may be practicable to the adjustments provided for in this Section 3.
- 3.2 Stock Splits, Dividends and Combinations. In the event that the Company shall at any time subdivide the outstanding shares of Common Stock, or shall issue a stock dividend on its outstanding shares of Common Stock, the number of Shares issuable upon exercise of this Warrant immediately prior to such subdivision or to the issuance of such stock dividend shall be proportionately increased and the Exercise Price shall be proportionately decreased so that the Holder of the Warrant after such time shall be entitled to receive the number of shares of Common Stock which such Holder would have owned or been entitled to receive had such Warrant been exercised immediately prior to such event, and in the event that the Company shall at any time combine the outstanding shares of Common Stock, the number of Shares issuable upon exercise of this Warrant immediately prior to such combination shall be proportionately decreased, and the Exercise Price shall be proportionately increased so that the Holder of the Warrant after such time shall be entitled to receive the number of shares of Common Stock which such Holder would have owned or been entitled to receive had such Warrant been exercised prior to such event, in either case effective at the close of business on the date of such subdivision, stock dividend or combination, as the case may be.
- 4. Notice of Adjustments. In the event that: (i) the Company shall declare any dividend or distribution upon its Common Stock, whether in cash, property, stock or other securities; (ii) the Company shall offer for subscription pro rata to the holders of its Common Stock any additional shares of stock of any class or other rights; (iii) there shall be any merger or consolidation of the Company with or into a third party pursuant to which the Company's stockholders prior to the transaction own less than fifty percent (50%) of the surviving entity or the sale of all or substantially all of the assets of the Company (a "Merger Event"); or (iv) there shall be any voluntary or involuntary dissolution, liquidation or winding up of the Company; then, in connection with each such event, the Company shall send to the Holder:

- (a) At least ten (10) days prior written notice of the date on which the books of the Company shall close or a record shall be taken for such dividend, distribution, subscription rights (specifying the date on which the holders of Common Stock shall be entitled thereto) or for determining rights to vote in respect of such Merger Event, dissolution, liquidation or winding up; and
- (b) In the case of any such Merger Event, dissolution, liquidation or winding up, at least ten (10) days prior written notice of the date when the same shall take place (and specifying the date on which the holders of Common Stock shall be entitled to exchange their Common Stock for securities or other property deliverable upon such Merger Event, dissolution, liquidation or winding up).

Each such written notice shall set forth, in reasonable detail, (i) the event requiring the adjustment, (ii) the amount of the adjustment, (iii) the method by which such adjustment was calculated, (iv) the Exercise Price, and (v) the number of shares subject to purchase hereunder after giving effect to such adjustment, and shall be given by first class mail, postage prepaid, addressed to the Holder, at the address as shown on the books of the Company.

- 5. <u>Fractional Shares</u>. No fractional shares of Common Stock will be issued in connection with any exercise hereunder. In lieu of such fractional shares the Company shall make a cash payment therefor based upon the Exercise Price then in effect.
 - 6. Representations and Warranties of the Holder. The Holder hereby represents and warrants to the Company, with respect to its acquisition of the Warrant, as follows:
- 6.1 Experience. The Holder has sufficient knowledge and experience in financial and business matters so that he is capable of evaluating the merits and risks of his investment in the Company and has the capacity to protect his own interests.
- 6.2 <u>Investment</u>. The Holder is acquiring the Warrant and the Shares for investment for its own account, not as a nominee or agent, and not with the view to, or for resale in connection with, any distribution thereof. The Holder understands that the Warrant and the Shares have not been, and will not be, registered under the Act by reason of a specific exemption from the registration provisions of the Act, the availability of which depends upon, among other things, the bona fide nature of the investment intent and the accuracy of such Holder's representations as expressed herein.
- 6.3 Rule 144. The Holder acknowledges that the Warrant and the Shares must be held indefinitely unless subsequently registered under the Act or unless an exemption from such registration is available. The Holder is aware of the provisions of Rule 144 promulgated under the Act which permit limited resale of shares purchased in a private placement subject to the satisfaction of certain conditions, including, among other things, the existence of a public market for the shares, the availability of certain current public information about the Company, the resale occurring not less than one (1) year after a party has purchased and paid for the security to be sold, the sale being effected through a "broker's transaction" or in transactions directly with a "market maker" and the number of shares being sold during any three-month period not exceeding specified limitations.
- 6.4 No Public Market. The Holder understands that no public market now exists for any of the securities issued by the Company and that the Company has made no assurances that a public market will ever exist for the Company's securities.

- 6.5 No Solicitation. The Holder knows of no public solicitation or advertisement of an offer in connection with the proposed issuance and sale of the Warrant or the Shares.
- 6.6 Residence. The residence of the Holder for securities law purposes is set forth herein on page 6.

7 Restrictions on Transfer

7.1 <u>Restrictive Legend</u>. Each certificate representing (i) the Shares and (ii) any other securities issued in respect of the Shares upon any stock split, stock dividend, recapitalization, merger, consolidation or similar event, (collectively, the "**Restricted Securities**") shall (unless otherwise permitted by the provisions of Section 7.2 below) be stamped or otherwise imprinted with a legend in substantially the following form (in addition to any legend required under applicable state securities laws):

THE SHARES REPRESENTED BY THIS CERTIFICATE HAVE BEEN ACQUIRED FOR INVESTMENT AND HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933. THESE SHARES MAY NOT BE SOLD OR TRANSFERRED IN THE ABSENCE OF SUCH REGISTRATION OR AN EXEMPTION THEREFROM UNDER SAID ACT.

7.2 Notice of Proposed Transfers. The holder of each certificate representing Restricted Securities by acceptance thereof agrees to comply in all respects with the provisions of this Section 7. Prior to any proposed transfer of any Restricted Securities, unless there is in effect a registration statement under the Act covering the proposed transfer, the holder thereof shall give written notice to the Company of such Holder's intention to effect such transfer. Each such notice shall describe the manner and circumstances of the proposed transfer in sufficient detail, and shall, if the Company so requests, be accompanied by either (i) an unqualified written opinion of legal counsel who shall be reasonably satisfactory to the Company, addressed to the Company and reasonably satisfactory in form and substance to the Company's counsel, to the effect that the proposed transfer of the Restricted Securities may be effected without registration under the Act, or (ii) a "No Action" letter from the Securities and Exchange Commission") to the effect that the transfer of such securities without registration will not result in a recommendation by the staff of the Commission that action be taken with respect thereto, whereupon the holder of such Restricted Securities shall be entitled to transfer such Restricted Securities in accordance with the terms of the notice delivered by the holder to the Company; provided, however, that no opinion or No Action letter need be obtained with respect to a transfer to (A) the immediate family of Holder upon the Holder's death, by will or intestacy, or to a trust for the benefit of the Holder's immediate family, (B) a partner, active or retired, of a holder of Restricted Securities, (C) the estate of any such partner, or (D) an "affiliate" of a holder of Restricted Securities as that term is defined in Rule 405 promulgated by the Commission under the Act, provided that in such cases the transferre agrees in writing to be subject to the terms hereof. Each certificate evidencing the Restricted

- 8. <u>Rights of Stockholders</u>. No holder of this Warrant shall be entitled, as a Warrant holder, to vote or receive dividends or be deemed the holder of Common Stock or any other securities of the Company which may at any time be issuable on the exercise hereof for any purpose, nor shall anything contained herein be construed to confer upon the Holder, as such, any of the rights of a stockholder of the Company or any right to vote for the election of directors or upon any matter submitted to stockholders at any meeting thereof, or to give or withhold consent to any corporate action (whether upon any recapitalization, issuance of stock, reclassification of stock, change of par value, consolidation, merger, conveyance, or otherwise) or to receive notice of meetings, or to receive dividends or subscription rights or otherwise until the Warrant shall have been exercised and the Shares purchasable upon the exercise hereof shall have become deliverable, as provided herein.
- 9. Market Stand-Off Provision. The holder of this Warrant agrees to be bound by the "Market Stand-Off" provision in section 1(l) of that certain Amended and Restated Investor Rights Agreement dated as of July 5, 2000 (the "Rights Agreement") between the Company and certain Investors.
 - 10. Expiration of Warrant. Notwithstanding any other provision of this "Warrant, this Warrant shall expire and shall no longer be exercisable upon the earlier to occur of:
 - (a) 5:00 p.m., California time, on June 26, 2011;
 - (b) immediately prior to the initial underwritten public offering of the Company's Common Stock; or
 - (c) immediately upon the closing of a sale, conveyance, disposal or encumbrance of all or substantially all of the Company's property or business or the Company's merger into or consolidation with any other corporation or any other transaction or series of related transactions in which more than fifty percent (50%) of the voting power of the Company is disposed (a "Change of Control"); provided, however, that a Change of Control shall not include any sale of stock directly by the Company to professional venture capital investors in connection with a transaction the primary purpose of which is to raise financing for the Company.

11. Miscellaneous

- $11.1 \, \underline{\text{Governing Law}}$. This Agreement shall be governed in all respects by the laws of the State of Delaware.
- 11.2 <u>Successors and Assigns</u>. Except as otherwise provided herein, the provisions hereof shall inure to the benefit of, and be binding upon, the successors, assigns, heirs, executors, and administrators of the Company and the Holder.
- 11.3 Entire Agreement; Amendment. This Warrant constitutes the full and entire understanding and agreement between the parties with regard to the subjects hereof. Neither this Warrant nor any term hereof may be amended, waived, discharged, or terminated other than by a written

instrument signed by the parry against whom enforcement of any such amendment, waiver, discharge or termination is sought.

11.4 Notices, etc.. All notices and other communications required or permitted hereunder shall be in writing and shall be deemed effectively given upon delivery to the party to be notified in person or by courier service or by registered or certified mail, postage prepaid, addressed (a) to the Holder, at the address set forth on the last page of this Warrant or at such other address as such Holder shall have furnished the Company in writing, or (b) if to the Company, at the address set forth on the last page of this Warrant and addressed to the attention of the Chief Executive Officer, or at such other address as the Company shall have furnished to the Holder.

Issued this 26th day of June, 2001

COMSCORE NETWORKS, INC.

By: /s/ James A. Powers

James A. Powers, Corporate Secretary

COMMON STOCK PURCHASE WARRANT

THIS WARRANT AND THE SHARES OF COMMON STOCK WHICH MAY BE PURCHASED PURSUANT TO THE EXERCISE OF THIS WARRANT HAVE BEEN ACQUIRED SOLELY FOR INVESTMENT AND HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR ANY STATE SECURITIES LAWS. SUCH SECURITIES MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED OR HYPOTHECATED IN THE ABSENCE OF SUCH REGISTRATION OR AN OPINION OF COUNSEL SATISFACTORY TO THE COMPANY AND ITS COUNSEL THAT SUCH SALE, OFFER, PLEDGE OR HYPOTHECATION IS EXEMPT FROM THE REGISTRATION AND PROSPECTUS DELIVERY REQUIREMENTS OF THE ACT AND OF ANY APPLICABLE STATE SECURITIES LAWS UNLESS SOLD PURSUANT TO RULE 144 OF THE ACT.

Void After September 30, 2005

COMSCORE NETWORKS, INC.

WARRANT TO PURCHASE SHARES OF COMMON STOCK

For value received, Comstock Homes, LLC (the "Holder") is entitled to subscribe for and purchase up to 10,000 shares (as adjusted pursuant to Section 3 hereof) of the Common Stock (the "Common Stock"), \$0.001 par value (the "Shares"), of comScore Networks, inc., a Delaware corporation (the "Company"), at the price of \$5.90 per share (the "Exercise Price") (as adjusted pursuant to Section 3 hereof), subject to the provisions and upon the terms and conditions hereinafter set forth.

1. Exercise and Payment.

- 1.1 Exercise. The purchase rights represented by this Warrant may be exercised by the Holder, in whole or in part, by the surrender of a duly executed exercise notice in the form attached hereto as **Exhibit A** at the principal office of the Company, and by the payment to the Company, by check or wire transfer, of an amount equal to the aggregate Exercise Price of the shares being purchased.
- 1.2 <u>Stock Certificate</u>. In the event of the exercise of this Warrant, a certificate for the shares of Common Stock so purchased shall be delivered to the Holder within a reasonable time, which shall in no event be later than thirty (30) days thereafter.
- 2. <u>Stock Fully Paid</u>; <u>Reservation of Shares</u>. All of the Shares issuable upon the exercise of this Warrant will, upon issuance and receipt of the Exercise Price therefor, be fully paid and nonassessable. During the period within which this Warrant maybe exercised, the Company shall at all times have authorized and reserved for issuance sufficient shares of its Common Stock to provide for the exercise of this Warrant.
 - 3. Adjustment of Exercise Price and Number of Shares. Subject to Section 9 hereof, the number and kind of securities purchasable upon the exercise of this Warrant and the Exercise Price

therefor shall be subject to adjustment from time to time upon the occurrence of certain events, as follows:

- 3.1 <u>Reclassification</u>, <u>Consolidation or Merger</u>. In case of any reclassification or change of the Common Stock (other than a change in par value, or as a result of a subdivision or combination), or in case of any Merger Event (as defined herein), the Company or the successor corporation, as the case may be, shall execute a new warrant, providing that the Holder shall have the right to exercise such new warrant, and procure upon such exercise and payment of the same aggregate Exercise Price, in lieu of the shares of Common Stock theretofore issuable upon exercise of this Warrant, the kind and amount of shares of stock, other securities, money and property receivable upon such reclassification, change, or Merger Event by a holder of an equivalent number of shares of Common Stock. Such new warrant shall provide for adjustments which shall be as nearly equivalent as may be practicable to the adjustments provided for in this Section 3.
- 3.2 Stock Splits, Dividends and Combinations. In the event that the Company shall at any time subdivide the outstanding shares of Common Stock, or shall issue a stock dividend on its outstanding shares of Common Stock, the number of Shares issuable upon exercise of this Warrant immediately prior to such subdivision or to the issuance of such stock dividend shall be proportionately increased and the Exercise Price shall be proportionately decreased so that the Holder of the Warrant after such time shall be entitled to receive the number of shares of Common Stock which such Holder would have owned or been entitled to receive had such Warrant been exercised immediately prior to such event, and in the event that the Company shall at any time combine the outstanding shares of Common Stock, the number of Shares issuable upon exercise of this Warrant immediately prior to such combination shall be proportionately decreased, and the Exercise Price shall be proportionately increased so that the Holder of the Warrant after such time shall be receive the number of shares of Common Stock which such Holder would have owned or been entitled to receive had such Warrant been exercised prior to such event, in either case effective at the close of business on the date of such subdivision, stock dividend or combination, as the case may be.
- 4. <u>Notice of Adjustments</u>. In the event that: (i) the Company shall declare any dividend or distribution upon its Common Stock, whether in cash, property, stock or other securities; (ii) the Company shall offer for subscription pro rata to the holders of its Common Stock any additional shares of stock of any class or other rights; (iii) there shall be any merger or consolidation of the Company with or into a third party pursuant to which the Company's stockholders prior to the transaction own less than fifty percent (50%) of the surviving entity or the sale of all or substantially all of the assets of the Company (a "Merger Event"); or (iv) there shall be any voluntary or involuntary dissolution, liquidation or winding up of the Company; then, in connection with each such event, the Company shall send to the Holder:
- (a) At least ten (10) days prior written notice of the date on which the books of the Company shall close or a record shall be taken for such dividend, distribution, subscription rights Specifying the date on which the holders of Common Stock shall be entitled thereto) or for determining rights to vote in respect of such Merger Event, dissolution, liquidation or winding up; and
 - (b) In the case of any such Merger Event, dissolution, liquidation or winding up, at least ten (10) days prior written notice of the date when the same shall take place (and specifying the date

on which the holders of Common Stock shall be entitled to exchange their Common Stock for securities or other property deliverable upon such Merger Event, dissolution, liquidation or winding up).

Each such written notice shall set forth, in reasonable detail, (i) the event requiring the adjustment, (ii) the amount of the adjustment, (iii) the method by which such adjustment was calculated, (iv) the Exercise Price, and (v) the number of shares subject to purchase hereunder after giving effect to such adjustment, and shall be given by first class mail, postage prepaid, addressed to the Holder, at the address as shown on the books of the Company.

- 5. <u>Fractional Shares</u>. No fractional shares of Common Stock will be issued in connection with any exercise hereunder. In lieu of such fractional shares the Company shall make a cash payment therefor based upon the Exercise Price then in effect.
 - 6. Representations and Warranties of the Holder. The Holder hereby represents and warrants to the Company, with respect to its acquisition of the Warrant, as follows:
- 6.1 Experience. The Holder has sufficient knowledge and experience in financial and business matters so that he is capable of evaluating the merits and risks of his investment in the Company and has the capacity to protect his own interests.
- 6.2 <u>Investment</u>. The Holder is acquiring the Warrant and the Shares for investment for its own account, not as a nominee or agent, and not with the view to, or for resale in connection with, any distribution thereof. The Holder understands that the Warrant and the Shares have not been, and will not be, registered under the Act by reason of a specific exemption from the registration provisions of the Act, the availability of which depends upon, among other things, the bona fide nature of the investment intent and the accuracy of such Holder's representations as expressed herein.
- 6.3 Rule 144. The Holder acknowledges that the Warrant and the Shares must be held indefinitely unless subsequently registered under the Act or unless an exemption from such registration is available. The Holder is aware of the provisions of Rule 144 promulgated under the Act which permit limited resale of shares purchased in a private placement subject to the satisfaction of certain conditions, including, among other things, the existence of a public market for the shares, the availability of certain current public information about the Company, the resale occurring not less than one (1) year after a party has purchased and paid for the security to be sold, the sale being effected through a "broker's transaction" or in transactions directly with a "market maker" and the number of shares being sold during any three-month period not exceeding specified limitations.
- 6.4 No Public Market. The Holder understands that no public market now exists for any of the securities issued by the Company and that the Company has made no assurances that a public market will ever exist for the Company's securities.
 - 6.5 No Solicitation. The Holder knows of no public solicitation or advertisement of an offer in connection with the proposed issuance and sale of the Warrant or the Shares.
 - 6.6 Residence. The residence of the Holder for securities law purposes is set forth herein on page 6.

7. Restrictions on Transfer.

7.1 <u>Restrictive Legend</u>. Each certificate representing (i) the Shares and (ii) any other securities issued in respect of the Shares upon any stock split, stock dividend, recapitalization, merger, consolidation or similar event, (collectively, the "**Restricted Securities**") shall (unless otherwise permitted by the provisions of Section 7.2 below) be stamped or otherwise imprinted with a legend in substantially the following form (in addition to any legend required under applicable state securities laws):

THE SHARES REPRESENTED BY THIS CERTIFICATE HAVE BEEN ACQUIRED FOR INVESTMENT AND HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933. THESE SHARES MAY NOT BE SOLD OR TRANSFERRED IN THE ABSENCE OF SUCH REGISTRATION OR AN EXEMPTION THEREFROM UNDER SAID ACT.

7.2 Notice of Proposed Transfers. The holder of each certificate representing Restricted Securities by acceptance thereof agrees to comply in all respects with the provisions of this Section 7. Prior to any proposed transfer of any Restricted Securities, unless there is in effect a registration statement under the Act covering the proposed transfer, the holder thereof shall give written notice to the Company of such Holder's intention to effect such transfer. Each such notice shall describe the manner and circumstances of the proposed transfer in sufficient detail, and shall, if the Company so requests, be accompanied by either (i) an unqualified written opinion of legal counsel who shall be reasonably satisfactory to the Company, addressed to the Company and reasonably satisfactory in form and substance to the Company's counsel, to the effect that the proposed transfer of the Restricted Securities may be effected without registration under the Act, or (ii) a "No Action" letter from the Securities and Exchange Commission (the "Commission") to the effect that the transfer of such securities without registration will not result in a recommendation by the staff of the Commission that action be taken with respect thereto, whereupon the holder of such Restricted Securities shall be entitled to transfer such Restricted Securities in accordance with the terms of the notice delivered by the holder to the Company; provided, however, that no opinion or No Action letter need be obtained with respect to a transfer to (A) the immediate family of Holder upon the Holder's death, by will or intestacy, or to a trust for the benefit of the Holder's immediate family, (B) a partner, active or retired, of a holder of Restricted Securities, (C) the estate of any such partner, or (D) an "affiliate" of a holder of Restricted Securities as that term is defined in Rule 405 promulgated by the Commission under the Act, provided that in such cases the transferree agrees in writing to be subject to the terms hereof. Each certificate evidenc

8. <u>Rights of Stockholders</u>. No holder of this Warrant shall be entitled, as a Warrant holder, to vote or receive dividends or be deemed the holder of Common Stock or any other securities of the Company which may at any time be issuable on the exercise hereof for any purpose, nor shall anything contained herein be construed to confer upon the Holder, as such, any of the rights of a stockholder of the Company or any right to vote for the election of directors or upon any matter submitted to

stockholders at any meeting thereof, or to give or withhold consent to any corporate action (whether upon any recapitalization, issuance of stock, reclassification of stock, change of par value, consolidation, merger, conveyance, or otherwise) or to receive notice of meetings, or to receive dividends or subscription rights or otherwise until the Warrant shall have been exercised and the Shares purchasable upon the exercise hereof shall have become deliverable, as provided herein.

- 9. Expiration of Warrant. Notwithstanding any other provision of this Warrant, this Warrant shall expire and shall no longer be exercisable at 5:00 p.m., California time, on September 30, 2009.
- 10. Miscellaneous.
 - 10.1 Governing Law. This Agreement shall be governed in all respects by the laws of the State of Delaware.
- 10.2 <u>Successors and Assigns</u>. Except as otherwise provided herein, the provisions hereof shall inure to the benefit of, and be binding upon, the successors, assigns, heirs, executors, and administrators of the Company and the Holder.
- 10.3 Entire Agreement; Amendment. This Warrant constitutes the full and entire understanding and agreement between the parties with regard to the subjects hereof. Neither this Warrant nor any term hereof may be amended, waived, discharged, or terminated other than by a written instrument signed by the party against whom enforcement of any such amendment, waiver, discharge or termination is sought.
- 10.4 Notices, etc.. All notices and other communications required or permitted hereunder shall be in writing and shall be deemed effectively given upon delivery to the party to be notified in person or by courier service or by registered or certified mail, postage prepaid, addressed (a) to the Holder, at the address set forth on the last page of this Warrant or at such other address as such Holder shall have furnished the Company in writing, or (b) if to the Company, at the address set forth on the last page of this Warrant and addressed to the attention of the Chief Executive Officer, or at such other address as the Company shall have furnished to the Holder.

Issued this 30^{th} day of November 2001.

COMSCORE NETWORKS, INC.

By: Name: /s/ James A. Powers

James A. Powers General Counsel and Corporate Secretary Title:

COMSTOCK PARTNERS, L.C.

Address: 1313 Beverly Road

Suite 302 McLean, VA 22101

By: Christopher Clemente

Title: Manager

6

EXHIBIT A

TO: COMSCORE NETWORKS, INC.

Reston, VA 20190	
1. The undersigned hereby elects to purchaseshares of Common Stock (the "Common Stock	${f k}$ ") of COMSCORE NETWORKS, INC. pursuant to the terms of the attached Warrant.
2. The undersigned elects to exercise the attached Warrant and tenders herewith payment in full for the pro-	urchase price of the shares being purchased, together with all applicable transfer taxes, if any.
3. Please issue a certificate or certificates representing said shares of Common Stock in the name of the u	indersigned or in such other name as is specified below:
(Nam	ue)
(Addre	ess)
4. The undersigned hereby represents and warrants that the aforesaid shares of Common Stock are being connection with the distribution thereof, and that the undersigned has no present intention of distributing or rest Representation Statement in substantially the form attached to the Warrant as Exhibit B .	
(S	Signature)
Ti	ïtle:
(Date)	

The Securities maybe resold in certain limited circumstances subject to the provisions of Rule 144, which requires among other things: (1) the availability of certain public information about the Company, (2) the resale occurring not less than one year after the party has purchased, and made full payment for, within the meaning of Rule 144, the securities to be sold; and, in the case of an affiliate, or of a nonaffiliate who has held the securities less than two years, (2) the sale being made through a broker in an unsolicited "**broker's transaction**" or in transactions directly with a market maker (as said term is defined under the Securities Exchange Act of 1934) and the amount of securities being sold during any three month period not exceeding the specified limitations stated therein, if applicable.

(e) Purchaser agrees, in connection with the Company's initial underwritten public offering of the Company's securities (1) not to sell, make short sale of, loan, grant any options for the purchase of, or otherwise dispose of any shares of Common Stock of the Company held by me (other than those shares included in the registration) without the prior written consent of the Company and the underwriters managing such initial underwritten public offering of the Company's securities for one hundred eighty (180) days from the effective date of such registration, and (2) Purchaser further agrees to execute any agreement reflecting the above as may be requested by the underwriters at the time of the public offering.

(f) Purchaser further understands that in the event all of the applicable requirements of Rule 144 are not satisfied, registration under the Securities Act, compliance with Regulation A, or some other registration exemption will be required; and that, notwithstanding the fact that Rule 144 and Regulation A are not exclusive, the Staff of the SEC has expressed its opinion that persons proposing to sell private placement securities other than in a registered offering and otherwise than pursuant to Rule 144 will have a substantial burden of proof in establishing that an exemption from registration is available for such offers or sales, and that such persons and their respective brokers who participate in such transactions do so at their own risk.

Exhibit 4.8

THIS WARRANT HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 AS AMENDED, OR ANY STATE SECURITIES LAWS. NO SALE OR DISPOSITION MAY BE EFFECTED WITHOUT (i) EFFECTIVE REGISTRATION STATEMENTS RELATED THERETO, (ii) AN OPINION OF COUNSEL OR OTHER EVIDENCE, REASONABLY SATISFACTORY TO THE COMPANY, THAT SUCH REGISTRATIONS ARE NOT REQUIRED, (iii) RECEPT OF NO-ACTION LETTERS FROM THE APPROPRIATE GOVERNMENTAL AUTHORITIES, OR (iv) OTHERWISE COMPLYING WITH THE PROVISIONS OF SECTION 7 OF THIS WARRANT.

COMSCORE NETWORKS, INC.

WARRANT TO PURCHASE 12,000 SHARES OF COMMON STOCK

THIS CERTIFIES THAT, for value received, Comstock Partners LLC is entitled to subscribe for and purchase 12,000 shares of fully paid and nonassessable Common Stock (as adjusted pursuant to Section 4 hereof, the "Shares") of COMSCORE NETWORKS, INC., a Delaware corporation (the "Company"), at the price of \$3.00 per share (such price and such other price as shall result, from time to time, from the adjustments specified in Section 4 hereof is herein referred to as the "Warrant Price"), subject to the provisions and upon the terms and conditions hereinafter set forth. As used herein, the term "Date of Grant" shall mean July 3, 2002

1. <u>Term</u>. The purchase right represented by this Warrant is exercisable, in whole or in part, at any time and from time to time from the Date of Grant until July 3, 2012; provided, however, if the underwriter of an initial public offering of the Company's stock provides the holder of this Warrant reasonable prior written notice requesting such holder to exercise its option to purchase the Shares, the holder shall within a reasonable period of time either exercise its rights under this Warrant or waive its right to exercise.

Notwithstanding the term of this Warrant fixed pursuant to the above paragraph, the right to purchase the Shares as granted herein shall expire, if not previously exercised, immediately upon the closing of a sale, conveyance, disposal or encumbrance of all or substantially all of the Company's property or business, or the Company's merger into or consolidation with any other corporation, or any other transaction or series of related transactions in which more than fifty percent (50%) of the voting power of the Company is disposed of (a "Merger"), provided that the term "Merger" shall not apply to a merger effected exclusively for the purpose of changing the domicile of the Company.

The Company shall provide the holder of this Warrant with at least twenty (20) days' written notice prior to the closing thereof of the terms and conditions of a Merger. However, if the Company fails to deliver such written notice, then notwithstanding anything to the contrary in this Warrant, the rights to purchase the Shares shall not expire until twenty (20) days after the Company complies with such notice provisions. If such closing does not take place, the Company shall promptly notify the holder of the Warrant that such proposed transaction has been terminated, and the holder of the Warrant may rescind any exercise of its purchase rights promptly after such notice

of termination of the proposed transaction if the exercise of the Warrant has occurred after the Company notified the holder that the Merger was proposed. In the event of such rescission, the Warrant will continue to be exercisable on the same terms and conditions contained herein.

- 2. Method of Exercise: Payment; Issuance of New Warrant. Subject to Section 1 hereof, the purchase right represented by this Warrant may be exercised by the holder hereof, in whole or in part and from time to time, at the election of the holder hereof, by (a) the surrender of this Warrant (with the notice of exercise substantially in the form attached hereto as Exhibit A-l duly completed and executed) at the principal office of the Company and by the payment to the Company, by certified or bank check, or by wire transfer to an account designated by the Company (a "Wire Transfer") of an amount equal to the then applicable Warrant Price multiplied by the number of Shares then being purchased; or (b) exercise of the "net issuance" right provided for in Section 9.1 hereof. The person or persons in whose name(s) any certificate(s) representing any of the Shares shall be issuable upon exercise of this Warrant shall be deemed to have become the holder(s) of record of, and shall be treated for all purposes as the record holder(s) of, the Shares represented thereby (and such Shares shall be deemed to have been issued) immediately prior to the close of business on the date or dates upon which this Warrant is exercised. In the event of any exercise of the rights represented by this Warrant, certificates for the Shares so purchased shall be delivered to the holder hereof as soon as possible and in any event within thirty (30) days after such exercise and, unless this Warrant has been fully exercised or expired, a new Warrant representing the portion of the Shares, if any, with respect to which this Warrant shall not then have been exercised shall also be issued to the holder hereof as soon as possible and in any event within such thirty-day period; provided, however, at such time as the Company is subject to the reporting requirements of the Securities Exchange Act of 1934, as amended, if requested by the holder of this Warrant, the Company shall use reasonable commercial efforts to cause its transfer agent to deliver the
- 3. <u>Stock Fully Paid</u>; <u>Reservation of Shares</u>. All Shares that may be issued upon the exercise of the rights represented by this Warrant will, upon issuance pursuant to the terms and conditions herein, be fully paid and nonassessable, and free from all preemptive rights and taxes, liens and charges with respect to the issue thereof. During the period within which the rights represented by this Warrant may be exercised, the Company will at all times have authorized and reserved for the purpose of issuance upon exercise of the purchase rights evidenced by this Warrant, a sufficient number of shares of its Common Stock to provide for the exercise of the rights represented by this Warrant.
- 4. <u>Adjustment of Warrant Price and Number of Shares</u>. The number and kind of securities purchasable upon the exercise of this Warrant and the Warrant Price shall be subject to adjustment from time to time upon the occurrence of certain events, as follows:
- (a) Reclassifications or Changes. In case of any reclassification or change of securities of the class issuable upon exercise of this Warrant (other than a change in par value, or from par value to no par value, or from no par value, or as a result of a subdivision or

combination), the Company shall duly execute and deliver to the holder of this Warrant a new Warrant (in form and substance satisfactory to the holder of this Warrant), or the Company shall make appropriate provision without the issuance of a new Warrant, so that the holder of this Warrant shall have the right to receive upon exercise of this Warrant, at a total purchase price not to exceed that payable upon the exercise of the unexercised portion of this Warrant, and in lieu of the Shares theretofore issuable upon exercise of this Warrant, (i) the kind and amount of shares of stock, other securities, money and property receivable upon such reclassification or change by a holder of the number of Shares then purchasable under this Warrant. Any new Warrant shall provide for adjustments that shall be as nearly equivalent as may be practicable to the adjustments provided for in this Section 4. The provisions of this Section 4(a) shall similarly apply to successive reclassifications and changes.

- (b) <u>Subdivision or Combination of Shares</u>. If the Company at any time while this Warrant remains outstanding and unexpired shall subdivide or combine its outstanding shares of Common Stock, the Warrant Price shall be proportionately decreased and the number of Shares issuable hereunder shall be proportionately increased in the case of a subdivision, and the Warrant Price shall be proportionately increased and the number of Shares issuable hereunder shall be proportionately decreased in the case of a combination.
- (c) <u>Stock Dividends and Other Distributions</u>. If the Company at any time while this Warrant is outstanding and unexpired shall (i) pay a dividend with respect to its Common Stock payable in Common Stock, then the Warrant Price shall be adjusted, from and after the date of determination of stockholders entitled to receive such dividend or distribution, to that price determined by multiplying the Warrant Price in effect immediately prior to such date of determination by a fraction (A) the numerator of which shall be the total number of shares of Common Stock outstanding immediately prior to such dividend or distribution, and (B) the denominator of which shall be the total number of shares of Common Stock outstanding immediately after such dividend or distribution; or (ii) make any other distribution with respect to Common Stock (except any distribution specifically provided for in Sections 4(a) and 4(b)), then, in each such case, provision shall be made by the Company such that the holder of this Warrant shall receive upon exercise of this Warrant a proportionate share of any such dividend or distribution as though it were the holder of the Shares as of the record date fixed for the determination of the stockholders of the Company entitled to receive such dividend or distribution.
- (d) <u>Adjustment of Number of Shares</u>. Upon each adjustment in the Warrant Price, the number of Shares of Common Stock purchasable hereunder shall be adjusted, to the nearest whole share, to the product obtained by multiplying the number of Shares purchasable immediately prior to such adjustment in the Warrant Price by a fraction, the numerator of which shall be the Warrant Price immediately prior to such adjustment and the denominator of which shall be the Warrant Price immediately thereafter.
- 5. Notice of Adjustments. Whenever the Warrant Price or the number of Shares purchasable hereunder shall be adjusted pursuant to Section 4 hereof, the Company shall issue a certificate signed by its chief financial officer setting forth, in reasonable detail, the event requiring the adjustment, the amount of the adjustment, the method by which such adjustment was calculated,

and the Warrant Price and the number of Shares purchasable hereunder after giving effect to such adjustment, and shall cause copies of such certificate to be mailed (without regard to Section 13 hereof, by first class mail, postage prepaid) to the holder of this Warrant.

6. <u>Fractional Shares</u>. No fractional shares of Common Stock will be issued in connection with any exercise hereunder, but in lieu of such fractional shares the Company shall make a cash payment therefor based on the fair market value of the Common Stock on the date of exercise as reasonably determined in good faith by the Company's Board of Directors.

7. Compliance with Act; Disposition of Warrant or Shares of Common Stock.

(a) <u>Compliance with Act</u>. The holder of this Warrant, by acceptance hereof, agrees that this Warrant, and the Shares to be issued upon exercise hereof are being acquired for investment and that such holder will not offer, sell or otherwise dispose of this Warrant, or any Shares to be issued upon exercise hereof except under circumstances which will not result in a violation of the Act or any applicable state securities laws. Upon exercise of this Warrant, unless the Shares being acquired are registered under the Securities Act of 1933, as amended (the "Act"), and any applicable state securities laws or an exemption from such registration is available, the holder hereof shall confirm in writing that the Shares so purchased are being acquired for investment and not with a view toward distribution or resale in violation of the Act and shall confirm such other matters related thereto as maybe reasonably requested by the Company. This Warrant and all Shares issued upon exercise of this Warrant (unless registered under the Act and any applicable state securities laws) shall be stamped or imprinted with a legend in substantially the following form:

"THE SECURITIES EVIDENCED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES LAWS. NO SALE OR DISPOSITION MAY BE EFFECTED WITHOUT (i) EFFECTIVE REGISTRATION STATEMENTS RELATED THERETO, (ii) AN OPINION OF COUNSEL OR OTHER EVIDENCE, REASONABLY SATISFACTORY TO THE COMPANY, THAT SUCH REGISTRATIONS ARE NOT REQUIRED, (iii) RECEIPT OF NO-ACTION LETTERS FROM THE APPROPRIATE GOVERNMENTAL AUTHORITIES, OR (iv) OTHERWISE COMPLYING WITH THE PROVISIONS OF SECTION 7 OF THE WARRANT UNDER WHICH THESE SECURITIES WERE ISSUED, DIRECTLY OR INDIRECTLY."

Said legend shall be removed by the Company, upon the request of a holder, at such time as the restrictions on the transfer of the applicable security shall have terminated. In addition, in connection with the issuance of this Warrant, the holder specifically represents to the Company by acceptance of this Warrant as follows:

(1) The holder is aware of the Company's business affairs and financial condition, and has acquired information about the Company sufficient to reach an informed and knowledgeable decision to acquire this Warrant. The holder is acquiring this Warrant for its own account for investment purposes only and not with a view to, or for the resale in connection with, any "distribution" thereof in violation of the Act.

- (2) The holder understands that this Warrant has not been registered under the Act in reliance upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of the holder's investment intent as expressed herein.
- (3) The holder further understands that this Warrant must be held indefinitely unless subsequently registered under the Act and qualified under any applicable state securities laws, or unless exemptions from registration and qualification are otherwise available. The holder is aware of the provisions of Rule 144 promulgated under the Act.
 - (4) The holder is an "accredited investor" as such term is defined in Rule 501 of Regulation D promulgated under the Act.
- (b) <u>Disposition of Warrant or Shares</u>. With respect to any offer, sale or other disposition of this Warrant or any Shares acquired pursuant to the exercise of this Warrant prior to registration of such Warrant or Shares, the holder hereof agrees to give written notice to the Company prior thereto, describing briefly the manner thereof, together with a written opinion of such holder's counsel, or other evidence, if reasonably satisfactory to the Company, to the effect that such offer, sale or other disposition may be effected without registration or qualification (under the Act as then in effect or any federal or state securities law then in effect) of this Warrant or such Shares and indicating whether or not under the Act certificates for this Warrant or such Shares to be sold or otherwise disposed of require any restrictive legend as to applicable restrictions on transferability in order to ensure compliance with such law. Upon receiving such written notice and reasonably satisfactory opinion or other evidence, the Company, as promptly as practicable but no later than fifteen (15) days after receipt of the written notice, shall notify such holder may sell or otherwise dispose of this Warrant or such Shares, all in accordance with the terms of the notice delivered to the Company. If a determination has been made pursuant to this Section 7(b) that the opinion of counsel for the holder or other evidence is not reasonably satisfactory to the Company, the Company shall so notify the holder promptly with details thereof after such determination has been made. Notwithstanding the foregoing, this Warrant or such Shares may, as to such federal laws, be offered, sold or otherwise disposed of in accordance with Rule 144 or 144A under the Act, provided that the Company shall have been furnished with such information as the Company may reasonably request to provide a reasonable assurance that the provisions of Rule 144 or 144A have been satisfied. Each certificate representing this Warrant or the Shares thus transferred (
- (c) <u>Applicability of Restrictions</u>. Neither any restrictions of any legend described in this Warrant nor the requirements of Section 7(b) above shall apply to any transfer of, or grant of a security interest in, this Warrant (or the Shares obtainable upon exercise thereof) or any part hereof (i) to a partner of the holder if the holder is a partnership or to a member of the holder if the holder is a limited liability company, (ii) to a partnership of which the holder is a partner or to a limited liability company of which the holder is a member, or (iii) to any affiliate of the holder if the holder is a corporation; <u>provided</u>, <u>however</u>, in any such transfer, if applicable, the transferee shall on the

Company's request agree in writing to be bound by the terms of this Warrant as if an original holder hereof.

8. <u>Rights as Stockholders</u>. No holder of this Warrant, as such, shall be entitled to vote or receive dividends or be deemed the holder of the Shares or any other securities of the Company which may at any time be issuable upon the exercise hereof for any purpose, nor shall anything contained herein be construed to confer upon the holder of this Warrant, as such, any of the rights of a stockholder of the Company or any right to vote for the election of directors or upon any matter submitted to stockholders at any meeting thereof, or to receive notice of meetings, or to receive dividends or subscription rights or otherwise until this Warrant shall have been exercised and the Shares purchasable upon the exercise hereof shall have become deliverable, as provided herein.

9. Additional Rights.

9.1 Right to Convert Warrant into Stock: Net Issuance.

(a) <u>Right to Convert</u>. In addition to and without limiting the rights of the holder under the terms of this Warrant, the holder shall have the right to convert this Warrant or any portion thereof (the "Conversion Right") into Shares as provided in this Section 9.1 at any time or from time to time during the term of this Warrant. Upon exercise of the Conversion Right With respect to a particular number of Shares subject to this Warrant (the "Converted Warrant Shares"), the Company shall deliver to the holder (without payment by the holder of any exercise price or any cash or other consideration) that number of shares of fully paid and nonassessable Common Stock as is determined according to the following formula:

$$X = \frac{B - A}{Y}$$

Where: X = the number of shares of Common Stock that shall be issued to holder

Y = the fair market value of one share of Common Stock

A = the aggregate Warrant Price of the specified number of Converted Warrant Shares immediately prior to the exercise of the Conversion Right (i.e., the number of Converted Warrant Shares multiplied by the Warrant Price)

B = the aggregate fair market value of the specified number of Converted Warrant Shares (i.e., the number of Converted Warrant Shares multiplied by the fair market value of one Converted Warrant Share)

No fractional shares shall be issuable upon exercise of the Conversion Right, and, if the number of shares to be issued determined in accordance with the foregoing formula is other than a whole number, the Company shall pay to the holder an amount in cash equal to the fair market value of the resulting fractional share on the Conversion Date (as hereinafter defined). For purposes of

Section 9 of this Warrant, shares issued pursuant to the Conversion Right shall be treated as if they were issued upon the exercise of this Warrant.

- (b) Method of Exercise. The Conversion Right may be exercised by the holder by the surrender of this Warrant at the principal office of the Company together with a written statement (which may be in the form of Exhibit A-I) specifying that the holder thereby intends to exercise the Conversion Right and indicating the number of shares subject to this Warrant which are being surrendered (referred to in Section 9.1(a) hereof as the Converted Warrant Shares) in exercise of the Conversion Right. Such conversion shall be effective upon receipt by the Company of this Warrant together with the aforesaid written statement, or on such later date as is specified therein (the "Conversion Date"), and, at the election of the holder hereof, may be made contingent upon the closing of the sale of the Company's Common Stock to the public in a public offering pursuant to a Registration Statement under the Act (a "Public Offering"). Certificates for the shares issuable upon exercise of the Conversion Right and, if applicable, a new warrant evidencing the balance of the shares remaining subject to this Warrant, shall be issued as of the Conversion Date and shall be delivered to the holder within thirty (30) days following the Conversion Date.
 - (c) Determination of Fair Market Value. For purposes of this Section 10.1, "fair market value" of a share of Common Stock as of a particular date (the "Determination Date") shall mean:
- (i) If the Conversion Right is exercised in connection with and contingent upon a Public Offering, and if the Company's registration statement relating to such Public Offering ("Registration Statement") has been declared effective by the Securities and Exchange Commission, then the initial "Price to Public" specified in the final prospectus with respect to such offering.
 - (ii) If the Conversion Right is not exercised in connection with and contingent upon a Public Offering, then as follows:
- (A) If traded on a securities exchange, the fair market value of the Common Stock shall be deemed to be the average of the closing prices of the Common Stock on such exchange over the five trading days immediately prior to the Determination Date;
- (B) If traded on the Nasdaq National Market or other over-the-counter system, the fair market value of the Common Stock shall be deemed to be the average of the closing bid prices of the Common Stock over the five trading days immediately prior to the Determination Date; and
 - (C) If there is no public market for the Common Stock, then the fair market value of the Common Stock shall be reasonably determined by the board of directors of the Company.

In making a determination under clauses (A) or (B) above, if on the Determination Date, five trading days had not passed since the IPO, then the fair market value of the Common Stock shall be the average closing prices or closing bid prices, as applicable, for the shorter period beginning on and including the date of the IPO and ending on the trading day prior to the Determination Date (or if such period includes only one trading day the closing price or closing bid price, as applicable, for

such trading day). If closing prices or closing bid prices are no longer reported by a securities exchange or other trading system, the closing price or closing bid price shall be that which is reported by such securities exchange or other trading system at 4:00 p.m. New York City time on the applicable trading day.

- 9.2 Exercise Prior to Expiration. To the extent this Warrant is not previously exercised as to all of the Shares subject hereto, and if the fair market value of one Share is greater than the Warrant Price then in effect, this Warrant shall be deemed automatically exercised pursuant to Section 9.1 above (even if not surrendered) immediately before its expiration. For purposes of such automatic exercise, the fair market value of one Share upon such expiration shall be determined pursuant to Section 9.1(c). To the extent this Warrant or any portion thereof is deemed automatically exercised pursuant to this Section 9.2, the Company agrees to promptly notify the holder hereof of the number of Shares, if any, the holder hereof is to receive by reason of such automatic exercise.
 - 10. Representations and Warranties. The Company represents and warrants to the holder of this Warrant as follows:
- (a) This Warrant has been duly authorized and executed by the Company and is a valid and binding obligation of the Company enforceable in accordance with its terms, subject to laws of general application relating to bankruptcy, insolvency and the relief of debtors and the rules of law or principles at equity governing specific performance, injunctive relief and other equitable remedies.
- (b) The Shares have been duly authorized and reserved for issuance by the Company and, when issued in accordance with the terms hereof, will be validly issued, fully paid and nonassessable and free from preemptive rights.
- (c) The execution and delivery of this Warrant are not, and the issuance of the Shares upon exercise of this Warrant in accordance with the terms hereof will not be, inconsistent with the Company's Sixth Amended and Restated Certificate of Incorporation (the "Charter") or bylaws, do not and will not contravene any law, governmental rule or regulation, judgment or order applicable to the Company, and do not and will not conflict with or contravene any provision of, or constitute a default under, any indenture, mortgage, contract or other instrument of which the Company is a party or by which it is bound or require the consent or approval of, the giving of notice to, the registration or filing with or the taking of any action in respect of or by, any Federal, state or local government authority or agency or other person, except for the filing of notices pursuant to federal and state securities laws, which filings will be effected by the time required thereby.
- (d) There are no actions, suits, audits, investigations or proceedings pending or, to the knowledge of the Company, threatened against the Company in any court or before any governmental commission, board or authority which, if adversely determined, could have a material adverse effect on the ability of the Company to perform its obligations under this Warrant.

- 11. Market Stand-Off Provision. The holder of this Warrant agrees to be bound by the "Market Stand-Off" provision in section 1(l) of the Third Amended and Restated Investor Rights Agreement dated June 6, 2002 between the Company and certain of its investors and founders.
- 12. <u>Modification and Waiver</u>. This Warrant and any provision hereof may be changed, waived, discharged or terminated only by an instrument in writing signed by the party against which enforcement of the same is sought.
- 13. <u>Notices</u>. Any notice, request, communication or other document required or permitted to be given or delivered to the holder hereof or the Company shall be delivered, or shall be sent by certified or registered mail, postage prepaid, to each such holder at its address as shown on the books of the Company or to the Company at the address indicated therefor on the signature page of this Warrant.
- 14. <u>Lost Warrants or Stock Certificates</u>. The Company covenants to the holder hereof that, upon receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant or any stock certificate and, in the case of any such loss, theft or destruction, upon receipt of an indemnity reasonably satisfactory to the Company, or in the case of any such mutilation upon surrender and cancellation of such Warrant or stock certificate, the Company will make and deliver a new Warrant or stock certificate, of like tenor, in lieu of the lost, stolen, destroyed or mutilated Warrant or stock certificate.
- 15. <u>Descriptive Headings</u>. The descriptive headings of the various Sections of this Warrant are inserted for convenience only and do not constitute a part of this Warrant. The language in this Warrant shall be construed as to its fair meaning without regard to which party drafted this Warrant.
 - 16. Governing Law. This Warrant shall be construed and enforced in accordance with, and the rights of the parties shall be governed by, the laws of the State of Delaware.
- 17. <u>Survival of Representations</u>, <u>Warranties and Agreements</u>. All representations and warranties of the Company and the holder hereof contained herein shall survive the Date of Grant, the exercise or conversion of this Warrant (or any part hereof) or the termination or expiration of rights hereunder. All agreements of the Company and the holder hereof contained herein shall survive indefinitely until, by their respective terms, they are no longer operative.
- 18. Remedies. In case any one or more of the covenants and agreements contained in this Warrant shall have been breached, the holders hereof (in the case of a breach by the Company), or the Company (in the case of a breach by a holder), may proceed to protect and enforce their or its rights either by suit in equity and/or by action at law, including, but not limited to, an action for damages as a result of any such breach and/or an action for specific performance of any such covenant or agreement contained in this Warrant.

- 19. <u>Severability</u>. The invalidity or unenforceability of any provision of this Warrant in any jurisdiction shall not affect the validity or enforceability of such provision in any other jurisdiction, or affect any other provision of this Warrant, which shall remain in full force and effect.
- 20. Recovery of Litigation Costs. If any legal action or other proceeding is brought for the enforcement of this Warrant, or because of an alleged dispute, breach, default, or misrepresentation in connection with any of the provisions of this Warrant, the successful or prevailing party or parties shall be entitled to recover reasonable attorneys' fees and other costs incurred in that action or proceeding, in addition to any other relief to which it or they may be entitled.
- 21. Entire Agreement; Modification. This Warrant constitutes the entire agreement between the parties pertaining to the subject matter contained in it and supersedes all prior and contemporaneous agreements, representations, and undertakings of the parties, whether oral or written, with respect to such subject matter.

The Company has caused this Warrant to be duly executed and delivered as of the Date of Grant specified above.

COMSCORE NETWORKS, INC.

By: /s/ James A. Powers

Title: General Counsel and Corporate Secretary

Address:

11465 Sunset Hills Road Suite 200 Reston, VA 20190

EXHIBIT A-l

NOTICE OF EXERCISE

To: COMSCORE NETWORKS, INC. (the "Company")

1.	The un	idetsigned neteby.
	0	elects to purchase shares of Common Stock of the Company pursuant to the terms of the attached Warrant, and tenders herewith payment of the purchase price of such shares in full, or
	0	elects to exercise its net issuance rights pursuant to Section 9.1 of the attached Warrant with respect to shares of Common Stock.
2. Pleas	e issue a	certificate or certificates representing shares in the name of the undersigned or in such other name or names as are specified below:
		(Name)
		(Address)
		ned represents that the aforesaid shares are being acquired for the account of the undersigned for investment and not with a view to, or for resale in connection with, the distribution thereof and that the present intention of distributing or reselling such shares, all except as in compliance with applicable securities laws.
		(Signature)
(Da	ite)	_

Exhibit 4.9

THIS WARRANT HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED OR ANY STATE SECURITIES LAWS. NO SALE OR DISPOSITION MAY BE EFFECTED WITHOUT (i) EFFECTIVE REGISTRATION STATEMENTS RELATED THERETO, (ii) AN OPINION OF COUNSEL OR OTHER EVIDENCE, REASONABLY SATISFACTORY TO THE COMPANY, THAT SUCH REGISTRATIONS ARE NOT REQUIRED, (iii) RECEIPT OF NO-ACTION LETTERS FROM THE APPROPRIATE GOVERNMENTAL AUTHORITIES, OR (iv) OTHERWISE COMPLYING WITH THE PROVISIONS OF SECTION 7 OF THIS WARRANT.

COMSCORE NETWORKS, INC.

WARRANT TO PURCHASE 36,127 SHARES OF SERIES D PREFERRED STOCK

THIS CERTIFIES THAT, for value received, SILICON VALLEY BANK and its assignees are entitled to subscribe for and purchase 36,127 shares of the fully paid and nonassessable Series D Preferred Stock (as adjusted pursuant to Section 4 hereof, the "Shares") of COMSCORE NETWORKS, INC., a Delaware corporation (the "Company"), at the price \$0.8996 per share (such price and such other price as shall result, from time to time, from the adjustments specified in Section 4 hereof is herein referred to as the "Warrant Price"), subject to the provisions and upon the terms and conditions hereinafter set forth. As used herein, (a) the term "Series Preferred" shall mean the Company's presently authorized Series D Preferred Stock may hereafter be converted or exchanged, and after the automatic conversion of the Series D Preferred Stock to Common Stock shall mean the Company's Common Stock, (b) the term "Date of Grant" shall mean July 31, 2002, and (c) the term "Other Warrants" shall mean any other warrants issued by the Company in connection with the transaction with respect to which this Warrant was issued, and any warrant issued upon transfer or partial exercise of or in lieu of this Warrant. The term "Warrant" as used herein shall be deemed to include Other Warrants unless the context clearly requires otherwise.

- 1. <u>Term.</u> The purchase right represented by this Warrant is exercisable, in whole or in part, at any time and from time to time from the Date of Grant through the later of (i) ten (10) years after the Date of Grant or (ii) five (5) years after the closing of the Company's Initial public offering of its Common Stock ("IPO") effected pursuant to a Registration Statement on Form S-1 (or its successor) filed under the Securities Act of 1933, as amended (the "Act").
- 2. <u>Method of Exercise: Payment: Issuance of New Warrant.</u> Subject to Section 1 hereof, the purchase right represented by this Warrant may be exercised by the holder hereof, in whole or in part and from time to time, at the election of the holder hereof, by (a) the surrender of this Warrant (with the notice of exercise substantially in the form attached hereto as Exhibit A-l duly completed and executed) at the principal office of the Company and by the payment to the Company, by certified or bank check, or by wire transfer to an account designated by the Company (a "Wire Transfer") of an amount equal to the then applicable Warrant Price multiplied by the number of Shares then being purchased; or (b) exercise of the "net issuance" right provided for in Section 10.2

hereof. The person or persons in whose name(s) any certificate(s) representing shares of Series Preferred shall be issuable upon exercise of this Warrant shall be deemed to have become the holder(s) of, record of, and shall be treated for all purposes as the record holder(s) of, the shares represented thereby (and such shares shall be deemed to have been issued) immediately prior to the close of business on the date or dates upon which this Warrant is exercised. In the event of any exercise of the rights represented by this Warrant, certificates for the shares of stock so purchased shall be delivered to the holder hereof as soon as possible and in any event within thirty (30) days after such exercise and, unless this Warrant has been fully exercised or expired, a new Warrant representing the portion of the Shares, if any with respect to which this Warrant shall not have been exercised shall also be issued to the holder hereof as soon as possible and in any event within such thirty-day period; provided, however, at such time as the Company is subject to the reporting requirements of the Securities Exchange Act of 1934, as amended, if requested by the holder of this Warrant, the Company shall use its best efforts to cause its transfer agent to deliver the certificate representing Shares issued upon exercise of this Warrant to a broker or other person (as directed by the holder exercising this Warrant) within the time period required to settle any trade made by the holder after exercise of this Warrant.

- 3. <u>Stock Fully Paid</u>; <u>Reservation of Shares</u>. All Shares that may be issued upon the exercise of the rights represented by this Warrant will, upon issuance pursuant to the terms and conditions herein, be fully paid and nonassessable, and free from all preemptive rights and taxes, liens and charges with respect to the issue thereof. During the period within which the rights represented by this Warrant may be exercised, the Company will at all times have authorized, and reserved for the purpose of the issue upon exercise of the purchase rights evidenced by this Warrant, a sufficient number of shares of its Series Preferred to provide for the exercise of the rights represented by this Warrant and a sufficient number of shares of its Common Stock to provide for the conversion of the Series Preferred into Common Stock.
- 4. <u>Adjustment of Warrant Price and Number of Shares</u>. The number and kind of securities purchasable upon the exercise of this Warrant and the Warrant Price shall be subject to adjustment from time to time upon the occurrence of certain events, as follows:
- (a) <u>Reclassification or Merger</u>. In case of any reclassification or change of securities of the class issuable upon exercise of this Warrant (other than a change in par value, or from par value to no par value, or from par value to par value, or as a result of a subdivision or combination), or in case of any merger of the Company with or into another corporation (other than a merger with another corporation in which the Company is the acquiring and the surviving corporation and which does not result in any reclassification or change of outstanding securities issuable upon exercise of this Warrant), or in case of any sale of all or substantially all of the assets of the Company, the Company, or such successor or purchasing corporation, as the case may be, shall duly execute and deliver to the holder of this Warrant (in form and substance satisfactory to the holder of this Warrant), or the Company shall make appropriate provision without the issuance of a new Warrant, so that the holder of this Warrant shall have the right to receive upon exercise of this Warrant, at a total purchase price not to exceed that payable upon the exercise of the unexercised portion of this Warrant, and in lieu of the shares of Series Preferred theretofore issuable upon exercise of this Warrant, the kind and amount of shares of stock, other securities, money and

property receivable upon such classification, change, merger or sale by a holder of the number of shares of Series Preferred then purchasable under this Warrant. Any new Warrant shall provide for adjustments that shall be as nearly equivalent as may be practicable to the adjustments provided for in this Section 4. The provisions of this Section 4(a) shall similarly apply to successive reclassifications, changes, mergers and sales.

- (b) <u>Subdivision or Combination of Shares</u>. If the Company at any time while this Warrant remains outstanding and unexpired shall subdivide or combine its outstanding shares of Series Preferred, the Warrant Price shall be proportionately decreased and the number of Shares issuable hereunder shall be proportionately increased in the case of a subdivision and the Warrant Price shall be proportionately increased and the number of Shares issuable hereunder shall be proportionately decreased in the case of a combination.
- (c) <u>Stock Dividends and Other Distributions</u>. If the Company at any time while this Warrant is outstanding and unexpired shall (i) pay a dividend with respect to Series Preferred payable in Series Preferred, then the Warrant Price shall be adjusted, from and after the date of determination of shareholders entitled to receive such dividend or distribution, to that price determined by multiplying the Warrant Price in effect immediately prior to such date of determination by a fraction (A) the numerator of which shall be the total number of shares of Series Preferred outstanding immediately prior to such dividend or distribution, and (B) the denominator of which shall be the total number of shares of Series Preferred outstanding immediately prior to such dividend or distribution with respect to Series Preferred (except any distribution specifically provided for in Sections 4(a) and 4(b)), then, in each such case, provision shall be made by the Company such that the holder of this Warrant shall receive upon exercise of this Warrant a proportionate share of any such dividend or distribution as though it were the holder of the Series Preferred (or Common Stock issuable upon conversion thereof) as of the record date fixed for the determination of the shareholders of the Company entitled to receive such dividend or distribution.
- (d) <u>Adjustment of Number of Shares</u>. Upon each adjustment in the Warrant Price, the number of Shares of Series Preferred purchasable hereunder shall be adjusted, to the nearest whole share, to the product obtained by multiplying the number of Shares purchasable immediately prior to such adjustment in the Warrant Price by a fraction, the numerator of which shall be the Warrant Price immediately prior to such adjustment and the denominator of which shall be the Warrant Price immediately thereafter.
- (e) <u>Antidilution Rights</u>. The other antidulation rights applicable to the Shares of Series Preferred purchasable hereunder are set forth in the Company's Certificate of Incorporation, as amended through the Date of Grant, a true and complete copy of which is attached hereto as Exhibit B (the "Charter"). Such antidilution rights shall not be restated, amended, modified or waived in any manner that is adverse to the holder hereof without such holder's prior written consent, unless such amendment, modification or waiver affects such holder in the same manner as it affects other holders of only the Series Preferred. The Company shall promptly provide the holder hereof with any restatement, amendment, modification or waiver of the Charter promptly after the same has been made.

- 5. Notice of Adjustments. Whenever the Warrant Price or the number of Shares purchasable hereunder shall be adjusted pursuant to Section 4 hereof, the Company shall make a certificate signed by its chief financial officer setting forth, in reasonable detail, the event requiring the adjustment, the amount of the adjustment, the method by which such adjustment was calculated, and the Warrant Price and the number of Shares purchasable hereunder after giving effect to such adjustment, and shall cause copies of such certificate to be mailed (without regard to Section 13 hereof, by first class mail, postage prepaid) to this holder of this Warrant. In addition, whenever the conversion price or conversion ratio of "the Series Preferred shall be adjustment, and shall cause copies of such certificate to be mailed (without regard to Section 13 hereof, by first class mail, postage prepaid) to the holder of this Warrant.
- 6. <u>Fractional Shares</u>. No fractional shares of Series Preferred will be issued in connection, with any exercise hereunder, but in lieu of such fractional shares the Company shall make a cash payment therefor based on the fair market value of the Series Preferred on the date of exercise as reasonably determined in good faith by the Company's Board of Directors.
 - Compliance with Act; Disposition of Warrant or Shares of Series Preferred
- (a) Compliance with Act. The holder of this Warrant by acceptance hereof, agrees that this Warrant, and the shares of Series Preferred to be issued upon exercise hereof and any Common Stock issued upon conversion thereof are being acquired for investment and that such holder will not offer, sell or otherwise dispose of this Warrant, or any shares of Series Preferred to be issued upon exercise hereof or any Common Stock issued upon conversion thereof except under circumstances which will not result in a violation of the Act or any applicable state securities laws. Upon exercise of this Warrant, unless the Shares being acquired are registered under the Act and any applicable state securities laws or an exemption from such registration is available, the holder hereof shall confirm in writing that the shares of Series Preferred so purchased (and any shares of Common Stock issued upon conversion thereof) are being acquired for investment and not with a view toward distribution or resale in violation of the Act and shall confirm such other matters related thereto as may be reasonably requested by the Company. This Warrant and all shares of Series Preferred issued upon exercise of this Warrant and all shares of Common Stock issued upon conversion thereof (unless registered under the Act and any applicable state securities laws) shall be stamped or imprinted with a legend in substantially the following form:

"THE SECURITIES EVIDENCED HERBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES LAWS. NO SALE OR DISPOSITION MAY BE EFFECTED WITHOUT (i) EFFECTIVE REGISTRATION STATEMENTS RELATED THERETO, (ii) AN OPINION OF COUNSEL OR OTHER EVIDENCE REASONABLY SATISFACTORY TO THE COMPANY, THAT SUCH REGISTRATIONS ARE NOT REQUIRED, (iii) RECEIPT OF NO-ACTION LETTERS FROM THE APPROPRIATE GOVERNMENTAL AUTHORITIES, OR (iv) OTHERWISE COMPLYING

WITH THE PROVISIONS OF SECTION 7 OF THE WARRANT UNDER WHICH THESE SECURITIES WERE ISSUED, DIRECTLY OR INDIRECTLY."

Said legend shall be removed by the Company, upon the request of a holder, at such time as the restrictions on the transfer of the applicable security shall have terminated. In addition, in connection with this issuance of this Warrant, the holder specifically represents to the Company by acceptance of this Warrant as follows:

- (1) The holder is aware of this Company's business affairs and financial condition, and has acquired information about the Company sufficient to reach an informed and knowledgeable decision to acquire this Warrant. The holder is acquiring this Warrant for its own account for investment purposes only and not with a view to, or for the resale in connection with, any "distribution" thereof in violation of the Act.
- (2) The holder understands that this Warrant has not been registered under the Act in reliance upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of the holder's investment intent as expressed herein.
- (3) The holder further understands that this Warrant must be held indefinitely unless subsequently registered under the Act and qualified under any applicable state securities laws, or unless exemptions from registration and qualification are otherwise available. The holder is aware of the provisions of Rule 144 promulgated under the Act.
 - (4) The holder is an "accredited investor" as such term is defined in Rule 501 of Regulation D promulgated under the Act.
- (b) <u>Disposition of Warrant or Shares</u>. With respect to any offer, sale or other disposition of this Warrant or any shares of Series Preferred acquired pursuant to the exercise of this Warrant prior to registration of such Warrant or shares, the holder hereof agrees to give written notice to the Company prior thereto, describing briefly the manner thereof, together with a written opinion of such holder's counsel, or other evidence, if reasonably satisfactory to the Company, to the effect that such offer, sale or other disposition may be effected without registration or qualification (under the Act as then in effect or any federal or state securities law then in effect) of this Warrant or such shares of Series Preferred or Common Stock and indicating whether or not under the Act certificates for this Warrant or such shares of Series Preferred to be sold or otherwise disposed of require any restrictive legend as to applicable restrictions on transferability in order to ensure compliance with such law. Upon receiving such written notice and reasonably satisfactory opinion or other evidence, the Company, as promptly as practicable but no later than fifteen (15) days after receipt of the written notice, shall notify such holder that such holder may sell or otherwise dispose of this Warrant or such shares of Series Preferred or Common Stock, all in accordance with the terms of the notice delivered to the Company. If a determination has been made pursuant to this Section 7(b) that the opinion of counsel for the holder or other evidence is not reasonably satisfactory to the Company, the Company shall so notify the holder promptly with details thereof after such determination has been made. Notwithstanding the foregoing, this Warrant or such shares of Series Preferred or Common Stock may, as to such federal laws, be offered, sold or otherwise disposed of

in accordance with Rule 144 or 144A under the Act, provided that the Company shall have been furnished with such information as the Company may reasonably request to provide a reasonable assurance that the provisions of Rule 144 or 144A have been satisfied. Each certificate representing this Warrant or the shares of Series Preferred thus transferred (except a transfer pursuant to Rule 144 or 144A) shall bear a legend as to the applicable restrictions on transferability in order to ensure compliance with such laws, unless in the aforesaid opinion of counsel for the holder, such legend is not required in order to ensure compliance with such laws. The Company may issue stop transfer instructions to its transfer agent in connection with such restrictions.

- (c) <u>Applicability of Restrictions</u>. Neither any restrictions of any legend described in this Warrant nor the requirements of Section 7(b) above shall apply to any transfer of, or grant of a security interest in, this Warrant (or the Series Preferred or Common Stock obtainable upon exercise thereof) or any part hereof (i) to a partner of the holder if the holder is a partnership or to a member of the holder if the holder is a limited liability company, or (ii) to Silicon Valley Bancshares (holder's parent company) or any affiliate of the holder if the holder is a corporation or a bank; <u>provided</u>, <u>however</u>, in any such transfer, (x) the transferee shall on the Company's request agree in writing to be bound by the terms of this Warrant as if an original holder hereof, and (z) other than the transfer to Silicon Valley Bancshares the transferor shall give the Company prior written notice thereof in reasonable detail, including the name of the transferee and the extent of the rights and/or number of shares to be transferred. Subject to the provisions of this Section 7(c), upon receipt by holder of the executed Warrant, holder will transfer all or part of this Warrant or the Shares issuable, directly or indirectly, upon conversion of the Shares, if any) to Silicon Valley Bancshares, holder's parent company. Subject to the provisions of this Section 7(c) and upon providing Company with written notice, holder or Silicon Valley Bancshares may transfer all or part of this Warrant or the Shares issuable upon exercise of this Warrant (or the securities issuable, directly or indirectly, upon conversion of the Shares, if any) to The Silicon Valley Bank Foundation.
- 8. <u>Rights as Shareholders</u>; <u>Information</u>. No holder of this Warrant, as such, shall be entitled to vote or receive dividends or be deemed the holder of Series Preferred or any other securities of the Company which may at any time be issuable upon the exercise hereof for any purpose, nor shall anything contained herein be construed to confer upon the holder of this Warrant, as such, any of the rights of a shareholder of the Company or any right to vote for the election of directors or upon any matter submitted to shareholders at any meeting thereof, or to receive notice of meetings, or to receive dividends or subscription rights or otherwise until this Warrant shall have been exercised and the Shares purchasable upon the exercise hereof shall have become deliverable, as provided herein. Notwithstanding the foregoing, the Company will transmit to the holder of this Warrant such information, documents and reports as are generally distributed to the holders of any class or series of the Securities of the Company concurrently with the distribution to the shareholders.
- 9. <u>Registration Rights</u>. The Company grants registration rights to the holder of this Warrant for any Common Stock of the Company obtained upon conversion of the Series Preferred, comparable to the registration rights granted to the investors in that certain Second Amended and

Restated Investor Rights Agreement dated as of August 8, 2001, (the "Registration Rights Agreement"), with the following exceptions and clarifications:

- (1) The holder will have not have the right to demand registration, but can otherwise participate in any registration demanded by others other holders of at least a majority of the Registrable Securities (as defined in the Rights Agreement).
 - (2) The holder will be subject to the same provisions regarding indemnification as contained in the Registration Rights Agreement.
 - (3) The registration rights are freely assignable by the holder offers Warrant in connection with a permitted transfer of this Warrant or the Shares.

10. Additional Rights.

10.1 Acquisition Transactions. The Company shall provide the holder of this Warrant with at least twenty (20) days' written notice prior to closing thereof of the terms and conditions of any of the following transactions (to the extent the Company has notice thereof): (i) the sale, lease, exchange, conveyance or other disposition of all or substantially all of the Company's property or business, or (ii) its merger into or consolidation with any other corporation (other than a wholly-owned subsidiary of the Company), or any transaction (including a merger or other reorganization) or series of related transactions, in which more than 50% of the voting power of the Company is disposed of.

10.2 Right to Convert Warrant into Stock: Net Issuance.

(a) <u>Right to Convert</u>. In addition to and without limiting the rights of the holder under the terms of this Warrant, the holder shall have the right to convert this Warrant or any portion thereof (the "Conversion Right") into shares of Series Preferred as provided in this Section 10.2 at any time or from time to time during the term of this Warrant. Upon exercise of the Conversion Right with respect to a particular number of shares subject to this Warrant (the "Convened Warrant Shares"), the Company shall deliver to the holder (without payment by the holder of any exercise price or any cash or other consideration) that number of shares of fully paid and nonassessable Series Preferred as is determined according to the following formula:

$$X = \frac{B - A}{Y}$$

Where: X = the number of shares of Series Preferred that shall be issued to holder

Y = the fair market value of one share of Series Preferred

A = the aggregate Warrant Price of the specified number of Converted Warrant Shares immediately prior to the exercise of the Conversion Right (i.e., the number of Converted Warrant Shares multiplied by the Warrant Price)

B = the aggregate fair market value of the specified number of Converted Warrant Shares (i.e., the number of Converted Warrant Shares multiplied by the fair market value of one Convened Warrant Share)

No fractional shares shall be issuable upon exercise of the Conversion Right, and, if the number of shares to be issued determined in accordance with the foregoing formula is other than a whole number, the Company shall pay to the holder an amount in cash equal to the fair market value of the resulting fractional share on the Conversion Date (as hereinafter defined). For purposes of Section 10 of this Warrant, shares issued pursuant to the Conversion Right shall be treated as if they were issued upon the exercise of this Warrant.

- (b) Method of Exercise. The Conversion Right may be exercised by the holder by the surrender of this Warrant at the principal office of the Company together with a written statement (which may be in the form of Exhibit A-1 or Exhibit A-2 hereto) specifying that the holder thereby intends to exercise the Conversion Right and indicating the number of shares subject to this Warrant which are being surrendered (referred to in Section 10.2(a) hereof as the Converted Warrant Shares) in exercise of the Conversion Right. Such conversion shall be effective upon receipt by the Company of this Warrant together with the aforesaid written statement, or on such later date as is specified therein (the "Conversion Date"), and, at the election of the holder hereof, may be made contingent upon the closing of the sale of the Company's Common Stock to the public in a public offering pursuant to a Registration Statement under the Act (a "Public Offering"). Certificates for the shares issuable upon exercise of the Conversion Right and, if applicable, a new warrant evidencing the balance of the shares remaining subject to this Warrant, shall be issued as of the Conversion Date and shall be delivered to the holder within thirty (30) days following the Conversion Date.
- (c) <u>Determination of Fair Market Value</u>. For purposes of this Section 10.2, "fair market value" of a share of Series Preferred (or Common Stock if the Series Preferred has been automatically converted into Common Stock) as of a particular date (the "Determination Date") shall mean:
- (i) If the Conversion Right is exercised in connection with and contingent upon a Public Offering, and if the Company's Registration Statement relating to such Public Offering ("Registration Statement") has been declared effective by the Securities and Exchange Commission, then the initial "Price to Public" specified in the final prospectus with respect to such offering.
 - (ii) If the Conversion Right is not exercised in connection with and contingent upon a Public Offering, then as follows:
- (A) If traded on securities exchange, the fair market value of the Common Stock shall be deemed to be the average of the closing prices of the Common Stock on such exchange over the five trading days immediately prior to the Determination Date, and the fair market value of the Series Preferred shall be deemed to be such fair market value of the Common Stock multiplied by the number of shares of Common Stock into which each share of Series Preferred is then convertible;

- (B) If traded on the Nasdaq Stock Market or other over-the-counter system, the fair market value of the Common Stock shall be deemed to be the average of the closing bid prices of the Common Stock over the five trading days immediately prior to the Determination Date, and the fair market value of the Series Preferred shall be deemed to be such fair market value of the Common Stock multiplied by the number of shares of Common Stock into which each share of Series Preferred is then convertible; and
 - (C) If there is no public market for the Common Stock, then fair marker value shall be determined in good faith by the board of directors of the Company.

In making a determination under clauses (A) or (B) above, if on the Determination Date, five trading days had not passed since the IPO, then the fair market value of the Common Stock shall be the average closing prices or closing bid prices, as applicable, for the shorter period beginning on and including the date of the IPO and ending on the trading day prior to the Determination Date (or if such period includes only one trading day the closing price or closing bid price, as applicable, for such trading day). If closing prices or closing bid prices are no longer reported by a securities exchange or other trading system, the closing price or closing bid price shall be that which is reported by such securities exchange or other trading system at 4:00 p.m. New York City time on the applicable trading day.

- 10.3 Exercise Prior to Expiration. To the extent this Warrant is not previously exercised as to all of the Shares subject hereto, and if the fair market value of one share of the Series Preferred is greater than the Warrant Price then in effect, this Warrant shall be deemed automatically exercised pursuant to Section 10.2 above (even if not surrendered) immediately before its expiration. For purposes of such automatic exercise, the fair market value of one share of the Series Preferred upon such expiration shall be determined pursuant to Section 10.2(c). To the extent this Warrant or any portion thereof is deemed automatically exercised pursuant to this Section 10.3, the Company agrees to promptly notify the holder hereof of the number of Shares, if any, the holder hereof is to receive by reason of such automatic exercise.
 - 11. Representations and Warranties. The Company represents and warrants to the holder of this Warrant as follows:
- (a) This Warrant has been duly authorized and executed by the Company and is a valid and binding obligation of the Company enforceable in accordance with its terms, subject to laws of general application relating to bankruptcy, insolvency and the relief of debtors and the rules of law or principles at equity governing specific performance, injunctive relief and other equitable remedies.
- (b) The Shares have been duly authorized and reserved for issuance by the Company and, when issued in accordance with the terms hereof, will be validly issued, fully paid and nonassessable and free from preemptive rights.
 - (c) The rights, preferences, privileges and restrictions granted to or imposed upon the Series Preferred and the holders thereof are as set forth in the Charter, and on the Date of Grant,

each share of the Series Preferred represented by this Warrant is convertible into one share of Common Stock.

- (d) The shares of Common Stock issuable upon conversion of the Shares have been duly authorized and reserved for issuance by the Company and, when issued in accordance with the terms of the Charter will be validly issued, fully paid and nonassessable.
- (e) The execution and delivery of this Warrant are not, and the issuance of the Shares upon exercise of this Warrant in accordance with the terms hereof will not be inconsistent with the Company's Charter or by-laws, do not and will not contravene any law, governmental rule or regulation, judgment or order applicable to the Company, and do not and will not conflict with or contravene any provision of, or constitute a default under, any indenture, mortgage, contact or other instrument of which the Company is a party or by which it is bound or require the consent or approval of, the giving of notice to, the registration or filing with or the taking of any action in respect of or by, any Federal, state or local government authority or agency or other person, except for the filing of notices pursuant to federal and state securities laws, which filings will be effected by the time required thereby.
- (f) There are no actions, suits, audits, investigations or proceedings pending or, to the knowledge of the Company, threatened against the Company in any court or before any governmental commission, board or authority which, if adversely determined, could have a material adverse effect on the ability of the Company to perform its obligations under this Warrant.
- (g) The number of shares of Common Stock of the Company outstanding on the date hereof, on a fully dilated basis (assuming the conversion of all outstanding convertible securities and the exercise of all outstanding options and warrants), does not exceed 48,000,000 shares.
- 12. <u>Modification and Waiver</u>. This Warrant and any provision hereof may be changed, waived, discharged or terminated only by an instrument in writing signed by the party against which enforcement of the same is sought.
 - 13. Market Stand-off. The holder of this Warrant agrees to be bound by the "Market Stand-Off provision in Section 1(1) of the Rights Agreement.
- 14. <u>Notices</u>. Any notice, request, communication or other document required or permitted to be given or delivered to the holder hereof or the Company shall be delivered, or shall be sent by certified or registered mail, postage prepaid, to each such holder at its address as shown on the books of the Company or to the Company at the address indicated therefor on the signature page of this Warrant.
- 15. <u>Binding Effect on Successors</u>. This Warrant shall be binding upon any corporation succeeding the Company by merger, consolidation or acquisition of all or substantially all of the Company's assets, and all of the obligations of the Company relating to the Series Preferred issuable upon the exercise or conversion of this Warrant shall survive the exercise, conversion and

termination of this Warrant and all of the covenants and agreements of the Company shall inure to the benefit of the successors and assigns of the holder hereof.

- 16. Lost Warrants or Stock Certificates. The Company covenants to the holder hereof that, upon receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant or any stock certificate and, in the case of any such loss, theft or destruction, upon receipt of an indemnity reasonably satisfactory to the Company, or in the case of any such mutilation upon surrender and cancellation of such Warrant or stock certificate, the Company will make and deliver a new Warrant or stock certificate, of like tenor, in lieu of the lost, stolen, destroyed or mutilated Warrant or stock certificate.
- 17. <u>Descriptive Headings</u>. The descriptive headings of the various Sections of this Warrant are inserted for convenience only and do not constitute a part of this Warrant. The language in this Warrant shall be construed as to its fair meaning without regard to which party drafted this Warrant.
 - 18. Governing Law, This Warrant shall be construed and enforced in accordance with, and the rights of the parties shall be governed by, the laws of the State of California.
- 19. <u>Survival of Representations</u>, <u>Warranties and Agreements</u>. All representations and warranties of the Company and the holder hereof contained herein shall survive the Date of Grant, the exercise or conversion of this Warrant (or any part hereof) or the termination or expiration of rights hereunder. All agreements of the Company and the holder hereof contained herein shall survive indefinitely until, by their respective terms, they are no longer operative.
- 20. Remedies. In case any one or more of the covenants and agreements contained in this Warrant shall have been breached, the holders hereof (in the case of a breach by the Company), or the Company (in the case of a breach by a holder), may proceed to protect and enforce their or its rights either by suit in equity and/or by action at law, including, but not limited to, an action for damages as a result of any such breach and/or an action for specific performance of any such covenant or agreement contained in this Warrant.
- 21. No Impairment of Rights. The Company will not, by amendment of its Charter or through any other means, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, but will at all times in good faith assist in the carrying out of all such terms and in the taking of all such action as may be necessary or appropriate in order to protect the rights of the holder of this Warrant against impairment
- 22. <u>Severability</u>. The invalidity or unenforceability of any provision, of this Warrant in any jurisdiction shall not affect the validity or enforceability of such provision in any other jurisdiction, or affect any other provision of this Warrant, which shall remain in full force and effect.
- 23. Recovery of Litigation Costs. If any legal action or other proceeding is brought for the enforcement of this Warrant, or because of an alleged dispute, breach, default, or misrepresentation in connection with any of the provisions of this Warrant, the successful or

prevailing party or parties shall be entitled to recover reasonable attorneys' fees and other costs incurred in that action or proceeding, in addition to any other relief to which it or they may be entitled.

24. Entire Agreement: Modification. This Warrant constitutes the entire agreement between the parries pertaining to the subject matter contained in it and supersedes all prior and contemporaneous agreements, representations, and undertakings of the parties, whether oral or written, with respect to such subject matter.

The Company has caused this Warrant to be duly executed and delivered as of the Date of Grant specified above.

COMSCORE NETWORKS, INC.

By: /s/ James A. Powers

Title: General Counsel & Corporate Secretary

Address:

11465 Sunset Hills Road Suite 200 Reston, VA 20190

EXHIBIT A-1

NOTICE OF EXERCISE

To: COMSCORE NETWORKS, INC. (the "Company")

1.	The un	dersigned hereby:
	0	elects to purchase shares of [Series Preferred Stock] [Common Stock] of the Company pursuant to the terms of the attached Warrant, and tenders herewith payment of the purchase price of such shares in full, or
	0	elects to exercise its net issuance rights pursuant to Section 10.2 of the attached Warrant with respect to Shares of [Series Preferred Stock] [Common Stock].
2. Pleas	e issue a	certificate or certificates representing shares in the name of the undersigned or in such other name or names as are specified below:
		(Name)
		
		(Address)
		ed represents that the aforesaid shares are being acquired for the account of the undersigned for investment and not with a view to, or for resale in connection with, the distribution thereof and that tipresent intention of distributing or reselling such shares, all except as in compliance with applicable securities laws.
		(Signature)
(Da	ite)	_

EXHIBIT A-2

NOTICE OF EXERCISE

To: COMSCORE NETWORKS, INC. (the "Company")

1. Contingent upon and effective immediately prior to the closing (the "Closing") of the Company's public offering contemplated by the Registration Statement on Form S ______, filed _____, 200_, the undersigned hereby:

o elects to purchase______ shares of [Series Preferred Stock] [Common Stock] of the Company (or such lesser number of shares as may be sold on behalf of the undersigned at the Closing) pursuant to the terms of the attached Warrant, or

o elects to exercise its net issuance rights pursuant to Section 10.2 of the attached Warrant with respect to ______ Shares of [Series Preferred Stock] [Common Stock].

2. Please deliver to the custodian for the selling shareholders a stock certificate representing such______ shares.

3. The undersigned has instructed the custodian for the selling shareholders to deliver to the Company \$_____ or, if less, the net proceeds due the undersigned from the sale of shares in the aforesaid public offering. If such net proceeds are less than the purchase price for such shares, the undersigned agrees to deliver the difference to the Company prior to the Closing.

(Signature)

EXHIBIT B

CHARTER

THIS WARRANT HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED OR ANY STATE SECURITIES LAWS. NO SALE OR DISPOSITION MAY BE EFFECTED WITHOUT (i) EFFECTIVE REGISTRATION STATEMENTS RELATED THERETO, (ii) AN OPINION OF COUNSEL OR OTHER EVIDENCE, REASONABLY SATISFACTORY TO THE COMPANY, THAT SUCH REGISTRATIONS ARE NOT REQUIRED, (iii) RECEIPT OF NO-ACTION LETTERS FROM THE APPROPRIATE GOVERNMENTAL AUTHORITIES, OR (iv) OTHERWISE COMPLYING WITH THE PROVISIONS OF SECTION 7 OF THIS WARRANT.

COMSCORE NETWORKS, INC.

WARRANT TO PURCHASE 108,382 SHARES OF SERIES D PREFERRED STOCK

THIS CERTIFIES THAT, for value received, GATX VENTURES, INC. and its assignees are entitled to subscribe for and purchase 108,382 shares of the fully paid and nonassessable Series D Preferred Stock (as adjusted pursuant to Section 4 hereof, the "Shares") of COMSCORE NETWORKS, INC., a Delaware corporation (the "Company"), at the price of \$0.8996 per share (such price and such other price as shall result, from time to time, from the adjustments specified in Section 4 hereof is herein referred to as the "Warrant Price"), subject to the provisions and upon the terms and conditions hereinafter set forth. As used herein, (a) the term "Series Preferred" shall mean the Company's presently authorized Series D Preferred Stock and any stock into or for which such Series D Preferred Stock may hereafter be converted or exchanged, and after the automatic conversion of the Series D Preferred Stock to Common Stock shall mean the Company's Common Stock, (b) the term "Date of Grant" shall mean July 31, 2002, and (c) the term "Other Warrants" shall mean any other warrants issued by the Company in connection with the transaction with respect to which this Warrant was issued, and any warrant issued upon transfer or partial exercise of or in lieu of this Warrant. The term "Warrant" as used herein shall be deemed to include Other Warrants unless the context clearly requires otherwise.

- 1. <u>Term</u>. The purchase right represented by this Warrant is exercisable, in whole or in part, at any time and from time to time from the Date of Grant through the later of (i) ten (10) years after the Date of Grant or (ii) five (5) years after the closing of the Company's initial public offering of its Common Stock ("IPO") effected pursuant to a Registration Statement on Form S-1 (or its successor) filed under the Securities Act of 1933, as amended (the "Act").
- 2. <u>Method of Exercise</u>; <u>Payment</u>; <u>Issuance of New Warrant</u>. Subject to Section 1 hereof, the purchase right represented by the Warrant may be exercised by the holder hereof, in whole or in part and from time to time, at the election of the holder hereof, by (a) the surrender of this Warrant (with the notice of exercise substantially in the form attached hereto as Exhibit A-1 duly completed and executed) at the principal office of the Company and by the payment to the Company, by certified or bank check, or by wire transfer to an account designated by the Company (a "Wire Transfer") of an amount equal to the then applicable Warrant Price multiplied by the number of Shares then being purchased; or (b) exercise of the "net issuance" right provided for in Section 10.2

hereof. The person or persons in whose name(s) any certificate(s) representing shares of Series Preferred shall be issuable upon exercise of this Warrant shall be deemed to have become the holder(s) of, record of, and shall be treated for all purposes as the record holder(s) of, the shares represented thereby (and such shares shall be deemed to have been issued) immediately prior to the close of business on the date or dates upon which this Warrant is exercised. In the event of any exercise of the rights represented by this Warrant, certificates for the shares of stock so purchased shall be delivered to the holder hereof as soon as possible and in any event within thirty (30) days after such exercise and, unless this Warrant has been fully exercised or expired, a new Warrant representing the portion of the Shares, if any, with respect to which this Warrant shall not then have been exercised shall also be issued to the holder hereof as soon as possible and in any event within such thirty-day period; provided, however, at such time as the Company is subject to the reporting requirements of the Securities Exchange Act of 1934, as amended, if requested by the holder of this Warrant, the Company shall use its best efforts to cause its transfer agent to deliver the certificate representing Shares issued upon exercise of this Warrant to a broker or other person (as directed by the holder exercising this Warrant) within the time period required to settle any trade made by the holder after exercise of this Warrant.

- 3. <u>Stock Fully Paid; Reservation of Shares</u>. All Shares that may be issued upon the exercise of the rights represented by this Warrant will, upon issuance pursuant to the terms and conditions herein, be fully paid and nonassessable, and free from all preemptive rights and taxes, liens and charges with respect to the issue thereof. During the period within which the rights represented by this Warrant may be exercised, the Company will at all times have authorized, and reserved for the purpose of the issue upon exercise of the purchase rights evidenced by this Warrant, a sufficient number of shares of its Series Preferred to provide for the exercise of the rights represented by this Warrant and a sufficient number of shares of its Common Stock to provide for the conversion of the Series Preferred into Common Stock.
- 4. <u>Adjustment of Warrant Price and Number of Shares</u>. The number and kind of securities purchasable upon the exercise of this Warrant and the Warrant Price shall be subject to adjustment from time to time upon the occurrence of certain events, as follows:
- (a) <u>Reclassification or Merger</u>. In case of any reclassification or change of securities of the class issuable upon exercise of this Warrant (other than a change in par value, or from par value to no par value, or from par value to par value, or as a result of a subdivision or combination), or in case of any merger of the Company with or into another corporation (other than a merger with another corporation in which the Company is the acquiring and the surviving corporation and which does not result in any reclassification or change of outstanding securities issuable upon exercise of this Warrant), or in case of any sale of all or substantially all of the assets of the Company, the Company, or such successor or purchasing corporation, as the case may be, shall duly execute and deliver to the holder of this Warrant (in form and substance satisfactory to the holder of this Warrant), or the Company shall make appropriate provision without the issuance of a new Warrant, so that the holder of this Warrant shall have the right to receive upon exercise of this Warrant, at a total purchase price not to exceed that payable upon the exercise of the unexercised portion of this Warrant, and in lieu of the shares of Series Preferred theretofore issuable upon exercise of this Warrant, the kind and amount of shares of stock, other securities, money and

property receivable upon such reclassification, change, merger or sale by a holder of the number of shares of Series Preferred then purchasable under this Warrant. Any new Warrant shall provide for adjustments that shall be as nearly equivalent as may be practicable to the adjustments provided for in this Section 4. The provisions of this Section 4(a) shall similarly apply to successive reclassifications, changes, mergers and sales.

- (b) <u>Subdivision or Combination of Shares</u>. If the Company at any time while this Warrant remains outstanding and unexpired shall subdivide or combine its outstanding shares of Series Preferred, the Warrant Price shall be proportionately decreased and the number of Shares issuable hereunder shall be proportionately increased in the case of a subdivision and the Warrant Price shall be proportionately increased and the number of Shares issuable hereunder shall be proportionately decreased in the case of a combination.
- (c) <u>Stock Dividends and Other Distributions</u>. If the Company at any time while this Warrant is outstanding and unexpired shall (i) pay a dividend with respect to Series Preferred payable in Series Preferred, then the Warrant Price shall be adjusted, from and after the date of determination of shareholders entitled to receive such dividend or distribution, to that price determined by multiplying the Warrant Price in effect immediately prior to such date of determination by a fraction (A) the numerator of which shall be the total number of shares of Series Preferred outstanding immediately prior to such dividend or distribution, and (B) the denominator of which shall be the total number of shares of Series Preferred outstanding immediately prior to such dividend or distribution with respect to Series Preferred (except any distribution specifically provided for in Sections 4(a) and 4(b)), then, in each such case, provision shall be made by the Company such that the holder of this Warrant shall receive upon exercise of this Warrant a proportionate share of any such dividend or distribution as though it were the holder of the Series Preferred (or Common Stock issuable upon conversion thereof) as of the record date fixed for the determination of the shareholders of the Company entitled to receive such dividend or distribution.
- (d) <u>Adjustment of Number of Shares</u>. Upon each adjustment in the Warrant Price, the number of Shares of Series Preferred purchasable hereunder shall be adjusted, to the nearest whole share, to the product obtained by multiplying the number of Shares purchasable immediately prior to such adjustment in the Warrant Price by a fraction, the numerator of which shall be the Warrant Price immediately prior to such adjustment and the denominator of which shall be the Warrant Price immediately thereafter.
- (e) <u>Antidilution Rights</u>. The other antidilution rights applicable to the Shares of Series Preferred purchasable hereunder are set forth in the Company's Certificate of Incorporation, as amended through the Date of Grant, a true and complete copy of which is attached hereto as Exhibit B (the "Charter"). Such antidilution rights shall not be restated, amended, modified or waived in any manner that is adverse to the holder hereof without such holder's prior written consent, unless such amendment, modification or waiver affects such holder in the same manner as it affects other holders of only the Series Preferred. The Company shall promptly provide the holder hereof with any restatement, amendment, modification or waiver of the Charter promptly after the same has been made.

- 5. Notice of Adjustments. Whenever the Warrant Price or the number of Shares purchasable hereunder shall be adjusted pursuant to Section 4 hereof, the Company shall make a certificate signed by its chief financial officer setting forth, in reasonable detail, the event requiring the adjustment, the amount of the adjustment, the method by which such adjustment was calculated, and the Warrant Price and the number of Shares purchasable hereunder after giving effect to such, adjustment, and shall cause copies of such certificate to be mailed (without regard to Section 13 hereof, by first class mail, postage prepaid) to the holder of this Warrant. In addition, whenever the conversion price or conversion ratio of the Series Preferred shall be adjusted, the Company shall make a certificate signed by its chief financial officer setting forth, in reasonable detail, the event requiring the adjustment, the amount of the adjustment the method by which such adjustment was calculated, and the conversion price or ratio of the Series Preferred after giving effect to such adjustment, and shall cause copies of such certificate to be mailed (without regard to Section 13 hereof, by first class mail, postage prepaid) to the holder of this Warrant.
- 6. <u>Fractional Shares</u>. No fractional shares of Series Preferred will be issued in connection with any exercise hereunder, but in lieu of such fractional shares the Company shall make a cash payment therefor based on the fair market value of the Series Preferred on the date of exercise as reasonably determined in good faith by the Company's Board of Directors.
 - 7. Compliance with Act: Disposition of Warrant or Shares of Series Preferred.
- (a) Compliance with Act. The holder of this Warrant, by acceptance hereof, agrees that this Warrant, and the shares of Series Preferred to be issued upon exercise hereof and any Common Stock issued upon conversion thereof are being acquired for investment and that such holder will not offer, sell or otherwise dispose of this Warrant, or any shares of Series Preferred to be issued upon exercise hereof or any Common Stock issued upon conversion thereof except under circumstances which will not result in a violation of the Act or any applicable state securities laws. Upon exercise of this Warrant, unless the Shares being acquired are registered under the Act and any applicable state securities laws or an exemption from such registration is available, the holder hereof shall confirm in writing that the shares of Series Preferred so purchased (and any shares of Common Stock issued upon conversion thereof) are being acquired for investment and not with a view toward distribution or resale in violation of the Act and shall confirm such other matters related thereto as may be reasonably requested by the Company. This Warrant and all shares of Series Preferred issued upon exercise of this Warrant and all shares of Common Stock issued upon conversion thereof (unless registered under the Act and any applicable state securities laws) shall be stamped or imprinted with a legend in substantially the following form:

"THE SECURITIES EVIDENCED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES LAWS. NO SALE OR DISPOSITION MAY BE EFFECTED WITHOUT (i) EFFECTIVE REGISTRATION STATEMENTS RELATED THERETO, (ii) AN OPINION OF COUNSEL OR OTHER EVIDENCE, REASONABLY SATISFACTORY TO THE COMPANY, THAT SUCH REGISTRATIONS ARE NOT REQUIRED, (iii) RECEIPT OF NO-ACTION LETTERS FROM THE APPROPRIATE GOVERNMENTAL AUTHORITIES, OR (iv) OTHERWISE COMPLYING

WITH THE PROVISIONS OF SECTION 7 OF THE WARRANT UNDER WHICH THESE SECURITIES WERE ISSUED, DIRECTLY OR INDIRECTLY."

Said legend shall be removed by the Company, upon the request of a holder, at such time as the restrictions on the transfer of the applicable security shall have terminated. In addition, in connection with the issuance of this Warrant, the holder specifically represents to the Company by acceptance of this Warrant as follows:

- (1) The holder is aware of the Company's business affairs and financial condition, and has acquired information about the Company sufficient to reach an informed and knowledgeable decision to acquire this Warrant. The holder is acquiring this Warrant for its own account for investment purposes only and not with a view to, or for the resale in connection with, any "distribution" thereof in violation of the Act.
- (2) The holder understands that this Warrant has not been registered under the Act in reliance upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of the holder's investment intent as expressed herein.
- (3) The holder further understands that this Warrant must be held indefinitely unless subsequently registered under the Act and qualified under any applicable state securities laws, or unless exemptions from registration and qualification are otherwise available. The holder is aware of the provisions of Rule 144, promulgated under the Act.
 - (4) The holder is an "accredited investor" as such term is defined in Rule 501 of Regulation D promulgated under the Act.
- (b) <u>Disposition of Warrant or Shares</u>. With respect to any offer, sale or other disposition of this Warrant or any shares of Series Preferred acquired pursuant to the exercise of this Warrant prior to registration of such Warrant or shares, the holder hereof agrees to give written notice to the Company prior thereto, describing briefly the manner thereof, together with a written opinion of such holder's counsel, or other evidence, if reasonably satisfactory to the Company, to the effect that such offer, sale or other disposition may be effected without registration or qualification (under the Act as then in effect or any federal or state securities law then in effect) of this Warrant or such shares of Series Preferred to be sold or otherwise disposed of require any restrictive legend as to applicable restrictions on transferability in order to ensure compliance with such law. Upon receiving such written notice and reasonably satisfactory opinion or other evidence, the Company, as promptly as practicable but no later than fifteen (15) days after receipt of the written notice, shall notify such holder that such holder may sell or otherwise dispose of this Warrant or such shares of Series Preferred or Common Stock, all in accordance with the terms of the notice delivered to the Company. If a determination has been made pursuant to this Section 7(b) that the opinion of counsel for the holder or other evidence is not reasonably satisfactory to the Company, the Company shall so notify the holder promptly with details thereof after such determination has been made. Notwithstanding the foregoing, this Warrant or such shares of Series Preferred or Common Stock may, as to such federal laws, be offered, sold or otherwise disposed of

in accordance with Rule 144 or 144A under the Act, provided that the Company shall have been furnished with such information as the Company may reasonably request to provide a reasonable assurance that the provisions of Rule 144 or 144A have been satisfied. Each certificate representing this Warrant or the shares of Series Preferred thus transferred (except a transfer pursuant to Rule 144 or 144A) shall bear a legend as to the applicable restrictions on transferability in order to ensure compliance with such laws, unless in the aforesaid opinion of counsel for the holder, such legend is not required in order to ensure compliance with such laws. The Company may issue stop transfer instructions to its transfer agent in connection with such restrictions.

- (c) <u>Applicability of Restrictions</u>. Neither any restrictions of any legend described in this Warrant nor the requirements of Section 7(b) above shall apply to any transfer of, or grant of a security interest in, this Warrant (or the Series Preferred or Common Stock obtainable upon exercise thereof) or any part hereof (i) to a partner of the holder if the holder is a partnership or to a member of the holder is a limited liability company, (ii) to a partnership of which the holder is a partner or to a limited liability company of which the holder is a member, or (iii) to any affiliate of the holder is a corporation; <u>provided</u>, <u>however</u>, in any such transfer, if applicable, the transferee shall on the Company's request agree in writing to be bound by the terms of this Warrant as if art original holder hereof.
- 8. <u>Rights as Shareholders</u>; <u>Information</u>. No holder of this Warrant, as such, shall be entitled to vote or receive dividends or be deemed the holder of Series Preferred or any other securities of the Company which may at any time be issuable upon the exercise hereof for any purpose, nor shall anything contained herein be construed to confer upon the holder of this Warrant, as such, any of the rights of a shareholder of the Company or any right to vote for the election of directors or upon any matter submitted to shareholders at any meeting thereof, or to receive notice of meetings, or to receive dividends or subscription rights or otherwise until this Warrant shall have been exercised and the Shares purchasable upon the exercise hereof shall have become deliverable, as provided herein. Notwithstanding the foregoing, the Company will transmit to the holder of this Warrant such information, documents and reports as are generally distributed to the holders of any class or series of the securities of the Company concurrently with the distribution thereof to the shareholders.
- 9. <u>Registration Rights</u>. The Company grants registration rights to the holder of this Warrant for any Common Stock of the Company obtained upon conversion of the Series Preferred, comparable to the registration rights granted to the investors in that certain Second Amended and Restated Investor Rights Agreement dated as of August 8, 2001, (the "Registration Rights Agreement"), with the following exceptions and clarifications:
- (1) The holder will have not have the right to demand registration, but can otherwise participate in any registration demanded by others other holders of at least a majority of the Registrable Securities (as defined in the Rights Agreement).
 - (2) The holder will be subject to the same provisions regarding indemnification as contained in the Registration Rights Agreement.

(3) The registration rights are freely assignable by the holder of this Warrant in connection with a permitted transfer of this Warrant or the Shares.

10. Additional Rights.

10.1 Acquisition Transactions. The Company shall provide the holder of this Warrant with at least twenty (20) days' written notice prior to closing thereof of the terms and conditions of any of the following transactions (to the extent the Company has notice thereof): (i) the sale, lease, exchange, conveyance or other disposition of all or substantially all of the Company's property or business, or (ii) its merger into or consolidation with any other corporation (other than a wholly-owned subsidiary of the Company), or any transaction (including a merger or other reorganization) or series of related transactions, in which more than 50% of the voting power of the Company is disposed of.

10.2 Right to Convert Warrant into Stock: Net Issuance.

(a) <u>Right to Convert</u>. In addition to and without limiting the rights of the holder under the terms of this Warrant, the holder shall have the right to convert this Warrant or any portion thereof (the "Conversion Right") into shares of Series Preferred as provided in this Section 10.2 at any time or from time to time during the term of this Warrant. Upon exercise of the Conversion Right with respect to a particular number of shares subject to this Warrant (the "Converted Warrant Shares"), the Company shall deliver to the holder (without payment by the holder of any exercise price or any cash or other consideration) that number of shares of fully paid and nonassessable Series Preferred as is determined according to the following formula;

$$X = B - A$$

$$Y$$

Where: X = the number of shares of Series Preferred that shall be issued to holder

Y = the fair market value of one share of Series Preferred

A = the aggregate Warrant Price of the specified number of Converted Warrant Shares immediately prior to the exercise of the Conversion Right (i.e., the number of Converted Warrant Shares multiplied by the Warrant Price)

B = the aggregate fair market value of the specified number of Converted Warrant Shares (i.e., the number of Converted Warrant Shares multiplied by the fair market value of one Converted Warrant Shares (i.e., the number of Converted Warrant Shares)

No fractional shares shall be issuable upon exercise of the Conversion Right, and, if the number of shares to be issued determined in accordance with the foregoing formula is other than a whole number, the Company shall pay to the holder an amount in cash equal to the fair market value of the resulting fractional share on the Conversion Date (as hereinafter defined). For purposes of

Section 10 of this Warrant, shares issued pursuant to the Conversion Right shall be treated as if they were issued upon the exercise of this Warrant.

- (b) Method of Exercise. The Conversion Right may be exercised by the holder by the surrender of this Warrant at the principal office of the Company together with a written statement (which may be in the form of Exhibit A-1 or Exhibit A-2 hereto) specifying that the holder thereby intends to exercise the Conversion Right and indicating the number of shares subject to this Warrant which are being surrendered (referred to in Section 10.2(a) hereof as the Converted Warrant Shares) in exercise of the Conversion Right. Such conversion shall be effective upon receipt by the Company of this Warrant together with the aforesaid written statement, or on such later date as is specified therein (the "Conversion Date"), and, at the election of the holder hereof, may be made contingent upon the closing of the sale of the Company's Common Stock to the public in a public offering pursuant to a Registration Statement under the Act (a "Public Offering"), Certificates for the shares issuable upon exercise of the Conversion Right and, if applicable, a new warrant evidencing the balance of the shares remaining subject to this Warrant, shall be issued as of the Conversion Date and shall be delivered to the holder within thirty (30) days following the Conversion Date.
- (c) <u>Determination of Fair Market Value</u>. For purposes of this Section 10.2, "fair market value" of a share of Series Preferred (or Common Stock if the Series Preferred has been automatically converted into Common Stock) as of a particular date (the "Determination Date") shall mean:
- (i) If the Conversion Right is exercised in connection with and contingent upon a Public Offering, and if the Company's Registration Statement relating to such Public Offering ("Registration Statement") has been declared effective by the Securities and Exchange Commission, then the initial "Price to Public" specified in the final prospectus with respect to such offering.
 - (ii) If the Conversion Right is not exercised in connection with and contingent upon a Public Offering, then as follows:
- (A) If traded on a securities exchange, the fair market value of the Common Stock shall be deemed to be the average of the closing prices of the Common Stock on such exchange over the five trading days immediately prior to the Determination Date, and the fair market value of the Series Preferred shall be deemed to be such fair market value of the Common Stock multiplied by the number of shares of Common Stock into which each share of Series Preferred is then convertible;
- (B) If traded on the Nasdaq Stock Market or other over-the-counter system, the fair market value of the Common Stock shall be deemed to be the average of the closing bid prices of the Common Stock over the five trading days immediately prior to the Determination Date, and the fair market value of the Series Preferred shall be deemed to be such fair market value of the Common Stock multiplied by the number of shares of Common Stock into which each share of Series Preferred is then convertible; and

(C) If there is no public market for the Common Stock, then fair market value shall be determined in good faith by the board of directors of the Company.

In making a determination under clauses (A) or (B) above, if on the Determination Date, five trading days had not passed since the IPO, then the fair market value of the Common Stock shall be the average closing prices or closing bid prices, as applicable, for the shorter period beginning on and including the date of the IPO and ending on the trading day prior to the Determination Date (or if such period includes only one trading day the closing price or closing bid price, as applicable, for such trading day). If closing prices or closing bid prices are no longer reported by a securities exchange or other trading system, the closing price or closing bid price shall be that which is reported by such securities exchange or other trading system at 4:00 p.m. New York City time on the applicable trading day.

- 10.3 Exercise Prior to Expiration. To the extent this Warrant is not previously exercised as to all of the Shares subject hereto, and if the fair market value of one share of the Series Preferred is greater than the Warrant Price then in effect, this Warrant shall be deemed automatically exercised pursuant to Section 10.2 above (even if not surrendered) immediately before its expiration. For purposes of such automatic exercise, the fair market value of one share of the Series Preferred upon such expiration shall be determined pursuant to Section 10.2(c). To the extent this Warrant or any portion thereof is deemed automatically exercised pursuant to this Section 10.3, the Company agrees to promptly notify the holder hereof of the number of Shares, if any, the holder hereof is to receive by reason of such automatic exercise.
 - 11. Representations and Warranties. The Company represents and warrants to the holder of this Warrant as follows:
- (a) This Warrant has been duly authorized and executed by the Company and is a valid and binding obligation of the Company enforceable in accordance with its terms, subject to laws of general application relating to bankruptcy, insolvency and the relief of debtors and the rules of law or principles at equity governing specific performance, injunctive relief and other equitable remedies.
- (b) The Shares have been duly authorized and reserved for issuance by the Company and, when issued in accordance with the terms hereof, will be validly issued, fully paid and nonassessable and free from preemptive rights.
- (c) The rights, preferences, privileges and restrictions granted to or imposed upon the Series Preferred and the holders thereof are as set forth in the Charter, and on the Date of Grant, each share of the Series Preferred represented by this Warrant is convertible into one share of Common Stock.
- (d) The shares of Common Stock issuable upon conversion of the Shares have been duly authorized and reserved for issuance by the Company and, when issued in accordance with the terms of the Charter will be validly issued, fully paid and nonassessable.

- (e) The execution and delivery of this Warrant are not, and the issuance of the Shares upon exercise of this Warrant in accordance with the terms hereof will not be, inconsistent with the Company's Charter or by-laws, do not and will not contravene any law, governmental rule or regulation, judgment or order applicable to the Company, and do not and will not conflict with or contravene any provision of, or constitute a default under, any indenture, mortgage, contract or other instrument of which the Company is a party or by which it is bound or require the consent or approval of, the giving of notice to, the registration or filing with or the taking of any action in respect of or by, any Federal, state or local government authority or agency or other person, except for the filing of notices pursuant to federal and state securities laws, which filings will be effected by the time required thereby.
- (f) There are no actions, suits, audits, investigations or proceedings pending or, to the knowledge of the Company, threatened against the Company in any court or before any governmental commission, board or authority which, if adversely determined, could have a material adverse effect on the ability of the Company to perform its obligations under this Warrant.
- (g) The number of shares of Common Stock of the Company outstanding on the date hereof, on a fully diluted basis (assuming the conversion of all outstanding convertible securities and the exercise of all outstanding options and warrants), does not exceed 48,000,000 shares.
- 12. <u>Modification and Waiver</u>. This Warrant and any provision hereof may be changed, waived, discharged or terminated only by an instrument in writing signed by the party against which enforcement of the same is sought.
 - 13. Market Stand-Off. The holder of this Warrant agrees to be bound by the "Market Stand-Off" provision in Section 1(1) of the Rights Agreement.
- 14. <u>Notices</u>. Any notice, request, communication or other document required or permitted to be given or delivered to the holder hereof or the Company shall be delivered, or shall be sent by certified or registered mail, postage prepaid, to each such holder at its address as shown on the books of the Company or to the Company at the address indicated therefor on the signature page of this Warrant.
- 15. <u>Binding Effect on Successors</u>. This Warrant shall be binding upon any corporation succeeding the Company by merger, consolidation or acquisition of all or substantially all of the Company's assets, and all of the obligations of the Company relating to the Series Preferred issuable upon the exercise or conversion of this Warrant shall survive the exercise, conversion and termination of this Warrant and all of the covenants and agreements of the Company shall inure to the benefit of the successors and assigns of the holder hereof.
- 16. <u>Lost Warrants or Stock Certificates</u>. The Company covenants to the holder hereof that, upon receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant or any stock certificate and, in the case of any such loss, theft or destruction, upon receipt of an indemnity reasonably satisfactory to the Company, or in the case of any such mutilation upon surrender and cancellation of such Warrantor stock certificate, the

Company will make and deliver a new Warrant or stock certificate, of like tenor, in lieu of the lost, stolen, destroyed or mutilated Warrant or stock certificate.

- 17. <u>Descriptive Headings</u>. The descriptive headings of the various Sections of this Warrant are inserted for convenience only and do not constitute a part of this Warrant. The language in this Warrant shall be construed as to its fair meaning without regard to which party drafted this Warrant.
 - 18. Governing Law. This Warrant shall be construed and enforced in accordance with, and the rights of the parties shall be governed by, the laws of the State of California.
- 19. <u>Survival of Representations, Warranties and Agreements</u>. All representations and warranties of the Company and the holder hereof contained herein shall survive the Date of Grant, the exercise or conversion of this Warrant (or any part hereof) or the termination, or expiration of rights hereunder. All agreements of the Company and the holder hereof contained herein shall survive indefinitely until, by their respective terms, they are no longer operative.
- 20. Remedies. In case any one or more of the covenants and agreements contained in this Warrant shall have been breached, the holders hereof (in the case of a breach by the Company), or the Company), or the Company (in the case of a breach by a holder), may proceed to protect and enforce their or its rights either by suit in equity and/or by action at law, including, but not limited to, an action for damages as a result of any such breach and/or an action for specific performance of any such covenant or agreement contained in this Warrant.
- 21. No Impairment of Rights. The Company will not, by amendment of its Charter or through any other means, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, but will at all times in good faith assist in the carrying out of all such terms and in the taking of all such action as may be necessary or appropriate in order to protect the rights of the holder of this Warrant against impairment.
- 22. <u>Severability</u>. The invalidity or unenforceability of any provision of this Warrant in any jurisdiction shall not affect the validity or enforceability of such provision in any other jurisdiction, or affect any other provision of this Warrant, which shall remain in full force and effect.
- 23. Recovery of Litigation Costs. If any legal action or other proceeding is brought for the enforcement of this Warrant, or because of an alleged dispute, breach, default, or misrepresentation in connection with any of the provisions of this Warrant, the successful or prevailing party or parties shall be entitled to recover reasonable attorneys' fees and other costs incurred in that action or proceeding, in addition to any other relief to which it or they may be entitled.
- 24. Entire Agreement Modification. This Warrant constitutes the entire agreement between the parties pertaining to the subject matter contained in it and supersedes all prior and contemporaneous agreements, representations, and undertakings of the parties, whether oral or written, with respect to such subject matter.

The Company has caused this Warrant to be duly executed and delivered as of the Date of Grant specified above.

COMSCORE NETWORKS, INC.

By /s/ James A. Powers

Title: /s/ General Counsel / Secretary

Address:

11465 Sunset Hills Road Suite 200 Reston, VA 20190

EXHIBIT A-I NOTICE OF EXERCISE

To: COMSCORE NETWORKS, INC. (the "Company")

- 1. The undersigned hereby:
 - o elects to purchase ___shares of [Series Preferred Stock] [Common Stock] of the Company pursuant to the terms of the attached Warrant, and tenders herewith payment of the purchase price of such shares in full, or

o elects to exercise its net issuance rights	ursuant to Section 10.2 of the attached Warrant with respect toShares of [Series Preferred Stock] [Common Stock].	
2. Please issue a certificate or certificates representing	shares in the name of the undersigned or in such other name or names as are specified below:	
	(Name)	
	(Address)	
	are being acquired for the account of the undersigned for investment and not with a view to, or for resale in connection with, the distribution thereof and that	t th
ndersigned has no present intention of distributing or re	elling such shares, all except as in compliance with applicable securities laws.	
	(Signature)	
Date)		
,		

EXHIBIT A-2 NOTICE OF EXERCISE

To: COMSCORE NETWORKS, INC. (the "Company")

1. Contingent upon and effective immediately prior to the closing (the "Closing") of the Company's public offering contemplated by the Registration Statement on Form S, filed200, the undersigned hereby:
o elects to purchaseshares of [Series Preferred Stock] [Common Stock] of the Company (or such lesser number of shares as may be sold on behalf of the undersigned at the Closing) pursuant to the terms of tattached Warrant, or
o elects to exercise its net issuance rights pursuant to Section 10.2 of the attached Warrant with respect toShares of [Series Preferred Stock] [Common Stock].
2. Please deliver to the custodian for the selling shareholders a stock certificate representing suchshares.
3. The undersigned has instructed the custodian for the selling shareholders to deliver to the Company Sor, if less, the net proceeds due the undersigned from the sale of shares in the aforesaid public offering such net proceeds are less than the purchase price for such shares, the undersigned agrees to deliver the difference to the Company prior to the Closing.
(Signature)
(Date)

WARRANT TO PURCHASE STOCK

THIS WARRANT AND THE SHARES ISSUABLE HEREUNDER HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR THE SECURITIES LAWS OF ANY STATE AND, EXCEPT AND PURSUANT TO THE PROVISIONS OF ARTICLE 5 BELOW, MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, PLEDGED OR HYPOTHECATED UNLESS AND UNTIL REGISTERED UNDER SAID ACT AND APPLICABLE STATE SECURITIES LAW OR, IN THE OPINION OF LEGAL COUNSEL IN FORM AND SUBSTANCE SATISFACTORY TO THE ISSUER OF THESE SECURITIES, SUCH OFFER, SALE OR TRANSFER, PLEDGE OR HYPOTHECATION IS EXEMPT FROM REGISTRATION.

WARRANT TO PURCHASE STOCK

Company: comScore Networks, Inc., a Delaware corporation

Number of Shares: 45,854

Class of Stock:

Warrant Price:

Issue Date:

Expiration Date:

Series D Preferred
\$.8996 per share
December 5, 2002
December 4, 2012

THIS WARRANT CERTIFIES THAT, for the agreed upon value of \$1.00 and for other good and valuable consideration, SILICON VALLEY BANK ("Holder") is entitled to purchase the number of fully paid and nonassessable shares of the class of securities (the "Shares") of the company (the "Company") at the Warrant Price, all as set forth above and as adjusted pursuant to Article 2 of this Warrant, subject to the provisions and upon the terms and conditions set forth in this Warrant.

ARTICLE 1. EXERCISE.

- 1.1 <u>Method of Exercise</u>. Holder may exercise this Warrant by delivering a duly executed Notice of Exercise in substantially the form attached as Appendix 1 to the principal office of the Company. Unless Holder is exercising the conversion right set forth in Article 1.2, Holder shall also deliver to the Company a check, wire transfer (to an account designated by the Company), or other form of payment acceptable to the Company for the aggregate Warrant Price for the Shares being purchased.
- 1.2 <u>Conversion Right</u>. In lieu of exercising this Warrant as specified in Article 1.1, Holder may from time to time convert this Warrant, in whole or in part, into a number of Shares determined by dividing (a) the aggregate fair market value of the Shares or other securities otherwise issuable upon exercise of this Warrant minus the aggregate Warrant Price of such Shares by (b) the fair market value of one Share. The fair market value of the Shares shall be determined pursuant to Article 1.3.
- 1.3 Fair Market Value. If the Company's common stock is traded in a public market and the shares are common stock, the fair market value of each Share shall be the closing price of a Share reported for the business day immediately before Holder delivers its Notice of Exercise to the Company (or in the instance where the Warrant is exercised immediately prior to the effectiveness of the Company's initial

public offering, the "price to public" per share price specified in the final prospectus relating to such offering). If the Company's common stock is traded in a public market and the Shares are preferred stock, the fair market value of a Share shall be the closing price of a share of the Company's common stock reported for the business day immediately before Holder delivers its Notice of Exercise to the Company (or, in the instance where the Warrant is exercised immediately prior to the effectiveness of the Company's initial public offering, the initial "price to public" per share price specified in the final prospectus relating to such offering), in both cases, multiplied by the number of shares of the Company's common stock into which a Share is convertible. If the Company's common stock is not traded in a public market, the Board of Directors of the Company shall determine fair market value in its reasonable good faith judgment.

- 1.4 <u>Delivery of Certificate and New Warrant</u>. Promptly after Holder exercises or converts this Warrant, but in any event within twenty (20) days of such exercise or conversion, and, if applicable, the Company receives payment of the aggregate Warrant Price, the Company shall deliver to Holder certificates for the Shares acquired and, if this Warrant has not been fully exercised or converted and has not expired, a new Warrant representing the Shares not so acquired.
- 1.5 <u>Replacement of Warrants</u>. On receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant and, in the case of loss, theft or destruction, on delivery of an indemnity agreement reasonably satisfactory in form and amount to the Company or, in the case of mutilation, or surrender and cancellation of this Warrant, the Company shall, at its expense, execute and deliver, in lieu of this Warrant, a new warrant of like tenor.
 - 1.6 Treatment of Warrant Upon Acquisition of Company.
- 1.6.1 "Acquisition". For the purpose of this Warrant, "Acquisition" means any sale, license, or other disposition of all or substantially all of the assets of the Company, or any reorganization, consolidation, or merger of the Company where the holders of the Company's securities before the transaction beneficially own less than 50% of the outstanding voting securities of the surviving entity after the transaction.

1.6.2 Treatment of Warrant at Acquisition.

(a) Upon the written request of the Company, Holder agrees that, in the event of an Acquisition in which the sole consideration is cash, either (a) Holder shall exercise its conversion or purchase right under this Warrant and such exercise will be deemed effective immediately prior to the consummation of such Acquisition or (b) if Holder elects not to exercise the Warrant, this Warrant will expire upon the consummation of such Acquisition. The Company shall provide the Holder with written notice of its request relating to the foregoing (together with such reasonable information as the Holder may request in connection with such contemplated Acquisition giving rise to such notice), which is to be delivered to Holder not less than ten (10) days prior to the closing of the proposed Acquisition. However, if the Company fails to deliver such written notwithstanding anything to the contrary in the Warrant, the Holder's right to exercise its conversion or purchase right pursuant to this Section 1.6.2(a) shall not expire until twenty (20) days after the Company complies with such notice provisions. If such closing does not take place, the Company shall promptly notify the Holder that such proposed transaction has been terminated, and the Holder may rescind any exercise of its conversion or purchase right promptly after such notice of

termination of the proposed transaction if the exercise of the Warrant has occurred after the Company notified the Holder that the Acquisition was proposed. In the event of such rescission, the Warrant will continue to be exercisable on the same terms and conditions contained herein.

- (b) Upon the written request of the Company, Holder agrees that, in the event of an Acquisition that is an "arms length" sale of all or substantially all of the Company's assets (and only its assets) to a third party that is not an Affiliate (as defined below) of the Company (a "True Asset Sale"), either (a) Holder shall exercise its conversion or purchase right under this Warrant and such exercise will be deemed effective immediately prior to the consummation of such Acquisition or (b) if Holder elects not to exercise the Warrant, this Warrant will continue until the Expiration Date if the Company continues as a going concern following the closing of any such True Asset Sale. The Company shall provide the Holder with written notice of its request relating to the foregoing (together with such reasonable information as the Holder may request in connection with such contemplated Acquisition giving rise to such notice), which is to be delivered to Holder not less than ten (10) days prior to the closing of the proposed Acquisition.
- (c) Upon the closing of any Acquisition other than those particularly described in subsections (a) and (b) above, the successor entity shall assume the obligations of this Warrant, and this Warrant shall be exercisable for the same securities, cash, and property as would be payable for the Shares issuable upon exercise of the unexercised portion of this Warrant as if such Shares were outstanding on the record date for the Acquisition and subsequent closing. The Warrant Price and/or number of Shares shall be adjusted accordingly.

As used herein "Affiliate" shall mean any person or entity that owns or controls directly or indirectly ten (10) percent or more of the stock of Company, any person or entity that controls or is controlled by or is under common control with such persons or entities, and each of such person's or entity's officers, directors, joint venturers or partners, as applicable.

ARTICLE 2. ADJUSTMENTS TO THE SHARES.

- 2.1 Stock Dividends, Splits, Etc. If the Company declares or pays a dividend on the Shares payable in common stock, or other securities, then upon exercise of this Warrant, for each Share acquired, Holder shall receive, without cost to Holder, the total number and kind of securities to which Holder would have been entitled had Holder owned the Shares of record as of the date the dividend occurred. If the Company subdivides the Shares by reclassification or otherwise into a greater number of shares or takes any other action which increase the amount of stock into which the Shares are convertible, the number of shares purchasable hereunder shall be proportionately increased and the Warrant Price shall be proportionately increased and the number of Shares shall be proportionately decreased. If the outstanding shares are combined or consolidated, by reclassification or otherwise, into a lesser number of shares, the Warrant Price shall be proportionately increased and the number of Shares shall be proportionately decreased.
- 2.2 <u>Reclassification, Exchange, Combinations or Substitution</u>. Upon any relassification, exchange, substitution, or other event that results in a change of the number and/or class of the securities issuable upon exercise or conversion of this Warrant, Holder shall be entitled to receive, upon exercise or conversion of this Warrant, the number and kind of securities and property that Holder would have received for the

Shares if this Warrant had been exercised immediately before such reclassification, exchange, substitution, or other event. Such an event shall include any automatic conversion of the outstanding or issuable securities of the Company of the same class of series as the Shares to common stock pursuant to the terms of the Company's Certificate of Incorporation upon the closing of a registered public offering of the Company's common stock. The Company or its successor shall promptly issue to Holder an amendment to this Warrant setting forth the number and kind of such new securities or other property issuable upon exercise or conversion of this Warrant as a result of such reclassification, exchange, substitution or other event that results in a change of the number and/or class of securities issuable upon exercise or conversion of this Warrant. The amendment to this Warrant shall provide for adjustments which shall be as nearly equivalent as may be practicable to the adjustments provided for in this Article 2 including, without limitation, adjustments to the Warrant Price and to the number of securities or property issuable upon exercise of the new Warrant. The provisions of this Article 2.2 shall similarly apply to successive reclassifications, exchanges, substitutions, or other events.

- 2.3 <u>Adjustments for Diluting Issuances</u>. The Warrant Price and the number of Shares issuable upon exercise of this Warrant or, if the Shares are Preferred Stock, the number of shares of common stock issuable upon conversion of the Shares, shall be subject to adjustment, from time to time in the manner set forth in the Company's Certificate of Incorporation as if the Shares were issued and outstanding on and as of the date of any such required adjustment. The provisions set forth for the Shares in the Company's Certificate of Incorporation relating to the above in effect as of the Issue Date may not be amended, modified or waived, without the prior written consent of Holder unless such amendment modification or waiver affects the rights associated with all other shares of the same series and class as the Shares granted to the Holder.
- 2.4 No Impairment. The Company shall not, by amendment of its Certificate of Incorporation or through a reorganization, transfer of assets, consolidation, merger, dissolution, issue, or sale of securities or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms to be observed or performed under this Warrant by the Company, but shall at all times in good faith assist in carrying out of all the provisions of this Article 2 and in taking all such action as may be necessary or appropriate to protect Holder's rights under this Article against impairment.
- 2.5 <u>Fractional Shares</u>. No fractional Shares shall be issuable upon exercise or conversion of the Warrant and the number of Shares to be issued shall be rounded down to the nearest whole Share. If a fractional share interest arises upon any exercise or conversion of the Warrant, the Company shall eliminate such fractional share interest by paying Holder the amount computed by multiplying the fractional interest by the fair market value of a full Share.
- 2.6 <u>Certificate as to Adjustments</u>. Upon each adjustment of the Warrant Price, the Company shall promptly notify Holder in writing, and, at the Company's expense, promptly compute such adjustment, and furnish Holder with a certificate of its Chief Financial Officer setting forth such adjustment and the facts upon which such adjustment is based. The Company shall, upon written request, furnish Holder a certificate setting forth the Warrant Price in effect upon the date thereof and the series of adjustments leading to such Warrant Price.

ARTICLE 3. REPRESENTATIONS AND COVENANTS OF THE COMPANY.

- 3.1 Representations and Warranties. The Company represents and warrants to the Holder as follows:
- (a) The initial Warrant Price referenced on the first page of this Warrant is not greater than (i) the price per share at which the Shares were last issued in an arms-length transaction in which at least \$500,000 of the Shares were sold and (ii) the fair market value of the Shares as of the date of this Warrant.
- (b) All Shares which may be issued upon the exercise of the purchase right represented by this Warrant, and all securities, if any, issuable upon conversion of the Shares, shall, upon issuance, be duly authorized, validly issued, fully paid and nonassessable, and free of any liens and encumbrances except for restrictions on transfer provided for herein or under applicable federal and state securities laws.
- (c) The Company further covenants and agrees that, during the period within which the rights represented by this Warrant may be exercised, the Company will at all times have authorized and reserved for the purpose of issue or transfer upon exercise of the subscription rights evidenced by this Warrant, a sufficient number of Shares of authorized but unissued stock, or other securities and property, when and as required to provide for the exercise of the rights represented by this Warrant.
 - (d) The capitalization table previously provided to Holder remains true and complete as of the Issue Date.
- 3.2 Notice of Certain Events. If the Company proposes at any time (a) to declare any dividend or distribution upon any of its stock, whether in cash, property, stock, or other securities and whether or not a regular cash dividend; (b) to offer for sale additional shares of any class or series of the Company's stock; (c) to effect any reclassification or recapitalization of any of its stock; (d) to merge or consolidate with or into any other corporation, or sell, lease, license, or convey all or substantially all of its assets, or to liquidate, dissolve or wind up; or (e) offer holders of registration rights the opportunity to participate in an underwritten public offering of the company's securities for cash, then, in connection with each such event, the Company shall give Holder: (1) at least 10 days prior written notice of the date on which a record will be taken for such dividend, distribution, or subscription rights (and specifying the date on which the holders of common stock will be entitled thereto) or for determining rights to vote, if any, in respect of the matters referred to in (a) and (b) above; (2) in the case of the matters referred to in (c) and (d) above at least 10 days prior written notice of the date when the same will take place (and specifying the date on which the holders of common stock will be entitled to exchange their common stock for securities or other property deliverable upon the occurrence of such event); and (3) in the case of the matter referred to in (e) above, the same notice as is given to the holders of such registration rights.
- 3.3 <u>Registration Under Securities Act of 1933, as amended.</u> The Company agrees that the Shares or, if the Shares are convertible into common stock of the Company, such common stock, shall have certain incidental, or "Piggyback," and Form S-3 registration rights pursuant to and as set forth in the Company's Third Amended and Restated Investor Rights Agreement, dated June 6, 2002 (the "Investor Rights Agreement"). The Company further agrees that the Holder shall be deemed a "Holder" and the common stock issued or issuable upon conversion of the Shares shall

be deemed "Registrable Securities" (as those terms are defined and used in the Investor Rights Agreement) for the purpose of exercising the registration rights granted to the Holder under this Section 3.3. The provisions set forth in the Investor Rights Agreement relating to the above in effect as of the Issue Date may not be amended, modified or waived without the prior written consent of Holder unless such amendment, modification or waiver affects the rights associated with the Shares in the same manner as such amendment, modification, or waiver affects the rights associated with all other shares of the same series and class as the Shares granted to the Holder.

3.4 No Shareholder Rights. Except as provided in this Warrant, the Holder will not have any rights as a shareholder of the Company until the exercise of this Warrant.

ARTICLE 4. REPRESENTATIONS, WARRANTIES OF THE HOLDER. The Holder represents and warrants to the Company as follows:

- 4.1 <u>Purchase for Own Account</u>. This Warrant and the securities to be acquired upon exercise of this Warrant by the Holder will be acquired for investment for the Holder's account, not as a nominee or agent, and not with a view to the public resale or distribution within the meaning of the Act. Holder also represents that the Holder has not been formed for the specific purpose of acquiring this Warrant or the Shares.
- 4.2 <u>Disclosure of Information</u>. The Holder has received or has had full access to all the information it considers necessary or appropriate to make an informed investment decision with respect to the acquisition of this Warrant and its underlying securities. The Holder further has had an opportunity to ask questions and receive answers from the Company regarding the terms and conditions of the offering of this Warrant and its underlying securities and to obtain additional information (to the extent the Company possessed such information or could acquire it without unreasonable effort or expense) necessary to verify any information furnished to the Holder or to which the Holder has access.
- 4.3 <u>Investment Experience</u>. The Holder understands that the purchase of this Warrant and its underlying securities involves substantial risk. The Holder has experience as an investor in securities of companies in the development stage and acknowledges that the Holder can bear the economic risk of such Holder's investment in this Warrant and its underlying securities and has such knowledge and experience in financial or business matters that the Holder is capable of evaluating the merits and risks of its investment in this Warrant and its underlying securities and/or has a preexisting personal or business relationship with the Company and certain of its officers, directors or controlling persons of a nature and duration that enables the Holder to be aware of the character, business acumen and financial circumstances of such persons.
 - 4.4 Accredited Investor Status. The Holder is an "accredited investor" within the meaning of Regulation D promulgated under the Act.
- 4.5 <u>The Act</u>. The Holder understands that this Warrant and the Shares issuable upon exercise or conversion hereof have not been registered under the Act in reliance upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of the Holder's investment intent as expressed herein. The Holder understands that this Warrant and the Shares issued upon any exercise or conversion hereof must be held indefinitely unless subsequently registered

under the 1933 Act and qualified under applicable state securities laws, or unless exemption from such registration and qualification are otherwise available.

ARTICLE 5. MISCELLANEOUS

- 5.1 Term: This Warrant is exercisable in whole or in part at any time and from time to time on or before the Expiration Date.
- 5.2 Legends. This Warrant and the Shares (and the securities issuable, directly or indirectly, upon conversion of the Shares, if any) shall be imprinted with a legend in substantially the following form:

THIS WARRANT AND THE SHARES ISSUABLE HEREUNDER HAVE NOT BEEN REGISTERED UNDER THE ACT, OR THE SECURITIES LAWS OF ANY STATE AND, EXCEPT AND PURSUANT TO THE PROVISIONS OF ARTICLE 5 BELOW, MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, PLEDGED OR HYPOTHECATED UNLESS AND UNTIL REGISTERED UNDER SAID ACT AND APPLICABLE STATE SECURITIES LAW OR, IN THE OPINION OF LEGAL COUNSEL IN FORM AND SUBSTANCE SATISFACTORY TO THE ISSUER OF THESE SECURITIES, SUCH OFFER. SALE OR TRANSFER. PLEDGE OR HYPOTHECATION IS EXEMPT FROM REGISTRATION.

- 5.3 Compliance with Securities Laws on Transfer. This Warrant and the Shares Issuable upon exercise of this Warrant (and the securities issuable, directly or indirectly, upon conversion of the Shares, if any) may not be transferred or assigned in whole or in part without compliance with applicable federal and state securities laws by the transferor and the transfere (including, without limitation, the delivery of investment representation letters and legal opinions reasonably satisfactory to the Company, as reasonably requested by the Company). The Company shall not require Holder to provide an opinion of counsel if the transfer is to Silicon Valley Bancshares (Holder's parent company) or any other affiliate of Holder. Additionally, the Company shall also not require an opinion of counsel if there is no material question as to the availability of current information as referenced in Rule 144(c), Holder represents that it has complied with Rule 144(d) and (e) in reasonable detail, the selling broker represents that it has compiled with Rule 144(f), and the Company is provided with a copy of Holder's notice of proposed sale. At the written request of the Holder, who proposes to sell Shares issuable upon the exercise of the Warrant in compliance with Rule 144, within ten (10) days after receipt of such request, a written statement confirming the Company's compliance with the filing requirements of the Securities and Exchange Commission as set forth in such Rule, as such Rule may be amended from time to time. In addition, the Company agrees to provide the Holder within ten (10) days of written request such additional documents as the Holder may require in order to exercise its rights under this Warrant and transfer the Shares issued hereunder and carry out the intent of this Warrant including, without limitation, an opinion of counsel for the benefit of the Holder or any underwriter or broker.
- 5.4 <u>Transfer Procedure</u>. Upon receipt by Holder of the executed Warrant, Holder will transfer all of this Warrant to Silicon Valley Bancshares, Holder's parent company, by execution of an Assignment substantially in the form of Appendix 2. Subject to the provisions of Article 5.3 and upon providing Company with written notice, Silicon Valley Bancshares and any subsequent Holder may transfer all or part of this

Warrant or the Shares issuable upon exercise of this Warrant (or the Shares issuable directly or indirectly, upon conversion of the Shares, if any) to any transferee, provided, however, in connection with any such transfer, Silicon Valley Bancshares or any subsequent Holder will give the Company notice of the portion of the Warrant being transferred with the name, address and taxpayer identification number of the transferee and Holder will surrender this Warrant to the Company for reissuance to the transferee(s) (and Holder if applicable). The Company may refuse to transfer this Warrant or the Shares to any person who directly competes with the Company, unless, in either case, the stock of the Company is publicly traded.

5.5 Notices. All notices and other communications from the Company to the Holder, or vice versa, shall be deemed delivered and effective when given personally or mailed by first-class registered or certified mail, postage prepaid, at such address as may have been furnished to the Company or the Holder, as the case may (or on the first business day after transmission by facsimile) be, in writing by the Company or such holder from time to time. Effective upon receipt of the fully executed Warrant and the initial transfer described in Article 5.4 above, all notices to the Holder shall be addressed as follows until the Company receives notice of a change of address in connection with a transfer or otherwise:

Silicon Valley Bancshares Attn: Treasury Department 3003 Tasman Drive, HA 200 Santa Clara, CA 95054 Telephone: 408-654-7400 Facsimile: 408-496-2405

Notice to the Company shall be addressed as follows until the Holder receives notice of a change in address:

comScore Networks, Inc. Attn: General Counsel 11465 Sunset Hills Road, Inc., Suite 200 Reston, VA 20190 Telephone: 703-438-2049 Facsimile: 703-438-2051

5.6 Waiver. This Warrant and any term hereof may be changed, waived, discharged or terminated only by an instrument in writing signed by the party against which enforcement of such change, waiver, discharge or termination is sought.

5.7 Attorney's fees. In the event of any dispute between the parties concerning the terms and provisions of this Warrant, the party prevailing in such dispute shall be entitled to collect from the other party all costs

- incurred in such dispute, including reasonable attorney's fees.
- 5.8 <u>Automatic Conversion upon Expiration</u>. In the event that, upon the Expiration Date, the fair market value of one Share (or other security issuable upon the exercise hereof) as determined in accordance with Section 1.3 above is greater than the Exercise Price in effect on such date, then this Warrant shall automatically be deemed on and as of such date to be converted pursuant to Section 1.2 above as to all

Shares (or such other securities) for which it shall not previously have been exercised or converted, and the Company shall promptly deliver a certificate representing the Shares (or such other securities) issued upon such conversion to the Holder.

5.9 Counterparts. This Warrant may be executed in counterparts, all of which together shall constitute one and the same agreement.

5.10 Governing Law. This Warrant shall be governed by and construed in accordance with the laws of the State of California, without giving effect to its principles regarding conflicts of law.

5.11 Market Stand-Off Provision. The Holder agrees to be bound by the "Market Stand-Off" provision (the "Market Stand-Off Provision") in section 1(1) of the Investors Rights Agreement. The Market Stand-Off Provision provisions set forth in the Rights Agreement may not be amended, modified or waived without the prior written consent of the Holder unless such amendment, modification or waiver affects the Holder in the same manner as they affect all other shareholders of the series of Shares granted pursuant to this Warrant.

[SIGNATURES APPEAR ON THE FOLLOWING PAGE]

"COMPANY"

COMSCORE NETWORKS, INC.

/s/ James A. Powers By:

Name: /s/ James A. Powers V.P and

(Print) & General Counsel

Title:

Chairman of the Board, President or Vice President

"HOLDER" SILICON VALLEY BANK

By: /s/ Larry Singer

Name: /s/ Larry Singer

(Print)

Title: /s/ Silicon Valley Bank

/s/Sheri L. Huston By:

/s/Sheri L. Huston Name:

(Print)

Chief Financial Officer, Secretary, Assistant Treasurer or Assistant Secretary Title:

APPENDIX 1

NOTICE OF EXERCISE

ASSIGNMENT

For value received, Silicon Valley Bank hereby sells, assigns and transfers unto

Name: Address:	Silicon Valley Bancshares 3003 Tasman Drive (HA-200) Santa Clara, CA 95054				
Tax ID:	91-1962278				
that certain Warrant to Purchase Stock issued by comSco	ore Networks. Inc. (the "Company"), on Decembe	r, 2002	(the "Warrant") together with all rights, title and interest therein.		
		SILICON VALLEY BANK			
		By:	/s/ Larry Singer		
		Name:	/s/ Larry Singer		
		Title:	/s/ Senior Vice President		
Date: December, 2002					
By its execution below, and for the benefit of the Company, Silicon Valley Bancshares makes each of the representations and warranties set forth in Article 4 of the Warrant as of the date hereof.					
		SILICON	I VALLEY BANCSHARES		
		By:			
		Name:			
		Title:			

Exhibit 4.12

THIS WARRANT HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES LAWS. NO SALE OR DISPOSITION MAY BE EFFECTED WITHOUT (i) EFFECTIVE REGISTRATION STATEMENTS RELATED THERETO, (ii) AN OPINION OF COUNSEL OR OTHER EVIDENCE, REASONABLY SATISFACTORY TO THE COMPANY, THAT SUCH REGISTRATIONS ARE NOT REQUIRED, (iii) RECEIPT OF NO-ACTION LETTERS FROM THE APPROPRIATE GOVERNMENTAL AUTHORITIES, OR (iv) OTHERWISE COMPLYING WITH THE PROVISIONS OF SECTION 7 OF THIS WARRANT.

COMSCORE NETWORKS, INC.

WARRANT TO PURCHASE 100,000 SHARES OF COMMON STOCK

THIS CERTIFIES THAT, for value received, Comstock Partners LLC is entitled to subscribe for and purchase 100,000 shares of fully paid and nonassessable Common Stock (as adjusted pursuant to Section 4 hereof, the "Shares") of COMSCORE NETWORKS, INC., a Delaware corporation (the "Company"), at the price of \$.60 per share (such price and such other price as shall result, from time to time, from the adjustments specified in Section 4 hereof is herein referred to as the "Warrant Price"), subject to the provisions and upon the terms and conditions hereinafter set forth. As used herein, the term "Date of Grant" shall mean June 24, 2003.

1. <u>Term.</u> The purchase right represented by this Warrant is exercisable, in whole or in part, at any time and from time to time from the Date of Grant until June 24, 2013; provided, however, if the underwriter of an initial public offering of the Company's stock provides the holder of this Warrant reasonable prior written notice requesting such holder to exercise its option to purchase the Shares, the holder shall within a reasonable period of time either exercise its rights under this Warrant or waive its right to exercise.

Notwithstanding the term of this Warrant fixed pursuant to the above paragraph, the right to purchase the Shares as granted herein shall expire, if not previously exercised, immediately upon the closing of a sale, conveyance, disposal or encumbrance of all or substantially all of the Company's property or business, or the Company's merger into or consolidation with any other corporation, or any other transaction or series of related transactions in which more than fifty percent (50%) of the voting power of the Company is disposed of (a "Merger"), provided that the term "Merger" shall not apply to a merger effected exclusively for the purpose of changing the domicile of the Company.

The Company shall provide the holder of this Warrant with at least twenty (20) days' written notice prior to the closing thereof of the terms and conditions of a Merger. However, if the Company fails to deliver such written notice, then notwithstanding anything to the contrary in this Warrant, the rights to purchase the Shares shall not expire until twenty (20) days after the Company complies with such notice provisions. If such closing does not take place, the Company shall promptly notify the holder of the Warrant that such proposed transaction has been terminated, and the holder of the Warrant may rescind any exercise of its purchase rights promptly after such notice of termination of

the proposed transaction if the exercise of the Warrant has occurred after the Company notified the holder that the Merger was proposed. In the event of such rescission, the Warrant will continue to be exercisable on the same terms and conditions contained herein.

- 2. Method of Exercise: Payment: Issuance of New Warrant. Subject to Section 1 hereof, the purchase right represented by this Warrant may be exercised by the holder hereof, in whole or in part and from time to time, at the election of the holder hereof, by (a) the surrender of this Warrant (with the notice of exercise substantially in the form attached hereto as Exhibit A-1 duly completed and executed) at the principal office of the Company and by the payment to the Company, by certificate or bank check, or by wire transfer to an account designated by the Company (a "Wire Transfer") of an amount equal to the then applicable Warrant Price multiplied by the number of Shares then being purchased; or (b) exercise of the "net issuance" right provided for in Section 9.1 hereof. The person or persons in whose name(s) any certificate(s) representing any of the Shares shall be issuable upon exercise of this Warrant shall be deemed to have become the holder(s) of record of, and shall be treated for all purposes as the record holder(s) of, the Shares represented thereby (and such Shares shall be deemed to have been issued) immediately prior to the close of business on the date or dates upon which this Warrant is exercised. In the event of any exercise of the rights represented by this Warrant, certificates for the Shares so purchased shall be delivered to the holder hereof as soon as possible and in any event within thirty (30) days after such exercise and, unless this Warrant has been fully exercised or expired, a new Warrant representing the portion of the Shares, if any, with respect to which this Warrant shall not then have been exercised shall also be issued to the holder hereof as soon as possible and in any event within such thirty-day period; provided, however, at such time as the Company is subject to the reporting requirements of the Securities Exchange Act of 1934, as amended, if requested by the holder of this Warrant, the Company shall use reasonable commercial efforts to cause its transfer agent to deliver th
- 3. <u>Stock Fully Paid</u>; <u>Reservation of Shares</u>. All Shares that may be issued upon the exercise of the rights represented by this Warrant will, upon issuance pursuant to the terms and conditions herein, be fully paid and nonassessable, and free from all preemptive rights and taxes, liens and charges with respect to the issue thereof. During the period within which the rights represented by this Warrant may be exercised, the Company will at all times have authorized and reserved for the purpose of issuance upon exercise of the purchase rights evidenced by this Warrant, a sufficient number of shares of its Common Stock to provide for the exercise of the rights represented by this Warrant.
- 4. <u>Adjustment of Warrant Price and Number of Shares</u>. The number and kind of securities purchasable upon the exercise of this Warrant and the Warrant Price shall be subject to adjustment from time to time upon the occurrence of certain events, as follows:
- (a) Reclassifications or Changes. In case of any reclassification or change of securities of the class issuable upon exercise of this Warrant (other than a change in par value, or from par value to no par value, or from no par value, or as a result of a subdivision or

combination), the Company shall duly execute and deliver to the holder of this Warrant a new Warrant (in form and substance satisfactory to the holder of this Warrant), or the Company shall make appropriate provision without the issuance of a new Warrant, so that the holder of this Warrant shall have the right to receive upon exercise of this Warrant, at a total purchase price not to exceed that payable upon the exercise of the unexercised portion of this Warrant, and in lieu of the Shares theretofore issuable upon exercise of this Warrant, (i) the kind and amount of shares of stock, other securities, money and property receivable upon such reclassification or change by a holder of the number of Shares then purchasable under this Warrant. Any new Warrant shall provide for adjustments that shall be as nearly equivalent as may be practicable to the adjustments provided for in this Section 4. The provisions of this Section 4(a) shall similarly apply to successive reclassifications and changes.

- (b) <u>Subdivision or Combination of Shares</u>. If the Company at any time while this Warrant remains outstanding and unexpired shall subdivide or combine its outstanding shares of Common Stock, the Warrant Price shall be proportionately decreased and the number of Shares issuable hereunder shall be proportionately increased in the case of a subdivision, and the Warrant Price shall be proportionately increased and the number of Shares issuable hereunder shall be proportionately decreased in the case of a combination.
- (c) <u>Stock Dividends and Other Distributions</u>. If the Company at any time while this Warrant is outstanding and unexpired shall (i) pay a dividend with respect to its Common Stock payable in Common Stock, then the Warrant Price shall be adjusted, from and after the date of determination of stockholders entitled to receive such dividend or distribution, to that price determined by multiplying the Warrant Price in effect immediately prior to such date of determination by a fraction (A) the numerator of which shall be the total number of shares of Common Stock outstanding immediately prior to such dividend or distribution, and (B) the denominator of which shall be the total number of shares of Common Stock outstanding immediately after such dividend or distribution; or (ii) make any other distribution with respect to Common Stock (except any distribution specifically provided for in Sections 4(a) and 4(b)), then, in each such case, provision shall be made by the Company such that the holder of this Warrant shall receive upon exercise of this Warrant a proportionate share of any such dividend or distribution as though it were the holder of the Shares as of the record date fixed for the determination of the stockholders of the Company entitled to receive such dividend or distribution.
- (d) <u>Adjustment of Number of Shares</u>. Upon each adjustment in the Warrant Price, the number of Shares of Common Stock purchasable hereunder shall be adjusted, to the nearest whole share, to the product obtained by multiplying the number of Shares purchasable immediately prior to such adjustment in the Warrant Price by a fraction, the numerator of which shall be the Warrant Price immediately prior to such adjustment and the denominator of which shall be the Warrant Price immediately thereafter.
- 5. Notice of Adjustments. Whenever the Warrant Price or the number of Shares purchasable hereunder shall be adjusted pursuant to Section 4 hereof, the Company shall issue a certificate signed by its chief financial officer setting forth, in reasonable detail, the event requiring the adjustment, the amount of the adjustment, the method by which such adjustment was calculated,

and the Warrant Price and the number of Shares purchasable hereunder after giving effect to such adjustment, and shall cause copies of such certificate to be mailed (without regard to Section 13 hereof, by first class mail, postage prepaid) to the holder of this Warrant.

6. <u>Fractional Shares</u>. No fractional shares of Common Stock will be issued in connection with any exercise hereunder, but in lieu of such fractional shares the Company shall make a cash payment therefor based on the fair market value of the Common Stock on the date of exercise as reasonably determined in good faith by the Company's Board of Directors.

7. Compliance with Act; Disposition of Warrant or Shares of Common Stock.

(a) <u>Compliance with Act</u>. The holder of this Warrant, by acceptance hereof, agrees that this Warrant, and the Shares to be issued upon exercise hereof are being acquired for investment and that such holder will not offer, sell or otherwise dispose of this Warrant, or any Shares to be issued upon exercise hereof except under circumstances which will not result in a violation of the Act or any applicable state securities laws. Upon exercise of this Warrant, unless the Shares being acquired are registered under the Securities Act of 1933, as amended (the "Act"), and any applicable state securities laws or an exemption from such registration is available, the holder hereof shall confirm in writing that the Shares so purchased are being acquired for investment and not with a view toward distribution or resale in violation of the Act and shall confirm such other matters related thereto as may be reasonably requested by the Company. This Warrant and all Shares issued upon exercise of this Warrant (unless registered under the Act and any applicable state securities laws) shall be stamped or imprinted with a legend in substantially the following form:

"THE SECURITIES EVIDENCED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES LAWS. NO SALE OR DISPOSITION MAY BE EFFECTED WITHOUT (i) EFFECTIVE REGISTRATION STATEMENTS RELATED THERETO, (ii) AN OPINION OF COUNSEL OR OTHER EVIDENCE, REASONABLY SATISFACTORY TO THE COMPANY, THAT SUCH REGISTRATIONS ARE NOT REQUIRED, (iii) RECEIPT OF NO-ACTION LETTERS FROM THE APPROPRIATE GOVERNMENTAL AUTHORITIES, OR (iv) OTHERWISE COMPLYING WITH THE PROVISIONS OF SECTION 7 OF THE WARRANT UNDER WHICH THESE SECURITIES WERE ISSUED, DIRECTLY OR INDIRECTLY."

Said legend shall be removed by the Company, upon the request of a holder, at such time as the restrictions on the transfer of the applicable security shall have terminated. In addition, in connection with the issuance of this Warrant, the holder specifically represents to the Company by acceptance of this Warrant as follows:

(1) The holder is aware of the Company's business affairs and financial condition, and has acquired information about the Company sufficient to reach an informed and knowledgeable decision to acquire this Warrant. The holder is acquiring this Warrant for its own account for investment purposes only and not with a view to, or for the resale in connection with, any "distribution" thereof in violation of the Act.

- (2) The holder understands that this Warrant has not been registered under the Act in reliance upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of the holder's investment intent as expressed herein.
- (3) The holder further understands that this Warrant must be held indefinitely unless subsequently registered under the Act and qualified under any applicable state securities laws, or unless exemptions from registration and qualification are otherwise available. The holder is aware of the provisions of Rule 144 promulgated under the Act.
 - (4) The holder is an "accredited investor" as such term is defined in Rule 501 of Regulation D promulgated under the Act.
- (b) <u>Disposition of Warrant or Shares</u>. With respect to any offer, sale or other disposition of this Warrant or any Shares acquired pursuant to the exercise of this Warrant prior to registration of such Warrant or Shares, the holder hereof agrees to give written notice to the Company prior thereto, describing briefly the manner thereof, together with a written opinion of such holder's counsel, or other evidence, if reasonably satisfactory to the Company, to the effect that such offer, sale or other disposition may be effected without registration or qualification (under the Act as then in effect or any federal or state securities law then in effect) of this Warrant or such Shares and indicating whether or not under the Act certificates for this Warrant or such Shares to be sold or otherwise disposed of require any restrictive legend as to applicable restrictions on transferability in order to ensure compliance with such law. Upon receiving such written notice and reasonably satisfactory opinion or other evidence, the Company, as promptly as practicable but no later than fifteen (15) days after receipt of the written notice, shall notify such holder may sell or otherwise dispose of this Warrant or such Shares, all in accordance with the terms of the notice delivered to the Company. If a determination has been made pursuant to this Section 7(b) that the opinion of counsel for the holder or other evidence is not reasonably satisfactory to the Company, shall so notify the holder promptly with details thereof after such determination has been made. Notwithstanding the foregoing, this Warrant or such Shares may, as to such federal laws, be offered, sold or otherwise disposed of in accordance with Rule 144 or 144A under the Act, provided that the Company shall have been furnished with such information as the Company may reasonably request to provide a reasonable assurance that the provisions of Rule 144 or 144A have been satisfied. Each certificate representing this Warrant or the Shares thus transferred (except a tra
- (c) <u>Applicability of Restrictions</u>. Neither any restrictions of any legend described in this Warrant nor the requirements of Section 7(b) above shall apply to any transfer of, or grant of a security interest in, this Warrant (or the Shares obtainable upon exercise thereof) or any part hereof (i) to a partner of the holder if the holder is a partnership or to a member of the holder if the holder is a limited liability company, (ii) to a partnership of which the holder is a partner or to a limited liability company of which the holder is a member, or (iii) to any affiliate of the holder if the holder is a corporation; <u>provided</u>, <u>however</u>, in any such transfer, if applicable, the transferee shall on the

Company's request agree in writing to be bound by the terms of this Warrant as if an original holder hereof.

8. <u>Rights as Stockholders</u>. No holder of this Warrant, as such, shall be entitled to vote or receive dividends or be deemed the holder of the Shares or any other securities of the Company which may at any time be issuable upon the exercise hereof for any purpose, nor shall anything contained herein be construed to confer upon the holder of this Warrant, as such, any of the rights of a stockholder of the Company or any right to vote for the election of directors or upon any matter submitted to stockholders at any meeting thereof, or to receive notice of meetings, or to receive dividends or subscription rights or otherwise until this Warrant shall have been exercised and the Shares purchasable upon the exercise hereof shall have become deliverable, as provided herein.

9. Additional Rights.

9.1 Right to Convert Warrant into Stock: Net Issuance.

(a) Right to Convert. In addition to and without limiting the rights of the holder under the terms of this Warrant, the holder shall have the right to convert this Warrant or any portion thereof (the "Conversion Right") into Shares as provided in this Section 9.1 at any time or from time to time during the term of this Warrant. Upon exercise of the Conversion Right with respect to a particular number of Shares subject to this Warrant (the "Converted Warrant Shares"), the Company shall deliver to the holder (without payment by the holder of any exercise price or any cash or other consideration) that number of shares of fully paid and nonassessable Common Stock as is determined according to the following formula:



Where: X = the number of shares of Common Stock that shall be issued to holder

- Y = the fair market value of one share of Common Stock
- A = the aggregate Warrant Price of the specified number of Converted Warrant Shares immediately prior to the exercise of the Conversion Right (i.e., the number of Converted Warrant Shares multiplied by the Warrant Price)
- B = the aggregate fair market value of the specified number of Converted Warrant Shares (i.e., the number of Converted Warrant Shares multiplied by the fair market value of one Converted Warrant Share)

No fractional shares shall be issuable upon exercise of the Conversion Right, and, if the number of shares to be issued determined in accordance with the foregoing formula is other than a whole number, the Company shall pay to the holder an amount in cash equal to the fair market value of the resulting fractional share on the Conversion Date (as hereinafter defined). For purposes of

Section 9 of this Warrant, shares issued pursuant to the Conversion Right shall be treated as if they were issued upon the exercise of this Warrant.

- (b) Method of Exercise. The Conversion Right may be exercised by the holder by the surrender of this Warrant at the principal office of the Company together with a written statement (which may be in the form of Exhibit A-1) specifying that the holder thereby intends to exercise the Conversion Right and indicating the number of shares subject to this Warrant which are being surrendered (referred to in Section 9.1(a) hereof as the Converted Warrant Shares) in exercise of the Conversion Right. Such conversion shall be effective upon receipt by the Company of this Warrant together with the aforesaid written statement, or on such later date as is specified therein (the "Conversion Date"), and, at the election of the holder hereof, may be made contingent upon the closing of the Sale of the Company's Common Stock to the public in a public offering pursuant to a Registration Statement under the Act (a "Public Offering"). Certificates for the shares issuable upon exercise of the Conversion Right and, if applicable, a new warrant evidencing the balance of the shares remaining subject to this Warrant, shall be issued as of the Conversion Date and shall be delivered to the holder within thirty (30) days following the Conversion Date.
 - (c) Determination of Fair Market Value. For purposes of this Section 10.1, "fair market value" of a share of Common Stock as of a particular date (the "Determination Date") shall mean:
- (i) If the Conversion Right is exercised in connection with and contingent upon a Public Offering, and if the Company's registration statement relating to such Public Offering ("Registration Statement") has been declared effective by the Securities and Exchange Commission, then the initial "Price to Public" specified in the final prospectus with respect to such offering.
 - (ii) If the Conversion Right is not exercised in connection with and contingent upon a Public Offering, then as follows:
- (A) If traded on a securities exchange, the fair market value of the Common Stock shall be deemed to be the average of the closing prices of the Common Stock on such exchange over the five trading days immediately prior to the Determination Date;
- (B) If traded on the Nasdaq National Market or other over-the-counter system, the fair market value of the Common Stock shall be deemed to be the average of the closing bid prices of the Common Stock over the five trading days immediately prior to the Determination Date; and
 - (C) If there is no public market for the Common Stock, then the fair market value of the Common Stock shall be reasonably determined by the board of directors of the Company.

In making a determination under clauses (A) or (B) above, if on the Determination Date, five trading days had not passed since the IPO, then the fair market value of the Common Stock shall be the average closing prices or closing bid prices, as applicable, for the shorter period beginning on and including the date of the IPO and ending on the trading day prior to the Determination Date (or if such period includes only one trading day the closing price or closing bid price, as applicable, for

such trading day). If closing prices or closing bid prices are no longer reported by a securities exchange or other trading system, the closing price or closing bid price shall be that which is reported by such securities exchange or other trading system at 4:00 p.m. New York City time on the applicable trading day.

- 9.2 Exercise Prior to Expiration. To the extent this Warrant is not previously exercised as to all of the Shares subject hereto, and if the fair market value of one Share is greater than the Warrant Price then in effect, this Warrant shall be deemed automatically exercised pursuant to Section 9.1 above (even if not surrendered) immediately before its expiration. For purposes of such automatic exercise, the fair market value of one Share upon such expiration shall be determined pursuant to Section 9.1 (c). To the extent this Warrant or any portion thereof is deemed automatically exercised pursuant to this Section 9.2, the Company agrees to promptly notify the holder hereof of the number of Shares, if any, the holder hereof is to receive by reason of such automatic exercise.
 - 10. Representations and Warranties. The Company represents and warrants to the holder of this Warrant as follows:
- (a) This Warrant has been duly authorized and executed by the Company and is a valid and binding obligation of the Company enforceable in accordance with its terms, subject to laws of general application relating to bankruptcy, insolvency and the relief of debtors and the rules of law or principles at equity governing specific performance, injunctive relief and other equitable remedies.
- (b) The Shares have been duly authorized and reserved for issuance by the Company and, when issued in accordance with the terms hereof, will be validly issued, fully paid and nonassessable and free from preemptive rights.
- (c) The execution and delivery of this Warrant are not, and the issuance of the Shares upon exercise of this Warrant in accordance with the terms hereof will not be, inconsistent with the Company's Sixth Amended and Restated Certificate of Incorporation (the "Charter") or bylaws, do not and will not contravene any law, governmental rule or regulation, judgment or order applicable to the Company, and do not and will not conflict with or contravene any provision of, or constitute a default under, any indenture, mortgage, contract or other instrument of which the Company is a party or by which it is bound or require the consent or approval of, the giving of notice to, the registration or filing with or the taking of any action in respect of or by, any Federal, state or local government authority or agency or other person, except for the filing of notices pursuant to federal and state securities laws, which filings will be effected by the time required thereby.
- (d) There are no actions, suits, audits, investigations or proceedings pending or, to the knowledge of the Company, threatened against the Company in any court or before any governmental commission, board or authority which, if adversely determined, could have a material adverse effect on the ability of the Company to perform its obligations under this Warrant.

- 11. Market Stand-Off Provision. The holder of this Warrant agrees to be bound by the "Market Stand-Off" provision in section 1(l) of the Third Amended and Restated Investor Rights Agreement dated June 6, 2002 between the Company and certain of its investors and founders.
- 12. <u>Modification and Waiver</u>. This Warrant and any provision hereof may be changed, waived, discharged or terminated only by an instrument in writing signed by the party against which enforcement of the same is sought.
- 13. <u>Notices</u>. Any notice, request, communication or other document required or permitted to be given or delivered to the holder hereof or the Company shall be delivered, or shall be sent by certified or registered mail, postage prepaid, to each such holder at its address as shown on the books of the Company or to the Company at the address indicated therefor on the signature page of this Warrant.
- 14. <u>Lost Warrants or Stock Certificates</u>. The Company covenants to the holder hereof that, upon receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant or any stock certificate and, in the case of any such loss, theft or destruction, upon receipt of an indemnity reasonably satisfactory to the Company, or in the case of any such mutilation upon surrender and cancellation of such Warrant or stock certificate, the Company will make and deliver a new Warrant or stock certificate, of like tenor, in lieu of the lost, stolen, destroyed or mutilated Warrant or stock certificate.
- 15. <u>Descriptive Headings</u>. The descriptive headings of the various Sections of this Warrant are inserted for convenience only and do not constitute a part of this Warrant. The language in this Warrant shall be construed as to its fair meaning without regard to which party drafted this Warrant.
 - 16. Governing Law. This Warrant shall be construed and enforced in accordance with, and the rights of the parties shall be governed by, the laws of the State of Delaware.
- 17. <u>Survival of Representations</u>, <u>Warranties and Agreements</u>. All representations and warranties of the Company and the holder hereof contained herein shall survive the Date of Grant, the exercise or conversion of this Warrant (or any part hereof) or the termination or expiration of rights hereunder. All agreements of the Company and the holder hereof contained herein shall survive indefinitely until, by their respective terms, they are no longer operative.
- 18. Remedies. In case any one or more of the covenants and agreements contained in this Warrant shall have been breached, the holders hereof (in the case of a breach by the Company), or the Company (in the case of a breach by a holder), may proceed to protect and enforce their or its rights either by suit in equity and/or by action at law, including, but not limited to, an action for damages as a result of any such breach and/or an action for specific performance of any such covenant or agreement contained in this Warrant.

- 19. <u>Severability</u>. The invalidity or unenforceability of any provision of this Warrant in any Jurisdiction shall not affect the validity or enforceability of such provision in any other Jurisdiction, or affect any other provision of this Warrant, which shall remain in full force and effect.
- 20. Recovery of Litigation Costs. If any legal action or other proceeding is brought for the enforcement of this Warrant, or because of an alleged dispute, breach, default, or misrepresentation in connection with any of the provisions of this Warrant, the successful or prevailing party or parties shall be entitled to recover reasonable attorneys' fees and other costs incurred in that action or proceeding, in addition to any other relief to which it or they may be entitled.
- 21. Entire Agreement: Modification. This Warrant constitutes the entire agreement between the parties pertaining to the subject matter contained in it and supersedes all prior and contemporaneous agreements, representations, and undertakings of the parties, whether oral or written, with respect to such subject matter.

The Company has caused this Warrant to be duly executed and delivered as of the Date of Grant specified above.

COMSCORE NETWORKS, INC.
By: /s/Sheri L. Huston
Title: CFO

Address:

- 11 -

11465 Sunset Hills Road Suite 200 Reston, VA 20190

EXHIBIT A-I NOTICE OF EXERCISE

To: COMSCORE NETWORKS, INC. (the "Company")

1. The undersigned hereby:
o elects to purchase shares of Common Stock of the Company pursuant to the terms of the attached Warrant, and tenders herewith payment of the purchase price of such shares in full, or
o elects to exercise its net issuance rights pursuant to Section 9.1 of the attached Warrant with respect to shares of Common Stock.
2. Please issue a certificate or certificates representing shares in the name of the undersigned or in such other name or names as are specified below:
(Name)
(Address)
3. The undersigned represents that the aforesaid shares are being acquired for the account of the undersigned for investment and not with a view to, or for resale in connection with, the distribution thereof and that the undersigned has no present intention of distributing or reselling such shares, all except as incompliance with applicable securities laws.
(Signature)
(Date)

THIS WARRANT HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED OR ANY STATE SECURITIES LAWS. NO SALE OR DISPOSITION MAY BE EFFECTED WITHOUT (i) EFFECTIVE REGISTRATION STATEMENTS RELATED THERETO, (ii) AN OPINION OF COUNSEL OR OTHER EVIDENCE, REASONABLY SATISFACTORY TO THE COMPANY, THAT SUCH REGISTRATIONS ARE NOT REQUIRED, (iii) RECEIPT OF NO-ACTION LETTERS FROM THE APPROPRIATE GOVERNMENTAL AUTHORITIES, OR (iv) OTHERWISE COMPLYING WITH THE PROVISIONS OF SECTION 7 OF THIS WARRANT.

COMSCORE NETWORKS, INC.

WARRANT TO PURCHASE 240,000 SHARES OF SERIES E PREFERRED STOCK

THIS CERTIFIES THAT, for value received, Heller Financial Leasing, Inc., a GE Capital company, and its assignees are entitled to subscribe for and purchase 240,000 shares of the fully paid and nonassessable Series E Preferred Stock (as adjusted pursuant to Section 4 hereof, the "Shares") of COMSCORE NETWORKS, INC., a Delaware corporation (the "Company"), at the price of \$0.50 per share (such price and such other price as shall result, from time to time, from the adjustments specified in Section 4 hereof is herein referred to as the "Warrant Price"), subject to the provisions and upon the terms and conditions hereinafter set forth. As used herein, (a) the term "Series Preferred" shall mean the Company's presently authorized Series E Preferred Stock, and any stock into or for which such Series E Preferred Stock may hereafter be converted or exchanged, and after the automatic conversion of the Series E Preferred Stock to Common Stock shall mean the Company's Common Stock, (b) the term "Date of Grant" shall mean December 19, 2003, and (c) the term "Other Warrants" shall mean any other warrants issued by the Company in connection with the transaction with respect to which this Warrant was issued, and any warrant issued upon transfer or partial exercise of or in lieu of this Warrant. The term "Warrant" as used herein shall be deemed to include Other Warrants unless the context clearly requires otherwise.

1. <u>Term.</u> The purchase right represented by this Warrant is exercisable, in whole or in part, at any time and from time to time from the Date of Grant through the later of (i) ten (10) years after the Date of Grant or (ii) five (5) years after the closing of the Company's initial public offering of its Common Stock ("IPO") effected pursuant to a Registration Statement on Form S-I (or its successor) filed under the Securities Act of 1933, as amended (the "Act"); provided, however, if the underwriter of an initial public offering of the Company's stock provides the holder of this Warrant reasonable prior written notice, requesting such holder to exercise its option to purchase Shares, the holder shall within a reasonable period of time either exercise its rights under this warrant or waive its right to exercise.

Notwithstanding the term of this Warrant fixed pursuant to the above paragraph, the right to purchase Series Preferred as granted herein shall expire, if not previously exercised, immediately upon the closing of a sale, conveyance, disposal or encumbrance of all or substantially all of the Company's property or business or the Company's merger into or consolidation with any other

corporation or any other transaction or series of related transactions in which more than fifty percent (50%) of the voting power of the Company is disposed of (a "Merger"), provided that the term "Merger" shall not apply to a merger effected exclusively for the purpose of changing the domicile of the company.

The Company shall provide the holder of this Warrant with at least twenty (20) days' written notice prior to closing thereof of the terms and conditions of a Merger. However, if the Company fails to deliver such written notice, then notwithstanding anything to the contrary in this Warrant, the rights to purchase the Company's Series Preferred shall not expire until the Company complies with such notice provisions. If such closing does not take place, the Company shall promptly notify the holder of the Warrant that such proposed transaction has been terminated, and the holder of the Warrant may rescind any exercise of its purchase rights promptly after such notice of termination of the proposed transaction if the exercise of the Warrant has occurred after the Company notified the holder that the Merger was proposed. In the event of such rescission, the Warrant will continue to be exercisable on the same terms and conditions contained herein.

- 2. Method of Exercise; Payment; Issuance of New Warrant. Subject to Section 1 hereof, the purchase right represented by this Warrant may be exercised by the holder hereof, in whole or in part and from time to time, at the election of the holder hereof, by (a) the surrender of this Warrant (with the notice of exercise substantially in the form attached hereto as Exhibit A-1 duly completed and executed) at the principal office of the Company and by the payment to the Company, by certified or bank check, or by wire transfer to an account designated by the Company (a "Wire Transfer") of an amount equal to the then applicable Warrant Price multiplied by the number of Shares then being purchased; or (b) exercise of the "net issuance" right provided for in Section 10.1 hereof. The person or persons in whose name(s) any certificate(s) representing shares of Series Preferred shall be issuable upon exercise of this Warrant shall be deemed to have become the holder(s) of record of, and shall be treated for all purposes as the record holder(s) of, the shares represented thereby (and such shares shall be deemed to have been issued) immediately prior to the close of business on the date or dates upon which this Warrant is exercised. In the event of any exercise of the rights represented by this Warrant, certificates for the shares of stock so purchased shall be delivered to the holder hereof as soon as possible and in any event within thirty (30) days after such exercise and, unless this Warrant has been fully exercised or expired, a new Warrant representing the portion of the Shares, if any, with respect to which this Warrant shall not then have been exercised shall also be issued to the holder hereof as soon as possible and in any event within such thirty-day period; provided, however, at such time as the Company is subject to the reporting requirements of the Securities Exchange Act of 1934, as amended, if requested by the holder of this Warrant, the Company shall use its best efforts to cause its transfer agent to delive
- 3. <u>Stock Fully Paid; Reservation of Shares</u>. All Shares that may be issued upon the exercise of the rights represented by this Warrant will, upon issuance pursuant to the terms and conditions herein, be fully paid and nonassessable, and free from all preemptive rights and taxes,

liens and charges with respect to the issue thereof, other than preemptive rights to which all holders of Series Preferred are subject pursuant to the Rights Agreement (as defined below). During the period within which the rights represented by this Warrant may be exercised, the Company will at all times have authorized, and reserved for the purpose of the issue upon exercise of the purchase rights evidenced by this Warrant, a sufficient number of shares of its Series Preferred to provide for the exercise of the rights represented by this Warrant and a sufficient number of shares of its Common Stock to provide for the conversion of the Series Preferred into Common Stock.

- 4. <u>Adjustment of Warrant Price and Number of Shares</u>. The number and kind of securities purchasable upon the exercise of this Warrant and the Warrant Price shall be subject to adjustment from time to time upon the occurrence of certain events, as follows:
- (a) <u>Reclassification</u>. In case of any reclassification or change of securities of the class issuable upon exercise of this Warrant (other than a change in par value, or from par value to no par value, or from no par value to par value, or as a result of a subdivision or combination), the Company, or such successor corporation, shall duly execute and deliver to the holder of this Warrant a new Warrant (in form and substance satisfactory to the holder of this Warrant), or the Company shall make appropriate provision without the issuance of a new Warrant, so that the holder of this Warrant shall have the right to receive upon exercise of this Warrant, at a total purchase price not to exceed that payable upon the exercise of the unexercised portion of this Warrant, and in lieu of the shares of Series Preferred theretofore issuable upon exercise of this Warrant, the kind and amount of shares of stock, other securities, money and property receivable upon such reclassification or change, by a holder of the number of shares of Series Preferred then purchasable under this Warrant. Any new Warrant shall provide for adjustments that shall be as nearly equivalent as may be practicable to the adjustments provided for in this Section 4 The provisions of this Section 4(a) shall similarly apply to successive rectifications and changes.
- (b) <u>Subdivision or Combination of Shares</u>. If the Company at any time while this Warrant remains outstanding and unexpired shall subdivide or combine its outstanding shares of Series Preferred, the Warrant Price shall be proportionately decreased and the number of Shares issuable hereunder shall be proportionately increased in the case of a subdivision and the Warrant Price shall be proportionately increased and the number of Shares issuable hereunder shall be proportionately decreased in the case of a combination.
- (c) <u>Stock Dividends and Other Distributions</u>. If the Company at any time while this Warrant is outstanding and unexpired shall (i) pay a dividend with respect to Series Preferred payable in Series Preferred, then the Warrant Price shall be adjusted, from and after the date of determination of shareholders entitled to receive such dividend or distribution, to that price determined by multiplying the Warrant Price in effect immediately prior to such date of determination by a fraction (A) the numerator of which shall be the total number of shares of Series Preferred outstanding immediately prior to such dividend or distribution, and (B) the denominator of which shall be the total number of shares of Series Preferred outstanding immediately after such dividend or distribution; or (ii) make any other distribution with respect to Series Preferred (except any distribution specifically provided for in Sections 4(a) and 4(b)), then, in each such case, provision shall be made by the Company such that the holder of this Warrant shall receive upon

exercise of this Warrant a proportionate share of any such dividend or distribution as though it were the holder of the Series Preferred (or Common Stock issuable upon conversion thereof) as of the record date fixed for the determination of the shareholders of the Company entitled to receive such dividend or distribution.

- (d) <u>Adjustment of Number of Shares</u>. Upon each adjustment in the Warrant Price, the number of Shares of Series Preferred purchasable hereunder shall be adjusted, to the nearest whole share, to the product obtained by multiplying the number of Shares purchasable immediately prior to such adjustment in the Warrant Price by a fraction, the numerator of which shall be the Warrant Price immediately prior to such adjustment and the denominator of which shall be the Warrant Price immediately thereafter.
- (e) <u>Antidilution Rights</u>. The other antidilution rights applicable to the Shares of Series Preferred purchasable hereunder are set forth in the Company's Certificate of Incorporation, as amended through the Date of Grant, a true and complete copy of which is attached hereto as Exhibit B (the "Charter"). Such antidilution rights shall not be restated, amended, modified or waived in any manner that is adverse to the holder hereof without such holder's prior written consent, unless such amendment, modification or waiver affects such holder in the same manner as it affects other holders of only the Series Preferred. The Company shall promptly provide the holder hereof with any restatement, amendment, modification or waiver of the Charter promptly after the same has been made.
- 5. Notice of Adjustments. Whenever the Warrant Price or the number of Shares purchasable hereunder shall be adjusted pursuant to Section 4 hereof, the Company shall make a certificate signed by its chief financial officer setting forth, in reasonable detail, the event requiring the adjustment, the amount of the adjustment, the method by which such adjustment was calculated, and the Warrant Price and the number of Shares purchasable hereunder after giving effect to such adjustment, and shall cause copies of such certificate to be mailed (without regard to Section 14 hereof, by first class mail, postage prepaid) to the holder of this Warrant. In addition, whenever the conversion price or conversion ratio of the Series Preferred shall be adjustment was calculated, and the conversion price or ratio of the Series Preferred after giving effect to such adjustment, and shall cause copies of such certificate to be mailed (without regard to Section 14 hereof, by first class mail, postage prepaid) to the holder of this Warrant.
- 6. <u>Fractional Shares</u>. No fractional shares of Series Preferred will be issued in connection with any exercise hereunder, but in lieu of such fractional shares the Company shall make a cash payment therefor based on the fair market value of the Series Preferred on the date of exercise as reasonably determined in good faith by the Company's Board of Directors.
 - 7. Compliance with Act; Disposition of Warrant or Shares of Series Preferred.
 - (a) Compliance with Act. The holder of this Warrant, by acceptance hereof, agrees that this Warrant, and the shares of Series Preferred to be issued upon exercise hereof and any

Common Stock issued upon conversion thereof are being acquired for investment and that such holder will not offer, sell or otherwise dispose of this Warrant, or any shares of Series Preferred to be issued upon exercise hereof or any Common Stock issued upon conversion thereof except under circumstances which will not result in a violation of the Act or any applicable state securities laws. Upon exercise of this Warrant, unless the Shares being acquired are registered under the Act and any applicable state securities laws or an exemption from such registration is available, the holder hereof shall confirm in writing that the shares of Series Preferred so purchased (and any shares of Common Stock issued upon conversion thereof) are being acquired for investment and not with a view toward distribution or resale in violation of the Act and shall confirm such other matters related thereto as may be reasonably requested by the Company. This Warrant and all shares of Series Preferred issued upon exercise of this Warrant and all shares of Common Stock issued upon conversion thereof (unless registered under the Act and any applicable state securities laws) shall be stamped or imprinted with a legend in substantially the following form:

"THE SECURITIES EVIDENCED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES LAWS. NO SALE OR DISPOSITION MAY BE EFFECTED WITHOUT (i) EFFECTIVE REGISTRATION STATEMENTS RELATED THERETO, (ii) AN OPINION OF COUNSEL OR OTHER EVIDENCE, REASONABLY SATISFACTORY TO THE COMPANY, THAT SUCH REGISTRATIONS ARE NOT REQUIRED, (iii) RECEIPT OF NO-ACTION LETTERS FROM THE APPROPRIATE GOVERNMENTAL AUTHORITIES, OR (iv) OTHERWISE COMPLYING WITH THE PROVISIONS OF SECTION 7 OF THE WARRANT UNDER WHICH THESE SECURITIES WERE ISSUED, DIRECTLY OR INDIRECTLY."

Said legend shall be removed by the Company, upon the request of a holder, at such time as the restrictions on the transfer of the applicable security shall have terminated. In addition, in connection with the issuance of this Warrant, the holder specifically represents to the Company by acceptance of this Warrant as follows:

- (1) The holder is aware of the Company's business affairs and financial condition, and has acquired information about the Company sufficient to reach an informed and knowledgeable decision to acquire this Warrant. The holder is acquiring this Warrant for its own account for investment purposes only and not with a view to, or for the resale in connection with, any "distribution" thereof in violation of the Act.
- (2) The holder understands that this Warrant has not been registered under the Act in reliance upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of the holder's investment intent as expressed herein.
- (3) The holder further understands that this Warrant must be held indefinitely unless subsequently registered under the Act and qualified under any applicable state securities laws, or unless exemptions from registration and qualification are otherwise available. The holder is aware of the provisions of Rule 144, promulgated under the Act.

- (4) The holder is an "accredited investor" as such term is defined in Rule 501 of Regulation D promulgated under the Act.
- (b) <u>Disposition of Warrant or Shares</u>. With respect to any offer, sale or other disposition of this Warrant or any shares of Series Preferred acquired pursuant to the exercise of this Warrant prior to registration of such Warrant or shares, the holder hereof agrees to give written notice to the Company prior thereto, describing briefly the manner thereof, together with a written opinion of such holder's counsel, or other evidence, if reasonably satisfactory to the Company, to the effect that such offer, sale or other disposition may be effected without registration or qualification (under the Act as then in effect or any federal or state securities law then in effect) of this Warrant or such shares of Series Preferred or Common Stock and indicating whether or not under the Act certificates for this Warrant or such shares of Series Preferred to be sold or otherwise disposed of require any restrictive legend as to applicable restrictions on transferability in order to ensure compliance with such law. Upon receiving such written notice and reasonably satisfactory opinion or other evidence, the Company, as promptly as practicable but no later than fifteen (15) days after receipt of the written notice, shall notify such holder that such holder may sell or otherwise dispose of this Warrant or such shares of Series Preferred or Common Stock, all in accordance with the terms of the notice delivered to the Company. If a determination has been made pursuant to this Section 7(b) that the opinion of counsel for the holder or other evidence is not reasonably satisfactory to the Company, the Company shall so notify the holder promptly with details thereof after such determination has been made. Notwithstanding the foregoing, this Warrant or such shares of Series Preferred or Common Stock may, as to such federal laws, be offered, sold or otherwise disposed of in accordance with Rule 144 or 144A nader the Act, provided that the Company shall have been furnished with such information as the Company may reasonably request to pr
- (c) <u>Applicability of Restrictions</u>. Neither any restrictions of any legend described in this Warrant nor the requirements of Section 7(b) above shall apply to any transfer of, or grant of a security interest in, this Warrant (or the Series Preferred or Common Stock obtainable upon exercise thereof) or any part hereof (i) to a partner of the holder if the holder is a partnership or to a member of the holder is a limited liability company, (ii) to a partnership of which the holder is a partner or to a limited liability company of which the holder is a member, or (iii) to any affiliate of the holder if the holder is a corporation; <u>provided</u>, <u>however</u>, in any such transfer, if applicable, the transferee shall on the Company's request agree in writing to be bound by the terms of this Warrant as if an original holder hereof.
- 8. <u>Rights as Shareholders</u>. No holder of this Warrant, as such, shall be entitled to vote or receive dividends or be deemed the holder of Series Preferred or any other securities of the Company which may at any time be issuable upon the exercise hereof for any purpose, nor shall anything contained herein be construed to confer upon the holder of this Warrant, as such, any of the

rights of a shareholder of the Company or any right to vote for the election of directors or upon any matter submitted to shareholders at any meeting thereof, or to receive notice of meetings, or to receive dividends or subscription rights or otherwise until this Warrant shall have been exercised and the Shares purchasable upon the exercise hereof shall have become deliverable, as provided herein.

- 9. <u>Registration Rights</u>. The Company grants registration rights to the holder of this Warrant for any Common Stock of the Company obtained upon conversion of the Series Preferred, comparable to the registration rights granted to the investors in that certain Fourth Amended and Restated Investor Rights Agreement dated as of August 1, 2001 (the "Rights Agreement"), with the following exceptions and clarifications:
- (1) The holder will have not have the right to demand registration, but can otherwise participate in any registration demanded by other holders of at least a majority of the Registrable Securities (as defined in the Rights Agreement).
 - (2) The holder will be subject to the same provisions regarding indemnification as contained in the Rights Agreement.
 - (3) The registration rights are freely assignable by the holder of this Warrant in connection with a permitted transfer, in accordance with Section 7 above, of this Warrant or the Shares.

10. Additional Rights.

10.1 Right to Convert Warrant into Stock: Net Issuance.

(a) Right to Convert. In addition to and without limiting the rights of the holder under the terms of this Warrant, the holder shall have the right to convert this Warrant or any portion thereof (the "Conversion Right") into shares of Series Preferred as provided in this Section 10.1 at any time or from time to time during the term of this Warrant. Upon exercise of the Conversion Right with respect to a particular number of shares subject to this Warrant (the "Converted Warrant Shares"), the Company shall deliver to the holder (without payment by the holder of any exercise price or any cash or other consideration) that number of shares of fully paid and nonassessable Series Preferred as is determined according to the following formula:

$$X = B - A$$

$$Y$$

Where: X = the number of shares of Series Preferred that shall be issued to holder

Y = the fair market value of one share of Series Preferred

A = the aggregate Warrant Price of the specified number of Converted Warrant Shares immediately prior to the exercise of the Conversion Right (i.e., the number of Converted Warrant Shares multiplied by the Warrant Price)

B = the aggregate fair market value of the specified number of Converted Warrant Shares (i.e., the number of Converted Warrant Shares multiplied by the fair market value of one Converted Warrant Share)

No fractional shares shall be issuable upon exercise of the Conversion Right, and, if the number of shares to be issued determined in accordance with the foregoing formula is other than a whole number, the Company shall pay to the holder an amount in cash equal to the fair market value of the resulting fractional share on the Conversion Date (as hereinafter defined). For purposes of Section 10 of this Warrant, shares issued pursuant to the Conversion Right shall be treated as if they were issued upon the exercise of this Warrant.

- (b) Method of Exercise. The Conversion Right may be exercised by the holder by the surrender of this Warrant at the principal office of the Company together with a written statement (which may be in the form of Exhibit A-I) specifying that the holder thereby intends to exercise the Conversion Right and indicating the number of shares subject to this Warrant which are being surrendered (referred to in Section 10.1 (a) hereof as the Converted Warrant Shares) in exercise of the Conversion Right. Such conversion shall be effective upon receipt by the Company of this Warrant together with the aforesaid written statement, or on such later date as is specified therein (the "Conversion Date"), and, at the election of the holder hereof, may be made contingent upon the closing of the sale of the Company's Common Stock to the public in a public offering pursuant to a Registration Statement under the Act (a "Public Offering"). Certificates for the shares issuable upon exercise of the Conversion Right and, if applicable, a new warrant evidencing the balance of the shares remaining subject to this Warrant, shall be issued as of the Conversion Date and shall be delivered to the holder within thirty (30) days following the Conversion Date.
- (c) <u>Determination of Fair Market Value</u>. For purposes of this Section 10.1, "fair market value" of a share of Series Preferred (or Common Stock if the Series Preferred has been automatically converted into Common Stock) as of a particular date (the "Determination Date") shall mean:
- (i) If the Conversion Right is exercised in connection with and contingent upon a Public Offering, and if the Company's Registration Statement relating to such Public Offering ("Registration Statement") has been declared effective by the Securities and Exchange Commission, then the initial "Price to Public" specified in the final prospectus with respect to such offering.
 - (ii) If the Conversion Right is not exercised in connection with and contingent upon a Public Offering, then as follows:
- (A) If traded on a securities exchange, the fair market value of the Common Stock shall be deemed to be the average of the Cosing prices of the Common Stock on such exchange over the five trading days immediately prior to the Determination Date, and the fair market value of the Series Preferred shall be deemed to be such fair market value of the Common Stock multiplied by the number of shares of Common Stock into which each share of Series Preferred is then convertible;

- (B) If traded on the Nasdaq Stock Market or other over-the-counter system, the fair market value of the Common Stock shall be deemed to be the average of the closing bid prices of the Common Stock over the five trading days immediately prior to the Determination Date, and the fair market value of the Series Preferred shall be deemed to be such fair market value of the Common Stock multiplied by the number of shares of Common Stock into which each share of Series Preferred is then convertible; and
 - (C) If there is no public market for the Common Stock, then fair market value shall be reasonably determined by the board of directors of the Company.

In making a determination under clauses (A) or (B) above, if on the Determination Date, five trading days had not passed since the IPO, then the fair market value of the Common Stock shall be the average closing prices or closing bid prices, as applicable, for the shorter period beginning on and including the date of the IPO and ending on the trading day prior to the Determination Date (or if such period includes only one trading day the closing price or closing bid price, as applicable, for such trading day). If closing prices or closing bid prices are no longer reported by a securities exchange or other trading system, the closing price or closing bid price shall be that which is reported by such securities exchange or other trading system at 4:00 p.m. New York City time on the applicable trading day.

- 10.2 Exercise Prior to Expiration. To the extent this Warrant is not previously exercised as to all of the Shares subject hereto, and if the fair market value of one share of the Series Preferred is greater than the Warrant Price then in effect, this Warrant shall be deemed automatically exercised pursuant to Section 10.1 above (even if not surrendered) immediately before its expiration. For purposes of such automatic exercise, the fair market value of one share of the Series Preferred upon such expiration shall be determined pursuant to Section 10.1(c). To the extent this Warrant or any portion thereof is deemed automatically exercised pursuant to this Section 10.2, the Company agrees to promptly notify the holder hereof of the number of Shares, if any, the holder hereof is to receive by reason of such automatic exercise.
 - 11. Representations and Warranties. The Company represents and warrants to the holder of this Warrant as follows:
- (a) This Warrant has been duly authorized and executed by the Company and is a valid and binding obligation of the Company enforceable in accordance with its terms, subject to laws of general application relating to bankruptcy, insolvency and the relief of debtors and the rules of law or principles at equity governing specific performance, injunctive relief and other equitable remedies,
- (b) The Shares have been duly authorized and reserved for issuance by the Company and, when issued in accordance with the terms hereof, will be validly issued, fully paid and nonassessable and free from preemptive rights.
 - (c) The rights, preferences, privileges and restrictions granted to or imposed upon the Series Preferred and the holders thereof are as set forth in the Charter, and on the Date of Grant,

each share of the Series Preferred represented by this Warrant is convertible into one share of Common Stock.

- (d) The shares of Common Stock issuable upon conversion of the Shares have been duly authorized and reserved for issuance by the Company and, when issued in accordance with the terms of the Charter will be validly issued, fully paid and nonassessable.
- (e) The execution and delivery of this Warrant are not, and the issuance of the Shares upon exercise of this Warrant in accordance with the terms hereof will not be, inconsistent with the Company's Charter or by-laws, do not and will not contravene any law, governmental rule or regulation, judgment or order applicable to the Company, and do not and will not conflict with or contravene any provision of, or constitute a default under, any indenture, mortgage, contract or other instrument of which the Company is a party or by which it is bound or require the consent or approval of, the giving of notice to, the registration or filing with or the taking of any action in respect of or by, any Federal, state or local government authority or agency or other person, except for the filing of notices pursuant to federal and state securities laws, which filings will be effected by the time required thereby.
- (f) There are no actions, suits, audits, investigations or proceedings pending or, to the knowledge of the Company, threatened against the Company in any court or before any governmental commission, board or authority which, if adversely determined, could have a material adverse effect on the ability of the Company to perform its obligations under this Warrant.
- (g) The number of shares of Common Stock of the Company outstanding on the date hereof, on a fully diluted basis (assuming the conversion of all outstanding convertible securities and the exercise of all outstanding options and warrants), does not exceed 117,000,000 shares.
 - 12. Market Stand-Off Provision. The holder of this Warrant agrees to be bound by the "Market Stand-Off" provision in section 1(1) of the Rights Agreement.
- 13. Modification and Waiver. This Warrant and any provision hereof may be changed, waived, discharged or terminated only by an instrument in writing signed by the party against which enforcement of the same is ought.
- 14. <u>Notices</u>. Any notice, request, communication or other document required or permitted to be given or delivered to the holder hereof or the Company shall be delivered, or shall be sent by certified or registered mail, postage prepaid, to each such holder at its address as shown on the books of the Company or to the Company at the address indicated therefor on the signature page of this Warrant.
- 15. <u>Lost Warrants or Stock Certificates</u>. The Company covenants to the holder hereof that, upon receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant or any stock certificate and, in the case of any such loss, theft or destruction, upon receipt of an indemnity reasonably satisfactory to the Company, or in the case of any such mutilation upon surrender and cancellation of such Warrant or stock certificate, the

Company will make and deliver a new Warrant or stock certificate, of like tenor, in lieu of the lost, stolen, destroyed or mutilated Warrant or stock certificate.

- 16. <u>Descriptive Headings</u>. The descriptive headings of the various Sections of this Warrant are inserted for convenience only and do not constitute a part of this Warrant. The language in this Warrant shall be construed as to its fair meaning without regard to which party drafted this Warrant.
 - 17. Governing Law. This Warrant shall be construed and enforced in accordance with, and the rights of the parties shall be governed by, the laws of the State of Delaware.
- 18. <u>Survival of Representations</u>, <u>Warranties and Agreements</u>. All representations and warranties of the Company and the holder hereof contained herein shall survive the Date of Grant, the exercise or conversion of this Warrant (or any part hereof) or the termination or expiration of rights hereunder. All agreements of the Company and the holder hereof contained herein shall survive indefinitely until, by their respective terms, they are no longer operative.
- 19. Remedies. In case any one or more of the covenants and agreements contained in this Warrant shall have been breached, the holders hereof (in the case of a breach by the Company), or the Company (in the case of a breach by a holder), may proceed to protect and enforce their or its rights either by suit in equity and/or by action at law, including, but not limited to, an action for damages as a result of any such breach and/or an action for specific performance of any such covenant or agreement contained in this Warrant.
- 20. No Impairment of Rights. The Company will not, by amendment of its Charter or through any other means, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, but will at all times in good faith assist in the carrying out of all such terms and in the taking of all such action as may be necessary or appropriate in order to protect the rights of the holder of this Warrant against impairment.
- 21. <u>Severability</u>. The invalidity or unenforceability of any provision of this Warrant in any jurisdiction shall not affect the validity or enforceability of such provision in any other jurisdiction, or affect any other provision of this Warrant, which shall remain in full force and effect.
- 22. <u>Recovery of Litigation Costs</u>. If any legal action or other proceeding is brought for the enforcement of this Warrant, or because of an alleged dispute, breach, default, or misrepresentation in connection with any of the provisions of this Warrant, the successful or prevailing party or parties shall be entitled to recover reasonable attorneys' fees and other costs incurred in that action or proceeding, in addition to any other relief to which it or they may be entitled.
- 23. Entire Agreement: Modification. This Warrant constitutes the entire agreement between the parties pertaining to the subject matter contained in it and supersedes all prior and contemporaneous agreements, representations, and undertakings of the parties, whether oral or written, with respect to such subject matter.

The Company has caused this Warrant to be duly executed and delivered as of the Date of Grant specified above.

COMSCORE NETWORKS, INC.

By /s/ Sheri L. Huston

Title /s/ CFO

Address:

11465 Sunset Hills Road Suite 200 Reston, VA 20190

- 12 -

EXHIBIT A-1 NOTICE OF EXERCISE

To: COMSCORE NETWORKS, INC. (the "Company")

. COMBCORE RET WORKS, 111C. (the	Company
1. The undersigned hereby:	

- o elects to purchase __shares of [Series E Preferred Stock] [Common Stock] of the Company pursuant to the terms of the attached Warrant, and tenders herewith payment of the purchase price of such shares in full, or

[GE Logo]

GE Commercial Finance

NEITHER THIS WARRANT NOR THE SHARES OF STOCK ISSUABLE UPON EXERCISE HEREOF HAVE BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"). NO SALE, TRANSFER OR OTHER DISPOSITION OF THIS WARRANT OR SAID SHARES MAY BE EFFECTED WITHOUT (i) AN EFFECTIVE REGISTRATION STATEMENT RELATED THERETO OR (ii) AN OPINION OF COUNSEL FOR THE HOLDER, REASONABLY SATISFACTORY TO THE COMPANY, THAT SUCH REGISTRATION IS NOT REQUIRED OR (iii) OTHERWISE COMPLYING WITH THE PROVISIONS OF SECTION 7 OF THIS WARRANT.

WARRANT

TO PURCHASE SHARES OF COMMON STOCK

THIS CERTIFIES THAT, for good and valuable consideration received from Heller Financial Leasing, Inc. ("Warrantholder"), Warrantholder is entitled to subscribe for and purchase 68,182 shares (as adjusted pursuant to provisions hereof, the "Shares") of the fully paid and non-assessable Common Stock of comScore Networks, Inc., a Delaware corporation with its principal place of business at 11465 Sunset Hills Road, Suite 200, Reston, VA 20190 (the "Company"), at an exercise price per share of \$1.10 (such price and such other price as shall result, from time to time, from adjustments specified herein, is hereafter referred to as the "Exercise Price"), subject to the provisions and upon the terms and conditions hereinafter set forth. As used herein, the term "Common Stock" or "Shares" shall mean the Company's presently authorized Comon Stock, and any stock into or for which such Common Stock may hereafter be converted or exchanged pursuant to the Seventh Amended and Restated Certificate of Incorporation of comScore Networks, Inc., as amended by the Certificate of Amendment (collectively, the "Charter") as from time to time amended as provided by law and in such Certificate. As used herein, the term "Grant Date" shall mean April 29, 2005. The Company acknowledges that the cash consideration paid by Warrantholder for this Warrant is \$10.00 for income tax purposes, that such amount has been duly received by the Company, and that this Warrant is issued in connection with that certain financial accommodation entered into by and between Company as the obligor and Warrantholder as the obligee thereunder (the "Financing Arrangement").

1. <u>Term.</u> The purchase rights represented by this Warrant are exercisable, in whole or in part, at any time and from time to time, from and after the Grant Date through the later of (i) ten (10) years after the Date of Grant or (ii) five (5) years after the closing of the Company's

initial public offering of its Common Stock ("IPO") effected pursuant to a Registration Statement on Form S-1 (or its successor) filed under the Securities Act of 1933, as amended (the "Act"); provided, however, if the underwriter of an initial public offering of the Company's stock provides the holder of this Warrant reasonable prior written notice, requesting such holder to exercise its option to purchase Shares, the holder shall within a reasonable period of time either exercise its rights under this warrant or waive its right to exercise.

Notwithstanding the term of this Warrant fixed pursuant to the above paragraph, the right to purchase Shares as granted herein shall expire, if not previously exercised, immediately upon the closing of a sale, conveyance, disposal or encumbrance of all or substantially all of the Company's property or business or the Company's merger into or consolidation with any other corporation or any other transaction or series of related transactions in which more than fifty percent (50%) of the voting power of the Company is disposed of (a "Merger"), provided that the term "Merger" shall not apply to a merger effected exclusively for the purpose of changing the domicile of the Company.

The Company shall provide the holder of this Warrant with at least twenty (20) days' written notice prior to closing thereof of the terms and conditions of a Merger. However, if the Company fails to deliver such written notice, then notwithstanding anything to the contrary in this Warrant, the rights to purchase the Shares shall not expire until the Company complies with such notice provisions. If such closing does not take place, the Company shall promptly notify the holder of the Warrant that such proposed transaction has been terminated, and the holder of the Warrant may rescind any exercise of its purchase rights promptly after such notice of termination of the proposed transaction if the exercise of the Warrant has occurred after the Company notified the holder that the Merger was proposed. In the event of such rescission, the Warrant will continue to be exercisable on the same terms and conditions contained herein.

2. Method of Exercise; Net Issue Exercise.

2.1 Method of Exercise; Payment; Issuance of New Warrant. Subject to Section 1, the purchase rights represented by this Warrant may be exercised by the Warrantholder, in whole or in part and from time to time, by the surrender of this Warrant (with the notice of exercise in the form attached hereto as Exhibit A duly completed and executed) at the principal office of the Company and by (a) the payment to the Company, by certified or bank check, or by wire transfer to an account designated by the Company of an amount equal to the then applicable Exercise Price per share multiplied by the number of Shares then being purchased. The Warrantholder shall be deemed to have become the holder(s) of record of, and shall be treated for all purposes as the record holder(s) of, the Shares represented thereby (and such Shares shall be deemed to have been issued) immediately prior to the close of business on the date or dates upon which this Warrant is exercised. In the event of any exercise of the rights represented by this Warrant, certificates for the Shares so purchased shall be promptly delivered to the holder hereof as soon as possible (and in any event within thirty (30) days of receipt of such notice) and, unless this Warrant has been fully exercised or expired, a new Warrant representing the portion of the Shares, if any, with respect to which this Warrant shall not then have been exercised shall also be issued to the holder hereof as soon as possible (and in any event within such thirty-day

day period); provided, however, at such time as the Company is subject to the reporting requirements of the Securities Exchange Act of 1934, as amended, if requested by the Warrantholder, the Company shall use its best efforts to cause its transfer agent to deliver the certificate representing Shares issued upon exercise of this Warrant to a broker or other person (as directed by the Warrantholder exercising this Warrant) within the time period required to settle any trade made by the holder after exercise of this Warrant.

2.2 Non-Cash Exercise.

(a) Subject to Section 1, in lieu of payment in cash, the rights represented by this Warrant may also be exercised by a written notice of exercise, in the form of Exhibit A attached hereto, duly completed and executed, providing for the non-cash exercise of this Warrant for the Shares equal to the Fair Market Value (as determined below) of this Warrant (or the portion thereof being exercised), specifying that this non-cash exercise election has been made, and the net number of Shares to be issued after giving effect to such non-cash exercise. In the event the Warrantholder makes such election, Company shall issue to the holder a number of Shares computed using the following formula:

$$X = B-A$$

Where:

- X = the number of shares of Common Stock that shall be issued to holder
- Y = the Fair Market Value of one share of Common Stock
- A = the aggregate Exercise Price of the specified number of Shares to be converted immediately prior to the non-cash exercise (the "Converted Warrant Shares") (i.e., the number of Converted Warrant Shares *multiplied by* the Exercise Price)
- b = the aggregate Fair Market Value of the specified number of Converted Warrant Shares (i.e., the number of Converted Warrant Shares multiplied by the Fair Market Value of one Converted Warrant Share)

For purposes of Section 2 of this Warrant, shares issued pursuant to the non-cash exercise right shall be treated as if they were issued upon the exercise of this Warrant.

- (b) For purposes of this Section 2.2, the "Fair Market Value" of one share of the Company's Common Stock as of a particular date (the "Determination Date") shall be equal to
- (i) If the non-cash exercise right is exercised in connection with and contingent upon a Public Offering, and if the Company's Registration Statement relating to such Public Offering ("Registration Statement") has been declared effective by the Securities and Exchange Commission, then the initial "Price to Public" specified in the final prospectus with

respect to such offering.

- (ii) If the non-cash exercise right is not exercised in connection with and contingent upon a Public Offering, then as follows:
- (A) If traded on a securities exchange, the fair market value of the Common Stock shall be deemed to be the average of the closing prices of the Common Stock on such exchange over the five trading days immediately prior to the Determination Date;
- (B) If traded on the Nasdaq Stock Market or other over-the-counter system, the fair market value of the Common Stock shall be deemed to be the average of the closing bid prices of the Common Stock over the five trading days immediately prior to the Determination Date; and

(C) If there is no public market for the Common Stock, then fair market value shall be reasonably determined by the board of directors of the Company. If the Warrantholder hereof does not agree with the determination of Fair Market Value as determined by the Company's board of directors, the Company and the holder hereof shall negotiate an appropriate Fair Market Value. If after ten (10) days, the Company and the Warrantholder cannot agree, then the Warrantholder may request that the Fair Market Value be determined by an investment banker or other independent third party analyst of national reputation selected by the Company and reasonably acceptable to the Warrantholder. The fees and expenses of such investment banker shall be borne by the Company unless the Fair Market Value as determined by the Company, in which event the fees and expenses of such investment banker shall be borne by the Warrantholder hereof.

In making a determination under clauses (A) or (B) above, if on the Determination Date, five trading days had not passed since the Public Offering, then the fair market value of the Common Stock shall be the average closing prices or closing bid prices, as applicable, for the shorter period beginning on and including the date of the Public Offering and ending on the trading day prior to the Determination Date (or if such period includes only one trading day the closing price or closing bid price, as applicable, for such trading day). If closing prices or closing bid prices are no longer reported by a securities exchange or other trading system, the closing price or closing bid price shall be that which is reported by such securities exchange or other trading system at 4:00 p.m. New York City time on the applicable trading day.

2.3 <u>Automatic Exercise.</u> Immediately before the expiration or termination of this Warrant, to the extent this Warrant is not previously exercised, and if the Fair Market Value of one share of the Company's Common Stock subject to this Warrant is greater than the Exercise Price, then in effect as adjusted pursuant to this Warrant, this Warrant shall be deemed automatically exercised pursuant to Section 2.2 above, even if not surrendered. For purposes of such automatic exercise, the Fair Market Value of the Company's Common Stock upon such expiration shall be determined pursuant to Section 2.2 (b) above. To the extent this Warrant or

any portion thereof is deemed automatically exercised pursuant to this Section, the Company agrees to promptly notify the Warrantholder of the number of Shares, if any, the holder hereof is to receive by reason of such automatic exercise.

- 3. <u>Stock Fully Paid: Reservation of Shares</u>. All Shares that may be issued upon the exercise of the rights represented by this Warrant will, upon issuance pursuant to the terms and conditions herein, be validly issued, fully paid and non-assessable, assuming the accuracy of the Warrantholder's representations and warranties contained in Section 7 below, issued in compliance with all applicable federal and state securities laws, and free from all taxes, liens and charges with respect to the issue thereof. During the period within which the rights represented by this Warrant may be exercised, the Company will at all times have authorized and reserved for the purpose of issuance upon exercise of the purchase rights evidenced by this Warrant, a sufficient number of shares of its Common Stock to provide for the exercise of the rights represented by this Warrant.
- 4. <u>Adjustment of Exercise Price and Number of Shares</u>. The number and kind of securities purchasable upon the exercise of this Warrant and the Exercise Price shall be subject to adjustment from time to time upon the occurrence of certain events, as follows:
- (a) <u>Reclassification</u>. Reorganization, <u>Change or Conversion</u>. In case of any reclassification, reorganization, change or conversion of securities of the class issuable upon exercise of this Warrant (other than a change in par value, or from par value to no par value, or from no par value to par value, or as a result of a subdivision or combination), then in any of these events, the Company shall execute a replacement warrant (a "New Warrant"), in form and substance reasonably satisfactory to the holder of this Warrant, or the Company shall make appropriate provision without the issuance of a new Warrant, so the holder of this Warrant shall have the right to receive upon the exercise of this Warrant and at otal purchase price not to exceed that payable upon the exercise of the unexercised portion of this Warrant and in lieu of the Shares receivable upon the exercise of this Warrant, the same kind and amount of shares of stock, other securities, money and property receivable by a holder of the number of Shares then purchasable under this Warrant upon such reclassification, reorganization, change or conversion. Any such New Warrant shall provide for adjustments that shall be as nearly equivalent as may be practicable to the adjustments provided for in this Section 4. The provisions of this Section 4(a) shall similarly apply to successive reclassifications, reorganizations, changes, or conversions.
- (b) <u>Subdivisions or Combination of Shares: Stock Dividends</u>. In the event that the Company shall at any time while this Warrant remains outstanding and unexpired subdivide the outstanding shares of Common Stock, or shall issue a stock dividend on its outstanding shares of Common Stock, the number of Shares issuable upon exercise of this Warrant immediately prior to such subdivision or immediately prior to the issuance of such stock dividend shall be proportionately increased, and the Exercise Price shall be proportionately decreased, and in the event that the Company shall at any time combine the outstanding shares of Common Stock, the number of Shares issuable upon exercise of this Warrant immediately prior

to such combination shall be proportionately decreased, and the Exercise Price shall be proportionately increased, effective at the close of business on the date of such subdivision, stock dividend or combination, as the case may be.

- (f) Notices of Record Date. In case at any time:
 - (i) the Company shall declare any dividend upon its Common Stock payable in cash or stock or make any other distribution to the holders of its Common Stock;
 - (ii) the Company shall offer for subscription pro rata to the holders of its Common Stock any additional shares of stock of any class, or other rights;
- (iii) there shall be any capital reorganization or reclassification of the capital stock of the Company, or a consolidation or merger of the Company with or into, or a sale of all or substantially all its assets to another entity or entities: or
 - (iv) there shall be a voluntary or involuntary dissolution, liquidation or winding up of the Company;

then, in any one or more of said cases, the Company shall give notice as provided in Section 1 l(e) hereunder as follows: (A) at least 20 days' prior written notice of the date on which the books of the Company shall close or a record shall be taken for such dividend, distribution or subscription rights or for determining rights to vote in respect of any such reorganization, reclassification, consolidation, merger, sale, dissolution, liquidation or winding up, and (B) in the case of any such reorganization, reclassification, consolidation, merger, sale, dissolution, liquidation or winding up, at least 20 days' prior written notice of the date when the same shall take place. Such notice in accordance with the foregoing clause (A) shall also specify the date on which the holders of Common Stock shall be entitled thereto, and such notice in accordance with the foregoing clause (B) shall also specify the date on which the holders of Common Stock shall be entitled to exchange their Common Stock for securities or other property deliverable upon such reorganization, reclassification, consolidation, merger, sale, dissolution, liquidation or winding up, as the case may be.

- 5. Notice of Adjustments. Whenever the Exercise Price shall be adjusted pursuant to the provisions hereof, the Company shall within ten (10) business days of such adjustment deliver a certificate signed on behalf of the Company by its chief financial officer to the holder of this Warrant setting forth, in reasonable detail, the event requiring the adjustment, the amount of the adjustment, the method by which such adjustment was calculated, and the Exercise Price after giving effect to such adjustment.
- 6. <u>Fractional Shares</u>. No fractional shares of Common Stock will be issued in connection with any exercise hereunder, but in lieu of such fractional shares the Company shall make a cash payment therefor upon the basis of the Exercise Price then in effect.

7. Compliance with Securities Act: Disposition of Warrant or Shares of Common Stock.

(a) Compliance with Act, The holder of this Warrant, by acceptance hereof, agrees that this Warrant, and the Common Stock to be issued upon exercise hereof, are being acquired for investment purposes only and that such holder will not offer, sell or otherwise dispose of this Warrant or any shares of Common Stock to be issued upon exercise hereof except under circumstances which will not result in a violation of the Act or any applicable federal or state securities law. This Warrant and all shares of Common Stock issued upon exercise of this Warrant (unless registered under the Act) shall be stamped or imprinted with a legend in substantially the following form:

THE SECURITIES EVIDENCED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT") OR ANY STATE SECURITIES LAWS. NO SALE OR DISPOSITION MAY BE EFFECTED WITHOUT (i) EFFECTIVE REGISTRATION STATEMENTS RELATED THERETO; (ii) AN OPINION OF COUNSEL FOR THE HOLDER, REASONABLY SATISFACTORY TO THE COMPANY, THAT SUCH REGISTRATIONS ARE NOT REQUIRED; OR (iii) OTHERWISE COMPLYING WITH THE PROVISIONS OF SECTION 7 OF THE WARRANT UNDER WHICH THESE SECURITIES WERE ISSUED. DIRECTLY, OR INDIRECTLY.

(b) <u>Disposition of Warrant and Shares</u>. With respect to any offer, sale or other transfer or disposition of this Warrant or any shares of Common Stock acquired pursuant to the exercise of this Warrant prior to registration of such Shares, the holder hereof and each subsequent holder of this Warrant agrees to give written notice to the Company prior thereto, describing briefly the manner thereof, together with a written opinion of such holder's counsel (if reasonably requested by the Company and reasonably satisfactory to the Company) to the effect that (i) such offer, sale or other transfer or disposition may be effected without registration or qualification of this Warrant or such shares of Common Stock under the Act as then in effect or any federal or state securities law then in effect, and (ii) indicating whether or not under the Act this Warrant or the certificates representing such shares of Common Stock to be sold or otherwise transferred or disposed of require any restrictive legend thereon in order to ensure compliance with such law, Notwithstanding the foregoing, this Warrant or such shares of Common Stock may, as to such federal laws, be offered, sold or otherwise disposed of in accordance with Rule 144 or 144A under the Act, provided that the Company shall have been furnished with such information as the Company may reasonably request to provide a reasonable assurance that the provisions of Rule 144 or 144A have been satisfied. This Warrant or the certificates representing the shares of Common Stock thus transferred (except a transfer pursuant to Rule 144) shall bear a legend as to the applicable restrictions on transferability in order to insure compliance with such law, unless in the aforesaid opinion of counsel for the holder (reasonably satisfactory to the Company), such legend is not required in order to insure compliance with the such law. Upon any valid transfer of this Warrant or portion thereof, Company agrees to reissue the Warrant (or Warrants in the case of a partial transfer) and/or th

transfer of this Warrant (or any portion hereof) or the certificates representing the shares of Common Stock acquired pursuant to the exercise of this Warrant by the initial holder hereof or any successor holder to (i) any affiliate of such holder, including without limitation any partnership affiliated with such holder, any partner of any such partnership, (ii) any legal entity or natural person (hereinafter "Person") in a Public Offering pursuant to an effective Registration Statement under the Act, (iii) to any other Person to the extent that the transfer to such Person is exempt from the registration requirements of such laws and such Person agrees in writing to be bound by all of the restrictions on transfer contained herein and such Person is not a direct competitor of the Company, or (iv) any Person or Persons if the holder hereof shall also transfer or assign all or transfer its in the Financing Arrangement, and such Person agrees in writing to be bound by all of the restrictions on transfer contained herein and such Person is not a direct competitor of the Company. Any transfer described above must be made in compliance with all applicable federal and state securities laws and in any such transfer, the transferee shall on the Company's request agree in writing to be bound by the terms of this Warrant as if an original holder hereof. The Company may issue stop transfer instructions to its transfer agent in connection with the foregoing restrictions.

8. Warrantholder's Representations The Warrantholder acknowledges that it has had access to all material information concerning the Company which it has requested. The Warrantholder also acknowledges that it has had the opportunity to, and has to its satisfaction, questioned the officers of the Company with respect to its investment hereunder. The Warrantholder represents that it understands that the Warrant and the Common Stock are speculative investments, that it is aware of the Company's business affairs and financial condition and that it has acquired sufficient information about the Company to reach an informed and knowledgeable decision to acquire the Warrant. The Warrantholder is purchasing the Warrant and any Common Stock issued upon exercise thereof for investment for its own account only and not with a view to, or for resale in connection with, any "distribution" thereof in violation of the Act or applicable state securities laws. The Warrantholder further represents that it understands that the Warrant and Common Stock have not been registered under the Act or applicable state securities laws by reason of specific exemptions therefrom, which exemptions depend upon, among other things, the bona fide nature of the Warrantholder's investment intent as expressed herein. The Warrantholder further understands that this Warrant must be held indefinitely unless subsequently registered under the Act and qualification are otherwise available. The holder is aware of the provisions of Rule 144, promulgated under the Act. The Warrantholder is an "accredited investor" as defined in Regulation D promulgated under the Act.

9. Company's Representations.

The Company hereby represents and warrants to the Warrantholder as follows:

(a) This Warrant has been duly authorized and executed by the Company and is a valid and binding obligation of the Company enforceable in accordance with its terms, subject to laws of general application relating to bankruptcy, insolvency and the relief of debtors

and the rules of law or principles at equity governing specific performance, injunctive relief and other equitable remedies.

- (b) The Shares have been duly authorized and reserved for issuance by the Company and, when issued in accordance with the terms hereof, will be validly issued, fully paid and nonassessable and free from preemptive rights.
 - (c) The rights and restrictions granted to or imposed upon the Common Stock and the holders thereof are as set forth in the Charter.
- (d) The shares of Common Stock issuable upon exercise of the Warrant have been duly authorized and reserved for issuance by the Company and, when issued in accordance with the terms of the Charter will be validly issued, fully paid and nonassessable.
- (e) The execution and delivery of this Warrant are not, and the issuance of the Shares upon exercise of this Warrant in accordance with the terms hereof will not be, inconsistent with the Company's Charter or by-laws, do not and will not contravene any law, governmental rule or regulation, judgment or order applicable to the Company, and do not and will not conflict with or contravene any provision of, or constitute a default under, any indenture, mortgage, contract or other instrument of which the Company is a party or by which it is bound or require the consent or approval of, the giving of notice to, the registration or filing with or the taking of any action in respect of or by, any Federal, state or local government authority or agency or other person, except for the filing of notices pursuant to federal and state securities laws, which filings will be effected by the time required thereby.
- (f) There are no actions, suits, audits, investigations or proceedings pending or, to the knowledge of the Company, threatened against the Company in any court or before any governmental commission, board or authority which, if adversely determined, could have a material adverse effect on the ability of the Company to perform its obligations under this Warrant.
- (g) The number of shares of Common Stock of the Company outstanding on the date hereof, on a fully diluted basis (assuming the conversion of all outstanding convertible securities and the exercise of all outstanding options and warrants), does not exceed 121,000,000 shares.
 - 10. Market Stand-Off Provision. The holder of this Warrant agrees to be bound by the "Market Stand-Off" provision in Section 1(1) of the Rights Agreement.
 - 11. Miscellaneous
- (a) <u>Rights as Shareholders.</u> No holder of this Warrant, as such, shall be entitled to vote or receive dividends or be deemed the holder of Common Stock or otherwise be entitled to any voting or other rights as a shareholder of the Company or any right to vote for the election of directors or upon nay matter submitted to stockholders at any meeting thereof, or to receive notice of meetings, or to receive dividend or subscription rights, until this Warrant shall have

been exercised and the Shares purchasable upon the exercise shall have become deliverable, as provided herein.

- (b) Issuance Tax. The issuance of certificates for shares of Common Stock upon exercise of this Warrant shall be made without charge to the holder hereof for any issuance tax in respect hereof, provided that the Company shall not be required to pay any tax which may be payable in respect of any transfer involved in the issuance and delivery of any certificate in a name other than that of the holder of this Warrant.
 - (c) Modification and Waiver. This Warrant and any provision hereof may be changed, waived, discharged or terminated only by an instrument in writing signed by the Company and the holder of this Warrant.
- (d) Attorneys' Fees. In the event of an action, suit or proceeding brought under or in connection herewith, the prevailing party therein shall be entitled to recover from, and the other party hereto agrees to pay, the prevailing party's costs and expenses in connection therewith, including reasonably attorneys' fees.
- (e) Notices. All notices, demands or other communications required or permitted to be given or delivered under or by reason of the provisions hereof shall be in writing and shall be deemed to have been given when (i) delivered personally to the recipient, (ii) sent via facsimile transmission, (iii) the next business day after having been sent to the recipient by reputable overnight courier service (charges prepaid) or (iv) four business days after having been mailed to the recipient by certified or registered mail, return receipt requested and postage prepaid. Such notices, demands and other communications shall be sent to the Warrantholder and to the Company at the respective addresses and transmission numbers indicated on the signature page hereof, or to such other address or to the attention of such other person as the recipient party has specified by prior written notice to the sending party.
- (f) <u>Binding Effect on Successors</u>. Subject to the terms and conditions of this Warrant, all covenants and agreements in contained herein by or on behalf of any of the parties hereto shall bind and inure to the benefit of the respective successors and assigns of the parties hereto whether so expressed or not.
- (g) Lost Warrants or Stock Certificates. The Company covenants to the holder hereof that upon receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction, or mutilation of this Warrant or any stock certificate issued upon exercise hereof or in replacement thereafter and, in the case of any such loss, theft or destruction, upon receipt of an indemnity reasonably satisfactory to the Company and without requiring any bond, or in the case of any such mutilation upon surrender and cancellation of such Warrant or stock certificate, the Company will make and deliver a new Warrant or stock certificate, of like tenor, in lieu of the lost, stolen, destroyed or mutilated Warrant or stock certificate.
 - (h) Registration Rights. The Company grants registration rights to the holder of this Warrant for any Common Stock of the Company obtained upon exercise of this Warrant

- (i) The holder will have not have the right to demand registration, but can otherwise participate in any registration demanded by other holders of at least a majority of the Registrable Securities (as defined in the Rights Agreement).
 - (ii) The holder will be subject to the same provisions regarding indemnification as contained in the Rights Agreement
 - (iii) The registration rights are freely assignable by the holder of this Warrant in connection with a permitted transfer, in accordance with Section 7 above, of this Warrant or the Shares.
- (i) <u>Descriptive Headings</u>. The descriptive headings of the several paragraphs of this Warrant are inserted for convenience only and do not constitute a part of this Warrant. The language in this Warrant shall be construed as to its fair meaning without regard to which party drafted this Warrant
- (j) Governing Law. THIS WARRANT SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, AND THE RIGHTS OF THE PARTIES SHALL BE GOVERNED BY, THE LAWS OF THE STATE OF DELAWARE.
- (k) Remedies. In case any one or more of the covenants and agreements contained in this Warrant shall have been breached, the holders hereof (in the case of a breach by the Company), or the Company (in the case of a breach by a holder), may proceed to protect and enforce their or its rights either by suit in equity and/or by action at law, including, but not limited to, an action for damages as a result of any such breach and/or an action for specific performance of any such covenant or agreement contained in this Warrant.
- (1) Entire Agreement. This Warrant constitutes the entire agreement between the parties pertaining to the subject matter contained in it and supersedes all prior and contemporaneous agreements, representations, and undertakings of the parties, whether oral or written, with respect to such subject matter.

SIGNATURE PAGE FOLLOWS

In Witness Whereof, this Warrant to purchase Common Stock has been duly executed as of the Grant Date hereinabove set forth.

Issued By: comScore Networks, Inc. By: /s/ Sheri L. Huston Name: Sheri L. Huston Title: Chief Financial Officer Address for Notices:

11465 Sunset Hills Road, Suite 200 Reston, VA 20190 Attention: Chief Financial Officer

With a copy to: comScore Networks, Inc. 11465 Sunset Hills Road, Suite 200 Reston, VA 20190 Attention: Legal Department

Fax:

Accepted By:
Heller Financial Leasing, Inc.
By: /s/ Frank A. Van De Zande
Name: Frank A. Van De Zande
Title: Vice President
Address for Notices:
500 West Monroe
Chicago, IL 60661
Attention: Portfolio Management,
GE Technology Lending

Fax: 312-441-7715

[1. The undersigned hereby elects to purchase full.]	shares of Common Stock of Company pursuant to the terms of the attached Warrant, and tenders herewith payment of the purchase price of such shares in
[1. The undersigned hereby elects to purchase	shares of Common Stock of Company pursuant to a non-cash exercise of the Warrant as provided in Section 2.2 of the Warrant.]
	hat this exercise is made in connection with the occurrence of a public offering, sale or merger of the Company, and the undersigned further elects to of said public offering, sale or merger of the Company. This exercise shall not be deemed to be effective until the consummation of such transaction.
2. Please issue a certificate or certificates representing sai	shares in the name of the undersigned or in such other name or names as are specified below:
Heller Financial Leasing, Inc. 500 West Monroe Chicago, IL 60661	
3. The undersigned represents that the aforesaid shares aroundersigned has no present intention of distributing such shares.	eing acquired for the account of the undersigned for investment and not with a view to, or for resale in connection with, the distribution thereof and that the s, all except in compliance with applicable securities laws.
	Heller Financial Leasing, Inc.
	Ву:
	(Signature) Its:
	Date:

Attention: Chief Financial Officer

STOCK RESTRICTION AND PUT RIGHT AGREEMENT

This Stock Restriction Agreement (the "Agreement") is made as of July 28, 2004 by and among comScore Networks, Inc., a Delaware corporation (the "Company") and Lawrence Denaro ("Denaro").

RECITALS

- A. Company, Denaro and certain other parties have entered into an Agreement and Plan of Merger and Reorganization (the "Merger Agreement") which provides for the acquisition of Denaro & Associates, Inc. d/b/a Q2 Brand Intelligence, Inc. ("Q2") by Company. Pursuant to the Merger Agreement and as consideration for his interest in Q2, Denaro shall receive, in part, shares of common stock of Company, and a corresponding Put Right, as defined below, to sell such shares back to the Company. The Put Right is an integral part of the consideration under the Merger Agreement, without which Denaro would not have sold his interest in Q2.
 - B. Capitalized terms used but not defined in this Agreement shall have the meanings ascribed to such terms in the Merger Agreement.
 - C. As a condition to the Merger Agreement, Denaro and Company agreed to enter into this Agreement.

NOW THEREFORE, in consideration of the mutual covenants and representations set forth herein, intending to be legally bound hereby, the parties agree as follows:

ACREEMENT

1. <u>Issuance of the Shares</u>. Pursuant to the terms and conditions of the Merger Agreement, at the Closing, the Company shall issue to Denaro an aggregate of 1,060,000 shares of common stock of the Company (the "Common Stock"). For purposes hereof, the term "Shares" shall mean the Common Stock as adjusted for stock dividends, stock distributions, combinations, consolidations or splits with respect to such shares.

2. Denaro's Put Right

a. Subject to the terms and conditions of this Section 2, on the date that is the third anniversary of the Closing Date (the "Exercise Date"), Denaro shall have the right to require the Company to repurchase some or all of the Shares at a repurchase price of \$2.50 per share (as adjusted for stock dividends, stock distributions, combinations, consolidations or splits with respect to such Shares) (the "Put Price"), with an aggregate value of \$2,650,000 (the "Put Right"), Denaro shall have ninety (90) days from and including the Exercise Date to exercise this Put Right (the "Exercise Period"). Upon expiration of the Exercise Period, this Put Right shall terms and the exercise Period "Not the Exercise Period" (the "Put Price"), Denaro shall have ninety (90) days from and including the Exercise Date to exercise this Put Right (the "Exercise Period").

b. To exercise the Put Right, during the Exercise Period, Denaro shall deliver a written notice to the Company in accordance with Section 7(i) below, of his election to exercise this Put Right that includes the number of Shares he is electing to have repurchased. Within 10 business days of receipt of such written notice and upon the receipt from Denaro of the Company stock certificate(s) duly endorsed for transfer, the Company shall pay to Denaro the aggregate Put Price in cash (by cashier's check or wire transfer). If less than all of the Shares are repurchased, the Company also shall issue to Denaro a new stock

certificate representing the remaining Shares held by Denaro after such repurchase and a corresponding portion of the Put Right shall survive with respect to the remaining shares for the remainder of the Exercise

c. Notwithstanding anything to the contrary contained in Sections 2(a) and 2(b) above, if, at any time prior to the Exercise Date, the Company shall be involved in (i) a Liquidation Event (as defined below); and (ii) the holders of the Common Stock receive consideration valued at less than \$2.50 per share for the shares of stock outstanding immediately prior to the Liquidation Event (valued in accordance with Article IV.B.2(d) (ii) of the Company's Seventh Amended and Restated Certificate of Incorporation (as such may be amended from time to time after the date hereof) (the "Restated Certificate")), then the Put Right shall accelerate and become exercisable for cash by Denaro immediately prior to the consummation of such Liquidation Event. The Company shall provide at least 20 days written notice to Denaro of any such Liquidation Event (the "Liquidation Event Notice"). Denaro shall have the option, but shall not be required, to exercise all or any portion of the Put Right for 20 days following the receipt of the Liquidation Event Notice. Within 10 business days of receipt of written notice of such exercise and upon the receipt from Denaro of the Company stock certificate(s) duly endorsed for transfer, the Company shall pay to Denaro the aggregate Put Price in cash (by cashier's check or wire transfer) with respect to such with respect to such exercise. If Denaro exercises any portion of the Put Right pursuant to this Section 2(c) during such 20 day period, then the Put Right shall remain in full force and effect under the terms and conditions of this Agreement.

For the purposes of this Section 2(c), "Liquidation Event" shall be deemed to mean (i) a liquidation, dissolution or winding up of the Company; (ii) a consolidation or merger of the Company with or into any other corporations (or entity or entities) (unless the Company's stockholders of record as constituted immediately prior to such transaction will, immediately after such transaction, hold (solely in respect of their equity interests in the Company before the transaction) at least a majority of the voting power of the surviving or successor entity to the business and assets of the corporation); (iii) a sale, conveyance or disposition of all or substantially all of the assets of the Company (other than a pledge of assets or grant of security interest therein to a commercial lender or similar entity in connection with commercial lending or similar transactions) (unless the Company's stockholders of record as constituted immediately prior to such transaction will, immediately after such transaction, hold (solely in respect of their equity interests in the Company before the transaction) at least a majority of the voting power of the surviving entity or successor to the business and assets of the Company); (iv) any sale, transfer or issuance or series of sales, transfers or issuances of shares of the Company's capital stock by the Company or the existing holders thereof to new holders, as a result of which the holders of the Company's outstanding capital stock possessing the voting power to elect a majority of the Company's Board immediately prior to such sale, transfer or issuance cease to own the requisite amount of the Company's outstanding capital stock to possess the voting power to elect a majority of the Company's Board immediately prior to such sale, transfer or issuance cease to own the requisite amount of the Company's outstanding capital stock to possess the voting power to elect a majority of the Company's board immediately prior to such sale, transfer or issuance cease to own the requisite amount of

d. Notwithstanding anything to the contrary contained in this Section 2, the Company's obligation to repurchase any Shares under this Section 2 is subject to Delaware General Corporation Law and any other applicable local, state or federal law, statute or rule.

- e. Without prejudice to the remedies available to Denaro under this Agreement or otherwise, if the Company fails to make payment when due hereunder, the Company shall pay in respect of such due and unpaid amount, a late payment calculated from the date of default until all amounts due hereunder are paid at an annual rate of twelve percent (12%) or the maximum rate permitted by applicable usury law, whichever is lower.
- 3. Company's Right of First Refusal. Before any Shares held by Denaro or any transfere (either being sometimes referred to herein as the "Holder") may be sold or otherwise transferred (including transfer by gift or operation of law), the Company or its assignee(s) shall have a right of first refusal to purchase the Shares on the terms and conditions set forth in this Section 3 (the "Right of First Refusal").
- a. Notice of Proposed Transfer. The Holder of the Shares shall deliver to the Company a written notice (the "Notice") in accordance with Section 7(i) below stating: (i) the Holder's bona fide intention to sell or otherwise transfer such Shares; (ii) the name of each proposed purchaser or other transferee ("Proposed Transferee"); (iii) the number of Shares to be transferred to each Proposed Transferee; and (iv) the bona fide cash price or other consideration for which the Holder proposes to transfer the Shares (the "Offered Price"), and the Holder shall offer the Shares at the Offered Price to the Company or its assignee(s).
- b. Exercise of Right of First Refusal. At any time within thirty (30) days after receipt of the Notice, the Company and/or its assignee(s) may, by giving written notice to the Holder, elect to purchase some or all of the Shares proposed to be transferred to any one or more of the Proposed Transferees, at the purchase price determined in accordance with subsection (c) below.
- c. <u>Purchase Price</u>. The purchase price ("Purchase Price") for the Shares purchased by the Company or its assignee(s) under this Section shall be the Offered Price. If the Offered Price includes consideration other than cash, the officers of the Company and Denaro (or his heirs or representatives in the case of any transfer by operation of law) in good faith shall agree on the cash equivalent value of the non-cash consideration.
 - d. Payment. Payment of the Purchase Price shall be made in cash (by cashier's check or wire transfer) within thirty (30) days after receipt of the Notice.
- e. Holder's Right to Transfer. To the extent all of the Shares proposed in the Notice to be transferred to a given Proposed Transferee are not purchased by the Company and/or its assignee(s) as provided in this Section 3, then the Holder may sell or otherwise transfer such unpurchased Shares to that Proposed Transferee at the Offered Price or at a higher price, provided that such sale or other transfer is consummated within 120 days after the date of the Notice, that any such sale or other transfer is effected in accordance with any applicable securities laws and that the Proposed Transferee grees in writing that the provisions of this Agreement shall continue to apply to the Shares in the hands of such Proposed Transferee. If the Shares described in the Notice are not transferred to the Proposed Transferee within such period, a new Notice shall be given to the Company, and the Company and/or its assignees shall again be offered the Right of First Refusal before any Shares held by the Holder may be sold or otherwise transferred.
- f. Exception for Certain Transfers. Notwithstanding anything to the contrary contained in this Section 3, the transfer of any or all of the Shares during Denaro's lifetime or on Denaro's death by will or intestacy to Denaro's immediate family or a trust for the benefit of Denaro's immediate family shall be exempt from the provisions of this Section 3; provided, that any such transferee agrees in writing that the provisions of this Agreement shall continue to apply to the Shares in the hands of such transferee. "Immediate family" as used herein shall mean spouse, lineal descendant or antecedent, father, mother, brother

or sister or a trust formed for the sole benefit of Denaro and/or any of the above or a limited liability company solely owned by Denaro and/or any of the above.

- g. Attachment of Put Right. The Put Right is attached to the Shares and accrues to the benefit of any and all holders of such Shares.
- h. <u>Termination of Right of First Refusal</u>. The Right of First Refusal shall terminate as to any Shares upon the earlier of (i) the first sale of Common Stock of the Company to the general public pursuant to a registration statement filed with and declared effective by the Securities and Exchange Commission under the Securities Act of 1933, as amended (the "Securities Act"), and (ii) a change of control of the Company in which the successor corporation has equity securities that are publicly traded on a national securities exchange or on the Nasdaq Stock Market.
- 4. <u>Lock-Up Period</u>. Denaro hereby agrees that, if so requested by the representative of the managing underwriter (the "Managing Underwriter") in connection with any registration of the offering of any securities of the Company under the Securities Act, Denaro shall not sell or otherwise transfer any Shares or other securities of the Company during the 180-day period (the "Market Standoff Period") following the effective date of a registration statement of the Company filed under the Securities Act. Such restriction shall apply only to the first registration statement of the Company to become effective under the Securities Act that includes securities to be sold on behalf of the Company to the public in an underwritten public offering under the Securities Act. The Company may impose stop-transfer instructions with respect to securities subject to the foregoing restrictions until the end of such Market Standoff Period. Such restrictions shall not apply unless:
 - a. all officers and directors of the Company enter into similar agreements or are bound by similar agreements with the Company; and
 - b. the Company uses all reasonable efforts to obtain from persons who hold in excess of one percent (1%) of the Company's outstanding capital stock, a lock-up agreement similar to that set forth in this Section 4.

Notwithstanding the foregoing, the obligations described in this Section 4 shall not apply to a registration relating solely to employee benefit plans on Form S-l or Form S-8 or similar forms which may be promulgated in the future, or a registration relating solely to a transaction under Rule 145 of the Act.

5. Restrictive Legends and Stop-Transfer Orders

a. <u>Legends</u>. Denaro understands and agrees that the Company shall cause the legends set forth below or legends substantially equivalent thereto, to be placed upon any certificate(s) evidencing ownership of the Shares together with any other legends that may be required by the Company or by state or federal securities laws:

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 (THE "ACT") AND MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, PLEDGED OR HYPOTHECATED UNLESS AND UNTIL REGISTERED UNDER THE ACT OR, IN THE OPINION OF COUNSEL REASONABLY SATISFACTORY TO THE ISSUER OF THESE SECURITIES, SUCH OFFER, SALE OR TRANSFER, PLEDGE OR HYPOTHECATION IS IN COMPLIANCE THEREWITH.

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN RESTRICTIONS ON TRANSFER AND A RIGHT OF FIRST REFUSAL HELD BY THE ISSUER OR ITS ASSIGNEE(S) AS SET FORTH IN THE STOCK RESTRICTION AGREEMENT BETWEEN THE ISSUER AND THE ORIGINAL HOLDER OF THESE SHARES, A COPY OF WHICH MAY BE OBTAINED AT THE PRINCIPAL OFFICE OF THE ISSUER. SUCH TRANSFER RESTRICTIONS AND RIGHT OF FIRST REFUSAL ARE BINDING ON TRANSFEREES OF THESE SHARES.

- b. <u>Stop-Transfer Notices</u>. Denaro agrees that, in order to ensure compliance with the restrictions referred to herein, the Company may issue appropriate "stop transfer" instructions to its transfer agent, if any, and that, if the Company transfers its own securities, it may make appropriate notations to the same effect in its own records.
- c. <u>Refusal to Transfer</u>. The Company shall not be required (i) to transfer on its books any Shares that have been sold or otherwise transferred in violation of any of the provisions of this Agreement or (ii) to treat as owner of such Shares, or to accord the right to vote or pay dividends, or accord the Put Right to any purchaser or other transferee to whom such Shares shall have been so transferred.
 - 6. Representations and Warranties. The Company and Denaro hereby represents and warrants to the other that:
 - a. it has the full power, authority and legal right to incur the obligations provided for in this Agreement, to execute and deliver this Agreement, and to perform and observe the terms and provisions hereof;
- b. this Agreement constitutes its legal, valid and binding obligation, enforceable against it in accordance with the terms thereof, subject to applicable bankruptcy or other laws affecting creditors' rights generally, and subject to general principles of equity, regardless of whether considered in a proceeding in equity or at law;
 - c. the execution, delivery and performance of this Agreement have been duly authorized by all necessary actions on its part;
- d. the execution, delivery and performance of this Agreement does not violate or exceed its or his powers or contravene (i) any provision of any applicable law, regulation, decree or order to which it is subject, (ii) any provision of its charter documents, or (iii) any provision of any mortgage, deed, contract, agreement or other instrument to which it is a party, or which is binding upon it or any of its assets;
 - e. all authorizations, consents, approvals and licenses required for the execution, delivery and performance of this Agreement have been duly obtained or granted and are in full force and effect.

7. General Provisions

a. <u>Set-Off.</u> All payments which the Company is required to make under this Agreement shall be (i) subject to set-off in accordance with Section 6.5 of the Merger Agreement; and (ii) made without deduction or withholding for any tax unless the Company is required to make a deduction or withholding by applicable law, regulation or rule; provided, however, that Denaro shall be responsible for any applicable transfer tax on the sale of the Shares.

- b. <u>Further Documents</u>. Each party agrees upon request of the other party to execute any further documents or instruments necessary or reasonably desirable in the view of the other party to carry out the purposes or intent of this Agreement.
- c. <u>Waiver</u>. A party's failure to enforce any provision of this Agreement shall not in any way be construed as a waiver of any such provision, nor prevent that party from thereafter enforcing any other provision of this Agreement. The rights granted the parties hereunder are cumulative and shall not constitute a waiver of any party's right to assert any other legal remedy available to it. The waiver of a breach of any term or provision of this Agreement, which must be in writing, shall not operate as or be construed to be a waiver of any other previous or subsequent breach of this Agreement.
- d. Severability. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction to be invalid, void or unenforceable, then the remainder of the terms, provisions, covenants and restrictions of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.
- e. Binding Effect and Assignment. The Company may assign any of its rights under this Agreement to single or multiple assignees, and this Agreement shall inure to the benefit of the successors and assigns of the Company. Notwithstanding any such assignment, the Company shall remain subject to its obligations set forth herein. Subject to the restrictions on transfer herein set forth, this Agreement shall be binding upon Denaro and his or her heirs, executors, administrators, successors and assigns.
 - f. Amendments and Modification. This Agreement may not be modified, amended, altered or supplemented except upon the execution and delivery of a written agreement executed by the Company and Denaro.
- g. Remedies. Notwithstanding any other provision of this Agreement, in the event the Company defaults under Section 2 above or is unable to honor the Put Right because of Section 2(d) above or fails to timely pay the Put Price hereunder for any other reason, and such failure or default is uncured within thirty (30) days after written notice from Denaro of his election to exercise his rights under this Section 7(g), then Denaro shall be entitled to pursue any and all remedies available to him, at law or in equity or otherwise, against the Company with respect to such default or failure to pay.
- h. <u>Specific Performance</u>; <u>Injunctive Relief</u>. The parties hereto acknowledge that Company will be irreparably harmed and that there will be no adequate remedy at law for a violation of any of the covenants or agreements of Denaro set forth in Section 3 or Section 4. Therefore, it is agreed that, in addition to any other remedies that may be available to Company upon any such violation, Company shall have the right to enforce such covenants and agreements by specific performance, injunctive relief or by any other means available to Company at law or in equity.
- i. Notices. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or by commercial messenger or courier service, or mailed by registered or certified mail (return receipt requested) or sent via facsimile (with acknowledgment of complete transmission) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice, except that such notices of change of address shall only be effective upon receipt); provided, however, that notices sent by mail will not be deemed given until received:

If to the Company: comScore Networks, Inc.

11465 Sunset Hills Road, Suite 200 Reston, Virginia 20190 Attention: Chief Financial Officer Telephone Number: (703)438-2000

With a copy to: comScore Networks, Inc.

11465 Sunset Hills Road, Suite 200 Reston, Virginia 20190 Attention: Corporate Counsel Telephone Number:(703) 43 8-2000

If to Denaro: To the address for notice set forth on the signature page hereof.

The Company agrees to notify Denaro of any change in its address above.

- j. Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, regardless of the laws that might otherwise govern under applicable principles of onflicts of laws thereof
- k. Attorney Fees. If either party hereto brings any action to enforce his or its rights hereunder, the prevailing party in any such action shall be entitled to recover his or its reasonable attorneys' fees and other easonable costs incurred in connection with such action
- l. Interpretation. The captions and section headings herein are for convenience only and shall not affect the construction or interpretation of this Agreement. Whenever required by the context, the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; and the neuter gender shall include the masculine and feminine genders. As used in this General Release, the words "includes" and "including," and variations thereof, shall not be deemed to be terms of limitation, and shall be deemed to be followed by words "without limitation."
- m. Entire Agreement. This Agreement and the Merger Agreement (and the exhibits thereto), constitute the entire agreement of the parties with respect to the subject matter hereof and supersede in their entirety all prior undertakings and agreements of the Company and Denaro with respect to the subject matter hereof.
- n. Counterparts. This Agreement may be executed in several counterparts, each of which shall be an original, but all of which together shall constitute one and the same agreement. Facsimile signatures shall be valid and binding as original manual signatures.
 - o. Effect of Headings. The captions and section headings herein are for convenience only and shall not affect the construction or interpretation of this Agreement.

[Remainder of Page Intentionally Left Blank]

The parties represent that they have read this Agreement in its entirety, have had an opportunity to onotify the Company of any change in his address below.	btain the advice of counsel prior to executing this Agreement and fully understand this Agreement. Denaro agrees to
LAWRENCE DENARO	COMSCORE NETWORKS, INC
/s/ Lawrence Denaro	/s/ Sheri L Huston
Signature	Signature
Lawrence Denaro	Sheri L Huston
Print Name	Print Name
	CFO
	Print Title

[Signature Page to Stock Restriction Agreement]

STOCK RESTRICTION AND PUT RIGHT AGREEMENT

This Stock Restriction Agreement (the "Agreement") is made as of January 1, 2005 by and among comScore Networks, Inc., a Delaware corporation (the "Company") and 954253 Ontario Inc., an Ontario corporation, Rice and Associates Advertising Consultants, Inc., an Ontario corporation (each, a Stockholder" and collectively, the "Stockholder").

RECITAL

- A. Company, the Stockholders and certain other parties have entered into an Asset Purchase Agreement (the "*Purchase Agreement*") which provides for the acquisition of all or substantially all, of the assets of Stockholder by Company's wholly-owned subsidiary comScore Media Metrix, Canada. Inc., an Ontario corporation. Pursuant to the Purchase Agreement and as consideration for the assets, the Stockholders shall receive, in part, shares of common stock of Company, and a corresponding Put Right as defined below, to sell such shares back to the Company.
 - B. Capitalized terms used but not defined in this Agreement shall have the meaning ascribed to such terms in the Purchase Agreement.
 - C. As a condition to the Purchase Agreement, Stockholder and Company agreed to enter into this Agreement.

NOW THEREFORE, in consideration of the mutual covenants and representations set forth herein, intending to be legally bound hereby, the parties agree as follows:

AGREEMENT

1. <u>Issuance of the Shares</u>. Pursuant to the terms and conditions of the Purchase Agreement, at the Closing, the Company shall issue to the Stockholders an aggregate of 678,172 shares of common stock of the Company (the "Common Stock"). For purposes hereof, the term "Shares" shall mean the Common Stock as adjusted for stock dividends, stock distributions, combinations, consolidations or splits with respect to such shares.

2. Stockholder's Put Right

a. Subject to the terms and conditions of this Section 2, on the date that is the third anniversary of the Closing Date (the "Exercise Date"), each Stockholder shall have the right to require the Company to repurchase some or all of the Shares at a repurchase price per share of \$2.67 United States Dollars (as adjusted for stock dividends, stock distributions, combinations, consolidations or splits with respect to such Shares) (the "Put Price") (the "Put Right"). Each Stockholder shall have ninety (90) days from and including the Exercise Date to Exercise this Put Right (the "Exercise Period"). Upon expiration of the Exercise Period, this Put Right shall terminate.

- b. To exercise the Put Right, during the Exercise Period, the Stockholder shall deliver a written notice to the Company in accordance with Section 8(h) below, of its election to exercise this Put Right that includes the number of Shares it is electing to have repurchased. Within 10 business days of receipt of such written notice and upon the receipt from such Stockholder of the Company stock certificate(s) duly endorsed for transfer, the Company shall pay to such Stockholder the aggregate Put Price in cash (by check or wire transfer). If less than all of the Shares are repurchased, the Company also shall issue to such Stockholder a new stock certificate representing the remaining Shares held by such Stockholder after such repurchase and a corresponding portion of the Put Right shall survive with respect to the remaining shares for the remainder of the Exercise Period.
- c. Notwithstanding anything to the contrary contained in this Section 2, the Company's obligation to repurchase any Shares under this Section 2 is subject to Delaware General Corporation Law and any other applicable local, state, provincial or federal law, statute or rule of the United States or Canada (collectively, the "Laws"). If the Company is unable to honor its payment obligations because of application of such Laws, the Exercise Period shall remain in effect for a ninety (90) day period beginning on the day that the Company is no longer prevented from honoring the Put Right because of application of such Laws.
- 3. <u>Company's Right of First Refusal</u>. Before any Shares held by a Stockholder or any transferee (either being sometimes referred to herein as the "*Holder*") may be sold or otherwise transferred (including transfer by gift or operation of law), the Company or its assignee(s) shall have a right of first refusal to purchase the Shares on the terms and conditions set forth in this Section 3 (the "*Right of First Refusal*").
- a. <u>Notice of Proposed Transfer</u>. The Holder of the Shares shall deliver to the Company a written notice (the "*Notice*") in accordance with Section 8(h) below stating: (i) the Holder's bona fide intention to sell or otherwise transfer such Shares; (ii) the name of each proposed purchaser or other transferee ("*Proposed Transferee*"); (iii) the number of Shares to be transferred to each Proposed Transferee; and (iv) the bona fide cash price or other consideration for which the Holder proposes to transfer the Shares (the "*Offered Price*"), and the Holder shall offer the Shares at the Offered Price to the Company or its assignee(s).
- b. Exercise of Right of First Refusal. At any time within ten (10) days after receipt of the Notice, the Company and/or its assignee(s) may, by giving written notice to the Holder, elect to purchase some or all of the Shares proposed to be transferred to any one or more of the Proposed Transferees, at the purchase price determined in accordance with subsection (c) below.
- c. <u>Purchase Price</u>. The purchase price ("*Purchase Price*") for the Shares purchased by the Company or its assignee(s) under this Section shall be the Offered Price. If the Offered Price includes consideration other than cash, the officers of the Company and Stockholder in good faith shall agree on the cash equivalent value of the non-cash consideration.
 - d. Payment. Payment of the Purchase Price shall be made in cash (by check or wire/transfer) within thirty (30) days after receipt of the Notice.

- e. <u>Holder's Right to Transfer</u>. To the extent all of the Shares proposed in the Notice to be transferred to a given Proposed Transferee are not purchased by the Company and/or its assignee(s) as provided in this Section 3, then the Holder may sell or otherwise transfer such unpurchased Shares to that Proposed Transferee at the Offered Price or at a higher price, provided that (i) such sale or other transfer is consummated within 120 days after the date of the Notice, (ii) any such sale or other transfer is effected in accordance with any applicable securities laws (including Regulation S (Rule 901 through 905, and Preliminary Notes) of the Securities Act of 1936, as amended (the "Securities Act"), pursuant to registration under the Securities Act, or pursuant to an available exemption from registration); (iii) the Proposed Transferee agrees in writing to guarantee the obligations of Stockholder under the Purchase Agreement up to the value of the Shares so transferred (with such Shares remaining subject to Section 6.5 of the Purchase Agreement); and (iv) the Proposed Transferee agrees in writing that the provisions of this Agreement shall continue to apply to the Shares in the hands of such Proposed Transferee. If the Shares described in the Notice are not transferred to the Proposed Transferee within such period, a new Notice shall be given to the Company, and the Company and/or its assignees shall again be offered the Right of First Refusal before any Shares held by the Holder may be sold or other transferred.
- f. Exception for Certain Transfers. Notwithstanding anything to the contrary contained in this Section 3, the Right of First Refusal shall not apply to, (i) in the case of any Holder that is a partnership, limited liability company or corporation, a transferee or assignee who is a former constituent partner, member, stockholder or affiliate of that Holder, and (ii) in the case of any Holder, (y) a transferee or assignee who is a spouse, lineal descendant, father, mother, brother or sister (each, a "Family Member") of such transferring Holder or (z) or to a trust, the beneficiaries of which are exclusively the Holder and/or Family Members; provided that (i) any such sale or other transfer is effected in accordance with any applicable securities laws; (ii) the transferee or assignee agrees in writing to guarantee the obligations of Stockholder under the Purchase Agreement up to the value of the Shares so transferred (with such Shares remaining subject to Section 6.5 of the Purchase Agreement); and (iii) the Proposed Transferee agrees in writing that the provisions of this Agreement shall continue to apply to the Shares in the hands of such Proposed Transferee.
 - g. Attachment of Put Right. The Put Right is attached to the Shares and accrues to the benefit of any and all holders of such Shares.
- h. <u>Termination of Right of First Refusal</u>. The Right of First Refusal shall terminate as to any Shares upon the earlier of (i) the first sale of Common Stock of the Company to the general public pursuant to a registration statement filed with and declared effective by the Securities and Exchange Commission under the Securities Act, and (ii) a change of control of the Company in which the successor corporation has equity securities that are publicly traded on a national securities exchange or on the Nasdaq Stock Market.
- 4. <u>Lock-Up Period</u>. Stockholder hereby agrees that, if so requested by the representative of the managing' underwriter (the "Managing Underwriter") in connection with any registration of the offering of any securities of the Company under the Securities Act, Stockholder shall not sell or otherwise transfer any Shares or other securities of the Company during the 180-day period (the "Market Standoff Period") following the effective date of a registration statement of the Company filed under the Securities Act. Such restriction shall apply only to the first registration statement of

the Company to become effective under the Securities Act that includes securities to be sold on behalf of the Company to the public in an underwritten public offering under the Securities Act. The Company may impose stop-transfer instructions with respect to securities subject to the foregoing restrictions until the end of such Market Standoff Period,

- 5. Hedging Transactions. Stockholder agrees not to engage in hedging transactions with regard to the Shares unless in compliance with the Securities Act.
- 6. Restrictive Legends and Stop-Transfer Orders.
- a. <u>Legends</u>. Stockholder understands and agrees that the Company shall cause the legends set forth below or legends substantially equivalent thereto, to be placed upon any certificate(s) evidencing ownership of the Shares together with any other legends that may be required by the Company or by state, provincial or federal securities laws of the United States or Canada:

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 (THE "ACT") AND MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, PLEDGED OR HYPOTHECATED UNLESS AND UNTIL REGISTERED UNDER THE ACT OR, IN THE OPINION OF COUNSEL REASONABLY SATISFACTORY TO THE ISSUER OF THESE SECURITIES, SUCH OFFER, SALE OR TRANSFER, PLEDGE OR HYPOTHECATION IS IN COMPLIANCE WITH THE ACT, INCLUDING REGULATION S (RULE 901 THROUGH 905, AND PRELIMINARY NOTES) OR AN AVAILABLE EXEMPTION FROM REGISTRATION. HEDGING TRANSACTIONS INVOLVING THESE SHARES MAY NOT BE CONDUCTED UNLESS IN COMPLIANCE WITH THE ACT.

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN RESTRICTIONS ON TRANSFER AND A RIGHT OF FIRST REFUSAL HELD BY THE ISSUER OR ITS ASSIGNEE(S) AS SET FORTH IN THE STOCK RESTRICTION AGREEMENT BETWEEN THE ISSUER AND THE ORIGINAL HOLDER OF THESE SHARES, A COPY OF WHICH MAY BE OBTAINED AT THE PRINCIPAL OFFICE OF THE ISSUER. SUCH TRANSFER RESTRICTIONS AND RIGHT OF FIRST REFUSAL ARE BINDING ON TRANSFEREES OF THESE SHARES.

b. <u>Stop-Transfer Notices</u>. Each Stockholder agrees that, in order to ensure compliance with the restrictions referred to herein, the Company may issue appropriate "stop transfer" instructions to its transfer agent, if any, and that, if the Company transfers its own securities, it may make appropriate notations to the same effect in its own records.

- c. <u>Refusal to Transfer</u>. The Company shall not be required (i) to transfer on its books any Shares that have been sold or otherwise transferred in violation of any of the provisions of this Agreement or (ii) to treat as owner of such Shares, or to accord the right to vote or pay dividends, or accord the Put Right to any purchaser or other transferree to whom such Shares shall have been so transferred.
 - 7. Representations and Warranties. The Company and each Stockholder hereby represents and warrants to the other that:
 - a. it has the full power, authority and legal right to incur the obligations provided for in this Agreement, to execute and deliver this Agreement, and to perform and observe the terms and provisions hereof;
- b. this Agreement constitutes its legal, valid and binding obligation, enforceable against it in accordance with the terms thereof, subject to applicable bankruptcy or other laws affecting creditors' rights generally, and subject to general principles of equity, regardless of whether considered in a proceeding in equity or at law;
 - c. the execution, delivery and performance of this Agreement have been duly authorized by all necessary actions on its part; and
- d. the execution, delivery and performance of this Agreement does not violate or exceed its or his powers or contravene (i) any provision of any applicable law, regulation, decree or order to which it is subject, (ii) any provision of its charter documents, or (iii) any provision of any mortgage, deed, contract, agreement or other instrument to which it is a party, or which is binding upon it or any of its assets;

General Provisions.

- a. <u>Set-Off.</u> All payments which the Company is required to make under this Agreement shall be (i) subject to set-off in accordance with Section 6.5 of the Purchase Agreement; and (ii) made without deduction or withholding for any tax unless the Company is required to make a deduction or withholding by applicable law, regulation or rule; provided, however, that each Stockholder shall be responsible for any applicable transfer tax on the sale of the Shares.
- b. <u>Further Documents</u>. Each party agrees upon request of the other party to execute any further documents or instruments necessary or reasonably desirable in the view of the other party to carry out the purposes or intent of this Agreement.
- c. <u>Waiver</u>. A party's failure to enforce any provision of this Agreement shall not in any way be construed as a waiver of any such provision, nor prevent that party from thereafter enforcing any other provision of this Agreement. The rights granted the parties hereunder are cumulative and shall not constitute a waiver of any party's right to assert any other legal remedy available to it. The waiver of a breach of any term or provision of this Agreement, which must be in writing, shall not operate as or be construed to be a waiver of any other previous or subsequent breach of this Agreement.

- d. <u>Severability</u>, If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction to be invalid, void or unenforceable, then the remainder of the terms, provisions, covenants and restrictions of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.
- e. <u>Binding Effect and Assignment</u>. The Company may assign any of its rights under this Agreement to single or multiple assignees, and this Agreement shall inure to the benefit of the successors and assigns of the Company. Notwithstanding any such assignment, the Company shall remain subject to its obligations set forth herein. Subject to the restrictions on transfer herein set forth, this Agreement shall be binding upon each Stockholder and his, her or its heirs, executors, administrators, successors and assigns.
- f. Amendments and Modification. This Agreement may not be modified, amended, altered or supplemented except upon the execution and delivery of a written agreement executed by the Company and the Stockholders.
- g. <u>Specific Performance</u>: <u>Injunctive Relief</u>. The parties hereto acknowledge that the Company will be irreparably harmed and that there will be no adequate remedy at law for a violation of any of the covenants or agreements of a Stockholder set forth in Section 3 or Section 4. Therefore, it is agreed that, in addition to any other remedies that may be available to Company upon any such violation, Company shall have the right to enforce such covenants and agreements by specific performance, injunctive relief or by any other means available to Company at law or in equity.
- h. Notices. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or by commercial messenger or courier service, or mailed by registered or certified mail (return receipt requested) or sent via facsimile (with acknowledgment of complete transmission) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice, except that such notices of change of address shall only be effective upon receipt); provided, however, that notices sent by mail will not be deemed given until received;

If to the Company: comScore Networks, Inc.

11465 Sunset Hills Road, Suite 200 Reston, Virginia 20190 Attention: Chief Financial Officer Telephone Number: (703) 438-2000

With a copy to: comScore Networks, Inc.

11465 Sunset Hills Road, Suite 200 Reston, Virginia 20190

Attention: Corporate Counsel Telephone Number: (703) 438-2000

If to a Stockholder: To the address for notice set forth on the signature page hereof.

Each party agrees to notify the other party of any change in its address above.

- i. <u>Governing Law.</u> This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, regardless of the laws that might otherwise govern under applicable principles of conflicts of laws thereof.
- j. <u>Legal Counsel Fees</u>. If either party hereto brings any action to enforce his or its rights hereunder, the prevailing party in any such action shall be entitled to recover his or its reasonable legal counsel fees and other reasonable costs incurred in connection with such action.
- k. Interpretation. The captions and section headings herein are for convenience only and shall not affect the construction or interpretation of this Agreement. Whenever required by the context, the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; and the neuter gender shall include the masculine and feminine genders. As used in this Agreement, the words "includes" and "including," and variations thereof, shall not be deemed to be terms of limitation, and shall be deemed to be followed by words "without limitation."
- 1. Entire Agreement. This Agreement and the Purchase Agreement (and the exhibits thereto), constitute the entire agreement of the parties with respect to the subject matter hereof and supersede in their entirety all prior undertakings and agreements of the Company and the Stockholders with respect to the subject matter hereof.
- m. Counterparts. This Agreement may be executed in several counterparts, each of which shall be an original, but all of which together shall constitute one and the same agreement. Facsimile signatures shall be valid and binding as original manual signatures.
 - n. Effect of Headings. The captions and section headings herein are for convenience only and shall not affect the construction or interpretation of this Agreement.
- o. <u>Acknowledgment</u>. Each Stockholder acknowledges that such Stockholder has had independent legal counsel, has had sufficient time to, and has carefully read and fully understands all the provisions of this Agreement, and is knowingly and voluntarily entering into this Agreement.

[Signature Page Follows]

IN WITNESS WHEREOF, Company and Stockholder have caused this Agreement to be executed and delivered as of the date first above written.

COMPANY:

COMSCORE NETWORKS, INC.:

By: /s/ Sheri L. Huston

> /s/ Sheri L. Huston Name:

Title: /s/ CFO

STOCKHOLDERS

954253 ONTARIO, INC.

/s/ Jeff Hohner By: /s/ Jeff Hohner Name: Title: /s/ President

Address: 52 Parkhurst Boulevard Toronto, Ontario M4G 2C9 Attention: Jeff Hohner Telephone No.: (416) 642-1006 Facsimile No.: (416) 642-1007

with a copy to:

Goodmans LLP Barristers & Solicitors 250 Yonge Street, Suite 2400 Toronto, Ontario M5B 2M6 Attention: Neil Sheehy Telephone No.: (416) 597-4229 Facsimile No.: (416) 979-1234

RICE AND ASSOCIATES ADVERTISING CONSULTANTS, INC

By: /s/ Marshall Rice

Name: /s/ Marshall Rice Title: President

Address:

308 Hidden Trail Toronto, Ontario M2R 3R8

with a copy to:

Goodmans LLP

Barristers & Solicitors 250 Yonge Street, Suite 2400 Toronto, Ontario M5B 2M6 Attention: Neil Sheehy Telephone No.: (416) 597-4229 Facsimile No.: (416) 979-1234

[Signature Page to Stock Restriction Agreement]

COMSCORE, INC.

INDEMNIFICATION AGREEMENT

THIS AGREEMENT is entered into, effective as of ____, 2007 by and between comScore, Inc., a Delaware corporation (the "Company"), and ___("Indemnitee"), effective as of the date that the Registration Statement on Form S-1 related to the initial public offering of the Company's Common Stock is declared effective by the United States Securities and Exchange Commission.

WHEREAS, it is essential to the Company to retain and attract as directors and officers the most capable persons available;

WHEREAS, Indemnitee is a director and/or officer of the Company;

WHEREAS, both the Company and Indemnitee recognize the increased risk of litigation and other claims currently being asserted against directors and officers of corporations;

WHEREAS, the Amended and Restated Certificate of Incorporation (the "Certificate of Incorporation") and Amended and Restated Bylaws (the "Bylaws") of the Company require the Company to indemnify and advance expenses to its directors and officers to the fullest extent permitted under Delaware law, and the Indemnitee has been serving and continues to serve as a director and/or officer of the Company in part in reliance on the Company's Certificate of Incorporation and Bylaws; and

WHEREAS, in recognition of Indemnitee's need for (i) substantial protection against personal liability based on Indemnitee's reliance on the aforesaid Certificate of Incorporation and Bylaws, (ii) specific contractual assurance that the protection promised by the Certificate of Incorporation and Bylaws will be available to Indemnitee (regardless of, among other things, any amendment to or revocation of the Certificate of Incorporation and Bylaws or any change in the composition of the Company's Board of Directors or acquisition transaction relating to the Company) and (iii) an inducement to provide effective services to the Company as a director and/or officer, the Company wishes to provide in this Agreement for the indemnification of and the advancing of expenses to Indemnitee to the fullest extent (whether partial or complete) permitted under Delaware law and as set forth in this Agreement, and, to the extent insurance is maintained, to provide for the continued coverage of Indemnitee under the Company's directors' liability insurance policies.

NOW, THEREFORE, in consideration of the above premises and of Indemnitee continuing to serve the Company directly or, at its request, with another enterprise, and intending to be legally bound hereby, the parties agree as follows:

- 1. Certain Definitions:
- (a) "Affiliate" shall mean any corporation or other person or entity that directly, or indirectly through one or more intermediaries, controls or is controlled by or is under

common control with, the person specified, including, without limitation, with respect to the Company, any direct or indirect subsidiary of the Company.

- (b) "Board" shall mean the Board of Directors of the Company.
- (c) A "Change in Control" shall be deemed to have occurred if (i) any "person" (as such term is used in Sections 13(d) and 14(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act")) (other than a trustee or other fiduciary holding securities under an employee benefit plan of the Company or a corporation owned directly or indirectly by the stockholders of the Company in substantially the same proportions as their ownership of stock of the Company, and other than any person holding shares of the Company on the date that the Company first registers under the Securities Act of 1933, as amended, or any transferee of such individual if such transferee is a spouse or lineal descendant of the transferee or a trust for the benefit of the individual, his or her spouse or lineal descendants), is or becomes the "beneficial owner" (as defined in Rule 13d-3 under the Exchange Act), directly or indirectly, of securities of the Company representing 50% or more of the total voting power represented by the Company's then outstanding Voting Securities, (ii) during any period of two consecutive years, individuals who at the beginning of such period constitute the Board and any new director whose election by the Board or nomination for election by the Company's stockholders was approved, by a vote of at least two-thirds (2/3) of the directors then still in office who either were directors at the beginning of the period or whose election or nomination for election was previously so approved, cease for any reason to constitute a majority of the Board, (iii) the stockholders of the Company approve a merger or consolidation of the Company with any other entity, other than a merger or consolidation that would result in the Voting Securities of the Company outstanding immediately prior thereto continuing to represent (either by remaining outstanding or by being converted into Voting Securities of the Surviving entity) at least 80% of the total voting power represented by the Voting Securities of the Company or such su
- (d) "Expenses" shall mean any expense, liability or loss, including attorneys' fees, judgments, fines, ERISA excise taxes and penalties, amounts paid or to be paid in settlement, any interest, assessments or other charges imposed thereon, any federal, state, local or foreign taxes imposed as a result of the actual or deemed receipt of any payments under this Agreement and all other costs and obligations, paid or incurred in connection with investigating, defending, being a witness in, participating in (including on appeal) or preparing for any of the foregoing in, any Proceeding relating to any Indemnifiable Event.
- (e) "Indemnifiable Event" shall mean any event or occurrence that takes place either prior to or after the execution of this Agreement, related to the fact that Indemnitee is or was a director or officer of the Company or an Affiliate of the Company, or while a director or officer is or was serving at the request of the Company or an Affiliate of the Company as a director, officer, employee, trustee, agent or fiduciary of another foreign or domestic corporation, partnership, joint venture, employee benefit plan, trust or other enterprise or was a director, officer, employee or agent of a foreign or domestic corporation that was a predecessor corporation of the Company or of

another enterprise at the request of such predecessor corporation, or related to anything done or not done by Indemnitee in any such capacity, whether or not the basis of the Proceeding is alleged action in an official capacity as a director, officer, employee or agent of the Company or an Affiliate of the Company, as described above.

- (f) "Independent Counsel" shall mean the person or body appointed in connection with Section 3.
- (g) "Proceeding" shall mean any threatened, pending or completed action, suit or proceeding or any alternative dispute resolution mechanism (including an action by or in the right of the Company or an Affiliate of the Company) or any inquiry, hearing or investigation, whether conducted by the Company or an Affiliate of the Company or any other party, that Indemnitee in good faith believes might lead to the institution of any such action, suit or proceeding, whether civil, criminal, administrative, investigative or other.
 - (h) "Reviewing Party" shall mean the person or body appointed in accordance with Section 3.
 - (i) "Voting Securities" shall mean any securities of the Company that vote generally in the election of directors.
 - 2. Agreement to Indemnify.
- (a) General Agreement. In the event Indemnitee was, is or becomes a party to or witness or other participant in, or is threatened to be made a party to or witness or other participant in, a Proceeding by reason of (or arising in part out of) an Indemnifiable Event, the Company shall indemnify Indemnitee from and against any and all Expenses to the fullest extent permitted by law, as the same exists or may hereafter be amended or interpreted (but in the case of any such amendment or interpretation, only to the extent that such amendment or interpretation permits the Company to provide broader indemnification rights than were permitted prior thereto). The parties hereto intend that this Agreement shall provide for indemnification in excess of that expressly permitted by statute, including, without limitation, any indemnification provided by the Company's Certificate of Incorporation, its Bylaws, vote of its stockholders or disinterested directors or applicable law.
- (b) <u>Initiation of Proceeding</u>. Notwithstanding anything in this Agreement to the contrary, Indemnitee shall not be entitled to indemnification pursuant to this Agreement in connection with any Proceeding initiated by Indemnitee against the Company or any director or officer of the Company unless (i) the Company has joined in or the Board has consented to the initiation of such Proceeding, (ii) the Proceeding is one to enforce indemnification rights under Section 5 or (iii) the Proceeding is instituted after a Change in Control (other than a Change in Control approved by a majority of the directors on the Board who were directors immediately prior to such Change in Control) and Independent Counsel has approved its initiation.

- (c) Expense Advances. If so requested by Indemnitee, the Company shall advance (within thirty (30) days of such request) any and all Expenses to Indemnitee (an "Expense Advance"). The Indemnitee shall qualify for such Expense Advances upon the execution and delivery to the Company of this Agreement which shall constitute an undertaking providing that the Indemnitee undertakes to repay such Expense Advances if and to the extent that it is ultimately determined by a court of competent jurisdiction in a final judgment, not subject to appeal, that Indemnitee is not entitled to be indemnified by the Company. Indemnitee's obligation to reimburse the Company for Expense Advances shall be unsecured and no interest shall be charged thereon. This Section 2(c) shall not apply to any claim made by Indemnitee for which indemnity is excluded pursuant to Section 2(b) or 2(f).
- (d) <u>Mandatory Indemnification</u>. Notwithstanding any other provision of this Agreement, to the extent that Indemnitee has been successful on the merits or otherwise in defense of any Proceeding relating in whole or in part to an Indemnifiable Event or in defense of any issue or matter therein, Indemnitee shall be indemnified against all Expenses incurred in connection therewith.
- (e) <u>Partial Indemnification</u>. If Indemnitee is entitled under any provision of this Agreement to indemnification by the Company for some or a portion of Expenses, but not, however, for the total amount thereof, the Company shall nevertheless indemnify Indemnitee for the portion thereof to which Indemnitee is entitled.
- (f) <u>Prohibited Indemnification</u>. No indemnification pursuant to this Agreement shall be paid by the Company on account of any Proceeding in which a final judgment is rendered against Indemnitee or Indemnitee enters into a settlement, in each case (i) for an accounting of profits made from the purchase or sale by Indemnitee of securities of the Company pursuant to the provisions of Section 16(b) of the Exchange Act or similar provisions of any federal, state or local laws; (ii) for which payment has actually been made to or on behalf of Indemnitee under any insurance policy or other indemnity provision, except with respect to any excess beyond the amount paid under any insurance policy or other indemnity provision; or (iii) for which payment is prohibited by law. Notwithstanding anything to the contrary stated or implied in this Section 2(f), indemnification pursuant to this Agreement relating to any Proceeding against Indemnitee for an accounting of profits made from the purchase or sale by Indemnitee of securities of the Company pursuant to the provisions of Section 16(b) of the Exchange Act or similar provisions of any federal, state or local laws shall not be prohibited if Indemnitee ultimately establishes in any Proceeding that no recovery of such profits from Indemnitee is permitted under Section 16(b) of the Exchange Act or similar provisions of any federal, state or local laws.
- 3. <u>Reviewing Party</u>. Prior to any Change in Control, the Reviewing Party shall be any appropriate person or body consisting of a member or members of the Board or any other person or body appointed by the Board who is not a party to the particular Proceeding with respect to which Indemnitee is seeking indemnification; provided that if all members of the Board are parties to the particular Proceeding with respect to which Indemnitee is seeking indemnification, the Independent Counsel referred to below shall become the Reviewing Party; after a Change in Control, the Independent Counsel referred to below shall become the Reviewing Party. With respect to all matters arising before a Change in Control for which Independent Counsel shall be the Reviewing

Party and all matters arising after a Change in Control, in each case concerning the rights of Indemnitee to indemnity payments and Expense Advances under this Agreement or any other agreement or under applicable law or the Company's Certificate of Incorporation or Bylaws now or hereafter in effect relating to indemnification for Indemnifiable Events, the Company shall seek legal advice only from Independent Counsel selected by Indemnitee and approved by the Company (which approval shall not be unreasonably withheld or delayed), and who has not otherwise performed services for the Company or the Indemnitee (other than in connection with indemnification matters) within the last five years. The Independent Counsel shall not include any person who, under the applicable standards of professional conduct then prevailing, would have a conflict of interest in representing either the Company or Indemnitee in an action to determine Indemnitee's rights under this Agreement. Such counsel, among other things, shall render its written opinion to the Company and Indemnitee as to whether and to what extent the Indemnitee should be permitted to be indemnified under applicable law. The Company agrees to pay the reasonable fees of the Independent Counsel and to indemnify fully such counsel against any and all expenses (including attorneys' fees), claims, liabilities, loss and damages arising out of or relating to this Agreement or the engagement of Independent Counsel pursuant hereto.

4. Indemnification Process and Appeal.

- (a) <u>Indemnification Payment</u>. Indemnitee shall be entitled to indemnification of Expenses, and shall receive payment thereof, from the Company in accordance with this Agreement as soon as practicable after Indemnitee has made written demand on the Company for indemnification, but in no event later than thirty (30) business days after such demand, unless the Reviewing Party has given a written opinion to the Company that Indemnitee is not entitled to indemnification under applicable law. Indemnitee shall cooperate with the Reviewing Party making a determination with respect to Indemnitee's entitlement to indemnification, including providing to the Reviewing Party upon reasonable advance request any documentation or information which is not privileged or otherwise protected from disclosure and which is reasonably available to Indemnitee and reasonably necessary to such determination.
- (b) Suit to Enforce Rights. Regardless of any action by the Reviewing Party, if Indemnitee has not received full indemnification within thirty (30) days after making a demand in accordance with Section 4(a), Indemnitee shall have the right to enforce its indemnification rights under this Agreement by commencing litigation in any court in the State of California or the State of Delaware having subject matter jurisdiction thereof seeking an initial determination by the court or challenging any determination by the Reviewing Party or any aspect thereof. The Company hereby consents to service of process and to appear in any such proceeding. Any determination by the Reviewing Party not challenged by the Indemnitee shall be binding on the Company and Indemnitee. The Company shall be precluded from asserting in any such proceeding that the procedures and presumptions of this Agreement are not valid, binding and enforceable and shall stipulate in any such court that the Company is bound by all the provisions of this Agreement. The remedy provided for in this Section 4 shall be in addition to any other remedies available to Indemnitee at law or in equity.

(c) Defense to Indemnification, Burden of Proof, and Presumptions. It shall be a defense to any action brought by Indemnitee against the Company to enforce this

Agreement (other than an action brought to enforce a claim for Expenses incurred in defending a Proceeding in advance of its final disposition) that it is not permissible under applicable law for the Company to indemnify Indemnitee for the amount claimed. In connection with any such action or any determination by the Reviewing Party or otherwise as to whether Indemnitee is entitled to be indemnified hereunder, the burden of proving such a defense or determination shall be on the Company. Neither the failure of the Reviewing Party or the Company (including its Board, independent legal counsel or its stockholders) to have made a defense to the action by Indemnitee that indemnification of the claimant is proper under the circumstances because Indemnitee has met the standard of conduct set forth in applicable law, nor an actual determination by the Reviewing Party or Company (including its Board, independent legal counsel or its stockholders) that the Indemnitee had not met such applicable standard of conduct. For purposes of this Agreement, the termination of any claim, action, suit or proceeding, by judgment, order, settlement (whether with or without court approval), conviction or upon a plea of nolo contendere or its equivalent, shall not create a presumption that Indemnitee did not meet any particular standard of conduct or have any particular belief or that a court has determined that indemnification is not permitted by applicable law. For purposes of any determination of good faith under any applicable standard of conduct, Indemnitee shall be deemed to have acted in good faith if Indemnitee's action is based on the records or books of account of the Company, including financial statements, or on information supplied to Indemnitee by the officers of the Company in the course of their duties, or on the advice of legal counsel for the Company or the Board or on information or records given or reports made to the Company by an independent certified public accountant or by an appraiser, investment banker or other

- 5. <u>Indemnification for Expenses Incurred in Enforcing Rights</u>. The Company shall indemnify Indemnitee against any and all Expenses that are incurred by Indemnitee in connection with any action brought by Indemnitee for
- (i) indemnification or advance payment of Expenses by the Company under this Agreement or any other agreement or under applicable law or the Company's Certificate of Incorporation or Bylaws now or hereafter in effect relating to indemnification for Indemnifiable Events, and/or;
- (ii) recovery under directors' and officers' liability insurance policies maintained by the Company; but only in the event that Indemnitee ultimately is determined to be entitled to such indemnification or insurance recovery, as the case may be. In addition, the Company shall, if so requested by Indemnitee, advance the foregoing Expenses to Indemnitee, subject to and in accordance with Section 2(c).
 - 6. Notification and Defense of Proceeding,

- (a) Notice. Promptly after receipt by Indemnitee of notice of the commencement of any Proceeding, Indemnitee shall, if a claim in respect thereof is to be made against the Company under this Agreement, notify the Company of the commencement thereof; but the omission so to notify the Company will not relieve the Company from any liability that it may have to Indemnitee, except as provided in Section 6(c).
- (b) <u>Defense</u>. With respect to any Proceeding as to which Indemnitee notifies the Company of the commencement thereof, the Company will be entitled to participate in the Proceeding at its own expense and except as otherwise provided below, to the extent the Company so wishes, it may assume the defense thereof with counsel reasonably satisfactory to Indemnitee. After notice from the Company to Indemnitee of its election to assume the defense of any Proceeding, the Company shall not be liable to Indemnitee under this Agreement or otherwise for any Expenses subsequently incurred by Indemnitee in connection with the defense of such Proceeding other than reasonable costs of investigation or as otherwise provided below. Indemnitee shall have the right to employ legal counsel in such Proceeding, but all Expenses related thereto incurred after notice from the Company of its assumption of the defense shall be at Indemnitee's expense unless: (i) the employment of legal counsel by Indemnitee has been authorized by the Company, (ii) Indemnitee has reasonably determined that there may be a conflict of interest between Indemnitee and the Company in the defense of the Proceeding, (iii) after a Change in Control, the employment of counsel by Indemnitee has been approved by the Independent Counsel or (iv) the Company shall not in fact have employed counsel to assume the defense of such Proceeding, in each of which cases all Expenses of the Proceeding shall be borne by the Company. The Company shall not be entitled to assume the defense of any Proceeding brought by or on behalf of the Company, or as to which Indemnitee shall have made the determination provided for in (iii) above or under the circumstances provided for in (iii) and (iv) above.
- (c) Settlement of Claims. The Company shall not be liable to indemnify Indemnitee under this Agreement or otherwise for any amounts paid in settlement of any Proceeding effected without the Company's written consent, such consent not to be unreasonably withheld; provided, however, that if a Change in Control has occurred, the Company shall be liable for indemnification of Indemnitee for amounts paid in settlement if the Independent Counsel has approved the settlement. The Company shall not settle any Proceeding in any manner that would impose any penalty or limitation on Indemnitee without Indemnitee's written consent. The Company shall not be liable to indemnify the Indemnitee under this Agreement with regard to any judicial award if the Company was not given a reasonable and timely opportunity as a result of Indemnitees' failure to provide notice, at its expense, to participate in the defense of such action, and the lack of such notice materially prejudiced the Company's ability to participate in defense of such action. The Company's liability hereunder shall not be excused if participation in the Proceeding by the Company was barred by this Agreement.
- 7. Non-Exclusivity. The rights of Indemnitee hereunder shall be in addition to any other rights Indemnitee may have under the Company's Certificate of Incorporation, Bylaws, applicable law or otherwise; provided, however, that this Agreement shall supersede any prior indemnification agreement between the Company and the Indemnitee. To the extent that a change in applicable law (whether by statute or judicial decision) permits greater indemnification than would be afforded currently under the Company's Certificate of Incorporation, Bylaws, applicable law or

this Agreement, it is the intent of the parties that Indemnitee enjoy by this Agreement the greater benefits so afforded by such change.

- 8. <u>Liability Insurance</u>. To the extent the Company maintains an insurance policy or policies providing general and/or directors' and officers' liability insurance, Indemnitee shall be covered by such policy or policies, in accordance with its or their terms, to the maximum extent of the coverage available for any Company director or officer.
- 9. <u>Period of Limitations</u>. No legal action shall be brought and no cause of action shall be asserted by or on behalf of the Company or any Affiliate of the Company against Indemnitee, Indemnitee's spouse, heirs, executors or personal or legal representatives after the expiration of two (2) years from the date of accrual of such cause of action or such longer period as may be required by state law under the circumstances. Any claim or cause of action of the Company or its Affiliate shall be extinguished and deemed released unless asserted by the timely filing and notice of a legal action within such period; provided, however, that if any shorter period of limitations is otherwise applicable to any such cause of action, the shorter period shall govern.
- 10. Amendment of this Agreement. No supplement, modification or amendment of this Agreement shall be binding unless executed in writing by both of the parties hereto. No waiver of any of the provisions of this Agreement shall be binding unless in the form of a writing signed by the party against whom enforcement of the waiver is sought, and no such waiver shall operate as a waiver of any other provisions hereof (whether or not similar), nor shall such waiver constitute a continuing waiver. Except as specifically provided herein, no failure to exercise or any delay in exercising any right or remedy hereunder shall constitute a waiver thereof.
- 11. <u>Subrogation</u>. In the event of payment under this Agreement, the Company shall be subrogated to the extent of such payment to all of the rights of recovery of Indemnitee, who shall execute all papers required and shall do everything that may be necessary to secure such rights, including the execution of such documents necessary to enable the Company effectively to bring suit to enforce such rights.
- 12. No Duplication of Payments. The Company shall not be liable under this Agreement to make any payment in connection with any claim made against Indemnitee to the extent Indemnitee has otherwise received payment (under any insurance policy, Bylaw or otherwise) of the amounts otherwise indemnifiable hereunder.
- 13. <u>Duration of Agreement</u>. This Agreement shall continue until and terminate upon the later of (a) six (6) years after the date that Indemnitee shall have ceased to serve as a director or officer of the Company or (b) one (1) year after the final termination of any Proceeding, including any appeal, then pending in respect of which Indemnitee is granted rights of indemnification or advancement of Expenses hereunder and of any proceeding commenced by Indemnitee pursuant to Section 4(b) of this Agreement relating thereto.
- 14. <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of and be enforceable by the parties hereto and their respective successors (including any direct or indirect successor by purchase, merger, consolidation or otherwise to all or substantially all

of the business and/or assets of the Company), assigns, spouses, heirs and personal and legal representatives. The Company shall require and cause any successor (whether direct or indirect by purchase, merger, consolidation or otherwise) to all, substantially all or a substantial part, of the business and/or assets of the Company, by written agreement in form and substance satisfactory to Indemnitee, expressly to assume and agree to perform this Agreement in the same manner and to the same extent that the Company would be required to perform if no such succession had taken place. The indemnification provided under this Agreement shall continue as to Indemnitee for any action taken or not taken while serving in an indemnified capacity pertaining to an Indemnifiable Event even though Indemnitee may have ceased to serve in such capacity at the time of any Proceeding.

- 15. <u>Severability</u>. If any provision (or portion thereof) of this Agreement shall be held by a court of competent jurisdiction to be invalid, void or otherwise unenforceable, (a) the remaining provisions shall remain enforceable to the fullest extent permitted by law; (b) such provision or provisions shall be deemed reformed to the extent necessary to conform to applicable law and to give the maximum effect to the intent of the parties hereto; and (c) to the fullest extent possible, the provisions of this Agreement (including, without limitation, each portion of this Agreement containing any provision held to be invalid, void or otherwise unenforceable, that is not itself invalid, void or unenforceable) shall be construed so as to give effect to the intent manifested by the provision held invalid, void or unenforceable.
- 16. <u>Contribution</u>. To the fullest extent permissible under applicable law, whether or not the indemnification provided for in this Agreement is available to Indemnitee for any reason whatsoever, the Company shall pay all or a portion of the amount that would otherwise be incurred by Indemnitee for Expenses in connection with any claim relating to an Indemnifiable Event, as is deemed fair and reasonable in light of all of the circumstances of such Proceeding in order to reflect (i) the relative benefits received by the Company and Indemnitee as a result of the event(s) and/or transaction(s) giving cause to such Proceeding; and/or (ii) the relative fault of the Company (and its directors, officers, employees and agents) and Indemnitee in connection with such event(s) and/or transaction(s).
- 17. Entire Agreement. This Agreement constitutes the entire agreement of the parties with respect to the subject matter hereof and supersedes in its entirety all prior undertakings and agreements, including the any prior agreement with respect to the subject matter hereof.
- 18. Governing Law. This Agreement shall be governed by and construed and enforced in accordance with the laws of the State of Delaware applicable to contracts made and to be performed in such State without giving effect to its principles of conflicts of laws. The Company and Indemnitee hereby irrevocably and unconditionally (i) agree that any action or proceeding arising out of or in connection with this Agreement may be brought in the Delaware Court of Chancery or in the applicable state or federal courts in the Commonwealth of Virginia for purposes of any action or proceeding arising out of or in connection with this Agreement, (iii) waive any objection to the laying of venue of any such action or proceeding in the Delaware Court of Chancery or in the applicable state or federal courts in the Commonwealth of Virginia, and (iv) waive, and agree not to plead or to make, any claim that any

such action or proceeding brought in the Delaware Court of Chancery or in the applicable state or federal courts in the Commonwealth of Virginia has been brought in an improper or inconvenient forum.

19. Notices. All notices, demands and other communications required or permitted hereunder shall be made in writing and shall be deemed to have been duly given if delivered by hand, against receipt or mailed, postage prepaid, certified or registered mail, return receipt requested and addressed to the Company at:

comScore, Inc. Attn: Chief Executive Officer 11465 Sunset Hills Road Suite 200 Reston, Virginia 20190 Facsimile: 703-438-2051

and to Indemnitee at the address set forth below Indemnitee's signature hereto.

Notice of change of address shall be effective only when given in accordance with this Section. All notices complying with this Section shall be deemed to have been received on the date of hand delivery or on the third business day after mailing.

20. Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

IN WITNESS WHEREOF, the parties hereto have duly executed and delivered this Agreement as of the day specified above.

COMSCORE, INC. a Delaware corporation

y:	
Print Name	:
Title:	
INDEMNITEE, an individual	
Signed:	
Print Name	:
Address:	
_	
_	

[Signature Page to Indemnification Agreement]

COMSCORE, INC.

1999 STOCK PLAN

(As amended March 25, 2007)

- 1. <u>Purposes of the Plan</u>. The purposes of this Stock Plan are to attract and retain the best available personnel for positions of substantial responsibility, to provide additional incentive to Employees, Directors and Consultants and to promote the success of the Company's business. Options granted under the Plan may be Incentive Stock Options or Nonstatutory Stock Options, as determined by the Administrator at the time of grant. Stock Purchase Rights and Restricted Stock Units may also be granted under the Plan.
 - $2.\ \underline{Definitions}.$ As used herein, the following definitions shall apply:
 - (a) "Administrator" means the Board or any of its Committees as shall be administering the Plan in accordance with Section 4 hereof.
- (b) "Applicable Laws" means the requirements relating to the administration of equity-based awards under U.S. state corporate laws, U.S. federal and state securities laws, the Code, any stock exchange or quotation system on which the Common Stock is listed or quoted and the applicable laws of any other country or jurisdiction where Awards are granted under the Plan.
 - (c) "Award" means, individually or collectively, a grant under the Plan of Options, Stock Purchase Rights, or Restricted Stock Units as the Administrator may determine.
- (d) "Award Agreement" means the written or electronic agreement setting forth the terms and provisions applicable to each Award granted under the Plan, including an Option Agreement. The Award Agreement is subject to the terms and conditions of the Plan.
 - (e) "Board" means the Board of Directors of the Company.
 - (f) "Code" means the Internal Revenue Code of 1986, as amended.
 - (g) "Committee" means a committee of Directors or another individual or individuals appointed by the Board in accordance with Section 4 hereof.
 - (h) "Common Stock" means the Common Stock of the Company.
 - (i) "Company" means comScore, Inc., a Delaware corporation.
 - (j) "Consultant" means any person who is engaged by the Company or any Parent or Subsidiary to render consulting or advisory services to such entity.
 - (k) "Director" means a member of the Board of Directors of the Company.

- (l) "Disability" means total and permanent disability as defined in Section 22(e)(3) of the Code.
- (m) "Employee" means any person, including Officers and Directors, employed by the Company or any Parent or Subsidiary of the Company. A Service Provider shall not cease to be an Employee in the case of (i) any leave of absence approved by the Company or (ii) transfers between locations of the Company or between the Company, its Parent, any Subsidiary, or any successor. For purposes of Incentive Stock Options, no such leave may exceed ninety days, unless reemployment upon expiration of such leave is guaranteed by statute or contract. If reemployment upon expiration of a leave of absence approved by the Company is not so guaranteed, on the 181st day of such leave any Incentive Stock Option held by the Optionee shall cease to be treated as an Incentive Stock Option and shall be treated for tax purposes as a Nonstatutory Stock Option. Neither service as a Director nor payment of a director's fee by the Company shall be sufficient to constitute "employment" by the Company.
 - (n) "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- (o) "Exchange Program" means a program under which (i) outstanding Awards are surrendered or cancelled in exchange for Awards of the same type (which may have higher or lower exercise prices and different terms), Awards of a different type, and/or cash, (ii) Optionees would have the opportunity to transfer any outstanding Awards to a financial institution or other person or entity selected by the Administrator, and/or (iii) the exercise price of an outstanding Award is reduced. The Administrator will determine the terms and conditions of any Exchange Program in its sole discretion.
 - (p) "Fair Market Value" means, as of any date, the value of Common Stock determined as follows:
- (i) If the Common Stock is listed on any established stock exchange or a national market system, including without limitation the Nasdaq Global Market, the Nasdaq Global Select Market or the Nasdaq Capital Market, its Fair Market Value shall be the closing sales price for such stock (or the closing bid, if no sales were reported) as quoted on such exchange or system for the last market trading day prior to the time of determination, as reported in *The Wall Street Journal* or such other source as the Administrator deems reliable;
- (ii) If the Common Stock is regularly quoted by a recognized securities dealer but selling prices are not reported, its Fair Market Value shall be the mean between the high bid and low asked prices for the Common Stock on the last market trading day prior to the day of determination; or
 - (iii) In the absence of an established market for the Common Stock, the Fair Market Value thereof shall be determined in good faith by the Administrator.
 - (q) "Incentive Stock Option" means an Option intended to qualify as an incentive stock option within the meaning of Section 422 of the Code.

- (r) "Nonstatutory Stock Option" means an Option not intended to qualify as an Incentive Stock Option.
- (s) "Officer" means a person who is an officer of the Company within the meaning of Section 16 of the Exchange Act and the rules and regulations promulgated thereunder.
- (t) "Option" means a stock option granted pursuant to the Plan.
- (u) "Option Agreement" means a written or electronic agreement between the Company and an Optionee evidencing the terms and conditions of an individual Option grant. The Option Agreement is subject to the terms and conditions of the Plan.
 - (v) "Optioned Stock" means the Common Stock subject to an Award.
 - (w) "Optionee" means the holder of an outstanding Award granted under the Plan.
 - (x) "Parent" means a "parent corporation," whether now or hereafter existing, as defined in Section 424(e) of the Code.
 - (y) "Plan" means this 1999 Stock Plan.
 - (z) "Restricted Stock" means shares of Common Stock acquired pursuant to a grant of a Stock Purchase Right under Section 11 below or upon early exercise of an Option.
- (aa) "Restricted Stock Unit" means a bookkeeping entry representing the right to receive one Share or an amount equal to the Fair Market Value of one Share, granted pursuant to Section 12. Each Restricted Stock Unit represents an unfunded and unsecured obligation of the Company.
 - (bb) "Service Provider" means an Employee, Director or Consultant.
 - (cc) "Share" means a share of the Common Stock, as adjusted in accordance with Section 13 below.
 - (dd) "Stock Purchase Right" means a right to purchase Common Stock pursuant to Section 11 below.
 - (ee) "Subsidiary" means a "subsidiary corporation," whether now or hereafter existing, as defined in Section 424(f) of the Code.
- 3. Stock Subject to the Plan. Subject to the Plan is 26,760,284 Shares. The Shares may be authorized but unissued, or reacquired Common Stock.

If an Award expires or becomes unexercisable without having been exercised in full, or is surrendered pursuant to an Exchange Program, the unpurchased Shares which were subject thereto shall become available for future grant or sale under the Plan (unless the Plan has terminated). However, Shares that have actually been issued under the Plan, upon exercise of either an Award,

shall not be returned to the Plan and shall not become available for future distribution under the Plan, except that if unvested Shares of Restricted Stock or Shares acquired pursuant to Restricted Stock Units are repurchased by or forfeited to the Company or , such Shares shall become available for future grant under the Plan.

4. Administration of the Plan.

- (a) Administrator. The Plan shall be administered by the Board or a Committee appointed by the Board, which Committee shall be constituted to comply with Applicable Laws.
- (b) <u>Powers of the Administrator</u>. Subject to the provisions of the Plan and, in the case of a Committee, the specific duties delegated by the Board to such Committee, and subject to the approval of any relevant authorities, the Administrator shall have the authority in its discretion:
 - (i) to determine the Fair Market Value;
 - $(ii) \ to \ select \ the \ Service \ Providers \ to \ whom \ Awards \ may \ from \ time \ to \ time \ be \ granted \ hereunder;$
 - (iii) to determine the number of Shares to be covered by each Award granted hereunder;
 - (iv) to approve forms of agreement for use under the Plan;
- (v) to determine the terms and conditions, of any Award granted hereunder. Such terms and conditions include, but are not limited to, the exercise price, the time or times when Awards may be exercised (which may be based on performance criteria), any vesting acceleration or waiver of forfeiture restrictions, and any restriction or limitation regarding any Award or the Common Stock relating thereto, based in each case on such factors as the Administrator, in its sole discretion, shall determine;
 - (vi) to determine whether and under what circumstances an Option may be settled in cash under subsection 9(e) instead of Common Stock;
 - (vii) to reduce the exercise price of any Award to the then current Fair Market Value if the Fair Market Value of the Common Stock covered by such Award has declined since the date the Award was granted;
 - (viii) to initiate an Exchange Program;
- (ix) to prescribe, amend and rescind rules and regulations relating to the Plan, including rules and regulations relating to sub-plans established for the purpose of qualifying for preferred tax treatment under foreign tax laws;
- (x) to allow Optionees to satisfy withholding tax obligations by electing to have the Company withhold from the Shares to be issued upon exercise of an Award that number of Shares having a Fair Market Value equal to the minimum amount required to be withheld. The Fair Market Value of the Shares to be withheld shall be determined on the date that the amount of tax to

be withheld is to be determined. All elections by Optionees to have Shares withheld for this purpose shall be made in such form and under such conditions as the Administrator may deem necessary or advisable; and

- (xi) to construe and interpret the terms of the Plan and Awards granted pursuant to the Plan.
- (c) Effect of Administrator's Decision. All decisions, determinations and interpretations of the Administrator shall be final and binding on all Optionees.

5. Eligibility.

- (a) Nonstatutory Stock Options, Stock Purchase Rights and Restricted Stock Units may be granted to Service Providers. Incentive Stock Options may be granted only to Employees.
- (b) Each Option shall be designated in the Option Agreement as either an Incentive Stock Option or a Nonstatutory Stock Option. However, notwithstanding such designation, to the extent that the aggregate Fair Market Value of the Shares with respect to which Incentive Stock Options are exercisable for the first time by the Optionee during any calendar year (under all plans of the Company and any Parent or Subsidiary) exceeds \$100,000, such Options shall be treated as Nonstatutory Stock Options. For purposes of this Section 5(b), Incentive Stock Options shall be taken into account in the order in which they were granted. The Fair Market Value of the Shares shall be determined as of the time the Option with respect to such Shares is granted.
- (c) Neither the Plan nor any Award shall confer upon any Optionee any right with respect to continuing the Optionee's relationship as a Service Provider with the Company, nor shall it interfere in any way with his or her right or the Company's right to terminate such relationship at any time, with or without cause.
 - 6. Term of Plan. The Plan shall become effective upon its adoption by the Board. It shall continue in effect for a term of ten (10) years unless sooner terminated under Section 15 of the Plan.
- 7. <u>Term of Option</u>. The term of each Option shall be stated in the Option Agreement; provided, however, that the term shall be no more than ten (10) years from the date of grant thereof. In the case of an Incentive Stock Option granted to an Optionee who, at the time the Option is granted, owns stock representing more than ten percent (10%) of the voting power of all classes of stock of the Company or any Parent or Subsidiary, the term of the Option shall be five (5) years from the date of grant or such shorter term as may be provided in the Option Agreement.

8. Option Exercise Price and Consideration

- (a) The per share exercise price for the Shares to be issued upon exercise of an Option shall be such price as is determined by the Administrator, but shall be subject to the following:
 - (i) In the case of an Incentive Stock Option
- (A) granted to an Employee who, at the time of grant of such Option, owns stock representing more than ten percent (10%) of the voting power of all classes of stock of the Company or any Parent or Subsidiary, the exercise price shall be no less than 110% of the Fair Market Value per Share on the date of grant.
 - (B) granted to any other Employee, the per Share exercise price shall be no less than 100% of the Fair Market Value per Share on the date of grant.
 - (ii) In the case of a Nonstatutory Stock Option, the per Share exercise price shall be no less than 100% of the Fair Market Value per Share on the date of grant.
- (iii) Notwithstanding the foregoing, Options may be granted with a per Share exercise price other than as required above pursuant to a transaction described in, and in a manner consistent with, Section 424(a) of the Code and the regulations promulgated thereunder.
- (b) The consideration to be paid for the Shares to be issued upon exercise of an Option, including the method of payment, shall be determined by the Administrator (and, in the case of an Incentive Stock Option, shall be determined at the time of grant). Such consideration may consist of (without limitation) (1) cash, (2) check, (3) promissory note, (4) other Shares which (x) in the case of Shares acquired upon exercise of an Option, have been owned by the Optionee for more than six months on the date of surrender, and (y) have a Fair Market Value on the date of surrender equal to the aggregate exercise price of the Shares as to which such Option shall be exercised, (5) consideration received by the Company under a cashless exercise program implemented by the Company in connection with the Plan, or (6) any combination of the foregoing methods of payment. In making its determination as to the type of consideration to accept, the Administrator shall consider if acceptance of such consideration may be reasonably expected to benefit the Company.

Exercise of Option

(a) <u>Procedure for Exercise</u>; <u>Rights as a Shareholder</u>. Any Option granted hereunder shall be exercisable according to the terms hereof at such times and under such conditions as determined by the Administrator and set forth in the Option Agreement. Unless the Administrator provides otherwise, vesting of Options granted hereunder to Officers and Directors shall be tolled during any unpaid leave of absence. An Option may not be exercised for a fraction of a Share.

An Option shall be deemed exercised when the Company receives: (i) written or electronic notice of exercise (in accordance with the Option Agreement) from the person entitled to exercise the Option, and (ii) full payment for the Shares with respect to which the Option is exercised. Full payment may consist of any consideration and method of payment authorized by the

Administrator and permitted by the Option Agreement and the Plan. Shares issued upon exercise of an Option shall be issued in the name of the Optionee or, if requested by the Optionee, in the name of the Optionee and his or her spouse. Until the Shares are issued (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company), no right to vote or receive dividends or any other rights as a shareholder shall exist with respect to the Shares, notwithstanding the exercise of the Option. The Company shall issue (or cause to be issued) such Shares promptly after the Option is exercised. No adjustment will be made for a dividend or other right for which the record date is prior to the date the Shares are issued, except as provided in Section 13 of the Plan.

Exercise of an Option in any manner shall result in a decrease in the number of Shares thereafter available, both for purposes of the Plan and for sale under the Option, by the number of Shares as to which the Option is exercised.

- (b) <u>Termination of Relationship as a Service Provider</u>. If an Optionee ceases to be a Service Provider, such Optionee may exercise his or her Option within such period of time as is specified in the Option Agreement to the extent that the Option is vested on the date of termination (but in no event later than the expiration of the term of the Option as set forth in the Option Agreement). In the absence of a specified time in the Option Agreement, the Option shall remain exercisable for three (3) months following the Optionee's termination. If, on the date of termination, the Optionee is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option shall revert to the Plan. If, after termination, the Optionee does not exercise his or her Option within the time specified by the Administrator, the Option shall terminate, and the Shares covered by such Option shall revert to the Plan.
- (c) <u>Disability of Optionee</u>. If an Optionee ceases to be a Service Provider as a result of the Optionee's Disability, the Optionee may exercise his or her Option within such period of time as is specified in the Option Agreement to the extent the Option is vested on the date of termination (but in no event later than the expiration of the term of such Option as set forth in the Option Agreement). In the absence of a specified time in the Option Agreement, the Optione hall remain exercisable for twelve (12) months following the Optionee's termination. If, on the date of termination, the Optionee is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option shall revert to the Plan. If, after termination, the Optionee does not exercise his or her Option within the time specified herein, the Option shall terminate, and the Shares covered by such Option shall revert to the Plan.
- (d) <u>Death of Optionee</u>. If an Optionee dies while a Service Provider, the Option may be exercised within such period of time as is specified in the Option Agreement to the extent that the Option is vested on the date of death (but in no event later than the expiration of the term of such Option as set forth in the Option Agreement) by the Optionee's setate or by a person who acquires the right to exercise the Option by bequest or inheritance. In the absence of a specified time in the Option Agreement, the Option shall remain exercisable for twelve (12) months following the Optionee's termination. If, at the time of death, the Optionee is not vested as to the entire Option, the Shares covered by the unvested portion of the Option shall immediately revert to the Plan. If the Option is not so exercised within the time specified herein, the Option shall terminate, and the Shares covered by such Option shall revert to the Plan.

- (e) <u>Buyout Provisions</u>. The Administrator may at any time offer to buy out for a payment in cash or Shares, an Option previously granted, based on such terms and conditions as the Administrator shall establish and communicate to the Optionee at the time that such offer is made.
- 10. Limited Transferability of Awards. Unless determined otherwise by the Administrator, Awards may not be sold, pledged, assigned, hypothecated, transferred, or disposed of in any manner other than by will or the laws of descent and distribution, and may be exercised during the lifetime of the Optionee, only by the Optionee. If the Administrator makes an Award transferable, such Award shall contain such additional terms and conditions as the Administrator deems appropriate.

11 Stock Purchase Rights

- (a) <u>Rights to Purchase</u>. Stock Purchase Rights may be issued either alone, in addition to, or in tandem with other Awards granted under the Plan and/or cash awards made outside of the Plan. After the Administrator determines that it will offer Stock Purchase Rights under the Plan, it shall advise the offeree in writing or electronically of the terms, conditions and restrictions related to the offer, including the number of Shares that such person shall be entitled to purchase, the price to be paid (if any), and the time within which such person must accept such offer. The offer shall be accepted by execution of an Award Agreement in the form determined by the Administrator.
- (b) <u>Repurchase Option</u>. Unless the Administrator determines otherwise, the Award Agreement shall grant the Company a repurchase option exercisable upon the voluntary or involuntary termination of the purchaser's service with the Company for any reason (including death or disability). Unless the Administrator provides otherwise, the purchase price for Shares repurchased pursuant to the Award Agreement shall be the original price paid by the purchaser and may be paid by cancellation of any indebtedness of the purchaser to the Company. The repurchase option shall lapse at such rate as the Administrator may determine.
 - (c) Other Provisions. The Award Agreement shall contain such other terms, provisions and conditions not inconsistent with the Plan as may be determined by the Administrator in its sole discretion.
- (d) <u>Rights as a Shareholder</u>. Once the Stock Purchase Right is exercised, the Optionee shall have rights equivalent to those of a shareholder and shall be a shareholder when his or her purchase is entered upon the records of the duly authorized transfer agent of the Company. No adjustment shall be made for a dividend or other right for which the record date is prior to the date the Stock Purchase Right is exercised, except as provided in Section 13 of the Plan.

Restricted Stock Units

(a) <u>Grant</u>. Restricted Stock Units may be granted at any time and from time to time as determined by the Administrator. Each Restricted Stock Unit grant will be evidenced by an Award Agreement that will specify such other terms and conditions as the Administrator, in its sole discretion, will determine, including all terms, conditions, and restrictions related to the grant, the number of Restricted Stock Units and the form of payout, which, subject to Section 12(d), may be left to the discretion of the Administrator.

- (b) <u>Vesting Criteria and Other Terms</u>. The Administrator will set vesting criteria in its discretion, which, depending on the extent to which the criteria are met, will determine the number of Restricted Stock Units that will be paid out to the Optionee. After the grant of Restricted Stock Units, the Administrator, in its sole discretion, may reduce or waive any restrictions for such Restricted Stock Units. Each Award of Restricted Stock Units will be evidenced by an Award Agreement that will specify the vesting criteria, and such other terms and conditions as the Administrator, in its sole discretion, will determine.
- (c) <u>Earning Restricted Stock Units</u>. Upon meeting the applicable vesting criteria, the Optionee will be entitled to receive a payout as specified in the Award Agreement. Notwithstanding the foregoing, at any time after the grant of Restricted Stock Units, the Administrator, in its sole discretion, may reduce or waive any vesting criteria that must be met to receive a payout.
- (d) Form and Timing of Payment. Payment of earned Restricted Stock Units will be made as soon as practicable after the date(s) set forth in the Award Agreement. The Administrator, in its sole discretion, may pay earned Restricted Stock Units in cash, Shares, or a combination thereof. Shares represented by Restricted Stock Units that are fully paid in cash again will be available for grant under the Plan.
 - (e) Cancellation. On the date set forth in the Award Agreement, all unearned Restricted Stock Units will be forfeited to the Company.
 - 13. Adjustments Upon Changes in Capitalization, Merger or Asset Sale.
- (a) <u>Changes in Capitalization</u>. Subject to any required action by the shareholders of the Company, the number of shares of Common Stock covered by each outstanding Option and Stock Purchase Right or issuable pursuant to a Restricted Stock Unit, and the number of shares of Common Stock which have been authorized for issuance under the Plan but as to which no Awards have yet been granted or which have been returned to the Plan upon cancellation or expiration of an Award, as well as the price per share of Common Stock covered by each such outstanding Award, shall be proportionately adjusted for any increase or decrease in the number of issued shares of Common Stock resulting from a stock split, reverse stock split, stock dividend, combination or reclassification of the Common Stock, any other increase or decrease in the number of issued shares of Common Stock effected without receipt of consideration by the Company or other change in the Common Stock as the result of a spin-off, extraordinary dividend or similar transaction; provided, however, that conversion of any convertible securities of the Company shall not be deemed to have been "effected without receipt of consideration." Such adjustment shall be made by the Board, whose determination in that respect shall be final, binding and conclusive. Except as expressly provided herein, no issuance by the Company of shares of stock of any class, or securities convertible into shares of stock of any class, shall affect, and no adjustment by reason thereof shall be made with respect to, the number or price of shares of Common Stock subject to an Award.
- (b) <u>Dissolution or Liquidation</u>. In the event of the proposed dissolution or liquidation of the Company, the Administrator shall notify each Optionee as soon as practicable prior to the effective date of such proposed transaction. The Administrator in its discretion may provide for an Optionee to have the right to exercise his or her Award until fifteen (15) days prior to

such transaction as to all of the Optioned Stock covered thereby, including Shares as to which the Award would not otherwise be exercisable. In addition, the Administrator may provide that any Company repurchase option or reacquisition right applicable to any Shares purchased upon exercise of an Award shall lapse as to all such Shares, provided the proposed dissolution or liquidation takes place at the time and in the manner contemplated. To the extent it has not been previously exercised, an Award will terminate immediately prior to the consummation of such proposed action.

- (c) Merger or Asset Sale. In the event of a merger of the Company with or into another corporation, or the sale of substantially all of the assets of the Company, each outstanding Award shall be assumed or an equivalent award substituted by the successor corporation or a Parent or Subsidiary of the successor corporation. In the event that the successor corporation refuses to assume or substitute for the Award, the Participant will fully vest in and have the right to exercise all of his or her outstanding Options and Stock Purchase Rights, including Shares as to which such Awards would not otherwise be vested or exercisable, all restrictions or corporation refuses to assume or substitute an Option or Stock Units, all performance goals or other vesting criteria will be deemed achieved at target levels and all other terms and conditions met. If the successor corporation refuses to assume or substitute an Option or Stock Purchase Right in the event of a merger or sale of assets, the Administrator shall notify the Participant in writing or electronically that the Award shall be fully vested and exercisable for a period of fifteen (15) days from the date of such notice, and the Award shall terminate upon the expiration of such period. For the purposes of this paragraph, the Award shall be considered assumed if, following the merger or sale of assets, the award confers the right to purchase or receive, for each Share of Optioned Stock immediately prior to the merger or sale of assets, the consideration (whether stock, cash, or other securities or property) received in the merger or sale of assets by holders of Common Stock for each Share held on the effective date of the transaction (and if holders were offered a choice of consideration, the type of consideration chosen by the holders of a majority of the outstanding Shares); provided, however, that if such consideration received upon the exercise of the Award, for each Share of Optioned Stock, to be solely common stock of the successor corporation or its Parent, the
- 14. Time of Granting Awards. The date of grant of an Award shall, for all purposes, be the date on which the Administrator makes the determination granting such Award, or such other date as is determined by the Administrator. Notice of the determination shall be given to each Service Provider to whom an Award is so granted within a reasonable time after the date of such grant.
 - 15. Amendment and Termination of the Plan
 - (a) Amendment and Termination. The Board may at any time amend, alter, suspend or terminate the Plan.
 - (b) Shareholder Approval. The Board shall obtain shareholder approval of any Plan amendment to the extent necessary and desirable to comply with Applicable Laws.

(c) Effect of Amendment or Termination. No amendment, alteration, suspension or termination of the Plan shall impair the rights of any Optionee, unless mutually agreed otherwise between the Optionee and the Administrator, which agreement must be in writing and signed by the Optionee and the Company. Termination of the Plan shall not affect the Administrator's ability to exercise the powers granted to it hereunder with respect to Awards granted under the Plan prior to the date of such termination.

16. Conditions Upon Issuance of Shares.

- (a) <u>Legal Compliance</u>. Shares shall not be issued pursuant an Award unless the exercise of such Award, if applicable, and the issuance and delivery of such Shares shall comply with Applicable Laws and shall be further subject to the approval of counsel for the Company with respect to such compliance.
- (b) Investment Representations. As a condition to the exercise of an Award or the issuance of Shares thereunder, the Administrator may require the person exercising such Award or receiving Shares thereunder to represent and warrant at the time of any such exercise or issuance that the Shares are being purchased or received only for investment and without any present intention to sell or distribute such Shares if, in the opinion of counsel for the Company, such a representation is required.
- 17. <u>Inability to Obtain Authority</u>. The inability of the Company to obtain authority from any regulatory body having jurisdiction, which authority is deemed by the Company's counsel to be necessary to the lawful issuance and sale of any Shares hereunder, shall relieve the Company of any liability in respect of the failure to issue or sell such Shares as to which such requisite authority shall not have been obtained.
 - 18. Reservation of Shares. The Company, during the term of this Plan, shall at all times reserve and keep available such number of Shares as shall be sufficient to satisfy the requirements of the Plan.
- 19. Shareholder Approval. The Plan shall be subject to approval by the shareholders of the Company within twelve (12) months after the date the Plan is adopted. Such shareholder approval shall be obtained in the degree and manner required under Applicable Laws.

APPENDIX A

TO

COMSCORE NETWORKS, INC. 1999 STOCK PLAN, AS AMENDED

(for California residents only)

This Appendix A to the comScore Networks, Inc. 1999 Stock Plan, as amended, shall apply only to Optionees who are residents of the State of California and who are receiving an Option or Stock Purchase Right under the Plan. Capitalized terms contained herein shall have the same meanings given to them in the Plan unless otherwise provided by this Appendix A. Notwithstanding any provisions contained in the Plan to the contrary and to the extent required by Applicable Laws, the following terms shall apply to all Options and Stock Purchase Rights granted to residents of the State of California, until such time as the Administrator amends this Appendix A or the Administrator otherwise provides.

Nonstatutory Stock Options granted to a person who, at the time of grant of such Option, owns stock representing more than ten percent (10%) of the voting power of all classes of stock of the Company or any Parent or Subsidiary, shall have an exercise price not less than 110% of the Fair Market Value per Share on the date of grant. Nonstatutory Stock Options granted to any other person shall have an exercise price that is not less than 100% of the Fair Market Value per Share on the date of grant. Notwithstanding the foregoing, Options may be granted with a per Share exercise price other than as required above pursuant to a merger or other corporate transaction.

- (a) The term of each Option shall be stated in the Option Agreement, provided, however, that the term shall be no more than ten (10) years from the date of grant thereof. The term of each Restricted Stock Purchase Agreement shall be no more than ten (10) years from the date the agreement is entered into.
- (b) Unless determined otherwise by the Administrator, Options or Stock Purchase Rights may not be sold, pledged, assigned, hypothecated, transferred, or disposed of in any manner other than by will or the laws of descent and distribution, and may be exercised during the lifetime of the Optionee, only by the Optionee. If the Administrator in its sole discretion makes an Option or Stock Purchase Right transferable, such Option or Stock Purchase Right may only be transferred (i) by will, (ii) by the laws of descent and distribution, or (iii) as permitted by Rule 701 of the Securities Act of 1933, as amended.
- (c) Except in the case of Options granted to officers, Directors and Consultants, Options shall become exercisable at a rate of no less than twenty percent (20%) per year over five (5) years from the date the Options are granted.
- (d) If an Optionee ceases to be a Service Provider, such Optionee may exercise his or her Option within thirty (30) days of termination, or such longer period of time as specified in the Option Agreement, to the extent that the Option is vested on the date of termination (but in no event later than the expiration of the term of the Option as set forth in the Option Agreement).

- (e) If an Optionee ceases to be a Service Provider as a result of the Optionee's Disability, Optionee may exercise his or her Option within six (6) months of termination, or such longer period of time as specified in the Option Agreement, to the extent the Option is vested on the date of termination (but in no event later than the expiration of the term of such Option as set forth in the Option Agreement).
- (f) If an Optionee dies while a Service Provider, the Option may be exercised within six (6) months following the Optionee's death, or such longer period of time as specified in the Option Agreement, to the extent the Option is vested on the date of termination (but in no event later than the expiration of the term of such Option as set forth in the Option Agreement) by the Optionee's designated beneficiary, personal representative, or by the person(s) to whom the Option is transferred pursuant to the Optionee's will or in accordance with the laws of descent and distribution.
- (g) The terms of any Stock Purchase Rights offered under this Appendix A shall comply in all respects with Section 260.140.42 of Title 10 of the California Code of Regulations including, without limitation:
- (i) that except with respect to Shares purchased by officers, Directors and Consultants, the repurchase option shall in no case lapse at a rate of less than twenty percent (20%) per year over five (5) years from the date of purchase; and
- (ii) Stock Purchase Rights granted to a person who, at the time of grant of such Stock Purchase Right, owns stock representing more than ten percent (10%) of the voting power of all classes of stock of the Company or any Parent or Subsidiary, shall have a purchase price not less than 100% of the Fair Market Value per Share on the date of grant or on the date of purchase. Stock Purchase Rights granted to any other person shall have a purchase price that is not less than 100% of the Fair Market Value per Share on the date of grant or on the date of purchase.
 - (h) No Option or Stock Purchase Right shall be granted to a resident of California more than ten (10) years after the earlier of the date of adoption of the Plan or the date the Plan is approved by the stockholders.
- (i) The Company shall provide to each Optionee and to each individual who acquires Shares pursuant to the Plan, not less frequently than annually during the period such Optionee has one or more Options or Stock Purchase Rights outstanding, and, in the case of an individual who acquires Shares pursuant to the Plan, during the period such individual owns such Shares, copies of annual financial statements. The Company shall not be required to provide such statements to key Employees whose duties in connection with the Company assure their access to equivalent information.
- (j) In the event that any dividend or other distribution (whether in the form of cash, Shares, other securities, or other property), recapitalization, stock split, reverse stock

adjust the number and class of shares of common stock that may be delivered under the Plan and/or the number, class, and price of shares covered by each outstanding Option. The Administrator shall also make such adjustments to the extent required by Section 25102(o) of the California Corporations Code.

(k) This Appendix A shall be deemed to be part of the Plan and the Administrator shall have the authority to amend this Appendix A in accordance with Section 15 of the Plan.

COMSCORE, INC. 1999 STOCK PLAN

STOCK OPTION AGREEMENT

Unless otherwise defined herein, the terms defined in the 1999 Stock Plan shall have the same defined meanings in this Stock Option Agreement.

I. NOTICE OF STOCK OPTION GRANT

[Optionee's Name and Address]

The undersigned Optionee has been granted an Option to purchase Common Stock of the Company, subject to the terms and conditions of the Plan and this Option Agreement, as follows:

Grant Number	
Date of Grant	
Vesting Commencement Date	
Exercise Price per Share	\$
Total Number of Shares Granted	
Total Exercise Price	\$
Type of Option:	o Incentive Stock Option
	o Nonstatutory Stock Option
Term/Expiration Date:	
<u>Vesting Schedule</u> :	
This Option shall be exercisable, in whole or in part, according to the following vesting	schedule:
[25% of the Shares subject to the Option shall vest twelve months after the Vesting	Commencement Date, and 1/48 of the Shares subject to the Option shall vest each month thereafter, subject to

Optionee's continuing to be a Service Provider on such dates.]

Termination Period:

This Option shall be exercisable for [three months] after Optionee ceases to be a Service Provider. Upon Optionee's death or Disability, this Option may be exercised for [one year] after Optionee ceases to be a Service Provider. In no event may Optionee exercise this Option after the Term/Expiration Date as provided above.

II. AGREEMENT

1. <u>Grant of Option</u>. The Plan Administrator of the Company hereby grants to the Optionee named in the Notice of Grant (the "Optionee"), an option (the "Option") to purchase the number of Shares set forth in the Notice of Grant, at the exercise price per Share set forth in the Notice of Grant (the "Exercise Price"), and subject to the terms and conditions of the Plan, which is incorporated herein by reference. Subject to Section 14(c) of the Plan, in the event of a conflict between the terms and conditions of the Plan and this Option Agreement, the terms and conditions of the Plan shall prevail.

If designated in the Notice of Grant as an Incentive Stock Option ("ISO"), this Option is intended to qualify as an Incentive Stock Option as defined in Section 422 of the Code. Nevertheless, to the extent that it exceeds the \$100,000 rule of Code Section 422(d), this Option shall be treated as a Nonstatutory Stock Option ("NSO").

2. Exercise of Option

- (a) Right to Exercise. This Option shall be exercisable during its term in accordance with the Vesting Schedule set out in the Notice of Grant and with the applicable provisions of the Plan and this Option Agreement.
- (b) <u>Method of Exercise</u>. This Option shall be exercisable by delivery of an exercise notice in the form attached as <u>Exhibit A</u> (the "Exercise Notice") which shall state the election to exercise the Option, the number of Shares with respect to which the Option is being exercised, and such other representations and agreements as may be required by the Company. The Exercise Notice shall be accompanied by payment of the aggregate Exercise Price as to all Exercise Shares. This Option shall be deemed to be exercised upon receipt by the Company of such fully executed Exercise Notice accompanied by the aggregate Exercise Price.

No Shares shall be issued pursuant to the exercise of an Option unless such issuance and such exercise comply with Applicable Laws. Assuming such compliance, for income tax purposes the Shares shall be considered transferred to the Optione on the date on which the Option is exercised with respect to such Shares.

3. <u>Optionee's Representations</u>. In the event the Shares have not been registered under the Securities Act of 1933, as amended, at the time this Option is exercised, the Optionee shall, if required by the Company, concurrently with the exercise of all or any portion of this Option, deliver to the Company his or her Investment Representation Statement in the form attached hereto as <u>Exhibit B</u>.

- 4. <u>Lock-Up Period</u>. Optionee hereby agrees that, if so requested by the Company or any representative of the underwriters (the "Managing Underwriter") in connection with any registration of the offering of any securities of the Company under the Securities Act, Optionee shall not sell or otherwise transfer any Shares or other securities of the Company during the 180-day period (or such other period as may be requested in writing by the Managing Underwriter and agreed to in writing by the Company) (the "Market Standoff Period") following the effective date of a registration statement of the Company filed under the Securities Act. Such restriction shall apply only to the first registration statement of the Company to become effective under the Securities Act that includes securities to be sold on behalf of the Company to the public in an underwritten public offering under the Securities Act. The Company may impose stop-transfer instructions with respect to securities subject to the foregoing restrictions until the end of such Market Standoff Period.
 - 5. Method of Payment. Payment of the aggregate Exercise Price shall be by any of the following, or a combination thereof, at the election of the Optionee:
 - (a) cash or check:
 - (b) consideration received by the Company under a formal cashless exercise program adopted by the Company in connection with the Plan; or
- (c) surrender of other Shares which, (i) in the case of Shares acquired upon exercise of an option, have been owned by the Optionee for more than six (6) months on the date of surrender, and (ii) have a Fair Market Value on the date of surrender equal to the aggregate Exercise Price of the Exercised Shares.
- 6. <u>Restrictions on Exercise</u>. This Option may not be exercised until such time as the Plan has been approved by the shareholders of the Company, or if the issuance of such Shares upon such exercise or the method of payment of consideration for such shares would constitute a violation of any Applicable Law.
- 7. Non-Transferability of Option. This Option may not be transferred in any manner otherwise than by will or by the laws of descent or distribution and may be exercised during the lifetime of Optionee only by Optionee. The terms of the Plan and this Option Agreement shall be binding upon the executors, administrators, heirs, successors and assigns of the Optionee.
 - 8. Term of Option. This Option may be exercised only within the term set out in the Notice of Grant, and may be exercised during such term only in accordance with the Plan and the terms of this Option.
- 9. <u>Tax Consequences</u>. Set forth below is a brief summary as of the date of this Option of some of the federal tax consequences of exercise of this Option and disposition of the Shares. THIS SUMMARY IS NECESSARILY INCOMPLETE, AND THE TAX LAWS AND REGULATIONS ARE SUBJECT TO CHANGE. THE OPTIONEE SHOULD CONSULT A TAX ADVISER BEFORE EXERCISING THIS OPTION OR DISPOSING OF THE SHARES.

- (a) Exercise of NSO. There may be a regular federal income tax liability upon the exercise of an NSO. The Optionee will be treated as having received compensation income (taxable at ordinary income tax rates) equal to the excess, if any, of the Fair Market Value of the Shares on the date of exercise over the Exercise Price. If Optionee is an Employee or a former Employee, the Company will be required to withhold from Optionee's compensation or collect from Optionee and pay to the applicable taxing authorities an amount in cash equal to a percentage of this compensation income at the time of exercise, and may refuse to honor the exercise and refuse to deliver Shares if such withholding amounts are not delivered at the time of exercise.
- (b) Exercise of ISO. If this Option qualifies as an ISO, there will be no regular federal income tax liability upon the exercise of the Option, although the excess, if any, of the Fair Market Value of the Shares on the date of exercise over the Exercise Price will be treated as an adjustment to the alternative minimum tax for federal tax purposes and may subject the Optionee to the alternative minimum tax in the year of exercise.
- (c) <u>Disposition of Shares</u>. In the case of an NSO, if Shares are held for at least one year, any gain realized on disposition of the Shares will be treated as long-term capital gain for federal income tax purposes. In the case of an ISO, if Shares transferred pursuant to the Option are held for at least one year after exercise and of at least two years after the Date of Grant, any gain realized on disposition of the Shares will also be treated as long-term capital gain for federal income tax purposes. If Shares purchased under an ISO are disposed of within one year after exercise or two years after the Date of Grant, any gain realized on such disposition will be treated as compensation income (taxable at ordinary income rates) to the extent of the difference between the Exercise Price and the lesser of (1) the Fair Market Value of the Shares on the date of exercise, or (2) the sale price of the Shares. Any additional gain will be taxed as capital gain, short-term or long-term depending on the period that the ISO Shares were held.
- (d) Notice of Disqualifying Disposition of ISO Shares. If the Option granted to Optionee herein is an ISO, and if Optionee sells or otherwise disposes of any of the Shares acquired pursuant to the ISO on or before the later of (1) the date two years after the Date of Grant, or (2) the date one year after the date of exercise, the Optionee shall immediately notify the Company in writing of such disposition. Optionee agrees that Optionee may be subject to income tax withholding by the Company on the compensation income recognized by the Optionee.
- 10. Entire Agreement; Governing Law. The Plan is incorporated herein by reference. The Plan and this Option Agreement constitute the entire agreement of the parties with respect to the subject matter hereof and supersede in their entirety all prior undertakings and agreements of the Company and Optionee with respect to the subject matter hereof, and may not be modified adversely to the Optionee's interest except by means of a writing signed by the Company and Optionee. The internal substantive laws but not the choice of law rules of Virginia govern this agreement.
- 11. No Guarantee of Continued Service. OPTIONEE ACKNOWLEDGES AND AGREES THAT THE VESTING OF SHARES PURSUANT TO THE VESTING SCHEDULE HEREOF IS EARNED ONLY BY CONTINUING AS A SERVICE PROVIDER AT THE WILL OF THE COMPANY (NOT THROUGH THE ACT OF BEING HIRED, BEING GRANTED THIS

OPTION OR ACQUIRING SHARES HEREUNDER). OPTIONEE FURTHER ACKNOWLEDGES AND AGREES THAT THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREUNDER AND THE VESTING SCHEDULE SET FORTH HEREIN DO NOT CONSTITUTE AN EXPRESS OR IMPLIED PROMISE OF CONTINUED ENGAGEMENT AS A SERVICE PROVIDER FOR THE VESTING PERIOD, FOR ANY PERIOD, OR AT ALL, AND SHALL NOT INTERFERE IN ANY WAY WITH OPTIONEE'S RIGHT OR THE COMPANY'S RIGHT TO TERMINATE OPTIONEE'S RELATIONSHIP AS A SERVICE PROVIDER AT ANY TIME, WITH OR WITHOUT CAUSE.

Optionee acknowledges receipt of a copy of the Plan and represents that he or she is familiar with the terms and provisions thereof, and hereby accepts this Option subject to all of the terms and provisions thereof. Optionee has reviewed the Plan and this Option in their entirety, has had an opportunity to obtain the advice of counsel prior to executing this Option and fully understands all provisions of the Option. Optionee hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions arising under the Plan or this Option. Optionee further agrees to notify the Company upon any change in the residence address indicated below.

OPTIONEE	COMSCORE, INC.
Signature	Ву
Print Name	Title
Residence Address	

EXHIBIT A 1999 STOCK PLAN EXERCISE NOTICE

comScore, inc. 1761 Business Center Drive Reston, VA 20190

 Exercise of Option. Effective as of today, 	, 19, the undersigned ("Optionee") hereby elects to exercise Optic	onee's option to purchasesh	ares of the Common Stock (the "Shares") of
comScore, inc. (the "Company") under and pursuant to	the 1999 Stock Plan (the "Plan") and the Stock Option Agreement dated	, 19 (the "Option Agreement	").

- 2. Delivery of Payment. Purchaser herewith delivers to the Company the full purchase price of the Shares, as set forth in the Option Agreement.
- 3. Representations of Optionee. Optionee acknowledges that Optionee has received, read and understood the Plan and the Option Agreement and agrees to abide by and be bound by their terms and conditions.
- 4. <u>Rights as Shareholder</u>. Until the issuance of the Shares (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company), no right to vote or receive dividends or any other rights as a shareholder shall exist with respect to the Optioned Stock, notwithstanding the exercise of the Option. The Shares shall be issued to the Optionee as soon as practicable after the Option is exercised. No adjustment shall be made for a dividend or other right for which the record date is prior to the date of issuance except as provided in Section 12 of the Plan.
- 5. <u>Company's Right of First Refusal</u>. Before any Shares held by Optionee or any transferee (either being sometimes referred to herein as the "Holder") may be sold or otherwise transferred (including transfer by gift or operation of law), the Company or its assignee(s) shall have a right of first refusal to purchase the Shares on the terms and conditions set forth in this Section (the "Right of First Refusal").
- (a) Notice of Proposed Transfer. The Holder of the Shares shall deliver to the Company a written notice (the "Notice") stating: (i) the Holder's bona fide intention to sell or otherwise transfer such Shares; (ii) the name of each proposed purchaser or other transferee ("Proposed Transferee"); (iii) the number of Shares to be transferred to each Proposed Transferee; and (iv) the bona fide cash price or other consideration for which the Holder proposes to transfer the

Shares (the "Offered Price"), and the Holder shall offer the Shares at the Offered Price to the Company or its assignee(s).

- (b) Exercise of Right of First Refusal. At any time within thirty (30) days after receipt of the Notice, the Company and/or its assignee(s) may, by giving written notice to the Holder, elect to purchase all, but not less than all, of the Shares proposed to be transferred to any one or more of the Proposed Transferees, at the purchase price determined in accordance with subsection (c) below.
- (c) <u>Purchase Price</u>. The purchase price ("Purchase Price") for the Shares purchased by the Company or its assignee(s) under this Section shall be the Offered Price. If the Offered Price includes consideration other than cash, the Board of Directors of the Company in good faith shall determine the cash equivalent value of the non-cash consideration.
- (d) <u>Payment</u>. Payment of the Purchase Price shall be made, at the option of the Company or its assignee(s), in cash (by check), by cancellation of all or a portion of any outstanding indebtedness of the Holder to the Company (or, in the case of repurchase by an assignee, to the assignee), or by any combination thereof within thirty (30) days after receipt of the Notice or in the manner and at the times set forth in the Notice.
- (e) <u>Holder's Right to Transfer</u>. If all of the Shares proposed in the Notice to be transferred to a given Proposed Transferee are not purchased by the Company and/or its assignee(s) as provided in this Section, then the Holder may sell or otherwise transfer such Shares to that Proposed Transferee at the Offered Price or at a higher price, provided that such sale or other transfer is consummated within 120 days after the date of the Notice, that any such sale or other transfer is effected in accordance with any applicable securities laws and that the Proposed Transferee agrees in writing that the provisions of this Section shall continue to apply to the Shares in the hands of such Proposed Transferee. If the Shares described in the Notice are not transferred to the Proposed Transferee, a new Notice shall be given to the Company, and the Company and/or its assignees shall again be offered the Right of First Refusal before any Shares held by the Holder may be sold or otherwise transferred.
- (f) Exception for Certain Family Transfers. Anything to the contrary contained in this Section notwithstanding, the transfer of any or all of the Shares during the Optionee's lifetime or on the Optionee's death by will or intestacy to the Optionee's immediate family or a trust for the benefit of the Optionee's immediate family shall be exempt from the provisions of this Section. "Immediate Family" as used herein shall mean spouse, lineal descendant or antecedent, father, mother, brother or sister. In such case, the transferee or other recipient shall receive and hold the Shares so transferred subject to the provisions of this Section, and there shall be no further transfer of such Shares except in accordance with the terms of this Section.
- (g) <u>Termination of Right of First Refusal</u>. The Right of First Refusal shall terminate as to any Shares upon the first sale of Common Stock of the Company to the general public pursuant to a registration statement filed with and declared effective by the Securities and Exchange Commission under the Securities Act of 1933, as amended.

6. <u>Tax Consultation</u>. Optionee understands that Optionee may suffer adverse tax consequences as a result of Optionee's purchase or disposition of the Shares. Optionee represents that Optionee has consulted with any tax consultants Optionee deems advisable in connection with the purchase or disposition of the Shares and that Optionee is not relying on the Company for any tax advice.

Restrictive Legends and Stop-Transfer Orders

(a) <u>Legends</u>. Optionee understands and agrees that the Company shall cause the legends set forth below or legends substantially equivalent thereto, to be placed upon any certificate(s) evidencing ownership of the Shares together with any other legends that may be required by the Company or by state or federal securities laws:

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 (THE "ACT") AND MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, PLEDGED OR HYPOTHECATED UNLESS AND UNTIL REGISTERED UNDER THE ACT OR, IN THE OPINION OF COMPANY COUNSEL SATISFACTORY TO THE ISSUER OF THESE SECURITIES, SUCH OFFER, SALE OR TRANSFER, PLEDGE OR HYPOTHECATION IS IN COMPLIANCE THEREWITH.

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN RESTRICTIONS ON TRANSFER AND A RIGHT OF FIRST REFUSAL HELD BY THE ISSUER OR ITS ASSIGNEE(S) AS SET FORTH IN THE EXERCISE NOTICE BETWEEN THE ISSUER AND THE ORIGINAL HOLDER OF THESE SHARES, A COPY OF WHICH MAY BE OBTAINED AT THE PRINCIPAL OFFICE OF THE ISSUER. SUCH TRANSFER RESTRICTIONS AND RIGHT OF FIRST REFUSAL ARE BINDING ON TRANSFEREES OF THESE SHARES.

- (b) Stop-Transfer Notices. Optionee agrees that, in order to ensure compliance with the restrictions referred to herein, the Company may issue appropriate "stop transfer" instructions to its transfer agent, if any, and that, if the Company transfers its own securities, it may make appropriate notations to the same effect in its own records.
- (c) <u>Refusal to Transfer</u>. The Company shall not be required (i) to transfer on its books any Shares that have been sold or otherwise transferred in violation of any of the provisions of this Exercise Notice or (ii) to treat as owner of such Shares or to accord the right to vote or pay dividends to any purchaser or other transferee to whom such Shares shall have been so transferred.
- 8. <u>Successors and Assigns</u>. The Company may assign any of its rights under this Exercise Notice to single or multiple assignees, and this Exercise Notice shall inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer herein set forth,

this Exercise Notice shall be binding upon Optionee and his or her heirs, executors, administrators, successors and assigns.

- 9. Interpretation. Any dispute regarding the interpretation of this Exercise Notice shall be submitted by Optionee or by the Company forthwith to the Administrator which shall review such dispute at its next regular meeting. The resolution of such a dispute by the Administrator shall be final and binding on all parties.
 - 10. Governing Law; Severability. This Exercise Notice is governed by the internal substantive laws but not the choice of law rules, of Virginia.
- 11. Entire Agreement. The Plan and Option Agreement are incorporated herein by reference. This Exercise Notice, the Plan, the Option Agreement and the Investment Representation Statement constitute the entire agreement of the parties with respect to the subject matter hereof and supersede in their entirety all prior undertakings and agreements of the Company and Optionee with respect to the subject matter hereof, and may not be modified adversely to the Optionee's interest except by means of a writing signed by the Company and Optionee.

Submitted by:	Accepted by:
OPTIONEE	COMSCORE, INC.
Signature	Ву
Print Name	Title
Address:	Address:
	1761 Business Center Drive, Suite 250
	Reston, VA 20190
	Date Received
	_/

EXHIBIT B

INVESTMENT REPRESENTATION STATEMENT

OPTIONEE:

COMPANY: COMSCORE, INC. SECURITY: COMMON STOCK

AMOUNT: DATE:

In connection with the purchase of the above-listed Securities, the undersigned Optionee represents to the Company the following:

(a) Optionee is aware of the Company's business affairs and financial condition and has acquired sufficient information about the Company to reach an informed and knowledgeable decision to acquire the Securities. Optionee is acquiring these Securities for investment for Optionee's own account only and not with a view to, or for resale in connection with, any "distribution" thereof within the meaning of the Securities Act of 1933, as amended (the "Securities Act")

(b) Optionee acknowledges and understands that the Securities constitute "restricted securities" under the Securities Act and have not been registered under the Securities Act in reliance upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of Optionee's investment intent as expressed herein. In this connection, Optionee understands that, in the view of the Securities and Exchange Commission, the statutory basis for such exemption may be unavailable if Optionee's representation was predicated solely upon a present intention to hold these Securities for the minimum capital gains period specified under tax statutes, for a deferred sale, for or until an increase or decrease in the market price of the Securities, or for a period of one year or any other fixed period in the future. Optionee further understands that the Securities must be held indefinitely unless they are subsequently registered under the Securities Act or an exemption from such registration is available. Optionee further acknowledges and understands that the Company is under no obligation to register the Securities. Optionee understands that the certificate evidencing the Securities will be imprinted with a legend which prohibits the transfer of the Securities unless they are registered or such registration is not required in the opinion of counsel satisfactory to the Company, and any other legend required under applicable state securities laws.

(c) Optionee is familiar with the provisions of Rule 701 and Rule 144, each promulgated under the Securities Act, which, in substance, permit limited public resale of "restricted securities" acquired, directly or indirectly from the issuer thereof, in a non-public offering subject to the satisfaction of certain conditions. Rule 701 provides that if the issuer qualifies under Rule 701 at the

time of the grant of the Option to the Optionee, the exercise will be exempt from registration under the Securities Act. In the event the Company becomes subject to the reporting requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, ninety (90) days thereafter (or such longer period as any market stand-off agreement may require) the Securities exempt under Rule 701 may be resold, subject to the satisfaction of certain of the conditions specified by Rule 144, including: (1) the resale being made through a broker in an unsolicited "broker's transaction" or in transactions directly with a market maker (as said term is defined under the Securities Exchange Act of 1934); and, in the case of an affiliate, (2) the availability of certain public information about the Company, (3) the amount of Securities being sold during any three month period not exceeding the limitations specified in Rule 144(e), and (4) the timely filing of a Form 144, if applicable.

In the event that the Company does not qualify under Rule 701 at the time of grant of the Option, then the Securities may be resold in certain limited circumstances subject to the provisions of Rule 144, which requires the resale to occur not less than one year after the later of the date the Securities were sold by the Company or the date the Securities were sold by an affiliate of the Company, within the meaning of Rule 144; and, in the case of acquisition of the Securities by an affiliate, or by a non-affiliate who subsequently holds the Securities less than two years, the satisfaction of the conditions set forth in sections (1), (2), (3) and (4) of the paragraph immediately above.

(d) Optionee further understands that in the event all of the applicable requirements of Rule 701 or 144 are not satisfied, registration under the Securities Act, compliance with Regulation A, or some other registration exemption will be required; and that, notwithstanding the fact that Rules 144 and 701 are not exclusive, the Staff of the Securities and Exchange Commission has expressed its opinion that persons proposing to sell private placement securities other than in a registered offering and otherwise than pursuant to Rules 144 or 701 will have a substantial burden of proof in establishing that an exemption from registration is available for such offers or sales, and that such persons and their respective brokers who participate in such transactions do so at their own risk. Optionee understands that no assurances can be given that any such other registration exemption will be available in such event.

Signature of Optionee:		
Date:	, 19	

COMSCORE, INC.

1999 STOCK PLAN, AS AMENDED

NOTICE OF GRANT OF STOCK PURCHASE RIGHT

Unless otherwise defined herein, the terms defined in the 1999 Stock Plan, as amended (the "Plan") will have the same defined meanings in this Notice of Grant of Stock Purchase Right (the "Notice of Grant") and Terms and Conditions of Stock Purchase Right, attached hereto as Exhibit A (together, the "Agreement").

Optionee has been granted the right to receive a Stock Purchase Right (the "Ar	ward"), subject to the terms and conditions of the Plan and this Agreement, as follows:
Grant Number	
Date of Grant	
Vesting Commencement Date	
Number of Shares Granted	
[Exercise Price Per Share	\$]
Term/Expiration Date	
<u>Vesting Schedule</u> :	
Subject to any acceleration provisions contained in the Plan or set forth below sollowing schedule:	the Stock Purchase Right will vest and the Company's right to repurchase the Stock Purchase Right will lapse in accordance with

[INSERT VESTING SCHEDULE]

Optionee: Address:

[OPTIONEE MUST PURCHASE THE SHARES BEFORE THE EXPIRATION DATE OR THE STOCK PURCHASE RIGHT WILL TERMINATE AND OPTIONEE WILL HAVE NO FURTHER RIGHT TO PURCHASE THE SHARES.]

Agreement. Optionee has reviewed the Plan and this Agreement in their entire	e below, Optionee and the Company agree that this Award is granted under and governed by the terms and conditions of the Plan and this ety, has had an opportunity to obtain the advice of counsel prior to executing this Agreement and fully understands all provisions of the Plan final all decisions or interpretations of the Administrator upon any questions relating to the Plan and Agreement. Optionee further agrees to w.
OPTIONEE	COMSCORE, INC.
Signature	Ву
Print Name	Title
Address:	
	-2-

EXHIBIT A

TERMS AND CONDITIONS OF STOCK PURCHASE RIGHT GRANT

1. <u>Purchase of Stock</u>. The Company hereby agrees to sell to the Optionee named in the Notice of Grant (the "Optionee") and Optionee hereby agrees to purchase the number of Shares (the "Restricted Stock"), at the per Share purchase price and as otherwise described in the Notice of Grant, subject to all of the terms and conditions in this Agreement and the Plan, which is incorporated herein by reference. Subject to Section 14(c) of the Plan, in the event of a conflict between the terms and conditions of the Plan will prevail. The purchase price for the Restricted Stock, if any, may be paid by delivery to the Company at the time of execution of this Agreement in cash, a check, or some combination thereof, together with any applicable tax withholding.

OF

Grant of Restricted Stock. The Company hereby grants to the Optionee named in the Notice of Grant (the "Optionee") under the Plan for past services and as a separate incentive in connection with his or her services and not in lieu of any salary or other compensation for his or her services, Shares (the "Restricted Stock"), at the per Share purchase price and as otherwise described in the Notice of Grant, subject to all of the terms and conditions in this Agreement and the Plan, which is incorporated herein by reference. Subject to Section 14(c) of the Plan, in the event of a conflict between the terms and conditions of the Plan and the terms and conditions of the Plan will prevail.

Escrow of Shares

- (a) All Shares of Restricted Stock will, upon execution of this Agreement, be delivered and deposited with an escrow holder designated by the Company (the "Escrow Holder"). The Shares of Restricted Stock will be held by the Escrow Holder until such time as the Shares of Restricted Stock vest or the date Optionee ceases to be a Service Provider.
 - (b) The Escrow Holder will not be liable for any act it may do or omit to do with respect to holding the Shares of Restricted Stock in escrow while acting in good faith and in the exercise of its judgment.
- (c) Upon Optionee's termination as a Service Provider for any reason, the Escrow Holder, upon receipt of written notice of such termination, will take all steps necessary to accomplish the transfer of the unvested Shares of Restricted Stock to the Company. Optionee hereby appoints the Escrow Holder with full power of substitution, as Optionee's true and lawful attorney-in-fact with irrevocable power and authority in the name and on behalf of Optionee to take any action and execute all documents and instruments, including, without limitation, stock powers which may be necessary to transfer the certificate or certificates evidencing such unvested Shares of Restricted Stock to the Company upon such termination.

- (d) The Escrow Holder will take all steps necessary to accomplish the transfer of Shares of Restricted Stock to Optionee after they vest following Optionee's request that the Escrow Holder do so.
- (e) Subject to the terms hereof, Optionee will have all the rights of a stockholder with respect to the Shares while they are held in escrow, including without limitation, the right to vote the Shares and to receive any cash dividends declared thereon.
- (f) In the event of a stock split, reverse stock split, stock dividend, combination, or reclassification, or any other increase or decrease in the number of shares of Common Stock effected without receipt of consideration, the Shares of Restricted Stock will be increased, reduced or otherwise changed, and by virtue of any such change Optionee will in his or her capacity as owner of unvested Shares of Restricted Stock be entitled to new or additional or different shares, cash or securities will thereupon be considered to be unvested Shares of Restricted Stock and will be subject to all of the conditions and restrictions which were applicable to the unvested Shares of Restricted Stock pursuant to this Agreement. If Optionee receives rights or warrants with respect to any unvested Shares of Restricted Stock, such rights or warrants may be held or exercised by Optionee, provided that until such exercise any such rights or warrants and after such exercise any shares or other securities acquired by the exercise of such rights or warrants will be considered to be unvested Shares of Restricted Stock and will be subject to all of the conditions and restrictions which were applicable to the unvested Shares of Restricted Stock pursuant to this Agreement. The Administrator in its absolute discretion at any time may accelerate the vesting of all or any portion of such new or additional shares of stock, cash or securities, rights or warrants to purchase securities or shares or other securities acquired by the exercise of such rights or warrants.
- (g) The Company may instruct the transfer agent for its Common Stock to place a legend on the certificates representing the Restricted Stock or otherwise note its records as to the restrictions on transfer set forth in this Agreement
- 3. Optionee's Representations. In the event the Shares have not been registered under the Securities Act of 1933, as amended, at the time this Stock Purchase Right is exercised, the Optionee will, if required by the Company, concurrently with the exercise of all or any portion of this Stock Purchase Right, deliver to the Company his or her Investment Representation Statement in the form attached hereto as Exhibit B.
- 4. <u>Lock-Up Period</u>. Optionee hereby agrees that Optionee will not offer, pledge, sell, contract to sell, sell any option or contract to purchase any option or contract to sell, grant any option, right or warrant to purchase, lend, or otherwise transfer or dispose of, directly or indirectly, any Common Stock (or other securities) of the Company or enter into any swap, hedging or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any Common Stock (or other securities) of the Company held by Optionee (other than those included in the registration) for a period specified by the representative of the underwriters of Common Stock (or other securities) of the Company not to exceed one hundred eighty (180) days following the effective date of any registration

statement of the Company filed under the Securities Act (or such other period as may be requested by the Company or the underwriters to accommodate regulatory restrictions on (i) the publication or other distribution of research reports and (ii) analyst recommendations and opinions, including, but not limited to, the restrictions contained in NASD Rule 2711(f)(4) or NYSE Rule 472(f)(4), or any successor provisions or amendments thereto).

Optionee agrees to execute and deliver such other agreements as may be reasonably requested by the Company or the underwriter which are consistent with the foregoing or which are necessary to give further effect thereto. In addition, if requested by the Company or the representative of the underwriters of Common Stock (or other securities) of the Company, Optionee will provide, within ten (10) days of such request, such information as may be required by the Company or such representative in connection with the completion of any public offering of the Company's securities pursuant to a registration statement filed under the Securities Act. The obligations described in this Section will not apply to a registration relating solely to employee benefit plans on Form S-0 or Form S-0 or similar forms that may be promulgated in the future, or a registration solely to a Commission Rule 145 transaction on Form S-4 or similar forms that may be promulgated in the future. The Company may impose stop-transfer instructions with respect to the shares of Common Stock (or other securities) subject to the foregoing restriction until the end of said one hundred eighty (180) day (or other) period. Optionee agrees that any transferee of the Stock Purchase Right or Shares acquired pursuant to the Stock Purchase Right will be bound by this Section.

- 5. <u>Vesting Schedule</u>. Except as provided in Section 6, and subject to Section 7, the Shares of Restricted Stock awarded by this Agreement will vest in accordance with the vesting provisions set forth in the Notice of Grant. Shares of Restricted Stock scheduled to vest on a certain date or upon the occurrence of a certain condition will not vest in Optionee in accordance with any of the provisions of this Agreement, unless Optionee will have been continuously a Service Provider from the Date of Grant until the date such vesting occurs.
- 6. Administrator Discretion. The Administrator, in its discretion, may accelerate the vesting of the balance, or some lesser portion of the balance, of the unvested Restricted Stock at any time, subject to the terms of the Plan. If so accelerated, such Restricted Stock will be considered as having vested as of the date specified by the Administrator.
- 7. <u>Forfeiture upon Termination of Status as a Service Provider.</u> Notwithstanding any contrary provision of this Agreement, the balance of the Shares of Restricted Stock that have not vested at the time of Optionee's termination as a Service Provider for any reason will be forfeited and automatically transferred to and reacquired by the Company at the original purchase price paid by the Optionee for such Shares (if any) (any may be paid by cancellation of indebtedness of the purchaser to the Company) and Optionee will have no further rights thereunder. Optionee hereby appoints the Escrow Agent with full power of substitution, as Optionee's true and lawful attorney-in-fact with irrevocable power and authority in the name and on behalf of Optionee to take any action and execute all documents and instruments, including, without limitation, stock powers which may be necessary to transfer the certificate or certificates evidencing such unvested Shares to the Company upon such termination of service.

Repurchase Option.

In the event Optionee ceases to be a Service Provider for any or no reason (including death or disability) before all of the Shares are released from the Company's Repurchase Option (see Notice of Grant of Stock Purchase Right ("Notice of Grant")), the Company shall, upon the date of such termination (as reasonably fixed and determined by the Company) have an irrevocable, exclusive option (the "Repurchase Option") for a period of sixty (60) days from such date to repurchase all or a portion of the unvested Shares of Restricted Stock at the original purchase price per share (the "Repurchase Price"). The Repurchase Option shall be exercised by the Company by delivering written notice to Optionee's executor (with a copy to the Escrow Holder) AND, at the Company's option, (i) by delivering to Optionee or Optionee's executor a check in the amount of the aggregate Repurchase Price, or (iii) by cancelling an amount of Optionee's indebtedness to the Company equal to the aggregate Repurchase Price, or (iii) by a combination of (i) and (ii) so that the combined payment and cancellation of indebtedness equals the aggregate Repurchase Price, the Company shall become the legal and beneficial owner of the unvested Shares of Restricted Stock being repurchased and all rights and interests therein or relating thereto, and the Company shall have the right to retain and transfer to its own name the number of Shares being repurchased by the Company.

Whenever the Company shall have the right to repurchase Shares hereunder, the Company may designate and assign one or more employees, officers, directors or shareholders of the Company or other persons or organizations to exercise all or a part of the Company's purchase rights under this Agreement and purchase all or a part of such Shares. If the Fair Market Value of the Shares to be repurchased on the date of such designation or assignment (the "Repurchase FMV") exceeds the aggregate Repurchase Price of such Shares, then each such designee or assignee shall pay the Company cash equal to the difference between the Repurchase FMV and the aggregate Repurchase Price of such Shares.

- 8. <u>Company's Right of First Refusal</u>. Before any Shares held by Optionee or any transferee (either being sometimes referred to herein as the "Holder") may be sold or otherwise transferred (including transfer by gift or operation of law), the Company or its assignee(s) will have a right of first refusal to purchase the Shares on the terms and conditions set forth in this Section 8 (the "Right of First Refusal").
- (a) Notice of Proposed Transfer. The Holder of the Shares will deliver to the Company a written notice (the "Notice") stating: (i) the Holder's bona fide intention to sell or otherwise transfer such Shares; (ii) the name of each proposed optionee or other transferee ("Proposed Transferee"); (iii) the number of Shares to be transferred to each Proposed Transferee; and (iv) the bona fide cash price or other consideration for which the Holder proposes to transfer the Shares (the "Offered Price"), and the Holder will offer the Shares at the Offered Price to the Company or its assignee(s).
 - (b) Exercise of Right of First Refusal. At any time within thirty (30) days after receipt of the Notice, the Company and/or its assignee(s) may, by giving written notice to

the Holder, elect to purchase all, but not less than all, of the Shares proposed to be transferred to any one or more of the Proposed Transferees, at the purchase price determined in accordance with subsection (c) below.

- (c) <u>Purchase Price</u>. The purchase price ("Purchase Price") for the Shares purchased by the Company or its assignee(s) under this Section 8 will be the Offered Price. If the Offered Price includes consideration other than cash, the cash equivalent value of the non-cash consideration will be determined by the Board of Directors of the Company in good faith.
- (d) <u>Payment</u>. Payment of the Purchase Price will be made, at the option of the Company or its assignee(s), in cash (by check), by cancellation of all or a portion of any outstanding indebtedness of the Holder to the Company (or, in the case of repurchase by an assignee, to the assignee), or by any combination thereof within thirty (30) days after receipt of the Notice or in the manner and at the times set forth in the Notice.
- (e) <u>Holder's Right to Transfer</u>. If all of the Shares proposed in the Notice to be transferred to a given Proposed Transferee are not purchased by the Company and/or its assignee(s) as provided in this Section 8, then the Holder may sell or otherwise transfer such Shares to that Proposed Transferee at the Offered Price or at a higher price, provided that such sale or other transfer is consummated within one hundred and twenty (120) days after the date of the Notice, that any such sale or other transfer is effected in accordance with any applicable securities laws and that the Proposed Transferee agrees in writing that the provisions of this Section 8 will continue to apply to the Shares in the hands of such Proposed Transferee. If the Shares described in the Notice are not transferred to the Proposed Transferee within such period, a new Notice will be given to the Company, and the Company and/or its assignees will again be offered the Right of First Refusal before any Shares held by the Holder may be sold or otherwise transferred.
- (f) Exception for Certain Family Transfers. Anything to the contrary contained in this Section notwithstanding, the transfer of any or all of the Shares during the Optionee's lifetime or on the Optionee's death by will or intestacy to the Optionee's immediate family or a trust for the benefit of the Optionee's immediate family will be exempt from the provisions of this Section 8. "Immediate Family" as used herein will mean spouse, lineal descendant or antecedent, father, mother, brother or sister. In such case, the transferee or other recipient will receive and hold the Shares so transferred subject to the provisions of this Section 8, and there will be no further transfer of such Shares except in accordance with the terms of this Section 8.
- (g) <u>Termination of Right of First Refusal</u>. The Right of First Refusal will terminate as to any Shares upon the earlier of (i) the first sale of Common Stock of the Company to the general public, or (ii) a Change in Control in which the successor corporation has equity securities that are publicly traded.
- 9. <u>Death of Optionee</u>. Any distribution or delivery to be made to Optionee under this Agreement will, if Optionee is then deceased, be made to Optionee's designated beneficiary, or if no beneficiary survives Optionee, the administrator or executor of Optionee's estate. Any such transferee must furnish the Company with (a) written notice of his or her status as

transferee, and (b) evidence satisfactory to the Company to establish the validity of the transfer and compliance with any laws or regulations pertaining to said transfer.

- 10. Withholding of Taxes. Notwithstanding any contrary provision of this Agreement, no certificate representing the Shares of Restricted Stock may be released from the escrow established pursuant to Section 2, unless and until satisfactory arrangements (as determined by the Administrator) will have been made by Optionee with respect to the payment of income, employment and other taxes which the Company determines must be withheld with respect to such Shares. To the extent determined appropriate by the Company in its discretion, it will have the right (but not the obligation) to satisfy any tax withholding obligations by reducing the number of Shares otherwise deliverable to Optionee. If Optionee fails to make satisfactory arrangements for the payment of any required tax withholding obligations because at the time any applicable Shares otherwise are scheduled to vest pursuant to Sections 5 or 6, Optionee will permanently forfeit such Shares and the Shares will be returned to the Company at no cost to the Company.
- 11. Rights as Stockholder. Neither Optionee nor any person claiming under or through Optionee will have any of the rights or privileges of a stockholder of the Company in respect of any Shares deliverable hereunder unless and until certificates representing such Shares will have been issued, recorded on the records of the Company or its transfer agents or registrars, and delivered to Optionee or the Escrow Agent. Except as provided in Section 2(f), after such issuance, recordation and delivery, Optionee will have all the rights of a stockholder of the Company with respect to voting such Shares and receipt of dividends and distributions on such Shares.
- 12. No Guarantee of Continued Service, OPTIONEE ACKNOWLEDGES AND AGREES THAT THE VESTING OF THE SHARES OF RESTRICTED STOCK PURSUANT TO THE VESTING SCHEDULE HEREOF IS EARNED ONLY BY CONTINUING AS A SERVICE PROVIDER AT THE WILL OF THE COMPANY (OR THE PARENT OR SUBSIDIARY EMPLOYING OR RETAINING OPTIONEE) AND NOT THROUGH THE ACT OF BEING HIRED, BEING GRANTED THIS RESTRICTED STOCK OR ACQUIRING SHARES HEREUNDER. OPTIONEE FURTHER ACKNOWLEDGES AND AGREES THAT THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREUNDER AND THE VESTING SCHEDULE SET FORTH HEREIN DO NOT CONSTITUTE AN EXPRESS OR IMPLIED PROMISE OF CONTINUED ENGAGEMENT AS A SERVICE PROVIDER FOR THE VESTING PERIOD, FOR ANY PERIOD, OR AT ALL, AND WILL NOT INTERFERE IN ANY WAY WITH OPTIONEE'S RIGHT OR THE RIGHT OF THE COMPANY (OR THE PARENT OR SUBSIDIARY EMPLOYING OR RETAINING OPTIONEE) TO TERMINATE OPTIONEE'S RELATIONSHIP AS A SERVICE PROVIDER AT ANY TIME, WITH OR WITHOUT CAUSE.
- 13. Address for Notices. Any notice to be given to the Company under the terms of this Agreement will be addressed to the Company at comScore, Inc., 11465 Sunset Hill Road, Suite 200, Reston, Virginia 20190, or at such other address as the Company may hereafter designate in writing.

- 14. <u>Grant is Not Transferable</u>. Except to the limited extent provided in Section 8, the unvested Shares subject to this grant and the rights and privileges conferred hereby will not be transferred, assigned, pledged or hypothecated in any way (whether by operation of law or otherwise) and will not be subject to sale under execution, attachment or similar process. Upon any attempt to transfer, assign, pledge, hypothecate or otherwise dispose of any unvested Shares of Restricted Stock subject to this grant, or any right or privilege conferred hereby, or upon any attempted sale under any execution, attachment or similar process, this grant and the rights and privileges conferred hereby immediately will become null and void.
- 15. <u>Binding Agreement</u>. Subject to the limitation on the transferability of this grant contained herein, this Agreement will be binding upon and inure to the benefit of the heirs, legatees, legal representatives, successors and assigns of the parties hereto.
- 16. Additional Conditions to Release from Escrow. The Company will not be required to issue any certificate or certificates for Shares hereunder or release such Shares from the escrow established pursuant to Section 2 prior to fulfillment of all the following conditions: (a) the admission of such Shares to listing on all stock exchanges on which such class of stock is then listed; (b) the completion of any registration or other qualification of such Shares under any state or federal law or under the rulings or regulations of the Securities and Exchange Commission or any other governmental regulatory body, which the Administrator will, in its absolute discretion, deem necessary or advisable; (c) the obtaining of any approval or other clearance from any state or federal governmental agency, which the Administrator will, in its absolute discretion, determine to be necessary or advisable; and (d) the lapse of such reasonable period of time following the date of grant of the Restricted Stock as the Administrator may establish from time to time for reasons of administrative convenience.
- 17. <u>Plan Governs</u>. This Agreement is subject to all terms and provisions of the Plan. In the event of a conflict between one or more provisions of this Agreement and one or more provisions of the Plan will govern. Capitalized terms used and not defined in this Agreement will have the meaning set forth in the Plan.
- 18. <u>Administrator Authority</u>. The Administrator will have the power to interpret the Plan and this Agreement and to adopt such rules for the administration, interpretation and application of the Plan as are consistent therewith and to interpret or revoke any such rules (including, but not limited to, the determination of whether or not any Shares of Restricted Stock have vested). All actions taken and all interpretations and determinations made by the Administrator in good faith will be final and binding upon Optionee, the Company and all other interested persons. No member of the Administrator will be personally liable for any action, determination or interpretation made in good faith with respect to the Plan or this Agreement.
- 19. <u>Electronic Delivery.</u> The Company may, in its sole discretion, decide to deliver any documents related to the Shares of Restricted Stock awarded under the Plan or future Restricted Stock that may be awarded under the Plan by electronic means or request Optionee's consent to participate in the Plan by electronic means. Optionee hereby consents to receive such documents by electronic delivery and agrees to participate in the Plan through any on-line or electronic system established and maintained by the Company or another third party designated by the Company.

- 20. Section 83(b) Election. Optionee hereby acknowledges that he or she has been informed that, with respect to the exercise of an Option for unvested Shares, an election (the "Election") may be filed by the Optionee with the Internal Revenue Service, within thirty (30) days of the purchase of the exercised Shares. A form of Election under Section 83(b) is attached hereto as Exhibit C-2 for reference.
- OPTIONEE ACKNOWLEDGES THAT IT IS OPTIONEE'S SOLE RESPONSIBILITY AND NOT THE COMPANY'S TO FILE TIMELY THE ELECTION UNDER SECTION 83(b) OF THE CODE, EVEN IF OPTIONEE REQUESTS THE COMPANY OR ITS REPRESENTATIVE TO MAKE THIS FILING ON OPTIONEE'S BEHALF.
 - 21. Captions. Captions provided herein are for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.
- 22. Agreement Severable. In the event that any provision in this Agreement will be held invalid or unenforceable, such provision will be severable from, and such invalidity or unenforceability will not be construed to have any effect on, the remaining provisions of this Agreement.
- 23. <u>Modifications to the Agreement</u>. This Agreement constitutes the entire understanding of the parties on the subjects covered. Optionee expressly warrants that he or she is not accepting this Agreement in reliance on any promises, representations, or inducements other than those contained herein. Modifications to this Agreement or the Plan can be made only in an express written contract executed by a duly authorized officer of the Company. Notwithstanding anything to the contrary in the Plan or this Agreement, the Company reserves the right to revise this Agreement as it deems necessary or advisable, in its sole discretion and without the consent of Optionee, to comply with Section 409A of the Internal Revenue Code of 1986, as amended (the "Code") or to otherwise avoid imposition of any additional tax or income recognition under Section 409A of the Code in connection to this Award.
- 24. <u>Amendment, Suspension or Termination of the Plan</u>. By accepting this Award, Optionee expressly warrants that he or she has received a Stock Purchase Right under the Plan, and has received, read and understood a description of the Plan. Optionee understands that the Plan is discretionary in nature and may be amended, suspended or terminated by the Company at any time.
- 25. <u>Governing Law</u>. This Agreement will be governed by the laws of the [Commonwealth of Virginia], without giving effect to the conflict of law principles thereof. For purposes of litigating any dispute that arises under this Award or this Agreement, the parties hereby submit to and consent to the jurisdiction of the [Commonwealth of Virginia], and agree that such litigation will be conducted in the courts of [Fairfax County], [Virginia], or the federal courts for the United States for the [Eastern District of Virginia], and no other courts, where this Award is made and/or to be performed.

EXHIBIT B

INVESTMENT REPRESENTATION STATEMENT

OPTIONEE

COMPANY : COMSCORE, INC.

SECURITY : COMMON STOCK

AMOUNT :

DATE

In connection with the purchase of the above-listed Securities, the undersigned Optionee represents to the Company the following:

(a) Optionee is aware of the Company's business affairs and financial condition and has acquired sufficient information about the Company to reach an informed and knowledgeable decision to acquire the Securities. Optionee is acquiring these Securities for investment for Optionee's own account only and not with a view to, or for resale in connection with, any "distribution" thereof within the meaning of the Securities Act of 1933 as amended (the "Securities Act")

(b) Optionee acknowledges and understands that the Securities constitute "restricted securities" under the Securities Act and have not been registered under the Securities Act in reliance upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of Optionee's investment intent as expressed herein. In this connection, Optionee understands that, in the view of the Securities and Exchange Commission, the statutory basis for such exemption may be unavailable if Optionee's representation was predicated solely upon a present intention to hold these Securities for the minimum capital gains period specified under tax statutes, for a deferred sale, for or until an increase or decrease in the market price of the Securities, or for a period of one year or any other fixed period in the future. Optionee further understands that the Securities must be held indefinitely unless they are subsequently registered under the Securities Act or an exemption from such registration is available. Optionee further acknowledges and understands that the Company is under no obligation to register the Securities. Optionee understands that the certificate evidencing the Securities will be imprinted with any legend required under applicable state

(c) Optionee is familiar with the provisions of Rule 701 and Rule 144, each promulgated under the Securities Act, which, in substance, permit limited public resale of "restricted securities" acquired, directly or indirectly from the issuer thereof, in a non-public offering subject to the satisfaction of certain conditions. Rule 701 provides that if the issuer qualifies under Rule 701 at the time of the grant of the Stock Purchase Right to the Optionee, the exercise will be exempt from registration under the Securities Act. In the event the Company becomes subject to the reporting requirements of Section 13 or 15(d) of the Securities Exchange

Act of 1934, ninety (90) days thereafter (or such longer period as any market stand-off agreement may require) the Securities exempt under Rule 701 may be resold, subject to the satisfaction of certain of the conditions specified by Rule 144, including: (1) the resale being made through a broker in an unsolicited "broker's transaction" or in transactions directly with a market maker (as said term is defined under the Securities Exchange Act of 1934); and, in the case of an affiliate, (2) the availability of certain public information about the Company, (3) the amount of Securities being sold during any three (3) month period not exceeding the limitations specified in Rule 144(e), and (4) the timely filling of a Form 144, if applicable.

In the event that the Company does not qualify under Rule 701 at the time of grant of the Option, then the Securities may be resold in certain limited circumstances subject to the provisions of Rule 144, which requires the resale to occur not less than one (1) year after the later of the date the Securities were sold by the Company or the date the Securities were sold by an affiliate of the Company, within the meaning of Rule 144; and, in the case of acquisition of the Securities by an affiliate, or by a non-affiliate who subsequently holds the Securities less than two (2) years, the satisfaction of the conditions set forth in sections (1), (2), (3) and (4) of the paragraph immediately above.

(d) Optionee further understands that in the event all of the applicable requirements of Rule 701 or 144 are not satisfied, registration under the Securities Act, compliance with Regulation A, or some other registration exemption will be required; and that, notwithstanding the fact that Rules 144 and 701 are not exclusive, the Staff of the Securities and Exchange Commission has expressed its opinion that persons proposing to sell private placement securities other than in a registered offering and otherwise than pursuant to Rules 144 or 701 will have a substantial burden of proof in establishing that an exemption from registration is available for such offers or sales, and that such persons and their respective brokers who participate in such transactions do so at their own risk. Optionee understands that no assurances can be given that any such other registration exemption will be available in such event.

	OPTIONEE
	Signature
	Print Name
	Date
-12-	

EXHIBIT C-1

ASSIGNMENT SEPARATE FROM CERTIFICATE

OR VALUE RECEIVED I,, hereby sell, assign and transfer unto comScore, Inc() shares of the Common Stock of comScore, Inc. standing in my name of the books of said corporation represent Certificate No herewith and do hereby irrevocably constitute and appoint to transfer the said stock on the books of the within named corporation with full power of substitution in the premises.	ted
is Stock Assignment may be used only in accordance with the Stock Purchase Right Agreement between comScore, Inc. and the undersigned dated,(the "Agreement").	
signature:	
INSTRUCTIONS: Please do not fill in any blanks other than the signature line. The purpose of this assignment is to enable the Company to exercise its "repurchase option," as set forth in the Agreement, without puring additional signatures on the part of the Optionee.	:
-13-	

EXHIBIT C-2

ELECTION UNDER SECTION 83(b) OF THE INTERNAL REVENUE CODE OF 1986

The undersigned taxpayer hereby elects, pursuant to Sections 55 and 83(b) of the Internal Revenue Code of 1986, as amended, to include in taxpayer's gross income or alternative minimum taxable income, as the case may be, for the current taxable year the amount of any compensation taxable to taxpayer in connection with taxpayer's receipt of the property described below.

 $1. \ \ \, \text{The name, address, taxpayer identification number and taxable year of the undersigned are as follows:}$

	NAM	Œ:	TAXPAYER:		SPOUSE:			
	ADD	RESS:						
	IDEN	VIIFICATION NO.:	TAXPAYER:		SPOUSE:			
	TAX	ABLE YEAR:						
2.	The property wi	The property with respect to which the election is made is described as follows: shares (the "Shares") of the Common Stock of comScore, Inc. (the "Company").						
3.	The date on whi	date on which the property was transferred is:,						
4.	The property is	The property is subject to the following restrictions:						
		The Shares may not be transferred and are subject to forfeiture under the terms of an agreement between the taxpayer and the Company. These restrictions lapse upon the satisfaction of certain conditions containe in such agreement.						
5.	The fair market	fair market value at the time of transfer, determined without regard to any restriction other than a restriction which by its terms will never lapse, of such property is: \$						
6.	6. The amount (if any) paid for such property is: \$							
		s submitted a copy of this statement to the person for whom the s ming the services in connection with the transfer of said property.						
Da	ated:			Taxpayer				
Th	he undersigned spo	ouse of taxpayer joins in this election.						
Da	ated:			Spouse of Taxpayer				

COMSCORE, INC.

1999 STOCK PLAN

NOTICE OF GRANT OF RESTRICTED STOCK UNITS

Unless otherwise defined herein, the terms defined in the 1999 Stock Plan (the "Plan") will have the same defined meanings in this Notice of Grant of Restricted Stock Units (the "Notice of Grant") and Terms and Conditions of Restricted Stock Grant, attached hereto as Exhibit A (together, the "Restricted Stock Unit Agreement").

Optionee:

Address:				
Optionee has been granted the right to receive an Award of Restricted Stock Units, subject to the terms and conditions of the Plan and this Agreement as follows:				
Grant Number				
Date of Grant				
Vesting Commencement Date				
Number of Restricted Stock Units				
Vesting Schedule:				
Subject to any acceleration provisions contained in the Plan or set forth below, the Restricted Stock Unit shall vest in accordance with the following schedule:				
[INSERT VESTING SCHEDULE.]				

By Optionee's signature and the signature of the Company's representative below, Optionee and the Company agree that this Award of Restricted Stock Units is granted under and governed by the terms and conditions of the Plan and this Agreement. Optionee has reviewed the Plan and this Agreement in their entirety, has had an opportunity to obtain the advice of counsel prior to executing this Agreement and fully understands all provisions of the Plan and Agreement. Optionee hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions relating to the Plan and Agreement. Optionee further agrees to notify the Company upon any change in the residence address indicated below.

OPTIONEE		COMSCORE, INC.
Signature	-	By
		,
Print Name	-	Title
Address:		
	<u>-</u>	
	-	

EXHIBIT A

TERMS AND CONDITIONS OF RESTRICTED STOCK UNIT GRANT

- 1. <u>Grant</u>. The Company hereby grants to Optionee under the Plan for future services and as a separate incentive in connection with his or her future services and not in lieu of any salary or other compensation for his or her future services, an Award of Restricted Stock Units, subject to all of the terms and conditions in this Agreement and the Plan, which is incorporated herein by reference. Subject to Section 15(c) of the Plan, in the event of a conflict between the terms and conditions of the Plan and the terms and conditions of the Plan will prevail.
- 2. Company's Obligation to Pay. Each Restricted Stock Unit represents the right to receive a Share on the date it vests. Unless and until the Restricted Stock Units will have vested in the manner set forth in Section 5, Optionee will have no right to payment of any such Restricted Stock Units. Prior to actual payment of any vested Restricted Stock Units, such Restricted Stock Unit will represent an unsecured obligation of the Company, payable (if at all) only from the general assets of the Company. Any Restricted Stock Units that vest in accordance with Sections 5 or 6 will be paid to Optionee (or in the event of Optionee's death, to his or her estate) in whole Shares, subject to Optionee satisfying any applicable tax withholding obligations as set forth in Section 10.
- 3. Optionee's Representations. In the event the Shares have not been registered under the Securities Act of 1933, as amended, at the time they are to be issued hereunder, the Optionee will, if required by the Company, concurrently with the issuance of such Shares, deliver to the Company his or her Investment Representation Statement in the form attached hereto as Exhibit B.
- 4. <u>Lock-Up Period</u>. Optionee hereby agrees that Optionee will not offer, pledge, sell, contract to sell, sell any option or contract to purchase any option or contract to sell, grant any option, right or warrant to purchase, lend, or otherwise transfer or dispose of, directly or indirectly, any Common Stock (or other securities) of the Company or enter into any swap, hedging or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any Common Stock (or other securities) of the Company held by Optionee (other than those included in the registration) for a period specified by the representative of the underwriters of Common Stock (or other securities) of the Company following the effective date of any registration statement of the Company filed under the Securities Act (or such other period as may be requested by the Company or the underwriters to accommodate regulatory restrictions on (i) the publication or other distribution of research reports and (ii) analyst recommendations and opinions, including, but not limited to, the restrictions contained in NASD Rule 2711(f)(4) or NYSE Rule 472(f)(4), or any successor provisions or amendments thereto).

Optionee agrees to execute and deliver such other agreements as may be reasonably requested by the Company or the underwriter which are consistent with the foregoing or which are necessary to give further effect thereto. In addition, if requested by the Company or the representative of the underwriters of Common Stock (or other securities) of the Company, Optionee will provide, within ten (10) days of such request, such information as may be required

by the Company or such representative in connection with the completion of any public offering of the Company's securities pursuant to a registration statement filed under the Securities Act. The obligations described in this Section will not apply to a registration relating solely to employee benefit plans on Form S-1 or Form S-8 or similar forms that may be promulgated in the future, or a registration relating solely to a Commission Rule 145 transaction on Form S-4 or similar forms that may be promulgated in the future. The Company may impose stop-transfer instructions with respect to the shares of Common Stock (or other securities) subject to the foregoing restriction until the end of said one hundred eighty (180) day (or other) period. Optionee agrees that any transferee of the Stock Purchase Right or Shares acquired pursuant to the Stock Purchase Right will be bound by this Section.

- 5. <u>Vesting Schedule</u>. Except as provided in Section 6, and subject to Section 7, the Restricted Stock Units awarded by this Agreement will vest in accordance with the vesting provisions set forth in the Notice of Grant. Restricted Stock Units scheduled to vest on a certain date or upon the occurrence of a certain condition will not vest in Optionee in accordance with any of the provisions of this Agreement, unless Optionee will have been continuously a Service Provider from the Date of Grant until the date such vesting occurs.
- 6. <u>Administrator Discretion</u>. The Administrator, in its discretion, may accelerate the vesting of the balance, or some lesser portion of the balance, of the unvested Restricted Stock Units at any time, subject to the terms of the Plan. If so accelerated, such Restricted Stock Units will be considered as having vested as of the date specified by the Administrator.
- 7. <u>Forfeiture upon Termination of Status as a Service Provider</u>. Notwithstanding any contrary provision of this Agreement, the balance of the Restricted Stock Units that have not vested as of the time of Optionee's termination as a Service Provider for any or no reason and Optionee's right to acquire any Shares hereunder will immediately terminate.
- 8. <u>Death of Optionee</u>. Any distribution or delivery to be made to Optionee under this Agreement will, if Optionee is then deceased, be made to Optionee's designated beneficiary, or if no beneficiary survives Optionee, the administrator or executor of Optionee's estate. Any such transferee must furnish the Company with (a) written notice of his or her status as transferee, and (b) evidence satisfactory to the Company to establish the validity of the transfer and compliance with any laws or regulations pertaining to said transfer.
- 9. <u>Company's Right of First Refusal</u>. Before any Shares held by Optionee or any transferee (either being sometimes referred to herein as the "Holder") may be sold or otherwise transferred (including transfer by gift or operation of law), the Company or its assignee(s) will have a right of first refusal to purchase the Shares on the terms and conditions set forth in this Section 9 (the "Right of First Refusal").
- (a) Notice of Proposed Transfer. The Holder of the Shares will deliver to the Company a written notice (the "Notice") stating: (i) the Holder's bona fide intention to sell or otherwise transfer such Shares; (ii) the name of each proposed optionee or other transferee ("Proposed Transferee"); (iii) the number of Shares to be transferred to each Proposed Transferee; and (iv) the bona fide cash price or other consideration for which the Holder proposes to transfer the Shares (the "Offered Price"), and the Holder will offer the Shares at the Offered Price to the Company or its assignee(s).

- (b) Exercise of Right of First Refusal. At any time within thirty (30) days after receipt of the Notice, the Company and/or its assignee(s) may, by giving written notice to the Holder, elect to purchase all, but not less than all, of the Shares proposed to be transferred to any one or more of the Proposed Transferees, at the purchase price determined in accordance with subsection (c) below.
- (c) <u>Purchase Price</u>. The purchase price ("Purchase Price") for the Shares purchased by the Company or its assignee(s) under this Section 9 will be the Offered Price. If the Offered Price includes consideration other than cash, the cash equivalent value of the non-cash consideration will be determined by the Board of Directors of the Company in good faith.
- (d) <u>Payment</u>. Payment of the Purchase Price will be made, at the option of the Company or its assignee(s), in cash (by check), by cancellation of all or a portion of any outstanding indebtedness of the Holder to the Company (or, in the case of repurchase by an assignee, to the assignee), or by any combination thereof within thirty (30) days after receipt of the Notice or in the manner and at the times set forth in the Notice.
- (e) <u>Holder's Right to Transfer</u>. If all of the Shares proposed in the Notice to be transferred to a given Proposed Transferee are not purchased by the Company and/or its assignee(s) as provided in this Section 9, then the Holder may sell or otherwise transfer such Shares to that Proposed Transferee at the Offered Price or at a higher price, provided that such sale or other transfer is consummated within one hundred and twenty (120) days after the date of the Notice, that any such sale or other transfer is effected in accordance with any applicable securities laws and that the Proposed Transferee agrees in writing that the provisions of this Section 9 will continue to apply to the Shares in the hands of such Proposed Transferee. If the Shares described in the Notice are not transferred to the Proposed Transferee within such period, a new Notice will be given to the Company, and the Company and/or its assignees will again be offered the Right of First Refusal before any Shares held by the Holder may be sold or otherwise transferred.
- (f) Exception for Certain Family Transfers. Anything to the contrary contained in this Section notwithstanding, the transfer of any or all of the Shares during the Optionee's lifetime or on the Optionee's death by will or intestacy to the Optionee's immediate family or a trust for the benefit of the Optionee's immediate family will be exempt from the provisions of this Section 9. "Immediate Family" as used herein will mean spouse, lineal descendant or antecedent, father, mother, brother or sister. In such case, the transferee or other recipient will receive and hold the Shares so transferred subject to the provisions of this Section 9, and there will be no further transfer of such Shares except in accordance with the terms of this Section 9.
- (g) <u>Termination of Right of First Refusal</u>. The Right of First Refusal will terminate as to any Shares upon the earlier of (i) the first sale of Common Stock of the Company to the general public, or (ii) a Change in Control in which the successor corporation has equity securities that are publicly traded.
- 10. Withholding of Taxes. Notwithstanding any contrary provision of this Agreement, no certificate representing the Shares will be issued to Optionee, unless and until satisfactory arrangements (as determined by the Administrator) will have been made by Optionee with respect to the payment of income, employment and other taxes which the Company

determines must be withheld with respect to such Shares. To the extent determined appropriate by the Company in its discretion, it shall have the right (but not the obligation) to satisfy any tax withholding obligations by reducing the number of Shares otherwise deliverable to Optionee. If Optionee fails to make satisfactory arrangements for the payment of any required tax withholding obligations hereunder at the time any applicable Restricted Stock Units otherwise are scheduled to vest pursuant to Sections 5 or 6, Optionee will permanently forfeit such Restricted Stock Units and any right to receive Shares thereunder and the Restricted Stock Units will be returned to the Company at no cost to the Company.

- 11. <u>Rights as Stockholder</u>. Neither Optionee nor any person claiming under or through Optionee will have any of the rights or privileges of a stockholder of the Company in respect of any Shares deliverable hereunder unless and until certificates representing such Shares will have been issued, recorded on the records of the Company or its transfer agents or registrars, and delivered to Optionee. After such issuance, recordation and delivery, Optionee will have all the rights of a stockholder of the Company with respect to voting such Shares and receipt of dividends and distributions on such Shares.
- 12. No Guarantee of Continued Service. OPTIONEE ACKNOWLEDGES AND AGREES THAT THE VESTING OF THE RESTRICTED STOCK UNITS PURSUANT TO THE VESTING SCHEDULE HEREOF IS EARNED ONLY BY CONTINUING AS A SERVICE PROVIDER AT THE WILL OF THE COMPANY (OR THE PARENT OR SUBSIDIARY EMPLOYING OR RETAINING OPTIONEE) AND NOT THROUGH THE ACT OF BEING HIRED, BEING GRANTED THIS RESTRICTED STOCK OR ACQUIRING SHARES HEREUNDER. OPTIONEE FURTHER ACKNOWLEDGES AND AGREES THAT THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREUNDER AND THE VESTING SCHEDULE SET FORTH HEREIN DO NOT CONSTITUTE AN EXPRESS OR IMPLIED PROMISE OF CONTINUED ENGAGEMENT AS A SERVICE PROVIDER FOR THE VESTING PERIOD, FOR ANY PERIOD, OR AT ALL, AND WILL NOT INTERFERE IN ANY WAY WITH OPTIONEE'S RIGHT OR THE RIGHT OF THE COMPANY (OR THE PARENT OR SUBSIDIARY EMPLOYING OR RETAINING OPTIONEE) TO TERMINATE OPTIONEE'S RELATIONSHIP AS A SERVICE PROVIDER AT ANY TIME, WITH OR WITHOUT CAUSE.
- 13. Address for Notices. Any notice to be given to the Company under the terms of this Agreement will be addressed to the Company at ComScore, Inc., 11465 Sunset Hill Road, Suite 200, Reston, Virginia 20190, or at such other address as the Company may hereafter designate in writing.
- 14. Grant is Not Transferable. Except to the limited extent provided in Section 8, this grant and the rights and privileges conferred hereby will not be transferred, assigned, pledged or hypothecated in any way (whether by operation of law or otherwise) and will not be subject to sale under execution, attachment or similar process. Upon any attempt to transfer, assign, pledge, hypothecate or otherwise dispose of this grant, or any right or privilege conferred hereby, or upon any attempted sale under any execution, attachment or similar process, this grant and the rights and privileges conferred hereby immediately will become null and void.
- 15. Binding Agreement. Subject to the limitation on the transferability of this grant contained herein, this Agreement will be binding upon and inure to the benefit of the heirs,

legatees, legal representatives, successors and assigns of the parties hereto.

- 16. Additional Conditions to Issuance of Stock. If at any time the Company will determine, in its discretion, that the listing, registration or qualification of the Shares upon any securities exchange or under any state or federal law, or the consent or approval of any governmental regulatory authority is necessary or desirable as a condition to the issuance of Shares to Optionee (or his or her estate), such issuance will not occur unless and until such listing, registration, qualification, consent or approval will have been effected or obtained free of any conditions not acceptable to the Company. Where the Company determines that the delivery of the payment of any Shares will violate federal securities laws or other applicable laws, the Company will defer delivery until the earliest date at which the Company reasonably anticipates that the delivery of Shares will no longer cause such violation. The Company will make all reasonable efforts to meet the requirements of any such state or federal law or securities exchange and to obtain any such consent or approval of any such governmental authority.
- 17. <u>Plan Governs</u>. This Agreement is subject to all terms and provisions of the Plan. In the event of a conflict between one or more provisions of this Agreement and one or more provisions of the Plan will govern. Capitalized terms used and not defined in this Agreement will have the meaning set forth in the Plan.
- 18. <u>Administrator Authority</u>. The Administrator will have the power to interpret the Plan and this Agreement and to adopt such rules for the administration, interpretation and application of the Plan as are consistent therewith and to interpret or revoke any such rules (including, but not limited to, the determination of whether or not any Restricted Stock Units have vested). All actions taken and all interpretations and determinations made by the Administrator in good faith will be final and binding upon Optionee, the Company and all other interested persons. No member of the Administrator will be personally liable for any action, determination or interpretation made in good faith with respect to the Plan or this Agreement.
- 19. <u>Electronic Delivery.</u> The Company may, in its sole discretion, decide to deliver any documents related to Restricted Stock Units awarded under the Plan or future Restricted Stock Units that may be awarded under the Plan by electronic means or request Optionee's consent to participate in the Plan by electronic means. Optionee hereby consents to receive such documents by electronic delivery and agrees to participate in the Plan through any on-line or electronic system established and maintained by the Company or another third party designated by the Company.
 - 20. Captions. Captions provided herein are for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.
- 21. <u>Agreement Severable</u>. In the event that any provision in this Agreement will be held invalid or unenforceable, such provision will be severable from, and such invalidity or unenforceability will not be construed to have any effect on, the remaining provisions of this Agreement.
- 22. <u>Modifications to the Agreement</u>. This Agreement constitutes the entire understanding of the parties on the subjects covered. Optionee expressly warrants that he or she is not accepting this Agreement in reliance on any promises, representations, or inducements other than those contained herein. Modifications to this Agreement or the Plan can be made only

in an express written contract executed by a duly authorized officer of the Company. Notwithstanding anything to the contrary in the Plan or this Agreement, the Company reserves the right to revise this Agreement as it deems necessary or advisable, in its sole discretion and without the consent of Optionee, to comply with Section 409A of the Internal Revenue Code of 1986, as amended (the "Code") or to otherwise avoid imposition of any additional tax or income recognition under Section 409A of the Code in connection to this Award of Restricted Stock Units.

- 23. <u>Amendment, Suspension or Termination of the Plan</u>. By accepting this Award, Optionee expressly warrants that he or she has received an Award of Restricted Stock Units under the Plan, and has received, read and understood a description of the Plan. Optionee understands that the Plan is discretionary in nature and may be amended, suspended or terminated by the Company at any time.
- 24. Governing Law. This Agreement will be governed by the laws of the [Commonwealth of Virginia], without giving effect to the conflict of law principles thereof. For purposes of litigating any dispute that arises under this Award or this Agreement, the parties hereby submit to and consent to the jurisdiction of the [Commonwealth of Virginia], and agree that such litigation will be conducted in the courts of [Fairfax County], [Virginia], or the federal courts for the United States for the [Eastern District of Virginia], and no other courts, where this Award is made and/or to be performed.

EXHIBIT B

INVESTMENT REPRESENTATION STATEMENT

OPTIONEE

COMPANY : COMSCORE, INC.

SECURITY : COMMON STOCK

AMOUNT :

DATE

In connection with the purchase of the above-listed Securities, the undersigned Optionee represents to the Company the following:

(a) Optionee is aware of the Company's business affairs and financial condition and has acquired sufficient information about the Company to reach an informed and knowledgeable decision to acquire the Securities. Optionee is acquiring these Securities for investment for Optionee's own account only and not with a view to, or for resale in connection with, any "distribution" thereof within the meaning of the Securities Act of 1933, as amended (the "Securities Act").

(b) Optionee acknowledges and understands that the Securities constitute "restricted securities" under the Securities Act and have not been registered under the Securities Act in reliance upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of Optionee's investment intent as expressed herein. In this connection, Optionee understands that, in the view of the Securities and Exchange Commission, the statutory basis for such exemption may be unavailable if Optionee's representation was predicated solely upon a present intention to hold these Securities for the minimum capital gains period specified under tax statutes, for a deferred sale, for or until an increase or decrease in the market price of the Securities, or for a period of one year or any other fixed period in the future. Optionee further understands that the Securities must be held indefinitely unless they are subsequently registered under the Securities Act or an exemption from such registration is available. Optionee further acknowledges and understands that the Company is under no obligation to register the Securities. Optionee understands that the certificate evidencing the Securities will be imprinted with any legend required under applicable state

(c) Optionee is familiar with the provisions of Rule 701 and Rule 144, each promulgated under the Securities Act, which, in substance, permit limited public resale of "restricted securities" acquired, directly or indirectly from the issuer thereof, in a non-public offering subject to the satisfaction of certain conditions. Rule 701 provides that if the issuer qualifies under Rule 701 at the time of the grant of the Stock Purchase Right to the Optionee, the exercise will be exempt from registration under the Securities Act. In the event the Company becomes subject to the reporting requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, ninety (90) days thereafter (or such longer period as any market stand-off agreement may require) the Securities exempt under Rule 701 may be resold, subject to the satisfaction of certain of the conditions specified by Rule 144, including: (1) the resale being

made through a broker in an unsolicited "broker's transaction" or in transactions directly with a market maker (as said term is defined under the Securities Exchange Act of 1934); and, in the case of an affiliate, (2) the availability of certain public information about the Company, (3) the amount of Securities being sold during any three (3) month period not exceeding the limitations specified in Rule 144(e), and (4) the timely filing of a Form 144, if applicable.

In the event that the Company does not qualify under Rule 701 at the time of grant of the Option, then the Securities may be resold in certain limited circumstances subject to the provisions of Rule 144, which requires the resale to occur not less than one (1) year after the later of the date the Securities were sold by the Company or the date the Securities were sold by an affiliate of the Company, within the meaning of Rule 144; and, in the case of acquisition of the Securities by an affiliate, or by a non-affiliate who subsequently holds the Securities less than two (2) years, the satisfaction of the conditions set forth in sections (1), (2), (3) and (4) of the paragraph immediately above.

(d) Optionee further understands that in the event all of the applicable requirements of Rule 701 or 144 are not satisfied, registration under the Securities Act, compliance with Regulation A, or some other registration exemption will be required; and that, notwithstanding the fact that Rules 144 and 701 are not exclusive, the Staff of the Securities and Exchange Commission has expressed its opinion that persons proposing to sell private placement securities other than in a registered offering and otherwise than pursuant to Rules 144 or 701 will have a substantial burden of proof in establishing that an exemption from registration is available for such offers or sales, and that such persons and their respective brokers who participate in such transactions do so at their own risk. Optionee understands that no assurances can be given that any such other registration exemption will be available in such event.

OPTIONE	2		
Signature			
Print Name			
Date			

COMSCORE, INC.

2007 EQUITY INCENTIVE PLAN

- 1. Purposes of the Plan. The purposes of this Plan are:
 - to attract and retain the best available personnel for positions of substantial responsibility,
 - to provide additional incentive to Employees, Directors and Consultants, and
 - to promote the success of the Company's business.

The Plan permits the grant of Incentive Stock Options, Nonstatutory Stock Options, Restricted Stock, Restricted Stock Units, Stock Appreciation Rights, Performance Units and Performance Shares.

- 2. <u>Definitions</u>. As used herein, the following definitions will apply:
 - (a) "Administrator" means the Board or any of its Committees as will be administering the Plan, in accordance with Section 4 of the Plan.
- (b) "Applicable Laws" means the requirements relating to the administration of equity-based awards under U.S. state corporate laws, U.S. federal and state securities laws, the Code, any stock exchange or quotation system on which the Common Stock is listed or quoted and the applicable laws of any foreign country or jurisdiction where Awards are, or will be, granted under the Plan.
 - (c) "Award" means, individually or collectively, a grant under the Plan of Options, Stock Appreciation Rights, Restricted Stock, Restricted Stock Units, Performance Units or Performance Shares.
- (d) "Award Agreement" means the written or electronic agreement setting forth the terms and provisions applicable to each Award granted under the Plan. The Award Agreement is subject to the terms and conditions of the Plan.
 - (e) " \underline{Board} " means the Board of Directors of the Company.
 - (f) "Change in Control" means the occurrence of any of the following events:
- (i) Any "person" (as such term is used in Sections 13(d) and 14(d) of the Exchange Act) becomes the "beneficial owner" (as defined in Rule 13d-3 of the Exchange Act), directly or indirectly, of securities of the Company representing fifty percent (50%) or more of the total voting power represented by the Company's then outstanding voting securities;

- (ii) The consummation of the sale or disposition by the Company of all or substantially all of the Company's assets;
- (iii) A change in the composition of the Board occurring within a two (2)-year period, as a result of which fewer than a majority of the directors are Incumbent Directors. "Incumbent Directors" means directors who either (A) are Directors as of the effective date of the Plan, or (B) are elected, or nominated for election, to the Board with the affirmative votes of at least a majority of the Incumbent Directors at the time of such election or nomination (but will not include an individual whose election or nomination is in connection with an actual or threatened proxy contest relating to the election of directors to the Company); or
- (iv) The consummation of a merger or consolidation of the Company with any other corporation, other than a merger or consolidation which would result in the voting securities of the Company outstanding immediately prior thereto continuing to represent (either by remaining outstanding or by being converted into voting securities of the surviving entity or its parent) at least fifty percent (50%) of the total voting power represented by the voting securities of the Company or such surviving entity or its parent outstanding immediately after such merger or consolidation.
 - (g) "Code" means the Internal Revenue Code of 1986, as amended. Any reference to a section of the Code herein will be a reference to any successor or amended section of the Code.
 - (h) "Committee" means a committee of Directors or of other individuals satisfying Applicable Laws appointed by the Board in accordance with Section 4 hereof.
 - (i) "Common Stock" means the common stock of the Company.
 - (j) "Company" means comScore, Inc., a Delaware corporation, or any successor thereto.
 - (k) "Consultant" means any person, including an advisor, engaged by the Company or a Parent or Subsidiary to render services to such entity.
 - (l) "Director" means a member of the Board.
- (m) "Disability" means total and permanent disability as defined in Section 22(e)(3) of the Code, provided that in the case of Awards other than Incentive Stock Options, the Administrator in its discretion may determine whether a permanent and total disability exists in accordance with uniform and non-discriminatory standards adopted by the Administrator from time to time.
- (n) "Employee" means any person, including Officers and Directors, employed by the Company or any Parent or Subsidiary of the Company. Neither service as a Director nor payment of a director's fee by the Company will be sufficient to constitute "employment" by the Company.
 - (o) " $\underline{\text{Exchange Act}}$ " means the Securities Exchange Act of 1934, as amended.

- (p) "Exchange Program" means a program under which (i) outstanding Awards are surrendered or cancelled in exchange for Awards of the same type (which may have higher or lower exercise prices and different terms), Awards of a different type, and/or cash, (ii) Participants would have the opportunity to transfer any outstanding Awards to a financial institution or other person or entity selected by the Administrator, and/or (iii) the exercise price of an outstanding Award is reduced. The Administrator will determine the terms and conditions of any Exchange Program in its sole discretion.
 - (q) "Fair Market Value" means, as of any date, the value of Common Stock determined as follows:
- (i) If the Common Stock is listed on any established stock exchange or a national market system, including without limitation the Nasdaq Global Select Market, the Nasdaq Global Market or the Nasdaq Capital Market of The Nasdaq Stock Market, its Fair Market Value will be the closing sales price for such stock (or the closing bid, if no sales were reported) as quoted on such exchange or system on the day of determination, as reported in *The Wall Street Journal* or such other source as the Administrator deems reliable;
- (ii) If the Common Stock is regularly quoted by a recognized securities dealer but selling prices are not reported, the Fair Market Value of a Share will be the mean between the high bid and low asked prices for the Common Stock on the day of determination, as reported in *The Wall Street Journal* or such other source as the Administrator deems reliable;
- (iii) For purposes of any Awards granted on the Registration Date, the Fair Market Value will be the initial price to the public as set forth in the final prospectus included within the registration statement in Form S-1 filed with the Securities and Exchange Commission for the initial public offering of the Company's Common Stock; or
 - (iv) In the absence of an established market for the Common Stock, the Fair Market Value will be determined in good faith by the Administrator.
 - (r) "Fiscal Year" means the fiscal year of the Company.
 - (s) "Incentive Stock Option" means an Option intended to qualify as an incentive stock option within the meaning of Section 422 of the Code and the regulations promulgated thereunder.
 - (t) "Inside Director" means a Director who is an Employee.
 - (u) "Nonstatutory Stock Option" means an Option that by its terms does not qualify or is not intended to qualify as an Incentive Stock Option.
 - (v) "Officer" means a person who is an officer of the Company within the meaning of Section 16 of the Exchange Act and the rules and regulations promulgated thereunder.
 - (w) " $\underline{\mathrm{Option}}$ " means a stock option granted pursuant to the Plan.
 - (x) "Outside Director" means a Director who is not an Employee.

- (y) "Parent" means a "parent corporation," whether now or hereafter existing, as defined in Section 424(e) of the Code.
- (z) "Participant" means the holder of an outstanding Award.
- (aa) "Performance Share" means an Award denominated in Shares which may be earned in whole or in part upon attainment of performance goals or other vesting criteria as the Administrator may determine pursuant to Section 10.
- (bb) "Performance Unit" means an Award which may be earned in whole or in part upon attainment of performance goals or other vesting criteria as the Administrator may determine and which may be settled for cash, Shares or other securities or a combination of the foregoing pursuant to Section 10.
- (cc) "Period of Restriction" means the period during which the transfer of Shares of Restricted Stock are subject to restrictions and therefore, the Shares are subject to a substantial risk of forfeiture. Such restrictions may be based on the passage of time, the achievement of target levels of performance, or the occurrence of other events as determined by the Administrator.
 - (dd) "Plan" means this 2007 Equity Incentive Plan.
- (ee) "Registration Date" means the effective date of the first registration statement that is filed by the Company and declared effective pursuant to Section 12(g) of the Exchange Act, with respect to any class of the Company's securities.
 - (ff) "Restricted Stock" means Shares issued pursuant to a Restricted Stock award under Section 7 of the Plan, or issued pursuant to the early exercise of an Option.
- (gg) "Restricted Stock Unit" means a bookkeeping entry representing an amount equal to the Fair Market Value of one Share, granted pursuant to Section 8. Each Restricted Stock Unit represents an unfunded and unsecured obligation of the Company.
 - (hh) "Rule 16b-3" means Rule 16b-3 of the Exchange Act or any successor to Rule 16b-3, as in effect when discretion is being exercised with respect to the Plan.
 - (ii) "Section 16(b)" means Section 16(b) of the Exchange Act.
 - (jj) "Service Provider" means an Employee, Director or Consultant.
 - (kk) "Share" means a share of the Common Stock, as adjusted in accordance with Section 13 of the Plan.
 - (II) "Stock Appreciation Right" means an Award, granted alone or in connection with an Option, that pursuant to Section 9 is designated as a Stock Appreciation Right.
 - (mm) "Subsidiary" means a "subsidiary corporation", whether now or hereafter existing, as defined in Section 424(f) of the Code.

3. Stock Subject to the Plan.

- (a) Stock Subject to the Plan. Subject to the provisions of Section 13 of the Plan, the maximum aggregate number of Shares that may be issued under the Plan is 7,000,000 Shares, plus (i) any Shares that, as of the Registration Date, have been reserved but not issued pursuant to any awards granted under the Company's Incentive Plan (the "Existing Plan") and are not subject to any awards granted thereunder, and (ii) any Shares subject to stock options or similar awards granted under the Existing Plan that expire or otherwise terminate without having been exercised in full and Shares issued pursuant to awards granted under the Existing Plan that are forfeited to or repurchased by the Company, with the maximum number of Shares to be added to the Plan pursuant to clauses (i) and (ii) equal to 5,000,000 Shares. The Shares may be authorized, but unissued, or reaccuired Common Stock.
- (b) <u>Automatic Share Reserve Increase</u>. The number of Shares available for issuance under the Plan will be increased on the first day of each Fiscal Year beginning with the 2008 Fiscal Year, in an amount equal to the least of (i) 9,000,000 (pre-split) Shares, (ii) 4% of the outstanding Shares on the last day of the immediately preceding Fiscal Year or (iii) such number of Shares determined by the Board.
- (c) <u>Lapsed Awards</u>. If an Award expires or becomes unexercisable without having been exercised in full, is surrendered pursuant to an Exchange Program, or, with respect to Restricted Stock, Restricted Stock Units, Performance Units or Performance Shares, is forfeited to or repurchased by the Company due to failure to vest, the unpurchased Shares (or for Awards other than Options or Stock Appreciation Rights the forfeited or repurchased Shares) which were subject thereto will become available for future grant or sale under the Plan (unless the Plan has terminated). With respect to Stock Appreciation Rights, only Shares actually issued pursuant to a Stock Appreciation Right will cease to be available under the Plan all remaining Shares under Stock Appreciation Rights will remain available for future grant or sale under the Plan (unless the Plan has terminated). Shares that have actually been issued under the Plan under any Award will not be returned to the Plan and will not become available for future distribution under the Plan; provided, however, that if Shares issued pursuant to Awards of Restricted Stock, Restricted Stock Units, Performance Shares or Performance Units are repurchased by the Company or are forfeited to the Company, such Shares will become available for future grant under the Plan. Shares used to pay the exercise price of an Award or to satisfy the tax withholding obligations related to an Award will become available for future grant or sale under the Plan. To the extent an Award under the Plan is paid out in cash rather than Shares, such cash payment will not result in reducing the number of Shares available for issuance under the Plan. Notwithstanding the foregoing and, subject to adjustment as provided in Section 13, the maximum number of Shares that may be issued upon the exercise of Incentive Stock Options will equal the aggregate Share number stated in Section 3(a), plus, to the extent allowable under Section 422 of the Code and the Treasury Regulations promulgated thereunder, any Shares
 - (d) Share Reserve. The Company, during the term of this Plan, will at all times reserve and keep available such number of Shares as will be sufficient to satisfy the requirements of the Plan.

4. Administration of the Plan.

- (a) Procedure
 - (i) Multiple Administrative Bodies. Different Committees with respect to different groups of Service Providers may administer the Plan.
- (ii) <u>Section 162(m)</u>. To the extent that the Administrator determines it to be desirable to qualify Awards granted hereunder as "performance-based compensation" within the meaning of Section 162(m) of the Code, the Plan will be administered by a Committee of two (2) or more "outside directors" within the meaning of Section 162(m) of the Code.
- (iii) Rule 16b-3. To the extent desirable to qualify transactions hereunder as exempt under Rule 16b-3, the transactions contemplated hereunder will be structured to satisfy the requirements for exemption under Rule 16b-3.
 - (iv) Other Administration. Other than as provided above, the Plan will be administered by (A) the Board or (B) a Committee, which committee will be constituted to satisfy Applicable Laws.
- (b) <u>Powers of the Administrator</u>. Subject to the provisions of the Plan, and in the case of a Committee, subject to the specific duties delegated by the Board to such Committee, the Administrator will have the authority, in its discretion:
 - (i) to determine the Fair Market Value;
 - (ii) to select the Service Providers to whom Awards may be granted hereunder;
 - (iii) to determine the number of Shares to be covered by each Award granted hereunder;
 - (iv) to approve forms of Award Agreements for use under the Plan;
- (v) to determine the terms and conditions, not inconsistent with the terms of the Plan, of any Award granted hereunder. Such terms and conditions include, but are not limited to, the exercise price, the time or times when Awards may be exercised (which may be based on performance criteria), any vesting acceleration or waiver of forfeiture restrictions, and any restriction or limitation regarding any Award or the Shares relating thereto, based in each case on such factors as the Administrator will determine;
 - (vi) to determine the terms and conditions of any, and to institute any Exchange Program;
 - (vii) to construe and interpret the terms of the Plan and Awards granted pursuant to the Plan;

- (viii) to prescribe, amend and rescind rules and regulations relating to the Plan, including rules and regulations relating to sub-plans established for the purpose of satisfying applicable foreign laws;
- (ix) to modify or amend each Award (subject to Section 18(c) of the Plan), including but not limited to the discretionary authority to extend the post-termination exercisability period of Awards and to extend the maximum term of an Option (subject to Section 6(b) regarding Incentive Stock Options);
 - $(x) \ to \ allow \ Participants \ to \ satisfy \ withholding \ tax \ obligations \ in \ such \ manner \ as \ prescribed \ in \ Section \ 14;$
 - (xi) to authorize any person to execute on behalf of the Company any instrument required to effect the grant of an Award previously granted by the Administrator;
 - (xii) to allow a Participant to defer the receipt of the payment of cash or the delivery of Shares that would otherwise be due to such Participant under an Award; and
 - (xiii) to make all other determinations deemed necessary or advisable for administering the Plan.
 - (c) Effect of Administrator's Decision. The Administrator's decisions, determinations and interpretations will be final and binding on all Participants and any other holders of Awards.
- 5. <u>Eligibility</u>. Nonstatutory Stock Options, Stock Appreciation Rights, Restricted Stock, Restricted Stock Units, Performance Shares and Performance Units may be granted to Service Providers. Incentive Stock Options may be granted only to Employees.

6. Stock Options.

- (a) <u>Limitations</u>. Each Option will be designated in the Award Agreement as either an Incentive Stock Option or a Nonstatutory Stock Option. However, notwithstanding such designation, to the extent that the aggregate Fair Market Value of the Shares with respect to which Incentive Stock Options are exercisable for the first time by the Participant during any calendar year (under all plans of the Company and any Parent or Subsidiary) exceeds one hundred thousand dollars (\$100,000), such Options will be treated as Nonstatutory Stock Options. For purposes of this Section 6(a), Incentive Stock Options will be taken into account in the order in which they were granted. The Fair Market Value of the Shares will be determined as of the time the Option with respect to such Shares is granted.
- (b) <u>Term of Option</u>. The term of each Option will be stated in the Award Agreement. In the case of an Incentive Stock Option, the term will be ten (10) years from the date of grant or such shorter term as may be provided in the Award Agreement. Moreover, in the case of an Incentive Stock Option granted to a Participant who, at the time the Incentive Stock Option is granted, owns stock representing more than ten percent (10%) of the total combined voting power of all classes of stock of the Company or any Parent or Subsidiary, the term of the Incentive Stock

Option will be five (5) years from the date of grant or such shorter term as may be provided in the Award Agreement.

- (c) Option Exercise Price and Consideration.
 - (i) Exercise Price. The per share exercise price for the Shares to be issued pursuant to exercise of an Option will be determined by the Administrator, subject to the following:
 - (1) In the case of an Incentive Stock Option
- a) granted to an Employee who, at the time the Incentive Stock Option is granted, owns stock representing more than ten percent (10%) of the voting power of all classes of stock of the Company or any Parent or Subsidiary, the per Share exercise price will be no less than one hundred ten percent (110%) of the Fair Market Value per Share on the date of grant.
- b) granted to any Employee other than an Employee described in paragraph (A) immediately above, the per Share exercise price will be no less than one hundred percent (100%) of the Fair Market Value per Share on the date of grant.
 - (2) In the case of a Nonstatutory Stock Option, the per Share exercise price will be no less than one hundred percent (100%) of the Fair Market Value per Share on the date of grant.
- (3) Notwithstanding the foregoing, Options may be granted with a per Share exercise price of less than one hundred percent (100%) of the Fair Market Value per Share on the date of grant pursuant to a transaction described in, and in a manner consistent with, Section 424(a) of the Code.
- (ii) <u>Waiting Period and Exercise Dates</u>. At the time an Option is granted, the Administrator will fix the period within which the Option may be exercised and will determine any conditions that must be satisfied before the Option may be exercised.
- (iii) Form of Consideration. The Administrator will determine the acceptable form of consideration for exercising an Option, including the method of payment. In the case of an Incentive Stock Option, the Administrator will determine the acceptable form of consideration at the time of grant. Such consideration may consist entirely of: (1) cash; (2) check; (3) promissory note, (4) other Shares, provided that such Shares have a Fair Market Value on the date of surrender equal to the aggregate exercise price of the Shares as to which such Option will be exercised and provided that accepting such Shares, in the sole discretion of the Administrator, will not result in any adverse accounting consequences to the Company; (5) consideration received by the Company under a broker-assisted (or other) cashless exercise program implemented by the Company in connection with the Plan; (6) any combination of the foregoing methods of payment; or (7) such other consideration and method of payment for the issuance of Shares to the extent permitted by Applicable Laws.

(d) Exercise of Option.

(i) <u>Procedure for Exercise</u>; <u>Rights as a Stockholder</u>. Any Option granted hereunder will be exercisable according to the terms of the Plan and at such times and under such conditions as determined by the Administrator and set forth in the Award Agreement. An Option may not be exercised for a fraction of a Share.

An Option will be deemed exercised when the Company receives: (i) notice of exercise (in such form as the Administrator may specify from time to time) from the person entitled to exercise the Option, and (ii) full payment for the Shares with respect to which the Option is exercised (together with applicable withholding taxes). Full payment may consist of any consideration and method of payment authorized by the Administrator and permitted by the Award Agreement and the Plan. Shares issued upon exercise of an Option will be issued in the name of the Participant or, if requested by the Participant, in the name of the Participant and his or her spouse. Until the Shares are issued (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company), no right to vote or receive dividends or any other rights as a stockholder will exist with respect to the Shares subject to an Option, notwithstanding the exercise of the Option. The Company will issue (or cause to be issued) such Shares promptly after the Option is exercised. No adjustment will be made for a dividend or other right for which the record date is prior to the date the Shares are issued, except as provided in Section 13 of the Plan.

Exercising an Option in any manner will decrease the number of Shares thereafter available, both for purposes of the Plan and for sale under the Option, by the number of Shares as to which the Option is exercised.

- (ii) <u>Termination of Relationship as a Service Provider</u>. If a Participant ceases to be a Service Provider, other than upon the Participant's termination as the result of the Participant's death or Disability, the Participant may exercise his or her Option within such period of time as is specified in the Award Agreement to the extent that the Option is vested on the date of termination (but in no event later than the expiration of the term of such Option as set forth in the Award Agreement). In the absence of a specified time in the Award Agreement, the Option will remain exercisable for three (3) months following the Participant's termination. Unless otherwise provided by the Administrator, if on the date of termination the Participant is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option will revert to the Plan. If after termination the Participant does not exercise his or her Option within the time specified by the Administrator, the Option will terminate, and the Shares covered by such Option will revert to the Plan.
- (iii) <u>Disability of Participant</u>. If a Participant ceases to be a Service Provider as a result of the Participant's Disability, the Participant may exercise his or her Option within such period of time as is specified in the Award Agreement to the extent the Option is vested on the date of termination (but in no event later than the expiration of the term of such Option as set forth in the Award Agreement). In the absence of a specified time in the Award Agreement, the Option will remain exercisable for twelve (12) months following the Participant's termination. Unless otherwise provided by the Administrator, if on the date of termination the Participant is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option will revert to the Plan. If after termination the Participant does not exercise his or her Option within the

time specified herein, the Option will terminate, and the Shares covered by such Option will revert to the Plan.

(iv) <u>Death of Participant</u>. If a Participant dies while a Service Provider, the Option may be exercised following the Participant's death within such period of time as is specified in the Award Agreement to the extent that the Option is vested on the date of death (but in no event may the option be exercised later than the expiration of the term of such Option as set forth in the Award Agreement), by the Participant's designated beneficiary, provided such beneficiary has been designated prior to Participant's death in a form acceptable to the Administrator. If no such beneficiary has been designated by the Participant, then such Option may be exercised by the personal representative of the Participant's estate or by the person(s) to whom the Option is transferred pursuant to the Participant's will or in accordance with the laws of descent and distribution. In the absence of a specified time in the Award Agreement, the Option will remain exercisable for twelve (12) months following Participant's death. Unless otherwise provided by the Administrator, if at the time of death Participant is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option will immediately revert to the Plan. If the Option is not so exercised within the time specified herein, the Option will terminate, and the Shares covered by such Option will revert to the Plan.

7. Restricted Stock

- (a) <u>Grant of Restricted Stock</u>. Subject to the terms and provisions of the Plan, the Administrator, at any time and from time to time, may grant Shares of Restricted Stock to Service Providers in such amounts as the Administrator, in its sole discretion, will determine.
- (b) <u>Restricted Stock Agreement</u>. Each Award of Restricted Stock will be evidenced by an Award Agreement that will specify the Period of Restriction, the number of Shares granted, and such other terms and conditions as the Administrator, in its sole discretion, will determine. Unless the Administrator determines otherwise, the Company as escrow agent will hold Shares of Restricted Stock until the restrictions on such Shares have lapsed.
- (c) <u>Transferability</u>. Except as provided in this Section 7, Shares of Restricted Stock may not be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated until the end of the applicable Period of Restriction.
 - (d) Other Restrictions. The Administrator, in its sole discretion, may impose such other restrictions on Shares of Restricted Stock as it may deem advisable or appropriate.
- (e) <u>Removal of Restrictions</u>. Except as otherwise provided in this Section 7, Shares of Restricted Stock covered by each Restricted Stock grant made under the Plan will be released from escrow as soon as practicable after the last day of the Period of Restriction or at such other time as the Administrator may determine. The Administrator, in its discretion, may accelerate the time at which any restrictions will lapse or be removed.
- (f) <u>Voting Rights</u>. During the Period of Restriction, Service Providers holding Shares of Restricted Stock granted hereunder may exercise full voting rights with respect to those Shares, unless the Administrator determines otherwise.

- (g) <u>Dividends and Other Distributions</u>. During the Period of Restriction, Service Providers holding Shares of Restricted Stock will be entitled to receive all dividends and other distributions paid with respect to such Shares, unless the Administrator provides otherwise. If any such dividends or distributions are paid in Shares, the Shares will be subject to the same restrictions on transferability and forfeitability as the Shares of Restricted Stock with respect to which they were paid.
- (h) <u>Return of Restricted Stock to Company</u>. On the date set forth in the Award Agreement, the Restricted Stock for which restrictions have not lapsed will revert to the Company and again will become available for grant under the Plan.

8. Restricted Stock Units

- (a) <u>Grant</u>. Restricted Stock Units may be granted at any time and from time to time as determined by the Administrator. After the Administrator determines that it will grant Restricted Stock Units under the Plan, it will advise the Participant in an Award Agreement of the terms, conditions, and restrictions related to the grant, including the number of Restricted Stock Units.
- (b) <u>Vesting Criteria and Other Terms</u>. The Administrator will set vesting criteria in its discretion, which, depending on the extent to which the criteria are met, will determine the number of Restricted Stock Units that will be paid out to the Participant. The Administrator may set vesting criteria based upon the achievement of Company-wide, business unit, or individual goals (including, but not limited to, continued employment), or any other basis determined by the Administrator in its discretion.
- (c) <u>Earning Restricted Stock Units</u>. Upon meeting the applicable vesting criteria, the Participant will be entitled to receive a payout as determined by the Administrator. Notwithstanding the foregoing, at any time after the grant of Restricted Stock Units, the Administrator, in its sole discretion, may reduce or waive any vesting criteria that must be met to receive a payout.
- (d) Form and Timing of Payment. Payment of earned Restricted Stock Units will be made as soon as practicable after the date(s) determined by the Administrator and set forth in the Award Agreement. The Administrator, in its sole discretion, may only settle earned Restricted Stock Units in cash, Shares, or a combination of both.
 - (e) Cancellation. On the date set forth in the Award Agreement, all unearned Restricted Stock Units will be forfeited to the Company.

9. Stock Appreciation Rights.

- (a) <u>Grant of Stock Appreciation Rights</u>. Subject to the terms and conditions of the Plan, a Stock Appreciation Right may be granted to Service Providers at any time and from time to time as will be determined by the Administrator, in its sole discretion.
 - (b) Number of Shares. The Administrator will have complete discretion to determine the number of Stock Appreciation Rights granted to any Service Provider.

- (c) <u>Exercise Price and Other Terms</u>. The per share exercise price for the Shares to be issued pursuant to exercise of a Stock Appreciation Right will be determined by the Administrator and will be no less than one hundred percent (100%) of the Fair Market Value per Share on the date of grant. Otherwise, subject to Section 6(a) of the Plan, the Administrator, subject to the provisions of the Plan, will have complete discretion to determine the terms and conditions of Stock Appreciation Rights granted under the Plan.
- (d) <u>Stock Appreciation Right Agreement</u>. Each Stock Appreciation Right grant will be evidenced by an Award Agreement that will specify the exercise price, the term of the Stock Appreciation Right, the conditions of exercise, and such other terms and conditions as the Administrator, in its sole discretion, will determine.
- (e) Expiration of Stock Appreciation Rights. A Stock Appreciation Right granted under the Plan will expire upon the date determined by the Administrator, in its sole discretion, and set forth in the Award Agreement. Notwithstanding the foregoing, the rules of Section 6(d) also will apply to Stock Appreciation Rights.
 - (f) Payment of Stock Appreciation Right Amount. Upon exercise of a Stock Appreciation Right, a Participant will be entitled to receive payment from the Company in an amount determined by multiplying:
 - (i) The difference between the Fair Market Value of a Share on the date of exercise over the exercise price; times
 - (ii) The number of Shares with respect to which the Stock Appreciation Right is exercised.

At the discretion of the Administrator, the payment upon Stock Appreciation Right exercise may be in cash, in Shares of equivalent value, or in some combination thereof.

10. Performance Units and Performance Shares.

- (a) <u>Grant of Performance Units/Shares</u>. Performance Units and Performance Shares may be granted to Service Providers at any time and from time to time, as will be determined by the Administrator, in its sole discretion. The Administrator will have complete discretion in determining the number of Performance Units and Performance Shares granted to each Participant.
- (b) <u>Value of Performance Units/Shares</u>. Each Performance Unit will have an initial value that is established by the Administrator on or before the date of grant. Each Performance Share will have an initial value equal to the Fair Market Value of a Share on the date of grant.
- (c) <u>Performance Objectives and Other Terms</u>. The Administrator will set performance objectives or other vesting provisions (including, without limitation, continued status as a Service Provider) in its discretion which, depending on the extent to which they are met, will determine the number or value of Performance Units/Shares that will be paid out to the Service Providers. The time period during which the performance objectives or other vesting provisions must be met will be called the "Performance Period." Each Award of Performance Units/Shares will

be evidenced by an Award Agreement that will specify the Performance Period, and such other terms and conditions as the Administrator, in its sole discretion, will determine. The Administrator may set performance objectives based upon the achievement of Company-wide, divisional, or individual goals, applicable federal or state securities laws, or any other basis determined by the Administrator in its discretion.

- (d) <u>Earning of Performance Units/Shares</u>. After the applicable Performance Period has ended, the holder of Performance Units/Shares will be entitled to receive a payout of the number of Performance Units/Shares earned by the Participant over the Performance Period, to be determined as a function of the extent to which the corresponding performance objectives or other vesting provisions have been achieved. After the grant of a Performance Unit/Share, the Administrator, in its sole discretion, may reduce or waive any performance objectives or other vesting provisions for such Performance Unit/Share.
- (e) Form and Timing of Payment of Performance Units/Shares. Payment of earned Performance Units/Shares will be made as soon as practicable after the expiration of the applicable Performance Period. The Administrator, in its sole discretion, may pay earned Performance Units/Shares in the form of cash, in Shares (which have an aggregate Fair Market Value equal to the value of the earned Performance Units/Shares at the close of the applicable Performance Period) or in a combination thereof.
- (f) <u>Cancellation of Performance Units/Shares</u>. On the date set forth in the Award Agreement, all unearned or unvested Performance Units/Shares will be forfeited to the Company, and again will be available for grant under the Plan.
- 11. Leaves of Absence/Transfer Between Locations. Unless the Administrator provides otherwise, vesting of Awards granted hereunder will be suspended during any unpaid leave of absence. A Service Provider will not cease to be an Employee in the case of (i) any leave of absence approved by the Company or (ii) transfers between locations of the Company or between the Company, its Parent, or any Subsidiary. For purposes of Incentive Stock Options, no such leave may exceed ninety (90) days, unless reemployment upon expiration of such leave is guaranteed by statute or contract. If reemployment upon expiration of a leave of absence approved by the Company is not so guaranteed, then three (3) months following the ninety-first (91st) day of such leave any Incentive Stock Option held by the Participant will cease to be treated as an Incentive Stock Option and will be treated for tax purposes as a Nonstatutory Stock Option.
- 12. <u>Transferability of Awards</u>. Unless determined otherwise by the Administrator, an Award may not be sold, pledged, assigned, hypothecated, transferred, or disposed of in any manner other than by will or by the laws of descent or distribution and may be exercised, during the lifetime of the Participant, only by the Participant. If the Administrator makes an Award transferable, such Award will contain such additional terms and conditions as the Administrator deems appropriate.
 - 13. Adjustments: Dissolution or Liquidation: Merger or Change in Control.
- (a) <u>Adjustments</u>. In the event that any dividend or other distribution (whether in the form of cash, Shares, other securities, or other property), recapitalization, stock split, reverse stock split, reorganization, merger, consolidation, split-up, spin-off, combination, repurchase, or

exchange of Shares or other securities of the Company, or other change in the corporate structure of the Company affecting the Shares occurs, the Administrator, in order to prevent diminution or enlargement of the benefits or potential benefits intended to be made available under the Plan, will adjust the number and class of Shares that may be delivered under the Plan and/or the number, class, and price of Shares covered by each outstanding Award, and the numerical Share limits in Section 3 of the Plan.

- (b) <u>Dissolution or Liquidation</u>. In the event of the proposed dissolution or liquidation of the Company, the Administrator will notify each Participant as soon as practicable prior to the effective date of such proposed transaction. To the extent it has not been previously exercised, an Award will terminate immediately prior to the consummation of such proposed action.
- (c) Change in Control. In the event of a merger or Change in Control, each outstanding Award will be treated as the Administrator determines, including, without limitation, that each Award be assumed or an equivalent option or right substituted by the successor corporation or a Parent or Subsidiary of the successor corporation. The Administrator will not be required to treat all Awards similarly in the transaction.

In the event that the successor corporation does not assume or substitute for the Award, the Participant will fully vest in and have the right to exercise all of his or her outstanding Options and Stock Appreciation Rights, including Shares as to which such Awards would not otherwise be vested or exercisable, all restrictions on Restricted Stock and Restricted Stock Units will lapse, and, with respect to Awards with performance-based vesting, all performance goals or other vesting criteria will be deemed achieved at one hundred percent (100%) of target levels and all other terms and conditions met. In addition, if an Option or Stock Appreciation Right is not assumed or substituted in the event of a Change in Control, the Administrator will notify the Participant in writing or electronically that the Option or Stock Appreciation Right will be exercisable for a period of time determined by the Administrator in its sole discretion, and the Option or Stock Appreciation Right will terminate upon the expiration of such period.

For the purposes of this subsection (c), an Award will be considered assumed if, following the Change in Control, the Award confers the right to purchase or receive, for each Share subject to the Award immediately prior to the Change in Control, the consideration (whether stock, cash, or other securities or property) received in the Change in Control by holders of Common Stock for each Share held on the effective date of the transaction (and if holders were offered a choice of consideration, the type of consideration chosen by the holders of a majority of the outstanding Shares); provided, however, that if such consideration received in the Change in Control is not solely common stock of the successor corporation or its Parent, the Administrator may, with the consent of the successor corporation, provide for the consideration to be received upon the exercise of an Option or Stock Appreciation Right or upon the payout of a Restricted Stock Unit, Performance Unit or Performance Share, for each Share subject to such Award, to be solely common stock of the successor corporation or its Parent equal in fair market value to the per share consideration received by holders of Common Stock in the Change in Control.

Notwithstanding anything in this Section 13(c) to the contrary, an Award that vests, is earned or paid-out upon the satisfaction of one or more performance goals will not be considered assumed if the Company or its successor modifies any of such performance goals without the

Participant's consent; provided, however, a modification to such performance goals only to reflect the successor corporation's post-Change in Control corporate structure will not be deemed to invalidate an otherwise valid Award assumption.

(d) <u>Outside Director Awards</u>. With respect to Awards granted to an Outside Director that are assumed or substituted for, if on the date of or following such assumption or substitution the Participant's status as a Director or a director of the successor corporation, as applicable, is terminated other than upon a voluntary resignation by the Participant (unless such resignation is at the request of the acquirer), then the Participant will fully vest in and have the right to exercise Options and/or Stock Appreciation Rights as to all of the Shares underlying such Award, including those Shares which would not otherwise be vested or exercisable, all restrictions on Restricted Stock and Restricted Stock Units will lapse, and, with respect to Performance Units and Performance Shares, all performance goals or other vesting criteria will be deemed achieved at one hundred percent (100%) of target levels and all other terms and conditions met.

14. Tax Withholding.

- (a) Withholding Requirements. Prior to the delivery of any Shares or cash pursuant to an Award (or exercise thereof), the Company will have the power and the right to deduct or withhold, or require a Participant to remit to the Company, an amount sufficient to satisfy federal, state, local, foreign or other taxes (including the Participant's FICA obligation) required to be withheld with respect to such Award (or exercise thereof).
- (b) Withholding Arrangements. The Administrator, in its sole discretion and pursuant to such procedures as it may specify from time to time, may permit a Participant to satisfy such tax withholding obligation, in whole or in part by (without limitation) (a) paying cash, (b) electing to have the Company withhold otherwise deliverable cash or Shares having a Fair Market Value equal to the minimum statutory amount required to be withheld, or (c) delivering to the Company already-owned Shares having a Fair Market Value equal to the minimum statutory amount required to be withheld. The Fair Market Value of the Shares to be withheld or delivered will be determined as of the date that the taxes are required to be withheld.
- 15. No Effect on Employment or Service. Neither the Plan nor any Award will confer upon a Participant any right with respect to continuing the Participant's relationship as a Service Provider with the Company, nor will they interfere in any way with the Participant's right or the Company's right to terminate such relationship at any time, with or without cause, to the extent permitted by Applicable Laws.
- 16. <u>Date of Grant</u>. The date of grant of an Award will be, for all purposes, the date on which the Administrator makes the determination granting such Award, or such other later date as is determined by the Administrator. Notice of the determination will be provided to each Participant within a reasonable time after the date of such grant.

- 17. Term of Plan. Subject to Section 21 of the Plan, the Plan will become effective upon its adoption by the Board. It will continue in effect for a term of ten (10) years from the date adopted by the Board, unless terminated earlier under Section 18 of the Plan.
 - 18. Amendment and Termination of the Plan.
 - (a) Amendment and Termination. The Board may at any time amend, alter, suspend or terminate the Plan.
 - (b) Stockholder Approval. The Company will obtain stockholder approval of any Plan amendment to the extent necessary and desirable to comply with Applicable Laws.
- (c) Effect of Amendment or Termination. No amendment, alteration, suspension or termination of the Plan will impair the rights of any Participant, unless mutually agreed otherwise between the Participant and the Administrator, which agreement must be in writing and signed by the Participant and the Company. Termination of the Plan will not affect the Administrator's ability to exercise the powers granted to it hereunder with respect to Awards granted under the Plan prior to the date of such termination.

19. Conditions Upon Issuance of Shares.

- (a) <u>Legal Compliance</u>. Shares will not be issued pursuant to the exercise of an Award unless the exercise of such Award and the issuance and delivery of such Shares will comply with Applicable Laws and will be further subject to the approval of counsel for the Company with respect to such compliance.
- (b) <u>Investment Representations</u>. As a condition to the exercise of an Award, the Company may require the person exercising such Award to represent and warrant at the time of any such exercise that the Shares are being purchased only for investment and without any present intention to sell or distribute such Shares if, in the opinion of counsel for the Company, such a representation is required.
- 20. <u>Inability to Obtain Authority</u>. The inability of the Company to obtain authority from any regulatory body having jurisdiction, which authority is deemed by the Company's counsel to be necessary to the lawful issuance and sale of any Shares hereunder, will relieve the Company of any liability in respect of the failure to issue or sell such Shares as to which such requisite authority will not have been obtained.
- 21. <u>Stockholder Approval</u>. The Plan will be subject to approval by the stockholders of the Company within twelve (12) months after the date the Plan is adopted by the Board. Such stockholder approval will be obtained in the manner and to the degree required under Applicable Laws.

COMSCORE, INC.

2007 EQUITY INCENTIVE PLAN

NOTICE OF GRANT OF STOCK OPTION

Unless otherwise defined herein, the terms defined in the 2007 Equity Incentive Plan (the "Plan") will have the same defined meanings in this Notice of Grant of Stock Option (the "Notice of Grant") and Terms and Conditions of Stock Option Grant, attached hereto as Exhibit A (together, the "Agreement").

Participant:	
Address:	
Participant has been granted an Option to purchase	Common Stock of the Company, subject to the terms and conditions of the Plan and this Agreement, as follows:
Grant Number	
Date of Grant	
Vesting Commencement Date	
Number of Shares Granted	
Exercise Price per Share	\$
Total Exercise Price	\$
Type of Option	Incentive Stock Option
	Nonstatutory Stock Option
Term/Expiration Date	

Vesting Schedule:

Subject to accelerated vesting as set forth below or in the Plan, this Option will be exercisable, in whole or in part, in accordance with the following schedule:

[Twenty-five percent (25%) of the Shares subject to the Option will vest twelve (12) months after the Vesting Commencement Date, and one forty-eighth (1/48th) of the Shares subject to the Option will vest each month thereafter on the same day of the month as the Vesting Commencement Date (and if there is no corresponding day, on the last day of the month), subject to Participant continuing to be a Service Provider through each such date.]

Termination Period:

This Option will be exercisable for [three (3) months] after Participant ceases to be a Service Provider, unless such termination is due to Participant's death or Disability, in which case this Option will be exercisable for [twelve (12) months] after Participant ceases to be a Service Provider. Notwithstanding the foregoing, in no event may this Option be exercised after the Term/Expiration Date as provided above and may be subject to earlier termination as provided in Section 13(c) of the Plan.

By Participant's signature and the signature of the Company's representative below, Participant and the Company agree that this Option is granted under and governed by the terms and conditions of the Plan and this Agreement. Participant has reviewed the Plan and this Agreement in their entirety, has had an opportunity to obtain the advice of counsel prior to executing this Agreement and fully understands all provisions of the Plan and Agreement. Participant hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions relating to the Plan and Agreement. Participant further agrees to notify the Company upon any change in the residence address indicated below.

PARTICIPANT	COMSCORE, INC.	
Signature	By	
Print Name	Title	
Address:		
	-	
	-	

EXHIBIT A

TERMS AND CONDITIONS OF STOCK OPTION GRANT

1. <u>Grant</u>. The Company hereby grants to the Participant named in the Notice of Grant (the "Participant") an option (the "Option") to purchase the number of Shares, as set forth in the Notice of Grant (the "Exercise Price"), subject to all of the terms and conditions in this Agreement and the Plan, which is incorporated herein by reference. Subject to Section 18(c) of the Plan, in the event of a conflict between the terms and conditions of the Plan and the terms and conditions of the Plan will prevail.

If designated in the Notice of Grant as an Incentive Stock Option ("ISO"), this Option is intended to qualify as an Incentive Stock Option under Section 422 of the Code. However, if this Option is intended to be an Incentive Stock Option, to the extent that it exceeds the \$100,000 rule of Code Section 422(d) it will be treated as a Nonstatutory Stock Option ("NSO").

- 2. <u>Vesting Schedule</u>. Except as provided in Section 3, the Option awarded by this Agreement will vest in accordance with the vesting provisions set forth in the Notice of Grant. Shares scheduled to vest on a certain date or upon the occurrence of a certain condition will not vest in Participant in accordance with any of the provisions of this Agreement, unless Participant will have been continuously a Service Provider from the Date of Grant until the date such vesting occurs.
- 3. <u>Administrator Discretion</u>. The Administrator, in its discretion, may accelerate the vesting of the balance, or some lesser portion of the balance, of the unvested Option at any time, subject to the terms of the Plan. If so accelerated, such Option will be considered as having vested as of the date specified by the Administrator.
- 4. Exercise of Option. This Option may be exercised only within the term set out in the Notice of Grant, and may be exercised during such term only in accordance with the Plan and the terms of this Agreement.

This Option is exercisable by delivery of an exercise notice, in the form attached as Exhibit B (the "Exercise Notice") or in a manner and pursuant to such procedures as the Administrator may determine, which will state the election to exercise the Option, the number of Shares in respect of which the Option is being exercised (the "Exercised Shares"), and such other representations and agreements as may be required by the Company pursuant to the provisions of the Plan. The Exercise Notice will be completed by Participant and delivered to the Company. The Exercise Notice will be accompanied by payment of the aggregate Exercise Price as to all Exercised Shares together with any applicable tax withholding. This Option will be deemed to be exercised upon receipt by the Company of such fully executed Exercise Notice accompanied by such aggregate Exercise Price.

5. Method of Payment. Payment of the aggregate Exercise Price will be by any of the following, or a combination thereof, at the election of Participant:

(a) cash;

- (b) check;
- (c) consideration received by the Company under a formal cashless exercise program adopted by the Company in connection with the Plan; or
- (d) surrender of other Shares which have a Fair Market Value on the date of surrender equal to the aggregate Exercise Price of the Exercised Shares, provided that accepting such Shares, in the sole discretion of the Administrator, will not result in any adverse accounting consequences to the Company.

6. Tax Obligations

- (a) Withholding of Taxes. Notwithstanding any contrary provision of this Agreement, no certificate representing the Shares will be issued to Participant, unless and until satisfactory arrangements (as determined by the Administrator) will have been made by Participant with respect to the payment of income, employment and other taxes which the Company determines must be withheld with respect to such Shares. To the extent determined appropriate by the Company in its discretion, it will have the right (but not the obligation) to satisfy any tax withholding obligations by reducing the number of Shares otherwise deliverable to Participant. If Participant fails to make satisfactory arrangements for the payment of any required tax withholding obligations hereunder at the time of the Option exercise, Participant acknowledges and agrees that the Company may refuse to honor the exercise and refuse to deliver Shares if such withholding amounts are not delivered at the time of exercise.
- (b) Notice of Disqualifying Disposition of ISO Shares. If the Option granted to Participant herein is an ISO, and if Participant sells or otherwise disposes of any of the Shares acquired pursuant to the ISO on or before the later of (i) the date two (2) years after the Grant Date, or (ii) the date one (1) year after the date of exercise, Participant will immediately notify the Company in writing of such disposition. Participant agrees that Participant may be subject to income tax withholding by the Company on the compensation income recognized by Participant.
- (c) <u>Code Section 409A</u>. Under Code Section 409A, an option that vests after December 31, 2004 that was granted with a per Share exercise price that is determined by the Internal Revenue Service (the "IRS") to be less than the Fair Market Value of a Share on the date of grant (a "Discount Option") may be considered "deferred compensation." A Discount Option may result in (i) income recognition by Participant prior to the exercise of the option, (ii) an additional twenty percent (20%) federal income tax, and (iii) potential penalty and interest charges. The Discount Option may also result in additional state income, penalty and interest charges to the Participant acknowledges that the Company cannot and has not guaranteed that the IRS will agree that the per Share exercise price of this Option equals or exceeds the Fair Market Value of a Share on the Date of Grant in a later examination. Participant agrees that if the IRS determines that the Option was granted with a per Share exercise price that was less than the Fair Market Value of a Share on the date of grant, Participant will be solely responsible for Participant's costs related to such a determination.

- 7. <u>Rights as Stockholder</u>. Neither Participant nor any person claiming under or through Participant will have any of the rights or privileges of a stockholder of the Company in respect of any Shares deliverable hereunder unless and until certificates representing such Shares will have been issued, recorded on the records of the Company or its transfer agents or registrars, and delivered to Participant. After such issuance, recordation and delivery, Participant will have all the rights of a stockholder of the Company with respect to voting such Shares and receipt of dividends and distributions on such Shares.
- 8. No Guarantee of Continued Service. PARTICIPANT ACKNOWLEDGES AND AGREES THAT THE VESTING OF SHARES PURSUANT TO THE VESTING SCHEDULE HEREOF IS EARNED ONLY BY CONTINUING AS A SERVICE PROVIDER AT THE WILL OF THE COMPANY (OR THE PARENT OR SUBSIDIARY EMPLOYING OR RETAINING PARTICIPANT) AND NOT THROUGH THE ACT OF BEING HIRED, BEING GRANTED THE OPTION OR ACQUIRING SHARES HEREUNDER. PARTICIPANT FURTHER ACKNOWLEDGES AND AGREES THAT THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREUNDER AND THE VESTING SCHEDULE SET FORTH HEREIN DO NOT CONSTITUTE AN EXPRESS OR IMPLIED PROMISE OF CONTINUED ENGAGEMENT AS A SERVICE PROVIDER FOR THE VESTING PERIOD, FOR ANY PERIOD, OR AT ALL, AND WILL NOT INTERFERE IN ANY WAY WITH PARTICIPANT'S RIGHT OR THE RIGHT OF THE COMPANY (OR THE PARENT OR SUBSIDIARY EMPLOYING OR RETAINING PARTICIPANT) TO TERMINATE PARTICIPANT'S RELATIONSHIP AS A SERVICE PROVIDER AT ANY TIME, WITH OR WITHOUT CAUSE.
- 9. <u>Address for Notices</u>. Any notice to be given to the Company under the terms of this Agreement will be addressed to the Company at comScore, Inc., 11465 Sunset Hills Road, Suite 200, Reston, Virginia 20190, or at such other address as the Company may hereafter designate in writing.
- 10. <u>Grant is Not Transferable</u>. This Option may not be transferred in any manner otherwise than by will or by the laws of descent or distribution and may be exercised during the lifetime of Participant only by Participant.
- 11. <u>Binding Agreement</u>. Subject to the limitation on the transferability of this grant contained herein, this Agreement will be binding upon and inure to the benefit of the heirs, legatees, legal representatives, successors and assigns of the parties hereto.
- 12. <u>Additional Conditions to Issuance of Stock</u>. If at any time the Company will determine, in its discretion, that the listing, registration or qualification of the Shares upon any securities exchange or under any state or federal law, or the consent or approval of any governmental regulatory authority is necessary or desirable as a condition to the issuance of Shares to Participant (or his or her estate), such issuance will not occur unless and until such listing, registration, qualification, consent or approval will have been effected or obtained free of any conditions not acceptable to the Company. The Company will make all reasonable efforts to meet the requirements of any such state or federal law or securities exchange and to obtain any such consent or approval of any such governmental authority. Assuming such compliance, for income tax purposes the

Exercised Shares will be considered transferred to Participant on the date the Option is exercised with respect to such Exercised Shares.

- 13. <u>Plan Governs</u>. This Agreement is subject to all terms and provisions of the Plan. In the event of a conflict between one or more provisions of this Agreement and one or more provisions of the Plan will govern. Capitalized terms used and not defined in this Agreement will have the meaning set forth in the Plan.
- 14. <u>Administrator Authority</u>. The Administrator will have the power to interpret the Plan and this Agreement and to adopt such rules for the administration, interpretation and application of the Plan as are consistent therewith and to interpret or revoke any such rules (including, but not limited to, the determination of whether or not any Shares subject to the Option have vested). All actions taken and all interpretations and determinations made by the Administrator in good faith will be final and binding upon Participant, the Company and all other interested persons. No member of the Administrator will be personally liable for any action, determination or interpretation made in good faith with respect to the Plan or this Agreement.
- 15. <u>Electronic Delivery</u>. The Company may, in its sole discretion, decide to deliver any documents related to Options awarded under the Plan or future Options that may be awarded under the Plan by electronic means or request Participant's consent to participate in the Plan by electronic means. Participant hereby consents to receive such documents by electronic delivery and agrees to participate in the Plan through any on-line or electronic system established and maintained by the Company or another third party designated by the Company.
 - 16. Captions. Captions provided herein are for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.
- 17. <u>Agreement Severable</u>. In the event that any provision in this Agreement will be held invalid or unenforceable, such provision will be severable from, and such invalidity or unenforceability will not be construed to have any effect on, the remaining provisions of this Agreement.
- 18. <u>Modifications to the Agreement</u>. This Agreement constitutes the entire understanding of the parties on the subjects covered. Participant expressly warrants that he or she is not accepting this Agreement in reliance on any promises, representations, or inducements other than those contained herein. Modifications to this Agreement or the Plan can be made only in an express written contract executed by a duly authorized officer of the Company.
- 19. <u>Amendment, Suspension or Termination of the Plan</u>. By accepting this Award, Participant expressly warrants that he or she has received an Option under the Plan, and has received, read and understood a description of the Plan. Participant understands that the Plan is discretionary in nature and may be amended, suspended or terminated by the Company at any time.
- 20. Governing Law. This Agreement will be governed by the laws of the Commonwealth of Virginia, without giving effect to the conflict of law principles thereof. For purposes of litigating any dispute that arises under this Option or this Agreement, the parties hereby submit to and consent

to the jurisdiction of the Commonwealth of Virginia, and agree that such litigation will be conducted in the courts of Fairfax County, Virginia, or the federal courts for the United States for the Eastern District of Virginia, and no other courts, where this Option is made and/or to be performed.

- 7 -

EXHIBIT B

COMSCORE, INC.

2007 EQUITY INCENTIVE PLAN EXERCISE NOTICE

comScore, Inc. 11465 Sunset Hills Road Suite 200 Reston, Virginia 20190

Reston, Virginia 20190
Attention:
1. Exercise of Option. Effective as of today,, the undersigned ("Purchaser") hereby elects to purchaseshares (the "Shares") of the Common Stock of comScore, Inc. (the "Company") under and pursuant to the 2007 Equity Incentive Plan (the "Plan") and the Stock Option Agreement dated(the "Agreement"). The purchase price for the Shares will be \$, as required by the Agreement.
2. Delivery of Payment. Purchaser herewith delivers to the Company the full purchase price of the Shares and any required tax withholding to be paid in connection with the exercise of the Option.
3. Representations of Purchaser. Purchaser acknowledges that Purchaser has received, read and understood the Plan and the Agreement and agrees to abide by and be bound by their terms and conditions.
4. <u>Rights as Stockholder</u> . Until the issuance (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company) of the Shares, no right to vote or receive dividends or any other rights as a stockholder will exist with respect to the Optioned Stock, notwithstanding the exercise of the Option. The Shares so acquired will be issued to Participant as soon as practicable after exercise of the Option. No adjustment will be made for a dividend or other right for which the record date is prior to the date of issuance, except as provided in Section 13 of the Plan.
5. <u>Tax Consultation</u> . Purchaser understands that Purchaser may suffer adverse tax consequences as a result of Purchaser's purchase or disposition of the Shares. Purchaser represents that Purchaser has consulted with any tax consultants Purchaser deems advisable in connection with the purchase or disposition of the Shares and that Purchaser is not relying on the Company for any tax advice.
6. Entire Agreement; Governing Law. The Plan and Agreement are incorporated herein by reference. This Exercise Notice, the Plan and the Agreement constitute the entire agreement of

- 8

the parties with respect to the subject matter hereof and supersede in their entirety all prior undertakings and agreements of the Company and Purchaser with respect to the subject matter hereof, and may not be modi adversely to the Purchaser's interest except by means of a writing signed by the Company and Purchaser. This agreement is governed by the internal substantive laws, but not the choice of law rules, of the Commonwealth of Virginia.		
Submitted by:	Accepted by:	

PURCHASER	COMSCORE, INC.	
Signature	Ву	
Print Name	Its	
Address:		
	Date Received	

COMSCORE, INC.

2007 EQUITY INCENTIVE PLAN

NOTICE OF GRANT OF RESTRICTED STOCK

Unless otherwise defined herein, the terms defined in the 2007 Equity Incentive Plan (the "Plan") will have the same defined meanings in this Notice of Grant of Restricted Stock (the "Notice of Grant") and Terms and Conditions of Restricted Stock Grant, attached hereto as Exhibit A (together, the "Agreement").

Participant:	_	-
Address:		
Participant has been granted the right to receive an Award of Restricted Stock, subject	ct to the terms and conditions of the Plan and this Agreement, as f	ollows:
Grant Number		
Date of Grant		.
Vesting Commencement Date		-
Number of Shares Granted		•
[Exercise Price Per Share	<u>\$</u>]	
Term/Expiration Date		

Vesting Schedule:

Subject to any acceleration provisions contained in the Plan or set forth below, the Restricted Stock will vest and the Company's right to repurchase the Restricted Stock will lapse in accordance with the following schedule:

[VESTING SCHEDULE]

[PARTICIPANT MUST PURCHASE THE SHARES BEFORE THE EXPIRATION DATE OR THE RESTRICTED STOCK AWARD WILL TERMINATE AND PARTICIPANT WILL HAVE NO FURTHER RIGHT TO PURCHASE THE SHARES.]

of the Plan and this Agreement. Participant has reviewed the Plan and this Agreement in their entirety,	the Company agree that this Award of Restricted Stock is granted under and governed by the terms and conditions has had an opportunity to obtain the advice of counsel prior to executing this Agreement and fully understands all nal all decisions or interpretations of the Administrator upon any questions relating to the Plan and Agreement. low.
PARTICIPANT	COMSCORE, INC.
Signature	Ву
Print Name	Title
Address:	
	•
	•
	-2-

EXHIBIT A

TERMS AND CONDITIONS OF RESTRICTED STOCK GRANT

1. <u>Purchase of Stock</u>. The Company hereby agrees to sell to the Participant named in the Notice of Grant (the "Participant") and Participant hereby agrees to purchase the number of Shares (the "Restricted Stock"), at the per Share purchase price and as otherwise described in the Notice of Grant, subject to all of the terms and conditions in this Agreement and the Plan, which is incorporated herein by reference. Subject to Section 18(c) of the Plan, in the event of a conflict between the terms and conditions of the Plan will prevail. The purchase price for the Restricted Stock, if any, may be paid by delivery to the Company at the time of execution of this Agreement in cash, a check, or some combination thereof, together with any applicable tax withholding.

OR

Grant of Restricted Stock. The Company hereby grants to the Participant named in the Notice of Grant (the "Participant") under the Plan for past services and as a separate incentive in connection with his or her services and not in lieu of any salary or other compensation for his or her services, the number of Shares (the "Restricted Stock"), at the per Share purchase price and as otherwise described in the Notice of Grant, subject to all of the terms and conditions in this Agreement and the Plan, which is incorporated herein by reference. Subject to Section 18(c) of the Plan, in the event of a conflict between the terms and conditions of the Plan and the terms and conditions of this Agreement, the terms and conditions of the Plan will prevail.

Escrow of Shares

- (a) All Shares of Restricted Stock will, upon execution of this Agreement, be delivered and deposited with an escrow holder designated by the Company (the "Escrow Holder"). The Shares of Restricted Stock will be held by the Escrow Holder until such time as the Shares of Restricted Stock vest or the date Participant ceases to be a Service Provider.
 - (b) The Escrow Holder will not be liable for any act it may do or omit to do with respect to holding the Shares of Restricted Stock in escrow while acting in good faith and in the exercise of its judgment.
- (c) Upon Participant's termination as a Service Provider for any reason, the Escrow Holder, upon receipt of written notice of such termination, will take all steps necessary to accomplish the transfer of the unvested Shares of Restricted Stock to the Company. Participant hereby appoints the Escrow Holder with full power of substitution, as Participant's true and lawful attorney-in-fact with irrevocable power and authority in the name and on behalf of Participant to take any action and execute all documents and instruments, including, without limitation, stock powers which may be necessary to transfer the certificate or certificates evidencing such unvested Shares of Restricted Stock to the Company upon such termination.

-3-

- (d) The Escrow Holder will take all steps necessary to accomplish the transfer of Shares of Restricted Stock to Participant after they vest following Participant's request that the Escrow Holder do so.
- (e) Subject to the terms hereof, Participant will have all the rights of a stockholder with respect to the Shares while they are held in escrow, including without limitation, the right to vote the Shares and to receive any cash dividends declared thereon.
- (f) In the event of any dividend or other distribution (whether in the form of cash, Shares, other securities, or other property), recapitalization, stock split, reverse stock split, reorganization, merger, consolidation, split-up, spin-off, combination, repurchase, or exchange of Shares or ther securities of the Company, or other change in the corporate structure of the Company affecting the Shares, the Shares of Restricted Stock will be increased, reduced or otherwise changed, and by virtue of any such change Participant will in his or her capacity as owner of unvested Shares of Restricted Stock be entitled to new or additional or different shares, cash or securities (other than rights or warrants to purchase securities); such new or additional or different shares, cash or securities will thereupon be considered to be unvested Shares of Restricted Stock and will be subject to all of the conditions and restrictions which were applicable to the unvested Shares of Restricted Stock, such rights or warrants may be held or exercised by Participant, provided that until such exercise any such rights or warrants and after such exercise any shares or other securities acquired by the exercise of such rights or warrants will be considered to be unvested Shares of Restricted Stock and will be subject to all of the conditions and restrictions which were applicable to the unvested Shares of Restricted Stock pursuant to this Agreement. The Administrator in its absolute discretion at any time may accelerate the vesting of all or any portion of such new or additional shares of stock, cash or securities, rights or warrants to purchase securities or shares or other securities acquired by the exercise of such rights or warrants.
- (g) The Company may instruct the transfer agent for its Common Stock to place a legend on the certificates representing the Restricted Stock or otherwise note its records as to the restrictions on transfer set forth in this Agreement.
- 3. <u>Vesting Schedule</u>. Except as provided in Section 4, and subject to Section 5, the Shares of Restricted Stock awarded by this Agreement will vest in accordance with the vesting provisions set forth in the Notice of Grant. Shares of Restricted Stock scheduled to vest on a certain date or upon the occurrence of a certain condition will not vest in Participant in accordance with any of the provisions of this Agreement, unless Participant will have been continuously a Service Provider from the Date of Grant until the date such vesting occurs.
- 4. <u>Administrator Discretion</u>. The Administrator, in its discretion, may accelerate the vesting of the balance, or some lesser portion of the balance, of the unvested Restricted Stock at any time, subject to the terms of the Plan. If so accelerated, such Restricted Stock will be considered as having vested as of the date specified by the Administrator.
 - 5. Forfeiture upon Termination of Status as a Service Provider. Notwithstanding any contrary provision of this Agreement, the balance of the Shares of Restricted Stock that have

not vested at the time of Participant's termination as a Service Provider for any reason will be forfeited and automatically transferred to and reacquired by the Company at no cost to the Company upon the date of such termination and Participant will have no further rights thereunder. Participant will not be entitled to a refund of the price paid for the Shares of Restricted Stock, if any, returned to the Company pursuant to this Section 5. Participant hereby appoints the Escrow Agent with full power of substitution, as Participant's true and lawful attorney-in-fact with irrevocable power and authority in the name and on behalf of Participant to take any action and execute all documents and instruments, including, without limitation, stock powers which may be necessary to transfer the certificate or certificates evidencing such unvested Shares to the Company upon such termination of service.

- 6. <u>Death of Participant</u>. Any distribution or delivery to be made to Participant under this Agreement will, if Participant is then deceased, be made to Participant's designated beneficiary, or if no beneficiary survives Participant, the administrator or executor of Participant's estate. Any such transferee must furnish the Company with (a) written notice of his or her status as transferee, and (b) evidence satisfactory to the Company to establish the validity of the transfer and compliance with any laws or regulations pertaining to said transfer.
- 7. Withholding of Taxes. Notwithstanding any contrary provision of this Agreement, no certificate representing the Shares of Restricted Stock may be released from the escrow established pursuant to Section 5, unless and until satisfactory arrangements (as determined by the Administrator) will have been made by Participant with respect to the payment of income, employment and other taxes which the Company determines must be withheld with respect to such Shares. To the extent determined appropriate by the Company in its discretion, it will have the right (but not the obligation) to satisfy any tax withholding obligations by reducing the number of Shares otherwise deliverable to Participant. If Participant fails to make satisfactory arrangements for the payment of any required tax withholding obligations bereunder at the time any applicable Shares otherwise are scheduled to vest pursuant to Sections 3 or 4, Participant will permanently forfeit such Shares and the Shares will be returned to the Company at no cost to the Company.
- 8. <u>Rights as Stockholder</u>. Neither Participant nor any person claiming under or through Participant will have any of the rights or privileges of a stockholder of the Company in respect of any Shares deliverable hereunder unless and until certificates representing such Shares will have been issued, recorded on the records of the Company or its transfer agents or registrars, and delivered to Participant or the Escrow Agent. Except as provided in Section 2(f), after such issuance, recordation and delivery, Participant will have all the rights of a stockholder of the Company with respect to voting such Shares and receipt of dividends and distributions on such Shares.
- 9. No Guarantee of Continued Service. PARTICIPANT ACKNOWLEDGES AND AGREES THAT THE VESTING OF THE SHARES OF RESTRICTED STOCK PURSUANT TO THE VESTING SCHEDULE HEREOF IS EARNED ONLY BY CONTINUING AS A SERVICE PROVIDER AT THE WILL OF THE COMPANY (OR THE PARENT OR SUBSIDIARY EMPLOYING OR RETAINING PARTICIPANT) AND NOT THROUGH THE ACT OF BEING HIRED, BEING GRANTED THIS RESTRICTED STOCK OR ACQUIRING SHARES HEREUNDER. PARTICIPANT FURTHER ACKNOWLEDGES AND AGREES

THAT THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREUNDER AND THE VESTING SCHEDULE SET FORTH HEREIN DO NOT CONSTITUTE AN EXPRESS OR IMPLIED PROMISE OF CONTINUED ENGAGEMENT AS A SERVICE PROVIDER FOR THE VESTING PERIOD, FOR ANY PERIOD, OR AT ALL, AND WILL NOT INTERFERE IN ANY WAY WITH PARTICIPANT'S RIGHT OR THE RIGHT OF THE COMPANY (OR THE PARENT OR SUBSIDIARY EMPLOYING OR RETAINING PARTICIPANT) TO TERMINATE PARTICIPANT'S RELATIONSHIP AS A SERVICE PROVIDER AT ANY TIME, WITH OR WITHOUT CAUSE.

- 10. Address for Notices. Any notice to be given to the Company under the terms of this Agreement will be addressed to the Company at comScore, Inc., 11465 Sunset Hill Road, Suite 200, Reston, Virginia 20190, or at such other address as the Company may hereafter designate in writing.
- 11. <u>Grant is Not Transferable</u>. Except to the limited extent provided in Section 6, the unvested Shares subject to this grant and the rights and privileges conferred hereby will not be transferred, assigned, pledged or hypothecated in any way (whether by operation of law or otherwise) and will not be subject to sale under execution, attachment or similar process. Upon any attempt to transfer, assign, pledge, hypothecate or otherwise dispose of any unvested Shares of Restricted Stock subject to this grant, or any right or privilege conferred hereby, or upon any attempted sale under any execution, attachment or similar process, this grant and the rights and privileges conferred hereby immediately will become null and void.
- 12. <u>Binding Agreement</u>. Subject to the limitation on the transferability of this grant contained herein, this Agreement will be binding upon and inure to the benefit of the heirs, legatees, legal representatives, successors and assigns of the parties hereto.
- 13. Additional Conditions to Release from Escrow. The Company will not be required to issue any certificate or certificates for Shares hereunder or release such Shares from the escrow established pursuant to Section 2 prior to fulfillment of all the following conditions: (a) the admission of such Shares to listing on all stock exchanges on which such class of stock is then listed; (b) the completion of any registration or other qualification of such Shares under any state or federal law or under the rulings or regulations of the Securities and Exchange Commission or any other governmental regulatory body, which the Administrator will, in its absolute discretion, deem necessary or advisable; (c) the obtaining of any approval or other clearance from any state or federal governmental agency, which the Administrator will, in its absolute discretion, determine to be necessary or advisable; and (d) the lapse of such reasonable period of time following the date of grant of the Restricted Stock as the Administrator may establish from time to time for reasons of administrative convenience.
- 14. <u>Plan Governs</u>. This Agreement is subject to all terms and provisions of the Plan. In the event of a conflict between one or more provisions of this Agreement and one or more provisions of the Plan will govern. Capitalized terms used and not defined in this Agreement will have the meaning set forth in the Plan.
 - 15. Administrator Authority. The Administrator will have the power to interpret the Plan and this Agreement and to adopt such rules for the administration, interpretation and

application of the Plan as are consistent therewith and to interpret or revoke any such rules (including, but not limited to, the determination of whether or not any Shares of Restricted Stock have vested). All actions taken and all interpretations and determinations made by the Administrator in good faith will be final and binding upon Participant, the Company and all other interested persons. No member of the Administrator will be personally liable for any action, determination or interpretation made in good faith with respect to the Plan or this Agreement.

- 16. <u>Electronic Delivery.</u> The Company may, in its sole discretion, decide to deliver any documents related to the Shares of Restricted Stock awarded under the Plan or future Restricted Stock that may be awarded under the Plan by electronic means or request Participant's consent to participate in the Plan by electronic means. Participant hereby consents to receive such documents by electronic delivery and agrees to participate in the Plan through any on-line or electronic system established and maintained by the Company or another third party designated by the Company.
 - 17. Captions. Captions provided herein are for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.
- 18. <u>Agreement Severable</u>. In the event that any provision in this Agreement will be held invalid or unenforceable, such provision will be severable from, and such invalidity or unenforceability will not be construed to have any effect on, the remaining provisions of this Agreement.
- 19. <u>Modifications to the Agreement</u>. This Agreement constitutes the entire understanding of the parties on the subjects covered. Participant expressly warrants that he or she is not accepting this Agreement in reliance on any promises, representations, or inducements other than those contained herein. Modifications to this Agreement or the Plan can be made only in an express written contract executed by a duly authorized officer of the Company. Notwithstanding anything to the contrary in the Plan or this Agreement, the Company reserves the right to revise this Agreement as it deems necessary or advisable, in its sole discretion and without the consent of Participant, to comply with Section 409A of the Internal Revenue Code of 1986, as amended (the "Code") or to otherwise avoid imposition of any additional tax or income recognition under Section 409A of the Code in connection to this Award of Restricted Stock.
- 20. <u>Amendment, Suspension or Termination of the Plan</u>. By accepting this Award, Participant expressly warrants that he or she has received an Award of Restricted Stock under the Plan, and has received, read and understood a description of the Plan. Participant understands that the Plan is discretionary in nature and may be amended, suspended or terminated by the Company at any time.
- 21. <u>Governing Law.</u> This Agreement will be governed by the laws of the Commonwealth of Virginia, without giving effect to the conflict of law principles thereof. For purposes of litigating any dispute that arises under this Award of Restricted Stock or this Agreement, the parties hereby submit to and consent to the jurisdiction of the Commonwealth of Virginia, and agree that such litigation will be conducted in the courts of Fairfax County, Virginia, or the federal courts for the United States for the Eastern District of Virginia, and no other courts, where this Award of Restricted Stock is made and/or to be performed.

COMSCORE, INC.

2007 EQUITY INCENTIVE PLAN

NOTICE OF GRANT OF RESTRICTED STOCK UNITS

Unless otherwise defined herein, the terms defined in the 2007 Equity Incentive Plan (the "Plan") will have the same defined meanings in this Notice of Grant of Restricted Stock Units (the "Notice of Grant") and Terms and Conditions of Restricted Stock Unit Grant, attached hereto as Exhibit A (together, the "Agreement").

Participant:		
Address:		
Participant has been granted the right to receive an Award of Restricted S	Stock Units, subject to the terms and conditions of the Plan ar	nd this Agreement, as fo
Grant Number		
Date of Grant		
Vesting Commencement Date		
Number of Restricted Stock Units		
Vesting Schedule:		

Subject to any acceleration provisions contained in the Plan or set forth below, the Restricted Stock Unit will vest in accordance with the following schedule:

[VESTING SCHEDULE.]

In the event Participant ceases to be a Service Provider for any or no reason before Participant vests in the Restricted Stock Unit, the Restricted Stock Unit and Participant's right to acquire any Shares hereunder will immediately terminate.

By Participant's signature and the signature of the Company's representative below, Participant and the Company agree that this Award of Restricted Stock Units is granted under and governed by the terms and conditions of the Plan and this Agreement. Participant has reviewed the Plan and this Agreement in their entirety, has had an opportunity to obtain the advice of counsel prior to executing this Agreement and fully understands all provisions of the Plan and Agreement. Participant hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions relating to the Plan and Agreement. Participant further agrees to notify the Company upon any change in the residence address indicated below.

PARTICIPANT	COMSCORE, INC.	
Signature	By	
Print Name	Title	
Address:		
		-2-

EXHIBIT A

TERMS AND CONDITIONS OF RESTRICTED STOCK UNIT GRANT

- 1. <u>Grant</u>. The Company hereby grants to the Participant named in the Notice of Grant (the "Participant") under the Plan an Award of Restricted Stock Units, subject to all of the terms and conditions in this Agreement and the Plan, which is incorporated herein by reference. Subject to Section 18(c) of the Plan, in the event of a conflict between the terms and conditions of the Plan and the terms and conditions of the Plan will prevail.
- 2. Company's Obligation to Pay. Each Restricted Stock Unit represents the right to receive a Share on the date it vests. Unless and until the Restricted Stock Units will have vested in the manner set forth in Section 3, Participant will have no right to payment of any such Restricted Stock Units. Prior to actual payment of any vested Restricted Stock Units, such Restricted Stock Unit will represent an unsecured obligation of the Company, payable (if at all) only from the general assets of the Company. Any Restricted Stock Units that vest in accordance with Sections 3 or 4 will be paid to Participant (or in the event of Participant's death, to his or her estate) in whole Shares, subject to Participant satisfying any applicable tax withholding obligations as set forth in Section 6.
- 3. <u>Vesting Schedule</u>. Except as provided in Section 4, and subject to Section 5, the Restricted Stock Units awarded by this Agreement will vest in accordance with the vesting provisions set forth in the Notice of Grant. Restricted Stock Units scheduled to vest on a certain date or upon the occurrence of a certain condition will not vest in Participant in accordance with any of the provisions of this Agreement, unless Participant will have been continuously a Service Provider from the Date of Grant until the date such vesting occurs.
- 4. <u>Administrator Discretion</u>. The Administrator, in its discretion, may accelerate the vesting of the balance, or some lesser portion of the balance, of the unvested Restricted Stock Units at any time, subject to the terms of the Plan. If so accelerated, such Restricted Stock Units will be considered as having vested as of the date specified by the Administrator.
- 5. <u>Forfeiture upon Termination of Status as a Service Provider</u>. Notwithstanding any contrary provision of this Agreement, the balance of the Restricted Stock Units that have not vested as of the time of Participant's termination as a Service Provider for any or no reason and Participant's right to acquire any Shares hereunder will immediately terminate.
- 6. <u>Death of Participant</u>. Any distribution or delivery to be made to Participant under this Agreement will, if Participant is then deceased, be made to Participant's designated beneficiary, or if no beneficiary survives Participant, the administrator or executor of Participant's estate. Any such transferee must furnish the Company with (a) written notice of his or her status as transferee, and (b) evidence satisfactory to the Company to establish the validity of the transfer and compliance with any laws or regulations pertaining to said transfer.
- 7. Withholding of Taxes. Notwithstanding any contrary provision of this Agreement, no certificate representing the Shares will be issued to Participant, unless and until satisfactory arrangements (as determined by the Administrator) will have been made by

Participant with respect to the payment of income, employment and other taxes which the Company determines must be withheld with respect to such Shares. To the extent determined appropriate by the Company in its discretion, it will have the right (but not the obligation) to satisfy any tax withholding obligations by reducing the number of Shares otherwise deliverable to Participant. If Participant fails to make satisfactory arrangements for the payment of any required tax withholding obligations hereunder at the time any applicable Restricted Stock Units otherwise are scheduled to vest pursuant to Sections 3 or 4, Participant will permanently forfeit such Restricted Stock Units and any right to receive Shares thereunder and the Restricted Stock Units will be returned to the Company at no cost to the Company.

- 8. <u>Rights as Stockholder</u>. Neither Participant nor any person claiming under or through Participant will have any of the rights or privileges of a stockholder of the Company in respect of any Shares deliverable hereunder unless and until certificates representing such Shares will have been issued, recorded on the records of the Company or its transfer agents or registrars, and delivered to Participant. After such issuance, recordation and delivery, Participant will have all the rights of a stockholder of the Company with respect to voting such Shares and receipt of dividends and distributions on such Shares.
- 9. No Guarantee of Continued Service. PARTICIPANT ACKNOWLEDGES AND AGREES THAT THE VESTING OF THE RESTRICTED STOCK UNITS PURSUANT TO THE VESTING SCHEDULE HEREOF IS EARNED ONLY BY CONTINUING AS A SERVICE PROVIDER AT THE WILL OF THE COMPANY (OR THE PARENT OR SUBSIDIARY EMPLOYING OR RETAINING PARTICIPANT) AND NOT THROUGH THE ACT OF BEING HIRED, BEING GRANTED THIS AWARD OF RESTRICTED STOCK UNITS OR ACQUIRING SHARES HEREUNDER. PARTICIPANT FURTHER ACKNOWLEDGES AND AGREES THAT THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREUNDER AND THE VESTING SCHEDULE SET FORTH HEREIN DO NOT CONSTITUTE AN EXPRESS OR IMPLIED PROMISE OF CONTINUED ENGAGEMENT AS A SERVICE PROVIDER FOR THE VESTING PERIOD, FOR ANY PERIOD, OR AT ALL, AND WILL NOT INTERFERE IN ANY WAY WITH PARTICIPANT'S RIGHT OR THE RIGHT OF THE COMPANY (OR THE PARENT OR SUBSIDIARY EMPLOYING OR RETAINING PARTICIPANT) TO TERMINATE PARTICIPANT'S RELATIONSHIP AS A SERVICE PROVIDER AT ANY TIME, WITH OR WITHOUT CAUSE.
- 10. Address for Notices. Any notice to be given to the Company under the terms of this Agreement will be addressed to the Company at comScore, Inc., 11465 Sunset Hills Road, Suite 200, Reston, Virginia 20190, or at such other address as the Company may hereafter designate in writing.
- 11. <u>Grant is Not Transferable</u>. Except to the limited extent provided in Section 6, this grant and the rights and privileges conferred hereby will not be transferred, assigned, pledged or hypothecated in any way (whether by operation of law or otherwise) and will not be subject to sale under execution, attachment or similar process. Upon any attempt to transfer, assign, pledge, hypothecate or otherwise dispose of this grant, or any right or privilege conferred hereby, or upon any attempted sale under any execution, attachment or similar process, this grant and the rights and privileges conferred hereby immediately will become null and void.

- 12. <u>Binding Agreement</u>. Subject to the limitation on the transferability of this grant contained herein, this Agreement will be binding upon and inure to the benefit of the heirs, legatees, legal representatives, successors and assigns of the parties hereto.
- 13. Additional Conditions to Issuance of Stock. If at any time the Company will determine, in its discretion, that the listing, registration or qualification of the Shares upon any securities exchange or under any state or federal law, or the consent or approval of any governmental regulatory authority is necessary or desirable as a condition to the issuance of Shares to Participant (or his or her estate), such issuance will not occur unless and until such listing, registration, qualification, consent or approval will have been effected or obtained free of any conditions not acceptable to the Company. Where the Company determines that the delivery of the payment of any Shares will violate federal securities laws or other applicable laws, the Company will defer delivery until the earliest date at which the Company reasonably anticipates that the delivery of Shares will no longer cause such violation. The Company will make all reasonable efforts to meet the requirements of any such state or federal law or securities exchange and to obtain any such consent or approval of any such governmental authority.
- 14. <u>Plan Governs</u>. This Agreement is subject to all terms and provisions of the Plan. In the event of a conflict between one or more provisions of this Agreement and one or more provisions of the Plan will govern. Capitalized terms used and not defined in this Agreement will have the meaning set forth in the Plan.
- 15. <u>Administrator Authority</u>. The Administrator will have the power to interpret the Plan and this Agreement and to adopt such rules for the administration, interpretation and application of the Plan as are consistent therewith and to interpret or revoke any such rules (including, but not limited to, the determination of whether or not any Restricted Stock Units have vested). All actions taken and all interpretations and determinations made by the Administrator in good faith will be final and binding upon Participant, the Company and all other interested persons. No member of the Administrator will be personally liable for any action, determination or interpretation made in good faith with respect to the Plan or this Agreement.
- 16. <u>Electronic Delivery.</u> The Company may, in its sole discretion, decide to deliver any documents related to Restricted Stock Units awarded under the Plan or future Restricted Stock Units that may be awarded under the Plan by electronic means or request Participant's consent to participate in the Plan by electronic means. Participant hereby consents to receive such documents by electronic delivery and agrees to participate in the Plan through any on-line or electronic system established and maintained by the Company or another third party designated by the Company.
 - 17. Captions. Captions provided herein are for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.
- 18. <u>Agreement Severable</u>. In the event that any provision in this Agreement will be held invalid or unenforceable, such provision will be severable from, and such invalidity or unenforceability will not be construed to have any effect on, the remaining provisions of this Agreement.

- 19. <u>Modifications to the Agreement</u>. This Agreement constitutes the entire understanding of the parties on the subjects covered. Participant expressly warrants that he or she is not accepting this Agreement in reliance on any promises, representations, or inducements other than those contained herein. Modifications to this Agreement or the Plan can be made only in an express written contract executed by a duly authorized officer of the Company. Notwithstanding anything to the contrary in the Plan or this Agreement, the Company reserves the right to revise this Agreement as it deems necessary or advisable, in its sole discretion and without the consent of Participant, to comply with Section 409A of the Internal Revenue Code of 1986, as amended (the "Code") or to otherwise avoid imposition of any additional tax or income recognition under Section 409A of the Code in connection to this Award of Restricted Stock Units.
- 20. <u>Amendment, Suspension or Termination of the Plan</u>. By accepting this Award, Participant expressly warrants that he or she has received an Award of Restricted Stock Units under the Plan, and has received, read and understood a description of the Plan. Participant understands that the Plan is discretionary in nature and may be amended, suspended or terminated by the Company at any time.
- 21. <u>Governing Law</u>. This Agreement will be governed by the laws of the Commonwealth of Virginia, without giving effect to the conflict of law principles thereof. For purposes of litigating any dispute that arises under this Award of Restricted Stock Units or this Agreement, the parties hereby submit to and consent to the jurisdiction of the Commonwealth of Virginia, and agree that such litigation will be conducted in the courts of Fairfax County, Virginia, or the federal courts for the United States for the Eastern District of Virginia, and no other courts, where this Award of Restricted Stock Units is made and/or to be performed.

COMSCORE NETWORKS, INC. 1999 STOCK PLAN, AS AMENDED STOCK OPTION AGREEMENT

Unless otherwise defined herein, the terms defined in the 1999 Stock Plan, as amended, shall have the same defined meanings in this Stock Option Agreement.

I. NOTICE OF STOCK OPTION GRANT

Name Magid Abraham

Address Address

The undersigned Optionee has been granted an Option to purchase Common Stock of the Company, subject to the terms and conditions of the Plan and this Option Agreement, as follows:

Date of Grant December 16, 2003

Vesting Commencement Date Not applicable

Exercise Price per Share \$0.05

Total Number of Shares Granted 3,305,495

Total Exercise Price \$165,274.75

Type of Option:

o Nonstatutory Stock Option

Term/Expiration Date: December 16, 2013

Vesting Schedule:

100% of the Shares subject to the Option shall vest on the earlier of (i) December 16, 2009 and (ii) the consummation of a Change of Control (as defined below), subject to the earlier vesting of a number of Shares subject to the Option upon the attainment of each of the following performance objectives (the "Performance Objectives"):

1. 581,633 Shares will vest and become exercisable if the Company achieves EBITDA (as defined below) greater than \$0 for a full fiscal quarter;

- 2. 581,633 Shares will vest and become exercisable if the Company achieves Revenues (as defined below) of greater than or equal to \$40 million for the Trailing Twelve Month Period (as defined below);
- 3. 581,633 Shares will vest and become exercisable if the Company achieves Net Income (as defined below) greater than \$0 for the Trailing Twelve Month Period;
- 4. 520,199 Shares will vest and become exercisable if the Company achieves Revenues greater than or equal to \$50 million for the Trailing Twelve Month Period;
- 5. 520,199 Shares will vest and become exercisable if the Company achieves Pretax Net Income (as defined below) of greater than or equal to \$5,000,000 for the Trailing Twelve Month Period; and
- 6. 520,199 Shares will vest and become exercisable if the Company achieves Pretax Net Income of greater than or equal to \$10,000,000 for the Trailing Twelve Month Period.

With respect to any particular Performance Objective, the Board of Directors shall make a good faith determination of whether the Company has achieved such Performance Objective based upon the financial statements presented to the Board of Directors, which determination shall be binding upon the Optionee. Notwithstanding the order of the listing of the Performance Objectives in paragraphs 1 through 6, each Performance Objective is independent. For avoidance of doubt, the vesting of Shares upon the attainment of any particular Performance Objective is not dependent upon whether any other Performance Objective has been attained, or upon the chronological order in which such Performance Objective has been attained.

In the event any of the Performance Objectives listed in the above paragraphs 1 through 6 are not attained, Shares subject to the Option that would have vested in connection with such Performance Objective shall vest only on December 16, 2009, provided that the Optionee is a Service Provider to the Company or a Subsidiary of the Company on such date.

For purposes hereof,

"Change in Control" shall mean the occurrence of any of the following events:

(i) a consolidation or merger of the Company with or into any other corporation or corporations (or entity or entities) (unless the Company's stockholders of record as constituted immediately prior to such transaction will, immediately after such transaction, hold (solely in respect of their equity interests in the Company before the transaction) at least a majority of the voting power of the surviving or successor entity to the business and assets of the Company);

(ii) a sale, conveyance or disposition of all or substantially all of the assets of the Company (other than a pledge of assets or grant of security interest therein to a commercial lender or similar entity in connection with commercial lending or similar transactions) (unless the Company's stockholders of record as constituted immediately prior to such transaction will, immediately after such transaction, hold (solely in respect of their equity interests in the Company before the

transaction) at least a majority of the voting power of the surviving entity or successor to the business and assets of the Company);

- (iii) any sale, transfer or issuance or series of sales, transfers or issuances of shares of the Company's capital stock by the Company or the existing holders thereof to new holders, as a result of which the holders of the Company's outstanding capital stock possessing the voting power to elect a majority of the Company's Board immediately prior to such sale, transfer or issuance cease to own the requisite amount of the Company's outstanding capital stock to possess the voting power to elect a majority of the Company's Board; or
- (iv) the effectuation of a transaction or series of related transactions in which at least a majority of the Company's the outstanding voting power is transferred to another entity; provided that an Automatic Recapitalization (as defined in the Company's Amended and Restated Certificate of Incorporation) shall not be deemed a Change in Control.
 - "EBITDA" shall mean the Company's earnings before interest, taxes, depreciation and amortization.
 - "Net Income" shall mean the Company's gross Revenues minus taxes, interest, depreciation and all other expenses.
 - "Pretax Net Income" shall mean the Company's Net Income before federal income taxes are subtracted.
 - "Revenue" shall mean revenue generated from third parties and recognized in accordance with the Company's revenue recognition policy then in effect.
 - "Trailing Twelve Month Period" shall mean the twelve-month period preceding the date of measurement with respect to a particular Performance Objective.

Termination Period

This Option shall be exercisable for three months after Optionee ceases to be a Service Provider. Upon Optionee's death or Disability, this Option may be exercised for one year after Optionee ceases to be a Service Provider. In no event may Optionee exercise this Option after the Term/Expiration Date as provided above.

II. AGREEMENT

1. <u>Grant of Option</u>. The Plan Administrator of the Company hereby grants to the Optionee named in the Notice of Grant (the "Optionee"), an option (the "Option") to purchase the number of Shares set forth in the Notice of Grant, at the exercise price per Share set forth in the Notice of Grant (the "Exercise Price"), and subject to the terms and conditions of the Plan, which is incorporated herein by reference. Subject to Section 14(c) of the Plan, in the event of a conflict between the terms and conditions of the Plan and this Option Agreement, the terms and conditions of the Plan shall prevail.

If designated in the Notice of Grant as an Incentive Stock Option ("ISO"), this Option is intended to qualify as an Incentive Stock Option as defined in Section 422 of the Code. Nevertheless, to the extent that it exceeds the \$100,000 rule of Code Section 422(d), this Option shall be treated as a Nonstatutory Stock Option ("NSO").

2. Exercise of Option

- (a) Right to Exercise. This Option shall be exercisable during its term in accordance with the Vesting Schedule set out in the Notice of Grant and with the applicable provisions of the Plan and this Option Agreement.
- (b) <u>Method of Exercise</u>. This Option shall be exercisable by delivery of an exercise notice in the form attached as <u>Exhibit A</u> (the "Exercise Notice") which shall state the election to exercise the Option, the number of Shares with respect to which the Option is being exercised, and such other representations and agreements as may be required by the Company. The Exercise Notice shall be accompanied by payment of the aggregate Exercise Price as to all Exercised Shares. This Option shall be deemed to be exercised upon receipt by the Company of such fully executed Exercise Notice accompanied by the aggregate Exercise Price.

No Shares shall be issued pursuant to the exercise of an Option unless such issuance and such exercise comply with Applicable Laws. Assuming such compliance, for income tax purposes the Shares shall be considered transferred to the Optionee on the date on which the Option is exercised with respect to such Shares.

- 3. Optionee's Representations. In the event the Shares have not been registered under the Securities Act of 1933, as amended, at the time this Option is exercised, the Optionee shall, if required by the Company, concurrently with the exercise of all or any portion of this Option, deliver to the Company his or her Investment Representation Statement in the form attached hereto as Exhibit B.
- 4. <u>Lock-Up Period</u>. Optionee hereby agrees that, if so requested by the Company or any representative of the underwriters (the "Managing Underwriter") in connection with any registration of the offering of any securities of the Company under the Securities Act, Optionee shall not sell or otherwise transfer any Shares or other securities of the Company during the 180-day period (or such other period as may be requested in writing by the Managing Underwriter and agreed to in writing by the Company) (the "Market Standoff Period") following the effective date of a registration statement of the Company filed under the Securities Act. Such restriction shall apply only to the first registration statement of the Company to the public in an underwritten public offering under the Securities Act. The Company may impose stop-transfer instructions with respect to securities subject to the foregoing restrictions until the end of such Market Standoff Period.
 - 5. Method of Payment. Payment of the aggregate Exercise Price shall be by any of the following, or a combination thereof, at the election of the Optionee:

- (a) cash or check;
- (b) consideration received by the Company under a formal cashless exercise program adopted by the Company in connection with the Plan; or
- (c) surrender of other Shares which, (i) in the case of Shares acquired upon exercise of an option, have been owned by the Optionee for more than six (6) months on the date of surrender, and (ii) have a Fair Market Value on the date of surrender equal to the aggregate Exercise Price of the Exercised Shares.
- 6. <u>Restrictions on Exercise</u>. This Option may not be exercised until such time as the Plan has been approved by the shareholders of the Company, or if the issuance of such Shares upon such exercise or the method of payment of consideration for such shares would constitute a violation of any Applicable Law.
- 7. Non-Transferability of Option. This Option may not be transferred in any manner otherwise than by will or by the laws of descent or distribution and may be exercised during the lifetime of Optionee only by Optionee. The terms of the Plan and this Option Agreement shall be binding upon the executors, administrators, heirs, successors and assigns of the Optionee.
 - 8. Term of Option. This Option may be exercised only within the term set out in the Notice of Grant, and may be exercised during such term only in accordance with the Plan and the terms of this Option.
- 9. Tax Consequences. Set forth below is a brief summary as of the date of this Option of some of the federal tax consequences of exercise of this Option and disposition of the Shares. THIS SUMMARY IS NECESSARILY INCOMPLETE, AND THE TAX LAWS AND REGULATIONS ARE SUBJECT TO CHANGE. THE OPTIONEE SHOULD CONSULT A TAX ADVISER BEFORE EXERCISING THIS OPTION OR DISPOSING OF THE SHARES.
- (a) Exercise of NSO. There may be a regular federal income tax liability upon the exercise of an NSO. The Optionee will be treated as having received compensation income (taxable at ordinary income tax rates) equal to the excess, if any, of the Fair Market Value of the Shares on the date of exercise over the Exercise Price. If Optionee is an Employee or a former Employee, the Company will be required to withhold from Optionee's compensation or collect from Optionee and pay to the applicable taxing authorities an amount in cash equal to a percentage of this compensation income at the time of exercise, and may refuse to honor the exercise and refuse to deliver Shares if such withholding amounts are not delivered at the time of exercise.
- (b) Exercise of ISO. If this Option qualifies as an ISO, there will be no regular federal income tax liability upon the exercise of the Option, although the excess, if any, of the Fair Market Value of the Shares on the date of exercise over the Exercise Price will be treated as an adjustment to the alternative minimum tax for federal tax purposes and may subject the Optionee to the alternative minimum tax in the year of exercise

- (c) <u>Disposition of Shares</u>. In the case of an NSO, if Shares are held for at least one year, any gain realized on disposition of the Shares will be treated as long-term capital gain for federal income tax purposes. In the case of an ISO, if Shares transferred pursuant to the Option are held for at least one year after exercise and of at least two years after the Date of Grant, any gain realized on disposition of the Shares will also be treated as long-term capital gain for federal income tax purposes. If Shares purchased under an ISO are disposed of within one year after exercise or two years after the Date of Grant, any gain realized on such disposition will be treated as compensation income (taxable at ordinary income rates) to the extent of the difference between the Exercise Price and the lesser of (1) the Fair Market Value of the Shares on the date of exercise, or (2) the sale price of the Shares. Any additional gain will be taxed as capital gain, short-term or long-term depending on the period that the ISO Shares were held.
- (d) Notice of Disqualifying Disposition of ISO Shares. If the Option granted to Optionee herein is an ISO, and if Optionee sells or otherwise disposes of any of the Shares acquired pursuant to the ISO on or before the later of (1) the date two years after the Date of Grant, or (2) the date one year after the date of exercise, the Optionee shall immediately notify the Company in writing of such disposition. Optionee agrees that Optionee may be subject to income tax withholding by the Company on the compensation income recognized by the Optionee.
- 10. Entire Agreement; Governing Law. The Plan is incorporated herein by reference. The Plan and this Option Agreement constitute the entire agreement of the parties with respect to the subject matter hereof and supersede in their entirety all prior undertakings and agreements of the Company and Optionee with respect to the subject matter hereof, and may not be modified adversely to the Optionee's interest except by means of a writing signed by the Company and Optionee. The internal substantive laws but not the choice of law rules of Virginia govern this agreement.
- 11. No Guarantee of Continued Service. OPTIONEE ACKNOWLEDGES AND AGREES THAT THE VESTING OF SHARES PURSUANT TO THE VESTING SCHEDULE HEREOF IS EARNED ONLY BY CONTINUING AS A SERVICE PROVIDER AT THE WILL OF THE COMPANY (NOT THROUGH THE ACT OF BEING HIRED, BEING GRANTED THIS OPTION OR ACQUIRING SHARES HEREUNDER). OPTIONEE FURTHER ACKNOWLEDGES AND AGREES THAT THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREUNDER AND THE VESTING SCHEDULE SET FORTH HEREIN DO NOT CONSTITUTE AN EXPRESS OR IMPLIED PROMISE OF CONTINUED ENGAGEMENT AS A SERVICE PROVIDER FOR THE VESTING PERIOD, FOR ANY PERIOD, OR AT ALL, AND SHALL NOT INTERFERE IN ANY WAY WITH OPTIONEE'S RIGHT OR THE COMPANY'S RIGHT TO TERMINATE OPTIONEE'S RELATIONSHIP AS A SERVICE PROVIDER AT ANY TIME, WITH OR WITHOUT CAUSE.

Optionee acknowledges receipt of a copy of the Plan and represents that he or she is familiar with the terms and provisions thereof, and hereby accepts this Option subject to all of the terms and provisions thereof. Optionee has reviewed the Plan and this Option in their entirety, has had an opportunity to obtain the advice of counsel prior to executing this Option and fully understands all provisions of the Option. Optionee hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions arising under the Plan or this Option. Optionee further agrees to notify the Company upon any change in the residence address indicated below.

OPTIONEE

COMSCORE NETWORKS, INC.

/s/ Sheri Huston

By Sheri Huston

Magid Abraham

Chief Financial Officer

Title

1018 Murphy Drive, Great Falls, VA 22066

Residence Address

-7-

EXHIBIT A 1999 STOCK PLAN, AS AMENDED EXERCISE NOTICE

comScore Networks, Inc. 11465 Sunset Hills Road, Ste. 200 Reston, VA 20190

Attention: Corporate Secretary
1. Exercise of Option. Effective as of today,, 20, the undersigned ("Optionee") hereby elects to exercise Optionee's option to purchase shares of the Common Stock (the "Shares") of comScore Networks, Inc. (the "Company") under and pursuant to the 1999 Stock Plan, as amended (the "Plan"), and the Stock Option Agreement dated, 20 (the "Option Agreement").
2. Delivery of Payment. Purchaser herewith delivers to the Company the full purchase price of the Shares, as set forth in the Option Agreement.
3. Representations of Optionee. Optionee acknowledges that Optionee has received, read and understood the Plan and the Option Agreement and agrees to abide by and be bound by their terms and conditions.
4. <u>Rights as Shareholder</u> . Until the issuance of the Shares (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company), no right to vote or receive dividends or any other rights as a shareholder shall exist with respect to the Optioned Stock, notwithstanding the exercise of the Option. The Shares shall be issued to the Optionee as soon as practicable after the Option is exercised. No adjustment shall be made for a dividend or other right for which the record date is prior to the date of issuance except as provided in Section 12 of the Plan.
5. <u>Company's Right of First Refusal</u> . Before any Shares held by Optionee or any transferee (either being sometimes referred to herein as the "Holder") may be sold or otherwise transferred (including transfer by gift or operation of law), the Company or its assignee(s) shall have a right of first refusal to purchase the Shares on the terms and conditions set forth in this Section (the "Right of First Refusal").
(a) Notice of Proposed Transfer. The Holder of the Shares shall deliver to the Company a written notice (the "Notice") stating: (i) the Holder's bona fide intention to sell or otherwise transfer such Shares; (ii) the name of each proposed purchaser or other transferee ("Proposed Transferee"); (iii) the number of Shares to be transferred to each Proposed Transferee; and (iv) the bona fide cash price or other consideration for which the Holder proposes to transfer the

Shares (the "Offered Price"), and the Holder shall offer the Shares at the Offered Price to the Company or its assignee(s).

- (b) Exercise of Right of First Refusal. At any time within thirty (30) days after receipt of the Notice, the Company and/or its assignee(s) may, by giving written notice to the Holder, elect to purchase all, but not less than all, of the Shares proposed to be transferred to any one or more of the Proposed Transferees, at the purchase price determined in accordance with subsection (c) below.
- (c) <u>Purchase Price</u>. The purchase price ("Purchase Price") for the Shares purchased by the Company or its assignee(s) under this Section shall be the Offered Price. If the Offered Price includes consideration other than cash, the Board of Directors of the Company in good faith shall determine the cash equivalent value of the non-cash consideration.
- (d) <u>Payment</u>. Payment of the Purchase Price shall be made, at the option of the Company or its assignee(s), in cash (by check), by cancellation of all or a portion of any outstanding indebtedness of the Holder to the Company (or, in the case of repurchase by an assignee, to the assignee), or by any combination thereof within thirty (30) days after receipt of the Notice or in the manner and at the times set forth in the Notice.
- (e) <u>Holder's Right to Transfer</u>. If all of the Shares proposed in the Notice to be transferred to a given Proposed Transferee are not purchased by the Company and/or its assignee(s) as provided in this Section, then the Holder may sell or otherwise transfer such Shares to that Proposed Transferee at the Offered Price or at a higher price, provided that such sale or other transfer is consummated within 120 days after the date of the Notice, that any such sale or other transfer is effected in accordance with any applicable securities laws and that the Proposed Transferee is mixing that the provisions of this Section shall continue to apply to the Shares in the hands of such Proposed Transferee. If the Shares described in the Notice are not transferred to the Proposed Transferee within such period, a new Notice shall be given to the Company, and the Company and/or its assignees shall again be offered the Right of First Refusal before any Shares held by the Holder may be sold or otherwise transferred.
- (f) Exception for Certain Family Transfers. Anything to the contrary contained in this Section notwithstanding, the transfer of any or all of the Shares during the Optionee's lifetime or on the Optionee's death by will or intestacy to the Optionee's immediate family or a trust for the benefit of the Optionee's immediate family shall be exempt from the provisions of this Section. "Immediate Family" as used herein shall mean spouse, lineal descendant or antecedent, father, mother, brother or sister. In such case, the transferee or other recipient shall receive and hold the Shares so transferred subject to the provisions of this Section, and there shall be no further transfer of such Shares except in accordance with the terms of this Section.
- (g) <u>Termination of Right of First Refusal</u>. The Right of First Refusal shall terminate as to any Shares upon the first sale of Common Stock of the Company to the general public pursuant to a registration statement filed with and declared effective by the Securities and Exchange Commission under the Securities Act of 1933, as amended.

- 6. <u>IRA</u>. Optionee acknowledges that the Shares shall be subject to certain transfer restrictions, rights of first refusal and co-sale set forth in that certain Fourth Amended and Restated Investor Rights Agreement, made as of August 1, 2003 by and among the Company, certain stockholders of the Company listed on the signatures pages thereto and the founders listed on the signature pages thereto, as such agreement is amended from time to time.
- 7. <u>Tax Consultation</u>. Optionee understands that Optionee may suffer adverse tax consequences as a result of Optionee's purchase or disposition of the Shares. Optionee represents that Optionee has consulted with any tax consultants Optionee deems advisable in connection with the purchase or disposition of the Shares and that Optionee is not relying on the Company for any tax advice.

8. Restrictive Legends and Stop-Transfer Orders

(a) <u>Legends</u>. Optionee understands and agrees that the Company shall cause the legends set forth below or legends substantially equivalent thereto, to be placed upon any certificate(s) evidencing ownership of the Shares together with any other legends that may be required by the Company or by state or federal securities laws:

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 (THE "ACT") AND MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, PLEDGED OR HYPOTHECATED UNLESS AND UNTIL REGISTERED UNDER THE ACT OR, IN THE OPINION OF COMPANY COUNSEL SATISFACTORY TO THE ISSUER OF THESE SECURITIES, SUCH OFFER, SALE OR TRANSFER, PLEDGE OR HYPOTHECATION IS IN COMPLIANCE THEREWITH.

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN RESTRICTIONS ON TRANSFER AND A RIGHT OF FIRST REFUSAL HELD BY THE ISSUER OR ITS ASSIGNEE(S) AS SET FORTH IN THE EXERCISE NOTICE BETWEEN THE ISSUER AND THE ORIGINAL HOLDER OF THESE SHARES, A COPY OF WHICH MAY BE OBTAINED AT THE PRINCIPAL OFFICE OF THE ISSUER. SUCH TRANSFER RESTRICTIONS AND RIGHT OF FIRST REFUSAL ARE BINDING ON TRANSFEREES OF THESE SHARES.

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN RESTRICTIONS ON TRANSFER, A RIGHT OF FIRST REFUSAL AND A RIGHT OF CO-SALE HELD BY THE ISSUER AND CERTAIN STOCKHOLDERS OF THE ISSUER PURSUANT TO THE ISSUER'S FOURTH AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT, AS AMENDED FROM TIME TO TIME. COPIES OF SUCH AGREEMENT MAY BE OBTAINED UPON WRITTEN REQUEST TO THE SECRETARY OF

THE ISSUER. SUCH TRANSFER RESTRICTIONS, RIGHT OF FIRST REFUSAL AND RIGHT OF CO-SALE ARE BINDING ON TRANSFEREES OF THESE SHARES.

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO A VOTING AGREEMENT CONTAINED IN THE ISSUER'S FOURTH AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT, AS AMENDED FROM TIME TO TIME. COPIES OF SUCH AGREEMENT MAY BE OBTAINED UPON WRITTEN REQUEST TO THE SECRETARY OF THE ISSUER. BY ACCEPTING ANY INTEREST IN SUCH SHARES THE PERSON ACCEPTING SUCH INTEREST SHALL BE DEEMED TO AGREE TO AND SHALL BECOME BOUND BY ALL THE PROVISIONS OF SAID VOTING AGREEMENT.

- (b) Stop-Transfer Notices. Optionee agrees that, in order to ensure compliance with the restrictions referred to herein, the Company may issue appropriate "stop transfer" instructions to its transfer agent, if any, and that, if the Company transfers its own securities, it may make appropriate notations to the same effect in its own records.
- (c) <u>Refusal to Transfer</u>. The Company shall not be required (i) to transfer on its books any Shares that have been sold or otherwise transferred in violation of any of the provisions of this Exercise Notice or (ii) to treat as owner of such Shares or to accord the right to vote or pay dividends to any purchaser or other transferee to whom such Shares shall have been so transferred.
- 9. <u>Successors and Assigns</u>. The Company may assign any of its rights under this Exercise Notice to single or multiple assignees, and this Exercise Notice shall inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer herein set forth, this Exercise Notice shall be binding upon Optionee and his or her heirs, executors, administrators, successors and assigns.
- 10. <u>Interpretation</u>. Any dispute regarding the interpretation of this Exercise Notice shall be submitted by Optionee or by the Company forthwith to the Administrator which shall review such dispute at its next regular meeting. The resolution of such a dispute by the Administrator shall be final and binding on all parties.
 - 11. Governing Law; Severability. This Exercise Notice is governed by the internal substantive laws but not the choice of law rules, of Virginia.
- 12. Entire Agreement. The Plan and Option Agreement are incorporated herein by reference. This Exercise Notice, the Plan, the Option Agreement and the Investment Representation Statement constitute the entire agreement of the parties with respect to the subject matter hereof and supersede in their entirety all prior undertakings and agreements of the Company and Optionee with respect to the subject matter hereof, and may not be modified adversely to the Optionee's interest except by means of a writing signed by the Company and Optionee.

Accepted by:		
COMSCORE	NETWORKS, INC.	
Ву		
Title		
	Address:	
	11465 Sunset Hills Road, Ste 200	
	Reston, VA 20190 Date Received	
-5-		
	COMSCORE By Title	COMSCORE NETWORKS, INC. By Title Address: 11465 Sunset Hills Road, Ste 200 Reston, VA 20190 Date Received

EXHIBIT B

INVESTMENT REPRESENTATION STATEMENT

OPTIONEE: MAGID ABRAHAM

COMPANY: COMSCORE NETWORKS, INC.

SECURITY: COMMON STOCK

AMOUNT:

DATE:

In connection with the purchase of the above-listed Securities, the undersigned Optionee represents to the Company the following:

(a) Optionee is aware of the Company's business affairs and financial condition and has acquired sufficient information about the Company to reach an informed and knowledgeable decision to acquire the Securities. Optionee is acquiring these Securities for investment for Optionee's own account only and not with a view to, or for resale in connection with, any "distribution" thereof within the meaning of the Securities Act of 1933, as amended (the "Securities Act").

(b) Optionee acknowledges and understands that the Securities constitute "restricted securities" under the Securities Act and have not been registered under the Securities Act in reliance upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of Optionee's investment intent as expressed herein. In this connection, Optionee understands that, in the view of the Securities and Exchange Commission, the statutory basis for such exemption may be unavailable if Optionee's representation was predicated solely upon a present intention to hold these Securities for the minimum capital gains period specified under tax statutes, for a deferred sale, for or until an increase or decrease in the market price of the Securities, or for a period of one year or any other fixed period in the future. Optionee further understands that the Securities must be held indefinitely unless they are subsequently registered under the Securities Act or an exemption from such registration is available. Optionee further acknowledges and understands that the Company is under no obligation to register the Securities. Optionee understands that the certificate evidencing the Securities will be imprinted with a legend which prohibits the transfer of the Securities unless they are registered or such registration is not required in the opinion of counsel satisfactory to the Company, and any other legend required under applicable state securities laws.

(c) Optionee is familiar with the provisions of Rule 701 and Rule 144, each promulgated under the Securities Act, which, in substance, permit limited public resale of "restricted securities" acquired, directly or indirectly from the issuer thereof, in a non-public offering subject to the satisfaction of certain conditions. Rule 701 provides that if the issuer qualifies under Rule 701 at the

time of the grant of the Option to the Optionee, the exercise will be exempt from registration under the Securities Act. In the event the Company becomes subject to the reporting requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, ninety (90) days thereafter (or such longer period as any market stand-off agreement may require) the Securities exempt under Rule 701 may be resold, subject to the satisfaction of certain of the conditions specified by Rule 144, including: (1) the resale being made through a broker in an unsolicited "broker's transaction" or in transactions directly with a market maker (as said term is defined under the Securities Exchange Act of 1934); and, in the case of an affiliate, (2) the availability of certain public information about the Company, (3) the amount of Securities being sold during any three month period not exceeding the limitations specified in Rule 144(e), and (4) the timely filing of a Form 144, if applicable.

In the event that the Company does not qualify under Rule 701 at the time of grant of the Option, then the Securities may be resold in certain limited circumstances subject to the provisions of Rule 144, which requires the resale to occur not less than one year after the later of the date the Securities were sold by the Company or the date the Securities were sold by an affiliate of the Company, within the meaning of Rule 144; and, in the case of acquisition of the Securities by an affiliate, or by a non-affiliate who subsequently holds the Securities less than two years, the satisfaction of the conditions set forth in sections (1), (2), (3) and (4) of the paragraph immediately above.

(d) Optionee further understands that in the event all of the applicable requirements of Rule 701 or 144 are not satisfied, registration under the Securities Act, compliance with Regulation A, or some other registration exemption will be required; and that, notwithstanding the fact that Rules 144 and 701 are not exclusive, the Staff of the Securities and Exchange Commission has expressed its opinion that persons proposing to sell private placement securities other than in a registered offering and otherwise than pursuant to Rules 144 or 701 will have a substantial burden of proof in establishing that an exemption from registration is available for such offers or sales, and that such persons and their respective brokers who participate in such transactions do so at their own risk. Optionee understands that no assurances can be given that any such other registration exemption will be available in such event.

Signature of Optionee:		
Date:	, 20	
2		

COMSCORE NETWORKS, INC. 1999 STOCK PLAN, AS AMENDED STOCK OPTION AGREEMENT

Unless otherwise defined herein, the terms defined in the 1999 Stock Plan, as amended, shall have the same defined meanings in this Stock Option Agreement.

I. NOTICE OF STOCK OPTION GRANT

Name Gian Fulgoni Address

Address Address

The undersigned Optionee has been granted an Option to purchase Common Stock of the Company, subject to the terms and conditions of the Plan and this Option Agreement, as follows:

Date of Grant December 16, 2003

Vesting Commencement Date Not applicable

Exercise Price per Share \$0.05

Total Number of Shares Granted 2,377,637

Total Exercise Price \$118,881.85

Type of Option:

o Nonstatutory Stock Option

Term/Expiration Date: December 16, 2013

Vesting Schedule:

100% of the Shares subject to the Option shall vest on the earlier of (i) December 16, 2009 and (ii) the consummation of a Change of Control (as defined below), subject to the earlier vesting of a number of Shares subject to the Option upon the attainment of each of the following performance objectives (the "Performance Objectives"):

1. 418,367 Shares will vest and become exercisable if the Company achieves EBITDA (as defined below) greater than \$0 for a full fiscal quarter;

- 2. 418,367 Shares will vest and become exercisable if the Company achieves Revenues (as defined below) of greater than or equal to \$40 million for the Trailing Twelve Month Period (as defined below);
- 3. 418,367 Shares will vest and become exercisable if the Company achieves Net Income (as defined below) greater than \$0 for the Trailing Twelve Month Period;
- 4. 374,178 Shares will vest and become exercisable if the Company achieves Revenues greater than or equal to \$50 million for the Trailing Twelve Month Period;
- 5. 374,178 Shares will vest and become exercisable if the Company achieves Pretax Net Income (as defined below) of greater than or equal to \$5,000,000 for the Trailing Twelve Month Period; and
- 6. 374,178 Shares will vest and become exercisable if the Company achieves Pretax Net Income of greater than or equal to \$10,000,000 for the Trailing Twelve Month Period.

With respect to any particular Performance Objective, the Board of Directors shall make a good faith determination of whether the Company has achieved such Performance Objective based upon the financial statements presented to the Board of Directors, which determination shall be binding upon the Optionee. Notwithstanding the order of the listing of the Performance Objectives in paragraphs 1 through 6, each Performance Objective is independent. For avoidance of doubt, the vesting of Shares upon the attainment of any particular Performance Objective is not dependent upon whether any other Performance Objective has been attained, or upon the chronological order in which such Performance Objective has been attained.

In the event any of the Performance Objectives listed in the above paragraphs 1 through 6 are not attained, Shares subject to the Option that would have vested in connection with such Performance Objective shall vest only on December 16, 2009, provided that the Optionee is a Service Provider to the Company or a Subsidiary of the Company on such date.

For purposes hereof,

"Change in Control" shall mean the occurrence of any of the following events:

(i) a consolidation or merger of the Company with or into any other corporation or corporations (or entity or entities) (unless the Company's stockholders of record as constituted immediately prior to such transaction will, immediately after such transaction, hold (solely in respect of their equity interests in the Company before the transaction) at least a majority of the voting power of the surviving or successor entity to the business and assets of the Company);

(ii) a sale, conveyance or disposition of all or substantially all of the assets of the Company (other than a pledge of assets or grant of security interest therein to a commercial lender or similar entity in connection with commercial lending or similar transactions) (unless the Company's stockholders of record as constituted immediately prior to such transaction will, immediately after such transaction, hold (solely in respect of their equity interests in the Company before the

transaction) at least a majority of the voting power of the surviving entity or successor to the business and assets of the Company);

- (iii) any sale, transfer or issuance or series of sales, transfers or issuances of shares of the Company's capital stock by the Company or the existing holders thereof to new holders, as a result of which the holders of the Company's outstanding capital stock possessing the voting power to elect a majority of the Company's Board immediately prior to such sale, transfer or issuance cease to own the requisite amount of the Company's outstanding capital stock to possess the voting power to elect a majority of the Company's Board; or
- (iv) the effectuation of a transaction or series of related transactions in which at least a majority of the Company's the outstanding voting power is transferred to another entity; provided that an Automatic Recapitalization (as defined in the Company's Amended and Restated Certificate of Incorporation) shall not be deemed a Change in Control.
 - "EBITDA" shall mean the Company's earnings before interest, taxes, depreciation and amortization.
 - "Net Income" shall mean the Company's gross Revenues minus taxes, interest, depreciation and all other expenses.
 - "Pretax Net Income" shall mean the Company's Net Income before federal income taxes are subtracted.
 - "Revenue" shall mean revenue generated from third parties and recognized in accordance with the Company's revenue recognition policy then in effect.
 - "Trailing Twelve Month Period" shall mean the twelve-month period preceding the date of measurement with respect to a particular Performance Objective.

Termination Period:

This Option shall be exercisable for three months after Optionee ceases to be a Service Provider. Upon Optionee's death or Disability, this Option may be exercised for one year after Optionee ceases to be a Service Provider. In no event may Optionee exercise this Option after the Term/Expiration Date as provided above.

II ACDEEMENT

1. <u>Grant of Option</u>. The Plan Administrator of the Company hereby grants to the Optionee named in the Notice of Grant (the "Optionee"), an option (the "Option") to purchase the number of Shares set forth in the Notice of Grant, at the exercise price per Share set forth in the Notice of Grant (the "Exercise Price"), and subject to the terms and conditions of the Plan, which is incorporated herein by reference. Subject to Section 14(c) of the Plan, in the event of a conflict between the terms and conditions of the Plan and this Option Agreement, the terms and conditions of the Plan shall prevail.

If designated in the Notice of Grant as an Incentive Stock Option ("ISO"), this Option is intended to qualify as an Incentive Stock Option as defined in Section 422 of the Code. Nevertheless, to the extent that it exceeds the \$100,000 rule of Code Section 422(d), this Option shall be treated as a Nonstatutory Stock Option ("NSO").

2. Exercise of Option

- (a) <u>Right to Exercise</u>. This Option shall be exercisable during its term in accordance with the Vesting Schedule set out in the Notice of Grant and with the applicable provisions of the Plan and this Option Agreement.
- (b) <u>Method of Exercise</u>. This Option shall be exercisable by delivery of an exercise notice in the form attached as <u>Exhibit A</u> (the "Exercise Notice") which shall state the election to exercise the Option, the number of Shares with respect to which the Option is being exercised, and such other representations and agreements as may be required by the Company. The Exercise Notice shall be accompanied by payment of the aggregate Exercise Price as to all Exercised Shares. This Option shall be deemed to be exercised upon receipt by the Company of such fully executed Exercise Notice accompanied by the aggregate Exercise Price.

No Shares shall be issued pursuant to the exercise of an Option unless such issuance and such exercise comply with Applicable Laws. Assuming such compliance, for income tax purposes the Shares shall be considered transferred to the Optionee on the date on which the Option is exercised with respect to such Shares.

- 3. Optionee's Representations. In the event the Shares have not been registered under the Securities Act of 1933, as amended, at the time this Option is exercised, the Optionee shall, if required by the Company, concurrently with the exercise of all or any portion of this Option, deliver to the Company his or her Investment Representation Statement in the form attached hereto as Exhibit B.
- 4. <u>Lock-Up Period</u>. Optionee hereby agrees that, if so requested by the Company or any representative of the underwriters (the "Managing Underwriter") in connection with any registration of the offering of any securities of the Company under the Securities Act, Optionee shall not sell or otherwise transfer any Shares or other securities of the Company during the 180-day period (or such other period as may be requested in writing by the Managing Underwriter and agreed to in writing by the Company) (the "Market Standoff Period") following the effective date of a registration statement of the Company filed under the Securities Act. Such restriction shall apply only to the first registration statement of the Company to the public in an underwritten public offering under the Securities Act. The Company may impose stop-transfer instructions with respect to securities subject to the foregoing restrictions until the end of such Market Standoff Period.
 - 5. Method of Payment. Payment of the aggregate Exercise Price shall be by any of the following, or a combination thereof, at the election of the Optionee:

- (a) cash or check;
- (b) consideration received by the Company under a formal cashless exercise program adopted by the Company in connection with the Plan; or
- (c) surrender of other Shares which, (i) in the case of Shares acquired upon exercise of an option, have been owned by the Optionee for more than six (6) months on the date of surrender, and (ii) have a Fair Market Value on the date of surrender equal to the aggregate Exercise Price of the Exercised Shares.
- 6. <u>Restrictions on Exercise</u>. This Option may not be exercised until such time as the Plan has been approved by the shareholders of the Company, or if the issuance of such Shares upon such exercise or the method of payment of consideration for such shares would constitute a violation of any Applicable Law.
- 7. Non-Transferability of Option. This Option may not be transferred in any manner otherwise than by will or by the laws of descent or distribution and may be exercised during the lifetime of Optionee only by Optionee. The terms of the Plan and this Option Agreement shall be binding upon the executors, administrators, heirs, successors and assigns of the Optionee.
 - 8. Term of Option. This Option may be exercised only within the term set out in the Notice of Grant, and may be exercised during such term only in accordance with the Plan and the terms of this Option.
- 9. Tax Consequences. Set forth below is a brief summary as of the date of this Option of some of the federal tax consequences of exercise of this Option and disposition of the Shares. THIS SUMMARY IS NECESSARILY INCOMPLETE, AND THE TAX LAWS AND REGULATIONS ARE SUBJECT TO CHANGE. THE OPTIONEE SHOULD CONSULT A TAX ADVISER BEFORE EXERCISING THIS OPTION OR DISPOSING OF THE SHARES.
- (a) Exercise of NSO. There may be a regular federal income tax liability upon the exercise of an NSO. The Optionee will be treated as having received compensation income (taxable at ordinary income tax rates) equal to the excess, if any, of the Fair Market Value of the Shares on the date of exercise over the Exercise Price. If Optionee is an Employee or a former Employee, the Company will be required to withhold from Optionee's compensation or collect from Optionee and pay to the applicable taxing authorities an amount in cash equal to a percentage of this compensation income at the time of exercise, and may refuse to honor the exercise and refuse to deliver Shares if such withholding amounts are not delivered at the time of exercise.
- (b) Exercise of ISO. If this Option qualifies as an ISO, there will be no regular federal income tax liability upon the exercise of the Option, although the excess, if any, of the Fair Market Value of the Shares on the date of exercise over the Exercise Price will be treated as an adjustment to the alternative minimum tax for federal tax purposes and may subject the Optionee to the alternative minimum tax in the year of exercise.

- (c) <u>Disposition of Shares</u>. In the case of an NSO, if Shares are held for at least one year, any gain realized on disposition of the Shares will be treated as long-term capital gain for federal income tax purposes. In the case of an ISO, if Shares transferred pursuant to the Option are held for at least one year after exercise and of at least two years after the Date of Grant, any gain realized on disposition of the Shares will also be treated as long-term capital gain for federal income tax purposes. If Shares purchased under an ISO are disposed of within one year after exercise or two years after the Date of Grant, any gain realized on such disposition will be treated as compensation income (taxable at ordinary income rates) to the extent of the difference between the Exercise Price and the lesser of (1) the Fair Market Value of the Shares on the date of exercise, or (2) the sale price of the Shares. Any additional gain will be taxed as capital gain, short-term or long-term depending on the period that the ISO Shares were held.
- (d) Notice of Disqualifying Disposition of ISO Shares. If the Option granted to Optionee herein is an ISO, and if Optionee sells or otherwise disposes of any of the Shares acquired pursuant to the ISO on or before the later of (1) the date two years after the Date of Grant, or (2) the date one year after the date of exercise, the Optionee shall immediately notify the Company in writing of such disposition. Optionee agrees that Optionee may be subject to income tax withholding by the Company on the compensation income recognized by the Optionee.
- 10. Entire Agreement; Governing Law. The Plan is incorporated herein by reference. The Plan and this Option Agreement constitute the entire agreement of the parties with respect to the subject matter hereof and supersede in their entirety all prior undertakings and agreements of the Company and Optionee with respect to the subject matter hereof, and may not be modified adversely to the Optionee's interest except by means of a writing signed by the Company and Optionee. The internal substantive laws but not the choice of law rules of Virginia govern this agreement.
- 11. No Guarantee of Continued Service. OPTIONEE ACKNOWLEDGES AND AGREES THAT THE VESTING OF SHARES PURSUANT TO THE VESTING SCHEDULE HEREOF IS EARNED ONLY BY CONTINUING AS A SERVICE PROVIDER AT THE WILL OF THE COMPANY (NOT THROUGH THE ACT OF BEING HIRED, BEING GRANTED THIS OPTION OR ACQUIRING SHARES HEREUNDER). OPTIONEE FURTHER ACKNOWLEDGES AND AGREES THAT THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREUNDER AND THE VESTING SCHEDULE SET FORTH HEREIN DO NOT CONSTITUTE AN EXPRESS OR IMPLIED PROMISE OF CONTINUED ENGAGEMENT AS A SERVICE PROVIDER FOR THE VESTING PERIOD, FOR ANY PERIOD, OR AT ALL, AND SHALL NOT INTERFERE IN ANY WAY WITH OPTIONEE'S RIGHT OR THE COMPANY'S RIGHT TO TERMINATE OPTIONEE'S RELATIONSHIP AS A SERVICE PROVIDER AT ANY TIME, WITH OR WITHOUT CAUSE.

Optionee has reviewed the Plan and this Option in their entirety, has had an opportunity	familiar with the terms and provisions thereof, and hereby accepts this Option subject to all of the terms and provisions thereof. to obtain the advice of counsel prior to executing this Option and fully understands all provisions of the Option. Optionee hereby dministrator upon any questions arising under the Plan or this Option. Optionee further agrees to notify the Company upon any
OPTIONEE	COMSCORE NETWORKS, INC.
/s/ Gian Fulgoni	/s/ Sheri Huston
Gian Fulgoni	By Sheri Huston
	Chief Financial Officer
	Title
Residence Address	

EXHIBIT A

1999 STOCK PLAN, AS AMENDED

EXERCISE NOTICE

comScore Networks, Inc. 11465 Sunset Hills Road, Ste. 200 Reston, VA 20190

Attention: Corporate Secretary
1. Exercise of Option. Effective as of today,, 20, the undersigned ("Optionee") hereby elects to exercise Optionee's option to purchase shares of the Common Stock (the "Shares") of comScore Networks, Inc. (the "Company") under and pursuant to the 1999 Stock Plan, as amended (the "Plan"), and the Stock Option Agreement dated, 20 (the "Option Agreement").
2. <u>Delivery of Payment</u> . Purchaser herewith delivers to the Company the full purchase price of the Shares, as set forth in the Option Agreement.
3. Representations of Optionee. Optionee acknowledges that Optionee has received, read and understood the Plan and the Option Agreement and agrees to abide by and be bound by their terms and conditions.
4. <u>Rights as Shareholder</u> . Until the issuance of the Shares (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company), no right to vote or receive dividends or any other rights as a shareholder shall exist with respect to the Optioned Stock, notwithstanding the exercise of the Option. The Shares shall be issued to the Optionee as soon as practicable after the Option is exercised. No adjustment shall be made for a dividend or other right for which the record date is prior to the date of issuance except as provided in Section 12 of the Plan.
5. <u>Company's Right of First Refusal</u> . Before any Shares held by Optionee or any transferee (either being sometimes referred to herein as the "Holder") may be sold or otherwise transferred (including transfer by gift or operation of law), the Company or its assignee(s) shall have a right of first refusal to purchase the Shares on the terms and conditions set forth in this Section (the "Right of First Refusal").
(a) Notice of Proposed Transfer. The Holder of the Shares shall deliver to the Company a written notice (the "Notice") stating: (i) the Holder's bona fide intention to sell or otherwise transfer such Shares; (ii) the name of each proposed purchaser or other transferee ("Proposed Transferee"); (iii) the number of Shares to be transferred to each Proposed Transferee; and (iv) the bona fide cash price or other consideration for which the Holder proposes to transfer the

Shares (the "Offered Price"), and the Holder shall offer the Shares at the Offered Price to the Company or its assignee(s).

- (b) Exercise of Right of First Refusal. At any time within thirty (30) days after receipt of the Notice, the Company and/or its assignee(s) may, by giving written notice to the Holder, elect to purchase all, but not less than all, of the Shares proposed to be transferred to any one or more of the Proposed Transferees, at the purchase price determined in accordance with subsection (c) below.
- (c) <u>Purchase Price</u>. The purchase price ("Purchase Price") for the Shares purchased by the Company or its assignee(s) under this Section shall be the Offered Price. If the Offered Price includes consideration other than cash, the Board of Directors of the Company in good faith shall determine the cash equivalent value of the non-cash consideration.
- (d) <u>Payment</u>. Payment of the Purchase Price shall be made, at the option of the Company or its assignee(s), in cash (by check), by cancellation of all or a portion of any outstanding indebtedness of the Holder to the Company (or, in the case of repurchase by an assignee, to the assignee), or by any combination thereof within thirty (30) days after receipt of the Notice or in the manner and at the times set forth in the Notice.
- (e) <u>Holder's Right to Transfer</u>. If all of the Shares proposed in the Notice to be transferred to a given Proposed Transferee are not purchased by the Company and/or its assignee(s) as provided in this Section, then the Holder may sell or otherwise transfer such Shares to that Proposed Transferee at the Offered Price or at a higher price, provided that such sale or other transfer is consummated within 120 days after the date of the Notice, that any such sale or other transfer is effected in accordance with any applicable securities laws and that the Proposed Transferee agrees in writing that the provisions of this Section shall continue to apply to the Shares in the hands of such Proposed Transferee. If the Shares described in the Notice are not transferred to the Proposed Transferee within such period, a new Notice shall be given to the Company, and the Company and/or its assignees shall again be offered the Right of First Refusal before any Shares held by the Holder may be sold or otherwise transferred.
- (f) Exception for Certain Family Transfers. Anything to the contrary contained in this Section notwithstanding, the transfer of any or all of the Shares during the Optionee's lifetime or on the Optionee's death by will or intestacy to the Optionee's immediate family or a trust for the benefit of the Optionee's immediate family shall be exempt from the provisions of this Section. "Immediate Family" as used herein shall mean spouse, lineal descendant or antecedent, father, mother, brother or sister. In such case, the transferee or other recipient shall receive and hold the Shares so transferred subject to the provisions of this Section, and there shall be no further transfer of such Shares except in accordance with the terms of this Section.
- (g) <u>Termination of Right of First Refusal</u>. The Right of First Refusal shall terminate as to any Shares upon the first sale of Common Stock of the Company to the general public pursuant to a registration statement filed with and declared effective by the Securities and Exchange Commission under the Securities Act of 1933, as amended.

- 6. <u>IRA</u>. Optionee acknowledges that the Shares shall be subject to certain transfer restrictions, rights of first refusal and co-sale set forth in that certain Fourth Amended and Restated Investor Rights Agreement, made as of August 1, 2003 by and among the Company, certain stockholders of the Company listed on the signatures pages thereto and the founders listed on the signature pages thereto, as such agreement is amended from time to time.
- 7. <u>Tax Consultation</u>. Optionee understands that Optionee may suffer adverse tax consequences as a result of Optionee's purchase or disposition of the Shares. Optionee represents that Optionee has consulted with any tax consultants Optionee deems advisable in connection with the purchase or disposition of the Shares and that Optionee is not relying on the Company for any tax advice.

8. Restrictive Legends and Stop-Transfer Orders

(a) <u>Legends</u>. Optionee understands and agrees that the Company shall cause the legends set forth below or legends substantially equivalent thereto, to be placed upon any certificate(s) evidencing ownership of the Shares together with any other legends that may be required by the Company or by state or federal securities laws:

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 (THE "ACT") AND MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, PLEDGED OR HYPOTHECATED UNLESS AND UNTIL REGISTERED UNDER THE ACT OR, IN THE OPINION OF COMPANY COUNSEL SATISFACTORY TO THE ISSUER OF THESE SECURITIES, SUCH OFFER, SALE OR TRANSFER, PLEDGE OR HYPOTHECATION IS IN COMPLIANCE THEREWITH.

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN RESTRICTIONS ON TRANSFER AND A RIGHT OF FIRST REFUSAL HELD BY THE ISSUER OR ITS ASSIGNEE(S) AS SET FORTH IN THE EXERCISE NOTICE BETWEEN THE ISSUER AND THE ORIGINAL HOLDER OF THESE SHARES, A COPY OF WHICH MAY BE OBTAINED AT THE PRINCIPAL OFFICE OF THE ISSUER. SUCH TRANSFER RESTRICTIONS AND RIGHT OF FIRST REFUSAL ARE BINDING ON TRANSFEREES OF THESE SHARES.

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN RESTRICTIONS ON TRANSFER, A RIGHT OF FIRST REFUSAL AND A RIGHT OF CO-SALE HELD BY THE ISSUER AND CERTAIN STOCKHOLDERS OF THE ISSUER PURSUANT TO THE ISSUER'S FOURTH AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT, AS AMENDED FROM TIME TO TIME. COPIES OF SUCH AGREEMENT MAY BE OBTAINED UPON WRITTEN REQUEST TO THE SECRETARY OF

THE ISSUER. SUCH TRANSFER RESTRICTIONS, RIGHT OF FIRST REFUSAL AND RIGHT OF CO-SALE ARE BINDING ON TRANSFEREES OF THESE SHARES.

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO A VOTING AGREEMENT CONTAINED IN THE ISSUER'S FOURTH AMENDED AND RESTATED INVESTOR RIGHTS AGREEMENT, AS AMENDED FROM TIME TO TIME. COPIES OF SUCH AGREEMENT MAY BE OBTAINED UPON WRITTEN REQUEST TO THE SECRETARY OF THE ISSUER. BY ACCEPTING ANY INTEREST IN SUCH SHARES THE PERSON ACCEPTING SUCH INTEREST SHALL BE DEEMED TO AGREE TO AND SHALL BECOME BOUND BY ALL THE PROVISIONS OF SAID VOTING AGREEMENT.

- (b) Stop-Transfer Notices. Optionee agrees that, in order to ensure compliance with the restrictions referred to herein, the Company may issue appropriate "stop transfer" instructions to its transfer agent, if any, and that, if the Company transfers its own securities, it may make appropriate notations to the same effect in its own records.
- (c) <u>Refusal to Transfer</u>. The Company shall not be required (i) to transfer on its books any Shares that have been sold or otherwise transferred in violation of any of the provisions of this Exercise Notice or (ii) to treat as owner of such Shares or to accord the right to vote or pay dividends to any purchaser or other transferee to whom such Shares shall have been so transferred.
- 9. <u>Successors and Assigns</u>. The Company may assign any of its rights under this Exercise Notice to single or multiple assignees, and this Exercise Notice shall inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer herein set forth, this Exercise Notice shall be binding upon Optionee and his or her heirs, executors, administrators, successors and assigns.
- 10. <u>Interpretation</u>. Any dispute regarding the interpretation of this Exercise Notice shall be submitted by Optionee or by the Company forthwith to the Administrator which shall review such dispute at its next regular meeting. The resolution of such a dispute by the Administrator shall be final and binding on all parties.
 - 11. Governing Law; Severability. This Exercise Notice is governed by the internal substantive laws but not the choice of law rules, of Virginia.
- 12. Entire Agreement. The Plan and Option Agreement are incorporated herein by reference. This Exercise Notice, the Plan, the Option Agreement and the Investment Representation Statement constitute the entire agreement of the parties with respect to the subject matter hereof and supersede in their entirety all prior undertakings and agreements of the Company and Optionee with respect to the subject matter hereof, and may not be modified adversely to the Optionee's interest except by means of a writing signed by the Company and Optionee.

Submitted by:	Accepted by:
OPTIONEE	COMSCORE NETWORKS, INC.
Gian Fulgoni	By
	Title
Address:	Address:
	11465 Sunset Hills Road, Ste 200
	Reston, VA 20190 Date Received
	-5-

EXHIBIT B

INVESTMENT REPRESENTATION STATEMENT

OPTIONEE: GIAN FULGONI

COMPANY: COMSCORE NETWORKS, INC.

SECURITY: COMMON STOCK

AMOUNT:

DATE:

In connection with the purchase of the above-listed Securities, the undersigned Optionee represents to the Company the following:

(a) Optionee is aware of the Company's business affairs and financial condition and has acquired sufficient information about the Company to reach an informed and knowledgeable decision to acquire the Securities. Optionee is acquiring these Securities for investment for Optionee's own account only and not with a view to, or for resale in connection with, any "distribution" thereof within the meaning of the Securities Act of 1933, as amended (the "Securities Act")

(b) Optionee acknowledges and understands that the Securities constitute "restricted securities" under the Securities Act and have not been registered under the Securities Act in reliance upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of Optionee's investment intent as expressed herein. In this connection, Optionee understands that, in the view of the Securities and Exchange Commission, the statutory basis for such exemption may be unavailable if Optionee's representation was predicated solely upon a present intention to hold these Securities for the minimum capital gains period specified under tax statutes, for a deferred sale, for or until an increase or decrease in the market price of the Securities, or for a period of one year or any other fixed period in the future. Optionee further understands that the Securities must be held indefinitely unless they are subsequently registered under the Securities Act or an exemption from such registration is available. Optionee further acknowledges and understands that the Company is under no obligation to register the Securities. Optionee understands that the certificate evidencing the Securities will be imprinted with a legend which prohibits the transfer of the Securities unless they are registered or such registration is not required in the opinion of counsel satisfactory to the Company, and any other legend required under applicable state securities laws.

(c) Optionee is familiar with the provisions of Rule 701 and Rule 144, each promulgated under the Securities Act, which, in substance, permit limited public resale of "restricted securities" acquired, directly or indirectly from the issuer thereof, in a non-public offering subject to the satisfaction of certain conditions. Rule 701 provides that if the issuer qualifies under Rule 701 at the

time of the grant of the Option to the Optionee, the exercise will be exempt from registration under the Securities Act. In the event the Company becomes subject to the reporting requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, ninety (90) days thereafter (or such longer period as any market stand-off agreement may require) the Securities exempt under Rule 701 may be resold, subject to the satisfaction of certain of the conditions specified by Rule 144, including: (1) the resale being made through a broker in an unsolicited "broker's transaction" or in transactions directly with a market maker (as said term is defined under the Securities Exchange Act of 1934); and, in the case of an affiliate, (2) the availability of certain public information about the Company, (3) the amount of Securities being sold during any three month period not exceeding the limitations specified in Rule 144(e), and (4) the timely filing of a Form 144, if applicable.

In the event that the Company does not qualify under Rule 701 at the time of grant of the Option, then the Securities may be resold in certain limited circumstances subject to the provisions of Rule 144, which requires the resale to occur not less than one year after the later of the date the Securities were sold by the Company or the date the Securities were sold by an affiliate of the Company, within the meaning of Rule 144; and, in the case of acquisition of the Securities by an affiliate, or by a non-affiliate who subsequently holds the Securities less than two years, the satisfaction of the conditions set forth in sections (1), (2), (3) and (4) of the paragraph immediately above.

(d) Optionee further understands that in the event all of the applicable requirements of Rule 701 or 144 are not satisfied, registration under the Securities Act, compliance with Regulation A, or some other registration exemption will be required; and that, notwithstanding the fact that Rules 144 and 701 are not exclusive, the Staff of the Securities and Exchange Commission has expressed its opinion that persons proposing to sell private placement securities other than in a registered offering and otherwise than pursuant to Rules 144 or 701 will have a substantial burden of proof in establishing that an exemption from registration is available for such offers or sales, and that such persons and their respective brokers who participate in such transactions do so at their own risk. Optionee understands that no assurances can be given that any such other registration exemption will be available in such event.

Signature of Op	tionee:			
Date:		, 20 <u> </u>		
-2-				

LEASE AGREEMENT

by and between

COMSCORE NETWORKS. INC. as "Tenant"

and

COMSTOCK PARTNERS. L.C. as "Landlord"

June 23, 2003

TABLE OF CONTENTS

SECTION 1. Definitions SECTION 2. Completion of Leased Premises; Term SECTION 3. Rent and Additional Charges

SECTION 4. Common Areas

SECTION 5. Services and Utilities

SECTION 6. Use of Leased Premises SECTION 7. Care of Leased Premises

SECTION 8. Rules and Regulations SECTION 9. Tenant's Alterations and Installations

SECTION 10. Name of Building; Tenant's Signs SECTION 11. Liability Insurance SECTION 12. Fire Insurance

SECTION 13. Damage by Fire or Other Casualty

SECTION 14. Condemnation

SECTION 15. Assignment and Subletting

SECTION 16. Default Provisions
SECTION 17. Bankruptcy Termination Provisions
SECTION 18. Landlord May Perform Tenant's Obligations

SECTION 19. Security Deposit

SECTION 20. Subordination

SECTION 21. Attornment

SECTION 22. Quiet Enjoyment

SECTION 23. Landlords Right of Access to Leased Premises

SECTION 24. Limitation on Landlord's Liability

SECTION 25. Estoppel Certificates SECTION 26. Surrender of Leased Premises SECTION 27. Holding Over

SECTION 27. Hotding Over SECTION 28. Parking SECTION 29. Leasing Commission SECTION 30. General Provisions

EXH	IIDI	TC

A.	Floor Plan of Leased Premises
B.	Base Building Definition
C.	Owner Approved Architects
D-1.	Space Design of Leased Premises
D-2.	Tenant Improvement Plans
E.	Building Interior Finish Specifications
F.	Rules and Regulations
G.	Security Deposit Promissory Note
H.	List of Landlord Affiliates
I.	Acknowledgement of Sublease form
J.	Financial Statement Certification Form

LEASE AGREEMENT

THIS LEASE AGREEMENT (this "Lease") is made and entered into this 23rd day of June, 2003, by and between (i) COMSTOCK PARTNERS, L.C., a Virginia limited liability company (hereinafter referred to as "Landlord"), and (ii) COMSCORE NETWORKS, INC., a Delaware corporation_(hereinafter referred to as "Tenant"), and referred to by singular pronouns of the neuter gender, regardless of the number and gender of the parties involved.

WITNESSETH: Upon and subject to the terms of this Lease, Landlord hereby leases to Tenant, and Tenant hereby leases from Landlord, the Leased Premises (as defined below), for the Term (as defined below), except that Landlord reserves and Tenant shall have no right in and to (a) the use of the exterior faces of all perimeter walls and windows of the Building, (b) the use of the roof of the Building, or (c) the use of the air space above the Building, except as specifically set forth herein.

1 DEFINITIONS

(a) General Interpretive Principles. For purposes of this Lease, except as otherwise expressly provided or unless the context otherwise requires, (i) the terms defined in this Section have the meanings assigned to them in this Section and include the plural as well as the singular, and the use of any gender shall be deemed to include all other genders; (ii) accounting terms not otherwise defined herein have the meanings assigned to them in accordance with generally accepted accounting principles; (iii) references herein to "Sections," "subsections," "paragraphs," and other subdivisions without reference to a document are to designated Sections, subsections, paragraphs, and other subdivisions of this Lease; (iv) a reference to a subsection without further reference to a section is a reference to such subsection as contained in the same Section in which the reference appears, and this rule shall also apply to paragraphs and other subdivisions; (v) the words "herein," "hereof," "hereonder," and other words" of similar import refer to this Lease as a whole and not to any particular provisions; (vi) the word "including," means "including, but not limited to"; (vii) daily rent is calculated on a thirty (30) day month applied to the number of days being charged, (viii) all amounts due Landlord hereunder are in Unites States dollars; and (ix) the words "months" mean calendar months and calendar years..

(b) Special Lease Definitions. As used in this Lease the following words and phrases shall have the meanings indicated:

Advance Rent: Forty Eight Thousand Three Hundred and Thirty Two and 53/00 (\$48,332.53) representing the Basic Rent for the first full month of the Term after the Lease Commencement Date, which Tenant shall pay to Landlord, on or before, July 1, 2003 by wire transfer, pursuant to the terms of this Lease.

Basic Rent: For each Lease Year, an amount equal to the product obtained by multiplying the Rentable Area of the Leased Premises leased by Landlord to Tenant during such Lease Year by the Rent per Square Foot for such Lease Year. The Basic Rent shall increase each year by 3% over the immediately prior year's Basic Rent. Therefore the Basic Rent for the second year is determined by multiplying the Basic Rent for the first year by 103% and for each subsequent year by multiplying the Basic Rent for the immediately prior year's Basic Rent by 103%.

Accordingly, the Basic Rent during the Initial Term hereunder will be as follows:

	Annual Rent	Monthly Rent	Rent/S.F.
Lease Year 1	\$579,990.40	\$48,332.53	\$22.00
Lease Year 2	597,390.11	49,782.51	\$22.66
Lease Year 3	615,311.82	51,275.98	\$23.34
Lease Year 4	633,771.17	52,814.26	\$24.04
Lease Year 5	652,784.30	54,398.69	\$24.76

Basic Rent Escalation: See definition of Basic Rent.

Building: The existing office building located at 11465 Sunset Hills Road, Reston, Virginia, including the parking lots and parking garage and Landlord's right, title and interest in and to the underlying land.

Building Rentable Area: The total net rentable area in the Building, which (although greater than the actual usable area) is agreed to be 89,221 square feet, including core factor, except as

otherwise provided in Section 3(b).

Brokers: There are no brokers involved in this transaction and the parties hereby indemnify each other in connection therewith.

<u>Landlord's Contractor</u>: Any and all professionals or trades people engaged by or on behalf of Landlord, or by Tenant at Landlords direction and/or expense, in connection with alterations and construction in the Leased Premises, either before or during the Term of this Lease, including but not limited to general contractors, sub-contractors, architects, engineers, and any other professionals or trades people typically associated with construction and/or alterations.

Landlord's Notice Address: COMSTOCK PARTNERS, L.C. 11465 Sunset Hills Road, Suite 510, Reston, Virginia 20190, Attention: Mr. Christopher Clemente, Manager, with copy to Mr. Marc Bettius, Cohen, Gettings, & Caulkins, 2200 Wilson Blvd., Arlington, Virginia 22201 and a copy to the property management company, as selected by Landlord. Currently the property management company is; The Rockcrest Group, 14800 Conference Center Drive, Suite 201, Chantilly Virginia 22151-3180. Landlord may change the property management company at its option and will notify Tenant in such event.

Lease Commencement Date: July 1, 2003

<u>Leased Premises</u>: The area located on the first, and second floors of the Building which is outlined in black on the floor plan, attached hereto as Exhibit A and incorporated herein, and containing 26,363.2 square feet of Rentable Area:

Operating Expense Base: For each calendar year ending during the Term, the sum of the 2001 actual operating expenses for each square foot of Building Rentable Area. Notwithstanding the fact that the Lease Commencement Date hereunder is July 1, 2003, the parties have agreed that the Operating Expense Base will be based on 2001 expenses.

Operating Expense Increases: For calendar year 2002 and each calendar year thereafter during the Term, an amount equal to the excess of Landlord's Operating Expenses for such calendar year over the Operating Expense Base.

Original Lease: One certain lease dated September 20, 2000 by and between Comstock Partners, LC (as Landlord) and Comscore Networks, Inc. (as Tenant) covering a portion of the Building containing 57,792.10 square feet (including the Leased Premises as defined herein), as amended July 3, 2002 (the "Original Lease"), terminated by Landlord pursuant to the terms thereof.

Pre-ordered Rent Payments: As defined in section 3(a) hereof.

Rent Payment Account: As defined in section 3(a) hereof.

Rent Payment Account Minimum Balance: As defined in section 3(a) hereof.

Rent Per Square Foot: The Basic Rent shall be Twenty Two and no/00 Dollars (\$22.00) per square foot of the Leased Premises during the First Lease Year. For each Lease Year thereafter during the Term, the Rent per Square Foot of the Leased Premises shall be increased by three percent (3%) as provided for in this Lease.

Rentable Area: The total rentable area of the Leased Premises, which (although greater than the actual usable area) is agreed to be 26,363.2 square feet.

Security Deposit: Upon execution of this Lease, in addition to paying Landlord the Advance Rent set forth herein, Tenant shall deliver and pay to Landlord a Security Deposit as set forth in Paragraph 19 hereof.

Tenant's Financial Reports: Throughout the Lease term Tenant agrees to provide Landlord with regular financial reports regarding Tenant ("Tenant's Financial Reports") and any affiliates of Tenant within forty five (45) days after the close of each calendar month (and the end of each fiscal year), as follows; (i) a Balance Sheet, (ii) a Statement of Profit and Loss for such period, and (iii) a Cash Flow Statement prepared and certified (the certification form to be in form and content attached hereto as Exhibit J, incorporated herein by reference, and signed by an officer of the Tenant) by Tenant or by an independent certified public accounting firm using generally accepted

accounting practices. In the event Tenant fails to deliver the Tenant's Financial Reports in a timely fashion Landlord shall provide written notice of such Default and Tenant shall have fifteen (15) days to cure such Default prior to Landlord exercising its remedies as a result of such Default

Tenant's Board Reports: Throughout the Lease term Tenant agrees to provide Landlord with regular written reports containing such information and details as are provided to the investors holding a seat on the Board of Directors of Tenant ("Tenants Board Reports") within five (5) business days of each meeting of the Board of Directors of Tenant. In the event Tenant fails to deliver the Tenant's Board Reports in a timely fashion Landlord shall provide written notice of such Default and Tenant shall have fifteen (15) days to core such Default prior to Landlord exercising its remedies as a result of such Default.

Tenant's Notice Address: The Tenant's notice address is: COMSCORE NETWORKS, INC. 11465 Sunset Hills Road, Suite 200 Reston, Virginia 20190, Attention: Corporate Counsel.

Tenant's Proportionate Share: The percentage, which the Rentable Area of the Leased Premises is of the Building Rentable Area. The Tenant's Proportionate Share is agreed to be thirty-one and 10/00 percent (31.10%).

Term: The period commencing on the Lease Commencement Date and ending on the last day of the calendar month which completes FIVE (5) YEARS after the Lease Commencement Date, but in any event the Term shall end on any date when this Lease is sooner terminated by Landlord as provided for herein.

(c) General Definitions. As used in this Lease the following words and phrases shall have the meanings indicated:

Additional Charges: All amounts payable by Tenant to Landlord under this Lease other than Basic Rent (including but not limited to Tenant's Additional Costs). All Additional Charges shall, unless otherwise provided herein, be due and payable within thirty (30) days of invoice and shall be deemed to be additional rent and all remedies applicable to the non-payment of Basic Rent shall be applicable thereto. Additional Charges shall include, but not be limited to, electrical override usage.

Additional Compensation: Within fifteen (15) days of the execution of this Lease, as additional compensation and as an inducement to Landlord to enter into this Lease with Tenant, the Tenant agrees to provide Landlord, or its assigns, with warrants for the purchase of 100,000 shares of the common stock of Tenant at a price not to exceed \$0.60 per share. The form and content of the warrant agreement shall be identical to the form and content of the previous warrant agreements provided by Tenant to Landlord, except for the price. Additionally, Tenant hereby reaffirms the validity of all previous Warrants granted to Landlord by Tenant.

Alterations: As defined in Section 9(a).

Business Days: All days except Saturdays, Sundays, and the following legal holidays: New Year's Day, Martin Luther King's Birthday, President's Day, Memorial Day, Fourth of July, Labor Day, Columbus Day, Veterans Day, Thanksgiving Day, Christmas Day, and those holidays designated by an Executive Order of the President of the United States or by Act of Congress.

<u>Default Interest Rate</u>: A rate per annum equal to a) the greater of (i) the sum of the prime rate of interest from time to time established and publicly announced by The Chase Manhattan Bank. N.A., New York, in its sole discretion, as its then applicable prime rate of interest to be used in determining actual interest rates to be charged to certain of its borrowers, said prime rate to change from time to time as and when the change is announced as being effective, plus four percent (4%), or fifteen percent (15%).

Event of Default: Any of the events set forth in Section 16(a) as an event of default.

Landlord: The Landlord named herein, its successors or assigns and any subsequent owner, lessees, or transferees, from time to time, of the Landlord's interest in the Building and their respective successors and assigns.

Lease: This Lease Agreement, as amended from time to time, and all Exhibits

incorporated herein and/or attached hereto.

<u>Lease Year</u>: The period of twelve (12) months commencing on the Lease Commencement Date and ending on the last day of the month which completes twelve (12) full calendar months after the Lease Commencement Date, and each 12-month period thereafter commencing on the first day after the end of the immediately preceding Lease Year, except that the last Lease Year shall end on the last day of the Term.

<u>Legal Requirements</u>: All laws, statutes, ordinances, orders, rules, regulations, and requirements of all federal, state, and municipal governments, and the appropriate agencies, officers, departments, boards, and commissions thereof, and the board of fire underwriters and/or the fire insurance rating organization or similar organization performing the same or similar functions, whether now or hereafter in force, applicable to the Building or any part thereof and/or the Leased Premises, as to the manner of use or occupancy or the maintenance, repair, or condition of the Leased Premises and/or the Building, and the usual and customary requirements of the carriers of all fire insurance policies maintained by Landlord on the Building.

Mortgage: Any mortgage, deed of trust, or other security instrument of record creating an interest in or affecting title to the Building or the land on which it is constructed, or both, or any part thereof, including a leasehold mortgage or sub-leasehold mortgage, and any and all renewals, modifications, consolidations, or extensions of any such instrument; Mortgagee shall mean the holder or beneficiary of any Mortgage. Tenant shall comply with all reasonable notices from Landlord's Mortgagee as to the manner of use or occupancy or the maintenance, repair or condition of the Leased Premises and/or the Building.

Non-disturbance: Landlord will provide Tenant a suitable non-disturbance agreement from any current or future mortgagees. In connection therewith Tenant shall execute documents reasonably requested by such lender.

Operating Expenses: Tenant shall pay Tenant's Proportionate Share of annual increases in Real Estate Taxes and Operating Expenses above the Calendar 2001 Base Year. An itemized breakdown of 2001 estimated Operating Expenses will be delivered to Tenant upon completion of Landlord's year-end consolidation. Detailed breakdowns of all charges to Tenant will be provided. The aggregate of all costs and expenses reasonably and customarily paid or incurred on a cash basis by Landlord in connection with the ownership, operation, servicing, and maintenance of the Leased Premises, the Building, the land on which the Building is constructed and any ancillary improvements constructed on the land, the surface and garage parking areas, and ingress/egress easements and private roadways servicing the Building, including, but not limited to, employees' wages, salaries, welfare and pension benefits and other customary and usual employee fringe benefits; payroll taxes; Real Estate Taxes; property owner's association dues, fees and contributions of any kind, electricity and other utility charges; telephone service; painting of public or other common areas of the Building; exterminating service; security services; trash removal; sewer and water charges; premiums for fire and casualty, liability, rent loss, workmen's compensations, sprinkler, water damage and other insurance; repairs, maintenance, additions and improvements made by Landlord to the Building (properly depreciating any capital improvements); building, janitorial and cleaning services and supplies; uniforms and dry cleaning; snow removal; landscaping maintenance; window cleaning; service contracts for the maintenance of elevators, boilers, HVAC, and other mechanical, plumbing, and electrical equipment; legal fees (other than legal fees relating to the enforcement of Landlord's rights under leases with tenants for space in the Building); accounting fees; advertising (except for advertising expenses and leasing fees relating to leasing space in the building); management fe

Notwithstanding anything in this Lease to the contrary, for purposes of the calculations to be made pursuant to this paragraph, Operating Expenses shall exclude (i) capital improvements except as provided under this definition of Operating Expenses, (ii) repairs and replacements, which under

sound accounting principles and practices should be classified as capital expenditures as determined by Landlord's independent accounting firm, depreciated as provided for above, (iii) painting, redecorating, or other work which Landlord performs for any other tenant or prospective tenant of the building other than painting, redecorating, or other work which is standard for the building and performed for tenants subsequent to their initial occupancy, (iv) repairs or other work (including rebuilding) occasioned by fire, windstorms, or other casualty, to the extent covered by insurance, or condemnation, (v) any cost (such as repairs, improvements, electricity, special cleaning or overtime services) to the extent such costs are included in tenants' rent or are expressly reimbursable to Landlord by tenants (as opposed to partial reimbursement in the nature of rent escalation provisions) or are separately charged to and payable by tenants or to the extent Landlord is entitled to compensation by insurance proceeds, (vi) leasing commissions and expenses of procuring tenants, including lease concessions and lease take-over obligations, (vii) depreciation, (viii) interest on and amortization of debt, (ix) taxes of any nature, excluding real estate taxes, but including interest and penalties for late payment of taxes, except as provided herein, (x) rent payable under any lease to which this lease is subject, (xi) wages or salaries of employees other than on-site employees for the building or employees specifically employed, in whole or in part, in connection with the ownership and maintenance of the Building, (xii) costs and expenses of enforcing leases against tenants, including legal fees, (xiii) managing agents' commissions in excess of rates then customarily charged by managing agents for comparable office buildings and, (xiv) expenses resulting from any violation by Landlord of the terms of any lease of space in the building or of any ground or underlying lease or mortgage to which this lease is subordinate.

In the event that pursuant to the terms of this Lease, Tenant is obligated to pay its proportionate share of Operating Expenses, Tenant shall have the right to audit Landlord's books and records as follows:

- A. Tenant shall be entitled at any reasonable time during business hours, after giving at least five (5) days prior written notice, to inspect Landlord's books and records relating to Tenant's proportionate share of Operating Expenses at the site of the location of such books and Records and to obtain an audit thereof by an independent auditor selected by Tenant (and reasonably acceptable to Landlord) to determine the accuracy of such amounts billed to Tenant by Landlord for the last two (2) calendar years immediately preceding the calendar year in which such notice is given.
- B. If such audit discloses a liability for Tenant's proportionate share of Operating Expenses which is less then the amount billed to, and paid by, Tenant, then Landlord shall within thirty (30) days refund to Tenant all amounts paid by Tenant in excess of the amount Tenant is actually required to pay as provided for herein ("Refund Amount").
- C. All costs of such audit shall be paid by Tenant However, in the event the Refund Amount is greater then five (5) percent (5%) of the amount for which Tenant is actually liable (as disclosed by the audit), all reasonable actual costs of such audit shall be paid by Landlord.

Option to Renew/Expansion: Tenant shall not have an option to renew this Lease and shall have no right to expansion space in the Building.

Option to Terminate: Landlord shall have the sole and exclusive option of terminating this Lease upon five (5) months written notice to Tenant, such termination being effective at any time after the third anniversary of this Lease.

Person: A natural person, a partnership, a limited liability company, a corporation, and any other form of business or legal association or entity.

Real Estate Taxes: All taxes, assessments, vault rentals, water and sewer rents, if any, and other charges, if any, general, special, or otherwise, including all assessments for schools, public betterment, and general or local improvements, which are mandatory or legally compelled, levied or assessed upon or with respect to the ownership of and/or all other taxable interests in the Building and the land on which it is built imposed by any public or quasi-public authority (including The Reston Association and any related or similar organization having jurisdiction over the Building and the ability to assess fees to the owner of the Building whether now existing or created after the date hereof) having jurisdiction and personal property taxes levied or assessed on Landlord's personal property used in connection with the operation, maintenance, and repair of the Building. Except for taxes, fees, charges, and impositions described in the next succeeding sentence, Real Estate Taxes shall not include any inheritance, estate, succession, transfer, gift, franchise, corporation, income, or profit tax or capital levy. If at any time during the Term the methods of taxation shall be altered so

that in addition to or in lieu of or as a substitute for the whole or any part of any Real Estate Taxes levied, assessed or imposed there shall be levied, assessed or imposed (i) a tax, license fee, excise or other charge on the rents received by Landlord, or (ii) any other type of tax or other imposition in lieu of, or as a substitute for, or in addition to, the whole or any portion of any Real Estate Taxes, then the same shall be included as Real Estate Taxes. A tax bill or true copy thereof, together with any explanatory or detailed statement of the area or property covered thereby, submitted by Landlord to Tenant shall be conclusive evidence of the amount of taxes assessed or levied, as well as of the items taxed. If any real property tax or assessment levied against the land, buildings or improvements covered thereby or the rents reserved therefrom, shall be evidenced by improvement or other bonds, or in other form, which may be paid in annual installments, only the amount paid or payable in any Lease Year shall be included as Real Estate Taxes for that Lease Year.

Substantially Completed: The completion of the construction or installation, or both, of the Landlord's improvements in question, except for any special order or long-lead items, to the extent that (i) only minor items remain unfinished, and (ii) such minor items do not prevent Tenant from occupying the Leased Premises for the use specified herein. It is understood and agreed by the parties that all construction and other improvements that are the responsibility of landlord are complete and satisfactory.

Taking: A taking of property or any interest therein or right appurtenant or accruing thereto, by condemnation or eminent domain or by action, proceedings, or agreement in lieu thereof, pursuant to governmental authority.

Tenant: The tenant named herein and any permitted assignee under Section 15.

Tenant's Additional Costs: In additional to any other provision hereof that provides for the Tenant to be responsible for certain costs incurred by Landlord, it is agreed and understood that Tenant shall be responsible for all costs of any kind ("Tenant's Additional Costs") incurred by Landlord in connection with (i) any repairs deemed necessary by Landlord (not to include any reasonably determined to be repairs of normal wear and tear) in any of the leased premises under the Original Lease based on an inspection made by Landlord and Tenant within ten (10) days of the Lease Commencement resulting from Tenant's occupancy or sub-letting of the leased premises under the Original Lease, (ii), any repairs to the Building reasonably deemed necessary by Landlord, including modifications or repairs required as a direct or indirect result of Tenant sequipment and personnel within the Building, (iii) any repairs reasonably deemed necessary by Landlord or as a direct or indirect result of Tenant's failure to abide by the Rules and Regulations and other applicable lease provisions, regarding the use, upkeep and care of the Building or the leased premises under the Original Lease or to strictly enforce same upon its Subtenants under the Original Lease, (iv) any electrical override usage charges or electrical submeter charges and after hours HVAC charges for services provided Tenant hereunder and under the Original Lease, (v) any charges due for services provided to Tenant by Landlord hereunder or under the Original Lease, (vi) legal costs incurred by Landlord in connection with enforcement of this Lease, and (vii) legal costs incurred by Landlord in connection with enforcement of this Lease, and (vii) legal costs incurred by Landlord in connection with enforcement of the Original Lease. The Tenant's Additional Charges shall be payable within fifteen (15) days of invoice by Landlord regardless of when the cost is incurred by Landlord, including such costs incurred by Landlord prior to the Lease Commencement Date as defined h

Tenant's Special Installations: As defined in Section 9(d)

<u>Unavoidable Delays</u>: Delays caused by strikes, acts of God, lockouts, labor difficulties, riots, explosions, sabotage, accidents, inability to obtain labor or materials, governmental restrictions or delays in obtaining required building permits or occupancy permits, enemy action, civil commotion, fire, unavoidable casualty, or similar causes not caused by and beyond the reasonable control of the Landlord.

2. COMPLETION OF LEASED PREMISES, SCHEDULE AND INSPECTIONS

Notwithstanding anything to the contrary contained in this Lease, the Original Lease, or elsewhere provided, it is understood and agreed that the Leased Premises are currently occupied by Tenant and are hereby unconditionally accepted by Tenant, in all respects, in their

"AS IS, WHERE IS" condition, and further that Landlord has fully satisfied all of its obligations regarding the completion of construction of the Leased premises under this Lease or the Original Lease, completion of any repairs that are the responsibility of the Landlord under this Lease or the Original Lease, completion of the Building as required by this Lease or the Original Lease, providing the Tenant with the Tenant Improvement Allowance as required by the Original Lease, providing Tenant the services as required under the Original Lease, and that Landlord has fully complied with all requirements under this Lease (except those requirements which by their nature are not required to be complied with at this time) and the Original Lease.

- (a) <u>Base Building Definition</u>: To the best knowledge and belief of Landlord the building shell was completed in accordance with the project specifications set forth in the Base Building Definition attached hereto as, Exhibit B and incorporated herein, (the "Base Building Definition").
- (b) Tenant Improvement Allowance: It is agreed and understood that pursuant to the Original Lease Landlord previously provided Tenant a Tenant Improvement Allowance in connection with the construction of Tenant Improvements within the Building, including within the Leased Premises, in the amount of One Million Five Hundred and Fifteen Thousand Seven Hundred and Forty and 72/00 Dollars (the "Tenant Improvement Allowance"), receipt and sufficiency of which is hereby acknowledged by Tenant for space planning, preparation of the Tenant Improvement Plans (as described below), architectural and engineering services related to the Tenant Improvement Plans, permitting required in connection with the Tenant Improvements (including modifications to the existing building specifications required as a result of the Tenant Improvement Plans), Tenant's building signage, and other costs incurred by Landlord in connection with the Tenant Improvement Plans or construction of the Tenant's Improvements. Accordingly, Landlord shall not provide any additional allowance for improvements or alterations to the Leased Premises in connection with this Lease. In the event of Tenant's full and faithful compliance with each and every term and condition of this Lease the Landlord shall not be entitled to any return of the Tenant Improvement Allowance, however in the event of Tenant's abandonment of the Leased Premises or Tenant's Default hereunder resulting in Tenant being evicted from the Leased Premises (as evidenced by court order or Landlord's Notice of Default and Termination as provided for herein) within five (5) years of the Lease Commencement Date Landlord shall, among other remedies provided for in this Lease, be entitled to the full and immediate repayment of the Tenant Improvement Allowance which Tenant shall repay to Landlord upon demand therefore. However, it is agreed and understood that the amount of the Tenant Improvement Date hereunder.
- (c) <u>Tenant's Improvements</u>: In accordance with the Original Lease, Landlord and Tenant jointly developed a mutually acceptable space plan and finishing schedule for the Leased Premises that met Tenant's requirements (the "Space Design"). Upon completion of the Space Design, an architectural firm was selected from those listed on Exhibit C, attached hereto and incorporated herein, to prepare the complete construction documents (the "Tenant Improvement Plans"). The Tenant Improvement Plans fully describe all leasehold improvements in connection with the Leased Premises (the "Tenant's Improvements") and include all required construction documents and specifications, finishing schedules, structural designs and plans, electrical designs and plans, plumbing designs and obcuments or items connection with obtaining building permits for the Tenant Improvement Plans and occupancy certificates or use permits for the Leased Premises. The Space Design and the Tenant Improvement Plans created in connection with the Original Lease are attached hereto as Exhibit D-1 and D-2 respectively, each hereby being incorporated herein. Landlord's specifications for interior building finishes, (the Building Interior Finish Specifications), are attached hereto as Exhibit E, and incorporated herein.
- (d) <u>Tenant's Costs</u>: All costs of any modifications of any kind to the Leased Premises that the Tenant desires shall be the sole responsibility of Tenant and shall be payable as provided below. In the event the Tenant desires to make any alterations ("Tenant Alterations") to the Leased Premises all subject work shall be strictly in accordance with this paragraph 2(d) and Paragraph 9 below, and in such event Tenant shall provide Landlord a written request describing, in adequate detail, the nature of any proposed Tenant Alterations and Landlord shall secure a bid for the proposed Tenant Alterations from Landlord's General Contractor and provide same to Tenant in writing, provided that the amount payable to Landlord's General Contractor, is commercially reasonable. In the event Tenant desires to proceed with the proposed Tenant Alterations, Tenant shall deposit cash

in the amount of one hundred percent (100%) of the cost identified in the bid for the proposed tenant Alterations with Landlord prior to commencement of the Tenant Alterations (the "Tenant Alterations Deposit"). Landlord's General Contractor, Signet Construction, Inc. shall be used for all Tenant Improvements, or such other contractor as selected by Landlord.

- (e) <u>Plan Approvals</u>: All Tenant Alteration Plans, and any related modifications to the Building shall be subject to Landlord's sole but reasonable approval. Prior to any work commencing, Landlord shall approve all plans and specifications. Prior to any work commencing all required permits shall be obtained by Landlord.
 - (f) Schedule: RESERVED
 - (g) Delays: RESERVED
- (h) <u>Subcontractors and Suppliers</u>: All sub-contractors and material suppliers performing work or supplying materials to the Building shall be selected by the Landlord's general contractor and shall be subject to the Landlord's approval in its sole but reasonable discretion. In order to protect the integrity and efficiency of the mechanical, electrical and plumbing systems, all mechanical, electrical and plumbing within the Tenant's space shall be designed by the design build team responsible for the mechanical, electrical and plumbing systems in the building.
- (i) <u>Inspections</u>: At the time Tenant surrenders the Leased Premises or at the end of the Term, or within ten (10) business days thereafter, Landlord and Tenant, or their respective agents, shall make a similar inspection of the Leased Premises to note the condition of the Leased Premises at the time of surrender and shall prepare a punch list of any items of repair that Tenant shall be responsible for completing, reasonable wear and tear excepted (the "Tenant's Punchlist"). Landlord shall not be obligated to refund to Tenant all or any part of the Security Deposit then being held by Landlord until all repairs that are the responsibility of Tenant are completed to Landlord's reasonable satisfaction. In the event Tenant fails to attend the inspection, as reasonably scheduled by Landlord, the Tenant shall be bound by the Tenant Punchlist, as prepared by Landlord.
- (j) Acceptance of Space: Tenant is currently in possession of the Leased premises and hereby acknowledges that the condition of the Leased Premises is acceptable in its present "as is" condition. Accordingly no repairs are needed and Tenant accepts the Leased Premises in its "as is" condition on the date hereof.

3. RENT AND ADDITIONAL CHARGES

(a) <u>Payment of Rent and Additional Charges</u>. Tenant shall pay the Basic Rent for each Lease Year in equal monthly installments in advance on the first day of each month during the Term, commencing on the Lease Commencement Date. The Basic Rent and all Additional Charges shall be paid promptly when due, in lawful money of the United States, without notice or demand and without deduction, diminution, abatement, counterclaim, or setoff of any amount or for any reason whatsoever, except as otherwise expressly provided in subsection (b), to Landlord at Landlord's Notice Address or a such other address or to such other person as Landlord may from time to time designate in writing. If Tenant makes any payment to Landlord by check, such payment shall be by check of Tenant and Landlord shall not be required to accept the check of any other person, and any check received by Landlord shall be deemed received subject to collection. If any check is mailed by Tenant, Tenant shall post such check in sufficient time prior to the date when payment is due so that such check will be received by Landlord on or before the date when payment is due. Tenant shall assume the risk of lateness or failure of delivery of the mails, and no lateness or failure of the mails will excuse Tenant from its obligation to have made the payment in question when required under this Lease. If, during the Term, Landlord receives two or more checks from Tenant which are returned by Tenant's bank for insufficient funds or are otherwise returned unpaid, Tenant agrees that all checks thereafter shall be either bank certified, cashiers', or treasurers' checks. Landlord shall be reimbursed by Tenant an amount equal to one hundred and fifty percent (150%) of all bank service charges resulting from any returned checks plus a handling fee of five hundred dollars (\$500.00). The rent reserved under this Lease shall be the total of all Basic Rent and Additional Charges, increased and adjusted as elsewhere herein provided, payable during the entire Term and, accord

directly to Landlord's account no later then the first regular business day of each calendar month (the "Rent Payment Account"). It is further agreed that throughout the Lease Term, no later then the 10th calendar day of each month, the Landlord shall receive written verification from the bank or financial institution holding the Rent Payment Account that Tenant has irrevocably pre-ordered automatic wire transfers from the Rent Payment Account to Landlord's account to occur on the first business day of each of the next two (2) months for the full payment of Basic Rent ("Pre-ordered Rent Payments") for the subject months and that the Rent Payment Account has sufficient balances to provide for said payments ("Rent Payment Account Minimum Balance").

- (b) Payment of Operating Expense Increases. Tenant shall pay as Additional Charges its Proportionate Share of any Operating Expense Increases in accordance with Section 1(b) for each calendar year, commencing with the calendar year 2002, it being understood and agreed that Tenant shall pay such Additional Charges for 2002 in spite of the fact that the Lease Commences July 1, 2003. Landlord shall make a reasonable estimate of Tenant's Operating Expense Increase for each calendar year, and Tenant shall pay to Landlord 1/12th of the amount so estimated on the first day of each month in advance. If Landlord's estimate of Tenant's Operating Expense Increases for any calendar year is received by Tenant after January 1 of the calendar year. Tenant shall pay to Landlord in a lump sum, within thirty (30) days after receipt of the estimate. the arrearage in the monthly estimates for each month in the calendar year before receipt of the estimate and shall pay the remaining monthly installments on the first day of each month in advance during the balance of the calendar year. After the end of each calendar year, Landlord shall submit to Tenant a statement setting forth in reasonable detail the Operating Expenses for such calendar year and the amount (if any) of Tenant's Operating Expense Increases for such calendar year. If Tenant's Operating Expense Increases so stated are more than the amount (if any) theretofore paid by Tenant for Operating Increases based on Landlord's estimate, Tenant shall pay to Landlord the deficiency within thirty (30) days after the submission of such statement. If Tenant's Operating Expense Increases so stated are less than the amount (if any) theretofore paid by Tenant for Operating Expense Increases based on Landlord's estimate, Landlord shall refund to Tenant the excess within thirty (30) days after submission of such statement. If either the Lease Commencement Date shall not coincide with the beginning of a calendar year or the last day of the Term shall not coincide with the end of a calendar year, then the amount of Operating Expense Increases payable for the calendar year in which the Lease Commencement Date or the last day of the Term occurs, as the case may be, shall be pro-rated on a daily basis between Landlord and Tenant based on the number of days in such calendar year in which this Lease is in effect. Tenant's obligations under this subsection to pay Operating Expense Increases and Landlord's obligation to reimburse Tenant for an overpayment of Operating Expenses shall survive the expiration of the Term. If any part of the Building is leased to tenants (hereinafter referred to as "Special Tenants") which, in accordance with the terms of their leases, provide their own cleaning and janitorial services, electrical services, or are not required to pay Operating Expense Increases on the basis of operating expenses for the Building which include substantially the same components as the Operating Expenses (as defined in this Lease), the following provisions shall apply: (i) the Building Rentable Area shall be reduced by the rentable area of the space leased to Special Tenants; (ii) Tenant's Proportionate Share shall be the percentage which the Rentable Area is of the Building Rentable Area (determined after the reduction specified in clause (i); and (iii) Operating Expenses shall be reduced by the sum of the amounts payable to Landlord by Special Tenants, in accordance with the terms of their leases, as reimbursements for Real Estate Taxes and expenses of owning, operating, managing and maintaining the Building and the amount of the applicable operating expense base under such Special Tenants' leases.
- (c) Interest. If Tenant fails to make any payment of Basic Rent or Additional Charges on the due date thereof, interest shall, at Landlord's option, accrue on the unpaid portion thereof from the due date at the Default Interest Rate, but in no event at a rate higher than the maximum rate allowed by law, and shall be payable on demand.
- (d) <u>Accord and Satisfaction</u>. No payment by Tenant, receipt or acceptance by Landlord of any lesser amount than the amount stipulated to be paid hereunder shall be deemed other than on account of the earliest stipulated Basic Rent or Additional Charges; nor shall any endorsement or statement on any check or letter be deemed an accord and satisfaction, and Landlord may accept any check or payment without prejudice to Landlord's right to recover the balance due or to pursue any other remedy available to Landlord.
- (e) <u>Late Payment Charge</u>. If Tenant fails to pay any Basic Rent or Additional Charges within five (5) days after the same become due and payable, Tenant shall also pay to Landlord a late payment service charge within thirty (30) days of Landlord's notice that a late payment service charge is due (to cover Landlord's administrative and overhead expenses of processing late payments) equal to the greater of \$100.00 or five percent (5%) of such unpaid sum

for each and every calendar month or part thereof after the due date that such sum has not been paid to Landlord. Such payment shall be deemed liquidated damages and not a penalty, but shall not excuse the untimely payment of rent.

4. COMMON AREAS

Throughout the Term, Tenant and its agents, employees and business invitees shall have the nonexclusive right, in common with others, to use the public lobbies, parking lots, elevator, corridors, stairways, and other common areas in the Building and the toilet rooms in public areas of multi-tenant floors in the Building. Landlord shall have the right at any time, without the Tenant's consent, to make reasonable changes to the arrangement or location of entrances, passageways, doors, doorways, corridors, stairs, toilet rooms, or other public portions of the Building, provided any such change does not unreasonably obstruct Tenant's access to the Leased Premises

5. SERVICES AND UTILITIES

- (a) <u>Services Provided</u>: Throughout the Term, Landlord agrees that the Building will be maintained in a manner befitting comparable Class A rental office buildings in Northern Virginia, and that, subject to Legal Requirements, it will furnish to Tenant the following services:
 - (1) Subject to the provisions of subsections (b) and (c), normal and usual electricity for lighting purposes and the operation of ordinary office equipment;
 - (2) Adequate supplies for toilet rooms located in public areas of the Building;
 - (3) Normal and usual cleaning and janitorial services after business hours on Business Days;
 - (4) Hot and cold running water in the toilet rooms;
- (5) Subject to the provisions of subsection (d), heating and air-conditioning to the Leased Premises when required for the comfortable occupancy of the Leased Premises, at reasonable temperatures, pressures, and degrees of humidity, and in reasonable volumes and velocities, between the hours of 8:00 a.m. and 6:00 p.m. on Business Days and between the hours of 9:00 a.m. and 12:00 p.m. on Saturdays unless Saturday is a legal holiday:
 - (6) Automatically operated elevator service twenty-four (24) hours a day, seven (7) days a week throughout the Term;
 - (7) All electric bulbs and fluorescent tubes in building standard light fixtures in the public areas of the Building and building standard fixtures within the Leased Premises;
- (8) A reasonable number of keys to the Leased Premises have already been provided to Tenant at no cost to Tenant. All additional keys including replacements for lost keys shall be issued only upon the payment of a reasonable actual cost for each additional key; and
 - (9) A security access system for the public areas of the Building and card keys or other means of entry into the Building.
- (b) <u>Electrical Supply</u>: Landlord shall not be liable in any way to Tenant for any failure or defect in the supply or character of electrical energy furnished to the Leased Premises by reason of any requirement, act or omission of the public utility serving the Building with electricity. Tenant's use of electrical energy in the Leased Premises shall not at any time exceed the capacity of any of the electrical conductors and equipment in or otherwise serving the Leased Premises. Tenant shall not install or operate in the Leased Premises any electrically operated equipment, including lighting, which uses electric current in excess of the allocable share of the Building system capacity without Landlord's written consent, which consent may be conditioned upon Tenant's agreement to pay an additional charge to compensate Landlord for Tenant's excessive consumption of electricity and to pay the cost of any additional wiring which may be required for the operation of such equipment. Tenant shall not connect any equipment or other electrical device to the electrical system of the Building that would require unusual or excessive electrical service or that would interfere with the adequate supply of electrical service to (i) other tenants within the Building, or (ii) the Building common facilities. Any feeders or risers to supply Tenants electrical requirements in addition to those originally installed, and all other equipment proper and necessary in connection with such feeders or risers, shall be installed by Landlord upon Tenant's request, at the sole cost and expense of Tenant, provided that, in Landlord's reasonable judgment, such additional feeders or risers are permissible under applicable laws and insurance regulations and the installation of such feeders or risers will not cause permanent damage or injury to the Building or cause or create a dangerous condition or unreasonably interfere with other tenants of the Building.

Electrical Use Limits: If, at any time or from time to time, the estimated connected electrical load (including lighting and power) used by Tenant's electrically operated equipment exceeds an average of eight (8) watts (6 watts for low voltage and 2 watts for high voltage) per square foot of the Leased Premises on a 120/208 volt panel board. Landlord may either (i) install a separate electric meter for the Leased Premises, at Tenant's sole cost and expense, and Tenant shall reimburse Landlord for the cost of electricity it consumes, as recorded by such meter, in excess of the amount of electricity that would be consumed by a tenant whose consumption of electricity was equal to, but did not exceed, the specified limits, or (ii) from time to time have a survey made by an independent electrical engineer or electrical consulting firm to be selected and paid for by Landlord to determine the amount of electricity consumed by Tenant in excess of the amount of electricity was equal to, but did not exceed, then specified limits, and Tenant shall pay to Landlord the cost of excess electricity it consumes as determined by such electrical engineer or consulting firm.

- (d) After Hours HVAC: Landlord shall provide heat and air-conditioning at times in addition to those specified in paragraph (5) of subsection (a) at Tenant's expense, provided Tenant gives Landlord notice prior to 10:00 a.m. on Fridays or the day preceding a holiday (in the case of after-hours service on Saturdays, Sundays, or holidays). Landlord shall initially charge Tenant for after-hours service at the rate of \$45.00 per hour (or, if the Leased Premises include more than one HVAC zone on the same floor and such after hours service is provided for portions of the Leased Premises containing more than one HVAC zone, at the rate of \$45.00 per hour per HVAC zone). Landlord reserves the right from time to time, in its sole discretion, to increase the hourly charge for said after-hours service, but in no event will the rate per hour charged to Tenant be more than an amount per hour which represents Landlord's reasonable estimate of its actual cost of providing such after hours service, including labor, cost of electricity and wear and tear on equipment, plus an allowance of ten percent (10%) thereof to cover general overhead as an Additional Charge hereunder. Tenant shall be permitted to include in the Tenant Improvements a Tenant controlled after hours HVAC system provided it adequately provides Landlord a means of accounting for the after hours use by Tenant, as determined by Landlord and in such event any future increases in the charge for after hour use of that portion of the HVAC system that is under the Tenant's control shall not include a markup to cover general overhead.
- (e) <u>Landlord's Use Rights</u>: Landlord reserves the right to erect, use, maintain, and repair pipes, conduits, cables, plumbing, vents, and wires in, to and through the Leased Premises as and to the extent that Landlord may now or hereafter deem to be necessary or appropriate for the proper operation and maintenance of the Building, or other tenants' installations in the Building, and the right at all times to transmit water, heat, air-conditioning, and electric current through such pipes, conduits, cables, plumbing, vents, and wires, provided that Landlord, in the exercise of such rights, shall not unreasonably inconvenience Tenant or unreasonably interfere with Tenant's use of the Leased Premises.
- (f) <u>Maintenance Access</u>: Landlord shall have unrestricted access to any and all air-conditioning facilities in the Leased Premises for the purpose of repairs, maintenance, alterations, and improvements, but in exercising its rights under this subsection Landlord shall use its best efforts to minimize interference with Tenant's business in the Leased Premises.
- (g) <u>Tenant's Efforts</u>: Tenant agrees to use reasonable efforts to keep or cause to be kept closed all window draperies or venetian blinds in the Leased Premises as and when necessary because of the sun's position whenever the air-conditioning system is in operation, and Tenant agrees at all times to cooperate fully with Landlord and to abide by all the reasonable regulations and requirements which Landlord may prescribe for the proper functioning and protection of the Building air-conditioning system.
- (h) <u>Service Interruptions</u>: Landlord reserves the right to stop the service of heating, air-conditioning, ventilating, elevator, plumbing, electricity, or other mechanical systems or facilities in the Leased Premises or the Building, if necessary by reason of accident or emergency, or for repairs, alterations, replacements, additions, or improvements which, in the reasonable judgment of Landlord, are desirable or necessary, until said repairs, alterations, replacements, additions, or improvements shall have been completed. The exercise of such right by Landlord shall not constitute an actual or constructive eviction, in whole or in part, or relieve Tenant from any of its obligations under this Lease, or impose any liability upon Landlord or its agents by reason of inconvenience or annoyance to Tenant, or injury to, or interruption of, Tenant's business, or otherwise, or entitle Tenant to any abatement or diminution of rent. Except in cases of emergency repairs, Landlord will give Tenant reasonable advance notice of any contemplated stoppage of any such repairs, alterations, replacements, additions, or improvements promptly. Landlord shall also perform any such work in a

manner designated to minimize interference with Tenant's normal business operations.

- (i) <u>Service Delays</u>: If Landlord shall fail to supply, or be delayed in supplying, any service expressly or implied to be supplied under this Lease, or shall be unable to make, or be delayed in making, any repairs, alterations, additions, improvements, or decorations, or shall be unable to supply, or be delayed in supplying, any equipment or fixtures, and if such failure, delay or inability shall result from Unavoidable Delays, such failure, delay or inability shall not constitute an actual or constructive eviction, in whole or in part, or relieve Tenant form any of its obligations under this Lease, or impose any liability upon Landlord or its agents by reason of inconvenience to Tenant, or injury to, or interruption of, Tenant's business, or otherwise, or entitle Tenant to any abatement or diminution of rent unless directly resulting from the gross negligence or willful misconduct of Landlord as determined by a court of competent jurisdiction.
- (j) <u>Voice, Data and other Communications Services</u>: Landlord shall make reasonable efforts to accommodate Tenant's need for additional riser space between floors one and two during the Lease Term and any extensions thereof for the installation of voice/data and other communications devices. All voice/data and/or other communications services shall only be provided to tenants within the Building by reputable providers of such services, as reasonably determined by the Landlord. Tenant shall have the right to include in the Tenant Improvement Plans a reasonable number of risers to be used solely for Tenant's internal (only within the Leased Premises) voice and data communications purposes. Landlord shall not open the secure risers without the consent of Tenant, not to be unreasonably withheld, conditioned or delayed. At Tenant's option such opening of the secure risers shall be supervised by Tenant or its representative.

6. USE OF LEASED PREMISES

- (a) <u>Permitted Uses</u>: Tenant shall use and occupy the Leased Premises solely for general office purposes strictly in accordance with the applicable zoning regulations and consistent with the character and dignity of the Building, and shall not use or permit or suffer the use of the Leased Premises for any other purpose whatsoever without the prior written consent of the Landlord which shall not be unreasonably conditioned, delayed or withheld. Tenant shall not permit or suffer the Leased Premises to be occupied by anyone other than Tenant except as provided by Section 15. Tenant shall at all times have access to the Leased Premises twenty-four (24) hours a day, seven (7) days a week, subject, however, in all respects to all the terms, covenants and conditions contained in this Lease. However, Landlord may regulate and restrict access to the Building at times other than normal business hours on Business Days for security purposes so long as Tenant's employees and agents have reasonable access to the Leased Premises without unreasonable inconvenience. Throughout the Term, Tenant shall not use, or permit the Leased Premises to be used, for the business of selling food, beverages, or tobacco products, except that Tenant may operate on the Leased Premises vending machines for the sale of food, beverages, and tobacco products exclusively to its employees, agents, assignees or their respective visitors.
- (b) <u>Use Restrictions</u>: Throughout the Term, Tenant covenants and agrees: (i) to pay before delinquency any and all taxes, assessments and public charges levied, assessed or imposed upon Tenant's business conducted in the Leased Premises, upon the leasehold estate created by this Lease or upon Tenant's fixtures, furnishings or equipment in the Leased Premises; (ii) not to use or permit or suffer the use of any portion of the Leased Premises for any unlawful purpose; (iii) not to use the plumbing facilities for any purpose other than that for which they were constructed, or dispose of any foreign substances therein; (iv) not to place a load on any floor exceeding the floor load per square foot which such floor was designed to carry in accordance with the plans and specifications of the Building, and not to install, operate or maintain in the Leased Premises any heavy item of equipment except in such manner as to achieve a proper distribution of weight; (v) not to strip, over-load, damage, or deface the Leased Premises, or the hallways, stairways, elevators parking facilities, or other public areas of the Building, or the fixtures therein or used therewith; (vi) not to move any furniture or equipment into or out of the Leased Premises except at such times and in such locations as Landlord may from time to time designate; (vii) not to install any other equipment of any kind or nature which will or may necessitate any changes, replacements or additions to or in the use of, the water system, heating system, air-conditioning system, or electrical system of the Leased Premises or the Building, without first obtaining the written consent of Landlord; and (ix) at all times to comply with all Legal Requirements.
- (c) <u>Legal Requirements</u>: Tenant will not use or occupy the Leased Premises in violation of any Legal Requirements. If any governmental authority, after the commencement of the Term, shall contend or declare that the Leased Premises are being used for a purpose which is in violation of any Legal Requirements, then Tenant shall, upon thirty (30) days' notice from Landlord,

immediately discontinue such use of the Leased Premises. If thereafter the governmental authority asserting such violation threatens, commences, or continues criminal or civil proceedings against Landlord for Tenant's failure to discontinue such use in addition to any and all rights, privileges and remedies given to Landlord under this Lease for default therein, Landlord shall have the right to terminate this Lease forthwith. Tenant shall indemnify and hold Landlord harmless from and against any and all liability for any such violation or violations.

- (d) Fire Insurance Limitations: Tenant shall not do, permit or suffer to be done any act, matter, thing, or failure to act in respect of the Leased Premises and/or the Building that will invalidate or be in conflict with fire insurance policies covering the Building or any part thereof, and shall not do, or permit anything to be done, in or upon the Leased Premises and/or the Building, or bring or keep anything therein, which shall increase the rate of fire insurance on the Building or on any property located therein. If, by reason of the failure of Tenant to comply with the provisions of this subsection, the fire insurance rate shall at any time be higher than it otherwise would be, then Tenant shall reimburse Landlord and any other tenant of the Building, on demand, for that part of all premiums for any insurance coverage that shall have been charged because of such violations by Tenant and which Landlord or such other tenant, or both, shall have paid on account of an increase in the rate or rates in its own policies of insurance. Tenant shall not be responsible for any increase in fire insurance rates generally applicable to office space in Fairfax County, Virginia, and not resulting from the particular manner in which Tenant uses the Leased Premises.
- (e) <u>Restricted Materials</u>: Tenant shall not bring or permit to be brought or kept in or on the Leased Premises any flammable, combustible, or explosive fluid, material, chemical or substance except standard cleaning fluid, standard equipment and materials (including magnetic tape) customarily used in conjunction with business machines and equipment of the type used from time to time by Tenant in reasonable quantities.

7. CARE OF LEASED PREMISES

- (a) <u>Tenant Care and Maintenance</u>: Tenant shall act with care in its use and occupancy of the Leased Premises and the Building and the fixtures therein and, at Tenant's sole cost and expense, shall furnish its own electric bulbs and fluorescent tubes for all non-building standard light fixtures in the Leased Premises and shall make all repairs and replacements to the Leased Premises, structural or otherwise, necessitated or caused by the acts, omissions, or negligence of Tenant or any Person claiming through or under Tenant or by the use or occupancy or manner of use or occupancy of the Leased Premises by Tenant or any such Person; however, the foregoing provisions of this subsection shall be subject to the provisions of Section 13. Without affecting Tenant's obligations set forth in the preceding sentence, Tenant, at Tenant's sole cost and expense, shall also (i) make all repairs and replacements, as and when necessary, to Tenant's Special Installations and to any Alterations made or performed by or on behalf of Tenant or any Person claiming through or under Tenant, and (ii) perform all maintenance and make all repairs and replacements, as and when necessary, to any air conditioning equipment, private elevators, escalators, conveyors, or mechanical systems (other than the Building's standard equipment and systems and other then as specifically approved in writing by Landlord) which may be installed in the Leased Premises, or elsewhere in the Building and serving the Leased Premises, by Landlord, Tenant, or others. However, except as otherwise provided in this Lease, Tenant shall not have any right to install air-conditioning equipment, elevators, escalators, conveyors, or mechanical systems. In addition to the foregoing, all damage or injury to the Leased Premises and to its fixtures, appurtenances and equipment or to the Building or to its fixtures, appurtenances and equipment caused by Tenant moving property in or out of the Building or by installation or removal of furniture, fixtures, or other property by
- (b) <u>Landlord Repairs</u>: Except as otherwise provided in subsection (a), Landlord shall make the following repairs as and when necessary: (i) structural repairs to the Leased Premises and Building; (ii) repairs required in order to provide the elevator, plumbing, electrical, heating, and air-conditioning services to be furnished by Landlord pursuant to this Lease; (iii) repairs to exterior portions of the Building, including the windows, balconies, parking areas and roof thereof; and (iv) other repairs to the Building necessary for Tenant's permitted use and enjoyment of the Leased Premises. Landlord's obligations under the preceding sentence shall not accrue until after notice by Tenant to Landlord of the necessity for any specific repair.

RULES AND REGULATIONS

Tenant shall comply with, and shall cause its agents, employees and invitees to, comply with and observe all reasonable rules and regulations concerning the use, management, operation, safety, and good order of the Leased Premises, the Building and the Building parking areas which may from time to time be promulgated by Landlord, provided that such rules and regulations are not inconsistent with the provisions of this Lease and do not materially interfere with Tenant's permitted use of the Leased Premises. Initial rules and regulations, which shall be effective until amended by Landlord (provided such amendments do not unreasonably interfere with Tenant's business), are attached to this Lease as Exhibit F hereto and incorporated herein. Tenant shall be deemed to have received notice of any amendment to the rules and regulations when a copy of such amendment has been delivered to Tenant at the Leased Premises or has been mailed to Tenant in the manner prescribed for the giving of notices. Landlord shall not be responsible to Tenant for any violation of the rules and regulations, or the covenants or agreements contained in any other lease, by any other lease, by any other tenant of the Building, or such tenant's agents, employees or invitees, and Landlord may waive in writing, or otherwise, any or all of the rules or regulations in respect of any one or more tenants.

9. TENANT'S ALTERATIONS AND INSTALLATIONS

Notwithstanding anything to the contrary contained in this Lease, the Original Lease, or elsewhere provided, it is understood and agreed that the Leased Premises are currently occupied by Tenant and are hereby unconditionally accepted by Tenant, in all respects, in their "AS IS, WHERE IS" condition, and further that Landlord has fully satisfied all of its obligations regarding the completion of construction of the Leased premises under this Lease or the Original Lease, completion of any repairs that are the responsibility of the Landlord under this Lease or the Original Lease, completion of the Building as required by this Lease or the Original Lease, providing the Tenant with the Tenant Improvement Allowance as required by the Original Lease, providing Tenant the services as required under the Original Lease, and that Landlord has fully complied with all requirements under this Lease (except those requirements which by their nature are not required to be complied with at this time) and the Original Lease.

(a) <u>Alterations</u>: Tenant shall not make or perform, or permit the making or performance of, any alterations, installations, improvements, additions or other physical changes in or about the Leased Premises (referred to collectively as "<u>Alterations</u>") without Landlord's prior written consent... All plans, specifications and details for such Alterations, and all contractors performing the Alterations are subject to the prior written approval of Landlord. In the event Landlord grants such consent and permits Tenant to contract out such work, such Alterations shall be made and performed in conformity with and subject to the following provisions: (i) all Alterations shall be made and performed at Tenant's sole cost and expense and at such time and in such manner as Landlord may reasonably from time to time designate; (ii) all Alterations shall be performed by adequately insured contractors approved by Landlord and in a good and workmanlike manner in accordance with all applicable Legal Requirements, and Tenant shall indemnify and hold harmless Landlord from and against any and all costs, expenses, claims, liens and damages to person or property resulting from the making of any such alterations, decorations, additions or improvements in or to the Leased Premises or the Building; (iii) no Alteration shall affect any part of the Building other than the Leased Premises or adversely affect any service required to be furnished by Landlord to Tenant or to any other tenant or occupant of the Building; (iv) all business machines and mechanical equipment shall be placed and maintained by Tenant in settings sufficient in Landlord's reasonable judgment to absorb and prevent vibration, noise and annoyance to other tenants or occupants of the Building; (v) Tenant shall submit to Landlord reasonably detailed written plans and specification and shall not commence any such Alteration without first obtaining Landlord's written approval of such plans and specifications; (vi) all Alterations in or to the electrical facilities in or s

- (b) <u>Unauthorized Alterations</u>: If Tenant shall be in Default under this Section by reason of the making of any Alteration not hereby authorized or by reason of failure to give any notice or to obtain any approval required herein, Tenant may cure such default within the applicable grace period provided in this Lease, and if Tenant fails to do so Landlord may correct or remove the same and Tenant shall be liable for any and all costs and expenses incurred by Landlord in such removal.
- (c) <u>Installed Fixtures</u>: Except to the extent specifically provided in sub-section (d), all appurtenances, fixtures, improvements, additions and other property attached to or installed in the Leased Premises, whether by Landlord or Tenant or others, and whether at Landlord's expense, or Tenant's expense, or the joint expense of Landlord and Tenant, which are affixed to walls, floors or ceilings or which cannot be removed without structural damage to the Building, shall be and remain the property of Landlord. Any replacements of any property of Landlord, whether made at Tenant's expense or otherwise, shall be and remain the property of Landlord except as agreed to in writing by Landlord prior to Lease Execution or prior to commencing such Alterations. Notwithstanding anything to the contrary set forth herein or elsewhere provided, to the extent that Tenant installs, or previously installed in any of the Leased Premises (as defined herein and as defined in the Original Lease), any fixtures, including but not limited to built in shelving, cabinetry, desks or workstations of any kind including removable workstations shown on the Tenant Improvement Plans, appliances, light fixtures, and built-in audio-visual equipment and communication equipment (excluding phone sets) all such items are considered fixtures of the Building and shall be retained by Landlord. To the extent such items were installed on the 3rd or 4th floor of the Building under the Original Lease, Tenant shall not remove those items from the 3rd or 4th floor, as the case may be and to the extent furniture was located on the 3rd or 4th floor of the Building prior to the Effective Date hereof, Tenant shall not remove those items from the 3rd or 4th floor, and hereby conveys all of its right, title and interest in such items to Landlord free and clear of any liens.
- (d) <u>Tenant's Special Installations</u>: All furniture, furnishings and trade fixtures, excepting lighting fixtures and equipment, but including, without limitation, business machines and equipment, vaults, vault doors and door frames, and vault equipment, if any, safe deposit equipment, counterscreens, grillwork, cages, partitions which are moveable, railings, raised floors, equipment relating to food preparation, food storage and serving, dish washing and cleaning devices and any moveable property, installed by or at the expense of Tenant shall remain the property of Tenant and are referred to herein as "Tenant's Special Installations." Tenant may at its expense remove all or any part of said property at mules Landlord shall otherwise consent in writing. Upon removal of any or all of said property Tenant shall then repair all damage. Any of Tenant's Special Installations which are not removed from the Leased Premises at the expiration of the Term shall be deemed to have been abandoned by Tenant and may be disposed of by Landlord without liability to Tenant.
- (e) <u>Mechanic's Liens</u>: Notice is hereby given that Landlord shall not be liable for any labor or materials furnished or to be furnished to Tenant upon credit, and that no mechanic's, materialman's or other lien for any such labor or materials shall attach to or affect the reversion or other estate or interest of Landlord in and to the Leased Premises or the Building. Whenever and as often as any mechanic's lien or materialman's lien shall have been filed against the Leased Premises or the Building based upon any act or interest of Tenant or of anyone claiming through Tenant, or if any lien or security interest with respect thereto shall have been filed affecting any materials, machinery or fixtures used in the construction, repair or operation thereof or annexed thereto by Tenant or its successors in interest, Tenant shall forthwith take such action by bonding, deposit or payment as will remove or satisfy the lien or other security interest and in default thereof after the expiration of fifteen (15) days after notice to Tenant, Landlord, in addition to any other remedy under this Lease, may pay the amount secured by such lien or security interest or discharge the same by deposit and the amount so paid or deposited shall be collectible as additional rent. The provisions of this subsection shall not be applicable to liens filed with respect to work done for Tenant's account by Landlord.

10. NAME OF BUILDING; TENANT'S SIGNS

- (a) <u>Building Name</u>: The name of the Building shall be 11465 Sunset Hills Road or such other name selected by Landlord in its sole discretion. Landlord expressly reserves the right to have the Building designated by a street number or numbers and to affix to the Building, at locations designated by Landlord, signs indicating any such number or numbers and to change the name of the Building as selected from time to time by Landlord.
 - (b) Roof Rights: Landlord has not granted to Tenant any rights in or to the roof or

the outer side of the outside walls or windows of the Building, control of which is hereby reserved by Landlord except that Tenant shall have non-exclusive access to and the use of its pro-rata share of available building roof (as determined by Landlord) for the installation and maintenance of communications equipment of Tenant. Landlord will require detailed specifications for review and approval to be provided to Landlord and it's chosen consultant at least thirty (30) days prior to the date Tenant desires installation to commence. Any reasonable cost of landlord's consultant in connection with review and approval of the subject specifications and plans shall be reimbursed by Tenant promptly upon request therefore. All roof access will be coordinated with Landlord's management. Any building penetration shall be subject to the approval of Landlord (and its consultant's) in Landlord's sole and absolute discretion. Tenant will obtain all required permits and comply with all applicable restrictions at its sole cost and shall be solely responsible for all costs associated with installation, maintenance and removal of Tenant's roof top equipment and of any associated building penetrations. Tenant also agrees to be responsible for present and future damages to said roof as the result of Tenant's access and use of the roof as described herein.

(c) <u>Signage</u>: Tenant shall not display or erect any lettering, signs, advertisements, awnings or other projections on the exterior of the Leased Premises or in the interior of the Leased Premises if visible from a public way, except for customary hallway door lettering or interior suite signage visible to the public way (approved in writing in advance by Landlord), and except that Tenant shall be entitled to maintain its existing exterior building signage subject to Tenant continuing to occupy the Leased Premises in its entirety and provided Tenant has not been in default beyond any applicable cure period. Landlord shall provide Tenant with a prominent ("top billing") location of its name on the existing building monument sign incorporated into the project by Landlord provided Tenant continues to occupy the Leased Premises in its entirety and fully and fully amounts due Landlord hereunder. The Tenant shall not utilize more then its pro-rata share of signage square feet as provided for in local zoning ordinances. The Tenant shall be solely responsible for obtaining all required permits and approvals and shall be solely responsible for all costs associated with permitting, installation, maintenance and removal of its signage. Landlord will require detailed specifications for review and approval, and installation will be coordinated with Landlord's management. Any building penetration shall be subject to the approval of Landlord (and its consultant's) in Landlord's sole and absolute discretion. Landlord shall provide a directory tablet in the main lobby of the Building, at its expense, upon which Landlord, at Landlord's expense, will affix Tenant's name and a reasonable number of names of its officers, partners or employees, Landlord, at Landlord's expense, shall provide a reasonable number of building standard suite identification signs. Directory listings and suite signage for any sub-tenants of Tenant shall be at Tenant's expense. The size, color, and style of such directory and names affixed thereto shall be

11. LIABILITY INSURANCE

(a) General Liability Insurance: Tenant, at Tenant's sole cost and expense, shall obtain and maintain in effect at all times during the Term, a policy of comprehensive general public liability insurance with broad form property damage endorsement, naming Landlord and (at Landlord's request) any Mortgagee of the Building and any management agent as additional insured(s), protecting Landlord, Tenant and any such Mortgagee and management agent against any liability for bodily injury, death or property damage occurring upon, in or about any part of the Building or the land on which it is built, the Leased Premises or any appurtenances thereto, with such policies to afford protection to the limit of not less than One Million Dollars (\$1,000,000.00) with respect to bodily injury or death to any one person, to the limit of not less than Three Million Dollars (\$3,000,000.00) with respect to bodily injury or death to any number of persons in any one accident, to the limit of not less than One Million Dollars (\$1,000,000.00) with respect to damage to the property of any one owner from one occurrence, and with a deductible of no greater than One Thousand Dollars (\$1,000.00) per occurrence. Such comprehensive liability insurance may be effected by a policy or policies of blanket insurance which cover other property in addition to the Leased Premises, provided that the protection afforded thereunder shall be no less than that which would have been afforded under a separate policy or policies relating only to the Leased Premises and provided further that in all other respects any such policy shall comply with the other provisions of this Section.

(b) <u>Policy Restrictions</u>: The insurance policy required to be obtained by Tenant under this Section: (i) shall be issued by an insurance company of recognized responsibility licensed to do business in the jurisdiction in which the Building is located; and (ii) shall be written as primary policy coverage and not contributing with or in excess of any coverage which Landlord may carry. Neither the issuance of any insurance policy required under this Lease, nor the minimum limits

specified herein with respect to Tenant's insurance coverage, shall be deemed to limit or restrict in any way Tenant's liability arising under or out of this Lease. With respect to each insurance policy required to be obtained by Tenant under this Section, on or before the Lease Commencement Date, and at least thirty (30) days before the expiration of the expiring policy or certificate previously furnished, Tenant shall deliver to Landlord a certificate of insurance therefor, together with evidence of payment of all applicable premiums. Each insurance policy required to be carried hereunder by or on behalf of Tenant shall provide (and any certificate evidencing the existence of each such insurance policy shall certify) that such insurance policy shall not be cancelled unless Landlord shall have received thirty (30) days' prior written notice of cancellation.

- (c) <u>Hold Harmless</u>: Except for the willful gross negligent acts or omissions of Landlord or its agents or employees, Tenant hereby agrees to indemnify and hold harmless Landlord from and against any and all claims, losses, actions, damages, liabilities, and expenses (including attorneys' fees) that (i) arise from or are in connection with Tenant's possession, use, occupancy, management, repair, maintenance, or control of the Leased Premises, or any portion thereof, or (ii) arise from or are in connection with any willful or negligent act or omission of Tenant or Tenant's agents, employees, invitees, or subtenants, or (iii) result from any default, breach, violation, or nonperformance of this Lease or any provisions therein by Tenant, or (iv) arise from injury or death to persons or damage to property sustained on or about the Leased Premises. Tenant shall, at its own cost and expense, defend any and all actions, suits, and proceedings which may be brought against Landlord with respect to the foregoing or in which Landlord may be impleaded. Tenant shall pay, satisfy, and discharge any and all money judgments which may be recovered against Landlord in connection with the foregoing.
 - (d) Unavailability in the marketplace of any insurance required herein shall not be excused as a force majure.

12. FIRE INSURANCE

- (a) Landlord shall, throughout the Term, at its expense, keep the Building, but not Tenant's Special Installations and Alterations or Tenant's furniture, furnishings, trade fixtures or property removable by Tenant under the provisions of this Lease, insured against all loss or damage by fire with extended coverage in such amount as any first Mortgagee of the Building may from time to time require. Tenant shall, throughout the Term, at its expense, keep Tenant's Special Installations and Alterations and Tenant's personal property insured against all loss or damage by fire with extended coverage in an amount sufficient to prevent Tenant from becoming a co-insurer. Tenant's policies of insurance shall contain an appropriate clause or endorsement under which the insurer agrees that such policy shall not be cancelled without at least thirty (30) days notice to Landlord.
- (b) Landlord and Tenant will (i) if requested, advise the other as to the provisions of fire and extended coverage insurance policies obtained pursuant to this Section, and (ii) notify the other promptly of any change in the terms of any such policy which would affect such provisions.

13. DAMAGE BY FIRE OR OTHER CASUALTY

In the event of loss of, or damage to, the Leased Premises or the Building by fire or other casualty, the rights and obligations of the parties hereto shall be as follows:

- (a) If the Leased Premises or any part thereof shall be damaged by fire or other casualty, Tenant shall give prompt notice thereof to Landlord, and Landlord, upon receiving such notice, shall proceed promptly and with reasonable diligence, subject to Unavoidable Delays and a reasonable time for adjustment of insurance losses, to repair, or cause to be repaired, such damage in a manner designed to minimize interference with Tenant's occupancy (but with no obligation to employ labor at overtime or other premium pay rates). If the Leased Premises or any part thereof shall be rendered untenantable by reason of such damage, whether to the Leased Premises or the Building, the Basic Rent and Additional Charges shall propriorionately abate for the period from the date of such damage to the date when such damage shall have been repaired for the Premises or damaged shall be rendered untenantable. However, if, prior to the date when all of such damage shall have been repaired, any part of the Leased Premises so damaged shall be rendered tenantable and shall be used or occupied by Tenant, then the amount by which the Basic Rent and Additional Charges shall abate shall be equitably apportioned for the period from the date of any such use.
- (b) If as a result of fire or other casualty more than one-half (1/2) of the Building Rentable Area is rendered untenantable, Landlord within sixty (60) days from the date of such fire or casualty may terminate this Lease by notice to Tenant, specifying a date, not less than twenty (20)

nor more than forty (40) days after the giving of such notice, on which the Term shall expire as fully and completely as if such date were the date herein originally fixed for the expiration of the Term. If the Leased Premises are damaged as a result of fire or other casualty and if the damage to the Leased Premised (but not including Tenant's Special Installations or Alterations) is so extensive that such damage cannot be substantially repaired within one hundred and eighty (180) days from the date of the fire or other casualty may terminate this Lease by notice to the other, specifying a date, not less than twenty (20) nor more than forty (40) days after the giving of such notice, on which the Term shall expire as fully and completely as if such date were the date originally fixed for the expiration of the Term. If either Landlord or Tenant terminates this Lease, the Basic Rent and Additional Charges shall be apportioned as of the date of such fire or other casualty. If neither Landlord nor Tenant so elects to terminate this Lease, then Landlord shall proceed to repair the damage to the Building and the damage to the Leased Premises (but not Tenant's Special Installations or Alterations), if any shall have occurred, and the Basic Rent and Additional Charges shall meanwhile be apportioned and abated all as provided in subsection (a). However, if such damage is not repaired and the Leased Premises and the Building restored to reasonably the same condition as they were prior to such damage within two hundred and seventy (270) days from the date of such damage (such 270-day period to be extended by the period of any Unavoidable Delays plus a reasonable time for adjustment of insurance losses), Tenant, within thirty (30) days from the expiration of such 270-day period (as the same may be extended), may terminate this Lease by notice to Landlord, specifying a date not more than sixty (60) days after the giving of such notice on which the Term shall expire as fully and completely as if such date were the date herein

- (c) If the Leased Premises shall be rendered untenantable to the extent of eighty percent (80%) or more by fire or other casualty during the last six (6) months of the Term, Landlord or Tenant may terminate this Lease upon notice to the other party given within ninety (90) days after such fire or other casualty specifying a day, not less than twenty (20) days nor more than forty (40) days after the giving of such notice, on which the Term shall expire as fully and completely as if such date were the date originally fixed for the expiration of the Term. If either Landlord or Tenant terminates this Lease pursuant to this subsection, the Basic Rent and Additional Charges shall be apportioned as of the date of such fire or casualty.
- (d) Landlord shall not be required to repair or replace any of Tenant's Special Installations or Alterations or any other personal property of Tenant and no damages, compensation, or claim shall be payable by Landlord for inconvenience, loss of business, or annoyance arising from any repair or restoration of any portion of the Leased Premises or of the Building, but the foregoing shall not be deemed to relieve Landlord of liability for its breach of any covenant of this Lease.
- (e) The provisions of this Section shall be considered an express agreement governing any instance of damage or destruction of the Building or the Leased Premises by fire or other casualty, and any law now or hereafter in force providing for such a contingency in the absence of express agreement shall have no application.
- (f) Notwithstanding any other provisions of this Lease, Landlord shall not be liable or responsible for, and Tenant hereby releases Landlord and its partners, shareholders, officers, directors, agents, and employees from, any and all liability or responsibility to Tenant or any Person claiming by, through or under Tenant, unless caused by Landlord's gross negligence, by way of subrogation or otherwise, for any injury, loss, or damage to Tenant's property covered by a valid and collectible fire insurance policies which could give rise to a right of subrogation against Landlord a clause or endorsement whereby the insurer(s) shall waive any rights of subrogation against Landlord, and Tenant shall pay any additional premium required therefor.
- (g) Notwithstanding any other provision of this Lease, Tenant shall not be liable or responsible for, and Landlord hereby releases Tenant and its partners, shareholders, officers, directors, agents, and employees from, any and all liability or responsibility to Landlord or any Person claiming by, through or under Landlord, unless caused by Tenant's gross negligence, by way of subrogation or otherwise, for any injury, loss, or damage to Landlord's property covered by a valid and collectible fire insurance policy with extended coverage endorsement. Landlord shall require its insurer(s) to include in all of Landlord's insurance policies which could give rise to a right of subrogation against Tenant a clause or endorsement whereby the insurer(s) shall waive any rights of subrogation against Tenant, and Landlord shall pay any additional premium required therefor.
 - (h) The proceeds payable under all fire and other hazard insurance policies

maintained by Landlord on the Building shall belong to and be the property of Landlord, and Tenant shall not have any interest in such proceeds. Tenant agrees to look to its own fire and hazard insurance policies in the event of damage to Tenant's Special Installations or Alterations or its personal property.

14. CONDEMNATION

(a) In the event of a Taking of the whole of the Leased Premises, this Lease shall terminate as of the date of such Taking. If only a part of the Leased Premises shall be so taken then, except as otherwise provided in this subsection, this Lease shall continue in force and effect but, from and after the date of the Taking, the Basic Rent and Additional Charges shall be equitably reduced on the basis of the portion of the Leased Premises so taken. If a part of the Building shall be taken, and if either (i) the part of the Building so taken contains more than twenty-five percent (25%) of the Rentable Area of the Leased Premises inmediately prior to such Taking, or (ii) in Landlord's reasonable opinion it shall be impracticable to continue to operate the Building, then Landlord, at Landlord's option, may give to Tenant within sixty (60) days after the date upon which Landlord shall have received notice of the Taking, thirty (30) days notice of termination of this Lease. If a part of the Building so taken contains more than twenty-five percent (25%) of the Rentable Area of the Leased Premises immediately prior to such Taking, or (ii) by reason of such Taking, Tenant no longer has reasonable means of access to the Leased Premises, then Tenant, at Tenant's option, may give to Landlord within sixty (60) days after the date upon which Tenant shall have received notice of such Taking, thirty (30) days notice of termination of this Lease. If thirty (30) days notice of termination is given by Landlord or Tenant, this Lease shall terminate upon the expiration of the thirty (30) day period. If this Lease is terminated pursuant to the foregoing provisions of this subsection, then, to the extent permitted by applicable law and such Taking, Tenant shall have access to the Leased Premises in order to remove Tenant's Special Installations and any other personal property then owned by Tenant and which Tenant is entitled to remove pursuant to this Lease, Landlord shall repair, alter, and restore the remaining portions of the Lease

(b) Landlord shall have the exclusive right to receive any and all awards made for damages to the Leased Premises and the Building accruing by reason of a Taking or by reason of anything lawfully done in pursuance of public or other authority. Tenant hereby releases and assigns to Landlord all of Tenant's rights to such awards, and covenants to deliver such further assignments and assurances thereof as Landlord may from time to time request, hereby irrevocably designating and appointing Landlord as its attorney-in-fact to execute and deliver in Tenant's name and behalf all such further assignments thereof. However, Tenant shall have the right to make its own claim against the condemning authority for a separate award for the value of any of Tenant's Special Installations and Alterations, for moving and relocation expenses and for such business damages and/or consequential damages as may be allowed by law which do not constitute part of the compensation for the Building and do not diminish the amount of the award to which Landlord would otherwise be entitled.

15. ASSIGNMENT AND SUBLETTING

Tenant shall not mortgage, pledge, encumber, sell, assign, or transfer this Lease, in whole or in part, by operation of law or otherwise, or sublease all or any part of the Leased Premises, without Landlord's written consent, which consent may be withheld for any reason whatsoever except as specifically set forth in this paragraph 15. In all events no such assignment shall be valid unless, prior to the commencement of the subject sub-lease or the occupancy of the sub-tenant Landlord shall have received financial information and documents from the proposed sub-tenant and approved the proposed sublease and Tenant shall have delivered to Landlord (i) a duplicate original instrument of assignment in form reasonably satisfactory to Landlord, duly executed by Tenant, and (ii) an instrument in form attached hereto as Exhibit I, duly executed by the Tenant and the assignee or sub-tenant, in which such assignee or sub-tenant shall agree, among other things, to observe and perform, and to be personally bound by, all of the terms, covenants, and conditions of this Lease on Tenant's part to be observed and performed, whether or not accruing prior to or after the date of such assignment and whether or not relating to matters arising prior to such assignment. In the event of any monetary Defualt hereunder that remains uncured after the passage of any applicable cure period, as set forth herein, Tenant hereby irrevocably assigns to Landlord the right to collect all Rent and additional charges due Tenant as Sublandlord from any subtenant.

(a) \underline{Right} to $\underline{Sublease}$: Tenant shall not have the right to sublease or assign the Leased

Premises in whole or in part to any party without Landlord's prior written approval. If Landlord allows subleasing, in all events Tenant shall remain primarily liable under the lease. All additional rent and other compensation, in excess of the Basic Rent hereunder, provided by the subject sub-tenant under any sub-lease shall be the sole property of Landlord. In all events, any and all security deposit, paid to Tenant by any sub-tenant shall be promptly delivered to an escrow agent, selected by Landlord. Further, all payments due Tenant, as Sub-landlord, under any such Sub-lease shall be paid by joint check, made payable to Landlord and Tenant and if paid over to Landlord shall be credited against the then current amount due Landlord from Tenant.

- (b) <u>Restrictions on Sub-leasing</u>: It is understood and agreed that the overall make up of tenants and the size of sub-leased spaces within the Building is subject to the Landlord's sole and absolute discretion and in all events subject to Landlord's approval. The Landlord reserves the right to deny approval of a sub-lease to any party.
- (c) <u>Prior to Offering</u>: In connection with any request by Tenant for consent to sublet all or any portion of the Leased Premises, Tenant shall, at least ten (10) days prior to offering any space for sub-lease, submit to Landlord, in writing, a notice of Tenant's desire to sub-lease a portion of the Leased premises containing such information as the amount of proposed sub-lease space, the location of the proposed sub-lease space, an asbuilt floor plan of the proposed sub-lease space, the terms to be sought by Tenant under a sub-lease for the proposed sub-lease space, and the date of availability of the proposed sub-lease space.
- (d) <u>Sub-tenant Identification</u>: Upon identifying a proposed sub-lease tenant (a "Proposed Sub-tenant") or a proposed assignee (a "Proposed Assignee") Tenant shall submit to Landlord, in writing, a statement containing the name of the Proposed Assignee or Sub-tenant, such information as to its financial responsibility and standing of the Proposed Assignee or Sub-tenant as Landlord may require, and all of the terms and provisions upon which the proposed assignment or sublease is to be made, and a floor plan delineating the proposed sublet area.
- (e) <u>Sub-lease Profits</u>: In all events, any and all additional rent and other compensation in excess of the Basic Rent provided for herein, as provided for under any sublease or assignment permitted by Landlord, shall be the sole and exclusive property of Landlord without offset or deduction of any kind, including, but not limited to any offset for Tenant's costs of sub-leasing, legal expenses associated with the subject sub-lease, sub-tenant improvements paid for by Tenant, and any other concessions made to the subject sub-tenant (the "Sub-lease costs"). For the purposes of this provision Tenant agrees that all additional rent and other compensation in excess of the Basic Rent provided for herein generated from sub-leasing any of the Leased premises shall belong to, and be the sole property of Landlord. Further, in no event shall Tenant be entitled to any portion of any profits generated by the sale of special services (including additional services provided by Landlord) to any sub-tenant of Tenant. In all events any permitted sub-tenant shall be required to execute a sub-lease agreement that is acceptable to Landlord in Landlord's reasonable discretion and which incorporates all terms and conditions of this Lease.
 - (f) Mortgagee Approval: In all events all proposed subleases of the Leased Premises shall be subject to the approval of any lender of Landlord that holds a mortgage on the Building.
- (g) <u>Right of First Refusal</u>: The Landlord shall at all times have a right of first refusal upon any portion of the Leased Premises that Tenant desires to sublease which right of first refusal may be assigned by Landlord. In the event the Landlord exercises this option the terms of the subject sub-lease shall be those readily available to Tenant from a third party. However, Landlord's right of first refusal shall not apply where Tenant demonstrates that such sublease provides Tenant a reasonable business benefit.
- (h) Invalid Transfers: Any attempted transfer, assignment, sub-leasing, mortgaging or encumbering of this Lease in violation of the provisions of this Section shall be void and confer no rights upon any third person. No permitted assignment or subletting shall relieve Tenant of any of its obligations under this Lease. Landlord and Tenant agree that (i) any consideration paid to Tenant in connection with a sub-leasing of all or any part of the Leased Premises which is attributable to an increase in the rental value of the Leased Premises over and above the Basic Rent and Additional Charges payable under this Lease, and (ii) any consideration paid to Tenant or any sub-tenant or other Person claiming through or under Tenant in connection with an assignment of the Tenant's interest in this Lease or the interest of any sub-tenant or other Person claiming through or under Tenant, or the creditors of Tenant or of any such subtenant or other Person claiming through or under Tenant, or the creditors of Tenant or of any such subtenant or other Person claiming through or under Tenant, and Landlord shall have the

right to condition its consent to any such assignment or subletting upon receipt by Landlord of Tenant's or any subtenant's or other Person's written confirmation of or other evidence of compliance with, the provisions of clause (i) or (ii), as the case may be.

- (i) <u>Transfer of Control</u>: If Tenant is a corporation, any transfer of any of Tenant's issued and outstanding capital stock or any issuance of additional capital stock, as a result of which the majority of the issued and outstanding capital stock of Tenant is held by a Person or Persons who do not hold a majority of the issued and outstanding capital stock of Tenant on the date hereof, shall be deemed an assignment under this Section 15; <u>provided. however.</u> that this sentence shall not apply to a corporation if all of the outstanding voting stock of such corporation is registered under Sections 12(b) or 12(g) of the Securities Exchange Act of 1934, as amended. If Tenant is a partnership, or limited liability company, or any other change in the composition of the partnership, or limited liability company, or any other change in the control of Tenant from the Person or Persons controlling the partnership, or limited liability company, on the date hereof, shall be deemed an assignment under this Section 15. Notwithstanding this provision (15(i)) it is agreed that if such a transfer of stock occurs and within fifteen (15) days of such transfer (if not provided in advance of the transfer) the Landlord (a) receives financial statements and other related information, as required of Tenant hereunder, from the transferee, (b) acknowledges to the Tenant in writing that Landlord finds the financial condition of the transferee of the stock to be reasonably acceptable, and (c) the transferee of the stock affirms this Lease in writing, in a form acceptable to Landlord, then in such event such a transfer of control shall not be considered a Event of Default hereunder.
- (j) Obligations of Assignee or Subtenant: If Tenant's interest in this Lease is assigned in whole or in part, whether or not in violation of the provisions of this Section, Landlord, at its sole option, may collect all rent and other amounts due Tenant from the assignee, whether Tenant is in Default hereunder or not. If the Leased Premises or any part thereof are sub-leased to, or occupied by, or used by, any Person other than Tenant, whether or not in violation of this Section, Landlord, at its sole option, may collect all rent and other amounts due Tenant from the sub-tenant ("Sub-tenant Payments"), whether Tenant is in Default hereunder or not. In either case, Landlord shall provide written notice to the subject sub-tenant or occupant of the Leased Premises, at the address of the Leased Premises, and to Tenant under this Lease informing the subject parties of Landlord's election of its option to receive all Sub-tenant Payments and thereafter all Sub-tenant Payments due Tenant shall be paid directly to Landlord. In either case. Landlord shall apply the amount collected to the rents reserved in this Lease, but neither any such assignment, sub-leasing, occupancy, or use, whether with or without Landlord's prior consent, nor any such collection or application, shall be deemed a waiver of any term, covenant or condition of this Lease or the acceptance by Landlord of such assignment or user as tenant. The consent by Landlord to any assignment or sub-leasing shall not relieve Tenant from its obligation to obtain the express prior written consent of Landlord to any assignment or sub-leasing. The listing of any name other than that of Tenant on any door of the Leased Premises or on any directory in the Building, or otherwise, shall not operate to vest in the Person so named any right or interest in this Lease or in the Leased Premises or be deemed to constitute, or serve as a substitute for, any prior consent of Landlord required under this Section, and it is understood that any such listing shall constitute a privil

16. <u>DEFAULT PROVISIONS</u>

- (a) Each of the following events shall be deemed to be, and is referred to in this Lease as, an "Event of Default":
- (1) A default by Tenant in the due and punctual payment of (i) all Basic Rent due, by wire transfer, on the first business day of each calendar month, (ii) all Additional Charges (including but not limited to Tenant's Additional Costs) as and when they become due as set forth herein, (iii) any amounts that become due and the entire balance due under the Promissory Note, representing the Security Deposit as set forth in Paragraph 19 hereof as and when such amounts become due, if applicable,; or

- (2) The neglect or failure of Tenant to perform or observe any of the terms, covenants, or conditions contained in this Lease on Tenant's part to be performed or observed (other than those referred to in paragraph (1) above for which Tenant does not have the right to a cure period) which if not remedied by Tenant within fifteen (15) business days after Landlord shall have given to Tenant written notice specifying such neglect or failure; if such condition can not practically be remedied within said fifteen (15) day period Tenant shall have forty five days from the date of such notice to remedy the condition provided Tenant timely commences and diligently prosecutes such remedy unless the nature of such condition requires it to be remedied in a shorter period of time; or
 - (3) The assignment, transfer, mortgaging, or encumbering of this Lease or the sub-leasing of any or all of the Leased Premises in a manner not strictly in accordance with and permitted by Section 15; or
- (4) The taking of this Lease or the Leased Premises, or any part thereof, upon execution or by other process of law directed against Tenant, or upon or subject to any attachment at the instance of any creditor of or claimant against Tenant, which execution or attachment shall not be discharged or disposed of within thirty (30) days after the levy thereof; or
- (5) The abandonment of the Leased Premises, in whole or in part, by Tenant, provided that no abandonment shall be considered to have occurred so long as Tenant continues to pay all amounts due Landlord, as and when due, and continues to meet all other terms, conditions, and obligations of Tenant under this Lease, as reasonably determined by Landlord.
 - (6) The failure of any sub-tenant occupying any portion of the Leased Premises to comply with each and every provision of this Lease.
 - (b) Upon the occurrence of an Event of Default, Landlord shall have the right, at its election, then or at any time thereafter while such Event of Default shall continue, either:
- (1) To give Tenant written notice that this Lease will terminate on a date to be specified in such notice, which date shall not be less than ten (10) business days after such notice, and on the date of such notice Tenant's right to possession of the Leased Premises shall cease and this Lease shall thereupon be terminated, but Tenant shall remain liable as provided in subsection (c);or
- (2) Without demand or notice, to re-enter and take possession of the Leased Premises, or any part thereof, and repossess the same as of Landlord's former estate and expel Tenant and those claiming through or under Tenant a right to occupy the Leased Premises, and remove the effects of both or either, either by summary proceedings, or by action at law or in equity or by force (if necessary) or otherwise, without being deemed guilty of any manner of trespass and without prejudice to any remedies for arrears of rent or preceding breach of covenant.
- If Landlord elects to re-enter the Leased Premises as set forth above, Landlord may terminate this Lease, or, from time to time, without terminating this Lease, may re-lease the Leased Premises, or any part thereof, as agent for Tenant for such term or terms and conditions as Landlord may deem advisable, with the right to make alterations and repairs to the Leased Premises. No such re-entry or taking of possession of the Leased Premises by Landlord shall be construed as an election on Landlord's part to terminate this Lease unless a written notice of such intention is given to Tenant as set forth above or unless the termination thereof be decreed by a court of competent jurisdiction. Tenant waives any right to the service of any notice of Landlord's intention to re-enter provided for by any present or future law.
- (c) If Landlord terminates this Lease pursuant to subsection (b), Landlord shall have the option to accelerate and declare the entire amount of all Basic Rent and Additional Charges (including but not limited to Tenant's Additional Costs) provided for herein until the date this Lease would have expired had such termination not occurred as the total rental set forth in Section (a) (1) of this Paragraph as due and payable forthwith. Tenant shall be liable (in addition to accrued liabilities) to the extent legally permissible for (i) the sum of (A) all Basic Rent and Additional Charges (including but not limited to Tenant's Additional Costs) provided for in this Lease until the date this Lease would have expired had such termination not occurred, and (B) any and all reasonable expenses incurred by Landlord in re-entering the Leased Premises, repossessing the same, making good any default of Tenant, painting the same, adjoining the same with any adjacent space for any new tenants, putting the same in proper repair, re-letting the same (including any and all reasonable attorneys' fees and disbursements and reasonable brokerage fees incurred with so doing), and any and all expenses which Landlord may incur during the occupancy of any new tenant (other than expenses

of a type that are Landlord's responsibility under the terms of this Lease); less (ii) the net proceeds of any re-letting.

In addition to the foregoing, Tenant shall pay to Landlord such sums as the court which has jurisdiction thereover may adjudge reasonable as attorney's fees with respect to any successful law suit or action instituted by Landlord to enforce the provisions of this Lease. Landlord shall have the right, at its sole option, to release the whole or any part of the Leased Premises for the whole of the un-expired Term, or longer, or from time to time for shorter periods, for any rental then obtainable, giving such concessions of rent and making such special repairs, alterations, decorations, and paintings for any new tenant as Landlord, in its sole and absolute discretion, may deem advisable. Tenant's liability as aforesaid shall survive the institution of summary proceedings and the issuance of any warrant thereunder. Landlord shall be under no obligation to re-lease the Leased Premises, but agrees to use its best efforts to do so.

17. BANKRUPTCY TERMINATION PROVISION

At the sole and exclusive option of Landlord, evidenced by written notice from Landlord to Tenant, the Landlord may, without relieving Tenant from any of its obligations that survive termination, terminate this Lease, without the performance of any additional act or the giving of any additional notice to any other party, effective immediately upon the occurrence of any of the following events, even if the effective date of termination precedes the date of Landlord's notice, or on such later date as determined by Landlord: (1) Tenant's admitting in writing its inability to pay its debts generally as they become due, or (2) the commencement by Tenant of a voluntary case under the federal bankruptcy, insolvency, or other similar law, or (3) the entry of a decree or order for relief by a court having jurisdiction in the premises in respect of Tenant in an involuntary case under the federal bankruptcy laws, as now constituted or hereafter amended, or any other applicable federal or state bankruptcy, insolvency or other similar law, and the continuance of any such decree or order unstayed and in effect for a period of thirty (30) consecutive days, or (4) Tenant's making an assignment of all or a substantial part of its property for the benefit of its creditors in satisfaction of a pre-existing debt or obligation, or (5) Tenant's seeking or consenting to or acquiescing in the appointment of, or the taking of possession by, a receiver, trustee or custodian for all or a substantial part of its property, or (6) the entry of a court order without Tenant's consent, which order shall not be vacated, set aside or stayed within thirty (30) days from the date of entry, appointing a receiver, trustee, or custodian for all or a substantial part of Tenant's property. The provisions of this Section shall be construed with due recognition for the provisions of the federal bankruptcy laws, where applicable, but shall be interpreted in a manner which results in a termination of this Lease, at the option of Landlord, in each and every

18. LANDLORD MAY PERFORM TENANT'S OBLIGATIONS

If Tenant shall fail to keep or perform, any of its obligations as provided in this Lease in respect to (a) maintenance of insurance, (b) repairs and maintenance of Leased Premises, (c) compliance with Legal Requirements, or (d) the making of any other payment or performance of any other obligation (other then the payment of amounts due Landlord hereunder), then Landlord may (but shall not be obligated to) upon the continuance of such failure on Tenant's part for ten (10) days after written notice to Tenant (or after such additional period, if any, as Tenant may reasonably require to cure such failure if of a nature which cannot be cured within said 10-day period but not to exceed fifteen additional days), or without notice in the case of an emergency, and without waiving or releasing Tenant from any obligation, and as an additional but not exclusive remedy, make any such payment or perform any such obligation and all sums so paid by Landlord and all necessary incidental costs and expenses, including attorney's fees, incurred by Landlord in making such payment or performing such obligation, together with interest thereon from the date of payment at the Default Interest Rate, shall be deemed Additional Rent and shall be paid to the Landlord on demand, or at Landlord's option may be added to any installment of Basic Rent thereafter falling due, and if not so paid by Tenant, Landlord shall have the same rights and remedies as in the case of a default by Tenant in the payment of Basic Rent.

19. SECURITY DEPOSIT

(a) Upon execution of this Lease, Tenant shall pay Landlord a Security Deposit in the amount of Seven Hundred and Fifty Thousand and 00/100 Dollars (\$750,000.00), in the form of the

promissory note attached hereto as Exhibit G and incorporated herein (the "Security Deposit Promissory Note"). The Security Deposit Promissory Note shall provide for the total amount due thereunder to be due and payable not later then five (5) days after any Default by Tenant that is not cured within the applicable cure period, if any applies. Provided the Tenant has demonstrated financial viability, as reasonably determined by Landlord or Landlord's Mortgagee(s), on the fourth anniversary of the Lease Commencement, the principal amount of this Note shall be reduced from Seven Hundred and Fifty Thousand Dollars (\$750,000.00) to an amount equal to the Basic Rent for the last Lease Year of the Lease Term.

- (b) As consideration for Landlord entering into this Lease Agreement the Tenant hereby expressly waives and relinquishes any and all right Tenant may have (i) to any portion of the security deposit paid to Landlord pursuant to the Original Lease, (ii) to earn interest on any cash Security Deposit delivered to Landlord in connection with the Security Deposit Promissory Note hereunder.
- (c) Tenant hereby deposits with Landlord the Security Deposit, as security for the prompt, full, and faithful performance by Tenant of each and every provision of this Lease and of all obligations of Tenant hereunder. If an Event of Default occurs, Landlord may use, make demand for payment of the full amount of the Security Deposit Promissory Note and upon receipt thereof, apply, or retain the whole or any part of the Security Deposit for the payment of (i) any Basic Rent or Additional Charges which Tenant may not have paid or which may become due after the occurrence of such Event or Default, (ii) any sum expended by Landlord on Tenant's behalf in accordance with the provisions of this Lease (including the reimbursement of the un-amortized Tenant Improvement Allowance provided by Landlord), or (iii) any sum which Landlord may expend or be required to expend by reason of Tenant's default, including damages or deficiency in the re-leasing of the Leased Premises as provided in Section 16. The use, application, or retention of the Security Deposit, or any portion thereof, by Landlord shall not prevent Landlord from exercising any other right or remedy provided by this Lease or by law and shall not operate as a limitation on any recovery to which Landlord may otherwise be entitled. If any portion of the Security Deposit is used, applied or retained by Landlord for the purpose set forth above, Tenant agrees, within ten (10) days after a written demand therefore is made by Landlord, to deposit cash with the Landlord in an amount sufficient to restore the Security Deposit to its original amount.
- (d) If Tenant shall fully and faithfully comply with all of the provisions of this Lease, the Security Deposit, or any balance thereof, shall be returned to Tenant within thirty (30) days after the expiration of the Term, without interest. In the absence of evidence satisfactory to Landlord of any permitted assignment of the right to receive the Security Deposit, or the remaining balance thereof, Landlord may return the same to the original Tenant, regardless of one or more assignments of Tenant's interest in this Lease or the Security Deposit. In such event, upon the return of the Security Deposit (or balance thereof) to the original Tenant, Landlord shall be completely relieved of liability under this Section.
- (e) In the event of a transfer of Landlord's interest in the Leased Premises, Landlord shall have the right to transfer the Security Deposit to the transferee thereof. In such event, upon the delivery by Landlord to Tenant of such transferee's written acknowledgement of its receipt of such Security Deposit, Landlord shall be deemed to have been released by Tenant from all liability or obligation for the return of such Security Deposit, and Tenant agrees to look solely to such transferee for the return of the Security Deposit.
- (f) The Security Deposit shall not be mortgaged, assigned, or encumbered in any manner whatsoever by Tenant without the prior written consent of Landlord, which may be withheld by Landlord in its sole discretion.

(g) THE SECURITY DEPOSIT PROMISSORY NOTE SHALL CONTAIN A CONFESSION OF JUDGMENT PROVISION WHICH CONSTITUTES A WAIVER OF IMPORTANT RIGHTS TENANT MAY HAVE AS A DEBTOR AND ALLOWS THE CREDITOR TO OBTAIN A JUDGMENT AGAINST TENANT WITHOUT ANY FURTHER NOTICE.

(h) Should any sums become due and payable under the Security Deposit Promissory Note and such sums are not paid when and as due, time being of the essence, the Borrower hereby constitutes and appoints Marc Bettius and/or Christopher D. Clemente, either of whom may act, as its attorney-in-fact to confess judgment on the Borrower under the Security Deposit Promissory Note for the full sum due thereunder, plus attorney's fees of 20% of the total amount then outstanding under

the Security Deposit Promissory Note, and upon entry of the judgment, the Borrower under the Security Deposit Promissory Note waives the benefit of any and every statute, ordinance, or rule of court which may lawfully waived conferring upon the Borrower under the Security Deposit Promissory Note any right or privilege or exemption, stay of execution or supplemented proceedings, or other relief from the enforcement or immediate enforcement of a judgment or related proceedings on a judgment. The Borrower under the Security Deposit Promissory Note acknowledges that said sum is reasonable as evidenced by Borrowers signature on the Security Deposit Promissory Note consents to venue in the Circuit Court of Fairfax County with respect to the institution of an action confessing judgment hereon. The authority and power to appear for and enter judgment against the Borrower under the Security Deposit Promissory Note shall not be exhausted by one or more exercises thereof, or yany imperfect exercise thereof, and shall not be extinguished by any judgment entered pursuant thereto, such authority and power may be exercised on one or more occasions from time to time in the same or different jurisdictions as often as the Lender under the Security Deposit Promissory Note are paid in full.

20. SUBORDINATION

(a) This Lease and Tenant's interest hereunder shall have priority over, and be senior to, the lien of any Mortgage made by Landlord after the date of this Lease. However, if at any time or from time to time during the Term, a Mortgagee or prospective Mortgage requests that this Lease be subject and subordinate to its Mortgage, and if Landlord consents to such subordination, this Lease and Tenant's interest hereunder shall be subject and subordinate to the lien of such Mortgage and to all renewals, modifications, replacements, consolidations, and extensions thereof and to any and all advances made thereunder and the interest thereon. Tenant agrees that, within ten (10) days after receipt of a written request therefor from Landlord, it will, from time to time, execute and deliver any instrument or other document required by any such Mortgagee to subordinate this Lease and its interest in the Leased Premises to the lien of such Mortgage. If, at any time or from time to time during the Term, a Mortgage and aprior to the date of this Lease shall request that this Lease have priority over the lien of such Mortgage, and if Landlord consents thereto, this Lease shall have priority over the lien of such Mortgage and all renewals, modifications, replacements, consolidations, and extensions thereof and all advances made thereunder and the interest thereon, and Tenant shall, within ten (10) days after receipt of a written request therefor from Landlord, execute, acknowledge and deliver any and all documents and instruments confirming the priority of this Lease. In addition, the Mortgage which has priority over this Lease shall have the right, at its option, to subordinate the lien of its Mortgage, in whole or in part, to this Lease by recording a unilateral declaration to that effect among the applicable Land Records. In any event, however, if this Lease shall have priority over the lien of a first Mortgage, without the written consent of the first Mortgagee.

(b) This Lease and Tenant's interest hereunder shall be subject and subordinate to each and every ground or underlying lease hereafter made of the Building or the land on which it is constructed, or both, and to all renewals, modifications, replacements, and extensions thereof. Tenant agrees that, within ten (10) days after receipt of written request therefor from Landlord, it will, from time to time, execute, acknowledge and deliver any instrument or other document required by any such lessor to subordinate this Lease and its interest in the Leased Premises to such ground or underlying lease.

(c) If (i) the Building, or any part thereof, or the land on which the Building is constructed, or the Landlord's leasehold estate in the Building, is at any time subject to a first Mortgage, and Landlord has entered into an assignment of this Lease to the holder of said first Mortgage, and (ii) the Tenant is given written notice of such assignment, including the name and address of the assignee, then, in that event, Tenant shall not terminate this Lease or make any abatement in the Basic Rent payable hereunder for any default on the part of the Landlord without first giving written notice, in the manner provided elsewhere in this Lease for the giving of notice, to such first Mortgagee, specifying the default in reasonable detail, and affording such first Mortgagee a reasonable opportunity to make performance, at its election, for and on behalf of the Landlord.

21 ATTORNMENT

In the event of (a) a transfer of Landlord's interest in the Leased Premises, (b) the

termination of any ground or underlying lease of the Building or the land on which it is constructed, or both, or (c) the purchase of the Building or Landlord's interest therein in a foreclosure sale or by deed in lieu of foreclosure under any Mortgage or pursuant to a power of sale contained in any Mortgage, then in any of such events Tenant shall, upon demand by the owner of the Building or the land on which it is constructed, or both, attorn to and recognize the transferee or purchaser of Landlord's interest or the lessor under the terminated ground or underlying lease, as the case may be, as Landlord under this Lease for the balance then remaining of the Term, and thereafter this Lease shall continue as a direct lease between such person, as "Landlord," and Tenant, as "Tenant," except that such lessor, transferee or purchaser shall not be liable for any act or omission of Landlord prior to such lease termination or prior to such person's succession to title, nor be subject to any offset, defense or counterclaim accruing price to such lease termination or prior to such person's succession to title for more than one (1) month in advance. Tenant shall, upon request by Landlord or the transferee or purchaser of Landlord's interest or the lessor under the terminated ground or underlying lease, as the case may be, execute and deliver an instruments confirming the foregoing provisions of this Section. Tenant hereby waives the provisions of any present or future law or regulation which gives or purports to give Tenant any right to terminate or otherwise adversely affect this Lease, or the obligations of Tenant hereunder, upon or as a result of the termination of any such ground or underlying lease or the completion of any such foreclosure and sale.

22. OUIET ENJOYMENT

Landlord covenants that Tenant, upon paying the Basic Rent and the Additional Charges provided for in this Lease, and upon performing and observing all of the terms, covenants, conditions, and provisions of this Lease on Tenant's part to be kept, observed and performed, shall quietly hold, occupy, and enjoy the Leased Premises during the Term without hindrance, ejection, or molestation by Landlord or any party lawfully claiming through or under Landlord.

23. LANDLORD'S RIGHT OF ACCESS TO LEASED PREMISES

- (a) Landlord and its agents shall have the following rights in and about the Leased Premises: (i) to enter the Leased Premises at all reasonable times and with reasonable notice to examine the Leased Premises or for any of the purposes set forth in this section or for the purpose of performing any obligation of Landlord under this Lease or exercising any right or remedy reserved to Landlord in this Lease, and if Tenant, its officers, partners, agents, or employees shall not be personally present or shall not open and permit an entry into the Leased Premises at any time when such entry shall be necessary or permissible, to use a master key or forcibly to enter the Leased Premises; (ii) to erect, install, use, and maintain pipes, ducts, and conduits in and through the Leased Premises which, when completed, will not substantially interfere with the use or appearance or materially reduce the space afforded to Tenant in the Leased Premises; (iii) to exhibit the Leased Premises to others at reasonable times and for reasonable purposes; (iv) to make such decorations, repairs, alterations, improvements, or additions, or to perform such maintenance, including, but not limited to, the maintenance of all heating, air-conditioning, elevator, plumbing, electrical and other mechanical facilities installed by Landlord, as Landlord may deem necessary or desirable; and (v) to take all materials into an upon the Leased Premises that may be required in connection with any such decorations, repairs, alterations, improvements, additions, or maintenance. Landlord agrees to give prior notice before it exercises its rights under this subsection, except that Landlord may enter the Leased Premises without notice in the case of an emergency. In making such an entry, Landlord agrees to use reasonable efforts to avoid interfering with the regular and usual conduct of the Tenant's business.
- (b) All parts (except surfaces facing the interior of the Leased Premises) of all walls, windows, and doors bounding the Leased Premises (including exterior Building walls, corridor walls, doors, and entrances), all balconies, terraces, and roofs adjacent to the Leased Premises, all space in or adjacent to the Leased Premises used for shafts, stacks, stairways, chutes, pipes, conduits, ducts, fan rooms, heating, and air-conditioning, plumbing, electrical, and other mechanical facilities installed by Landlord, service closets and other Building facilities, and the use thereof, as well as access thereto through the Leased Premises for the purposes of operation, maintenance, alteration, and repair, are hereby reserved to Landlord. Nothing contained in this Section shall impose any obligation upon Landlord with respect to the operation, maintenance, alteration, or repair of the Leased Premises or the Building.
 - (c) The exercise by Landlord or its agents of any right reserved to Landlord in this Section shall not constitute an actual or constructive eviction, in whole or in part, or entitle Tenant to

any abatement or diminution of rent, or relieve Tenant from any of its obligations under this Lease, or impose any liability upon Landlord, or its agents, or upon any lessor under any ground or underlying lease, by reason of inconvenience or annoyance to Tenant, or injury to or interruption of Tenant's business, or otherwise. Landlord agrees to exercise its rights under this Section in a manner designed to minimize interference with Tenant's normal business operations, without any obligation, however, to employ labor at overtime or other premium pay rates.

24. LIMITATION ON LANDLORD'S LIABILITY

(a) Except for damages resulting from the willful or negligent act or omission of Landlord, its agents and employees, Landlord shall not be liable to Tenant, its employees, agents, business invitees, licensees, customers, guests or trespassers, for any damage or loss to the property of Tenant or others located on the Leased Premises, or in the Building or the land on which it is built, or for any accident or injury to Persons in the Leased Premises or the Building, resulting from the necessity of repairing any portion of the Building; the use or operation (by Tenant or any other Person or Persons whatsoever) of any elevators, or heating, cooling, electrical or plumbing equipment or apparatus; the termination of this Lease by reason of the destruction of the Building or the Leased Premises; any fire, robbery, theft, and/or any other casualty; any leaking in any part of the Leased Premises; any water, gas, steam, fire, explosion, electricity or falling plaster; the bursting, stoppage or leakage of any pipes, sewer pipes, drains, conduits, appliances or plumbing works; or any other cause whatsoever.

(b) Landlord shall not be required to perform any of its obligations hereunder, nor be liable for loss or damage for failure to do so, nor shall Tenant be released from any of its obligations under this Lease because of the Landlord's failure to perform, where such failure arises from or through Unavoidable Delays or Legal Requirements. If Landlord is so delayed or prevented from performing any of its obligations during the Term, the period of such delay or such prevention shall be deemed added to the time herein provided for the performance of any such obligation.

25. ESTOPPEL CERTIFICATES

Tenant agrees from time to time, within ten (10) days after written request therefor by Landlord, to execute, acknowledge and deliver to Landlord a statement in writing certifying to Landlord, any Mortgagee, assignee of a Mortgagee, or any purchaser of the Building or the land on which it is constructed, or both, or any other Person designated by Landlord, as of the date of such statement, (i) that Tenant is in possession of the Leased Premises; (ii) that this Lease is unmodified and in full force and effect (or, if there have been modifications, that this Lease is in full force and effect as modified and setting forth such modifications); (iii) whether or not there are then existing any set-offs or defenses known to Tenant against the enforcement of any right or remedy of Landlord, or any duty or obligation of Tenant, hereunder (and, if so, specifying the same in detail); (iv) the dates, if any, to which any Basic Rent or Additional Charges have been paid in advance; (v) that Tenant has no knowledge of any uncured defaults on the part of Landlord under this Lease (or, if Tenant has such knowledge, specifying the same in detail); (vii) the amount of any Security Deposit held by Landlord; and (viii) any additional facts reasonably requested by any such Mortgagee, assignee or a Mortgagee or purchaser.

26. SURRENDER OF LEASED PREMISES

- (a) Tenant shall, on or before the last day of the Term, except as otherwise expressly provided elsewhere in this Lease, remove all of its property and peaceably and quietly leave, surrender and yield up to the Landlord the Leased Premises, free of sub-tenancies, broom clean and in good order and condition except for reasonable wear and tear, damage by fire or other casualty, or conditions requiring repair by Landlord hereunder at Landlords expense.
 - (b) The provisions of this Section shall survive any expiration or termination of this Lease.

27. HOLDING OVER

If Tenant shall hold over possession of the Leased Premises after the end of the Term, Tenant shall be deemed to be occupying the Leased Premises as a Tenant from month to month, at 150% of the Basic Rent, adjusted to a monthly basis, and subject to all the other conditions, provisions and obligations of this Lease insofar as the same are applicable, or as the same shall be adjusted, to a month-to-month tenancy. Notwithstanding the immediately preceding sentence the Tenant shall have the right to hold over for a period of up to two (2) months following the expiration

of the Lease Term, or any extension thereof, at 150% of the Base Rent in effect during the last month of the previous Lease Term with six (6) months written notice. Thereafter the holdover rent will be at 175% of the Basic Rent in effect during the last month of the previous Lease Term plus consequential damages.

28 PARKING

Tenant will have the right to utilize it's pro-rata share of the parking spaces in the project's parking structure and in the surface parking lots at no cost during the initial Lease Term and any extensions, including up to 10 reserved spaces in a location that includes the ten closest spaces to the entry door into the Building from the covered garage that were assigned to Tenant pursuant to the Original Lease. Throughout the Term, Tenant and/or its employees shall have the right, without additional cost, to park their automobiles in the surface and garage parking areas provided for the Building. Such parking spaces shall be available on a first-come, first-served basis (except as otherwise provided above), subject, however, to the rights of any other tenant of the Building to park automobiles in reserved parking spaces as provided in its lease. Landlord reserves the right, at any time or from time to time during the Term, to control access to the surface and garage parking areas, by use of mechanical or electric devices or otherwise, to tenants of the Building and their employees, provided that Landlord shall reserve at least twelve (12) parking spaces for parking for visitors of the Building. If, at any time during the Term, Landlord implements a controlled access system for the Building parking areas, Tenant shall have the right without additional cost, to use unassigned parking spaces in the structured parking and the surface parking lots on a pro-rata basis based on the square footage of the Leased Premises. Neither Tenant nor any of its employees shall use any of the parking facilities for storage of vehicles (or any other item such as boats or trailers) or park its or their automobiles in any portion of the Building parking areas reserved for visitor or handicapped parking or for parking of automobiles belonging to other tenants of the Building. Tenant shall cause its employees to abide by any parking reservation system implemented from time to time by Landlord. Any failure to abide by the parking restri

29. LEASING COMMISSION

Landlord and Tenant each represent and warrant to the other that neither of them has employed any broker in carrying on the negotiations relative to this Lease. Landlord and Tenant shall each indemnify and hold harmless the other from and against any claims for brokerage or other commission arising from or out of any breach of the foregoing representation and warranty.

30. GENERAL PROVISIONS

- (a) The covenants, conditions, agreements, terms and provisions herein contained shall be binding upon, and shall inure to the benefit of, the parties hereto and, subject to the provisions of Section 15, each of their respective personal representatives, successors and assigns.
 - (b) It is the intention of the parties hereto that this Lease (and the terms and provisions hereof) shall be construed and enforced in accordance with the laws of the Commonwealth of Virginia.
- (c) No failure by Landlord to insist upon the strict performance of any term, covenant, agreement, provision, condition or limitation of this Lease or to exercise any right or remedy consequent upon a breach thereof, and no acceptance by the Landlord of full or partial rent during the continuance of any such breach, shall constitute a waiver of any such breach or of any such term, covenant, agreement, provision, condition, limitation, right or remedy. No term, covenant, agreement, provision, condition or limitation of this Lease to be kept, observed or performed by Landlord or by Tenant, and no breach thereof, shall be waived, altered or modified except by a written instrument executed by Landlord or by Tenant, as the case may be. No waiver of any breach shall affect or alter this Lease, but each and every term, covenant, agreement, provision, condition and limitation of this Lease shall continue in full force and effect with respect to any other then existing or subsequent breach thereof.
- (d) No notice, request, consent, approval, waiver or other communication which may be or is required or permitted to be given under this Lease shall be effective unless the same is in writing and is delivered in person or sent by registered or certified mail, return receipt requested,

first-class postage prepaid, (1) if to Landlord, at Landlord's Notice Address, or (2) if to Tenant, at Tenant's Notice Address, or at any other address that may be given by one party to the other by notice pursuant to this subsection. Such notices, if sent by registered or certified mail, shall be deemed to have been given at the time of mailing.

- (e) It is understood and agreed by and between the parties hereto that this Lease contains the final and entire agreement between said parties, and that they shall not be bound by any terms, statements, conditions or representations, oral or written, express or implied, not herein contained. It is understood and agreed, however, that the terms hereof shall be modified, if so required, for the purpose of complying with or fulfilling the requirements of any Mortgagee secured by a first Mortgage that may now be or hereafter become a lien on the Building, provided, however, that such modification shall not be in substantial derogation or diminution of any of the rights of the parties hereunder, nor increase any of the obligations or liabilities of the parties hereunder.
- (f) Tenant hereby waives all right to trial by jury in any claim, action, proceeding or counterclaim by Landlord on any matters arising out of or in any way connected with this Lease, the relationship of Landlord and Tenant and/or Tenant's use or occupancy of the Leased Premises. Tenant also agrees to waive any and all counterclaims Tenant may have in any suit for possession by Landlord; it being understood that the subject of any such counterclaim may be asserted by Tenant but only in a separate action brought by Tenant against Landlord.
- (g) Tenant hereby waives any objection to the venue of any action filed by Landlord against Tenant in any state or federal court in the jurisdiction in which the Building is located, and Tenant further waives any right, claim or power, under the doctrine of forum non conveniens or otherwise, to transfer any such action filed by Landlord to any other court.
- (h) In the event Tenant institutes an action at law or in equity against Landlord under the terms of this Lease and does not prevail, or the case is non-suited by Tenant, then in either event Tenant shall be liable for Landlord's attorney's fees and other expenses of litigation.
- (i) If Tenant is a corporation, within two (2) business days of the signing of this Lease, it shall furnish to Landlord certified copies of the resolutions of its Board of Directors (or of the executive committee of its Board of Directors) authorizing Tenant to enter into this Lease; and it shall furnish to Landlord evidence (reasonably satisfactory to Landlord and its counsel) that Tenant is a duly organized corporation in good standing under the laws of the jurisdiction of its incorporation, is qualified to do business in good standing in the Commonwealth of Virginia, has the power and authority to enter into this Lease, and that all corporate action requisite to authorize Tenant to enter into this Lease has been duly taken.
 - (j) Time is of the essence in the performance of all Tenant's obligations under this Lease.
 - (k) Wherever appropriate herein, the singular includes the plural and the plural includes the singular
 - (I) If any provision of this Lease shall be invalid, illegal or unenforceable, the validity, legality and enforceability of the remaining provisions shall not be affected thereby.
 - (m) The captions in this Lease are for convenience only and shall not affect the interpretation of the provisions hereof.
 - (n) This Lease is not intended to create a partnership or joint venture between Landlord and Tenant in the conduct of their respective business.
- (o) Notwithstanding any provision to the contrary, Tenant shall look solely to the estate and property of Landlord in and to the Building in the event of any claim against Landlord arising out of or in connection with this Lease, the relationship of Landlord and Tenant or Tenant's use of the Leased Premises, and Tenant agrees that the liability of Landlord arising out of or in connection with this Lease, the relationship of Landlord and Tenant or Tenant's use of the Leased Premises, shall be limited to such estate and property of Landlord. No other properties or assets of Landlord shall be subject to levy, execution or other enforcement procedures for the satisfaction of any judgment (or other judicial process) or for the satisfaction of any other remedy of Tenant arising out of or in connection with this Lease, the relationship of Landlord and Tenant or Tenant's use of the Leased Premises, and if Tenant shall acquire a lien on or interest in any other properties or assets by judgment or otherwise, Tenant shall promptly release such lien on or interest in such other properties

and assets by executing, acknowledging and delivering to Landlord an instrument to that effect prepared by Landlord's attorneys.

- (a) This Lease may be executed in several counterparts, but all counterparts shall constitute one and the same instrument.
- (b) Tenant shall use best efforts to deliver to Landlord prior to the Lease Commencement Date, in a form and content reasonably satisfactory to Landlord, the following: (i) an irrevocable and unconditional assignment of each of the subleases (the "Assignment of Subleases"), entered into by Tenant as Sub-landlord with (a) Iona Technologies, Inc, (b) I-Connect, LC, and (c) Opennet, Inc. as subtenants (individually or collectively the "Comscore Subtenants"), (ii) a written statement from each of the Comscore Subtenants acknowledging the assignment of their subject sublease and the attorning of the rent and sublease to Landlord, affirming the validity of the subject sublease, and confirming all amounts due from Sub landlord to Subtenant Acknowledgement and Affirmation of Sublease").

IN WITNESS WHEREOF, Landlord and Tenant have caused this Lease to be signed by their duly authorized partners or officers as set forth below:

seen and agreed this <u>24</u> day of <u>June,</u> 2005	
ComScore Networks, Inc.	
By: /s/ Sheri L. Huston	Witness: /s/ Sarah A. Schar
Print Name: /s/ Sheri L. Huston	
Title: /s/ CFO	Name: /s/ Sarah A. Schar
Seen and agreed this <u>24</u> day of <u>June</u> , 2003	
Comstock Partners, LC	
By: /s/ Christopher Clemente	Witness: /s/ Sarah A. Schar
Print Name: Christopher Clemente	
Citle: Managing Member	Name: /s/ Sarah A. Schar
STATE OF <u>Virginia</u> ,	
COUNTY OF <u>Fairfax</u> , to-wit:	
the undersigned, a Notary Public in and or the County and aforesaid, do hereby certify that <u>Sheri L</u> certification, has personally appeared before me in my County and State aforesaid and acknowledged	<u>. Huston</u> , and his official capacity as <u>CFO</u> of Comscore Networks, Inc., whose name is signed to the foregoing d the same.
GIVEN under my hand and seal this $\underline{24}$ day of $\underline{\text{June}}$, 2003.	
	/s/ Sarah A. Schar
	NOTARY PUBLIC
My commission expires: <u>10-31-05</u> .	
STATE OF <u>Virginia</u> ,	
COUNTY OF Fairfax, to-wit:	
the undersigned, a Notary Public in and or the County and aforesaid, do hereby certify that <u>Christo</u> oregoing certification, has personally appeared before me in my County and State aforesaid and ack	<u>pher Clemente</u> , and his official capacity as <u>Managing Member</u> of Comstock Partners, LC whose name is signed to knowledged the same.
GIVEN under my hand and seal this <u>24</u> day of <u>June</u> , 2003.	
	/s/ Sarah A. Schar
	NOTARY PUBLIC

My commission expires: 10-31-05.

11465 Sunset Hills

Base Building Definition

Project Overview

11465 Sunset Hills will be a six story Class A office building with approximately 89,221.2 rentable square feet of office space, with two levels of structured parking adjacent. The lower level of the parking structure will provide covered parking with direct access into the building. A typical office floor contains a compact centralized core; a 35-foot core to exterior wall dimension; and an 8-foot 10-inch finished ceiling height.

The building features a traditional brick and architectural block exterior with glass curtain wall marking both front and rear entrances. Access to 11465 Sunset Hills is provided through main lobby entrances on two separate levels, both lobbies being finished with patterned marble flooring and wall accents. The overall detailing and design of this new development establishes a new measure for buildings along this portion of the Dulles Toll Road corridor.

Delivery of 11465 Sunset Hills was February 2001.

Base Building Definition

Address

 11465 Sunset Hills Road Reston, Virginia

Structure

- Reinforced Concrete Superstructure
- 80 lbs. + 20 lbs. per square foot live load.
- 33-foot column free space core to perimeter span with 20-foot perimeter spacing.
- 12'-8" slab to slab typical floor with 13'-4" slab to slab from level 2 to level 3.

. F--

Roof

Four ply built up roofing system consisting of four (4) layers of Type V.I asphalt felt set in hot asphalt and surfaced with washed pea gravel in a flood coat of asphalt. The roof system carries a 15-year warranty.

Building Skin

Exterior highlights begin with corner bays of the facade expressed with blue-tinted butt-glazed ribbon windows and alternating horizontal bands of ground face and rock face limestone-colored architectural block. The main facade features red brick veneer with architectural block trim accents above and below punch windows. Front and rear entry bays are distinguished by vertically glazed curtain walls. The public face of the building features well-lit walkways and drop-offs with decorative pavers and landscaping. These features are provided to engage the passing pedestrian or vehicle with the intention of easing any possibility of traffic congestion.

Elevators

Two 3,000 lb. 350-foot/per minute speed travel time and floor-by-floor lock-off capability.

Communications

- · Fiber Optic Telecommunications is available in the building. Specific tenant requirements will be accommodated during tenant improvement construction
- There are two (2) separate fiber optic vaults located on the Property. One is owned and operated by MCI and the other by Bell Atlantic. There will be a total of eight (8) four-inch conduits connecting the building to the fiber optic vaults (four conduits to each fiber optic vault). The conduits will extend to the main communications closet on the first floor of the

building. The communications closets on each of the five upper floors will have sleeves installed to provide easy access for fiber optic telecommunications to be extended to each floor for tenant purposes.

Interior

- Typical office tenant area shall have a minimum of 8'-10" foot finished ceiling heights
- Exterior core walls ready for paint.
- Window coverings coordinated throughout building.

Core Areas

- Restrooms are designed with ceiling hung toilet partitions; ceramic tile floor; 6'-0" high glazed ceramic tile wet wall; painted drywall walls with brushed stainless steel accessories and a corian vanity top.
- All core doors are solid core wood with wood stain grade finish and painted hollow metal frames.
- The elevator lobbies for future multi-tenant floors are planned to include painted drywall ceilings and walls with reveals and a carpeted floor with a granite stone base.
- The entry lobby finishes contain the distinctive exterior glass curtain wall, which frames the entrance to each of the two lobbies. A two-tone natural stone floor, matching stone door surrounds, architecturally detailed walls and carefully detailed metal finishes characterize this modern and highly finished space. The two main lobby areas will have 10'-0" ceiling heights.

HVAC System

- The building mechanical system is a highly efficient, self-contained A/C unit system supported by rooftop cooling towers and central controls.
- There is one highly efficient, compressorized, self-contained air conditioning unit per floor.
- The base building will include VAV boxes for the main lobby and toilet rooms on levels one and two. Lobbies and toilet rooms on levels 3 through 6 will be handled by tenant VAV boxes. Base building includes VAV boxes with fan powered VAV boxes at perimeter and shut-off boxes on interior of tenant spaces. Accordingly, sixteen (16) VAV boxes per floor are included within the base building.
- The cooling design load is planned to provide up to 46 tons of cooling per floor, or approximately 325 gross square feet of floor space per ton.
- The HVAC design includes 17,500 CFM per floor, providing approximately 1.15 CFM per square foot
- The HVAC design provides each floor with as much as 2000 CFM outside air and the possibility of an additional 200-CFM for future use.
- Each floor has set of 1-1/2" valved and capped condenser water lines for 10 hour of supplemental water-cooled A/C equipment.
- The energy management system planned for the building will include a state of the art DDC energy monitoring control system with night setback features.

Electrical

- Size of switchgear: 1 2,000A, 3 phase, 4W switchgear.
- Type and size of risers: 1 at 300A, 480/277V, 3 phase, 4w feeder, serving the mechanical and lighting panel boards and the 75 KVA, k-13 rated transformers serving the receptacle load panel boards in each typical tenant floor electrical closet.
- The design of the building allows the tenant (8 watts/sf)
 - 6 watts/sf for low voltage
 - 2 watts/sf for high-voltage (lighting)
- Number of electrical and communication closets per floor: one (1) electrical closets and one (1) communications closet per typical tenant floor.
- Size, type and rating of each transformer: 75 KVA dry-type transformer, 480V, 3 phase primary, 208/120V, 3 phase, 5-w, secondary, K-13, located in each electrical closet per typical tenant floor. In addition, there is one (1) 15 KVA, 480V, 3 phase, 208/120V, 3 phase, 4-w transformer in main electric and elevator machine rooms serving the 120V emergency system panel boards.
- · Size, type and rating of low voltage panel boards: 225 A, 208/120V, 3-phase, 5-W, 84-circuit panel boards equipped with 200% neutral and isolated ground for harmonic current

mitigation.

- Two (2) sets of four (4) conduits (4" each) extending from on-site fiber optic vaults into first floor communications room offers infrastructure for tenant required fiber and copper telecommunications cables; vertical access through building accomplished via sleeves in typical communications closet.
- The base building has a 80KW, 48Q/277V, 3 phase, 4-W diesel powered generator. This generator through automatic transfer switches carries the 20 HP fire pump, the jockey pump, the emergency lighting, the fire alarm system, and one elevator (in sequence) in the elevator bank.

Plumbing

There are three wet stacks with-in the tenant area per floor.

Fire / life Safety

There is a fully automated sprinkler system in the building with code required coverage for unbuilt tenant areas.

Parking

There are 206 parking spaces in the 2 level structured parking garage located next to the office building and 83 parking spaces on grade (total 289 parking spaces). There are also 4 loading spaces that are approximately 20' wide. In addition, a Metro park-and-ride is located directly across the street.

Building Specifications Summary

• Site Area: 151,758 Square feet (3.48 acres)

Building Size: Approximately 89,221.02 Rentable Square Feet

Number of Floors:
 6 - Office Floors

• Floor Rentable Area (subject to change):

 1st Floor
 12,251.70 NRSF

 2nd Floor
 14,111.50 NRSF

 3rd Floor
 15,714.45 NRSF

 4th Floor
 15,714.45 NRSF

 5th Floor
 15,714.45 NRSF

 6th Floor
 15,714.45 NRSF

• Finished Ceiling Height: 8'-10"

Elevators: 2 Passenger Traction Elevators 3,000 lbs./350 fpm

Communications:
 On-site fiber optic vaults provide substantial telecommunications capabilities

• Parking Provided: 293 Spaces Total (206 in Parking Garage, 83 on grade, 4 loading)

Metro park-and-ride located directly across the street

Loading Bays Provided:

• Tenant Occupancy: February/March 2001

EXHIBIT C Owner Approved Architects

Davis, Carter, Scott Architecture & Design Associates, Inc.

36

IMPORTANT NOTICE

THIS INSTRUMENT CONTAINS A CONFESSION OF JUDGMENT PROVISION WHICH CONSTITUTES A WAIVER OF IMPORTANT RIGHTS YOU MAY HAVE AS A DEBTOR AND ALLOWS THE CREDITOR TO OBTAIN A JUDGMENT AGAINST YOU WITHOUT ANY FURTHER NOTICE.

COMSCORE NETWORKS, INC., a Delaware corporation (the "Borrower" or "Obligor"), for value received, hereby promises to pay to the order of COMSTOCK PARTNERS, L.C., a Virginia limited liability company, (together with any subsequent holder of this Note, the "Lender") at 11465 Sunset Hills Road, Suite 510, Reston, Virginia 20190, or at such other address as the Lender shall specify in writing to the Borrower, the principal sum of SEVEN HUNDRED FIFTY THOUSAND DOLLARS (\$750,000.00), or so much thereof as remains unpaid.

This Note is that certain Security Deposit Promissory Note referenced in paragraph 19 of one certain lease agreement dated June 20, 2003 by and between Comstock Partners, L.C. ("Landlord") and Comscore Networks, Inc. ("Tenant") covering a portion of the office building ("Leased Premises" or "Building") located at; 11465 Sunset Hills Road, Reston, Virginia, ("Lease").

If not sooner paid, the entire principal of this Note and all accrued and unpaid interest shall be due and payable not later then five (5) days after any Default (as defined in the Lease) by Tenant that is not cured within the applicable Cure Period (as defined in the Lease) after Notice from Landlord (when required pursuant to the Lease).

On the fourth anniversary of the Effective Date (as defined in the Lease) of the Lease, the principal amount of this Note shall be reduced from Seven Hundred and Fifty thousand Dollars (\$750,000.00) to an amount equal to the Basic Rent for the last Lease Year (as defined in the Lease).

Payments or prepayments on this Note shall be applied to pay or reimburse the Lender for any costs and expenses incurred by or on behalf of the Lender under this Note, then to accrued interest, and the remainder to reduce the principal balance hereof.

The Borrower may prepay this Note in whole or in part at any time without penalty or premium.

The failure to pay the principal or any other sum described herein when due shall constitute an Event of Default under this Note. Upon the occurrence of an Event of Default, the entire unpaid balance of this Note shall, at the option of the Lender, become immediately due and payable, without notice or demand and the Lender may, in addition to any other remedy the Lender may exercise, charge the Borrower interest on the outstanding principal balance at rate of 18% per annum from the date of such an Event of Default until the entire principal balance is paid in full.

The Borrower waives presentment, demand, protest and notice of dishonor, to the fullest extent permitted by law, waives all exemptions, whether homestead or otherwise, as to the obligations evidenced by this Note, waives any rights which it may have to require the Lender to first proceed against any other person, agrees that without notice to any Obligor and without affecting any Obligor's liability, the Lender, at any time or times, may grant extensions of the time for payment or other indulgences to any Obligor or permit the renewal of this Note, and may add or release any Obligor primarily or secondarily liable, and agrees that the Lender may apply all moneys made available to it from any Obligor either to this Note or to any other obligation to the Lender of any Obligor,

The Lender shall not be deemed to have waived any of the Lender's rights or remedies hereunder unless such waiver is express and in writing signed by the Lender; and no delay or omission by the Lender in exercising, or failure by the Lender on any one or more occasions to exercise, any of the Lender's rights hereunder, or at law or in equity, including, without limitation, the Lender's right, after any Event of Default, to declare the entire indebtedness evidenced hereby immediately due and payable, shall be construed as a novation of this Note or shall operate as a waiver or prevent the subsequent exercise of any or all of such rights. Acceptance by the Lender of any portion or all of any sum payable hereunder whether before, on or after the due date of such payment, shall not be a waiver of the Lender's right either to require prompt payment when due of all sums payable hereunder or to exercise any of the Lender's rights, powers and remedies hereunder. A waiver of any right in writing on one occasion shall not be construed as a waiver of the Lender's rights to insist thereafter upon strict compliance with the terms hereof and no exercise of any right by the Lender shall constitute or be deemed to constitute an election of remedies by the Lender precluding the subsequent exercise by the Lender of any or all of the rights, powers and remedies available to it hereunder, or at law or in equity.

This Note shall bind and inure to the benefit of the parties hereto and their respective successors and assigns. This Note shall be governed by, and shall be construed according to, the laws of the Commonwealth of Virginia.

Should any sums become due and payable hereunder and such sums are not paid when and as due, time being of the essence, the Borrower hereby constitutes and appoints Marc Bettius and/or Christopher D. Clemente, either of whom may act, as its attorney-in-fact to confess judgment on the Borrower for the full sum due hereunder, plus attorney's fees of 20% of the total amount then outstanding under this Note, and upon entry of the judgment, the Borrower waives the benefit of any and every statute, ordinance, or rule of court which may lawfully waived conferring upon the Borrower any right or privilege or exemption, stay of execution or supplemented proceedings, or other relief from the enforcement or immediate enforcement of a judgment or related proceedings on a judgment. The Borrower acknowledges that said sum is reasonable as evidenced by Borrowers signature hereto. The Borrower consents to venue in the Circuit Court of Fairfax County with respect to the institution of an action confessing judgment hereon. The authority and power to appear for and enter judgment against the Borrower shall not be exhausted by one or more exercises thereof, or by any imperfect exercise thereof, and shall not be extinguished by any judgment entered pursuant thereto, such authority and power may be exercised on one or more occasions from time to time in the same or different jurisdictions as often as the Lender or its assigns shall deem necessary or advisable until all sums due hereunder have been paid in full.

BORROWER HEREBY AGREES NOT TO ELECT A TRIAL BY JURY OF ANY ISSUE TRIABLE OF RIGHT BY JURY, AND WAIVES ANY RIGHT TO TRIAL BY JURY FULLY TO THE EXTENT THAT ANY SUCH RIGHT SHALL NOW OR HEREAFTER EXIST WITH REGARD TO THIS NOTE, OR ANY CLAIM, COUNTERCLAIM OR OTHER ACTION ARISING IN CONNECTION HEREWITH INCLUDING, BUT NOT LIMITED TO THOSE RELATING TO (A) ALLEGATIONS THAT A PARTNERSHIP EXISTS BETWEEN LENDER AND BORROWER; (B) USURY OR PENALTIES OR DAMAGES THEREFORE; (C) ALLEGATIONS OF UNCONSCIONABLE ACTS, DECEPTIVE TRADE PRACTICE, LACK OF GOOD FAITH OR FAIR DEALINGS, LACK OF COMMERCIAL REASONABLENESS, OR SPECIAL RELATIONSHIPS (SUCH AS FIDUCIARY, TRUST OR CONFIDENTIAL RELATIONSHIP); (D) ALLEGATIONS OF DOMINION, CONTROL, ALTER EGO, INSTRUMENTALITY FRAUD, REAL ESTATE FRAUD, MISREPRESENTATIONS,

DURESS, COERCION, UNDUE INFLUENCE, INTERFERENCE OR NEGLIGENCE; (E) ALLEGATIONS OF TORTUOUS INTERFERENCE WITH PRESENT OR PROSPECTIVE BUSINESS RELATIONSHIPS OR OF ANTITRUST; OR (F) SLANDER, LIBEL OR DAMAGE TO REPUTATION. THIS WAIVER OF RIGHT TO TRIAL BY JURY IS GIVEN KNOWINGLY AND VOLUNTARILY BY BORROWER, AND IS INTENDED TO ENCOMPASS INDIVIDUALLY EACH INSTANCE AND EACH ISSUE AS TO WHICH THE RIGHT TO A TRIAL BY JURY WOULD OTHERWISE ACCRUE. LENDER IS HEREBY AUTHORIZED TO FILE A COPY OF THIS PARAGRAPH IN ANY PROCEEDING AS CONCLUSIVE EVIDENCE OF THIS WAIVER BY BORROWER.

IN WITNESS WHEREOF, the Borrower has caused this Note to be executed under seal as of the date of the first above written.

COMSC	CORE NETWORKS, INC.
a	corporation
Ву:	
	Name:
	Title:
TATE OF	
COUNTY OF, to-wit:	
the undersigned, a Notary Public in and or the County and aforesaid, do hereby certify that, and hi ertification, has personally appeared before me in my County and State aforesaid and acknowledged the same.	is official capacity as of Comscore Networks, Inc., whose name is signed to the foregoing
GIVEN under my hand and seal this day of, 200	
NOT	ARY PUBLIC
Ay commission expires:	

4 of 4

TO FILE A COPY OF THIS PARAGRAPH IN ANY PROCEEDING AS CONCLUSIVE EVIDENCE OF THIS WAIVER BY BORROWER.

IN WITNESS WHEREOF, the Borrower has caused this Note to be executed under seal as of the date of the first above written.

COMSCORE NETWORKS, INC.

	aCorporation
	By:
	Name:
	Title:
TATE OF	
COUNTY OF, to-wit:	
the undersigned, a Notary Public in and or the County and aforesaid, do hereby certify thatertification, has personally appeared before me in my County and State aforesaid and acknowledged	, and his official capacity as of Comscore Networks, Inc., whose name is signed to the foregoing the same.
GIVEN under my hand and seal this day of, 200	
	NOTARY PUBLIC
My commission expires:	
	48

EXHIBIT H List of Landlord Affiliates

Comstock Homes, Inc. ,

49

FIRST AMENDMENT TO LEASE AGREEMENT

THIS FIRST AMENDMENT TO LEASE AGREEMENT (this "Amendment") made and entered into this 3rd day February, 2005, by and between COMSTOCK PARTNERS, L.C., a Virginia limited liability company, hereinafter referred to as "Landlord"; and COMSCORE NETWORKS, INC., a Delaware corporation, hereinafter referred to as "Tenant".

WHEREAS, Landlord and Tenant entered into a Lease Agreement dated June 23, 2003 (the "Lease") for the lease of certain commercial office space located in an office building (the "Building") located at 11465 Sunset Hills Road, Reston, Virginia, and more particularly described in the Lease (the "Premises"); and

WHEREAS, Landlord and Tenant desire to amend certain terms and conditions of the Lease.

NOW THEREFORE, for and in consideration of the mutual promises of the parties herein contained, the premises and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties hereto agree to amend the Lease as follows:

- 1. Notwithstanding the date on which this Amendment may be executed by either the Landlord or Tenant, all of the terms and conditions contained within this Amendment shall be effective as of February 1, 2005
- 2. The definition of the "Leased Premises", as continued in paragraph 1(b) of the Lease, shall be amended to include an additional seven thousand four hundred sixty six (7,466) square feet of office space located on the fourth (4th) floor of the Building as more particularly described on Schedule 1 attached hereto (the "Additional Premises"). As of the Effective Date, the total Leased Premises shall be thirty three thousand eight hundred twenty nine and two tenths (33,829.20) square feet of Rentable Area.
- 3. The definition of "Basic Rent", as contained in paragraph 1(b) of the Lease, shall be amended to provide that as of the Effective Date of this Amendment, the Basic Rent for the remainder of lease year two (February 1, 2005 through June 30, 2005) shall be Twenty Four and 20/100ths Dollars (\$24.20) per square foot or Sixty Eight Thousand Two Hundred Twenty Two and 22/100ths Dollars (\$68,222.22) per month. The definition of Basic Rent shall further be amended to provide that the Basic Rent shall increase each year by four percent (4%) over the immediately prior years' Basic Rent. Therefore, the Basic Rent for the third lease year shall be determined by multiplying the annualized amended Basic Rent for the remainder of the second lease year (February 1, 2005 through June 30,

2005) of Twenty Four and 20/100ths Dollars (\$24.20) per square foot by one hundred four percent (104%), and for each subsequent year by multiplying the Basic Rent for the immediately prior lease year's Basic Rent by one hundred four percent (104%). Accordingly, the Basic Rent during the Initial Term hereunder will be as follows (and the chart shown on page 1 of the Lease is hereby replaced with the following):

	Annual Rent	Monthly Rent	Rent/S.F.
From 7/1/03 until 6/30/04:	\$579,990.40	\$48,332.53	\$22.00
From 7/1/04 until 1/31/05:	\$597,390.11	\$49,782.51	\$22.66
From 2/1/05 until 6/30/05:	\$818,666.64	\$68,222.22	\$24.20
From 7/1/05 until 6/30/06:	\$851,413.31	\$70,951.11	\$25.17
From 7/1/06 until 6/30/07:	\$885,469.84	\$73,789.15	\$26.17
From 7/1/07 until 6/30/08:	\$920,888,63	\$76.740.72	\$27.22

Notwithstanding the foregoing, the Landlord agrees to reduce the Basic Rent due for the month of February 2005 in the amount of \$12,603.13. Accordingly, the total Basic Rent due for February 2005 is \$55,619.09.

- 4. The definition of "Rent Per Square Foot", as contained in paragraph 1(b) of the Lease, shall be amended to provide that the Basic Rent is Twenty Four and 20/100ths Dollars (\$24.20) per square foot commencing as of the Effective Date of this Amendment, and that Rent Per Square Foot of the Leased Premises shall be increased by four percent (4%) on the dates set forth in the chart contained in paragraph 3 above.
- 5. The definition of "Rentable Area", as contained in paragraph 1(b) of the Lease, shall be amended to provide that as of the Effective Date of this Amendment, the total Rentable Area of the Leased Premises is agreed to be thirty three thousand eight hundred twenty nine and two tenths (33,829.20) square feet.
- 6. The definition for "Tenant's Proportional Share", as contained in paragraph 1(b) of the Lease, shall be amended to provide that the Tenants Proportional Share as of the Effective Date of this Amendment is agreed to be thirty nine and ninety one hundredths percent (39.91%).
- 7. Landlord's option to terminate the Lease at any time after the third anniversary date of the Lease, as contained in the definition of Option to Terminate as set forth in paragraph 1(b) of the Lease, is deleted in its entirety.
- 8. First Right of Offer: Provided no Event of Default by Tenant occurs, and no circumstances then exist which, the lapse of time, the giving of notice or both, would constitute an Event of Default, Tenant shall have a First Right of Offer on any contiguous space on any floor

partially occupied by Tenant that may become available during the term of this Lease and any extensions thereof ("Expansion Space"). Such right excludes uses by Landlord and its affiliates. Landlord shall give written notice to Tenant of any Expansion Space that is anticipated to become available and the anticipated date of availability, Tenant shall have five (5) business days to provide written notice to Landlord that Tenant intends to accept the Expansion Space. If Tenant provides notice to Landlord that it desires to accept or lease the Expansion Space, then Landlord and Tenant shall promptly execute an amendment to the Lease to incorporate the Expansion space upon it becoming available. The terms for the Lease of the Expansion Space shall be the then current market rate (for comparable buildings in the general vicinity of the Building) but no less then the then current rental rate applicable under this Lease.

- 9. The Additional Premises shall be provided to the Tenant in its "as is, where is" condition, and the Landlord shall have no obligation to make any repairs or improvements to the Additional Premises and all such repairs and improvements shall be at the sole cost and expense of the Tenant. In addition, the Landlord agrees that the Tenant may use the modular furniture and certain other furniture as set forth on Schedule 2 attached hereto ("Landlord's Furniture"), currently located in the Additional Premises. The right to use the Landlord's Furniture is being provided to the Tenant as a courtesy by the Landlord and the Tenant accepts the Landlord's Furniture in its "as is, where is" condition. At the expiration of the term of the Lease, the Landlord's Furniture shall be returned to the Landlord in its condition as of the Effective Date of this Amendment, normal wear and tear accepted. Finally, the Landlord acknowledges and consents to the construction of a dividing wall in the Additional Premises so as to convert a conference room into two (2) smaller offices. All such construction activity shall be done at Tenant's sole cost and shall be performed subject to proper building permits by Signet Construction Company, Inc. No other modifications to the improvements contained within the Additional Premises shall be made without the written consent of the Landlord and, further, shall be subject to all of the terms and conditions of the Lease.
- 10. The Lease is otherwise ratified and reaffirmed in all respects and all terms and conditions thereof, unless otherwise modified by this Amendment, are in full force and effect. If there are any conflicts between the terms of the Lease and this Amendment, then this Amendment shall control.
 - 11. Unless otherwise stated herein, all terms defined in the Lease shall have the same meaning when used in this Amendment.

WITNESS the following signatures and seals:		
	LANDLORD:	
	COMSTOCK PARTNERS, L.C. a Virginia limited liability company	
/s/	By: /s/ Christopher Clemente	(SEAL)
WITNESS	Christopher D. Clemente Managing Member	
2/3/05		
DATE		
	TENANT:	
	COMSCORE NETWORKS, INC. a Delaware corporation	
/s/	By: /s/ Sheri L. Huston	(SEAL)
WITNESS	NAME: Sher. L. Huston TITLE:	
February 4, 2005		
DATE		

4 of 6

12. This Amendment may be executed in counterparts all of which when taken together shall constitute one amendment binding on all parties, signatories of the original or any counterpart. Each party shall become bound by this Amendment immediately upon fixing its signature thereto independently of the signature of the other party.

SCHEDULE 1

Attached hereto as Schedule 1 shall be a floorplan of the Additional Premises.

5 of 6

SEPARATION AGREEMENT

This Separation Agreement ("Agreement") is made between comScore Networks, Inc. ("Company"), a Delaware corporation, and Sheri Huston ("Employee").

In consideration of the mutual promises contained in this Agreement, and for other good and valuable consideration, the receipt and sufficiency of which are acknowledged, the undersigned, intending to be legally bound, state and agree as provided below.

- 1. **Separation**. Employee's last day of work with the Company and Employee's employment termination date will be February 28, 2006 (the "Separation Date"). Until the Separation Date, Employee agrees to provide reasonable transition assistance, including without limitation, commercially reasonable efforts to complete the projects listed under Paragraph 1 of the Transition Summary attached as Exhibit B, and handing off of the projects listed under Paragraphs 2-5. From time to time, after the Separation Date, Employee shall be available to respond to transition related questions, to the extent reasonable.
- 2. Accrued Salary and Paid Time Off. The Company will pay Employee for all accrued salary, and all accrued and unused vacation earned through the Separation Date, subject to standard payroll deductions and withholdings, on the Company's ordinary payroll dates. Employee is entitled to the payments described in this section even if Employee elects not to execute this Agreement.
- 3. **Severance Benefits**. The Company will pay severance to Employee in the form of a lump sum payment for an amount equivalent to six (6) months of the Employee's current base salary (the "Severance Payment"). The Severance Payment shall be made on the Separation Date. In addition, Employee is eligible; for and shall be paid a bonus payment of \$73,788 attributable to 2005 performance (the "Bonus") The Bonus shall be paid upon full execution of this Separation Agreement. Both the Severance Payment and the Bonus will be subject to standard payroll deductions and withholdings.
- 4. **Health Insurance**. Beginning the first month following the month of separation, to the extent provided by the federal COBRA law or, if applicable, state insurance laws, and by the Company's current group health insurance policies, Employee will be eligible to continue Employee's group health insurance benefits at Employee's own expense. On the Separation Date, Company will make a lump sum payment to Employee of \$9014.00, subject to standard payroll deductions and withholdings, intended to be equivalent to Company's portion of Employee's health insurance premiums for a six (6) month period.
- 5. **Other Compensation or Benefits.** If Employee elects to exercise Employee's vested stock options, Employee must exercise such vested stock options within ninety (90) days of the Separation Date in accordance with the terms and conditions of the comScore Networks, Inc. 1999 Stock Plan Stock Option Agreement. In addition, Company's Chief Executive Officer has agreed to designate Employee as a Participant pursuant to Section 2.1(iii) of the Company Incentive Plan created by Company's Board of Directors on August 1, 2003, attached hereto as Exhibit C.

Employee acknowledges that, except as expressly provided in this Agreement, Employee will not receive any additional compensation, severance or benefits from the Company after the Separation Date.

- 6. Expense Reimbursements. Employee agrees that, within ten (10) days of the Separation Date, Employee will submit Employee's final documented expense reimbursement reflecting all business expenses Employee incurred through the Separation Date, if any, for which Employee seeks reimbursement. The Company will reimburse Employee for these expenses pursuant to its regular business practice.
- 7. **Return of Company Property**. By the Separation Date, Employee agrees to return to the Company all Company documents (ad all copies thereof) and other Company property that Employee had in Employee's possession at any time, including, but not limited to, Company files, manuals, notes, drawings, records,: business plans and forecasts, financial information, specifications, computer-recorded information, tangible property (including, but not limited to, computers), credit cards, entry cards, identification badges and keys; and, any materials of any kind that contain or embody any proprietary or confidential information of the Company (and all reproductions thereof). Notwithstanding- anything to the contrary, Company shall remove all Company materials from Employee's laptop, and effective as of the Separation Date, Company hereby transfers ownership over Employee's laptop to Employee. In the event that Employee discovers any Company materials that failed to be removed from the laptop, Employee shall treat such information as Company confidential Information, promptly notify Company of such discovery and permit Company or an independent third party to take reasonable actions to remove or destroy such information. Company makes no warranties as to the operation of the laptop, and assumes no responsibility over its maintenance.
- 8. **Proprietary Information and Noncompetition Obligations**. Employee acknowledges Employee's continuing obligations under Employee's Employment, Invention Assignment and Non-disclosure Agreement, a copy of which is attached hereto as Exhibit A, including but not limited to, Employee's obligations related to confidentiality and noninterference with personnel relations.
- 9. Confidentiality. The provisions of this Agreement will be held in strictest confidence by Employee and the Company and will not be publicized or disclosed in any manner whatsoever; provided, however, that:
 (a) Employee may disclose this Agreement in confidence to Employee's immediate family; (b) the parties may disclose this Agreement in confidence to their respective attorneys, accountants, auditors, tax preparers, and financial advisors; (c) the Company may disclose this Agreement as necessary to fulfill standard or legally required corporate reporting or disclosure requirements; and (d) the parties may disclose this Agreement insofar as such disclosure may be necessary to enforce its terms or as otherwise required by law. In particular, and without limitation, Employee agrees not to disclose the terms of this Agreement to any current or former Company employee. Notwithstanding anything to the contrary, the parties shall mutually agree on the positioning if the communication regarding Employee's separation to Company personnel, and to any third parties, and such agreed-upon positioning may be disclosed by either party.

- 10. **Non-Disparagement**. Each party agrees to refrain from all conduct, verbal or otherwise, that disparages or damages or could disparage or damage the reputation, goodwill, or standing in the community of the other party, or damage or interfere with the business of the Company. For the purposes of this paragraph, "party" shall mean the Company's current officers and directors. This non-disparagement provision shall not in any way prevent the parties from disclosing any information to their attorneys or in response to a lawful subpoena or court order requiring disclosure of such information. Both parties agree that the separation was not a function of Employee's performance.
- 11. Release of All Claims/Indemnification. Company hereby releases, acquits and discharges Employee, and Employee hereby releases, acquits and discharges the Company and its affiliates, and their officers, directors,' agents, servants, employees, attorneys, shareholders, successors and assigns (collectively, the "Released Parties"), of and from any and all claims, liabilities, demands, causes of action, costs, expenses, attorneys' fees, damages, indemnities and obligations of every kind and nature, in law, equity or otherwise, known or unknown, suspected or unsuspected, disclosed and undisclosed, arising out of or in any way related to agreements, events, acts or conduct at any time prior to and including the execution date of this Agreement, including but not limited to: all such claims and demands directly or indirectly arising out of or in any way connected with Employee's employment with the Company or the termination of that employment; claims or demands related to salary, bonuses, commissions, stock, stock options, or any other ownership interests in the Company, vacation pay, fringe benefits, expense reimbursements, severance pay, or any other form of compensation; claims pursuant to federal, state or local law, statute or cause of action including, but not limited to, the federal Civil Rights Act of 1964, as amended; the federal Americans with Disabilities Act of 1990, as amended; the federal Age Discrimination in Employment Act of 1967, as amended ("ADEA"); the Virginia Human Rights Act; tort law; contract law; wrongful discharge; discrimination; harassment; fraud; defamation; emotional distress; and breach of the implied covenant of implied good faith and fair dealing. This release does not extend to the Company's right to pursue all available legal remedies against Employee for any intentional torts, gross negligence, illegal acts, or acts for which criminal penalties are available. Company shall indemnify Employee from and against any loss, damages, liabilities, judgments, settlements or costs and e
- 12. **Cooperation**. Employee agrees to reasonably cooperate with the Company in good faith in any internal investigation or administrative, regulatory, or judicial proceeding, including without limitation, making herself available to the Company upon reasonable notice for interviews and factual investigations; appearing at the Company's request to give testimony without requiring service of a subpoena or other legal process; volunteering to the Company pertinent information; and turning over to the Company all relevant documents which are or may come into my possession all at times and on schedules that are reasonably consistent with her other permitted activities and commitments. Employee understands that in the event the Company asks for her cooperation in 'accordance with this provision, the Company will reimburse her solely for (a) reasonable out-of-pocket expenses, including travel, lodging and meals, upon her submission of receipts. and (b) to the

extent that she is requested by the Company to cooperate in such an investigation or proceeding for more than 40 cumulative hours, a daily rate of \$950 per day, within 15 days of receipt of an invoice.

- 13. ADEA Waiver. Employee acknowledges that Employee is knowingly and voluntarily waiving and releasing any rights Employee may have under the ADEA. Employee also acknowledges that the consideration given for the waiver and the release in the preceding paragraph hereof is in addition to anything of value to which Employee was already entitled. Employee further acknowledges that Employee has been advised by this writing, as required by the ADEA, that: (a) Employee's waiver and release do not apply to any rights or claims that may arise after the execution date of this Agreement; (b) Employee has been advised hereby that Employee has the right to consult with an attorney prior to executing this Agreement; (c) Employee has twenty-one (21) days to consider this Agreement (although Employee may choose to voluntarily execute this Agreement earlier); (d) Employee has seven (7) days following execution of this Agreement by the parties to revoke the Agreement; and (e) this Agreement will become effective on the. date upon which the revocation period has expired, which will be the eighth day after this Agreement is executed by Employee. In addition, this Agreement specifically incorporates and includes by reference all other legally required federal and state notice and rescission periods applicable to Employee.
- 14. **Remedies**. Employee and the Company each agree that it would be impossible or inadequate to measure and calculate the other's damages from any breach of the covenants set forth in the "Confidentiality" Section above. Accordingly, Employee and the Company each agree that the non-breaching party will have available, in addition to any other right or remedy available, in law, in equity or otherwise, the right to obtain injunctive relief against the threatened breach of the "Confidentiality" Section or the continuation of such breach by the breaching party, without the necessity of proving damages.
- 15. **Enforcement**. Except as otherwise provided herein, if any action at law or in equity is necessary to enforce or interpret the terms of this Agreement, the prevailing party shall be entitled to reasonable attorneys' fees, costs and necessary disbursements in addition to any other relief to which such party may be entitled.
 - 16. Costs. The parties intend that each shall bear its own costs (including attorney's fees), if any, that may have been incurred relating to this Agreement.
 - 17. No Admission of Liability. This Agreement is not intended as an admission of liability by any party.
- 18. **Effective Date**. This Agreement will become effective on the latter of: (a) February 28, 2006; or (b) after seven days have passed since Employee signed the Agreement, assuming that Employee does not revoke the Agreement (the "Effective Date").
 - 19. Notice. In the event that any notice is to be given to any party under this Agreement, it shall be given by certified mail, return receipt requested, and addressed to the party as follows:

To Company: comScore Networks, Inc.

Attention: Corporate Counsel 11465 Sunset Hills Drive, Suite 200

Reston, VA 20190

To Employee: Sheri L. Huston

9541 Noory Court Vienna, VA 22182 773.25 1.7466

20. **Miscellaneous**. This Agreement, including Exhibits A, B and C, constitutes the full and entire understanding and agreement between the parties regarding the subjects hereof. It is entered into without reliance on any promise or representation, written or oral, other than those expressly contained herein, and it supersedes any other such promises, warranties or representations. This Agreement may not be modified or amended except in writing signed by both Employee and a duly authorized officer of the Company. This Agreement shall bind the heirs, personal representatives, successors and assigns of both Employee and the Company, and increase to the benefit of both Employee and the Company, their heirs, successors and assigns. If any provision of this Agreement is determined to be invalid or unenforceable, in whole or in part, this determination will not affect any other provision of this Agreement and the provision in question shall be modified by the court so as to be rendered enforceable. This Agreement shall be governed in all respects by the laws of the Commonwealth of Virginia as such laws are applied to agreements between Virginia residents entered into and performed entirely in Virginia.

In Witness Whereof, the undersigned have executed this Agreement as of the date written below.

COMPANY:

COMSCORE NETWORKS, INC.

By: /s/ Magid Abraham

Magid Abraham, Chief Executive Officer

EMPLOYEE:

/s/ Sheri Huston Sheri Huston 2.09.06 Date

Exhibit A — Employment, Invention Assignment and Non-disclosure Agreement Exhibit B — Transition Summary Exhibit C — comScore Networks, Inc. Incentive Plan created August 1, 2003

May 8, 2006

Mr. John M. Green 401 0 Fordham Road, NW Washington, DC 20016

Dear John:

On behalf of comScore Networks, Inc., I am pleased to offer you the position of **Chief Financial Officer** with an anticipated start date of **May 9, 2006**. We are confident that you will play a key role in comScore's continued success while helping us remain a leader in the provision of innovative and valuable marketing solutions.

In this position you will be reporting to Magid Abraham, Chief Executive Officer, and will work from comScore's offices located in Reston, VA.

Your base salary will be \$9615.40 bi-weekly (equivalent to an annual salary of \$250,000), payable in accordance with comScore3s standard payroll practice and subject to applicable payroll deductions and withholdings.

In addition, each calendar year, you will be eligible for a target performance bonus of up to 30% of your base salary for the applicable year. Such bonus will be based on performance goals established by your manager, and payable annually after the conclusion of the calendar year, in accordance with comScore's standard bonus payment practice and subject to applicable payroll deductions and withholdings. For the remaining period in 2006, your target bonus shall be prorated to account for the duration of your employment during 2006.

During your employment, you are eligible to participate in comScore's standard benefits such as medical insurance, life insurance, sick leave, 401k and paid time off consistent with any eligibility requirements and standard company policy. comScore's current vacation policy provides three (3) weeks paid vacation per year, earned on an accrual basis. The accompanying benefits brochure provides additional benefits information. Detailed benefits information will be provided in the comScore Employee Guide and in Summary Plan Descriptions, which will be available for your review on the first day of your employment.

Subject to approval of comScore's Board of Directors at the next Board meeting scheduled for option approval, you will be granted an option to acquire 650,000 shares of comScore's common stock at an exercise price equal to the fair market value of the stock on the date of the grant as

Mr. John M. Green Date: May 8, 2006 Page 2

determined by the Board, with certain option vesting acceleration rights, as further described below (the "Option Grant"). Unless such vesting is accelerated, the shares governed by the Option shall vest in equal monthly installments over four (4) years, beginning one month after your employment start date (e.g., 1148th of the Option shares shall vest two months after your employment start). In the event your employment with comScore ends before your options are fully vested, the vested portion may be exercised as provided in comScore's standard Stock Option Agreement. Any options granted by comScore will be governed by the comScore Networks, Inc. 1999 Stock Plan and related Agreement.

In the event of a change of control for comScore whereby you lose your position as Chief Financial Officer, and are not provided with an alternative senior finance or accounting position within the surviving entity, the vesting for the unvested portion of the Option Grant shall accelerate and become fully vested as of the effective date of your termination. In the event of a change of control for comScore whereby you lose you position as Chief Financial Officer, and are provided with an alternative, but diminished, senior finance or accounting position within the surviving entity, the vesting for 162,500 unvested options of the Option Grant or the remaining unvested portion of the Option Grant, whichever is less, shall accelerate and become fully vested as of the date that you assume this alternative position within the surviving entity.

In the event of a termination without cause by comScore, you will be provided with a severance payment of 6 pay periods (or \$57,692.40).

On your first day of employment, we will provide for your review a copy of a comScore Employee Guide via corn score's intranet. As a comScore employee, you will be expected to abide by corn score's rules and policies and acknowledge in writing that you have read the comScore Employee Guide. In addition, you are required to sign and comply with the attached Employment, Invention Assignment and Non-Disclosure Agreement which, among other things, prohibits unauthorized use or disclosure of comScore's proprietary information. Please keep in mind that your employment will be at-will, which simply means that either you or comScore may terminate the relationship at any time for any reason, with or without cause.

By signing below, you acknowledge that you have apprised comScore of any and all contractual obligations that you may have that would prevent you from working at comScore, or limit your activities with comScore, including but not limited to any non-competition obligations to past employers.

The terms described in this letter, if you accept this offer, will be the terms of your employment and supersede any other agreements or promises made to you by anyone from comScore regarding these terms. This letter can only be modified by a written agreement signed by you

Mr. John M. Green Date: May 8, 2006 Page 3

and an authorized official of comScore. This offer is contingent upon you submitting the legally required proof of your identity and authorization to work in the United States, and upon the completion and satisfactory review of a background check.

If you wish to accept this offer, please return a signed copy of this letter to me. You may fax your signed offer letter to me at 703-438-2091.

We are very excited about the possibility of you joining comScore. We hope that you will accept this offer and help us in making comScore all it can be, to the lasting credit and economic benefit of us all.

Best Regards,

/s/ Magid Abraham

Magid Abraham

Chief Executive Officer

ACKNOWLEDGEMENT:

In response to this offer of employment <u>please sign one only</u>.

/s/ John M. Green 5/9/06 \boldsymbol{I} accept the offer of employment.

__ **I do not** accept the offer of employment.

Exhibit 21.1- Subsidiaries of the Registrant

Name of Subsidiary

comScore Brand Awareness, L.L.C.
Creative Knowledge, Inc.
Gatesmith Enterprises, Inc.
Marketscore, Inc.
OpinionCounts, Inc.
TMRG, Inc.
VoiceFive Networks, Inc.
comScore Europe, Inc.
comScore Canada, Inc.

Delaware, U.S.A.
Ontario, Canada

Consent of Independent Registered Public Accounting Firm

We consent to the reference to our firm under the caption "Experts" and to the use of our reports dated March 29, 2007, in the Registration Statement (Form S-1 No. 333-00000) and related Prospectus of comScore, Inc. for the registration of 000,000 shares of its common stock.

/s/ Ernst & Young LLP

McLean, VA March 29, 2007